# POTRERO YARD MODERNIZATION PROJECT

Infrastructure Facility DBFOM Agreement

# INFRASTRUCTURE FACILITY

# DESIGN-BUILD-FINANCE-OPERATE-MAINTAIN AGREEMENT<sup>1</sup>

# BETWEEN

# THE CITY AND COUNTY OF SAN FRANCISCO

# ACTING BY AND THROUGH THE

# SAN FRANCISCO MUNICIPAL TRANSPORTATION AGENCY

AND

# [PRINCIPAL PROJECT COMPANY]<sup>2</sup>

DATED AS OF [\_\_\_\_\_, 2025]

Infrastructure Facility DBFOM Agreement

# TABLE OF CONTENTS

# Page

ARTICLE	1. DEFINITIONS; CONTRACT DOCUMENTS; ORDER OF PRECEDENCE; OTHER DOCUMENTS	3
1.1	Definitions and Interpretation	3
1.2	Contract Documents; Rules to Reconcile Conflicting Provisions	
1.3	Conflicts, Ambiguities or Inconsistency in Principal Project Company's	-
	Management Plans	7
1.4	Reference Documents	7
ARTICLE	2. NATURE OF AGREEMENT; TERM	9
2.1	Nature of Agreement; PDA Status Under This Agreement	9
2.2	Right of Entry; Condition of Site	9
2.3	Term1	0
2.4	Title1	
	2.4.1 General1	
	2.4.2 Possessory Interest Tax1	
	2.4.3 Passage of Title1	
	2.4.4 Intellectual Property1	
	2.4.5 Documents of Title1	
	2.4.6 Care, Custody and Control Responsibilities1	
2.5	Limitation on Principal Project Company's Rights1	2
ARTICLE	3. FINANCIAL CLOSE1	3
3.1	Requirements for Financial Close1	3
	3.1.1 Obligation to Achieve Financial Close1	3
	3.1.2 Date of Financial Close1	
3.2	Financial Close Requirements and Deliverables1	
	3.2.1 Delivery of Financing Agreements and Other Documents1	
	3.2.2 City Deliverables1	
	3.2.3 Base Interest Rate Fluctuation and Credit Spread Risk Mitigation1	
	3.2.4 Conditions Precedent to Financial Close1	6
	3.2.6 Mandatory Terms of Project Debt, Financing Agreements and Other Documents	8
3.3	Post-Financial Close Requirements and Deliverables1	
3.4	Potential Adverse Events and Mitigation	
3.5	Permitted Excuses from Achieving Financial Close	
3.6	No-Fault Termination	
3.7	Principal Project Company's Failure to Achieve Financial Close	

ARTICLE	I	PRINCIPAL PROJECT COMPANY FINANCING; LENDERS' RIGHTS; REFINANCING; PRIVATE CAPITAL INVESTMENTS; FINANCIAL MODEL	24
4.1	Principa	al Project Company Right and Responsibility to Finance	24
4.1	No City	or City Responsibility for Project Debt	24 24
4.3	•	s' Rights	
4.4		cing	
	4.4.1	Right of Refinancing	
	4.4.2		
	4.4.3	5	
4.5		cing Gain	
4.6		Requirements	
4.7		al Model and Financial Model Updates	
	4.7.1	Model Update Event	
	4.7.2	Updates to the Financial Model	
	4.7.3	Replacement of Financial Model	30
	4.7.4	Financial Model Audits	31
4.8	Escrow	of Financial Model and Cost and Pricing Data	31
ARTICLE	5.	SUBMITTALS; MANAGEMENT SYSTEMS AND OVERSIGHT	33
5.1	Submitt	al Review Terms and Procedures	33
5.1	5.1.1		
	5.1.2		
	5.1.3	Limitations on Principal Project Company's Right to Rely	
5.2		Management Plan	
5.3		Assurance, Quality Control, Generally	
5.4		ht, Inspection and Testing	
5.5		and Test Results	
5.6		JS	
5.7		, ng	
	5.7.1		
	5.7.2		
ARTICLE	6.	GENERAL PRINCIPAL PROJECT COMPANY OBLIGATIONS	40
6.1	Plannin	g and Engineering Activities	40
6.2		Site Conditions	
6.3	Regulat	ory Approvals	
	6.3.1	CEQA Approval and NEPA Approval	40
	6.3.2	Regulatory Approvals other than the CEQA Approval and the NEPA Approval	42
	6.3.3	Copies to City	
	6.3.4	Certain Risks Relating to Regulatory Approvals	
	6.3.5	Changes to Regulatory Approvals	
	6.3.6	City Assistance with Regulatory Approvals	
	6.3.7	Regulatory Approvals in City's Name	
	6.3.8	Major Approval Delay	44

6.4	Compliance with Laws 6.4.1 Applicable Laws	
	6.4.2 Environmental Compliance	47
6.5	Compliance with Regulatory Approvals	47
6.6	Communication and Public Outreach	47
6.7	Coordination, Cooperation and Access	47
6.8	Intentionally Deleted	48
6.9	Safety Compliance	48
6.10	Law Enforcement and Security	49
	6.10.1 Law Enforcement Services	49
	6.10.2 Security	49
6.11	Warranties	
	6.11.1 Warranties for SFMTA O&M Facilities	49
	6.11.2 Contractor Warranties and Guaranties	51
	6.11.3 Warranties for Utility Owners and Authorities Having Jurisdiction	51
6.12	Maintain Good Standing	
ARTICLI	E 7. DESIGN AND CONSTRUCTION	53
7.1	General Obligations of Principal Project Company Concerning D&C Work	53
7.2	Performance, Design and Construction Standards	
7.3	Design Implementation	
7.4	Schedule, Deadlines, Notices to Proceed and Commencement of Work	54
	7.4.1 Project Schedule	
	7.4.2 Float	
	7.4.3 Commencement of Non-Construction Work	55
	7.4.4 Commencement of Construction Work	55
	7.4.6 Work Before NTP 1	
7.5	Acquisition of Real Property	
	7.5.1 Additional Acquisitions	
	7.5.2 Temporary Interests in Property	
	7.5.3 Property Acquisitions and Scheduling Work	
7.6	Utilities, Utility Adjustments and Utility Delay	
-	7.6.1 Utilities to Serve Facility	
	7.6.2 Principal Project Company's General Responsibilities	
	7.6.3 Avoiding Utility Adjustments and Minimizing Costs	
	7.6.4 Allocation of Work Responsibility	
	7.6.5 Utility Adjustment Costs	
	7.6.6 Incidental Utility Work	
	7.6.7 Bonds and Insurance; Security for Utility Adjustment Costs	
	7.6.8 Utility Investigation by Principal Project Company	
	7.6.9 Claims for Inaccuracies in Utility Information	
	7.6.10 Changes in Design	
	7.6.11 Utility Betterments	
	7.6.12 Failure of Utility Owners to Cooperate	
	7.6.13 Utility Delays	
	7.6.14 Utility-Related Claims; Additional Restrictions on Change Orders	
	Relating to Utility Adjustments	
	7.6.15 FTA Requirements	
7.7	Hazardous Materials Management; Risk Allocation	
	<b>J</b> ,	

	7.7.1	Hazardous Materials Management	68
	7.7.2	Additional Hazardous Materials Obligations of Principal Project	
		Company	
	7.7.3	Hazardous Materials Generator Responsibilities	
7.8		tial Completion	
7.9		ceptance	
7.10		sibility for Loss or Damage	
7.11		orming Work	
	7.11.1	Obligation to Replace Nonconforming Work	
	7.11.2	Remedial Plan for Nonconforming Work	
7.40	7.11.3	City's Remedies	
7.12		and Construction of Joint Development Alternatives	
7.13		uring D&C Period Selection and Procurement of PPC-Furnished FF&E	
	7.13.1		
	7.13.2 7.13.3	Selection of City-Furnished Equipment and Existing FF&E Installation of PPC-Furnished FF&E	
	7.13.3	Installation of City-Furnished IT/Comms FF&E	
	7.13.4	Commissioning of FF&E for Substantial Completion	
7.14			
7.14	7.14.1	General	
	7.14.1	Move-In Subcommittee	
	7.14.3	Move-In Resource	
	7.14.0		
ARTICLE	8. II	NFRASTRUCTURE FACILITY MAINTENANCE	
8.1	General.		
	8.1.1	General Obligations	79
	8.1.2	Changes in Standards and Specifications	
	8.1.3	Hazardous Materials Management	
	8.1.4	Utility Accommodation	
8.2		I Project Company Inspection, Testing and Reporting	
8.3		sibility for Loss or Damage	
8.4		I Work; IFM Management Plan	
	8.4.1	Performance of Renewal Work	
	8.4.2	IFM Management Plan	
8.5		ck	
	8.5.1	Handback Condition	
	8.5.2	Independent Engineer	
	8.5.3	Handback Reserve Amount	
	8.5.4	Joint Technical Review and Handback Inspections and Report	
	8.5.5	Handback Assessment	
	8.5.6	City Right to Self-Perform and Recover Costs	
	8.5.7	Spare Parts	
8.6		ck Requirements Reserve Account	
	8.6.1	Establishment	
	8.6.2	Funding	
	8.6.3	Use of Handback Requirements Reserve Account	88
	8.6.4	Disbursement of Handback Requirements Reserve Account on	00
	005	Completion of Handback Renewal Work or Earlier Termination	
	8.6.5	Handback Requirements Letters of Credit	88

	8.6.6	Coordination with JDA Project Companies and Joint Development Alternatives	80
8.7		perations Committee	
8.8		During IFM Period	
0.0	8.8.1	Title and Responsibility for Risk	
	8.8.2	Maintenance and Replacement of FF&E During IFM Period	
8.9		Management	
0.9	8.9.1	Independent Energy Auditor	
	8.9.1	External Energy Auditor	
	8.9.2 8.9.3	Failure to meet LEED Requirements	
		·	
ARTICLE	9.	CONTRACTING AND LABOR PRACTICES	95
9.1	Disclos	ure of Contracts and Contractors	95
9.2		nsibility for Work, Contractors and Employees	
9.3		ntracts; Contractor Qualifications	
	9.3.1	Key Contract Approvals, Amendments and Termination; Use of and	
		Change in Key Contractors	96
	9.3.2	Key Contract Provisions	
9.4	Prompt	Payment to Contractors	
9.5		rsonnel	
0.0	9.5.1		
	9.5.2		
	9.5.3		
9.6		articipation	
9.7		cts with Affiliates	
9.8		Standards	
9.9		liring Requirements for Construction Work and Renewal Work	
9.10		burce Hiring Program	
9.11		Employment Training Program	
9.12		ions and Noncompliance Points Relating to Local Hire and SFMTA	100
5.12		g Program Requirements Development	106
9.13		Standards	
0.10	Lunoar		
ARTICLE		INSURANCE; PAYMENT AND PERFORMANCE SECURITY; INDEMNITY	108
10.1	Insuran	Ice	108
	10.1.1	Insurance Policies and Coverage	108
	10.1.2	General Insurance Requirements	108
	10.1.3	Insurance Premium Benchmarking	
	10.1.4	Insurance Unavailability	
	10.1.5	Review of Insurance Unavailability Risks	
	10.1.6	Lender Insurance Requirements	
	10.1.7	Prosecution of Claims	
	10.1.8	Application of Insurance Proceeds	
10.2		nance Security	
	10.2.1	Equity Letter of Credit	
	10.2.2	D&C Performance Security	
10.3		I Requirements for D&C Performance Security	

10.4	Letters of Credit 10.4.1 General Provisions	
10.5	Guarantees	
10.5	Indemnities	
10.0	10.6.1 General Indemnity	
	10.6.2 Design Defects	
	10.6.3 Limitations on Indemnification Obligations	
	10.6.4 Principal Project Company's Defense	
10.7	Indemnities by Contractors	
10.8	Notice of Claims by Third Parties	128
10.9	SFMTA O&M Facilities Warranty Bond	128
10.0		120
ARTICLE	11. PAYMENTS TO PRINCIPAL PROJECT COMPANY	130
11.1	Milestone Payment	
11.2	Availability Payments	
11.3	Pass-Through Costs	
11.4	Invoice, Other Amounts and Payments	
11.5	Disputed Amounts	
11.6	Withholding from Payments	
11.7	Interest on Late Payments and Overpayments	
11.8	Taxes	
11.9	Payment Not Evidence of Approval	
11.10	Other Adjustments; Full Compensation	
11.11	Appropriation; Certification of Funds	
11.12	Allowances	134
ARTICLE		
	DEVIATIONS	138
12.1	General	
12.2	City Changes	
ARTICLE	13. GENERAL PROVISIONS APPLYING TO DELAY EVENTS AN	D
	RELIEF EVENTS	139
13.1	Interface with Other Portions of the Facility	
13.2	Delay Event and Relief Event Process	139
13.3	Mitigation	140
13.4	Deductions for Relevant Events	140
13.5	Acts of a PPC-Related Entity	
13.6	Notification; Delay in Notification	
13.7	Multiple and Overlapping Claims	
13.8	Burden of Proof and Mitigation	
13.9	Sole Entitlement	
13.10	Compensation	
13.11	Waiver	

ARTICLE	14. COMPENSATION AND OTHER RELIEF FOR DELAY EVENTS AND RELIEF EVENTS	143
14.1	Relief During the D&C Period	143
	14.1.1 Claim for Delay Event	
	14.1.2 Extension of Deadlines for Delay Events	
	14.1.3 Concurrent Delays	
	14.1.4 Costs Payable for Compensable Delay Events	145
	14.1.5 Additional Limits Relating to Hazardous Materials Event During D&C	140
	Period	
	14.1.6 Costs Payable for Force Majeure Events During the D&C Term	146
	14.1.7 Costs Payable for Unavoidable Delay Events During the D&C Term	147
	14.1.8 Impact of Delay Event on Performance of D&C Work	147
	14.1.9 Relief for Adverse Weather Event	147
14.2	Relief During the IFM Period	148
	14.2.1 Overview	148
	14.2.2 Claim for a Relief Event	148
	14.2.3 Costs Payable for Compensable Relief Events	149
	14.2.4 Costs Payable for Force Majeure During IFM Period	150
	14.2.5 Additional Limits Relating to Hazardous Materials Event During IFM Period	150
	14.2.6 Impact of Relief Event on Performance of IFM Services	
14.3	Method of Payment of Compensation for Compensable Delay Events and	101
14.0	Compensable Relief Events	151
14.4	Open Book Basis	
14.5	Excavations; Public Contract Code 7104	
14.6	Loss or Damage Due to Force Majeure Event	
1110		102
ARTICLE	15. DEDUCTIONS AND NONCOMPLIANCE POINTS	153
15.1	Failure Events	153
15.2	Deductions	
15.3	Noncompliance and Failure Reporting, Notification and Cure Process	154
	15.3.1 Noncompliance Database	
	15.3.2 Notification Initiated by City	
	15.3.3 Performance Reports	
	15.3.4 Response and Rectification Times	155
	15.3.5 Notification of Response and Rectification	
15.4	Assessment of Noncompliance Points	
15.5	Performance Notice	
15.6	Increased Oversight	
15.7	Persistent PPC Default	
15.8	Cure Plan for Persistent PPC Default	159
15.9	Amendment of IFM Period Performance Measurements Table	
ARTICLE	16. DEFAULT; REMEDIES	162
	- ,	
16.1	Default by Principal Project Company; Cure Periods	162
	16.1.1 PPC Default	

	16.1.2	Default Notice and Cure Periods	165
	16.1.3	Warning Notices	
	16.1.4	Principal Project Company to Comply with Default Notice and	
	10.1.4	Provide Cure Plan	167
16.2	City Re	medies for PPC Default	
10.2	16.2.3	Immediate City Entry and Cure of Wrongful Use	
	16.2.4	Remedies for Failure to Meet Safety Standards or Perform Safety	
	10.2.1	Compliance	169
	16.2.5	City Step-in Rights	
	16.2.6	Damages; Offset	
	16.2.7	Performance Bond	
	16.2.8	Suspension of Work	
	16.2.9	Other Rights and Remedies	
	16.2.10	•	
16.3		by City; Cure Periods	
10.0	16.3.1	City Default	
	16.3.2	Cure Periods	
16.4		al Project Company Remedies for City Breach	
10.1	16.4.1	Termination	
	16.4.2	Interest on Late Payment	
	16.4.3	Limitations on Remedies	
	16.4.4	Remedies at Law and in Equity	
	16.4.5	Principal Project Company Right to Suspend	
16.5		licative Payment.	178
16.6		on on Consequential Damages	
ARTICLE	17.	TERMINATION	179
17.1		ation for Convenience; Condemnation	
17.2		ation for Force Majeure Termination Events or Insurance Unavailability	180
	17.2.1	Notice of Conditional Election to Terminate – Force Majeure	
		Termination Events	
	17.2.2	No Right to Termination Election	
	17.2.3	Principal Project Company Options Upon City Notice	
	17.2.4	City Options Upon Principal Project Company Notice	
	17.2.5	No Waiver	
	17.2.6	Termination for Insurance Unavailability	
	17.2.7	Concurrent Notices	183
	17.2.8	Termination Compensation for Force Majeure Termination Events	
		and Insurance Unavailability	
17.3		ation for PPC Default	
	17.3.1	PPC Defaults Triggering City Termination Rights	
	17.3.2	Compensation to Principal Project Company	
	17.3.3	Finality	
17.4		al Project Company Rights to Terminate	
	17.4.1	Termination for City Default	
	17.4.2	Termination for Suspension of Work	
<i>.</i> – –	17.4.3	Termination Due to Court Ruling	
17.5	Iermina	ation if Financial Close Fails to Occur	186
17.6		ation Procedures and Duties	

		17.6.1 17.6.2	Performance of Work Pending Early Termination Date	
		17.6.2	Relinquishment and Possession of Project	
		17.6.4	Continuance or Termination of Key Contracts Before Work	
		4705	Completion	
		17.6.5	Other Close-Out Activities	
47		17.6.6	Calculation of Compensation	
17 17			t of Termination Compensation	
	-		Termination	
17		•	after Termination; Final Release	
			e Termination Rightsto Information	
17	. ! !	Access	to mormation	193
ARTIC	CLE	18. F	PARTNERING; CONTRACT DISPUTE PROCEDURES	194
18	8.1	Introduc	tion	194
18	3.2	Partneri	ng	194
		18.2.1	Collaborative Partnering	
		18.2.2	Confidentiality	
18	3.3	Contrac	t Dispute Procedures	
		18.3.1	Disputes Generally	
		18.3.2	Notice of Contract Dispute	
		18.3.3	Contract Dispute	196
ARTIC	CLE	19. F	REPRESENTATIONS AND WARRANTIES	199
19	).1	Principa	I Project Company Representations and Warranties	199
19	).2		presentations and Warranties	
19	.3		Remedies for Mutual Breach of Warranty	
ARTI	CLE	20. A	ASSIGNMENT AND TRANSFER	204
20	).1	Restricti	ons on Equity Transfers and Change of Control	204
20			ds and Procedures for City Approval	
20			ons on Assignment, Subletting and Other Transfers of PPC's Interest	• .
			roject	206
20	).4		nent by City	
20	).5		of Organization or Name	
ARTI		24 [	RECORDS AND AUDITS; INTELLECTUAL PROPERTY	200
ANTN	OLE	ZI. I	CECORDS AND ADDITS, INTELLECTUAL PROPERTY	200
21 21			ance and Inspection of Records	
21			ecords Act and San Francisco Sunshine Ordinance	
21				
21	.4	21.4.1	ual Property Developed IP	
		21.4.1	Principal Project Company IP	
		21.4.2	Third Party IP	
		21.4.3	City IP and City Data	
		21.4.4	Delivery of IP Materials	
		- I. F.O		···· 🗲 I 🕂

_	21.5 21.6	21.4.6 Payments Inclusive Intellectual Property Escrows City's Use of IP Materials	214
ART	TICLE	22. ADVERTISING AND OTHER BUSINESS OPPORTUNITIES	216
	22.1	Rights and Interests in the Project and Project Site	216
	22.2	Advertising and Business Opportunities	
2	22.3	Remedies	
AR	TICLE	23. MISCELLANEOUS	218
	23.1	Standard for Approvals	218
	23.2	Amendments	
_	23.3	Waiver	
	23.4	Independent Contractor; No Joint Venture or Partnership	
-	23.5	Successors and Assigns	
	23.6	Designation of Representatives; Cooperation with Representatives	
	23.7	Survival	
	23.8	Limitation on Third Party Beneficiaries	
2	23.9	Governing Law; Venue	
2	23.10	Notices and Communications	
2	23.11	Severability	221
2	23.12	Construction and Interpretation of Agreement	
2	23.13	Further Assurances	
2	23.14	Entire Agreement	223
2	23.15	Counterparts	223

## LIST OF EXHIBITS

Exhibit 1 Abbreviations and Defi	initions
----------------------------------	----------

- Exhibit 2 Project Management Plan
- Exhibit 3 Implementation Proposal
  - 3A Financial Proposal
  - 3B Technical Proposal Elements
- Exhibit 4 Payment Mechanism
  - 4A Milestone Payment Mechanism
  - 4B Availability Payment Mechanism
- Exhibit 5 Finance Documents
  - 5A List of Initial Financing Documents

- 5B Form of Direct Agreement
- 5C Calculation of Refinancing Gain
- 5D Form of Opinion from City's Legal Counsel
- 5E Form of Opinion from Principal Project Company's Legal Counsel
- 5F Base Capital MaxAP Adjustment for Base Interest Rate Fluctuation and Credit Spread Risk Mitigation
- 5G Financing Document Terms
- Exhibit 6 Forms of Payment, Performance and Warranty Security
  - 6A Financial Close Security
  - 6B Form of D&C Payment Bond
  - 6C Form of D&C Performance Bond
  - 6D Form of Multiple Obligee Rider for D&C Payment Bond
  - 6E Form of Multiple Obligee Rider for D&C Performance Bond
  - 6F Form of SFMTA O&M Facilities Warranty Bond
- Exhibit 7 Insurance Requirements
- Exhibit 8 Form of Interface Agreement [TO FOLLOW FROM PNC ONCE FINALIZED]
- Exhibit 9 Change Procedures
- Exhibit 10 Initial Designation of Authorized Representatives
- Exhibit 11 Submittals Review Process
- Exhibit 12 Energy Management
- Exhibit 13 Costs Schedule
- Exhibit 14 Key Contract Provisions
- Exhibit 15 Conditions Precedent
  - 15A Conditions to NTP 1 Commencement of Non-Construction Work
  - 15B Conditions to NTP 2 Commencement of Construction Work
  - 15C Conditions to Substantial Completion

- 15D Conditions to Final Acceptance
- Exhibit 16 Federal, State and City Requirements
  - 16A Federal Requirements
  - 16B State Requirements
  - 16C City Requirements
  - 16D SFMTA's Surveillance Technology Policy
- Exhibit 17 Section 3.6 Invoice
- Exhibit 18 Technical Requirements
- Exhibit 19 Initial Schedule
- Exhibit 20 List of Key Personnel

#### INFRASTRUCTURE FACILITY DESIGN-BUILD-FINANCE-OPERATE-MAINTAIN AGREEMENT

## POTRERO YARD MODERNIZATION PROJECT

This Design-Build-Finance-Operate-Maintain Agreement (this "**Agreement**") is made and entered into this [\_\_] day of [\_\_\_\_\_], 2025 (the "**Effective Date**"), by and between the City and County of San Francisco ("**City**"), a municipal corporation acting in its proprietary capacity by and through the San Francisco Municipal Transportation Agency ("**SFMTA**"), and [\_\_\_\_\_] ("**Principal Project Company**" or "**PPC**"), a [\_\_\_\_\_], with reference to the following facts:

A. City, under the jurisdiction of the SFMTA, owns the real property commonly known as 2500 Mariposa Street in San Francisco, California, which is a 4.4-acre site comprised of Assessor's Block No. 3971-001, bounded by Bryant Streets, 17<sup>th</sup> Street, Hampshire Street, and Mariposa Street ("**Project Site**").

B. The Facility to be constructed at the Project Site is comprised of (i) a transit operations component ("**Bus Yard Component**"), (ii) a Joint Development Alternative; and (iii), and the common infrastructure shared by the two components ("**Common Infrastructure**").

C. This Agreement addresses the design, construction, financing, operations and maintenance of the Bus Yard Component and Common Infrastructure (collectively, the "Infrastructure Facility"), and the integration of the Infrastructure Facility, and its interface with, a future Joint Development Alternative (together, the "**Project**"). If implemented, (i) one or more Housing and Commercial Components will be delivered by a JDA Project Company, under one or more separate agreements; and (ii) the Paratransit Facility will be delivered by Principal Project Company pursuant to a Change Order or by Principal Project Company or another entity under a separate agreement. The Infrastructure Facility and the Joint Development Alternative are, collectively, the "**facility**".

D. On March 16, 2021, the Board of Supervisors of the City and County of San Francisco ("**Board of Supervisors**") adopted Ordinance No. 38-21 ("**Ordinance**"), which codifies SFMTA's authority to procure the Facility under a joint development approach.

E. On April 9, 2021, City issued a request for proposals to design, build, finance, and maintain the Infrastructure Facility and design, build, finance, operate and maintain the Housing and Commercial Component at the Project Site ("**RFP**"), and on September 12, 2022, City determined that the proposal submitted by Potrero Neighborhood Collective LLC ("**PNC**") offered the best value to City for the development of the Facility.

F. On November 2, 2022, City and PNC, a limited liability company organized under the laws of the State of Delaware, entered into a Predevelopment Agreement for the Potrero Yard Modernization Project (as amended, the "**Predevelopment Agreement**").

G. City and PNC agreed upon the terms provided in this Agreement for the development of the Infrastructure Facility pursuant to the processes described in the Predevelopment Agreement. On [\_\_\_\_\_], 2025, PNC formed the Principal Project

Company as a special purpose entity to enter into this Agreement. Principal Project Company is *[insert ownership structure – wholly owned by Equity Members of PNC?]*.

H. The SFMTA Board of Directors and the Board of Supervisors of the City have authorized the Work for the Project, as specified in the Contract Documents.

NOW, THEREFORE, for good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the Parties agree as follows:

#### ARTICLE 1. DEFINITIONS; CONTRACT DOCUMENTS; ORDER OF PRECEDENCE; OTHER DOCUMENTS

#### **1.1** Definitions and Interpretation

**1.1.1** Definitions for certain acronyms, abbreviations and terms used in this Agreement and the other Contract Documents are contained in <u>Exhibit 1</u> (Abbreviations and Definitions).

- **1.1.2** Unless the context otherwise requires, in this Agreement:
  - (a) The words "including", "includes" and "include" will be read as if followed by the words "without limitation";
  - (b) The meaning of "or" will be that of the inclusive "or", that is meaning one, some or all of a number of possibilities;
  - A reference to any Party or Person includes each of their legal representatives, trustees, executors, administrators, successors, and permitted substitutes and assigns, including any Person taking part by way of novation;
  - (d) References to days are references to calendar days, provided that if the date to perform any act or provide any Notice falls on a non-Business Day, such act or Notice may be timely performed on the next Business Day. Notwithstanding the foregoing, requirements contained in this Agreement relating to actions to be taken in the event of an emergency and other requirements for which it is clear that performance is intended to occur on a non-Business Day, shall be required to be performed as specified, even though the date in question may fall on a non-Business Day;
  - (e) A reference to any Governmental Entity, institute, association or body is:
    - (i) if that Governmental Entity, institute, association or body is reconstituted, renamed or replaced or if the powers or functions of that Governmental Entity, institute, association or body are transferred to another organization, a reference to the reconstituted, renamed or replaced organization or the organization to which the powers or functions are transferred, as applicable; and
    - (ii) if that Governmental Entity, institute, association or body ceases to exist, a reference to the organization which serves substantially the same purposes or objectives as that Governmental Entity, institute, association or body;
  - (f) A reference to this Agreement or to any other deed, agreement, document, or instrument includes a reference to this Agreement or such other deed, agreement, document or instrument as amended, revised, supplemented or otherwise modified from time to time;

- (g) A reference to any legislation or to any section or provision of it includes any amendment to or re-enactment of, or any statutory provision substituted for that legislation, section or provision;
- (h) Words in the singular include the plural (and vice versa) and words denoting any gender include all genders;
- (i) Headings are for convenience only and do not affect the interpretation of this Agreement;
- The captions of the articles, sections and subsections in the Contract Documents are for convenience only and are not to be treated or construed as part of this Agreement;
- (k) Where any word or phrase is given a defined meaning, any other part of speech or other grammatical form of that word or phrase has a corresponding meaning;
- (I) All monetary amounts and obligations in the Contract Documents are expressed and payable in U.S. dollars;
- (m) Each party must perform its obligations in accordance with this Agreement at its own cost, unless expressly provided otherwise;
- (n) If this Agreement requires calculation of an amount payable to a party there must be no double counting in calculating that amount;
- (o) A reference to time is a reference to Pacific Time Zone in the United States;
- (p) In the event of an ambiguity in or dispute regarding the interpretation of this Agreement, this Agreement shall not be interpreted or construed against the Person who prepared this Agreement, and, instead, other rules of interpretation and construction shall be used;
- (q) The Parties acknowledge and agree that: (a) the Contract Documents are the product of an extensive and thorough, arm's-length exchange of ideas, questions, answers, information and drafts during the PDA Term; (b) each Party has been given the opportunity to independently review the Contract Documents with legal counsel; and (c) each Party has the requisite experience and sophistication to negotiate, understand, interpret and agree to the particular language of the provisions of the Contract Documents. Accordingly, in the event of a conflict, ambiguity or inconsistency in or Contract Dispute regarding the interpretation of the Contract Documents, the Contract Documents shall not be interpreted or construed against the Party preparing it, and instead the Contract Dispute resolver shall consult other customary rules of interpretation and construction;
- (r) Unless otherwise expressly stated in the Contract Documents, words that have well-known technical or construction industry meanings are used in the Contract Documents in accordance with such recognized meaning;

(s) A reference to a Section or Article is a reference to the section or article in the document or Exhibit where that Section or Article appears.

#### 1.2 Contract Documents; Rules to Reconcile Conflicting Provisions

**1.2.1** The term "Contract Documents" shall mean this Agreement and all its exhibits, including the Technical Requirements, as listed in <u>Section 1.2.2</u>. Each of the Contract Documents is an essential part of the agreement between the Parties. The Contract Documents are intended to be complementary and to be read together as a complete agreement.

**1.2.2** Unless the context otherwise requires and except as provided otherwise in <u>Sections</u> <u>1.2.3</u> and <u>1.2.4</u>, in the event of any conflict, ambiguity or inconsistency between any terms or provisions within the Contract Documents, the order of precedence, from highest to lowest, shall be as follows:

- (a) For Design Work and other non-Construction Work:
  - (i) Change Orders and Unilateral Change Orders directing Principal Project Company to implement a City Change in accordance with this Agreement and amendments to this Agreement;
  - (ii) <u>Article 1</u> (Definitions; Contract Documents; Order of Precedence; Other Documents) through <u>Article 23</u> (Miscellaneous) and <u>Exhibit 1</u> (Abbreviations and Definitions);
  - (iii) the Exhibits to this Agreement (excluding <u>Exhibit 1</u> (Abbreviations and Definitions and <u>Exhibit 18</u> (Technical Requirements));
  - (iv) <u>Exhibit 18</u> (Technical Requirements);
  - (v) Principal Project Company's Implementation Proposal identified in <u>Exhibit 3</u> (Implementation Proposal); provided that if City determines, in its sole discretion, that the Implementation Proposal contains a provision that is more beneficial to City than is specified elsewhere in the Contract Documents, then that provision shall take precedence; and
  - (vi) Project Management Plan.
- (b) Without limiting <u>Section 1.2.2(a)</u>, for Construction Work, the same order of precedence shall apply as for non-Construction Work in <u>subsection (a)</u>, except that the Final Design Documents shall also be considered part of this Agreement and shall be included as <u>Section 1.2.2(a)(vii)</u> in the order of precedence subject to the following:
  - (i) Specifications have precedence over plans; and
  - (ii) Any other Deviations contained in the Final Design Documents take priority over conflicting requirements of other parts of this Agreement

but only to the extent that Principal Project Company specifically identifies the conflicts to City and City approves such Deviations by Notice to Principal Project Company.

**1.2.3** In the event of any conflict, ambiguity or inconsistency within the Contract Documents, the following rules of interpretation shall apply:

**1.2.3.1** If the Contract Documents contain differing provisions on the same subject matter and within the same order of precedence pursuant to <u>Section 1.2.2</u>, the provisions that provide greater detail or establish a higher quality, manner or method of performing the Work or use more stringent standards shall prevail.

**1.2.3.2** Additional details in a lower priority Contract Document shall be given effect except to the extent they irreconcilably conflict with requirements, provisions and practices contained in the higher priority Contract Document.

**1.2.3.3** If the Contract Documents contain differing provisions on the same subject matter that cannot be reconciled by applying the rules in <u>Section 1.2.3.1</u> or <u>1.2.3.2</u>, then the provisions contained in the document of higher order of precedence shall prevail over the provisions contained in the document of lower order of precedence, unless City, in its good faith discretion, approves or directs otherwise in writing.

**1.2.3.4** In the event of an irreconcilable conflict between specific provisions of the Contract Documents and any standards, criteria, requirements, conditions, procedures, specifications or other provisions applicable to the Project established by reference to a manual or publication within a lower priority Contract Document, the specific provision of the higher priority Contract Document shall prevail over said standards or other provisions established by reference, to the extent of the irreconcilable conflict, unless City, in its good faith discretion, approves or directs otherwise in writing.

**1.2.3.5** In the event of a conflict among any standards, criteria, requirements, conditions, procedures, specifications or other provisions applicable to the Project, established by reference to a manual or publication within a Contract Document or set of Contract Documents with the same order of priority, the standard, criterion, requirement, condition, procedure, specification or other provision offering the higher quality, manner or method or performing will apply unless City, in its good faith discretion, approves or directs otherwise in writing.

**1.2.3.6** In the event of a conflict among the standards, criteria, requirements, conditions, procedures, specifications or other provisions of the Technical Requirements and those established by reference to a manual or publication, the Technical Requirements shall prevail.

**1.2.3.7** In all other respects, in the event of a conflict, ambiguity or inconsistency within the Contract Documents, general rules concerning construction of contracts in the State shall be applicable.

**1.2.4** This <u>Section 1.2</u> (Contract Documents; Rules to Reconcile Conflicting Provisions) shall not apply to provisions in the Technical Requirements that are erroneous, create a potentially Unsafe Condition, or may be inconsistent with Good Industry Practice or applicable Law. Instead, such provisions shall be reconciled under <u>Section 7.2.3</u>.

**1.2.5** Principal Project Company acknowledges and agrees that it had the opportunity and obligation to review the terms and conditions of this Agreement and to bring to the attention of City any conflicts, ambiguities or inconsistencies of which it is aware contained within this Agreement.

**1.2.6** Incorporation into this Agreement of any part of the Implementation Proposal shall not (a) limit, modify, or alter City's right to review and approve any Submittal included in the Implementation Proposal, or submitted to City after the Implementation Proposal (including any Project Schedule), or (b) be deemed as acceptance or approval of any part of the Implementation Proposal by City that conflicts with this Agreement or the Technical Requirements.

**1.2.7** If Principal Project Company becomes aware of any error or any conflict, ambiguity or inconsistency between or among the documents forming this Agreement, Principal Project Company shall promptly provide Notice to City, including the item Principal Project Company considers should apply based on the applicable rules in <u>Section 1.1.2(p)</u>. Except as expressly stated in this Agreement, if (a) the conflict, ambiguity, inconsistency or error cannot be reconciled by applying the applicable rules or (b) the Parties disagree about (i) which rule applies and/or (ii) the results of the application of such applicable rule(s), then City will determine, in its reasonable discretion, which of the conflicting items is to apply and provide Notice to Principal Project Company before Principal Project Company proceeds with the applicable aspect of the Work.

## 1.3 Conflicts, Ambiguities or Inconsistency in Principal Project Company's Management Plans

In the event of any conflict between or among the Principal Project Company's Project Management Plan and any of the Contract Documents, or if any provisions in Principal Project Company's Project Management Plan are in conflict, ambiguous or inconsistent, Principal Project Company shall modify the Project Management Plan, as applicable, to be consistent with the Contract Documents or cure the ambiguity or inconsistency in a manner satisfactory to City.

# 1.4 Reference Documents

**1.4.1** City has provided the Reference Documents to Principal Project Company for the purposes of disclosure and, in the case of general industry and general governmental manuals and publications, for guidance regarding Good Industry Practice. Reference Documents are for information only, and are not mandatory or binding on Principal Project Company, except to the extent that the Contract Documents incorporate specific provisions of the Reference Documents by reference.

**1.4.2** Principal Project Company acknowledges and agrees that City does not give any warranty, representation or undertaking in respect of the Reference Documents including that the Reference Documents:

(a) Are complete, accurate or fit for purpose;

- (b) Are in conformity with the requirements of the Contract Documents, Regulatory Approvals or Laws; or
- (c) Represent all of the information in City's possession or power relevant or material in connection with the Project; <u>provided</u>, <u>however</u>, that City represents and warrants that it has not affirmatively and intentionally provided a Reference Document that is knowingly (based solely on City's Actual Knowledge) and materially false with the intent to mislead Principal Project Company.

**1.4.3** Principal Project Company acknowledges and agrees that, except as provided in this Agreement:

- (a) It has, before Commercial Close, conducted its own analysis and review of the Reference Documents upon which it places reliance;
- (b) Any use or reliance on such Reference Documents by Principal Project Company shall be solely at its own risk;
- (c) Except to the extent that the Contract Documents incorporate specific provisions of the Reference Documents by reference, no PPC-Related Entity is entitled to make any Claim, cause of action or Loss against City in connection with the Reference Documents, including on the grounds:
  - (i) of any misunderstanding or misapprehension in respect of the Reference Documents;
  - (ii) of any conclusions any PPC-Related Entity may draw from or any action or forbearance in reliance on the Reference Documents;
  - (iii) of any failure to disclose or make available to any PPC-Related Entity any information, documents or data or to review or update the Reference Documents; or
  - (iv) that the Reference Documents were inaccurate, incomplete, or not fit for purpose; and
- (d) The Reference Documents may include interpretations, extrapolations, analyses, and recommendations about data, design solutions, technical issues and solutions, construction and installation means and methods, and maintenance means and methods. Such interpretations, extrapolations, analyses, and recommendations are (i) preliminary in nature and, in many cases, obsolete, (ii) not intended to express the views or preferences of City, or represent any statement of approval or acceptance thereof by City, and (iii) not intended to form the basis of Principal Project Company's design solutions, technical solutions, construction or maintenance means and methods.

# ARTICLE 2. NATURE OF AGREEMENT; TERM

## 2.1 Nature of Agreement; PDA Status Under This Agreement

**2.1.1** City hereby grants to Principal Project Company the exclusive right, and Principal Project Company accepts the obligation and agrees, during the Term, to:

- (a) design, build, and finance the Infrastructure Facility, and operate and maintain certain elements of it; perform Renewal Work, and perform and undertake handback of the Infrastructure Facility at the end of the IFM Period; and
- (b) perform all other activities relating to the Infrastructure Facility not specifically retained by City for the Infrastructure Facility as specified in this Agreement.

**2.1.2** Principal Project Company acknowledges and agrees that the Predevelopment Agreement does not apply to this Agreement or the Parties' obligations hereunder, and that any deliverables provided by Lead Developer to City pursuant to the Predevelopment Agreement, or approvals obtained from City pursuant to the Predevelopment Agreement, do not apply to this Agreement unless any such deliverable, or approved plan or submittal, is expressly incorporated into this Agreement. The foregoing shall not apply to any obligations of the Parties under the Predevelopment Agreement that survive the completion or termination thereof, as more particularly set forth in the Predevelopment Agreement.

## 2.2 Right of Entry; Condition of Site

**2.2.1** Subject to <u>Section 7.5</u> (Acquisition of Real Property), PPC-Related Entities shall have the right to enter onto the Project Site from and after the Financial Close Date (the "**Access Date**") and throughout the remainder of the Term to carry out Principal Project Company's obligations under this Agreement; <u>provided</u>, <u>however</u>, that construction Work may not be mobilized or undertaken until the later of (i) 90 days after the Access Date; or (ii) the date on which Principal Project Company has completed the City Relocation Scope. After the Termination Date, PPC-Related Entities may enter onto the Project Site to perform posttermination obligations either based on the right of entry granted in <u>Section 17.6.3</u> (Relinquishment and Possession of Project) or under a separate right of entry provided by City in writing. Without otherwise limiting the rights of the Principal Project Company to relief under this Agreement, Principal Project Company shall comply with, and shall ensure that PPC-Related Entities comply with, all agreements, easements, rights of entry, covenants, conditions, restrictions and other instruments applicable to the Project Site.

**2.2.2** Within 90 days after the Financial Close Date, City shall ensure that the Project Site is in the City Ready for Move Condition. The Principal Project Company is responsible for notifying all Utility Owners and City in advance to turn off the Utilities and to obtain all required approvals prior to demolition. Additionally, if any power, water, and or other Utilities need to be maintained during demolition, such lines should be temporarily relocated as necessary and or protected by Principal Project Company. Should any furniture or portable equipment remain at the Project Site after City vacates the Project Site and after Principal Project Company fully performs the City Relocation Scope, those items shall be removed and/or disposed of by Principal Project Company. Removal of any Hazardous Materials, Utility infrastructure, walls, fixtures, flooring, paint, tanks, and or any other building features that currently exist and, upon

vacation by the City of the Project Site, will not be removed by City, are deemed released to Principal Project Company and are the responsibility for Principal Project Company in connection with demolition.

#### 2.3 Term

This Agreement shall take effect on the Effective Date and shall remain in effect until 30 years after the Substantial Completion Deadline, subject to the right of the Parties to terminate this Agreement earlier in accordance with the terms of this Agreement.

#### 2.4 Title

#### 2.4.1 General

The Parties acknowledge and agree that:

- (a) Principal Project Company is not the legal or equitable owner or lessee of the Project Site or the Infrastructure Facility improvements for any purpose;
- (b) Principal Project Company's rights under this Agreement are derived solely from its status as an independent contractor under this Agreement, and not as tenant, lessee, easement holder, optionee, lienor, mortgagee, purchaser or owner of any other interest in real property; and
- (c) The payments to be received by Principal Project Company under this Agreement are for services to be performed by Principal Project Company, and are not payments in the nature of rent, fees with respect to real property, or purchase price of real property.

## 2.4.2 Possessory Interest Tax

**2.4.2.1** City hereby informs Principal Project Company, in accordance with California Revenue and Taxation Code section 107.6, that, notwithstanding <u>Section 2.4.1</u> (General), (i) this Agreement may create one or more possessory interests, (ii) any possessory interest created by this Agreement may be subject to property taxation, and (iii) Principal Project Company may be subject to payment of property taxes levied on such possessory interest. Accordingly, Principal Project Company shall comply with the reporting requirements of San Francisco Administrative Code sections 23.38 and 23.39, and report (on behalf of the City) to the County Assessor the information required by California Revenue and Taxation Code section 480.5, and timely provide any information that City may request to ensure compliance with such requirements.

**2.4.2.2** If it is determined that a possessory interest is created by this Agreement and the County Assessor seeks to levy an assessment or tax on such possessory interest, then:

 Principal Project Company shall (i) provide Notice to City not later than 3 Business Days after it becomes aware of the assessment or tax bill, and (ii) deliver copies of all documentation relating thereto no later than 3 Business Days following receipt;

- (b) Principal Project Company shall cooperate and consult regularly with City, and follow City's reasonable directions, concerning tax protest, litigation strategy, refund claims and appeals process in any such circumstance. At City's request or in absence of a direction from the City sent to Principal Project Company no later than 10 days prior to any filing deadline, Principal Project Company shall timely (i) contest the assessment, and (ii) file refund claims.;
- (C) If the Board of Supervisor has specifically authorized and approved, by resolution, the City's assumption of the payment, in whole, of such possessory interest taxes under this Agreement, the City will make a payment of the tax liability on behalf of the Principal Project Company (including any penalties for late payment or nonpayment and interest imposed, to the extent such penalties or interest resulted from Principal Project Company's failure to comply or timely comply with Section 2.4.2.2(a) or (b)). To the extent that penalties and interest have been imposed as a result of Principal Project Company's failure to comply or timely comply with Section 2.4.2.2(a) or (b), Principal Project Company will reimburse City for any such payment of penalties and interest, including Recoverable Costs. If Principal Project Company receives any refund of the tax (including a refund of any penalties and interest) pursuant to a claim for refund or other appeal, Principal Project Company shall pay to City an amount equal to such refund within 15 days of receipt (but only to the extent of payments previously made by City pursuant to this Section 2.4.2.2(c));
- (d) Principal Project Company shall submit to City an invoice (in a format acceptable to City) on a monthly basis in arrears for all actual and reasonable costs and expenses incurred by Principal Project Company in opposing the imposition of any such possessory interest tax, together with detailed, supporting documentation evidencing same. If Principal Project Company is awarded attorneys' fees in connection with any related appeal or litigation, such costs will not also be reimbursed by the City to Principal Project Company. Within 30 Business Days of receipt of such invoice and supporting documentation, subject to <u>Section 11.5</u> (Disputed Amounts), City will reimburse Principal Project Company for such invoiced amount; and
- (e) City reserves the right to direct Principal Project Company to assign to the City any appeals rights available under applicable Law in order for City to directly assume the property tax appeal or litigation. Any such assignment shall be effected through an agent's authorization or similar document acceptable to both Principal Project Company and the City.

Nothing in the foregoing constitutes a notification to Principal Project Company of whether a potential possessory interest could be created under any HCC Agreement, the Housing and Commercial Component, any Housing and Commercial Component share of the Common Infrastructure, or any interest that Principal Project Company may acquire under <u>Section 7.5.2</u> (Temporary Interests in Property).

# 2.4.3 Passage of Title

Title to all materials, equipment, tools and supplies furnished under the Contract Documents for incorporation into the Project or that are required for operation or maintenance of the Project shall pass to City, free and clear of all liens or other charges of any kind or nature, upon incorporation into the Project or, for items that will not be incorporated into the Project, upon delivery to the Project Site.

#### 2.4.4 Intellectual Property

Except for PPC Intellectual Property and Third Party IP, title to any Intellectual Property to the extent made, conceived, prepared or reduced to practice as part of the Work, incorporated into the Project, including any improvements, modifications, enhancements or derivative works to or of the City IP shall vest in City at the earliest of creation, conception, preparation or reduction to practice.

#### 2.4.5 Documents of Title

Principal Project Company shall furnish or execute all necessary documents of title within 90 days of receiving a written request from City.

## 2.4.6 Care, Custody and Control Responsibilities

Passage of title to City shall not affect Principal Project Company's care, custody and control responsibilities. Principal Project Company shall be responsible for care, custody and control of all components of the Project, including all materials, equipment, tools and supplies described in <u>Section 2.4.3</u> (Passage of Title), until the Termination Date, except Principal Project Company shall be responsible for care, custody and control of (i) all elements of the Project that will be owned by Utility Owners or Authorities Having Jurisdiction until acceptance of such elements by the relevant Third Party; and (ii) the SFMTA O&M Facilities until the Substantial Completion Date.

#### 2.5 Limitation on Principal Project Company's Rights

Notwithstanding anything to the contrary in the Contract Documents, Principal Project Company has no power or authority to make any commitments on City's behalf or to execute agreements in the name of or on behalf of City. Principal Project Company shall not enter into any agreement with any Governmental Entity, Utility Owner, property owner or other Third Party having regulatory jurisdiction over any aspect of the Project or Work or having any property interest affected by the Project or the Work that in any way purports to obligate City, or states or implies that City has an obligation to carry out any installation, design, construction, maintenance, repair, operation, control, supervision, regulation or other activity during or after the end of the Term, unless City otherwise approves.

# ARTICLE 3. FINANCIAL CLOSE

#### 3.1 Requirements for Financial Close

#### 3.1.1 Obligation to Achieve Financial Close

Subject to the Parties' respective rights to terminate this Agreement before Financial Close pursuant to <u>Sections 3.6</u> (No-Fault Termination) and <u>3.7</u> (Principal Project Company's Failure to Achieve Financial Close), Principal Project Company shall achieve Financial Close by no later than the Financial Close Deadline.

#### 3.1.2 Date of Financial Close

Subject to <u>Section 3.4.2(a)</u>, Principal Project Company shall determine the date of Financial Close (the "**Scheduled Financial Close Date**"), which shall be subject to a minimum 60 calendar days' prior Notice to City and no later than the Financial Close Deadline. After such determination, Principal Project Company shall have the right to modify the Scheduled Financial Close Date, provided that (i) Principal Project Company must provide City with prompt Notice of any such change; (ii) Principal Project Company shall not modify the Scheduled Financial Close Date within 14 days of such date without City's approval, in its reasonable discretion; (iii) the new date for the Scheduled Financial Close Date shall not be sooner than 10 days after delivery of Notice to City; and (iv) the modified Scheduled Financial Close Date shall be no later than the Financial Close Deadline.

#### 3.2 Financial Close Requirements and Deliverables

## 3.2.1 Delivery of Financing Agreements and Other Documents

Not later than 30 days before the Scheduled Financial Close Date (or Delaved 3.2.1.1 Financial Close Date, as applicable), Principal Project Company shall deliver to City, for City's review and comment, near final drafts of the Initial Financing Documents, Direct Agreement, Equity Members Funding Agreements, D&C Contract and IFM Contract. Such drafts shall be sufficiently advanced such that all material terms are incorporated into the applicable Principal Project Document or Financing Document. Principal Project Company shall also deliver to City, for City's review and comment, a list of other Key Contracts, and a draft of the model audit report related to the proposed Financial Model Update in accordance with Section 4.7.4 (Financial Model Audits). City will review the list of other Key Contracts and may request, and Principal Project Company shall promptly provide to City, substantially final drafts of the Key Contracts. In the event any material term included in the drafts of the Principal Project Documents or Financing Documents provided to City under this Section 3.2.1.1 is amended prior to Financial Close, Principal Project Company shall deliver to City, for City's review, comment and approval, a copy of such proposed amendment(s), such submittal to be no later than 10 days before the Scheduled Financial Close Date. For purposes of this Section 3.2.1.1, a material amendment shall include the following: (i) a change that differs materially from any commitments or provisions set forth in the Finance Plan; (ii) a change which would alter the risk allocations in this Agreement or any Key Contract; (iii) a change which could increase the liability of City for termination compensation in the event of a termination; (iv) a change that would modify the gearing of the financing; (v) a change that would result in an Affordability Event: (vi) a change that would cause a Delayed Financial Close Date; (vii) a change in any of

the Security Documents; (viii) a change in the funding sources of Common Infrastructure; and/or (ix) a change under any of the conditions precedent to Financial Close under the Financing Documents.

**3.2.1.2** City will provide its comments on any initial submission of these documents no later than 10 Business Days after their submission and on any amendments to such documents no later than 5 Business Days after their submission, with such comments, in each case, being limited to addressing (i) changes in these documents from those approved by City during the PDA Term; (ii) changes in this Agreement or the commercial or legal structure of the Project; (iii) changes arising out of changed circumstances, new events or occurrences or additional information after the date of approval by City during the PDA Term; (iv) clarifications, errors, ambiguities or conflicts in such documents; (v) the relationship and interface of the Infrastructure Facility and the Joint Development Alternative; (vi) the impacts and implications of any decision by Principal Project Company to bifurcate the Financial Close for the Infrastructure Facility and the Joint Development Alternative; (vii) changes in applicable Law or Standards and Specifications; (viii) any change to a material term, as set forth in <u>Section 3.2.1.1</u>); and (ix) inconsistencies between such documents and the terms of this Agreement (including the risk allocations).

## 3.2.2 City Deliverables

To support the Principal Project Company's achievement of Financial Close, City shall:

- (a) on or before 30 days before the Scheduled Financial Close Date and in connection with the issuance of any bonds:
  - (i) provide and authorize Principal Project Company to include, in the preliminary and final official statement for such bonds or any other capital markets issuance with respect to the Project, SFMTA's most recent audited financial statements together with economic information with respect to SFMTA, and provide such documents and information as required to comply with disclosure requirements under applicable Laws, provided that Principal Project Company has provided City with a list of required documents and information at least 15 Business Days before such documents and information are required;
  - (ii) agree to customary continuing disclosure as reasonably acceptable to City and the underwriter of any Project Debt; and
  - (iii) provide customary certificates as reasonably requested by Principal Project Company and as reasonably acceptable to City;
- (b) on or before three Business Days prior to the Scheduled Financial Close Date, deliver each of the following items to Principal Project Company subject to customary escrow terms and to City's approval to release from escrow:
  - an executed legal opinion of the City's Attorney's Office substantially in the form attached hereto as <u>Exhibit 5D</u> (Form of Opinion from City's Legal Counsel);

- (ii) an executed certificate updating, as of the date of Financial Close, City's representations and warranties set forth in <u>Section 19.2</u> (City Representations and Warranties); and
- (iii) a counterpart signature page for the Direct Agreement executed on behalf of City;
- (c) take any reasonably required authorizing actions in a reasonably timely manner; and
- (d) provide customary documents, certificates or undertakings (or, as applicable, a copy of the same certified by City as true, complete and accurate) that Principal Project Company may reasonably request from City as necessary to comply with (i) disclosure requirements under applicable Laws, or (ii) customary underwriter requirements, in each case in connection with issuance of bonds or any other capital markets issuance; provided, however, that none of the documents supplied in accordance with this <u>Section 3.2.2(d)</u> shall be construed as providing a legal opinion binding on City.

# 3.2.3 Base Interest Rate Fluctuation and Credit Spread Risk Mitigation

**3.2.3.1** If Principal Project Company has complied with the submittal requirements under [Section • (Selection of Base Interest Rate(s)) of the Finance Plan], City will adjust the Base Capital MaxAP to offset 100% of the impact of fluctuations (increases and decreases) in the Base Interest Rates used in the Finance Plan that have occurred during the Bank Debt Rate Protection Period and/or the Bond Rate Protection Period, as applicable.

If Principal Project Company has complied with the submittal requirements under 3.2.3.2 [Section • (Credit Spread Risk Mitigation) of the Finance Plan], City will adjust the Base Capital MaxAP to offset a portion of the impact of fluctuations (increases and decreases) in the credit spreads for any Bond Financing proposed in the Finance Plan that have occurred during the Bond Rate Protection Period. Subject to the limitations described in [Section • of the Finance Plan and set forth in Exhibit 5F (Base Capital MaxAP Adjustment for Base Interest Rate Fluctuation and Credit Spread Risk Mitigation)], the credit spread risk sharing between City and Principal Project Company will be implemented on an 50:50 basis, with City assuming 50% of the credit spread fluctuation risk (relative to the Baseline Credit Spreads) and Principal Project Company assuming 50% (the "Credit Spread Risk Mitigation"). City will not provide Credit Spread Risk Mitigation with respect to Private Placements (except offerings under Rule 144A and Regulation S of the Securities Act of 1933), Bank Debt or any other debt facilities for which committed credit spreads or margins are available. City shall not accept increases in credit spreads in respect of bonds which are part of Principal Project Company's financing resulting from the final credit rating of such bonds being lower than the indicative investment grade rating(s) of such bonds provided in the Finance Plan.

**3.2.3.3** Adjustments to the Base Capital MaxAP and Original Equity IRR related to Base Interest Rates fluctuations and Credit Spread Risk Mitigation, as applicable, shall be implemented in accordance with <u>Exhibit 5F</u> (Base Capital MaxAP Adjustment for Base Interest Rate Fluctuation and Credit Spread Risk Mitigation).

## 3.2.4 Conditions Precedent to Financial Close

**3.2.4.1** Financial Close will occur upon Principal Project Company's satisfaction of each of the following conditions:

- (a) Principal Project Company has delivered to City, in form and substance similar to the drafts submitted to City pursuant to <u>Section 3.2.1.1</u> and including any changes made pursuant to City's comments under <u>Section 3.2.1.2</u>, fully executed versions of:
  - (i) Initial Financing Documents;
  - (ii) Direct Agreement(s);
  - (iii) Equity Members Funding Agreements; and
  - (iv) Key Contracts, together with an updated list of Key Contracts;
- (b) Principal Project Company has delivered a certification signed by its chief financial officer or equivalent officer, certifying to the following:
  - Principal Project Company has satisfied (or upon Financial Close, will satisfy) all conditions precedent to the effectiveness of commitments of the Lenders under the Initial Financing Documents or such conditions have been waived if not satisfied, as applicable;
  - each of the documents delivered by Principal Project Company to City pursuant to <u>Section 3.2.4.1(a)</u> is a true, complete and accurate copy of the original;
  - (iii) all representations and warranties of Principal Project Company under the Contract Documents remain true as of the Financial Close Date, except for any representation or warranty made as of a specified date, in which case such representation or warranty shall be true as of the specified date and Principal Project Company shall provide Notice to City if any such representation is not true and correct as of the Financial Close Date; and
  - Principal Project Company has performed and complied with all covenants and obligations of Principal Project Company under the Contract Documents to have been performed or complied with as of the Financial Close Date;
- (c) Principal Project Company has delivered to City, on or before one Business Day prior to the date of Financial Close, an interim Financial Model Update consistent with <u>Section 4.7</u> (Financial Model and Financial Model Updates) that incorporates any proposed amendments to the Base Case Financial Model agreed to by the Parties on or before such date, including those agreed to pursuant to any preliminary calculations under <u>Section 3.2.3</u> (Base Interest

Rate Fluctuation and Credit Spread Risk Mitigation), together with the related Financial Modeling Data;

- (d) Principal Project Company has delivered to City (i) one or more legal opinion(s) of Principal Project Company's in-house or external counsel substantially in the form attached hereto as <u>Exhibit 5E</u> (Form of Opinion from Principal Project Company's Legal Counsel) and with customary qualifications and assumptions reasonably acceptable to City;
- (e) A Financial Model in accordance with <u>Section 4.7</u> (Financial Model and Financial Model Updates) which reflects, to City's reasonable satisfaction, evidence of a Construction Equity Ratio greater than or equal to 5%;
- (f) Certificate from an insurer that meets the requirements in <u>Section 10.1.2.1</u> (Insurers), certifying that all Insurance Policies required to be effected and maintained in accordance with this Agreement in connection with the Work to be performed during the D&C Period (whether City is required to be an insured or not) are in force and effect (or will be in full force and effect following receipt of the insurance premium payable from the proceeds of Financial Close), and such evidence as is necessary to demonstrate the compliance of each such policy with the requirements of this Agreement;
- (g) Evidence that all Deferred Equity Amounts are secured by an Equity Letter of Credit (which may be annually renewable) in accordance with the approved equity contribution agreement guaranteeing the provision of the committed amount by a date which is not later than the Final Acceptance Date, and that Equity Letter of Credit has been issued and is in full force and effect; and
- (h) Each Payment Bond and Performance Bond required under <u>Section 10.2</u> (Performance Security) has been obtained, meets the requirements of <u>Section</u> <u>10.2</u> (Performance Security) and is in full force and effect, and Principal Project Company has delivered to City originals or copies, as required, of each Payment Bond and Performance Bond.

## 3.2.5 Waiver of Conditions Precedent to Financial Close

**3.2.5.1** A condition precedent is only waived if the Party for whom the benefit of the condition precedent is provided gives Notice to the other Party of the waiver prior to the Financial Close Deadline.

**3.2.5.2** When all of the Principal Project Company conditions precedent to Financial Close have been satisfied by Principal Project Company or waived by City, City shall provide Notice to Principal Project Company confirming that all of the Principal Project Company conditions precedent have been satisfied or waived and the date upon which the last of the conditions precedent was satisfied or waived.

**3.2.5.3** When all of the City conditions precedent to Financial Close have been satisfied by City or waived by Principal Project Company, Principal Project Company shall provide Notice to

City confirming that all of the City conditions precedent have been satisfied or waived and the date upon which the last of the City conditions precedent was satisfied or waived.

**3.2.5.4** Upon satisfaction of all conditions precedent to Financial Close, City will provide Notice to Principal Project Company confirming the date upon which Financial Close was achieved.

## 3.2.6 Mandatory Terms of Project Debt, Financing Agreements and Other Documents

The Project Debt and Financing Documents (including the Initial Financing Documents) shall comply with the Financing Document Terms.

## 3.3 **Post-Financial Close Requirements and Deliverables**

**3.3.1** Principal Project Company shall deliver to City, on or before two Business Days following Financial Close, each of the items required under <u>Section 4.7.3</u> (Replacement of Financial Model) replacing the Base Case Financial Model with a Financial Model Update that reflects all changes agreed by the Parties as of the date of Financial Close, as well as:

- (a) a Financial Model Update that includes any final revisions to the interim Financial Model Update delivered under <u>Section 3.2.4.1(c)</u> to incorporate the Base Interest Rates and credit spreads applicable on the date of Financial Close under the Initial Financing Agreements and any other agreed upon revisions;
- (b) a model audit report related to the proposed Financial Model Update in accordance with <u>Section 4.7.4</u> (Financial Model Audits); and
- (c) a form of written amendment that (i) effects the replacement of the Financial Model in effect with the proposed Financial Model Update, and (ii) addresses all other related amendments to this Agreement required as a result of the amended Financial Model, including any amendments to the definitions of Base Capital MaxAP, Base IFM MaxAP, Equity IRR and Key Ratios, as applicable.

**3.3.2** Upon the satisfaction of each of the conditions precedent to Financial Close set forth in <u>Section 3.2.4</u> (Conditions Precedent to Financial Close), and the delivery and mutual approval of the Financial Model Update and documents required by <u>Section 3.3.1</u>, City and Principal Project Company shall enter into the amendment.

**3.3.3** At Financial Close, Principal Project Company shall pay to City the amount of [\$•] to reimburse City for PDA-Related Costs, including all amounts paid by City to Principal Project Company or Lead Developer pursuant to any Early Works Agreement. [Note: City to confirm prior to Commercial Close whether any such reimbursement will be required and, if so, the amount.]

**3.3.4** City shall return to Principal Project Company the Financial Close Security within five Business Days after reaching Financial Close, provided that Principal Project Company has delivered the items required under this <u>Section 3.3</u> (Post-Financial Close Requirements and Deliverables).

## 3.4 Potential Adverse Events and Mitigation

**3.4.1** The Parties acknowledge that any one or more Adverse Events may occur during the period between the Effective Date and the Scheduled Financial Close Date. If any such Adverse Event is the sole cause for Principal Project Company's failure or inability to satisfy any of its obligations under <u>Section 3.2.4</u> (Conditions Precedent to Financial Close) by the Financial Close Deadline or Delayed Financial Close Date, as applicable, then <u>Section 3.4.2</u> shall apply.

**3.4.2** Each Party will promptly provide Notice to the other Party if an Adverse Event has occurred ("**Adverse Event Notice**"). The Party provide an Adverse Event Notice may indicate in the notice the potential impact of the event on the schedule for Financial Close. City may elect, in its sole discretion, to consult and work with Principal Project Company to mitigate the actual or anticipated impacts of the occurrence of such event(s). City will provide Notice to Principal Project Company of City's decision to either take mitigation actions, or to not take mitigation actions, within 15 days of the Adverse Event Notice. If City elects mitigation, the Parties will negotiate in good faith to mutually agree on the actual mitigation actions to be undertaken by each Party. City anticipates that the Parties may take one or more of the following actions to mitigate the impacts and for Principal Project Company to be able to achieve Financial Close:

- (a) City may agree to delay Financial Close to a date not later than 120 days after the Scheduled Financial Close Date (the "Delayed Financial Close Date"), in which case, if the Delayed Financial Close Date is after the last day of the original Finance Plan Validity Period, then:
  - Principal Project Company shall extend the validity of its Financial Close Security to a date no earlier than 15 days later than the Delayed Financial Close Date;
  - (ii) If the Adverse Event was an Adverse Event set forth in <u>clauses (a)-(h)</u> of the definition of "Adverse Event", City shall compensate Principal Project Company for the actual cost of extending the validity of its Financial Close Security to the Delayed Financial Close Date within 45 days after receiving Principal Project Company's request and supporting documentation for such payment;
  - (iii) If the Adverse Event was an Adverse Event set forth in <u>clauses (a)-(h)</u> of the definition of "Adverse Event", Principal Project Company shall be entitled to escalate its D&C Contract Amount to adjust for escalation of labor, materials and equipment costs, if any, based on the change in:

     (1) the average of the BCI published each calendar month during the 12 calendar months preceding the Scheduled Financial Close Date; and (2) the average of the BCI published each calendar month during the 12 calendar month period preceding the Financial Close Date; and
  - (iv) Principal Project Company shall prepare a Financial Model Update in accordance with <u>Section 4.7.2</u> (Updates to the Financial Model);

- (b) City may change the amount or timing of the Milestone Payment, provided that Principal Project Company shall be entitled to recover from City the reasonably incurred costs associated with such process;
- (c) City may increase the Base Capital MaxAP and/or Base IFM MaxAP by an amount required to mitigate the relevant Adverse Event, including by an amount greater than 10% of the aggregate of the Base Capital MaxAP and Base IFM MaxAP;
- (d) Principal Project Company may:
  - conduct negotiations for at least 30 days with any or all of Principal Project Company's existing Lenders to increase, renew or extend their commitments, as applicable, provided that, any material deviations from the terms and conditions of the original commitments in Principal Project Company's Financial Proposal may be accepted by Principal Project Company only with City's approval; or
  - (ii) conduct a timely, competitive process to obtain new financing commitments (a "Project Debt Competition") to supplement or replace any of the original financing commitments in Principal Project Company's Financial Proposal; in which case, any such negotiations or Project Debt Competition (A) shall be transparent and open to City and its advisors, and (B) shall have the key objective of obtaining debt financing for the Project at the lowest-cost commercially available (given the terms and conditions of the Contract Document) and on terms and conditions otherwise reasonably acceptable to Principal Project Company and City;
- (e) upon the occurrence of an Affordability Event, Principal Project Company may elect to assume the cost and expense of that portion of the increase in the aggregate of the Base Capital MaxAP and Base IFM MaxAP that exceeds 10%; and/or
- (f) either Party may take any other action mutually agreed upon by City and Principal Project Company.

**3.4.3** The Parties acknowledge that the objective of the mitigation actions described in <u>Section 3.4.2</u>, and any others mutually agreed upon by the Parties, is to create circumstances allowing Financial Close to be achieved on terms that at a minimum will allow Principal Project Company to satisfy its obligation to obtain all financing required for the Project on terms and conditions substantially similar to those in its Financial Proposal. The Parties further acknowledge that if an Adverse Event set forth in <u>clauses (i)-(j)</u> of the definition of "Adverse Event" occurs, the objective of the time extension and mitigation actions described in <u>Section 3.4.2</u> and any others mutually agreed upon by the Parties is for Principal Project Company to have an opportunity to address the impacts of such Adverse Event with its Lenders in order for Principal Project Company to comply with its obligations to achieve Financial Close by the Delayed Financial Close Date.

## 3.5 Permitted Excuses from Achieving Financial Close

**3.5.1** With respect to an Adverse Event set forth in <u>clauses (a)-(h)</u> of the definition of "Adverse Event", Principal Project Company's obligation to achieve Financial Close by the Financial Close Deadline shall be excused if one or more events described in <u>Section 3.4.1</u> has occurred, provided that:

- (a) City notifies Principal Project Company that it will not take any action described under <u>Section 3.4.2;</u>
- (b) City fails to provide Notice to Principal Project Company within 15 days of receipt of the Adverse Event Notice from Principal Project Company of City's decision to either take mitigation actions, or to not take mitigation actions;
- (c) the Parties undertake one or more actions under <u>Section 3.4.2</u>, but the effect of such actions does not allow Principal Project Company to satisfy its obligation to obtain all financing required for the Project on terms and conditions substantially similar to those in its Financial Proposal;
- (d) Principal Project Company has made good faith efforts to take the action described under <u>Section 3.4.2(d)</u> and Principal Project Company has diligently and timely conducted negotiations with existing Lenders and/or the Project Debt Competition but was unable to obtain sufficient financing to satisfy its obligations under this Agreement on terms and with conditions reasonably acceptable to the Parties; or
- (e) Principal Project Company has negotiated in good faith with City to mutually agree on mitigation actions, but the Parties fail to agree on any action under <u>Section 3.4.2</u>.

**3.5.2** With respect to an Adverse Event set forth in <u>clauses (i)-(j)</u> of the definition of "Adverse Event", Principal Project Company's obligation to achieve Financial Close by the Financial Close Deadline shall be excused if, following the occurrence of such Adverse Event until the Delayed Financial Close Date, a new Adverse Event set forth in <u>clauses (a)-(h)</u> occurs (in which case, the provisions of <u>Section 3.5.1</u> shall apply to such new Adverse Event (but not with respect to the original Adverse Event set forth in <u>clauses (i)-(j)</u> of the definition of "Adverse Event").

**3.5.3** With respect to an Adverse Event set forth in <u>clauses (a)-(h)</u> of the definition of "Adverse Event", Principal Project Company has no obligation to reach Financial Close during the period between delivery of an Adverse Event Notice and agreement of the Parties on the mitigation actions to be undertaken by each Party to achieve Financial Close. With respect to an Adverse Event set forth in <u>clauses (i)-(i)</u>, Principal Project Company has no obligation to reach Financial Close during the period between delivery of an Adverse Event Notice and up to 120 days after such delivery, but otherwise shall remain obligated to reach Financial Close unless excused pursuant to <u>Section 3.5.2</u>.

## 3.6 No-Fault Termination

**3.6.1** City may, by delivering to Principal Project Company a Notice specifying City's election to terminate and its effective date, terminate this Agreement prior to Financial Close if City determines, in its sole discretion, that termination is in City's best interest.

**3.6.2** Either Party may terminate this Agreement, without fault or penalty, upon 15 days prior Notice to the other Party, if Financial Close is not achieved by the Financial Close Deadline or Delayed Financial Close Date, as applicable, and such failure is excused pursuant to <u>Section 3.5</u> (Permitted Excuses from Achieving Financial Close).

- **3.6.3** With respect to a termination under <u>Section 3.6.1</u> or <u>Section 3.6.2</u>:
  - (a) City shall return the Financial Close Security within five Business Days after termination; and
  - (b) Principal Project Company shall be entitled to Termination Compensation in an amount not to exceed \$9,990,000 based on commercially reasonable evidence of Principal Project Company's costs incurred during (i) the PDA Term, and (ii) from the period commencing on the Effective Date and ending on the Early Termination Date, as indicated in City's Notice delivered pursuant to <u>Section 3.6.1</u>. Such payment shall be the exclusive compensation payable by City to Principal Project Company under this Agreement for a termination under <u>Section 3.6.1</u> or <u>Section 3.6.2</u>.

**3.6.4** Payment of Termination Compensation in accordance with <u>Section 3.6.3(b)</u> is conditioned upon City's receipt from Principal Project Company, within 30 days of termination of this Agreement under <u>Section 3.6.1</u> or <u>Section 3.6.2</u>, of (i) a complete and compliant invoice requesting payment in the form set forth in <u>Exhibit 17</u> (Section 3.6 Invoice); and (ii) all work product set forth in <u>Section 3.6.5</u>. Such Termination Compensation shall be due and payable no later than 30 Business Days following City's receipt of such invoice.

**3.6.5** Payment of Termination Compensation in accordance with <u>Section 3.6.3(b)</u> is in consideration for ownership and title to all work product (including, subject to the terms of <u>Section 21.4</u> (Intellectual Property), any Developed IP) produced by Principal Project Company related to the Project along with all IP Materials generated as of the date of Notice provided pursuant to <u>Section 3.6.1</u> or <u>Section 3.6.2</u>, including (a) all written and electronic correspondence, exhibits, photographs, reports, printed material, tapes, disks, designs, concepts, ideas, technology, techniques, methods, processes, drawings, plans, specifications and other graphic and visual aids generated or developed by or on behalf of Principal Project Company during the Project procurement process and during the PDA Term, including aesthetic design concepts, interim design submittals, and other items delivered or submitted in any medium, media or format by or on behalf of Principal Project Company to City during the Project procurement process, during the PDA Term, or in connection with the Technical Proposal, and (b) all design, planning, materials, equipment, tools, supplies and other Work developed in connection with the Contract Documents.

## 3.7 Principal Project Company's Failure to Achieve Financial Close

**3.7.1** If Financial Close is not achieved by the Financial Close Deadline or Delayed Financial Close Date, as applicable, then, unless such failure is excused under <u>Section 3.5</u> (Permitted Excuses from Achieving Financial Close), City shall have the right to:

- (a) terminate this Agreement in its entirety by written Notice to Principal Project Company with immediate effect; and/or
- (b) draw and retain the full amount of the Financial Close Security, as City's sole remedy against Principal Project Company under this Agreement for Principal Project Company's failure to achieve Financial Close; <u>provided</u>, <u>however</u>, that, nothing herein shall prejudice any rights or remedies City may have for actions or breaches, other than failure to achieve Financial Close, caused by Principal Project Company under this Agreement or the Early Works Agreement, if any.

**3.7.1.2** Exercising either remedy under this <u>Section 3.7.1</u> will not prejudice any rights or remedies the City may have for other actions or breaches Principal Project Company causes under this Agreement or, if applicable, any Early Works Agreement.

**3.7.2** City's right to draw upon the Financial Close Security is not intended to constitute a penalty, but is intended to be, and shall constitute, liquidated damages to compensate City for the cost of foregoing alternative opportunities and for other costs incurred by City in reliance upon Principal Project Company's agreement to enter into the transactions contemplated hereby.

## ARTICLE 4. PRINCIPAL PROJECT COMPANY FINANCING; LENDERS' RIGHTS; REFINANCING; PRIVATE CAPITAL INVESTMENTS; FINANCIAL MODEL

## 4.1 Principal Project Company Right and Responsibility to Finance

**4.1.1** Principal Project Company is solely responsible for obtaining and repaying, at its own cost and risk and without recourse to the City or SFMTA, all financing necessary for the Work that is the Principal Project Company's responsibility under the Contract Documents. Principal Project Company shall take all appropriate action to obtain the Project Debt and Committed Investment as described in the Financial Proposal on or before the Financial Close Deadline. If the Finance Plan includes any other tax-exempt financing, Principal Project Company bears all risks relating to securing a Conduit Issuer, receiving the necessary approvals and compliance with applicable federal requirements.

**4.1.2** Principal Project Company may grant security interests in or assign the entire PPC's Interest (but not a portion of such interest) to Lenders for purposes of securing the Project Debt, subject to the terms of the Contract Documents. Principal Project Company shall not pledge or encumber the PPC's Interest, or any portion of such interest, to secure any indebtedness of any Person other than (a) Principal Project Company, (b) any special purpose entity that owns Principal Project Company but no other assets and has purposes and powers limited to the Project and Work, (c) a special purpose entity subsidiary owned by either Principal Project Company or an entity described in <u>clause (b)</u> of this <u>Section 4.1.2</u>, or (d) a Conduit Issuer.

**4.1.3** Except as otherwise provided in <u>Section 3.2.3</u> (Base Interest Rate Fluctuation and Credit Spread Risk Mitigation), Principal Project Company bears all risk of any changes in the interest rate, payment provisions, collateral requirements, financing charges, make whole amounts, hedge agreements, prepayment premiums, breakage charges or the other terms of Project Debt and Committed Investment.

**4.1.4** Notwithstanding the foreclosure or other enforcement of any security interest created by a Security Document, Principal Project Company shall remain liable to City for the payment of all sums owing to City under this Agreement and the performance and observance of all of Principal Project Company's covenants and obligations under the Contract Documents.

# 4.2 No City or City Responsibility for Project Debt

**4.2.1** All Project Debt or other obligations issued or incurred by a PPC-Related Entity in connection with this Agreement or the Project shall be issued or incurred only in the name of a PPC-Related Entity or a Conduit Issuer or other entity acting on behalf of a PPC-Related Entity as the ultimate obligor. Except as otherwise expressly provided in this Agreement, the City shall have no obligation to pay debt service on any Project Debt or any other debt issued or incurred by a PPC-Related Entity. The City shall have no obligation to join in, execute or guarantee any note or other evidence of indebtedness of a PPC-Related Entity, any other Financing Agreement or any Security Document (other than the Direct Agreement). Project Debt does not constitute indebtedness, or a pledge of the faith and credit, of City or any department of the City. The Lenders, individually or collectively, have no right to have taxes levied or compel appropriations by City, including the Board of Supervisors, for the payment of any or all of the amount of such principal of, premium, if any, and interest on Project Debt.

**4.2.2** Except for a violation by City of its express obligations to Lenders in any Direct Agreement, no Lender is entitled to pursue any remedy against City, including any right to seek any damages or other amounts from the City, whether for Project Debt or any other amount.

**4.2.3** Section 4.2.2 does not affect City's liability to Principal Project Company under <u>Article</u> <u>17</u> (Termination) for Termination Compensation that is measured in whole or in part by reference to outstanding Project Debt.

**4.2.4** City shall have no obligation to any Lender under the Contract Documents, except to the extent of any express obligations of City to Lenders under any Direct Agreement or in any other instrument or agreement signed by City in favor of such Lender or Collateral Agent. This <u>Section 4.2.4</u> does not preclude Lender enforcement of this Agreement against City where the Lender has succeeded to the PPC's Interest, whether by way of assignment or subrogation.

## 4.3 Lenders' Rights

**4.3.1** This Agreement is exclusively for the benefit of City and Principal Project Company, and shall not provide any Lender with any remedy, claim, liability, reimbursement, cause of action or other right, except for the rights of any Lender as provided in any Direct Agreement.

**4.3.2** The rights of City under <u>Article 16</u> (Default; Remedies) and <u>Article 17</u> (Termination) are subject to the terms of any Direct Agreement.

## 4.4 Refinancing

## 4.4.1 Right of Refinancing

City's prior written approval is required for all Refinancings other than Exempt Refinancings and Rescue Refinancings. City shall have no obligations or liabilities in connection with any Refinancing other than its obligations relating to Lender's rights in any Direct Agreement. If the Refinancing is with a new Lender, the new Lender may be added to an existing Direct Agreement or City shall enter into a new Direct Agreement with the new Lender, if such Lender so elects.

## 4.4.2 Notice of Refinancing

**4.4.2.1** At least 60 days before the proposed date for closing any proposed Refinancing (except an Exempt Refinancing under <u>clause (b)</u>, (c) or (d) of the definition of Exempt Refinancing), Principal Project Company shall submit to City a summary of the proposed Refinancing, together with a schedule setting forth the various activities, each with schedule durations, to be accomplished from commencement through the close of the proposed Refinancing.

**4.4.2.2** Principal Project Company shall provide at least 30 days' advance Notice to City of any intended Exempt Refinancing under <u>clause (b)</u>, (c) or (d) of the definition thereof, including facts and documents of such Exempt Refinancing which shall include, at a minimum, the documents described in <u>Section 4.4.2.4</u> and the reason Principal Project Company considers it to be an Exempt Refinancing.

**4.4.2.3** Within 20 days after receipt of the materials required under <u>Section 4.4.2.2</u>, City will review and provide Notice to Principal Project Company as to whether, in its opinion, the proposed Refinancing is an Exempt Refinancing.

**4.4.2.4** At least 45 days before the proposed date for closing any Refinancing (except an Exempt Refinancing under <u>clause (b)</u>, <u>(c)</u> or <u>(d)</u> of the definition of Exempt Refinancing), Principal Project Company shall:

- (a) provide draft proposed Financing Agreements and Security Documents (or term sheets therefor, if drafts are not then available), available Refinancing Data, and any other submittals required by <u>Exhibit 5C</u> (Calculation of Refinancing Gain); and
- (b) if applicable, provide Notice to City setting out the facts to support the basis for characterization of the transaction as an Exempt Refinancing or Rescue Refinancing.

**4.4.2.5** Within 15 days after receipt of the materials required under <u>Section 4.4.2.4</u>, City will provide Notice to Principal Project Company of City's determinations regarding the following:

- (a) whether the proposed Refinancing is an Exempt Refinancing or Rescue Refinancing;
- (b) if the proposed Refinancing is neither an Exempt Refinancing or Rescue Refinancing, whether to approve or disapprove the proposed Refinancing; and
- (c) if City will approve the proposed Refinancing, whether the Refinancing Gain requirements apply.

**4.4.2.6** City's failure to deliver to Principal Project Company Notice of the determinations and selection within the time period set forth in <u>Section 4.4.2.5</u> shall not prejudice City's right to disapprove the proposed Refinancing, to receive any portion of Refinancing Gain, or its selection of the means for payment of such portion.

**4.4.2.7** At least 10 days before the proposed date for closing the Refinancing, Principal Project Company shall deliver to City substantially final drafts of the proposed Financing Agreements and Security Documents, together with updated versions of the Refinancing Data.

**4.4.2.8** Within five Business Days after close of the Refinancing, Principal Project Company shall deliver to City copies of all signed Financing Agreements and Security Documents in connection with the Refinancing, and the final Refinancing Data.

**4.4.2.9** Within 10 Business Days after close of the Refinancing, City and Principal Project Company shall meet and confer to agree upon the final calculation of the Refinancing Gain in accordance with <u>Section 4.5</u> (Refinancing Gain). Once the final calculation is made, Principal Project Company shall pay City its portion of the Refinancing Gain in accordance with City's selected method of payment. If there is any dispute regarding the amount owing, Principal Project Company shall pay the undisputed amount to City and the amount in dispute shall be subject to resolution under the Contract Dispute Procedures.

## 4.4.3 Refinancing Limitations, Requirements and Conditions

**4.4.3.1** If City renders any assistance or performs any requested activity in connection with a Refinancing apart from delivering a consent and estoppel certificate under any Direct Agreement, then concurrently with close of the Refinancing, and as a condition precedent to Principal Project Company's right to close the Refinancing, Principal Project Company shall reimburse City for Recoverable Costs incurred in connection with the Refinancing. City will deliver to Principal Project Company a written invoice and demand before the scheduled date of closing. If for any reason the Refinancing does not close, Principal Project Company shall reimburse City for Recoverable Costs incurred in connection with the proposed Refinancing within 10 days after City delivers to Principal Project Company a written invoice and demand for such costs.

**4.4.3.2** Principal Project Company shall bear all risks for any Refinancing that negatively affects its Equity IRR, Key Ratios or financial performance.

## 4.5 Refinancing Gain

**4.5.1** City shall be entitled to receive 50% of any Refinancing Gain attributable to any Refinancing other than an Exempt Refinancing. The Refinancing Gain amount shall be, calculated in accordance with Exhibit 5C (Calculation of Refinancing Gain).

**4.5.2** Commencing at least 40 days before the proposed date for closing any Refinancing (except an Exempt Refinancing under <u>clause (b)</u>, (c) or (d) of the definition of Exempt Refinancing), the Parties will negotiate in good faith to determine the method by which City will receive its portion of the Refinancing Gain, if applicable, which includes one or a combination of the following methods:

- (a) a single payment on or about the date of the Refinancing in an amount less than or equal to any Distribution made on or about the date of Refinancing;
- (b) a credit or payment by Principal Project Company to City that effectively reduces the Availability Payments over the remainder of the Term; or
- (c) a combination of <u>clauses (a)-(b)</u>.

#### 4.6 Equity Requirements

Principal Project Company shall have and maintain a Construction Equity Ratio totaling not less than 5% throughout the period between the Financial Close Date and the Substantial Completion Date, except to the extent:

- (a) City otherwise approves in writing, in its sole discretion;
- (b) Principal Project Company shall reduce the amount of the Construction Equity Ratio below 5% as part of a workout of a breach or default under the Initial Financing Agreements or Initial Security Documents; or

(c) the amount of Construction Equity Ratio is reduced below 5% because Principal Project Company incurs additional Project Debt pursuant to a Refinancing.

## 4.7 Financial Model and Financial Model Updates

## 4.7.1 Model Update Event

- **4.7.1.1** Model Update Event means any of the following events:
  - the implementation at Financial Close of the Base Interest Rate fluctuation and Credit Spread Risk Mitigation process pursuant to <u>Section 3.2.3</u> (Base Interest Rate Fluctuation and Credit Spread Risk Mitigation);
  - (b) the implementation of any mitigation actions set forth in <u>Section 3.4.2;</u>
  - (c) a Compensable Delay Event or a Compensable Relief Event for which City owes Principal Project Company compensation pursuant to <u>Sections 14.1.4</u> (Costs Payable for Compensable Delay Events), <u>Section 14.1.6</u> (Costs Payable for Force Majeure Events During the D&C Term), <u>Section 14.1.7</u> (Costs Payable for Unavoidable Delay Events During the D&C Term), or <u>Section 14.2.4</u> (Costs Payable for Force Majeure During IFM Period), respectively, which City has elected to pay as an adjustment to the Availability Payments over the Term;
  - (d) an event for which City is entitled to compensation from Principal Project Company pursuant to <u>Section 16.2</u> (City Remedies for PPC Default);
  - (e) a Refinancing resulting in a Refinancing Gain to which City is entitled to a share pursuant to <u>Section 4.5</u> (Refinancing Gain); and
  - (f) any amendments to the Contract Documents that the Parties agree has a material effect on the Financial Model, including any amendments agreed to by the Parties between the Finance Plan Due Date and Financial Close.

**4.7.1.2** Whenever a Model Update Event occurs (except as otherwise provided in this Agreement or where the Parties mutually agree otherwise), the financial consequence shall be determined in accordance with this <u>Section 4.7</u> (Financial Model and Financial Model Updates).

## 4.7.2 Updates to the Financial Model

**4.7.2.1** In connection with any Model Update Event that entitles either Party to the payment of any amount, Principal Project Company may, in consultation with City, prepare an updated Financial Model in accordance with this <u>Section 4.7.2</u> (Updates to the Financial Model) (each, a "**Financial Model Update**") to reflect the financial impacts of such Model Update Event and calculate the amount due to either Party as a result of the Model Update Event. Principal Project Company may address more than one Model Update Event within a single Financial Model Update, provided that City may request interim versions of a Financial Model Update that address the impact of only a single Model Update Event. No Financial Model Update shall have any effect on the rights and obligations of the Parties under this Agreement until both Parties

have agreed in writing to accept it as an amendment to this Agreement pursuant to <u>Section</u> <u>4.7.3</u> (Replacement of Financial Model).

**4.7.2.2** Principal Project Company shall propose the Financial Model Update by Notice to City giving the proposed revised Financial Model together with full and complete details and explanation of the assumptions and calculations used to reflect the financial impacts of the Model Update Event which may only include changes to Principal Project Company's cash revenues and expenses that arise directly from the Model Update Event and consequential changes to the Project Debt draw down schedule, funding and release of reserves, financing costs, debt service schedule and amounts, Committed Investment draw down schedule and Principal Project Company's Distributions schedule and amounts or any other impacts or consequential changes mutually agreed by the Parties.

**4.7.2.3** A Financial Model Update:

- (a) may incorporate only the following revisions:
  - (i) changes to Principal Project Company's cash revenues and expenses that arise directly from the Model Update Event; and
  - (ii) consequential changes to the Project Debt draw down schedule, funding and release of reserves, financing costs, debt service schedule and amounts, Committed Investment draw down schedule and Principal Project Company's Distributions schedule and amounts; and
- (b) may not:
  - (i) incorporate other information or assumptions based on Principal Project Company's actual Project Financial Performance, except as permitted by <u>Exhibit 5C</u> (Calculation of Refinancing Gain) for a Financial Model Update related to a Refinancing; or
  - (ii) generally update projections through the end of the Term based on current market conditions.

**4.7.2.4** Principal Project Company may amend the logic or formulae incorporated in a Financial Model Update to the extent necessary to permit adjustments to the Financial Model in accordance with this <u>Section 4.7</u> (Financial Model and Financial Model Updates). However, if any amendment is to be made to the logic or formulae in a Financial Model Update, the Key Ratios in the Financial Model Update must be maintained at levels that are neither lower nor higher than the Key Ratios existing in the Financial Model then in effect, and the difference in the Equity IRR after and immediately before making such amendment may not be greater than one basis point (being 0.01%) or as may be agreed upon by the Parties.

**4.7.2.5** Principal Project Company shall provide City with access, on an Open Book Basis, to the set of updated and revised assumptions and other data that comprise or are included in any Financial Model Update. Principal Project Company shall also provide a reconciliation and explanation of all the adjustments applied in the Financial Model Update. City may challenge the validity, accuracy or reasonableness of any Financial Model Update or any portion thereof

and may require Principal Project Company to correct any detected errors in the Financial Model Update or Financial Model, as applicable.

### 4.7.3 Replacement of Financial Model

**4.7.3.1** Following mutual agreement by Principal Project Company and City on interim versions of a Financial Model Update, Principal Project Company shall deliver to City, in form and substance reasonably acceptable to City:

- the final version of the Financial Model Update and the related amended Financial Modeling Data, in the same form as the versions delivered pursuant to <u>Section 3.3</u> (Post-Financial Close Requirements and Deliverables) or in such other form as may be agreed upon by the Parties;
- (b) an updated model audit report related to such Financial Model Update in accordance with <u>Section 4.7.4</u> (Financial Model Audits); and
- (c) a form of written amendment that (i) effects the replacement of the then current Financial Model then in effect with the proposed Financial Model Update, and (ii) addresses all other amendments to this Agreement that may be required as a result of the Model Update Event and the amended Financial Model, including any required amendments to the definitions of Base Capital MaxAP, Base IFM MaxAP, Equity IRR and Key Ratios.

**4.7.3.2** Upon approval by City of the materials provided under <u>Section 4.7.4.1</u> and execution of the written amendment by the Parties, the Financial Model Update shall become the Financial Model for the purposes of this Agreement until further amended, as applicable. Each Financial Model in effect under this Agreement shall be assigned an exclusive identification number, using chronological sequencing, and shall be clearly marked with Principal Project Company's name, date of submittal, contract number and the words, "Financial Model for Escrow".

**4.7.3.3** City reserves the right to approve any material changes in the debt structure (e.g., fixed or variable rate, bank financing or bond financing, call provisions) before Financial Close that constitute a deviation from the assumptions in <u>Exhibit 3A</u> (Financial Proposal). Material changes, as used in this <u>Section 4.7.3.3</u>, include (i) a change that differs materially from any commitments or provisions set forth in the Finance Plan; (ii) a change which would alter the risk allocations in this Agreement or any Key Contract; (iii) a change which could increase the liability of City for termination compensation in the event of a termination; (iv) a change that would modify the gearing of the financing; (v) a change that would result in an Affordability Event; (vi) a change that would cause a Delayed Financial Close Date; (vii) a change in any of the Security Documents; (viii) a change in the funding sources of Common Infrastructure; and/or (ix) a change under any of the conditions precedent to Financial Close under the Financing Documents.

**4.7.3.4** In the event of a Contract Dispute about the Financial Model, the Financial Model or the immediately preceding Financial Model Update (as applicable) that is not being disputed (or, if there has been no undisputed Financial Model Update, the Financial Model) will remain in effect until such Contract Dispute is resolved or a new Financial Model Update is issued and not disputed. If a proposed Financial Model or Financial Model Update (as applicable) has not been disputed, or if any such Contract Dispute has been so resolved, the proposed Financial Model

or Financial Model Update (as applicable) will serve as the Financial Model or the current Financial Model Update (as applicable).

## 4.7.4 Financial Model Audits

**4.7.4.1** Any model audit report delivered to City pursuant to <u>Sections 3.3</u> (Post-Financial Close Requirements and Deliverables) and <u>4.7</u> (Financial Model and Financial Model Updates) shall be prepared by an independent audit firm with a nationally recognized reputation. The cost of a model audit report shall be paid solely by City if it is prepared in connection with a Financial Model Update related to a Model Update Event described in <u>Section 4.7.1.1(c)</u> or <u>4.7.1.1(f)</u> when an amendment is proposed only by City. The cost of any other model audit report required hereunder shall be paid solely by Principal Project Company.

**4.7.4.2** Principal Project Company shall bear the entire risk of any errors or omissions contained in the Financial Model and shall not be entitled to any compensation or other relief from City in relation to any loss or damage that it suffers as a result of such error or omission.

## 4.8 Escrow of Financial Model and Cost and Pricing Data

**4.8.1** City and Principal Project Company shall, within 10 days after the Effective Date, jointly deposit the Base Case Financial Model in a locked cabinet at SFMTA or another location approved by City, with both Parties having a key to the cabinet. Replacement Financial Models shall be deposited within 10 days after the Parties have executed and delivered the amendment in accordance with <u>Section 4.7.3</u> (Replacement of Financial Model).

**4.8.2** The Parties acknowledge that the Cost and Pricing Data has been placed into escrow as provided in <u>Section 4.8.1</u>. Concurrently with approval of each Change Order or other amendment to the Contract Documents, Principal Project Company shall deposit in the Cost and Pricing Data escrow one copy of all documentary information used by Principal Project Company in connection with pricing for the Change Order or other amendment, including quotations from Contractors.

**4.8.3** Principal Project Company represents and warrants that:

- (a) the material initially delivered into escrow constitutes the Base Case Financial Model and Cost and Pricing Data provided in connection with the Financial Proposal and an authorized officer of Principal Project Company has personally examined the contents of the electronic file and/or electronic storage media, as applicable, and they are complete; and
- (b) the Cost and Pricing Data constitutes all of the information used by Principal Project Company in determining the cost of the D&C Work, IFM Services and Renewal Work in preparation of the Implementation Proposal and, unless City agrees or directs otherwise, Principal Project Company shall not use any other Implementation Proposal preparation information in the negotiation of Change Orders or in connection with the potential resolution or settlement of Claims or Contract Disputes.

**4.8.4** Whenever Principal Project Company makes an additional deposit of any replacement Financial Models or Cost and Pricing Data into escrow, Principal Project Company

shall certify to City in writing at the time of deposit that: (a) the material deposited into escrow constitutes the true replacement Financial Models or Cost and Pricing Data, as applicable; (b) an authorized officer of Principal Project Company has personally examined the contents of the deposit; and (c) the deposit is complete.

**4.8.5** City may conduct a review of the Cost and Pricing Data in accordance with the procedure set forth in <u>Section 4.8.6</u> to determine whether it is complete. In the event City determines that any Cost and Pricing Data is missing, City may request that Principal Project Company submit the missing data and Principal Project Company shall provide such Cost and Pricing Data within three Business Days of the request, and at that time it will be date stamped, labeled to identify it as supplementary information, and added to the escrowed Cost and Pricing Data. Principal Project Company shall have no right to add documents to the Cost and Pricing Data except as otherwise provided in this <u>Section 4.8</u> (Escrow of Financial Model and Cost and Pricing Data).

**4.8.6** Each of City and Principal Project Company shall have the right to examine, through one or more designated representatives, any and all components of the Financial Model and Cost and Pricing Data during the SFMTA's normal business hours. The Party undertaking an examination need not have or state a specific reason to examine such material. Without limiting the foregoing, the Parties recognize that examination of the escrowed material may assist in the negotiation of Change Orders, or may assist in the potential resolution or settlement of Claims or Contract Disputes.

**4.8.7** City will provide Notice to Principal Project Company in writing at least two Business Days in advance of City's examination of escrowed material, and shall allow Principal Project Company to be present at the examination. City may make or retain copies of escrowed material, subject to terms reasonably necessary to protect the confidentiality and proprietary nature of the contents, as may be agreed upon by the Parties and subject to applicable Laws.

**4.8.8** Subject to applicable Laws, the escrowed material is, and shall remain, the property of Principal Project Company or its Contractors.

**4.8.9** Principal Project Company agrees that the Cost and Pricing Data is not part of the Contract Documents and that nothing in the Cost and Pricing Data shall change or modify the Contract Documents.

**4.8.10** Either Party may introduce escrowed material into evidence in accordance with the Contract Dispute Procedures. The Parties shall promptly abide by any request from the court or other dispute resolver to receive, review and utilize the Financial Model and Cost and Pricing Data to assist the dispute resolver in its deliberations.

**4.8.11** The escrow shall remain in effect throughout the Term and thereafter until final resolution of all Contract Disputes, subject to any mutual agreement of the Parties to retrieve and/or discard materials therein from time to time.

**4.8.12** Principal Project Company shall not be entitled to any additional payment for compilation of materials to be deposited into escrow or any other Principal Project Company expenses for complying with this <u>Section 4.8</u> (Escrow of Financial Model and Cost and Pricing Data).

# ARTICLE 5. SUBMITTALS; MANAGEMENT SYSTEMS AND OVERSIGHT

## 5.1 Submittal Review Terms and Procedures

### 5.1.1 Terms and Procedures

Principal Project Company shall comply with the terms and procedures for Submittals review set forth in <u>Exhibit 11</u> (Submittals Review Process). Except as set forth in this Agreement, including <u>Exhibit 11</u> (Submittals Review Process), the standard for review and approval of design and construction Submittals that comply with the Technical Requirements shall be reasonable discretion; <u>provided</u>, <u>however</u>, that decisions regarding waivers, releases, consents, acceptance of Nonconforming Work, Deviations, PPC Change Requests and other matters that are not consistent with or comply with the Contract Documents shall be within City's sole discretion.

#### 5.1.2 Conflicting Provisions

<u>Exhibit 11</u> (Submittals Review Process) sets forth uniform terms and procedures for Submittals. In the event of any conflict between the provisions of <u>Exhibit 11</u> (Submittals Review Process) and any other provisions of the Contract Documents or with the Project Management Plan concerning procedures with respect to submission, review, comment, approval, consent, determination, decision or other actions with respect to Submittals, <u>Exhibit 11</u> (Submittals Review Process) shall exclusively govern and control the procedures, except to the extent that the conflicting provision expressly states that it supersedes <u>Exhibit 11</u> (Submittals Review Process).

## 5.1.3 Limitations on Principal Project Company's Right to Rely

**5.1.3.1** No action or failure to take action by or on behalf of City relating to Oversight (including review and approval of the Project Management Plan) or other act or omission of City shall:

- (a) constitute an approval or acceptance by City of Principal Project Company's performance of its obligations in accordance with the Contract Documents;
- (b) alter, waive, diminish, release or otherwise prejudice any rights, remedies or powers that City has under the Contract Documents or otherwise;
- (c) limit Principal Project Company's obligation to perform the Work in accordance with the Contract Documents; or
- (d) affect Principal Project Company's liabilities and obligations to fulfill the requirements of the Contract Documents (including its indemnity obligations).

**5.1.3.2** Principal Project Company acknowledges and agrees that Oversight, including review, comment, exception, objection, rejection, approval, disapproval, acceptance, concurrence, certification or failure to conduct any such activity by City:

(a) is solely for the benefit and protection of City;

- (b) does not relieve Principal Project Company of its responsibility for the selection and the competent performance of all PPC-Related Entities;
- (c) does not relieve Principal Project Company from compliance with the requirements of the Contract Documents or create or impose upon City any liability, duty or obligation toward Principal Project Company to cause it to fulfill the requirements of the Contract Documents;
- (d) shall not be deemed or construed as any kind of warranty, express or implied, by City;
- (e) may not be relied upon by Principal Project Company or used as evidence in determining whether Principal Project Company has fulfilled the requirements of the Contract Documents;
- (f) shall not relieve Principal Project Company from liability for, and responsibility to replace, Nonconforming Work (including Work based on Design Documents to the extent that they include a change, deviation, modification, alteration or exception from the Technical Requirements not approved as a Deviation) and to cure PPC Defaults;
- (g) shall not be deemed or construed as any assumption of risk by City as to design, construction, equipping, supply, operations, maintenance, performance or quality of the Project or performance of the Work; and
- (h) may not be asserted by Principal Project Company against City as a legal or equitable defense to, or as a waiver of or relief from, Principal Project Company's obligation to fulfill the requirements of the Contract Documents.

**5.1.3.3** Notwithstanding the provisions of <u>Sections 5.1.3.1</u> and <u>5.1.3.2</u>, Principal Project Company may rely on Notices that City gives under this Agreement for purposes of confirming City's approval or consent to an event or matter, but without prejudice to any of City's other rights and remedies under this Agreement, including regarding compliance by Principal Project Company with the requirements of the Contract Documents.

**5.1.3.4** Notwithstanding the provisions of <u>Sections 5.1.3.1</u> and <u>5.1.3.2</u>, Principal Project Company shall be entitled to rely on City's written approval of specific Deviations.

**5.1.3.5** City's approval of Release for Construction Documents shall constitute approval of the design by City for purposes of Government Code section 830.6, but shall not be deemed to relieve Principal Project Company of liability for the design.

## 5.2 Project Management Plan

**5.2.1** Principal Project Company shall comply with the Project Management Plan and its component parts, plans and other included at <u>Exhibit 2</u> (Project Management Plan) and Good Industry Practice, including those requirements applicable to Quality Assurance and Quality Control.

**5.2.2** Principal Project Company shall submit to City, in accordance with the procedures and timeline described in the Technical Requirements, any proposed changes or additions to or revisions of the Project Management Plan and its component parts.

**5.2.3** Principal Project Company shall not commence any aspect of Construction Work before approval of the D&C Management Plan applicable to such Construction Work. Principal Project Company shall not commence any aspect of the IFM Services before approval of the relevant component parts, plans and other documentation of the IFM Management Plan applicable to the IFM Services.

**5.2.4** If any part, plan or other documentation of the Project Management Plan refers to, relies on or incorporates any manual, plan, procedure or like document, then all such referenced or incorporated materials shall be submitted to City for review, marked to identify relevant provisions.

**5.2.5** Principal Project Company shall monitor the Work and prescribe times for internal audits of the Project Management Plan, and shall carry out such internal audits at the times prescribed and shall promptly remedy all findings to City's satisfaction.

**5.2.6** Principal Project Company shall comply with and cause all Contractors to comply with applicable requirements of the Project Management Plan.

**5.2.7** Principal Project Company shall update the Project Management Plan in accordance with the Technical Requirements.

## 5.3 Quality Assurance, Quality Control, Generally

Principal Project Company is primarily responsible for all Quality Assurance and Quality Control activities (including self-monitoring activities) necessary to manage the Project. Principal Project Company shall undertake the primary aspects of Quality Assurance and Quality Control for the Project and the Work in accordance with the Project Management Plan, the Technical Requirements (including <u>Section 1.4</u> of <u>Division 1</u> of the Technical Requirements), other applicable provisions of the Contract Documents, Good Industry Practice and applicable Law. The foregoing shall not limit City's Oversight rights and quality assurance/quality control verification, all as set forth in this Agreement.

## 5.4 Oversight, Inspection and Testing

**5.4.1** City shall have the right at all times to conduct Oversight as provided in this <u>Section 5.4</u> (Oversight, Inspection, and Testing) and <u>Division 6</u> of the Technical Requirements. Such Oversight may include assessments regarding compliance with the Contract Documents, Project Management Plan, and requirements of applicable Governmental Entities and applicable Law. City may designate any Person or Persons to carry out any Oversight on City's behalf.

**5.4.2** City's Oversight rights include, at City's sole option, the following:

(a) monitoring and auditing Principal Project Company and its Books and Records to determine compliance with requirements of the Contract Documents and the Project Management Plan, including (i) audit review of compliance with

quality procedures and processes under PPC's Design Quality Plan, PPC's Construction Quality Plan and IFM Quality Management Plan, and (ii) audit review of Design Documents, Construction Documents, field work plans, land surveys, mapping, other data collection tasks, other Submittals and other Books and Records;

- (b) conducting audits of all design and pre-design activities for the Project as needed to ascertain and evaluate Principal Project Company's design quality and safety control processes, including (i) review of engineering calculations, engineering reports, and findings, (ii) review of the work of Principal Project Company's environmental compliance personnel with the Environmental Compliance Plan, and (iii) review of certifications that Principal Project Company's Quality Control checks of final Construction Documents have been performed and documented, and that the Construction Documents conform to the requirements of the Contract Documents;
- (c) conducting audits of all construction-related activities for the Project as needed to audit Principal Project Company's construction quality and safety control processes, including (i) auditing the services of Principal Project Company's accredited laboratories and associated testing devices and equipment, (ii) reviewing Principal Project Company's construction quality procedures, including conducting field monitoring and inspections as needed for audit purposes of construction activities, materials, and system components, as indicated in the Contract Documents, (iii) auditing Principal Project Company's records of materials, materials tests, materials certifications, and performance tests for Project systems, (iv) reviewing and investigating Project progress, Project quality, Deviations, Defects, and repair and replacement of Nonconforming Work, and (v) conducting field monitoring and inspections;
- (d) conducting inspection and testing of materials or software, including witnessing factory tests, off-site lab tests, and first article inspection of manufactured items to verify Principal Project Company's compliance with testing frequencies and requirements, including (i) performance and acceptance testing, in the Contract Documents, and the Project Management Plan, (ii) the accuracy of the tests, inspections and audits performed in accordance with PPC's Design Quality Plan and PPC's Construction Quality Plan, and (iii) compliance of materials incorporated into the Project with the applicable requirements, conditions and standards of the Contract Documents, Regulatory Approvals, the Project Management Plan, IFM Quality Management Plan and applicable Law;
- (e) accompanying Principal Project Company on physical inspections associated with Principal Project Company's Performance Monitoring, conducting its own performance inspections, assessing and scoring Principal Project Company's IFM Records, and assessing and rating the condition of elements;
- (f) attending and witnessing Principal Project Company's other on-site and offsite tests and inspections, including system start-up and acceptance tests and

inspections, subject to the obligation to observe all applicable Safety Standards and requirements;

- (g) reviewing Principal Project Company's certification of Record Documents and surveys;
- (h) investigating, analyzing and reporting on Safety Compliance and performance of Safety Compliance Orders; and
- monitoring and auditing Principal Project Company's detection, reporting, response times and time to respond to and rectify breaches and failures for which Noncompliance Points may be assessed in accordance with <u>Article 15</u> (Deductions and Noncompliance Points) and <u>Exhibit 4</u> (Payment Mechanism).

**5.4.3** City also has the right, but not the obligation, to conduct "over-the-shoulder" reviews of Design Documents and other Submittals. All such "over-the-shoulder" reviews conducted are for City's sole benefit.

**5.4.4** Nothing in the Contract Documents shall preclude, and Principal Project Company shall not interfere with, any review, inspection or oversight of Submittals or of Work that any Authority Having Jurisdiction may desire to conduct in accordance with its agreements with City or applicable Law.

## 5.5 Testing and Test Results

**5.5.1** All tests shall be carried out in accordance with <u>Division 6</u> of the Technical Requirements, this <u>Section 5.5</u> (Testing and Test Results) and all other applicable provisions of and the Contract Documents.

**5.5.2** Principal Project Company shall develop a test recording system that permits ready retrieval of all test readings and shall provide information relating to tests proposed, test methodology and Test Reports to City on request.

**5.5.3** City may attend and witness any tests and verifications to be conducted with respect to the Project. Principal Project Company shall provide to City all test results and reports (which shall be provided in electronic format in accordance with the Technical Requirements) within 10 Business Days after Principal Project Company or its Contractor receives them. With respect to continuous testing operations (such as concrete quality, structural concrete strengths, aggregate quality, compaction tests and material quality), Principal Project Company shall provide to City at regular intervals (at least weekly unless otherwise agreed) test summary sheets and statistical analyses indicating strength and quality trends.

**5.5.4** Principal Project Company shall give City timely advance Notice (not less than 5 Business Days) of the date and specific location of such tests.

**5.5.5** City's Authorized Representative and any other designee may attend any test and will give advance Notice (not less than one Business Day) of their intent to attend the test. Any materials or plant that fail(s) such tests shall be rejected.

**5.5.6** Principal Project Company acknowledges that, where Principal Project Company's Work impacts a Utility or an element subject to the jurisdiction of an Authority Having Jurisdiction, then the affected Utility Owner or Authority Having Jurisdiction, as applicable, will have the same rights as City under this <u>Section 5.5</u> (Testing and Test Results), subject to the same obligations that apply to City under <u>Section 6.7.3</u>.

## 5.6 Meetings

**5.6.1** Principal Project Company shall conduct coordination and progress meetings with City, in accordance with <u>Section 1.1.3</u> of <u>Division 1</u> of the Technical Requirements during the course of design and construction, including any design and construction occurring during the IFM Period, and in accordance with the IFM Specifications during the performance of the IFM Services. At City's request, Principal Project Company shall require the Engineer of Record and Contractors responsible for or affected by such Work to attend the progress meetings. City and its designated representatives are authorized to attend all such meetings and are permitted to raise any questions, concerns or opinions without restriction.

**5.6.2** The Parties shall hold any other meetings, at such times, frequency and locations, as applicable, as stated in the Technical Requirements.

**5.6.3** City and Principal Project Company, through their respective Authorized Representatives, shall meet from time to time at the other Party's request to discuss and resolve issues relating to the Work.

**5.6.4** Principal Project Company shall schedule all meetings with City at a date, time and place reasonably convenient to both Parties.

## 5.7 Reporting

## 5.7.1 Relating to the Work

**5.7.1.1** Principal Project Company shall submit all reports relating to the Work in the form, with the content and within the time required under the Contract Documents.

**5.7.1.2** Principal Project Company shall make available to City information relating to the status of the Work, including non-proprietary information relating to the design, engineering and construction, estoppel certificates, and such other matters as City may reasonably request in accordance with the Technical Requirements.

## 5.7.2 Financial Reporting

**5.7.2.1** On the first anniversary of the Effective Date and on every subsequent anniversary thereof during the Term, Principal Project Company shall deliver to City certified copies of (a) Principal Project Company's most recent annual audited financial statements and (b) any other reporting and notifications provided to Lenders regarding material events (including any draws on Principal Project Company's debt service reserve account) under the Financing Documents.

**5.7.2.2** From the Effective Date until the Final Acceptance Date, Principal Project Company shall deliver to City, on a monthly basis, certified copies of (a) Principal Project Company's draw requests to Lenders (including corresponding payment applications by Key Contractors to

Principal Project Company), and (b) the LTA's reports, which shall clearly state the D&C Percentage as of the date of the report, and the invoices approved by the LTA in connection with the foregoing, in each case, within two Business Days following delivery or receipt, as applicable, by Principal Project Company of the relevant documentation.

**5.7.2.3** During the IFM Period, Principal Project Company shall, concurrently with delivery to any Lender and at least annually, deliver to City a budget for IFM Services for the upcoming year, actual costs incurred in performance of the IFM Services during the preceding year, together with any updates required by or delivered to any Lender.

## ARTICLE 6. GENERAL PRINCIPAL PROJECT COMPANY OBLIGATIONS

## 6.1 Planning and Engineering Activities

Principal Project Company, through appropriately qualified and licensed design professionals, as identified in the Project Management Plan, shall furnish or cause to be furnished all planning and engineering activities appropriate for design and development of the Project in accordance with the Contract Documents and Good Industry Practice.

#### 6.2 **Project Site Conditions**

- **6.2.1** Principal Project Company acknowledges and agrees that:
  - (a) it has investigated and satisfied itself as to the conditions affecting the D&C Work, including those bearing upon transportation, disposal, handling and storage of materials, availability of labor, supplies, materials, equipment, water, electric power, roads and uncertainties of weather or similar physical conditions at the Project Site, the conformation and conditions of the ground, and the character of equipment and facilities needed in connection with the D&C Work;
  - (b) it has satisfied itself as to the character, quality and quantity of surface and subsurface materials or obstacles to be encountered insofar as this information is reasonably ascertainable from an inspection of the Project Site, including the results of exploratory work (including investigations undertaken pursuant to the PDA) and other information publicly available or provided to Principal Project Company by City; and
  - (c) any failure by Principal Project Company to acquaint itself with the available information relating to the conditions affecting the D&C Work will not relieve Principal Project Company from responsibility for estimating properly the difficulty or cost of successfully performing the D&C Work and shall not be the basis of a Relief Event claim.

**6.2.2** During the progress of the Work, if Principal Project Company encounters any Hazardous Materials or other Project Site conditions that may entitle Principal Project Company to claim that a Relief Event has occurred, then Principal Project Company shall provide Notice to City of the specific condition promptly before it is disturbed, or as soon as practicable afterwards, and before the affected Work continues. City shall promptly investigate such conditions.

## 6.3 Regulatory Approvals

## 6.3.1 CEQA Approval and NEPA Approval

**6.3.1.1** The Parties acknowledge that City has obtained the CEQA Approval and Principal Project Company acknowledges receipt of a copy of the CEQA Approval. City is responsible for maintaining the CEQA Approval. City is responsible for costs of litigation relating to the CEQA Approval, subject to the exception provided in <u>Section 6.3.4.2(d)</u>.

**6.3.1.2** Principal Project Company shall provide support to City and undertake additional efforts as specified in <u>Section 6.3.6.1</u> with respect to any modifications, renewals and extensions of the CEQA Approval, including those required as the result of Principal Project Company's design and Relief Events.

[Note: Assuming the NEPA Approval is obtained prior to Commercial Close, the below provisions will be deleted. If not, (i) these provisions shall remain; (ii) the Agreement or the TRs will include an assumed set of conditions, mitigations, etc.; and (iii) material changes in such assumed conditions, mitigations, etc. will be added as a Compensable Delay Event. PNC must include this type of language in its DB RFP and DB Contract as well]

**6.3.1.3** The Parties acknowledge that City has not yet obtained the NEPA Approval. City is responsible for maintaining the NEPA Approval. City is responsible for costs of litigation relating to the NEPA Approval, subject to the exception provided in <u>Section 6.3.4.2(d)</u>.

**6.3.1.4** Given that the City has not yet obtained the NEPA Approval, performance by either Party of its obligations under this Agreement shall not limit:

- (a) City's independent evaluation and sole discretion when conducting the environmental review process required for the NEPA Approval and preparing the NEPA environmental documents; or
- (b) City's sole discretion to withhold issuance of NTP 2.
- 6.3.1.5 Principal Project Company acknowledges and agrees that City retains:
  - (a) sole discretion, exclusive control and decision-making authority over the description of the Project and any identification or evaluation of alternatives or mitigation measures for the purposes of NEPA; and
  - (b) sole discretion over whether to accept the Work, the Infrastructure Facility and Principal Project Company Submittals for purposes of the NEPA environmental review process.

**6.3.1.6** Nothing contained in this Agreement commits, or shall be construed to commit, City to any proposed project or any Project alternative, modification, or mitigation regarding the Project (including a no-build alternative) until, unless and before the NEPA Approval is obtained and City approves the Project, alternative, modification or mitigation.

**6.3.1.7** Principal Project Company shall have no right or obligation to perform, and shall not perform, any services or Work that would violate conflict of interest rules under NEPA regarding the preparation, review, revision and decisions on scope and content of the NEPA environmental documents. All references in this Agreement to Principal Project Company's involvement with the NEPA environmental review process or development of the NEPA environmental documents shall be subject to the limitation in the preceding sentence.

**6.3.1.8** Unless and before the NEPA Approval is obtained, Principal Project Company may not undertake final design or construction Work.

## 6.3.2 Regulatory Approvals other than the CEQA Approval and the NEPA Approval

**6.3.2.1** Principal Project Company shall obtain and maintain all Regulatory Approvals required for the Project and the Work, other than the CEQA Approval and the NEPA Approval, and shall bear the risk of obtaining such approvals and any delay in obtaining such approvals, except as provided in <u>Section 6.3.4.1</u>, as well as the risk of conditions imposed on performance of the Work by such approvals. Principal Project Company shall conduct all necessary environmental studies and prepare all necessary environmental documents in compliance with applicable Environmental Laws as needed to obtain Regulatory Approvals other than the CEQA Approval and the NEPA Approval, and shall obtain all necessary modifications, renewals and extensions thereof.

**6.3.2.2** Principal Project Company shall pay all costs associated with obtaining Regulatory Approvals, other than the CEQA Approval and the NEPA Approval.

#### 6.3.3 Copies to City

Within 10 days of submitting any application for a Regulatory Approval to a Governmental Entity (or any proposed modification, renewal, extension or waiver of a Regulatory Approval or provision thereof), Principal Project Company shall submit an electronic copy of same, together with any supporting studies, analyses and data, to City. Principal Project Company shall deliver to City true and complete copies of all new or amended Regulatory Approvals, other than the CEQA Approval and the NEPA Approval.

#### 6.3.4 Certain Risks Relating to Regulatory Approvals

**6.3.4.1** Except to the extent required as a direct result of (a) a City Change, (b) a City Fault, or (c) Major Approval Delay, Principal Project Company shall not be entitled to any Extra Work Costs, Financing Delay Costs, Delay Costs, time extensions or any other relief associated with securing Regulatory Approvals.

**6.3.4.2** As between City and Principal Project Company, Principal Project Company shall bear all risk arising out of, relating to or resulting from:

- (a) any differences between Principal Project Company's design for any portion of the Project and the design that served as the basis for the application for a Regulatory Approval, except to the extent required as a direct result of a City Change;
- (b) any differences between the means and methods (including temporary works) Principal Project Company chooses for performance of the Work and those stated in, referred to or contemplated in the CEQA Approval and/or the NEPA Approval;
- (c) any change in the Project due to Principal Project Company's design, except to the extent that the change was directly attributable to a Relief Event; and
- (d) cost of litigation associated with Regulatory Approvals, including the CEQA Approval and the NEPA Approval, to the extent that the litigation arises, in

whole or in part, due to Principal Project Company's failure to comply with the requirements of the Regulatory Approval.

#### 6.3.5 Changes to Regulatory Approvals

If Principal Project Company wishes to obtain, modify, renew or extend any Regulatory Approvals, Principal Project Company shall first comply with, and obtain any consent or waiver required in accordance with, then-existing agreements between City and other Governmental Entities.

#### 6.3.6 City Assistance with Regulatory Approvals.

**6.3.6.1** Principal Project Company may, by Notice to City, request City's reasonable assistance and cooperation in obtaining modifying, renewing or extending any Regulatory Approvals (including any modification, renewal or extension of an existing Regulatory Approval required as the result of Principal Project Company's design or construction methods). Upon receipt of such a Notice and agreement of the Parties regarding the scope of, and budget for, assistance to be provided as described in <u>Section 6.3.6.2</u>, City will reasonably assist and cooperate with Principal Project Company in seeking to obtain the Regulatory Approvals, including joining in conferences and meetings with the Governmental Entities with jurisdiction, and providing Principal Project Company data, information and documents available to City and relevant to the application for the Regulatory Approvals

**6.3.6.2** City and Principal Project Company shall work jointly to establish a scope of work and budget for City's Recoverable Costs (which shall be limited to reasonable out-of-pocket costs) incurred in connection with the assistance and cooperation that City agrees to provide in connection with modifications to Regulatory Approvals under <u>Section 6.3.6.1</u>. Subject to an agreed upon scope of work and budget and to any rights of Principal Project Company in the case of a Relief Event, Principal Project Company shall fully reimburse City for such Recoverable Costs incurred in providing such assistance and cooperation, including those incurred to conduct further or supplemental environmental studies.

**6.3.6.3** Assistance provided by City under <u>Section 6.3.6.1</u> shall not include any obligation to:

- (a) coordinate and work with elected and other public officials as necessary and appropriate;
- (b) act as the lead agency and directly coordinate with such Governmental Entities;
- (c) take a position which City believes to be inconsistent with the Contract Documents, the Project Management Plan, applicable Law, Regulatory Approval(s), the requirements of Good Industry Practice, or City policy;
- (d) take a position that is not usual and customary for City to take in addressing similar circumstances affecting its own projects (except for usual and customary arrangements that are incompatible with the Project's public-private contracting methodology);
- (e) take a position that increases the risk, obligations or liabilities of City; or

(f) refrain from concurring with a position taken by a Governmental Entity if City believes that position to be proper.

**6.3.6.4** Notwithstanding any assistance provided by City in accordance with <u>Section 6.3.6.1</u>, Principal Project Company shall remain responsible for, and bear all risks and costs associated with, obtaining all Regulatory Approvals, except as otherwise expressly provided in the Contract Documents.

## 6.3.7 Regulatory Approvals in City's Name

**6.3.7.1** Certain Regulatory Approvals are required to be applied for or issued in City's name and/or require City to directly coordinate with such Governmental Entities in connection with obtaining Regulatory Approvals. With respect to such approvals, City will assist and cooperate with Principal Project Company following receipt of a request under <u>Section 6.3.6.1</u>, and the Parties shall proceed in accordance with <u>Section 6.3.6.2</u> Principal Project Company shall lead all actions and efforts to apply for and obtain the Regulatory Approval including: (a) conducting necessary field investigations, (b) preparing mitigation analyses and studies and plans, (c) preparing surveys and required reports, applications and other documents in form approved by City, and (d) joint coordination and joint discussions and attendance at meetings with the applicable Governmental Entity.

**6.3.7.2** Principal Project Company shall be solely responsible for obtaining all Regulatory Approvals required in connection with, and for compliance with applicable Laws with respect to, Temporary Areas.

## 6.3.8 Major Approval Delay

Principal Project Company shall make diligent efforts to obtain the cooperation of 6.3.8.1 each Governmental Entity as necessary for the issuance of any Major Approval. Principal Project Company is responsible for verifying the progress of each Governmental Entity's review and progress towards granting such Major Approval. Principal Project Company shall provide Notice to City within five days after the occurrence of any of the following: (a) Principal Project Company reasonably believes for any reason that any Governmental Entity will not issue the Major Approval in a manner consistent with the timely completion of the Project or in accordance with Law, the Regulatory Approvals or the Contract Documents, (b) Principal Project Company becomes aware that any Governmental Entity is not cooperating in a timely manner to issue the Major Approval in accordance with the Contract Documents, or (c) any other dispute arises between Principal Project Company and such Governmental Entity with respect to the Major Approval, despite Principal Project Company's diligent efforts to obtain such Governmental Entity's cooperation or otherwise resolve such dispute. Such Notice may include a request that City assist in resolving the dispute or in otherwise obtaining the Governmental Entity's timely cooperation. Principal Project Company shall provide City with such information as City reasonably requests regarding the Governmental Entity's failure to cooperate and the effect of any resulting delay on the Project Schedule. After delivering Notice to City, Principal Project Company shall continue to use diligent efforts to pursue the Governmental Entity's cooperation.

**6.3.8.2** If Principal Project Company requests assistance of City pursuant to <u>Section 6.3.8.1</u>, the following provisions apply:

- Principal Project Company shall provide evidence satisfactory to City, in its good (a) faith determination, that (i) the time for obtaining the Major Approval in the Project Schedule (including any Major Approval Deadline) was, in its inception, a reasonable amount of time for completion of such work, (iii) Principal Project Company has made, and continues to make, diligent efforts to obtain the Governmental Entity's cooperation and commenced coordination at the earliest time, including during the PDA Term (including submitting and actively pursuing applications for the Major Approval, taking into account the Project Schedule), (iv) Principal Project Company has submitted a fully compliant and complete request and application to the applicable Governmental Entity in accordance with applicable Law and the policies and procedures of such Governmental Entity; (v) Principal Project Company has paid in full all fees and charges required for the review, consultation and issuance of the Major Approval by the applicable Governmental Entity; (vi) Principal Project Company has satisfied all other conditions to the commencement of review by the applicable Governmental Entity in accordance with applicable Law and the policies and procedures of such Governmental Entity; and (vii) the Governmental Entity is not cooperating (clauses (a)(i) through (v) above are referred to herein as the "Major Approval Conditions to Assistance").
- (b) Following receipt by City of satisfactory evidence, City shall take such reasonable steps as City may reasonably determine to obtain the cooperation of the Governmental Entity with respect to the Major Approval or resolve the dispute; <u>provided</u>, <u>however</u>, City shall have no obligation to prosecute any legal proceedings, or to exercise any other legal remedy available to it under Law or existing contract, unless City elects to do so in its sole discretion. City may, at its sole discretion, participate in the resolution of any dispute between Principal Project Company and such Governmental Entity, whether or not requested to do so by Principal Project Company.
- (c) Without limiting City's obligations under <u>clause (b)</u> above, if City holds contractual rights that might be used to enforce the Governmental Entity's obligation to cooperate, City shall have the right not to exercise those rights. The decision not to exercise those rights shall be in the sole discretion of City.

**6.3.8.3** Any assistance provided by City shall not relieve Principal Project Company of its sole and primary responsibility for the satisfactory compliance with its obligations under the Contract Documents and its obligations with respect to obtaining any Major Approval.

**6.3.8.4** Major Approval Delay shall only apply to an unreasonable and unjustified delay by a Governmental Entity in connection with a Major Approval beyond the Major Approval Deadline following receipt by City of proper Notice pursuant to <u>Section 6.3.8.1</u>, provided that all of the "Major Approval Conditions to Assistance" described in <u>Section 6.3.8.2</u> have been satisfied.

**6.3.8.5** Notwithstanding the foregoing, the term "Major Approval Delay" does not include and is not intended to address (i) City Changes relating to a Major Approval, the impact of which will be addressed in a Change Order, (ii) any Governmental Approvals other than a Major Approval; (iii) additional work associated with a Project design change unless such design change is as a direct and sole result of a City Change, Compensable Delay Event or Unavoidable Delay Event

(excluding Major Approval Delay); (iv) any event which results from or arises out of the actions or omissions of any PPC-Related Entity or any PPC Fault; or (v) any delay or impact relating to or arising out of the failure by any PPC-Related Entity to undertake the coordination activities with the Governmental Entity contemplated by the Predevelopment Agreement or based on the results of a Reasonable Investigation (even if such delay otherwise would have been considered a Major Approval Delay but for such failure). Principal Project Company shall not rely upon any proposed schedules, durations or deadlines, if any, included in the Reference Documents with respect to a Major Approval, and Principal Project Company may not base any Claims for a time extension or additional compensation upon such proposed schedules, durations, and deadlines.

**6.3.8.6** Subject to the limitations and restrictions in this <u>Section 6.3.8</u> (Major Approval Delay) and <u>Articles 13</u> (General Provisions Applying to Delay Events and Relief Events) and <u>14</u> (Compensation and Other Relief for Delay Events and Relief Events), any Contract Deadline(s) affected by a Major Approval Delay shall be extended in accordance with, and subject to, <u>Section 14.1.3.</u>

**6.3.8.7** Principal Project Company shall not be entitled to extension of any Contract Deadline or other relief for a Major Approval Delay pursuant to <u>Section 6.3.8</u> (Major Approval Delay) or otherwise unless all of the following conditions are satisfied (in addition to satisfaction of any conditions specified in <u>Section 6.3.8</u> ((Major Approval Delay)):

- (a) Principal Project Company has timely satisfied the "Major Approval Conditions to Assistance" requirements described in <u>Section 6.3.8.2;</u>
- (b) Principal Project Company has provided evidence satisfactory to City that (i) Principal Project Company took advantage of Float available early in the Project Schedule for coordination activities with respect to the Governmental Entity to which the Major Approval relates, (ii) Principal Project Company has fulfilled its obligation to coordinate with the Governmental Entity to prevent or reduce such delays, and (iii) Principal Project Company has otherwise made diligent efforts to obtain timely performance by the Governmental Entity but has been unable to obtain such timely performance;
- (c) There exist no circumstances which have delayed or are delaying the Major Approval, other than those that fit within the definition of a Major Approval Delay; and
- (d) The delay is otherwise allowable under <u>Articles 13</u> (General Provisions Applying to Delay Events and Relief Events) and <u>14</u> (Compensation and Other Relief for Delay Events and Relief Events).

**6.3.8.8** Principal Project Company shall not be entitled to any Extra Work Costs, Delay Costs or for any other increased costs or Claims attributable to delays described in this <u>Section 6.3.8</u> (Major Approval Delay).

## 6.4 Compliance with Laws

#### 6.4.1 Applicable Laws

**6.4.1.1** Principal Project Company shall comply with, and require that all Contractors comply with, all applicable Laws, including those set forth or described in <u>Exhibit 16</u> (Federal, State and City Requirements).

**6.4.1.2** All provisions required by applicable Laws to be included in this Agreement are incorporated by reference in this Agreement.

#### 6.4.2 Environmental Compliance

**6.4.2.1** Without limiting the generality of <u>Section 6.4.1.1</u>, Principal Project Company shall comply with, and require that all Contractors comply with, all Environmental Laws.

#### 6.5 Compliance with Regulatory Approvals

Throughout the Term and the course of the Work, Principal Project Company shall:

- (a) comply with all commitments, obligations, responsibilities, conditions and requirements imposed by all Regulatory Approvals, except, with respect to Environmental Approvals only, those obligations, commitments and responsibilities of City that are expressly allocated to City or a Third Party and are expressly excluded from Principal Project Company's scope of Work;
- undertake all actions required by, or necessary to maintain in full force and effect all Regulatory Approvals to be obtained by Principal Project Company; and
- (c) comply with and implement, and cause Contractors to comply with and implement, all mitigation, monitoring, and reporting measures listed in the CEQA MMRP and the NEPA document or associated decision document, as applicable.

#### 6.6 Communication and Public Outreach

**6.6.1** Principal Project Company shall prepare and implement City's Public Outreach Plan developed pursuant to <u>Section 1.15</u> of <u>Division 1</u> of the Technical Requirements and <u>Division 9</u> of the Technical Requirements.

**6.6.2** Principal Project Company is prohibited from making any public announcement or disclosure with respect to the Project, whether for publication in the press, radio, television or any other medium, unless Principal Project Company has obtained City's prior written approval.

#### 6.7 Coordination, Cooperation and Access

**6.7.1** Principal Project Company shall coordinate and cooperate with City, Third Parties, Utility Owners, Other Contractors and Governmental Entities with jurisdiction in matters relating to the Work, including facilitating their Oversight of the Work, as applicable, including pursuant

to <u>Sections 1.12</u> and <u>1.13</u> of <u>Division 1</u> of the Technical Requirements. Principal Project Company shall coordinate and cooperate with Other Contractors that may be carrying out work within the Project Site or in the land adjoining or near the Project Site.

- 6.7.2 Principal Project Company shall provide City and City's representatives with:
  - (a) safe and unrestricted access to the Project Site and the Project at all times; and
  - (b) safe access during normal business hours to Principal Project Company's Project offices, operations buildings, and Temporary Areas.

**6.7.3** Notwithstanding anything to the contrary in this Agreement, whenever City or its representatives are present on the Project Site and production facilities, including while conducting Oversight, they will abide by the applicable Contractor's reasonable, non-discriminatory safety policies and practices and will take appropriate measures to avoid unreasonable interference with normal construction activity or normal operation and maintenance activity.

**6.7.4** Principal Project Company shall not interfere with the work of or cause any delay to any Other Contractors that may be carrying out work within the Project Site or in the land adjoining or near the Project Site and will allow them reasonable access to the Project Site, provided that Principal Project Company shall not be in breach of this <u>Section 6.7.4</u> for any temporary interruption to the work of any Other Contractors that (a) has been agreed to in advance in accordance with procedures agreed to by Principal Project Company, such contractor and any relevant Third Party; or (b) is reasonably necessary in accordance with Law and Good Industry Practice to respond to emergencies creating an immediate and serious threat to public health, safety, security or the Environment.

## 6.8 Intentionally Deleted

## 6.9 Safety Compliance

**6.9.1** City may from time to time issue Safety Compliance Orders to Principal Project Company with respect to the Project to implement Safety Compliance.

**6.9.2** Promptly upon City obtaining Actual Knowledge of any circumstance or information relating to the Project that, in City's reasonable judgment, is likely to result in a Safety Compliance Order, City will provide Notice to Principal Project Company regarding the issue. Except in the case of an Emergency, City will consult with Principal Project Company before issuing a Safety Compliance Order concerning the risk to public or worker safety, alternative compliance measures, cost impacts, and the availability of Principal Project Company resources to fund the Safety Compliance work.

**6.9.3** Where a Governmental Entity or other regulatory authority other than City directs a Safety Compliance Order, in order for Principal Project Company to comply with the Safety Compliance Order, City shall issue a corresponding Safety Compliance Order to Principal Project Company.

Infrastructure Facility DBFOM Agreement

**6.9.4** Principal Project Company shall implement each Safety Compliance Order as expeditiously as reasonably possible following its issuance. Principal Project Company shall diligently prosecute the work necessary to achieve such Safety Compliance until completion.

## 6.10 Law Enforcement and Security

## 6.10.1 Law Enforcement Services

**6.10.1.1** Principal Project Company acknowledges and agrees that law enforcement agencies, including City in its regulatory capacity, are empowered to enforce all applicable Laws and to enter the Project, and that any person engaged by City to provide law enforcement services has the authority to enter the Project, at any and all times to carry out their duties. Principal Project Company shall ensure that law enforcement agencies have necessary access to the Project to carry out their duties, power and jurisdiction.

**6.10.1.2** City shall not have any liability or obligation to Principal Project Company arising out of, relating to or resulting from the failure of law enforcement agencies to provide services, or any of their, or their respective agents' or employees', acts, omissions, negligence or misconduct in providing services, except to the extent such failure, act, omission, negligence or misconduct otherwise qualifies as a Compensable Delay Event Relief under <u>clause (i)</u> of the definition thereof or a Compensable Relief Event under <u>clause (d)</u> of the definition thereof. The general indemnity in <u>Section 10.6.1</u> (General Indemnity) shall not apply to the extent that a claim, cause of action, suit, legal or administrative proceeding or any other occurrence, loss or damage of the type listed in <u>Section 10.6.1</u> (General Indemnity) is directly attributable to actions of a law enforcement agency designated by City to provide services for the Project, and is not due to any PPC Fault.

## 6.10.2 Security

**6.10.2.1** Principal Project Company is responsible for the security of the Project and safety of the workers and public within the Project Site and the Infrastructure Facility during the performance of the D&C Work. During the IFM Period, Principal Project Company is responsible for protecting the Infrastructure Facility from damage and providing safe operation of the Infrastructure Facility.

**6.10.2.2** Principal Project Company shall comply with all rules, directives and guidance of the U.S. Department of Homeland Security and other comparable agencies, and shall coordinate and cooperate with all Governmental Entities providing security, first responder and other public emergency response services.

## 6.11 Warranties

## 6.11.1 Warranties for SFMTA O&M Facilities

**6.11.1.1** Principal Project Company warrants each SFMTA O&M Facility against Defects during the period commencing on Substantial Completion of the Project and ending two years thereafter. The general warranty contained in this <u>Section 6.11.1.1</u> is in addition to any express warranties provided for elsewhere in the Contract Documents. Principal Project Company warrants with respect to the SFMTA O&M Facilities that:

- the Construction Work shall be free of Defects, except to the extent that such Defects are inherent in prescriptive specifications included in the Contract Documents;
- (b) materials and equipment furnished by or on behalf of any PPC-Related Entity under the Contract Documents shall be of good quality and when installed, shall be new;
- (c) equipment furnished by or on behalf of any PPC-Related Entity shall be in good working condition;
- (d) the Work shall meet all of the requirements of the Contract Documents and Good Industry Practice;
- (e) the Work shall be free of Deviations that have not been approved by City; and
- (f) the Project shall be fit for use for the intended function as contemplated by the Contract Documents.

**6.11.1.2** Principal Project Company shall perform, at Principal Project Company's sole cost and expense, warranty Work for any SFMTA O&M Facilities Defect:

- (a) with respect to which City delivers written Notice to Principal Project Company within the applicable warranty period; or
- (b) of which Principal Project Company otherwise has Actual Knowledge before the expiry of the applicable warranty period.

**6.11.1.3** Principal Project Company shall commence the applicable SFMTA O&M Facility warranty Work within 14 days of written Notice of the relevant SFMTA O&M Facilities Defect from City or Principal Project Company's Actual Knowledge thereof, whichever is earlier; or such shorter period as may be designated by City for emergency repairs. Principal Project Company shall thereafter diligently complete the SFMTA O&M Facility warranty Work as soon as reasonably practicable and promptly provide Notice to City in writing of completion of same.

**6.11.1.4** If Principal Project Company fails to commence or pursue with diligence and complete the SFMTA O&M Facility warranty Work as required, City may provide written Notice of such failure to the Principal Project Company. If the Principal Project Company's failure continues for three Business Days after the City delivers this Notice, the City may, in its sole discretion, perform the SFMTA O&M Facility warranty Work, and Principal Project Company shall reimburse City within seven days after any written demand from City for Recoverable Costs incurred by City in connection with the performance of the SFMTA O&M Facility warranty Work, including any related reasonable attorneys' and consultants' fees and expenses.

**6.11.1.5** In the event of an emergency constituting an immediate hazard to health or safety of Building Occupants or City property due to an SFMTA O&M Facilities Defect, City may undertake, at Principal Project Company's sole cost and expense and without prior Notice, all work necessary to correct such hazardous condition(s).

**6.11.1.6** Before expiry of the applicable warranty period, Principal Project Company shall execute and deliver to City a written assignment, in form and substance reasonably acceptable to City, of all Principal Project Company's and Contractors' right, title and interest in and to all warranties, and to the extent assignable, claims and causes of action held by Principal Project Company or its Contractors against Third Parties, concerning the SFMTA O&M Facilities.

## 6.11.2 Contractor Warranties and Guaranties

**6.11.2.1** Principal Project Company shall obtain from all Contractors representations, warranties, guarantees and obligations in accordance with Good Industry Practice for work of similar scope and scale, with respect to design, materials, workmanship, equipment, tools and supplies furnished by all such Contractors and Suppliers, which shall extend not only to Principal Project Company but also to City and any Utility Owner or Authority Having Jurisdiction for whom Work is being performed. The warranties from Key Contractors shall be for such periods as specified in the Technical Requirements or, if not specified, a period of not less than two years from the date of the Certificate of Substantial Completion. All representations, warranties, guarantees and obligations of Key Contractors: (a) shall be written so as to survive all City and any Third Party inspections, tests and approvals; and (b) shall provide that upon expiration or any earlier termination of this Agreement before the expiration of such representations, warranties, guarantees and obligations they shall automatically be for the benefit of and enforceable by City and its successors and assigns, and any Utility Owner or Authority Having Jurisdiction for whom Work is being performed, subject to the rights of the Lenders as provided in any Direct Agreement.

**6.11.2.2** To the extent that any Contractor warranty or guaranty is voided after termination of this Agreement by reason of Principal Project Company's negligence or failure to comply with the requirements of the Contract Documents in incorporating material or equipment into the Project, Principal Project Company shall correct any Defects which would otherwise have been covered by such warranty.

**6.11.2.3** Contractor warranties are in addition to all rights and remedies available under the Contract Documents or applicable Law, and shall not limit Principal Project Company's liability or responsibility imposed by the Contract Documents or applicable Law with respect to the Work, including liability for design Defects, construction Defects, strict liability, breach, negligence, willful misconduct or fraud.

**6.11.2.4** Principal Project Company hereby assigns to City all warranties and guaranties under each Contract, as well as Principal Project Company's rights under the Contracts, effective as of the end of the Term.

## 6.11.3 Warranties for Utility Owners and Authorities Having Jurisdiction

If required by the Utility Owner or Authority Having Jurisdiction, as applicable, Principal Project Company shall provide, or obtain and ensure performance under as if Principal Project Company provided, warranties and guaranties, for all Work performed for Utility Owners and Authorities Having Jurisdiction, for two years after the date of acceptance of such work by the Utility Owner or Authority Having Jurisdiction, as applicable or such longer term as provided in any agreement with the Utility Owner or Authority Having Jurisdiction, for the benefit (with rights of enforcement) of such Utility Owner or Authority Having Jurisdiction. City shall have, and shall be identified as a third party beneficiary of the right to enforce, all such warranties and guaranties of such work. Upon acceptance of such work by the Utility Owner or Authority Having Jurisdiction, as applicable, and delivery of an assignment of the relevant warranty and guaranty rights to the Principal Project Company shall be relieved of responsibility for maintenance of such work.

## 6.12 Maintain Good Standing

**6.12.1** Principal Project Company shall remain qualified to do business in the State and remain in and maintain good standing and shall undertake all actions to do so, including continued timely submission of all required information and payments when due to the California Secretary of State, Franchise Tax Board, Internal Revenue Service, or any other applicable agency or entity. Principal Project Company shall immediately provide Notice to City if it is no longer duly qualified and in good standing.

**6.12.2** Principal Project Company shall ensure that each Key Contractor shall remain duly qualified to do business in the State and in good standing throughout the term of the Key Contract and immediately provide Notice to City if a Key Contractor is no longer duly qualified or in good standing.

## ARTICLE 7. DESIGN AND CONSTRUCTION

### 7.1 General Obligations of Principal Project Company Concerning D&C Work

- 7.1.1 Principal Project Company shall:
  - (a) expeditiously and diligently progress performance of the D&C Work to achieve Substantial Completion by the Substantial Completion Deadline;
  - (b) carry out or do all things necessary to perform the D&C Work and design and construct the Infrastructure Facility in accordance with the Contract Documents and Good Industry Practice;
  - (c) ensure the Infrastructure Facility meets the requirements of the Contract Documents, including that the Infrastructure Facility be constructed in compliance with the San Francisco Green Build Code (Chapter 7 (Green Building Requirements for City Buildings) of the San Francisco Environment Code) and achieve LEED Gold certification;
  - (d) provide maintenance and other services as described in the Contract Documents;
  - (e) ensure adequate materials, equipment and resources are available to ensure compliance with the requirements of the Contract Documents under normal conditions and reasonably anticipated abnormal conditions;
  - (f) ensure all materials and equipment are of good quality and new unless otherwise expressly stated;
  - (g) ensure, as of the Substantial Completion Date, the Construction Work shall be free of Defects, errors and omissions, except as may be set out in the Punch List (which shall be fully resolved as of Final Acceptance of the Infrastructure Facility);
  - (h) ensure, as of the Substantial Completion Date, the Design Work meets each of the requirements set out in this Agreement;
  - (i) ensure the Project Site is kept in a safe, secure, neat and clean condition at all times;
  - (j) cooperate with City and Authorities Having Jurisdiction in all matters relating to the D&C Work, including their Oversight of D&C Work;
  - (k) remove and replace Nonconforming Work and/or materials, whether discovered or rejected by City or Principal Project Company, or otherwise remedy such Nonconforming Work and/or materials in an acceptable manner and in accordance with the requirements of the Contract Documents; and

 pay all direct and indirect costs for all Utility services required to perform and complete the D&C Work in accordance with the requirements of the Contract Documents.

## 7.2 Performance, Design and Construction Standards

**7.2.1** Principal Project Company shall construct and equip the Infrastructure Facility in accordance with the Project's Release for Construction Documents, taking into account the Project Site limits and other constraints affecting the Project.

**7.2.2** The Project design and construction shall be subject to certification in accordance with the procedures contained in the approved PPC's Design Quality Plan and PPC's Construction Quality Plan.

**7.2.3** Principal Project Company shall use reasonable efforts to identify and provide Notice to City of any specifications or other provisions in the Technical Requirements that are erroneous, create a potentially Unsafe Condition, or may be inconsistent with Good Industry Practice or applicable Law. Such Notice must include a request for City approval of a Deviation or changes to the provision that Principal Project Company believes are necessary to render it correct, safe and consistent with Good Industry Practice and applicable Law. If Principal Project Company commences or continues any D&C Work affected by the change after the need for the change was known, or should have been known through the exercise of reasonable care, Principal Project Company shall bear any additional costs and time associated with redoing the D&C Work already performed.

**7.2.4** After Commercial Close, City may modify relevant provisions of the Technical Requirements to incorporate any changed, added or replaced Standards and Specifications applicable to the D&C Work by delivering a City Change to Principal Project Company.

## 7.3 Design Implementation

Principal Project Company, through appropriately qualified professional engineers and architects registered and licensed in the State and identified in Principal Project Company's Project Management Plan, shall furnish designs, plans and specifications in accordance with the Contract Documents. Principal Project Company shall cause the architect(s) of record and/or Engineer(s) of Record for the Project to sign and seal all Final Design Documents.

## 7.4 Schedule, Deadlines, Notices to Proceed and Commencement of Work

## 7.4.1 Project Schedule

**7.4.1.1** Subject to occurrence of a Delay Event resulting in an extension of time in accordance with this Agreement, if any, Principal Project Company represents and warrants that, as of the Effective Date, its Project Schedule represents a practical schedule to complete performance of the Work through Final Acceptance and is consistent with applicable deadlines.

**7.4.1.2** Between the Effective Date and the earlier of NTP2 and approval of Principal Project Company's Project Schedule in accordance with Section7.4.1.3, Principal Project Company shall perform the Work in accordance with the Initial Schedule.

**7.4.1.3** Principal Project Company shall submit to City, for City review and approval, a proposed detailed Project Schedule in accordance with <u>Section 1.2</u> of <u>Division 1</u> of the Technical Requirements and <u>Exhibit 11</u> (Submittal Review Process).

**7.4.1.4** Upon City's approval, the detailed Project Schedule becomes the Project Schedule and Principal Project Company shall perform the remaining Work in accordance with such Project Schedule.

**7.4.1.5** The Parties shall use the Project Schedule for planning and monitoring the progress of the D&C Work. The Project Schedule shall include the deadlines for Substantial Completion and Final Acceptance of the Infrastructure Facility.

**7.4.1.6** Principal Project Company shall not be limited in the sequencing or staging of the D&C Work, except to the extent that the Contract Documents or applicable Law imposes limitations.

**7.4.1.7** Principal Project Company acknowledges and agrees that the Contract Deadlines provide reasonable and adequate time to perform the Work required within the Contract Deadlines, subject only to Principal Project Company's rights to obtain time extensions under <u>Article 14</u> (Compensation and Other Relief for Delay Events and Relief Events).

## 7.4.2 Float

**7.4.2.1** All Float contained in the Project Schedule, or as generated during the course of the Work, shall be a project resource and available to both City and Principal Project Company, and shall not be considered as time for exclusive use or benefit of either City or Principal Project Company. Principal Project Company shall cause each Prime Contractor to acknowledge Float to be available to City as well as Principal Project Company as needed to absorb delay caused by Delay Events or other events, achieve interim completion dates and achieve Contract Deadlines.

**7.4.2.2** All Float shall be identified as such in the Project Schedule on each affected schedule path. City shall have the right to examine the identification of (or failure to identify) Float on the Project Schedule in determining whether to approve the Project Schedule. Once identified, Principal Project Company shall monitor, account for and maintain Float in accordance with critical path methodology.

## 7.4.3 Commencement of Non-Construction Work

Except as provided in <u>Section 7.4.6</u> (Work Before NTP 1), Principal Project Company shall not commence any Work until City has issued NTP 1 authorizing commencement of non-Construction Work. City shall promptly issue NTP 1 when all of the conditions set forth in <u>Exhibit 15A</u> (Conditions to NTP 1 - Commencement of Non-Construction Work) have been satisfied.

## 7.4.4 Commencement of Construction Work

Principal Project Company shall not commence any portion of the Construction Work until City has issued NTP 2 authorizing commencement of the applicable portion of Construction Work.

City shall promptly issue NTP 2 when all of the conditions set forth in <u>Exhibit 15B</u> (Conditions to NTP 2 - Commencement of Construction Work) have been satisfied.

**7.4.5** References in the Technical Requirements to Standards and Specifications relating to the D&C Work shall mean the most recent editions in effect as of the Setting Date, unless expressly provided otherwise.

## 7.4.6 Work Before NTP 1

**7.4.6.1** Except as provided in <u>Section 7.4.6.2</u> and as may be provided pursuant to any Early Works Agreement between Principal Project Company and City, Principal Project Company shall not perform any Work prior to NTP 1.

**7.4.6.2** Before NTP 1, Principal Project Company shall perform all Work required to achieve Financial Close, and otherwise undertake all efforts to satisfy the conditions set forth in <u>Exhibit</u> <u>15A</u> (Conditions to NTP 1 - Commencement of Non-Construction Work). If Financial Close fails to occur, City shall have no obligation to reimburse Principal Project Company for any of its costs incurred relating to this Agreement, other than payments allowed under <u>Section 3.6</u> (No-Fault Termination).

## 7.5 Acquisition of Real Property

## 7.5.1 Additional Acquisitions

**7.5.1.1** If Principal Project Company identifies any property that is not subject to <u>Section</u> <u>7.5.1.2</u> but that Principal Project Company seeks to add to the Project Site to accommodate Principal Project Company's particular design or for Principal Project Company's convenience in performing the Work, then Principal Project Company may submit to City a request for acquisition of additional property interests and related documentation as reasonably requested by City. In such event, Principal Project Company shall prepare and submit to City for review and approval new or revised surveys, legal descriptions, a preliminary title report and a copy of all encumbrances of record, draft site plats, design and other appropriate documentation of basis of acquisition and justification of acquisition. Principal Project Company's request shall include an analysis identifying alternative approaches that could be adopted to avoid the need for the acquisition, including use of retaining walls and other design modifications. Following delivery of a request under this <u>Section 7.5.1.1</u>:

- (a) City will review the request and supporting documentation and will determine whether the proposed acquisition appears to be appropriate for the Project, whether any additional information or documentation is necessary for the acquisition, and the anticipated schedule for the acquisition;
- (b) upon agreement between City and Principal Project Company regarding the acquisition of any additional property, Principal Project Company shall support the acquisition as requested by City; and
- (c) prior to acquisition of any additional property interests under this <u>Section 7.5</u> (Acquisition of Real Property), Principal Project Company shall provide to City any additional documentation required by City for the acquisition.

**7.5.1.2** If City chooses to acquire property proposed for acquisition by Principal Project Company pursuant to <u>Section 7.5.2.1</u> ("**Additional Properties**"), City will proceed with the acquisition, subject to this <u>Section 7.5</u> (Acquisition of Real Property). In all situations, City has no obligation to acquire any Additional Property.

**7.5.1.3** Principal Project Company shall be responsible for the cost to acquire Additional Properties, together with all costs and expenses incurred by City in connection with acquiring Additional Properties. In paying all such costs and expenses, Principal Project Company is not acquiring, and shall not be deemed to be acquiring, any interest in real property for Principal Project Company. Such costs and expenses shall include:

- (a) the cost of acquisition services, relocation services and associated document preparation costs;
- (b) the cost of relocation assistance in accordance with applicable Law;
- (c) the cost of condemnation proceedings, including attorneys' and expert witness fees, and all fees and expenses for production of exhibits, transcripts, photos and other documents and materials;
- (d) the acquisition price of the Additional Properties and associated appraisals, costs, settlements, offers of judgment, court awards or judgments, including pre-judgment and post-judgment interest, costs, attorney's fees, claims for loss of business goodwill and any other consideration for the Additional Properties;
- the cost of permanent or temporary acquisition of leases, easements, rights of entry, licenses and other interests in real property, including for drainage, temporary work space, Temporary Areas, and any other convenience of Principal Project Company;
- (f) the cost of permitting; and
- (g) closing costs in accordance with City policies.

**7.5.1.4** City will submit statements to Principal Project Company regarding Recoverable Costs relating to acquisition of Additional Properties, not more often than monthly. Principal Project Company shall reimburse City for Recoverable Costs within 30 days of receipt of an invoice. In addition to any other remedy, City shall have the right to curtail or suspend acquisition activities if Principal Project Company for any reason fails to pay any such invoice in full when due. City will resume acquisition activities promptly after delinquent amounts are paid in full with interest.

**7.5.1.5** Principal Project Company shall bear all risk of delays related to acquisition of Additional Properties and the risk that an Additional Property is not able to be acquired. Each Additional Property shall become part of the Project Site only upon City's written Notice to Principal Project Company indicating that the Additional Property has been acquired and that City grants Principal Project Company access to the Additional Property.

Infrastructure Facility DBFOM Agreement

**7.5.1.6** Principal Project Company shall not be entitled to any compensation or any extension of any Contract Deadline under the Contract Documents or otherwise entitled to make a Claim as a result of (i) site conditions associated with any Additional Properties (including those relating to Hazardous Materials, differing site conditions, or Utilities)

## 7.5.2 Temporary Interests in Property

**7.5.2.1** Principal Project Company shall be solely responsible for acquisition of any temporary interests in property that Principal Project Company determines is necessary, desirable or advisable to obtain in connection with the Project or the Work. Principal Project Company shall pay directly the cost to acquire, maintain, operate, and/or dispose of all such property interests. If the property is within the limits of any real property scheduled for acquisition by City or is intended to be used for permanent improvements, or if Principal Project Company intends to request City to acquire such real property, Principal Project Company shall coordinate with City and shall not negotiate with the owner(s) of such interests except with express permission of City and in compliance with applicable Law.

**7.5.2.2** City shall have no obligation to acquire temporary interests in property. Principal Project Company shall solely bear the risk of any delays and cost impacts related to acquisition of temporary interests, regardless of whether City agrees to undertake any such acquisition.

**7.5.2.3** Principal Project Company shall promptly provide Notice to City regarding all temporary interests in property that it or any of its Contractors acquires in the vicinity of the Project.

## 7.5.3 Property Acquisitions and Scheduling Work

**7.5.3.1** Principal Project Company's Project Schedule shall not provide for any Work to be done on any Additional Properties acquired or to be acquired in accordance with <u>Section 7.5.1</u> (Additional Acquisitions) until after City has granted Principal Project Company access to the Additional Property in writing.

**7.5.3.2** Concurrently with the initial review of the Project Schedule, Principal Project Company and City shall meet to discuss:

- (a) Principal Project Company's access requirements associated with planned activities and the extent to which delay in access to property identified for acquisition under <u>Section 7.5.1</u> (Additional Acquisitions) is likely to affect a Critical Path;
- (b) what efforts (if any) could reasonably be undertaken by the Parties to accelerate acquisition of any critical real property interests;
- (c) whether schedule delays may be avoided by providing access to property subject to conditions or restrictions;
- (d) whether any changes should be made to the Project Schedule; and

- (e) whether anticipated covenants, conditions and restrictions affecting access will affect Principal Project Company's ability to perform Work as scheduled, and how to mitigate any such problems.
- **7.5.3.3** Principal Project Company shall coordinate with City regarding:
  - (a) completion of Project design and identification of final site requirements and construction impacts;
  - (b) any adjustments to the Project Schedule necessary to reflect City's expected timeline to acquire any property identified for acquisition under <u>Section 7.5.1</u> (Additional Acquisitions); and
  - (c) any design features that may impact properties for which no property acquisition is contemplated, with the goal of avoiding damages to properties not previously identified and addressed.

# 7.6 Utilities, Utility Adjustments and Utility Delay

# 7.6.1 Utilities to Serve Facility

**7.6.1.1** Principal Project Company shall be solely responsible for obtaining and implementing all new Utility services needed for the Infrastructure Facility, and must timely submit the appropriate applications to obtain electricity, water, sewer, cable, etc. This work expressly includes obtaining electricity and power from PG&E for the Project, including all applications and coordination required to obtain such Utility services.

**7.6.1.2** City shall, from time to time as required, enter into contracts with Energy suppliers for the supply of Energy to the Infrastructure Facility and shall be responsible for all payments due pursuant to such supply contracts. For avoidance of doubt, the Infrastructure Facility's electricity, natural gas, water, and other such Utility bills will be sent to City and City will be responsible for paying such utility bills. Principal Project Company will be provided with copies of the invoices.

# 7.6.2 Principal Project Company's General Responsibilities

**7.6.2.1** Principal Project Company shall be responsible for all Utility Adjustments required by the Project.

**7.6.2.2** City agrees to cooperate as reasonably requested by Principal Project Company in advancing Utility Adjustments, including attendance at negotiation sessions and review with Utility Owners. Principal Project Company shall timely keep City informed of the status of any such negotiations.

**7.6.2.3** Principal Project Company shall bear the sole risk, cost and schedule impact of Utility Adjustments other than, subject to <u>Articles 13</u> (General Provisions Applying to Delay Events and Relief Events) and <u>14</u> (Compensation and Other Relief for Delay Events and Relief Events), as a direct result of a City Change or Relief Event.

# 7.6.3 Avoiding Utility Adjustments and Minimizing Costs

**7.6.3.1** Principal Project Company shall minimize Utility Adjustments in the design and construction of the Infrastructure Facility and costs for which Principal Project Company may be entitled to additional compensation pursuant to the Contract Documents, including by taking into account existing Utilities (and their location) in its design and working around existing Utilities (including existing PG&E poles at or adjacent to the Project Site), including through its construction sequencing, means and methods. If a Utility Adjustment could have been avoided through Principal Project Company undertaking a Reasonable Investigation prior to the Setting Date, Principal Project Company shall bear the full risk and liability associated with such Utility Adjustment, irrespective of any other provisions in this Agreement that might provide compensation, time or other relief in connection with such Utility Adjustment.

**7.6.3.2** Principal Project Company shall reimburse City for any costs City incurs as a result of Principal Project Company's failure to comply with the requirements of this <u>Section 7.6.3</u> (Avoiding Utility Adjustments and Minimizing Costs).

# 7.6.4 Allocation of Work Responsibility

**7.6.4.1** The allocation of responsibility for performing Utility Adjustment design, construction, and/or materials procurement for each Utility Adjustment as between Principal Project Company and the Utility Owner shall be determined in accordance with <u>Section 1.13</u> of <u>Division 1</u> of the Technical Requirements (including pursuant to the Utility Coordination Work Plan and Utility Project Execution Plan). For purposes of this <u>Section 7.6</u> (Utilities, Utility Adjustments and Utility Delay) and <u>Section 1.13</u> of <u>Division 1</u> of the Technical Requirements, references to responsibility for performing Utility Adjustment design and construction include all tasks customarily associated therewith; <u>provided</u>, <u>however</u>, that Principal Project Company shall be responsible for all coordination with Utility Owners that is necessary in order to accomplish the Utility Adjustments in compliance with the requirements of the Contract Documents.

**7.6.4.2** Principal Project Company is responsible for scheduling all Utility Adjustments so as allow for the issuance of NTP 2 as soon as practicable and to meet all applicable Contract Deadlines, without regard to whether a Utility Adjustment is performed by Principal Project Company or by the affected Utility Owner (or its contractors). Accordingly, under no circumstances shall any reallocation of responsibility for Utility Work between Principal Project Company and a Utility Owner be considered grounds for a time extension hereunder.

**7.6.4.3** No increase or decrease in compensation payable to Principal Project Company shall be made pursuant to this <u>Section 7.6.4</u> (Allocation of Work Responsibility) on account of any change in the allocation of responsibility for Incidental Utility Work, or any other matter for which the Contract Documents specify how liability, cost or risk is to be allocated between City and Principal Project Company.

# 7.6.5 Utility Adjustment Costs

**7.6.5.1** Principal Project Company is responsible for all costs of the Utility Work, including costs of acquiring Utility easements and costs with respect to relinquishment or acquisition of Utility Owner property interests, but excluding (a) costs attributable to Utility Betterments, for which the Utility Owner is responsible and which are not paid to Utility Owner by City; and (b)

any other costs for which the Utility Owner is responsible under Law which are not paid to Utility Owner by City. Principal Project Company shall fulfill this responsibility either by performing the Utility Work at its own cost, if permitted by the Utility Owner, or by reimbursing the Utility Owner for the Utility Owner's performance of the Utility Work, in accordance with <u>Section 7.6.5.4</u>. Principal Project Company shall bear the costs due to the Utility Owner and all costs and expenses associated therewith, including the costs of Utility Owner inspections and any overtime charges incurred by the Utility Owner. Principal Project Company is solely responsible for collecting directly from the Utility Owner any amount due to Principal Project Company for Utility Betterment costs or other costs incurred by Principal Project Company for which the Utility Owner is responsible, whether under Law or otherwise.

**7.6.5.2** If for any reason Principal Project Company is unable to collect any amounts due to Principal Project Company from any Utility Owner, then as between City and Principal Project Company, (a) City shall have no liability for such amounts, (b) Principal Project Company shall have no right to collect such amounts from City or to offset such amounts against amounts otherwise owing from Principal Project Company to City, and (c) Principal Project Company shall have no right to stop Work, sue for *mandamus*, demand or plead in any court for City's participation in resolution of any dispute with the Utility Owner, or seek to exercise any other remedies against City on account of the Utility Owner's failure to pay.

**7.6.5.3** Principal Project Company shall maintain a complete set of records for the costs of Utility Work performed (whether incurred by Principal Project Company or by the Utility Owner), in a format reasonably satisfactory to City and in sufficient detail for analysis. For both Utility Owner costs and Principal Project Company costs, the totals for each cost category shall be shown in such manner as to permit comparison with the categories stated on the estimate. Principal Project Company. All records with respect to Utility Work shall comply with the record keeping and audit requirements of the Contract Documents.

**7.6.5.4** To the extent that City is required to initially pay a Utility Owner for any costs related to Utility Owner's performance of Utility Work under this <u>Section 7.6.5</u> (Utility Adjustment Costs), Principal Project Company, on a monthly basis, shall reimburse City for any costs incurred in connection with such Utility Work. The amounts reimbursed shall be due and payable within 10 days after receipt of City's invoice for such Utility Work. City, in its sole discretion, may also deduct the amount to be reimbursed from any payment due and payable to Principal Project Company should Principal Project Company fail to reimburse City for such Utility Work.

# 7.6.6 Incidental Utility Work

Notwithstanding any contrary provision of the Contract Documents, Principal Project Company shall be responsible for all Incidental Utility Work without regard to the allocation of Work responsibility otherwise established pursuant to this <u>Section 7.6</u> (Utilities, Utility Adjustments and Utility Delay). Without limiting the provision of <u>Section 7.6.2.3</u>, Principal Project Company also shall be responsible for furnishing all designs for Incidental Utility Work that Principal Project Company performs, unless such designs are included among that portion of the Utility Work to be performed by the Utility Owner. Principal Project Company shall not be entitled to any adjustment in its compensation or Contract Deadlines on account of costs incurred, cost savings or delays associated with the performance of Incidental Utility Work by Principal Project Company or by any Utility Owner.

# 7.6.7 Bonds and Insurance; Security for Utility Adjustment Costs

**7.6.7.1** All Utility Work shall automatically be covered by the Payment Bonds and Performance Bonds described in <u>Section 10.2</u> (Performance Security) and by the insurance described in <u>Section 10.1</u> (Insurance).

**7.6.7.2** Principal Project Company shall satisfy all requirements of Utility Owners to provide security for reimbursement of Utility Adjustment costs to which the Utility Owner is entitled and that are the responsibility of Principal Project Company hereunder.

#### 7.6.8 Utility Investigation by Principal Project Company

7.6.8.1 During the PDA Term, Principal Project Company has undertaken an investigation of Utilities affecting the Project Site, including Utilities located on, in, under, around or impacting the Project Site and the Project. Principal Project Company represents and warrants that it has analyzed Reference Documents, diligently commenced coordination concerning Utility Adjustments (including submitting and actively pursuing relocation applications to Utility Owners with respect to impacted Utilities, taking into account the Project Schedule), contacted and made inquiries of Utility Owners, performed substantial surface inspections of the Project Site and undertaken such additional inspections, including substantial subsurface utility investigations, as it deems appropriate and as are required by the Contract Documents and in keeping with Good Industry Practice to verify, fully and accurately identify all Utilities (including contacting and accessing Non-Dig Alert Utilities), addressing all field conditions, and validating the Utility Information. Principal Project Company acknowledges that its rights to any compensation, time or other relief in connection with Utilities, Unidentified Utilities, Utility Adjustments and Utility Delay is conditioned on the accuracy of the foregoing representations and warranties.

**7.6.8.2** Except as otherwise provided in <u>Sections 7.6.9</u> (Claims for Inaccuracies in Utility Information) and <u>7.6.13</u> (Utility Delays), the Parties specifically intend by <u>Section 7.6</u> (Utilities, Utility Adjustments and Utility Delay) to delegate to Principal Project Company the obligation to perform all responsibilities with respect to identification of all Utilities and to allocate to Principal Project Company all risk of increased costs and time of the Utility Work resulting from inaccuracies in the reputed locations of such facilities (and in any other relevant information with respect to such facilities).

# 7.6.9 Claims for Inaccuracies in Utility Information

**7.6.9.1** If during performance of Construction Work, Principal Project Company encounters any Non-Dig Alert Utilities located at the Project Site that requires a Utility Adjustment, and such Utility is (i) an Unidentified Utility; (ii) not a Service Line; and (iii) not identified or reflected in whole or in part in the Utility Information, then, subject to the provisions of <u>Articles 13</u> (General Provisions Applying to Delay Events and Relief Events) and <u>14</u> (Compensation and Other Relief for Delay Events and Relief Events), Principal Project Company shall be entitled to a Change Order to compensate Principal Project Company for any material increase in Principal Project Company's actual direct costs of performing the Utility Work that is directly attributable to undertaking the Utility Adjustment of the Non-Dig Alert Utility. The amount of such Change Order shall be determined in accordance with <u>Article 14</u> (Compensation and Other Relief for Delay Events and Relief Events).

**7.6.9.2** Principal Project Company shall not be entitled to any compensation pursuant to this <u>Section 7.6.9</u> (Claims for Inaccuracies in Utility Information) or otherwise for any of the following:

- (a) Increased costs of the Work attributable to Unidentified Utilities and Non-Dig Alert Utilities, to the extent that the existence of the facility was known to Principal Project Company as of the Setting Date or could have been inferred from a Reasonable Investigation or otherwise from the presence of other facilities, such as buildings, meters, junction boxes, manholes or identifying markers, visible during a surface inspection of the area conducted prior to the Effective Date;
- (b) Increased costs of the Work attributable to Unidentified Utilities and Non-Dig Alert Utilities where Principal Project Company failed to provide timely Notice in accordance with <u>Articles 13</u> (General Provisions Applying to Delay Events and Relief Events) and <u>14</u> (Compensation and Other Relief for Delay Events and Relief Events);
- (c) Increased costs of the Work attributable to Unidentified Utilities and Non-Dig Alert Utilities that can be Protected in Place or removed rather than physically relocated;
- (d) Any additional costs incurred by Utility Owners (that are not reimbursable or payable to the Utility Owner) as a result of the Unidentified Utility;
- (e) Increased costs of the Utility Work attributable to all other Utilities that are not Unidentified Utilities or Non-Dig Alert Utilities; and
- (f) Delay and disruption damages, except as specifically set forth in the Contract Documents.

# 7.6.10 Changes in Design

**7.6.10.1** For purposes of this <u>Section 7.6.10</u> (Changes in Design), a Project design change impacting Utility Adjustments is a change that (a) requires a Utility Adjustment that was not listed in the Utility Information; or (b) necessitates acquisition of a Utility Easement not included in the real property rights comprising the Project Site.

**7.6.10.2** Inasmuch as Principal Project Company is both furnishing the design of and constructing the Project, Principal Project Company may have opportunities to reduce the costs of certain portions of the Work, which may increase the costs of certain other portions of the Work or of Utility Work to be performed by Utility Owners. In considering such opportunities, Principal Project Company shall consider the impact of Project design changes on Utility Adjustments with the overall goal of minimizing the necessity for Utility Adjustments to the extent practical, in compliance with <u>Section 7.6.3</u> (Avoiding Utility Adjustments and Minimizing Costs). Accordingly, except for cost increases or decreases resulting from City Changes in Project design affecting Utility Work, and notwithstanding any other contrary provision of the Contract Documents, the following rules shall apply with respect to Project design changes during the course of the Project which either reduce the nature or extent of or eliminate any Utility Adjustment, or result in unanticipated Utility Adjustments or an increase in the nature, extent, or costs of anticipated Utility Adjustments:

- (a) Principal Project Company shall not be entitled to extension of any Contract Deadline on account of delays resulting from any such design changes (including delays in acquisition of Utility Easements by City or Utility Owners);
- (b) Principal Project Company shall not be entitled to any increase in compensation for any additional costs which Principal Project Company incurs as a result of such design changes (including additional costs of Utility Work, the costs of any additional Work on other aspects of the Project undertaken in order to facilitate the avoidance or minimization of Utility Adjustments, and/or increased costs resulting from any Project Site conditions associated with Utility Easements made necessary by such design changes); and
- (c) If City incurs any additional costs as a result of such design changes (including any increases in amounts owed by City to Utility Owners, e.g., for work which is unusable or which must be redone), then Principal Project Company shall reimburse City for such costs within 10 days after receipt of City's invoice therefor, or in the sole discretion of City, City may deduct the amount of reimbursement due from any payment due and payable to Principal Project Company under the Contract Documents should Principal Project Company fail to reimburse City for such costs.

This <u>Section 7.6.10</u> (Changes in Design) shall not apply to any changes in design made to accommodate any Change in Law.

# 7.6.11 Utility Betterments

Principal Project Company shall be responsible for addressing any requests by Utility Owners that Principal Project Company design and/or construct a Utility Betterment. Notwithstanding any other provision of this <u>Section 7.6</u> (Utilities, Utility Adjustments and Utility Delay), no Work, Utility Adjustments or other Utility-related items initially included in the Work shall be considered a Utility Betterment. Any Utility Betterment performed as part of a Utility Adjustment, whether by Principal Project Company or by the Utility Owner, shall be subject to the same standards and requirements as if it were a necessary Utility Adjustment. Under no circumstances shall Principal Project Company proceed with any Utility Betterment that is incompatible with the Project or is not in compliance with applicable Law, the Regulatory Approvals or the Contract Documents, including the Contract Deadlines. Under no circumstances will Principal Project Company be entitled to any additional compensation or time extension under the Contract Documents as the result of any Utility Betterment, whether performed by Principal Project Company or by the Utility Betterment, whether performed by Principal Project Company or by the Utility Betterment, whether performed by Principal Project Company or by the Utility Betterment, whether performed by Principal Project Company or by the Utility Betterment, whether performed by Principal Project Company or by the Utility Owner. Principal Project Company shall provide City with such information, analyses, and certificates as City may request in order to determine compliance with this <u>Section 7.6.11</u> (Utility Betterments).

# 7.6.12 Failure of Utility Owners to Cooperate

**7.6.12.1** Principal Project Company shall make diligent efforts to obtain the cooperation of each Utility Owner as necessary for the Project. Principal Project Company is responsible for verifying the progress of each Utility Owner's work. Principal Project Company shall provide Notice to City within five days after the occurrence of any of the following: (a) Principal Project Company reasonably believes for any reason that any Utility Owner would not undertake or

permit Utility Work in a manner consistent with the timely completion of the Project or in accordance with Law, the Regulatory Approvals or the Contract Documents, (b) Principal Project Company becomes aware that any Utility Owner is not cooperating in a timely manner to provide agreed-upon work or approvals in accordance with the Contract Documents, or (c) any other dispute arises between Principal Project Company and a Utility Owner with respect to the Project, despite Principal Project Company's diligent efforts to obtain such Utility Owner's cooperation or otherwise resolve such dispute. Such Notice may include a request that City assist in resolving the dispute or in otherwise obtaining the Utility Owner's timely cooperation. Principal Project Company shall provide City with such information as City reasonably requests regarding the Utility Owner's failure to cooperate and the effect of any resulting delay on the Project Schedule. After delivering Notice to City, Principal Project Company shall continue to use diligent efforts to pursue the Utility Owner's cooperation.

**7.6.12.2** If Principal Project Company requests assistance of City pursuant to <u>Section 7.6.12.1</u>, the following provisions apply:

- (a) Principal Project Company shall provide evidence satisfactory to City, in its good faith determination, that (i) the subject Utility Work is necessary, (ii) the time for completion of the Utility Adjustment in the Project Schedule (including any Major Utility Adjustment Deadline) was, in its inception, a reasonable amount of time for completion of such work, (iii) Principal Project Company has made, and continues to make, diligent efforts to obtain the Utility Owner's cooperation and commenced coordination at the earliest time, including during the PDA Term (including submitting and actively pursuing Utility Adjustment applications to Utility Owners with respect to impacted Utilities, taking into account the Project Schedule), (iv) the representation and warranty set forth in Section 7.6.8.1 was true and accurate as of the Effective Date; and (v) the Utility Owner is not cooperating (clauses (a)(i) through (v) above are referred to herein as the "Conditions to Assistance").
- (b) Following receipt by City of satisfactory evidence, City shall take such reasonable steps as City may reasonably determine to obtain the cooperation of the Utility Owner or resolve the dispute; <u>provided</u>, <u>however</u>, City shall have no obligation to prosecute eminent domain or other legal proceedings, or to exercise any other legal remedy available to it under Law or existing contract, unless City elects to do so in its sole discretion. City may, at its sole discretion, participate in the resolution of any dispute between Principal Project Company and a Utility Owner, whether or not requested to do so by Principal Project Company.
- (c) Without limiting City's obligations under <u>clause (b)</u> above, if City holds contractual rights that might be used to enforce the Utility Owner's obligation to cooperate, City shall have the right not to exercise those rights. The decision not to exercise those rights shall be in the sole discretion of City.

**7.6.12.3** Any assistance provided by City shall not relieve Principal Project Company of its sole and primary responsibility for the satisfactory compliance with its obligations under the Contract Documents and its obligations with respect to timely completion of all necessary Utility Adjustments.

Infrastructure Facility DBFOM Agreement

# 7.6.13 Utility Delays

- **7.6.13.1** Except as set forth in <u>clause (b)</u>, below, the term "Utility Delay" shall mean:
  - (a) Any unreasonable and unjustified delay by a Utility Owner in connection with a Utility Adjustment relating to an Unidentified Utility beyond the date set forth in the approved Project Schedule (including any update to the Project Schedule submitted and approved following identification of the Unidentified Utility) following receipt by City of proper Notice pursuant to <u>Section 7.6.12.1</u>, provided that all of the "Conditions to Assistance" described in <u>Section 7.6.12.2</u> have been satisfied.
  - (b) Any Major Utility Adjustment Delay following receipt by City of proper Notice pursuant to <u>Section 7.6.12.1</u>, provided that all of the "Conditions to Assistance" described in <u>Section 7.6.12.2</u> have been satisfied.
  - (c) Notwithstanding the foregoing, the term "Utility Delay" does not include and is not intended to address (i) City Changes relating to Utilities, the impact of which will be addressed in the Change Order, (ii) any Utility Adjustments other than (x) Major Utility Adjustments beyond the Major Utility Adjustment Deadline or (y) as a direct result of a City Change, Unidentified Utility or Unavoidable Delay Event (excluding Utility Delay); (iii) additional work associated with a Project design change as described in Section 7.6.10 (Changes in Design); (iv) any event described in this Section 7.6.13.1 which results from or arises out of the actions or omissions of any PPC-Related Entity or any PPC Fault; or (v) any delay or impact relating to or arising out of the failure by any PPC-Related Entity to undertake the coordination activities with Utilities contemplated by the Predevelopment Agreement or based on the results of a Reasonable Investigation (even if such delay otherwise would have been considered Utility Delay but for such failure). Principal Project Company shall not rely upon any proposed schedules, durations or deadlines, if any, included in the Reference Documents with respect to Utility Adjustments, and Principal Project Company may not base any Claims for a time extension or additional compensation upon such proposed schedules, durations, and deadlines. IAs noted in the definitions of Maior Utility Adjustment/Maior Utility Adjustment Deadline/Major Utility Adjustment Delay, City will consider deletion of (v) if and when PNC has engaged with utilities and developed a list of Major Utility Adjustments and durations which City has approved]

**7.6.13.2** Subject to the limitations and restrictions in this <u>Section 7.6.13</u> (Utility Delays) and <u>Articles 13</u> (General Provisions Applying to Delay Events and Relief Events) and <u>14</u> (Compensation and Other Relief for Delay Events and Relief Events), any Contract Deadline(s) affected by a Utility Delay shall be extended in accordance with, and subject to, <u>Section 14.1.3</u>. Furthermore, if two Utility Delays occur that are concurrent with each other but are not concurrent with any other delay, then the period of concurrent delay shall be considered a Utility Delay but shall only be counted once for purposes of any time extension.

**7.6.13.3** Principal Project Company shall not be entitled to extension of any Contract Deadline or other relief for a Utility Delay pursuant to <u>Section 7.6.13.2</u> or otherwise unless all of the following conditions are satisfied (in addition to satisfaction of any conditions specified in <u>Section 7.16.13.1</u>):

- (a) The Utility Adjustment is solely associated with an Unidentified Utility or a Major Utility Adjustment Delay;
- (b) Principal Project Company has timely satisfied the "Conditions to Assistance" requirements described in <u>Section 7.6.12.2;</u>
- (c) If applicable, Principal Project Company has provided a reasonable Utility Adjustment plan to the Utility Owner that has been approved by City;
- (d) Principal Project Company has provided evidence satisfactory to City that (i) Principal Project Company took advantage of Float available early in the Project Schedule for coordination activities with respect to the Utility(ies) to which such Utility Delay relates, (ii) Principal Project Company has fulfilled its obligation to coordinate with the Utility Owner to prevent or reduce such delays, and (iii) Principal Project Company has otherwise made diligent efforts to obtain timely performance by the Utility Owner but has been unable to obtain such timely performance;
- (e) There exist no circumstances which have delayed or are delaying the affected Utility Adjustment(s), other than those that fit within the definition of a Utility Delay; and
- (f) The delay is otherwise allowable under <u>Articles 13</u> (General Provisions Applying to Delay Events and Relief Events) and <u>14</u> (Compensation and Other Relief for Delay Events and Relief Events).

**7.6.13.4** Principal Project Company shall not be entitled to any Extra Work Costs, Delay Costs or for any other increased costs or Claims attributable to delays described in this <u>Section 7.6.13</u> (Utility Delays).

# 7.6.14 Utility-Related Claims; Additional Restrictions on Change Orders Relating to Utility Adjustments

In addition to all of the other requirements and limitations contained in this <u>Section 7.6</u> (Utilities, Utility Adjustments and Utility Delay), and <u>Articles 13</u> (General Provisions Applying to Delay Events and Relief Events) and <u>14</u> (Compensation and Other Relief for Delay Events and Relief Events), Principal Project Company's entitlement to any relief relating to Utility Adjustments shall be subject to the restrictions and limitations set forth in this <u>Section 7.6.14</u> (Utility-Related Claims; Additional Restrictions on Change Orders Relating to Utility Adjustments).

**7.6.14.1** Principal Project Company shall provide documentation satisfactory to City showing that the required analysis was performed and an appropriate determination made regarding the need for the Utility Adjustment, and shall also bear the burden of proving that the amount of any additional costs and/or time incurred by Principal Project Company are both necessary and reasonable.

**7.6.14.2** To the extent compensation is permitted under this <u>Section 7.6</u> (Utilities, Utility Adjustments and Utility Delay), any relief increasing compensation to Principal Project Company pursuant to this <u>Section 7.6</u> (Utilities, Utility Adjustments and Utility Delay) shall include only the incremental costs arising from the circumstances giving rise to such relief, i.e., the amount

payable shall take into account the costs that would have been incurred absent such circumstances and a credit shall be allowed for any avoided costs.

**7.6.14.3** Principal Project Company shall not be entitled to any increase in compensation for any costs of coordinating with Utility Owners.

### 7.6.15 FTA Requirements

The Project is funded in part with funds made available by the Federal Transit Administration. In carrying out the Utility Adjustments, Principal project Company shall comply with all Laws as necessary for Utility Adjustments to be eligible for reimbursement from such funding.

#### 7.7 Hazardous Materials Management; Risk Allocation

#### 7.7.1 Hazardous Materials Management

**7.7.1.1** Except as otherwise provided in this <u>Section 7.7.1</u> (Hazardous Materials Management), Principal Project Company shall, as part of the Work, perform, or cause to be performed, all Hazardous Materials Management required in connection with the Project in accordance with applicable Law, Regulatory Approvals, the approved Environmental Protection Program, and all applicable provisions of the Contract Documents.

**7.7.1.2** Principal Project Company shall have the following duties to identify, avoid, minimize and mitigate adverse monetary and non-monetary impacts to the Project and to City relating to Hazardous Materials:

- (a) Principal Project Company shall adopt design and construction techniques for the Project, using Good Industry Practice, that avoid, to the extent reasonably practicable, the need for Hazardous Materials Management;
- (b) when performing Hazardous Materials Management, Principal Project Company shall use Good Industry Practice, including design modifications and construction techniques, to minimize costs (including long-term costs) of Hazardous Materials Management; and
- (c) Principal Project Company shall use appropriately trained personnel to conduct Hazardous Materials Management activities.

**7.7.1.3** Principal Project Company shall promptly provide Notice to City of any Hazardous Materials encountered in connection with the Project, the Project Site or the Work that require (a) reporting or Notice to any Governmental Entity and/or (b) taking any response action (e.g., evaluating and addressing the circumstances and location of the Hazardous Materials) under applicable Law, Regulatory Approvals, the approved Environmental Protection Program and the Contract Documents, as applicable. A Notice provided under this <u>Section 7.7.1.3</u> shall advise City of any obligation to provide Notice to Governmental Entities under applicable Law. Principal Project Company shall make all such reports, or deliver all such Notices, to any Governmental Entity with respect to Hazardous Materials encountered in connection with the Project, the Project Site or the Work, providing concurrent Notice and copies of such reports and Notices to City.

# 7.7.1.4 Procedures and Compensation for Hazardous Materials Management

- (a) Principal Project Company shall manage, treat, handle, store, remediate, remove, transport (where applicable), document and dispose of all Hazardous Materials and perform all other aspects of Hazardous Materials Management as appropriate, in accordance with applicable Law, Regulatory Approvals, the approved plans required to be provided under <u>Section 01 35 44</u> of <u>Division 10</u> of the Technical Requirements, and this Agreement.
- (b) Principal Project Company is prohibited from starting Hazardous Material removal Work without a City-approved Hazardous Materials Submittal as described in <u>Section 02 80 13</u> of <u>Division 10</u> of the Technical Requirements. Principal Project Company shall not conduct any sampling or analysis of suspected building materials without prior permission from the City. The time utilized and costs incurred by Principal Project Company to undertake and obtain City approval of such Hazardous Materials Submittals shall not be eligible for relief, time extension or compensation under this Agreement.
- (c) Except where Principal Project Company is required to take immediate action under the Contract Documents or applicable Law, Principal Project Company shall afford City the opportunity to inspect sites containing Hazardous Materials and to consult with Principal Project Company about the recommended approach before any Hazardous Materials Management or other action is taken which would inhibit City's ability to ascertain the nature and extent of the contamination.
- (d) Where management, treatment, handling, storage, remediation, transport, documentation or disposing of Hazardous Materials constitutes a Hazardous Materials Event, <u>Section 14.1.5</u> (Additional Limits Relating to Hazardous Materials Event During D&C Period) and <u>Section 14.2.5</u> (Additional Limits Relating to Hazardous Materials Event During IFM Period) shall apply regarding Principal Project Company's rights to potential compensation and extension of any Contract Deadline.

**7.7.1.5** If Principal Project Company fails to undertake the Hazardous Materials Management required under this <u>Section 7.7.1</u> (Hazardous Materials Management) within a reasonable time after discovery of Hazardous Materials, taking into consideration the nature and extent of the contamination and action required and the potential impact upon Principal Project Company's schedule for use of and operations on the Project Site, City may provide Notice to Principal Project Company that City will undertake the Hazardous Materials Management. If Principal Project Company fails to remedy the failure within 3 Business Days following provision of a Notice under this <u>Section 7.7.1.5</u>:

- (a) City may itself undertake Hazardous Materials Management actions or procure a contractor to perform such work, in which case City will do so in accordance with all applicable Environmental Laws;
- (b) Principal Project Company shall reimburse City on a current basis, within 10 days of request, for Recoverable Costs from fines, penalties or other

Infrastructure Facility DBFOM Agreement

assessments against City or the Project by Governmental Entities due to Principal Project Company's delay or failure to undertake the Hazardous Materials Management), so long as City has performed in accordance with <u>Section 7.7.1.5(a)</u>; and

(c) City shall have no liability or responsibility to Principal Project Company arising out of, relating to or resulting from City's Hazardous Materials Management actions and such actions shall not constitute a Relief Event or other basis for a Claim.

**7.7.1.6** In the event of an emergency constituting an immediate hazard to health or safety of Building Occupants or City property due to Principal Project Company's failure to undertake the Hazardous Materials Management required under this <u>Section 7.7.1</u> (Hazardous Materials Management), City may undertake, at Principal Project Company's sole cost and expense and without prior Notice, all work necessary to correct such hazardous condition(s).

# 7.7.2 Additional Hazardous Materials Obligations of Principal Project Company

**7.7.2.1** Principal Project Company shall avoid exacerbating Hazardous Materials (including Pre-Existing Hazardous Materials as well as new Releases) in, on, under or migrating from the Project Site. For purposes of determining liability, as between City and Principal Project Company, under <u>Sections 7.7.1.4(a)</u> and <u>7.7.1.4(c)</u>, Principal Project Company shall only be liable for exacerbation of Hazardous Materials arising out of or relating to PPC Fault.

**7.7.2.2** Principal Project Company shall take all reasonable efforts to ensure that no act or omission of any PPC-Related Entity will result in an unlawful Release of Hazardous Materials to or into wastewater, storm or sanitary sewer systems, surface water, air, soils or groundwater in, on, under or migrating from the Project Site.

# 7.7.3 Hazardous Materials Generator Responsibilities

**7.7.3.1** As between Principal Project Company and City, City shall be considered the generator and assume generator responsibility for Hazardous Materials, other than any PPC Release, and City shall not assert that Principal Project Company has legal responsibility for such Hazardous Materials (other than a PPC Release) as a generator. City has approval rights of Principal Project Company's selection of the destination facility to which Hazardous Materials, other than any PPC Release, will be transported.

**7.7.3.2** With regard to Hazardous Materials other than any PPC Release, City shall comply with the applicable standards for generators including those found at 40 CFR Part 262, including the responsibility to sign manifests and other waste tracking records for the transport of Hazardous Materials.

**7.7.3.3** Sections 7.7.3.1 and 7.7.3.2 shall not preclude or limit any rights, remedies or defenses that City or Principal Project Company may have against any Governmental Entity or other third parties, including prior owners, lessees, and licensees nor shall Sections 7.7.3.1 and 7.7.3.2 be interpreted as an admission of City's liability as to any Governmental Entity or other third parties, or otherwise preclude City from asserting to any Governmental Entity or other third parties that another entity, other than Principal Project Company or City, is the generator of any

Hazardous Materials. Notwithstanding the foregoing, Principal Project Company (and not City) shall be considered the generator with respect to any PPC Release.

### 7.8 Substantial Completion

**7.8.1.1** The City and Principal Project Company mutually agree that time is of the essence with respect to the dates and times specified in this Agreement.

**7.8.1.2** Principal Project Company shall achieve Substantial Completion on or before the Substantial Completion Deadline. Failure to achieve Substantial Completion by the Long Stop Date is a PPC Default under <u>Section 16.1.1(c)</u>.

**7.8.1.3** The conditions to Substantial Completion are set forth in <u>Exhibit 15C</u> (Conditions to Substantial Completion).

**7.8.1.4** Approximately six calendar months before the date on which Principal Project Company expects to achieve Substantial Completion, Principal Project Company shall provide to City its anticipated schedule to achieve Substantial Completion. Principal Project Company shall promptly advise City if at any time Principal Project Company determines that its anticipated schedule to achieve Substantial Completion will change. Principal Project Company shall provide an updated schedule to City 21 days before the date Principal Project Company expects to achieve Substantial Completion. Principal Project Company expects to achieve Substantial Completion. Principal Project Company at a minimum, dates when Principal Project Company will submit all remaining documentation and evidence required by Exhibit 15C (Conditions to Substantial Completion) and the Technical Requirements with respect to Substantial Completion.

**7.8.1.5** When Principal Project Company determines that it has satisfied all conditions to Substantial Completion, it shall deliver to City a request for Certificate of Substantial Completion, in a form reasonably acceptable to City, stating the date that Principal Project Company determined it satisfied all conditions in <u>Exhibit 15C</u> (Conditions to Substantial Completion). Together with the request for Certificate of Substantial Completion, Principal Project Company shall submit a Punch List of items to be completed as a condition precedent to achievement Final Acceptance, obtain approval from Utility Owners and Authorities Having Jurisdiction of any Punch List items affecting Utilities and elements subject to the jurisdiction of an Authority Having Jurisdiction and obtain City's acceptance of the Punch List. The Punch List shall not include any items that adversely affect the safety, use or operability of the Project.

**7.8.1.6** During the 14-day period following the receipt of such request for Certificate of Substantial Completion:

- (a) Principal Project Company and City shall meet and confer to facilitate City's determination of whether Principal Project Company has met the conditions for Substantial Completion; and
- (b) City will conduct an inspection of the Project, review the applicable Final Design Documents, Construction Documents and other Submittals and conduct such other investigations as may be necessary to evaluate whether Substantial Completion has been achieved.

**7.8.1.7** As soon as reasonably practicable and, in any event, no later than 21 days after receipt of Principal Project Company's Notice under <u>Section 7.8.1.5</u>, City's Authorized Representative shall inspect, in conjunction with Principal Project Company, the D&C Work and City's Authorized Representative shall either:

- (a) if City's Authorized Representative agrees that Substantial Completion has been achieved, issue the Certificate of Substantial Completion to Principal Project Company within 10 days after the inspection:
  - (i) certifying that Substantial Completion has taken place;
  - (ii) stating the Substantial Completion Date;
  - (iii) listing any Punch List items which, in its opinion, remains to be performed; and
  - (iv) setting out reasonable details of the work remaining to be performed to achieve Final Acceptance; or
- (b) if Substantial Completion has, in City's opinion, not been achieved, issue a Notice to Principal Project Company within 10 days after the inspection:
  - (i) listing the work which, in its opinion, remains to be performed to achieve Substantial Completion; or
  - (ii) stating in its opinion, that Substantial Completion is so far from being achieved that it is not practicable to provide a list of the type referred to in <u>Section 7.8.1.7(b)(i)</u>.

**7.8.1.8** Principal Project Company shall give Notice to City's Authorized Representative when Principal Project Company considers that the work listed in a Notice issued by City's Authorized Representative under <u>Section 7.8.1.7(b)(i)</u> has been completed.

**7.8.1.9** Sections 7.8.1.4 through 7.8.1.8 will apply in respect of any new Principal Project Company Notice under Section 7.8.1.8 in the same way as if it were the original Notice given under Section 7.8.1.4.

**7.8.1.10** City's Authorized Representative's opinion as to whether Substantial Completion has been achieved will not be restricted by any Notice, list or opinion which City previously provided to Principal Project Company under <u>Section 7.8.1.7(b)(i)</u>.

**7.8.1.11** In the event that the Principal Project Company and City disagree as to whether or not Substantial Completion has been achieved, the Substantial Completion Date, the list of work remaining to be performed or the list of or estimated cost to perform Punch List items, City and Principal Project Company will meet to resolve such dispute, failing which such dispute shall be resolved in accordance with <u>Error! Reference source not found.</u> (Contract Dispute Procedures).

**7.8.1.12** Notwithstanding any other provision of this Agreement, City is under no obligation to certify Substantial Completion or commence the payment of Availability Payments prior to the

original Substantial Completion Deadline, regardless of whether Substantial Completion has been achieved prior to the original Substantial Completion Deadline.

# 7.9 Final Acceptance

**7.9.1.1** Promptly after achieving Substantial Completion, Principal Project Company shall perform all remaining D&C Work, including expeditiously and diligently correcting all of the Punch List items specified pursuant to <u>Section 7.8.1.7(a)(iii)</u>, to achieve Final Acceptance by the Final Acceptance Deadline.

**7.9.1.2** The conditions to Final Acceptance are set forth in <u>Exhibit 15D</u> (Conditions to Final Acceptance).

**7.9.1.3** When Principal Project Company determines that it has satisfied all conditions to Final Acceptance, it shall provide a certification to City, in a form reasonably acceptable to City, stating the date that Principal Project Company determined that it satisfied all the conditions in <u>Exhibit 15D</u> (Conditions to Final Acceptance).

**7.9.1.4** Within 21 days after receipt of Principal Project Company's Notice under <u>Section</u> <u>7.9.1.3</u>, City's Authorized Representative shall inspect, in conjunction with Principal Project Company, the D&C Work and shall either:

- (a) if City's Authorized Representative agrees that Final Acceptance has been achieved, issue a Certificate of Final Acceptance to Principal Project Company within 15 days after the inspection certifying that Final Acceptance has taken place and the Final Acceptance Date; or
- (b) if in City's Authorized Representative's opinion, Final Acceptance has not been achieved, issue a Notice to Principal Project Company within 15 days after the inspection listing the work it believes remains to be performed to achieve Final Acceptance.

**7.9.1.5** Without limiting Principal Project Company's other obligations under this Agreement, immediately upon receipt of a Notice under <u>Section 7.9.1.4(b)</u>, Principal Project Company shall expeditiously and diligently progress performance of the work specified in the Notice.

**7.9.1.6** Principal Project Company shall give Notice to City's Authorized Representative when Principal Project Company considers that the work listed in City's Authorized Representative's Notice under <u>Section 7.9.1.4(b)</u> has been completed.

**7.9.1.7** Sections 7.9.1.4 through 7.9.1.6 will apply in respect of any new Principal Project Company Notice under Section 7.9.1.6 in the same way as if it were the original Notice under Section 7.9.1.4

**7.9.1.8** City's Authorized Representative's opinion as to whether Final Acceptance has been achieved will not be restricted by any:

(a) Certificate of Substantial Completion, Notice, list or opinion already provided under this Agreement; or

(b) warranty provided by any PPC-Related Entity.

**7.9.1.9** In the event that City and Principal Project Company disagree as to whether or not Final Acceptance has been achieved or the Final Acceptance Date, City and Principal Project Company will meet to resolve such dispute, failing which such dispute shall be resolved in accordance with <u>Error! Reference source not found.</u> (Contract Dispute Procedures).

# 7.10 Responsibility for Loss or Damage

**7.10.1** The D&C Work includes having full charge and care of the Project Site and the D&C Work (including bearing risk of loss and damage to the D&C Work and Project Site) through Substantial Completion (or in the case of any Punch List Items, Final Acceptance), except to the extent that City or Third Parties have accepted elements of the D&C Work and expressly assumed responsibility for maintenance of such elements.

**7.10.2** Principal Project Company shall take every reasonable precaution against Loss or damage to any part of the Project from any cause, whether arising from the performance or nonperformance of the D&C Work.

**7.10.3** For so long as Principal Project Company bears the risk of loss and damage to D&C Work under <u>Section 7.10.1</u>, Principal Project Company shall repair, restore and replace Losses or damages to such D&C Work occasioned by any cause, subject to potential relief in accordance with, and subject to, <u>clause (r)</u> of the definition of "Compensable Delay Event". Principal Project Company shall ensure that such work is performed in accordance with the Contract Documents and applicable Law.

**7.10.4** Principal Project Company shall repair, restore and replace any Losses or damages to City property other than the D&C Work caused by any PPC-Related Entity.

# 7.11 Nonconforming Work

# 7.11.1 Obligation to Replace Nonconforming Work

Principal Project Company shall perform all Work in conformity with the Contract Documents. If Principal Project Company has not performed the Work in conformity with the Contract Documents, then, in addition to any other remedies available to City, City may direct Principal Project Company to, and Principal Project Company shall, remove and replace or otherwise remedy the Nonconforming Work. Principal Project Company shall not be entitled to make a Claim in connection with such D&C Work except as it relates to a dispute regarding whether Principal Project Company had performed the D&C Work in conformity with the Contract Documents.

# 7.11.2 Remedial Plan for Nonconforming Work

**7.11.2.1** Promptly after Nonconforming Work is identified but no later than 10 Business Days after the earlier of (i) Notice from City to Principal Project Company; and (ii) Principal Project Company first obtains Actual Knowledge of such Nonconforming Work, Principal Project Company shall submit a remedial plan to City, for review and approval, describing the error or Defect giving rise to the Nonconforming Work and describing Principal Project Company's planned remedial action. Such proposal shall address Infrastructure Facility integrity,

aesthetics, operational impact on City including MOT, impact on the public, Building Occupants and any applicable Governmental Entities, maintainability, the effect on the Project Schedule and other relevant issues.

**7.11.2.2** If City determines that a proposed plan of correction may infringe upon Infrastructure Facility integrity, operations or maintainability, then City may elect to perform a technical assessment of Principal Project Company's proposal. City shall provide Notice to Principal Project Company promptly upon determining that an assessment is required, and shall take reasonable efforts to expedite the assessment. Should City elect to perform any such technical assessment, (a) if so requested by City, Principal Project Company shall not proceed with the remedial plan until City has conducted its technical assessment and provided prior approval of the remedial plan and (b) Principal Project Company shall not be entitled to make any Claim in connection with the technical assessment or reasonable delay in the remedial plan pending City's approval.

# 7.11.3 City's Remedies

**7.11.3.1** City shall have the right and authority to cause Nonconforming Work to be removed, replaced or otherwise remedied and to withhold or deduct the costs from any monies due or that become due to Principal Project Company under the Contract Documents upon (a) any failure of Principal Project Company to provide a proposed remedial plan as described in <u>Section</u> <u>7.11.2.1</u> and obtain City's approval thereof, or (b) any failure of Principal Project Company to comply with City's direction under this Agreement relating to any safety issue, including Safety Compliance Orders.

**7.11.3.2** In addition to the right to cause the removal, replacement or remedy of the Nonconforming Work, at the request of Principal Project Company or upon its failure to remove, replace or otherwise remedy the Nonconforming Work within the timelines set forth in this Agreement, City may, in its sole discretion, accept such Nonconforming Work and receive reimbursement in an amount equal to the greater of: (a) the amount deemed appropriate by City (acting in good faith) to provide compensation for future impacts, maintenance and/or other costs relating to the Nonconforming Work, or (b) 100% of Principal Project Company's cost savings associated with its failure to perform the Work in accordance with the requirements of the Contract Documents.

# 7.12 Design and Construction of Joint Development Alternatives

Nothing contained in this Agreement shall obligate City to design and construct any Joint Development Alternative or utilize Principal Project Company or any PPC-Related Entity in connection with the design and construction of any Joint Development Alternative.

# 7.13 FF&E During D&C Period

#### 7.13.1 Selection and Procurement of PPC-Furnished FF&E

- 7.13.1.1 Principal Project Company shall:
  - (a) update the Equipment List from time to time as appropriate to:
    - (i) reflect decisions made during the design development process;

- (ii) ensure that like PPC-Furnished FF&E are grouped together;
- (iii) address any comments provided by the City's Authorized Representative with respect to PPC-Furnished FF&E in accordance with Exhibit 11 (Submittals Review Process); and
- (iv) ensure that Principal Project Company provides PPC-Furnished FF&E as necessary to satisfy the requirements of this Agreement;
- (b) defer final selection of those items of PPC-Furnished FF&E which have high technical obsolescence risk to a time as close as reasonably possible to Substantial Completion, to better ensure that Principal Project Company will provide the most technically up to date items of such PPC-Furnished FF&E as of Substantial Completion; and
- (c) procure and provide the PPC-Furnished FF&E, as it may be amended in accordance with this <u>Section 7.13.1</u> (Selection and Procurement of PPC-Furnished FF&E) or as a result of a Change Order.

**7.13.1.2** Unless necessary to meet its obligations under <u>Section 7.13.1(a)(iv)</u> or to carry out a Change Order, Principal Project Company shall not decrease the quantity or provide PPC-Furnished FF&E of a lesser quality (which may be determined based on attributes such as specification, brand, and place of manufacture) than an item identified in the Equipment List as of the Effective Date.

### 7.13.2 Selection of City-Furnished Equipment and Existing FF&E

**7.13.2.1** City will select City-Furnished Equipment and identify Existing FF&E to be included in the Infrastructure Facility and update the Equipment List when reasonably required by Principal Project Company prior to Substantial Completion.

**7.13.2.2** Principal Project Company shall defer as long as reasonably possible the time for City to select the City-Furnished Equipment and identify the Existing FF&E.

**7.13.2.3** City will procure the City-Furnished Equipment identified in the Equipment List, as it may be amended as a result of a Change Order.

#### 7.13.3 Installation of PPC-Furnished FF&E

Principal Project Company shall install or locate (as applicable depending on whether the FF&E is loose or fixed) all items of PPC-Furnished FF&E within the Project Site in accordance with the Release for Construction Documents and so that the Infrastructure Facility meets the requirements of the Principal Project Documents. If the Release for Construction Documents do not identify locations for the placement of loose items of PPC-Furnished FF&E, Principal Project Company shall locate such items in accordance with direction from City's Authorized Representative.

# 7.13.4 Installation of City-Furnished IT/Comms FF&E

**7.13.4.1** Principal Project Company must provide City or its authorized representative continued and uninterrupted access to the IT/Comms Site for no less than 14 consecutive days prior to the commencement of commissioning and testing of the Infrastructure Facility ("**City Access Period**") for the City to install the City-Furnished IT/Comms FF&E. The access provided shall be sufficient for City to complete such installation and transition and shall comply with the conditions in Section 1.11.7 of Division 1 of the Technical Requirements.

**7.13.4.2** Provided City complies with Project Site safety standards, Principal Project Company shall have no entitlement to make any claim against City for any delay to the Work caused by any access to or work undertaken by City under <u>Section 7.13.5.1</u>. Principal Project Company shall be entitled to relief in accordance with, and subject to, <u>clause (n)</u> of the definition of "Compensable Delay Event" in connection with City's access or work undertaken by City under <u>Section 7.13.5.1</u>.

**7.13.4.3** Subject to <u>Section 7.13.4.4</u>, City shall install City-IT/Comms FF&E to the IT/Comms Site during the City Access Period.

**7.13.4.4** City will release Principal Project Company from all claims and liability in connection with the transfer of and installation of the City-IT/Comms FF&, except to the extent that Principal Project Company or a PPC-Related Entity (i) damages such FF&E or (ii) fails to comply with its obligations under this Agreement concerning such FF&E.

#### 7.13.5 Commissioning of FF&E for Substantial Completion

**7.13.5.1** Following installation of the PPC-Furnished FF&E and the City-Furnished IT/Comms FF&E, Principal Project Company shall commission and undertake all tests of the Infrastructure Facility, in accordance with the requirements of this Agreement and <u>Division 6</u> of the Technical Requirements to achieve the requirements of Substantial Completion.

**7.13.5.2** City will release Principal Project Company from all claims and liability in connection with the City-Furnished IT/Comms FF&E, to the extent that Principal Project Company is unable to successfully commission or undertake tests of such FF&E in accordance with the requirements of this Agreement and <u>Division 06</u> of the Technical Requirements, except to the extent that the failure of such FF&E is caused by an act or omission of Principal Project Company or a PPC-Related Entity.

#### 7.14 Move-In

# 7.14.1 General

As provided in <u>Section B.25</u> (Move-In Services) of the IFM Specifications, Principal Project Company shall plan, coordinate, manage and execute the physical Move-In of the Infrastructure Facility, including Existing FF&E and City Furnished Equipment (excluding City-Furnished IT/Comms FF&E), following Substantial Completion in accordance with the Move-In Plan and in coordination with the Move-In Subcommittee ("**Move-In**").

7.13.4.4 City will release Principal Project Company from all claims and liability in connection with the transfer of and installation of the City Furnished Equipment and Existing

Infrastructure Facility DBFOM Agreement

FF&E, except to the extent that Principal Project Company or a PPC-Related Entity (i) damages such FF&E or (ii) fails to comply with its obligations under this Agreement concerning such FF&E.

# 7.14.2 Move-In Subcommittee

**7.14.2.1** The move-in subcommittee (the "Move-In Subcommittee") shall consist of three representatives of each Party. Members of the Move-In Subcommittee may invite, on prior notice to all members, such other advisors and consultants as they require from time to time to attend meetings and provide briefings to the Move-In Subcommittee.

**7.14.2.2** The Move-In Subcommittee shall assist the Parties by promoting cooperative and effective communication with respect to matters related to the Move-In, including issues related to the move-in into the Infrastructure Facility, the City Access Periods and the transfer and installation of all Existing FF&E.

**7.14.2.3** The primary role of the Move-In Subcommittee shall be to oversee and coordinate the Move-In in a timely and efficient manner and in accordance with the Project Schedule.

**7.14.2.4** The Move-In Subcommittee shall be responsible for receiving and reviewing all matters related to the Move-In.

**7.14.2.5** The members of the Move-In Subcommittee may adopt such procedures and practices for the conduct of the activities of the Move-In Subcommittee as they consider appropriate from time to time.

**7.14.2.6** Unless otherwise agreed, the Move-In Subcommittee shall operate only until the Final Acceptance Date.

# 7.14.3 Move-In Resource

**7.14.3.1** Principal Project Company shall, at least 24 months prior to Substantial Completion, prepare and submit to City a list of prospective candidates, including from existing resources, (each a "Move-In Resource Candidate") for appointment as Move-In Resource, each of which must have experience planning and executing staff and logistical transitions in projects of similar size, scope and complexity in the United States.

**7.14.3.2** Subject to <u>Section 24.17(c)</u>, Principal Project Company shall conduct a competitive bid process for the selection of the Move-In Resource from among the Move-In Resource Candidates, which bid process shall be completed by no later than 18 months prior to the Scheduled Substantial Completion Date. Principal Project Company shall consult with City in the design and implementation of such competitive bid process, including the development of the evaluation criteria, and shall accommodate any reasonable request of City with respect thereto. City may participate in the evaluation and selection of the successful Move-In Resource Candidate for appointment as Move-In Resource. City may, but is not required to, retain the Move-In Resource at City's cost to plan, coordinate, manage and execute the transition from the Existing Facilities to the Infrastructure Facility.

# ARTICLE 8. INFRASTRUCTURE FACILITY MAINTENANCE

#### 8.1 General

#### 8.1.1 General Obligations

**8.1.1.1** Principal Project Company is responsible for performance of IFM Services in accordance with requirements specified in the Contract Documents, including the IFM Management Plan approved by City during the PDA Term.

- 8.1.1.2 Principal Project Company shall ensure that:
  - (a) all IFM Services is performed in accordance with all applicable Laws, Regulatory Approvals and Good Industry Practice, as it may evolve over time;
  - (b) the Infrastructure Facility remains safe, fit for use for the intended functions, meets the requirements of the Contract Documents, remains free of Defects and meets the minimum performance standards for operations as specified in the Contract Documents, including the Technical Requirements, throughout the IFM Period;
  - (c) (i) all materials and equipment furnished during the IFM Period are of good quality and new and (ii) all Infrastructure Facility components and related consumables obtained as part of the IFM Services and supplied during the IFM Period are of good quality and new and fit for their intended purpose in accordance with the Contract Documents; and
  - (d) all IFM Services is performed in accordance with the IFM Plan and Cityapproved plans required in the IFM Specifications.

**8.1.1.3** Principal Project Company shall monitor compliance of the IFM Services with the requirements of Contract Documents and provide Notice to City if any noncompliance occurs.

**8.1.1.4** Principal Project Company shall cooperate with City (and Authorities Having Jurisdiction as applicable) in all matters relating to the IFM Services and required availability of the Infrastructure Facility, including any Oversight with respect to operation and maintenance of the Project.

**8.1.1.5** Principal Project Company shall obtain, maintain, repair and replace elements of the Infrastructure Facility, except SFMTA O&M Facilities, as appropriate throughout the duration of the IFM Period, including maintenance, repair and replacement of consumable and life-expired items and rehabilitation of the Infrastructure Facility, except SFMTA O&M Facilities.

**8.1.1.6** Principal Project Company shall, in keeping with Good Industry Practice, implement or incorporate Technology Enhancements for the Infrastructure Facility throughout the duration of the IFM Period, at no cost to City, to the extent such enhancements are (a) scheduled in the IFM Management Plan, or (b) needed to correct Defects in the Work. To the extent that the City directs the Principal Project Company to implement or incorporate Technology Enhancements in circumstances other than those set forth above, such direction shall be considered a City Change.

**8.1.1.7** City shall pay all direct and indirect costs for all Utility services relating to the operation and maintenance, throughout the Term, of the Infrastructure Facility in accordance with the Contract Documents.

# 8.1.2 Changes in Standards and Specifications

**8.1.2.1** The Parties anticipate that, from time to time, the Standards and Specifications will be changed, added or replaced, which may be implemented through revisions to existing standards and specifications or through new standards and specifications.

**8.1.2.2** Any Renewal Work required to be performed by Principal Project Company shall be performed in compliance with the then current Standards and Specifications. City may, through a City Change, require Principal Project Company to comply with updated Standards and Specifications prior to the date set forth in Principal Project Company's Renewal Work Schedule for Renewal Work affected by the updated Standards and Specifications.

# 8.1.3 Hazardous Materials Management

Principal Project Company shall comply with <u>Section 7.7</u> (Hazardous Materials Management; Risk Allocation) throughout the IFM Period.

#### 8.1.4 Utility Accommodation

**8.1.4.1** Principal Project Company acknowledges that from time to time during the course of the IFM Period, Utility Owners will apply for additional utility permits to install new Utilities that may affect the Project Site, or to modify, repair, upgrade, relocate or expand existing Utilities within the Project Site.

**8.1.4.2** Throughout the IFM Period, Principal Project Company shall monitor Utilities and Utility Owners within the Project Site for compliance with applicable Utility permits, easements, and applicable Law, and shall use diligent efforts to obtain the cooperation of each Utility Owner having Utilities within the Project Site. Principal Project Company shall promptly provide Notice to City if (a) Principal Project Company believes that any Utility Owner is not complying with the terms of a Utility permit, easement, or applicable Law affecting a Utility within the Project Site, or (b) any other dispute arises between Principal Project Company and a Utility Owner with respect to a Utility within the Project Site, despite Principal Project Company having exercised its diligent efforts to obtain the Utility Owner's cooperation. If Principal Project Company, despite diligent efforts, is unable to resolve any dispute with a Utility Owner, Principal Project Company may request City to provide reasonable assistance. Following delivery of such a request the Parties shall consult regarding measures to be undertaken.

# 8.2 Principal Project Company Inspection, Testing and Reporting

**8.2.1** Principal Project Company shall carry out inspections and tests in accordance with the Technical Requirements and IFM Management Plan. Principal Project Company shall use the results of such inspections and tests to develop and update the IFM Management Plan, to maintain asset condition and service levels, and to develop programs of maintenance and Renewal Work to minimize the effect of IFM Services on City employees, residents, and other members of the public.

**8.2.2** Principal Project Company shall submit all reports relating to the IFM Services, including the IFM annual reports, in the form, with the content and within the time required under the Contract Documents.

**8.2.3** The inspections and reports described above are in addition to maintenance of the Failure Event and Noncompliance Event database and related reports under <u>Section 15.3.1</u> (Noncompliance Database).

# 8.3 Responsibility for Loss or Damage

**8.3.1** The IFM Services includes having full charge and care of the Infrastructure Facility, except for the SFTMA O&M Facilities, from the date of Substantial Completion of the Infrastructure Facility through the end of the IFM Period.

**8.3.2** Principal Project Company shall take reasonable precautions against Loss or damage to the Infrastructure Facility and other improvements and assets within the Project Site due to any cause.

**8.3.3** Principal Project Company shall repair, restore and replace all Losses or damages to the Infrastructure Facility, excluding the SFMTA O&M Facilities. Principal Project Company shall ensure that such work is performed in accordance with the Contract Documents and applicable Law.

**8.3.4** If any repair, restoration or replacement of the Infrastructure Facility building envelope or exterior grounds is required due to Vandalism, Principal Project Company shall be responsible for the first \$150,000 of costs incurred in the aggregate per calendar year (the "**Annual Vandalism Deductible**") to perform such repair, restoration or replacement. The Annual Vandalism Deductible shall be index-linked and increased by the Escalation Factor calculated in accordance with <u>Section 2</u> of <u>Exhibit 4B</u> (Availability Payment Mechanism).

**8.3.5** For each calendar year of the Term, City will compensate Principal Project Company through a Change Order for costs incurred by Principal Project Company in performing repairs, restoration or replacements due to Vandalism that exceed the Annual Vandalism Deductible, if there are any such costs. If any repair, restoration or replacement of the Infrastructure Facility building interior, excluding any SFMTA O&M Facilities, is required due to Vandalism, such repair, restoration or replacement will be treated as a City Change.

**8.3.6** Principal Project Company shall repair, restore and replace any Losses or damages to City property other than the Infrastructure Facility caused by any PPC-Related Entity.

# 8.4 Renewal Work; IFM Management Plan

# 8.4.1 Performance of Renewal Work

**8.4.1.1** Principal Project Company shall diligently perform Renewal Work as and when necessary to comply with the Contract Documents, including the Handback Requirements. Principal Project Company shall use the IFM Management Plan developed in accordance with the Technical Requirements as the principal guide for scheduling and performing Renewal Work; except that Principal Project Company may perform Renewal Work not identified in IFM Management Plan as necessary to maintain compliance with the Contract Documents, subject

to scheduling the performance of such Renewal Work at times agreed to by City, in its reasonable discretion.

**8.4.1.2** If, at any time, Principal Project Company has failed to perform the Renewal Work in accordance with the then current IFM Management Plan and applicable requirements of the Contract Documents (including the Technical Requirements, including <u>Section E</u> of the IFM Specifications), and performance of such Renewal Work is required to ensure continued performance of the Project in accordance with the Contract Documents, City may give written Notice thereof to Principal Project Company. If Principal Project Company has failed to complete the Renewal Work within 30 days after City delivers such Notice, then City shall have the right to perform and complete such Renewal Work at the expense and for the account of Principal Project Company, and to deduct from payments otherwise payable to Principal Project Company by City under this Agreement to pay the costs of such action. The foregoing remedy is in addition to any other remedies available to City under the Contract Documents on account of such failure, including the assessment of Noncompliance Points, and City's right to intervene immediately and without Notice to address Safety Compliance.

# 8.4.2 IFM Management Plan

As a condition to Substantial Completion pursuant to <u>Exhibit 15C</u> (Conditions to Substantial Completion), Principal Project Company shall prepare and submit to City for review and approval an IFM Management Plan updated in accordance with the IFM Specifications. Coordination of Operations and Maintenance Responsibilities

At Principal Project Company's request from time to time, City will, at no cost to City, reasonably assist Principal Project Company in seeking the cooperation and coordination of any Authority Having Jurisdiction with respect to Principal Project Company's IFM Services. The objectives of such assistance will be to minimize disruptions to transit operations, City employees, residents, and traffic and ensure that all IFM activities are carried out in accordance with then-current maintenance standards and then-current traffic management standards, practices and procedures of such Authorities Having Jurisdiction.

# 8.5 Handback

# 8.5.1 Handback Condition

**8.5.1.1** Principal Project Company shall diligently perform and complete all Renewal Work required to be performed and completed before the Termination Date, based on the required adjustments and changes to the IFM Management Plan resulting from the inspections and analysis under this <u>Section 8.5</u>.

**8.5.1.2** Upon the Termination Date, Principal Project Company shall surrender the Project, including any Upgrades, to City, in the condition and meeting all of the requirements of the Handback Requirements.

**8.5.1.3** In the event of early termination of this Agreement, Principal Project Company shall only be required to comply with the requirements of this <u>Section 8.5</u> (Handback) to the extent that any Renewal Work was scheduled to have been performed before the Early Termination Date.

#### 8.5.2 Independent Engineer

**8.5.2.1** The Parties must select an Independent Engineer to:

- (a) perform joint technical reviews and handback inspections in accordance with <u>Section 8.5.4;</u>
- (b) make a determination regarding the estimated Handback Work costs in accordance with <u>Section 8.5.4.2(h)</u>; and
- (c) verify in accordance with <u>Section 8.5.5</u> (Handback Assessment) whether the Handback Requirements have been met.

**8.5.2.2** Prior to each Review Date, Principal Project Company must, in consultation with City, establish the schedule and process for the selection of an Independent Engineer for purposes of the Independent Engineer roles contemplated in <u>Section 8.5.2.1</u>. Principal Project Company must develop the solicitation package and draft contract terms, subject to approval by City, and Principal Project Company must set the solicitation schedule with the goal of selecting the Independent Engineer 6 months prior to each Review Date.

**8.5.2.3** The solicitations required under <u>Section 8.6.2</u> (Independent Engineer) must include issuance of a request for competitive proposals from a list of firms approved by City, review of proposals by the Parties, a joint determination regarding which firm is the best qualified to provide Independent Engineer services, and negotiation of a fair and reasonable price for performance of such services. If negotiations fail with the highest ranked firm, the Parties may elect to terminate negotiations and proceed with the next highest ranked firm. This process must be followed until a firm is selected to serve as the Independent Engineer. If the Parties fail to reach agreement regarding selection of the Independent Engineer, or regarding acceptable terms of the agreement with the Independent Engineer, the Claim must be subject to resolution under <u>Article 18</u> (Contract Dispute Procedures).

**8.5.2.4** The Independent Engineer will be appointed jointly by the Parties and will act independently from and not as an agent of either Party. Any advisor of City, a PPC-Related Entity, or a Lender is deemed to have an organizational conflict of interest and not eligible to respond to the solicitation.

**8.5.2.5** Each Party is responsible for its own costs related to the Independent Engineer solicitation processes. The Parties will be equally responsible for all amounts payable under the terms of the agreement with the Independent Engineer.

#### 8.5.3 Handback Reserve Amount

**8.5.3.1** Within 30 days of the commencement of the 26<sup>th</sup> Contract Year, and annually thereafter to the end of the IFM Period, Principal Project Company shall calculate the Handback Reserve Amount and prepare and deliver to City a report that:

(a) specifies Principal Project Company's estimate of the Handback Reserve Amount based on the inspections and the updated Handback Renewal Work Plan provided for under <u>Section 8.5.4.2(h)8.5.4.2(h)</u>; and

Infrastructure Facility DBFOM Agreement

(b) details how Principal Project Company's estimated Handback Reserve Amount was calculated.

**8.5.3.2** Within 15 Business Days of receipt of each report described in <u>Section 8.5.3.1</u>, City will review the report and provide a Notice to Principal Project Company setting forth its determination of the Handback Reserve Amount, including supporting materials.

**8.5.3.3** Following delivery of each update and report, the Parties shall meet to discuss whether any changes should be made to the scope or schedule for performance of the Handback Renewal Work or to the Handback Reserve Amount.

# 8.5.4 Joint Technical Review and Handback Inspections and Report

**8.5.4.1** The Parties and the Independent Engineer shall schedule and conduct joint technical inspections of the Infrastructure Facility:

- (a) within 90 days following the Substantial Completion Date;
- (b) each 5 year anniversary after the Substantial Completion Date; and
- (c) the 25th anniversary of the Effective Date of this Agreement ("Handback Inspection"),

each of the above constitutes a "Review Date".

**8.5.4.2** Within 30 days following each Review Date, the Independent Engineer must produce and deliver to City a report (the "**Condition Report**") for review and approval in accordance with <u>Exhibit 11</u> (Submittal Review Process). The Condition Report shall:

- (a) identify the condition of the Infrastructure Facility and the Project Site and each element for which Principal Project Company is responsible under the Agreement;
- (b) compare the condition of each element with the current Infrastructure Facility standard requirements in the IFM Specifications;
- (c) compare the actual Renewal Work replacement with the approved Handback Renewal Work Plan;
- (d) identify and documents deficiencies in the current condition or Renewal Work replacement activities and makes recommendations on actions required to bring the condition and Renewal Work replacement activities to the current required standard;
- (e) provides the FCI index as of the time of the Joint Technical Review;
- (f) determines and verifies the condition of all elements and their Residual Lives;
- (g) adjusts, to the extent necessary based on inspection and analysis, element Useful Lives, ages, Residual Lives; and

(h) in the case of the Handback Inspection, (a) determines the Handback Renewal Work required to be performed and completed before the Termination Date, based on the Handback Requirements for Residual Life at the conclusion of the Term, the foregoing adjustments and the foregoing changes to the IFM Management Plan; and (b) includes an estimate and details of the costs to perform the Handback Work and the Renewal Work ("Handback Reserve Amount").

**8.5.4.3** Principal Project Company must update the IFM Management Plan following each inspection of the Infrastructure Facility in accordance with <u>Section 8.5.4.2</u> and deliver the updated plan to City for City's review and approval in accordance with <u>Exhibit 11</u> (Submittal Review Process). Such update shall include a plan to rectify identified deficiencies, schedule of activities and any process or administrative changes and maintenance and Renewal Work activities. All planned activities related to updated IFM Management Plan must be completed within a maximum of 12 months from receipt of the applicable Condition Report.

**8.5.4.4** The Parties shall undertake a subsequent inspection of the remediation work completed by Principal Project Company in connection with the updated IFM Management Plan and, if required, issue a revised Condition Report.

# 8.5.5 Handback Assessment

**8.5.5.1** In the case of the Handback Inspection, when Principal Project Company determines that it has satisfied the Handback Requirements, it shall provide a certification to City and Independent Engineer, in a form reasonably acceptable to City, stating the date that Principal Project Company determined that it satisfied the Handback Requirements.

**8.5.5.2** During the 30-day period following receipt of such certification:

- (a) Principal Project Company, Independent Engineer, and City shall meet and confer to facilitate City's determination of whether Principal Project Company has met the Handback Requirements;
- (b) Independent Engineer will conduct a final inspection of the Project, review the Handback Renewal Work Plan and any other relevant Submittals and conduct such other investigations as may be necessary to evaluate whether the Handback Requirements have been met; and
- (c) Independent Engineer shall (i) deliver a report of findings and determination to Principal Project Company and City stating in Independent Engineer's opinion whether Principal Project Company has met the Handback Requirements; or (ii) provide Notice to Principal Project Company stating the reasons why the Handback Requirements have not been met.

**8.5.5.3** Within 10 days after City's receipt of a report given by Independent Engineer under <u>Section 8.6.5.2(c)</u> stating that the Handback Requirements have been met, City will either:

(a) provide Notice to Principal Project Company that it concurs with the Independent Engineer's determination that the Handback Requirements have been met; or (b) provide Notice to Principal Project Company regarding the reasons why City believes the Handback Requirements have not been met.

**8.5.5.4** If Independent Engineer provides Notice under <u>Section 8.5.5.2(c)(ii)</u>, then Principal Project Company shall take appropriate steps to satisfy the remaining conditions and provide a certification under <u>Section 8.5.5</u> once the conditions have been satisfied and otherwise repeat the process described in this Section 8.5.5 until City provides Principal Project Company the notice described in Section 8.5.3(a).

**8.5.5.5** The findings of the Independent Engineer shall be inadmissible and given no weight during the <u>Article 18</u> (Contract Dispute Procedures) process.

# 8.5.6 City Right to Self-Perform and Recover Costs

If City determines that the Project does not comply with any Handback Requirement, or Handback Renewal Work is not timely or properly performed, City may provide Principal Project Company Notice of such failure(s). Principal Project Company shall have 15 days from the date the City delivers this Notice to rectify such failure(s) to City's satisfaction; provided, however, that if (i) the nature of such rectification is such that the rectification cannot with diligence be completed within such 15 day period; (ii) such failure does not relate to a matter of public or facility safety or an Emergency; and (iii) Principal Project Company has commenced and is diligently pursuing meaningful steps to rectify immediately after receiving the Notice of such failure(s), Principal Project Company shall have such additional period of time, up to a maximum period of 60 days from the date City delivers the Notice, as is reasonably necessary to diligently effect such rectification. If Principal Project Company fails to rectify such failure(s) within this period, then, in addition to City's rights under the Contract Documents, City may bring the Project into compliance with such Handback Requirement(s) and Principal Project Company shall be liable for City's Recoverable Costs incurred in performing that Work. In recovering such amounts, City may (a) reduce any Availability Payment then due and owing from City to Principal Project Company, (b) invoice Principal Project Company for such amount, as a lump sum payment, (c) set off such amount against any other amount then due and owing from City to Principal Project Company, (d) draw against funds withheld under this Section 8.5 (Handback) or against the letter of credit described in Section 8.6.5 (Handback Requirements Letters of Credit), (e) require funds in the reserve account described in Section 8.5.7 (Handback Requirements Reserve Account) to be used to pay for required Renewal Work, or (f) any combination of clauses (a) through (e).

# 8.5.7 Spare Parts

Principal Project Company must comply with the requirements with respect to Spare Parts as set out in <u>Section F.1.2(2)</u> of the IFM Specifications.

# 8.6 Handback Requirements Reserve Account

# 8.6.1 Establishment

**8.6.1.1** Within 30 days of the commencement of the 26<sup>th</sup> Contract Year, Principal Project Company shall establish a reserve account (the "Handback Requirements Reserve Account") exclusively available for the uses set forth in <u>Section 8.6.3</u> (Use of Handback Requirements Reserve Account). Principal Project Company shall provide to City the details regarding the

account, including the name, address and contact information for the depository institution and the account number, and shall provide Notice to City regarding any change made from time to time to any such details and the effective date of such change immediately upon or prior to such change taking effect. City shall have a first priority perfected security interest in the Handback Requirements Reserve Account, and the right to receive monthly account statements directly from the depository institution. Principal Project Company shall deliver such Notices to the depository institution and execute such documents as may be required to establish and perfect City's interest in the Handback Requirements Reserve Account under the Uniform Commercial Code as adopted in the State.

**8.6.1.2** In lieu of Principal Project Company establishing the Handback Requirements Reserve Account, Principal Project Company may deliver to City one or more Handback Requirements Letters of Credit on the terms and conditions set forth in <u>Section 8.6.5</u> (Handback Requirements Letters of Credit) and <u>Section 10.4</u> (Letters of Credit).

# 8.6.2 Funding

Within 30 days after the date of City's first Notice delivered to Principal Project 8.6.2.1 Company pursuant to Section 8.5.3.2, Principal Project Company shall deposit to the Handback Requirements Reserve Account an amount equal to the Handback Reserve Amount specified in such Notice. If, at any time after the initial deposit of funds into the Handback Requirements Reserve Account, City determines, in its good faith discretion, that the balance of funds held in the Handback Requirements Reserve Account may be insufficient to pay for the Handback Renewal Work (as amended) yet to be performed or paid for, then City may give Notice to Principal Project Company setting forth the insufficient amount and requiring Principal Project Company to replenish the Handback Requirements Reserve Account with additional funds necessary to make up the insufficiency. Principal Project Company shall deposit such additional funds into the Handback Requirements Reserve Account not later than ten Business Days after such Notice from City. If after recalculation of the Handback Reserve Amount following any of the annual inspections provided for in Section 8.5.4 (Joint Technical Review and Report), the amount on deposit in the Handback Requirements Reserve Account exceeds the percentage of the Handback Reserve Amount required to be on deposit in the Handback Requirements Reserve Account in accordance with Section 8.6.2.1 on such calculation date, Principal Project Company shall be entitled to draw any surplus amount and no further deposits shall be made until the next inspection and determination of the Handback Reserve Amount.

**8.6.2.2** If Principal Project Company disputes, in good faith, any Handback Reserve Amount specified by City in City's Notices delivered to Principal Project Company pursuant to <u>Section</u> <u>8.5.3.2</u>, Principal Project Company shall fund the full portion (including the disputed amount) of the Handback Reserve Amount pending resolution of the Contract Dispute in accordance with <u>Error! Reference source not found.</u> (Contract Dispute Procedures).

**8.6.2.3** Funds held in the Handback Requirements Reserve Account may be invested and reinvested only in securities, obligations, bonds, funds, instruments or other investments listed under California Government Code section 16430 ("**Eligible Investments**"). Eligible Investments in the Handback Requirements Reserve Account must mature during the Term, and the principal of such Eligible Investments must be available for withdrawal at any time during the Term without penalty. Any interest earned on funds held in the Handback Requirements Reserve Account shall be earned on behalf of the Principal Project Company.

**8.6.2.4** If Principal Project Company fails to make the deposits required pursuant to this <u>Section 8.6.2</u> (Funding), when due, City shall be entitled to deduct the amount required to be deposited from the Quarterly Availability Payments due to Principal Project Company and shall deposit such amount to the Handback Requirements Reserve Account on behalf of Principal Project Company.

**8.6.2.5** If the amount on deposit in the Handback Requirements Reserve Account at any time exceeds the current calculation of the Handback Reserve Amount, Principal Project Company may withdraw any surplus amount in excess of the current calculation.

# 8.6.3 Use of Handback Requirements Reserve Account

**8.6.3.1** Principal Project Company shall be entitled to draw funds from the Handback Requirements Reserve Account in such amounts and at such times as needed only to pay for the Handback Renewal Work as required by the Handback Renewal Work Plan. Amounts in the Handback Requirements Reserve Account can only be used for the purposes described in this <u>Section 8.6.3.1</u> and are not available as security for repayment of Project Debt or making Distributions, except as permitted by <u>Section 8.7.4.1</u>. The use of amounts in the Handback Requirements Reserve Account by Principal Project Company for any purpose other than as permitted in this <u>Section 8.6.3.1</u> shall be a PPC Default.

**8.6.3.2** Before drawing funds from the Handback Requirements Reserve Account, Principal Project Company shall give written Notice to City of the amount to be drawn and the purpose for which funds will be used. City shall have 10 Business Days from the date of the receipt of such Notice to disapprove the draw from the Handback Requirements Reserve Account. City may disapprove the draw only if the requested amount and/or purposes for which the funds will be used do not comply with the Handback Renewal Work Plan. If City fails to disapprove the draw within the 10 Business Day period following receipt of Notice, Principal Project Company shall be entitled to draw funds in the manner described in the Notice to City.

#### 8.6.4 Disbursement of Handback Requirements Reserve Account on Completion of Handback Renewal Work or Earlier Termination

**8.6.4.1** Following completion of the Handback Renewal Work and after the expiration of the Term, any remaining Handback Requirements Reserve Balance shall be drawn and retained by Principal Project Company.

**8.6.4.2** If this Agreement is terminated for any reason before the completion of the Handback Renewal Work, any remaining Handback Requirements Reserve Balance shall be drawn and retained by City. *[PNC TO PROVIDE SUGGESTED SPECIFIC DRAFTING TO ADDRESS ISSUE REGARDING THAT IT IS NOT IN A WORSE POSITION IF IT USES LOC VS. CASH]* 

# 8.6.5 Handback Requirements Letters of Credit

**8.6.5.1** In lieu of establishing the Handback Requirements Reserve Account, Principal Project Company may deliver to City one or more letters of credit (each, a "**Handback Requirements Letter of Credit**"), on the terms and conditions set forth in this <u>Section 8.6.5</u> (Handback Requirements Letter of Credit) and <u>Section 10.4</u> (Letters of Credit). The Handback Requirements Letter of Credit shall name City as the sole beneficiary and shall not provide for any other dual or multiple beneficiaries. Principal Project Company shall execute, and shall

cause the financial institution issuing the Handback Requirements Letter of Credit to execute, such documents as may be required to establish and perfect City's interest in the Handback Requirements Letter of Credit under the Uniform Commercial Code as adopted in the State, which interest shall be a first priority security interest as provided in <u>Section 8.6.1.1</u> If the Handback Requirements Reserve Account has been previously established, Principal Project Company at any time thereafter may substitute one or more Handback Requirements Letters of Credit for all or any portion of the amounts required to be on deposit in the Handback Requirements Reserve Account, on the terms and conditions set forth in this <u>Section 8.6.5</u> (Handback Requirements Letters of Credit).

**8.6.5.2** Upon receipt of the required substitute Handback Requirements Letter of Credit, City shall authorize the release to Principal Project Company of amounts in the Handback Requirements Reserve Account equal to the face amount of the substitute Handback Requirements Letter of Credit. If the face amount of all Handback Requirements Letters of Credit is less than the total amount required to be funded to the Handback Requirements Reserve Account before expiration of the Handback Requirements Letter of Credit, Principal Project Company shall be obligated to pay, when due, the shortfall into the Handback Requirements Reserve Account. Alternatively, Principal Project Company may deliver a Handback Requirements Letter of Credit with a face amount equal to at least the total amount required to be funded to the Handback Requirements Letter of the Handback Requirements Letter of the Bandback Requirements Letter of the Handback Requirements Letter of the Handback Requirements Letter of the Handback Requirements Letter of the Bandback Requirements Letter of the Handback Requirements Letter of the Handback Requirements Letter of the Bandback Requirements Letter of Credit, or may deliver additional Handback Requirements Letters of Credit or cause the existing Handback Requirements Letter of Credit to be amended to cover the shortfall before deposits of the shortfall to the Handback Requirements Reserve Account are due.

**8.6.5.3** City shall have the right to draw on the Handback Requirements Letter of Credit, without prior Notice to Principal Project Company, for the reasons set forth in <u>Sections</u> <u>10.4.1.3(a)</u>, (b) and (c) and <u>Section 8.6.4.2</u>.

**8.6.5.4** Except when drawing upon the Handback Requirements Letter of Credit in accordance with <u>Section 8.6.4.2</u>, City shall deposit the proceeds from drawing on the Handback Requirements Letter of Credit into the Handback Requirements Reserve Account.

# 8.6.6 Coordination with JDA Project Companies and Joint Development Alternatives

Principal Project Company acknowledges and agrees that one or more Joint Development Alternatives may be constructed prior to or following Substantial Completion. Principal Project Company shall not interfere with the work of or cause any delay to any JDA Project Company that may be carrying out work within the Project Site or in the land adjoining or near the Project Site, including with respect to the Joint Development Alternatives, and will allow them reasonable access to the Project Site, provided that Principal Project Company shall not be in breach of this <u>Section 8.7.6</u> (Coordination with JDA Project Companies and Joint Development Alternatives) for any temporary interruption to the work of any JDA Project Company that (a) has been agreed to in advance in accordance with procedures agreed to by Principal Project Company, such contractor for the Joint Development Alternative and any relevant Third Party; or (b) is reasonably necessary in accordance with Law and Good Industry Practice to respond to emergencies creating an immediate and serious threat to public health, safety, security or the Environment.

# 8.7 IFM Operations Committee

**8.7.1** As provided in <u>Section B.9.1.2</u> (IFM Operations Committee) of the IFM Specifications, the IFM Operations Committee shall be responsible for receiving and reviewing matters related to operational interface and communications for IFM Services, including Maintenance Services and Renewal Work, and SFMTA O&M Services; and

**8.7.2** Any unanimous decision of the IFM Operations Committee shall be final and binding on the Parties. If the IFM Operations Committee is unable to reach a unanimous decision, either Party may refer the matter for resolution in accordance with <u>Article 18</u> (Contract Dispute Procedures). The findings of the IFM Operations Committee shall be inadmissible and given no weight during the <u>Article 18</u> (Contract Dispute Procedures) process.

# 8.8 FF&E During IFM Period

#### 8.8.1 Title and Responsibility for Risk

**8.8.1.1** Subject to <u>Section 2.4.3</u> (Passage of Title), City will own all FF&E during the IFM Period.

#### 8.8.2 Maintenance and Replacement of FF&E During IFM Period

**8.8.2.1** Until the end of the IFM Period, Principal Project Company shall maintain, replace and repair all PPC-Maintained FF&E.

**8.8.2.2** Where Principal Project Company is required to replace PPC-Maintained FF&E, the replacement FF&E must:

- meet standards relative to the current market commensurate with standards relative to the market for the FF&E being replaced as of the original purchase date;
- (b) be consistent with Good Industry Practices for quality, durability, materials and technology;
- (c) have a design life equal to or greater than the terms of the PPC-Maintained FF&E being replaced; and
- (d) not increase costs payable by City (as compared with PPC-Maintained FF&E where alternative FF&E is readily available on comparable terms and would not have such an effect).
- 8.8.2.3 Subject to <u>Section 8.9.2.4</u> and <u>Section 6.11</u>, City will, during the Term:
  - (a) replace and repair Existing FF&E and maintain, replace and repair City-Furnished Equipment and Office/Admin and Training Spaces FF&E if an item:
    - (i) is exhibiting continuous faults; or

Infrastructure Facility DBFOM Agreement

- (ii) is leading to faults which are causing continual disruptions to IFM Services as a result of being worn or tired
- (b) notify Principal Project Company of Existing FF&E, or City-Furnished Equipment (as applicable) being replaced; and
- (c) in connection with replacements and repairs under <u>Section 8.9.2.3(a)</u>, ensure that the replacement Existing FF&E, Office/Admin and Training Spaces FF&E or City-Furnished Equipment (as applicable) do not increase the cost to Principal Project Company or any PPC-Related Entity of performing the IFM Services.

**8.8.2.4** The replacement and repair of the Existing FF&E, Office/Admin and Training Spaces FF&E or City-Furnished Equipment in accordance with <u>Section 8.9.2.3</u> will be at City's cost unless the need to replace the Existing FF&E or City-Furnished Equipment is due to any PPC Fault or the failure of Principal Project Company or any PPC-Related Entity to comply with Principal Project Company's obligations under this Agreement with respect to that Existing FF&E or City-Furnished Equipment, including <u>Section 6.11</u>. In such case, the replacement costs will be an obligation due and payable by Principal Project Company to City.

- 8.8.2.5 Principal Project Company shall:
  - (a) provide all equipment for IFM Services;
  - (b) ensure all items of FF&E and equipment for IFM Services are securely stored to prevent unauthorized access and to ensure that the health and safety of the public or City are not adversely affected;
  - (c) ensure all items of FF&E are accounted for and present on the completion of each activity or use;
  - (d) ensure all fuel to be used in the performance of the IFM Services is dispensed and accounted for via the use of a log book;
  - (e) ensure machinery and dispensing mechanisms have lockable fuel caps; and
  - (f) advise City promptly and no later than three days after Principal Project Company becomes aware of any item of FF&E or equipment for IFM Services which has not been accounted for, and cooperate with City in conducting all necessary procedures to establish and take steps to ensure that items will be accounted for properly in the future;
  - (g) develop a tools management and safety policy and submit said policy for review in accordance with <u>Exhibit 11</u> (Submittals Review Process); and
  - (h) maintain records of FF&E and FM Services Equipment audits and make such records available to City on request.

# 8.9 Energy Management

# 8.9.1 Independent Energy Auditor

**8.9.1.1** After the second anniversary, but no later than the third anniversary, of the Final Acceptance Date, the Parties shall appoint an Independent Energy Auditor to:

- (a) provide an initial report inclusive of operating assumptions, current Weather Data and energy model simulation files regarding the IFM Facility's energy performance for the purposes of establishing the Annual Energy Target in accordance with <u>Section 3</u> (Calculation of Annual Energy Target) of <u>Exhibit 12</u> (Energy Management); and
- (b) provide an updated report, as a result of agreed operating changes to the Infrastructure Facility, to determine the adjusted Annual Energy Target in accordance with <u>Section 3</u> (Calculation of Annual Energy Target) of <u>Exhibit 12</u> (Energy Management)

**8.9.1.2** Principal Project Company shall, in consultation with, and subject to the approval of, City, establish the schedule and process for the selection of an Independent Energy Auditor for purposes of the Independent Energy Auditor roles contemplated in <u>Section 3</u> (Calculation of Annual Energy Target) of <u>Exhibit 12</u> (Energy Management). Principal Project Company shall develop the solicitation package and draft contract terms, subject to approval by City, and Principal Project Company shall set the solicitation schedule with the goal of selecting the Independent Energy Auditor at least one month prior to the second anniversary of the Final Acceptance Date, and as agreed between the Parties in accordance with <u>Section 3.2.2</u> of <u>Exhibit 12</u> (Energy Management).

**8.9.1.3** The solicitations required under this <u>Section 8.9.1</u> (Independent Energy Auditor) shall include issuance of a request for competitive proposals from a list of firms approved by City, review of proposals by the Parties, a joint determination regarding which firm is the best qualified to provide Independent Energy Auditor services, and negotiation of a fair and reasonable price for performance of such services. If negotiations fail with the highest ranked firm, the Parties may elect to terminate negotiations and proceed with the next highest ranked firm. This process shall be followed until a firm is selected to serve as the Independent Energy Auditor. If the Parties fail to reach agreement regarding selection of the Independent Energy Auditor, or regarding acceptable terms of the agreement with the Independent Energy Auditor, the Claim shall be subject to resolution under <u>Article 18</u> (Contract Dispute Procedures).

**8.9.1.4** The Independent Energy Auditor will be appointed jointly by the Parties and will act independently from and not as an agent of either Party. Any member or advisor of City, a PPC-Related Entity, or a Lender is deemed to have an organizational conflict of interest and not eligible to respond to the solicitation.

**8.9.1.5** Each Party is responsible for its own costs related to the Independent Energy Auditor solicitation processes. The Parties will be equally responsible for all amounts payable under the terms of the agreement with the Independent Energy Auditor.

# 8.9.2 External Energy Auditor

**8.9.2.1** On an as needed basis during the IFM Period, the City, in its sole discretion, may select an External Energy Auditor to:

- (a) undertake an energy audit of the Bus Yard inclusive of both SFMTA O&M Services and IFM Services; and
- (b) provide operational recommendations related to energy savings and maintenance considerations for Bus Yard inclusive of both SFMTA O&M Services and IFM Services
- **8.9.2.2** If the City asks PPC to undertake the solicitation of the External Energy Auditor:
  - (a) Principal Project Company shall, in agreement with City, establish the schedule and process for the selection of an External Energy Auditor. Principal Project Company shall develop the solicitation package and draft contract terms, subject to approval by City, and Principal Project Company shall set the solicitation schedule with the goal of selecting the External Energy Auditor, and as agreed between the Parties in accordance with Section 3.2.2 of Exhibit 12 (Energy Management).
  - (b) The solicitations required under this <u>Section 8.9.2.2</u> (External Energy Auditor) shall include issuance of a request for competitive proposals from a list of firms approved by City, review of proposals by the Parties, a joint determination regarding which firm is the best qualified to provide External Energy Auditor services, and negotiation of a fair and reasonable price for performance of such services. If negotiations fail with the highest ranked firm, the Parties may elect to terminate negotiations and proceed with the next highest ranked firm. This process shall be followed until a firm is selected to serve as the External Energy Auditor. If the Parties fail to reach agreement regarding selection of the Independent Energy Auditor, or regarding acceptable terms of the agreement with the Independent Energy Auditor, the Claim shall be subject to resolution under <u>Article 18</u> (Contract Dispute Procedures).

**8.9.2.3** If the City undertakes the solicitation of the External Energy Auditor by itself, the City will determine its procurement process in its sole discretion.

**8.9.2.4** The External Energy Auditor will be appointed by the City and will act independently from and not as an agent of City. Any advisor or member of City, a PPC-Related Entity, or a Lender is deemed to have an organizational conflict of interest and not eligible to respond to the solicitation.

**8.9.2.5** City will be responsible for all costs related to the External Energy Auditor solicitation processes and all amounts under the terms of the agreement with the External Energy Auditor.

**8.9.2.6** Principal Project Company shall, promptly and reasonably, provide any information necessary for the External Energy Auditor to reasonably undertake its obligations in providing the services in accordance with <u>Section 8.9.2.1</u>

**8.9.2.7** Principal Project Company shall comply with the operational recommendations of the External Energy Auditor at its own cost in accordance with <u>Section 6</u> (External Energy Auditor) of <u>Exhibit 12</u> (Energy Management).

# 8.9.3 Failure to meet LEED Requirements

**8.9.3.1** In the event that LEED BD+C Gold Rating, including the mandatory 12x EAc2 Optimize Energy Performance credits, is not obtained within 12 months after the Substantial Completion Date, other than as a sole and direct result of any affirmative act or omission of City or unless unreasonably withheld by USGBC or an equivalent body, Principal Project Company shall pay to City liquidated damages in the amount of \$100,000. In the event that LEED BD+C Gold Rating, including the mandatory 12x EAc2 Optimize Energy Performance credits, is not obtained within 24 months after the Final Acceptance Date, other than as a sole and direct result of any affirmative act or omission of City or unless unreasonably withheld by USGBC or an equivalent body, such event shall be a PPC Default pursuant to <u>Section 16</u>.

**8.9.3.2** Principal Project Company shall submit all necessary documentation to achieve LEED BD+C Gold within 6 months after the Substantial Completion Date.

**8.9.3.3** The Parties agree that such liquidated damages are not a penalty but represent a genuine and reasonable pre-estimate of the damages that the City will suffer as a result of the happening of the specified event and would be difficult or impossible to quantify upon the happening of the specified event. Such payment shall constitute full and final settlement of all damages that may be claimed by the City as a result of a failure by Project Company to achieve LEED BD+C Gold Rating and the minimum credit requirements and, for greater certainty, a failure by Project Co to achieve LEED BD+C Gold Rating shall not result in a PPC Default Event provided that such liquidated damages are promptly paid. The Parties agree that such liquidated damages shall be payable whether or not City incurs or mitigates its damages, and that City shall not have any obligation to mitigate any such damages.

# ARTICLE 9. CONTRACTING AND LABOR PRACTICES

# 9.1 Disclosure of Contracts and Contractors

**9.1.1** The provisions of this <u>Section 9.1</u> (Disclosure of Contracts and Contractors) apply with respect to (a) Prime Contracts and lower tier Contracts entered into by Prime Contractors, excluding personal services contracts and contracts with Suppliers other than Key Contractors, and (b) Contracts with Affiliates, regardless of the nature or tier of the Contract.

**9.1.2** With each Monthly Report required during the D&C Period and each Infrastructure Facility monthly report required during the IFM Period, Principal Project Company shall provide City with a list of all Contracts, including, for each Contract, (a) the name of the Contractor (indicating the Contractor is an Affiliate, if applicable), (b) a summary of the scope of work, and (c) the dollar value. If the Contract includes Construction Work or Renewal Work, include assurance that the prevailing wage for labor, as specified in the Contract Documents, is paid for all labor performed under such Contract.

**9.1.3** Principal Project Company shall allow City full access to all Contracts and records regarding Contracts and shall deliver to City, (a) within 10 days after execution, copies of all Key Contracts, guarantees thereof and amendments and supplements to Key Contracts and guarantees thereof, and (b) within 10 days after receipt of a request from City, copies of all other Contracts (including amendments and supplements) as may be requested.

# 9.2 Responsibility for Work, Contractors and Employees

**9.2.1** Principal Project Company may enter into one or more Contracts with Contractors to perform portions of the Work.

**9.2.2** The retention of Contractors by Principal Project Company does not relieve Principal Project Company of its responsibilities under this Agreement or for the quality of the Work or materials or services provided by it.

**9.2.3** Each Contract shall include (a) terms sufficient to ensure both the acknowledgement of and compliance by the Contractor with the applicable requirements of the Contract Documents and to ensure that City has the ability to exercise its rights specified in the Contract Documents, (b) those terms that are specifically required by the Contract Documents to be included in such Contract, and (c) all applicable Laws.

**9.2.4** Principal Project Company shall require each Contractor to familiarize itself with the requirements of any and all applicable Laws and the conditions of any required Regulatory Approvals.

**9.2.5** Nothing in this Agreement will create any contractual relationship between City and any Contractor. No Contract entered into by or under Principal Project Company shall impose any obligation or liability upon any Indemnitee to any Contractor or any of its employees. Principal Project Company shall include, or cause to be included, a provision in all Contracts acknowledging the same.

**9.2.6** Principal Project Company shall supervise and be fully responsible for any PPC Fault while performing Work under this Agreement or while on the Project Site, as though Principal Project Company directly employed all such individuals.

**9.2.7** If City has reasonable cause to disapprove of an employee of any PPC-Related Entity, Principal Project Company shall remove such employee within 10 Business Days after receipt of written Notice from City of such disapproval.

# 9.3 Key Contracts; Contractor Qualifications

# 9.3.1 Key Contract Approvals, Amendments and Termination; Use of and Change in Key Contractors

**9.3.1.1** Principal Project Company shall provide City, for City's review and comment, draft copies of (a) any Key Contracts not executed before the Effective Date and (b) proposed material amendments to Key Contracts (regardless of whether the Key Contract to be amended was executed before the Effective Date). Such drafts shall be provided at least 30 days before execution of a Key Contract or an amendment to a Key Contract, as applicable. Any proposed amendment to required terms described in <u>Section 9.3.2</u> (Key Contract Provisions) shall be considered a material amendment.

**9.3.1.2** Except as otherwise approved by City, Principal Project Company shall retain, employ and utilize the firms and organizations identified in the Implementation Proposal to fill the roles designated therein.

**9.3.1.3** Principal Project Company shall not terminate or permit termination of any Key Contract or permit any substitution, replacement or assignment of any Key Contractor, except with City's prior approval; provided, however, that City's prior approval is not required in the event of (a) any termination of this Agreement where City elects not to assume Principal Project Company's future obligations under such Key Contract, (b) any suspension, debarment, disqualification or removal (distinguished from ineligibility due to lack of financial qualifications) of the Key Contractor, or (c) any agreement for voluntary exclusion of the Key Contractor, from bidding, proposing or contracting with any federal, State or local department or agency. City agrees to act reasonably with respect to approval of a replacement in the case of material uncured default by the Key Contractor under the Key Contract.

# 9.3.2 Key Contract Provisions

**9.3.2.1** Each Key Contract shall comply with the requirements set forth in <u>Exhibit 14</u> (Key Contract Provisions).

# 9.4 Prompt Payment to Contractors

Principal Project Company shall comply, and shall cause each Contractor to comply, with the provisions of Business and Professions Code section 7108.5, California Civil Code sections 8122-8138, and any other applicable Law relating to prompt payment of contractors and/or subcontractors and waivers and releases by them of stop payment notices and payment bond rights.

#### 9.5 Key Personnel

Principal Project Company shall:

- (a) retain, employ and utilize the individuals specifically listed in the Implementation Proposal, and retain, employ and utilize individuals qualified for the positions described in <u>Section 1.1.4</u> of <u>Division 1</u> of the Technical Requirements to fill Key Personnel positions for the relevant period identified in the Technical Requirements; Principal Project Company acknowledges that if City reasonably determines that an individual is not qualified for a Key Personnel position, Principal Project Company shall at City's request replace that individual with one that meets the required qualifications;
- (b) not change or substitute any such individuals except due to retirement, death, disability, incapacity, or voluntary or involuntary termination of employment, or as otherwise approved by City under <u>Section 9.5(c)</u>;
- (c) promptly provide Notice to City of any proposed replacement for any Key Personnel position. City shall have the right to review the qualifications and character of each individual to be appointed to a Key Personnel position and shall act reasonably in approving or disapproving use of such individual in such position before the commencement of any Work by such individual;
- (d) cause each Key Person to dedicate the full amount of time necessary for the proper prosecution and performance of the Work, as more specifically described in <u>Section 1.1.4</u> of <u>Division 1</u> of the Technical Requirements;
- (e) commit each Key Person to the Project in accordance with <u>Section 1.1.4</u> of <u>Division 1</u> of the Technical Requirements;
- (f) provide City with office and cell phone numbers and email addresses for each Key Person. City may contact Key Personnel 24 hours per day, seven days per week;
- (g) ensure that the Project Manager identified in the Implementation Proposal or otherwise approved by City (i) will have full responsibility for the prosecution of the D&C Work, (ii) will act as agent and be a single point of contact in all matters relating to the D&C Work on behalf of Principal Project Company at least until Final Acceptance, (iii) will be present (or ensure that his or her approved designee is present) at the construction site at all times that Construction Work is performed, and (iv) will be available to respond promptly to City; and
- (h) ensure that the Quality Program Manager identified in the Implementation Proposal or otherwise approved by City (i) will have full responsibility for quality assurance and quality control with respect to D&C Work until Final Acceptance, (ii) is present (or his or her approved designee is present) at the construction site at all times that Construction Work is performed, and (iii) will be available to respond promptly to City.

## 9.5.1 Key Personnel Deductions

**9.5.1.1** If an individual filling a Key Personnel role is not available for, or actively involved in, the performance of the Work per their position requirement as required in <u>Section 1.1.4</u> of <u>Division 1</u> of the Technical Requirements, as determined by City in its good faith discretion, then:

- (a) Principal Project Company acknowledges and agrees that City and the Project will suffer significant and substantial damages and that it is impracticable and extremely difficult to ascertain and determine the actual damages which would accrue to City in such event;
- (b) Subject to Section 9.5.2 (Limitations on Deductions for Unavailability of Key Personnel), Principal Project Company shall pay City a Key Personnel deduction as follows, for each position held by such individual, as deemed compensation to City for such damages:

Category	KEY PERSONNEL DEDUCTION
Project Manager	\$150,000
Deputy Project Manager	\$75,000
Design Manager	\$75,000
Construction Manager	\$75,000
IFM Manager	\$75,000
Quality Program Manager	\$75,000
Project Safety Representative	\$75,000
Construction Superintendent	\$75,000
Third Party and Utility Coordinator Manager	\$75,000

and

(c) Principal Project Company agrees to pay a further Key Personnel deduction for all Key Personnel in the amount of 50% of the applicable amounts listed under <u>Section 9.5.1.1(b)</u> for each 6 calendar month period where any Key Personnel position is vacant or not being fulfilled in accordance with this Agreement as determined by City.

**9.5.1.2** Principal Project Company agrees that the Key Personnel deductions payable in accordance with <u>Section 9.5.1.1</u> are liquidated damages, not a penalty and are reasonable under the circumstances existing as of the Effective Date. The Parties have agreed to liquidated damages under this <u>Section 9.5.1</u> (Key Personnel Deductions) in order to fix and limit Principal Project Company's costs and to avoid later Contract Disputes over what amounts of damages are properly chargeable to Principal Project Company.

**9.5.1.3** City may deduct amounts owing from Principal Project Company to City under <u>Section 9.5.1.2</u> from amounts owing from City to Principal Project Company under this

Agreement (including the Milestone Payment), or to collect such liquidated damages from the Performance Bond.

#### 9.5.2 Limitations on Deductions for Unavailability of Key Personnel

**9.5.2.1** Principal Project Company is not liable for liquidated damages under <u>Section 9.5.1.1</u> if:

- (a) Principal Project Company removes or replaces such personnel at the direction of City; or
- (b) An individual filling a Key Personnel position is unavailable because of the application of applicable Law, or due to death, disability, family leave, retirement, injury or no longer being employed by the applicable PPC-Related Entity (provided that moving to an affiliated company or a Subcontractor is not considered grounds for avoiding liquidated damages),

provided Principal Project Company promptly proposes to City a replacement for such personnel for review and approval within 30 days of unavailability, in the case of <u>Sections</u> <u>9.5.2.1(a)</u> or <u>9.5.2.1(b)</u>.

**9.5.2.2** Principal Project Company may replace the Principal Project Company's Project Manager not more frequently than every three years during the IFM Period without incurring liquidated damages under <u>Section 9.5.1</u> (Key Personnel Deductions), but only if Principal Project Company replaces the outgoing Principal Project Company's Project Manager with a City-approved replacement before the outgoing individual vacates the position.

**9.5.2.3** Upon approval of any Key Personnel replacement, the new individual shall be considered a Key Personnel for all purposes under this Agreement, including this <u>Section 9.5.2</u> (Limitations on Deductions for Unavailability of Key Personnel).

# 9.5.3 Skilled Personnel; Removal at City Direction

All individuals performing the Work shall have the skill and experience and any licenses or certifications required to perform the Work assigned to them. If City determines, in its good faith discretion, that any individual employed by Principal Project Company or by any Subcontractor is not performing the Work in a proper, safe and skillful manner, then at the written request of City, Principal Project Company or such Subcontractor shall promptly remove such individual and such individual shall not be re-employed on the Project without the prior written approval of City, in its good faith discretion. If Principal Project Company or the Subcontractor fails to remove such individual or individuals or fails to furnish skilled and experienced personnel for the proper performance of the Work, then City may, in its good faith discretion, suspend the affected portion of the Work for cause by delivery of Notice of such suspension to Principal Project Company. Such suspension shall in no way relieve Principal Project Company of any obligation contained in the Contract Documents or entitle Principal Project Company to a time extension, compensation or other Change Order. Once compliance is achieved, as determined by City, City will deliver Notice to Principal Project Company and Principal Project Company shall be entitled to and shall promptly resume the Work. If City's determination is later found to not have

Infrastructure Facility DBFOM Agreement

been exercised in good faith, such suspension shall be considered a suspension for convenience.

## 9.6 DBE Participation

# [NOTE TO PNC: DBE PROVISIONS/PROGRAM UNDER DEVELOPMENT]

**9.6.1** Principal Project Company shall either achieve or demonstrate it has exercised good faith efforts to achieve the following Disadvantaged Business Enterprise ("DBE") participation goals:

- (a) [\_\_]% of Principal Project Company's costs for the Design Work;
- (b) [\_\_]% of Principal Project Company's costs for the Construction Work;

**9.6.2** Achievement of the DBE participation goal for each type of Work identified in <u>Sections 9.6.2(a)</u> and (b) will be measured as a percentage of Principal Project Company's costs for each such type of Work, where the total amount paid to DBE firms for each such type of Work will be measured against Principal Project Company's costs for each such type of Work, without reduction for Deductions or other offsets.

**9.6.3** Principal Project Company shall comply with the terms and conditions set forth in Exhibit [\_\_] (DBE Compliance Manual) and in the approved DBE Utilization Plan.

**9.6.4** If Principal Project Company fails to achieve any DBE participation goal, then Principal Project Company must demonstrate that it met all good faith efforts requirements for achieving the goal in accordance with 49 C.F.R. Appendix A to Part 26 - Guidance Concerning Good Faith Efforts and the DBE Compliance Manual. The efforts employed must at a minimum include those that one could reasonably expect a contractor to take if the contractor were actively and aggressively trying to obtain DBE participation sufficient to meet the DBE goals (See 49 C.F.R. Part 26, Appendix A).

**9.6.5** To obtain an accurate record of Principal Project Company's performance towards meeting its DBE participation commitments and DBE goals, Principal Project Company shall utilize City's reporting system to report DBE and non-DBE payments and all other reporting activities throughout the performance of Work. Principal Project Company shall maintain an accurate listing of all DBE and non-DBE firms in City's reporting system. Principal Project Company shall ensure that all relevant Contractors, Suppliers and/or brokers (at all tiers) participate in trainings in the use and operation of City's reporting system and comply with the verification of payments and other related reporting requirements through the reporting system. City may require additional reports to ensure adequate reporting of DBE participation for the Project.

**9.6.6** Principal Project Company shall complete all DBE and non-DBE reporting requirements on City's reporting system in accordance with the DBE Compliance Manual. If City provides a notice to Principal Project Company indicating that Principal Project Company has not achieved a DBE participation goal, Principal Project Company shall, within ten days of such notice, submit for City's consideration Principal Project Company's Good Faith Efforts documentation for the applicable time period.

**9.6.7** Failure by Principal Project Company to carry out the DBE requirements set forth in this <u>Section 9.6</u> (DBE Participation) is a material breach of this Agreement, which may result in:

- (a) the termination of this Agreement;
- (b) Noncompliance Points and Deductions as set forth in <u>Exhibit 4</u> (Payment Mechanism); and/or
- (c) other administrative remedies, including monetary liability, as set forth in Sections 801 and 802 of <u>Exhibit []</u> (DBE Compliance Manual) or in this Agreement.

**9.6.8** Principal Project Company acknowledges the importance of Principal Project Company and Contractors timely tracking and entering DBE compliance data into City's reporting system. City's comments, approvals, acceptances, NTPs, payments, including the Milestone Payment, Availability Payments, periodic payments, progress payments, and other actions shall not waive or limit any remedies City has under the Contract Documents should City determine that Principal Project Company has failed to carry out the DBE requirements pursuant to this <u>Section 9.6</u> (DBE Participation).

#### 9.6.9 DBE Utilization Plan

**9.6.9.1** Within [\_\_] days after issuance of NTP 1, Principal Project Company shall submit to City for approval in its sole discretion, a DBE Utilization Plan that describes Principal Project Company's robust plan outreach, marketing, recruitment, and other good faith efforts for achieving the DBE participation goals set forth in this <u>Section 9.6</u> (DBE Participation). The DBE Utilization Plan shall, at a minimum, include the following components:

- (a) provisions for submission of DBE forms and affidavits through City's DBE reporting system;
- (b) expanded descriptions of the types of proactive DBE and small business bidspecific marketing, recruitment, outreach and community engagement efforts that Principal Project Company will implement while preparing for and undertaking the D&C Work in order to include DBEs and on the Project, including:
  - (i) processes for timely communications and outreach methods that Principal Project Company will use;
  - (ii) processes to keep track of potential DBEs, small businesses and other Contractors on the Work;
  - (iii) proposed innovative methods for (A) involving new and emerging DBEs, and (B) identifying firms that might potentially be certified as DBEs, and for assisting them to become DBE-certified and involved in the Project; and
  - (iv) a discussion of how these efforts will flow through tiers of Subcontractors on the Project;

- (c) description of efforts Principal Project Company has made and will make to recruit and utilize DBE firms such as graphic design and printing, marketing, outreach, training, employment services, catering companies, maintenance companies, janitorial services, security companies and other trades to help meet the DBE business goals;
- (d) description of proposed DBE and small business capacity-building efforts to be implemented throughout the Work, including methods to assist DBEs with record-keeping and compliance, bonding, financing, access to supplies and other capabilities;
- description of the estimated DBE participation level for Design Work for each month of the Design Work and cumulatively, including a table or diagram of an estimated schedule that illustrates projected work sequencing of DBE utilization for the Design Work;
- (f) description of the estimated DBE participation level for Construction Work for each month of the Construction Work and cumulatively, including a table or diagram of an estimated schedule that illustrates projected work sequencing of DBE utilization for the Construction Work;
- (g) description of processes and procedures that Principal Project Company will use to monitor, track, document and report recruitment, progress and utilization of DBEs, and to maintain and adjust the DBE participation schedule to help ensure achievement of the DBE goals, including time intervals at which Principal Project Company will employ these processes and procedures;
- (h) description of specific measures that Principal Project Company will undertake throughout the duration of the D&C Work to achieve and manage the DBE goals, including training workshops, technical and financial assistance, support services, mentor/protégé relationships, recruiting and encouraging potential DBEs to obtain certification, etc. Include a proposed schedule of events/activities;
- (i) description of how Principal Project Company will manage DBEs on the Project, including processes for project management, technical performance reviews, feedback and dispute resolution to resolve issues that may arise;
- description of other procedures and processes for meeting DBE requirements, such as documenting and submitting affidavits for additional DBEs committed to the Project to meet or exceed the DBE goals, prompt pay requirements and substitution/replacement of DBEs; and
- (k) description of any other innovative or additional good faith efforts activities already undertaken or ones Principal Project Company plans to undertake that are not listed above or listed in 49 C.F.R. Part 26.

**9.6.10** Principal Project Company shall engage the DBE Contractors identified in the table set forth in Exhibit [\_\_] (List of Approved DBE Subcontractors and Suppliers) as part of Principal

Project Company's efforts to meet Principal Project Company's DBE participation commitments and DBE goals during the Term, subject to the DBE Contractor substitution procedures set forth in <u>Exhibit [\_\_]</u> (DBE Compliance Manual).

**9.6.11** Not later than [150] days following NTP 1, in accordance with the requirements set forth in Exhibit [\_\_] (DBE Compliance Manual), Principal Project Company shall provide to City for approval a list of all the DBE and small business firms that Principal Project Company proposes to engage to meet Principal Project Company's DBE participation commitments for the Construction Work.

**9.6.12** Unless City's prior written consent is provided for termination or substitution of a DBE Contractor, Principal Project Company shall not be entitled to any payment for work or materials identified for a DBE Contractor unless it is performed or supplied by such DBE Contractor.

### 9.6.13 DBE Compliance Contractor

Principal Project Company shall retain or cause to be retained a Contractor to provide DBE liaison and compliance services that:

- (a) is not corporately affiliated with Principal Project Company or a Key Contractor; and
- (b) has primary responsibility on behalf of Principal Project Company to coordinate Principal Project Company's DBE compliance efforts and interface with City regarding DBE and SBE matters.

#### 9.6.14 DBE Compliance Assessment

**9.6.14.1** Annually until Substantial Completion, following the anniversary of the Effective Date each year, City will assess Principal Project Company's utilization during the preceding 12-month period of DBE firms in performance of Design Work and Construction Work. If after the end of any such 12-month period Principal Project Company is behind in meeting a DBE participation goal under <u>Section 9.6.1(a)</u> or (b), City may require Principal Project Company to update its plan and efforts to achieve the goal, at no additional cost to City.

#### 9.7 Contracts with Affiliates

- 9.7.1 Affiliates may perform Work only if:
  - (a) Principal Project Company or a Contractor executes a written Contract with the Affiliate which:
    - complies with all applicable provisions of the Contract Documents, including this <u>Article 9</u> (Contracting and Labor Practices), is consistent with Good Industry Practice, and is in form and substance substantially similar to Contracts then being used by Principal Project Company or Affiliates for similar Work or services with unaffiliated contractors;
    - (ii) sets forth the scope of Work and all the pricing, terms respecting the scope of Work;

- (iii) contains pricing, scheduling and other terms no less favorable to Principal Project Company than those that Principal Project Company could reasonably obtain in an arms'-length, competitive transaction with an unaffiliated Contractor. Principal Project Company shall bear the burden of proving that the same are no less favorable to Principal Project Company; and
- (b) The Work to be performed by the Affiliate is not Work that any Contract Document, the Project Management Plan indicates is to be performed by an independent or unaffiliated party.

**9.7.2** Before Principal Project Company or a Contractor enters into a written Contract (including supplements and amendments) with an Affiliate, Principal Project Company shall submit a true and complete copy of the proposed Contract to City for comment. City shall have 20 days after receipt to deliver its comments to Principal Project Company. If the Contract with the Affiliate is a Key Contract, it shall be subject to City's approval as provided in <u>Section 9.3.1</u> (Key Contract Approvals, Amendments and Termination; Use of and Change in Key Contractors).

**9.7.3** Principal Project Company shall make no payments to Affiliates for work or services in advance of provision of such work or services, except for reasonable mobilization payments or other payments consistent with arm's length, competitive transactions of similar scope. Advance payments in violation of this provision shall be excluded from the calculation of Termination Compensation.

# 9.8 Labor Standards

**9.8.1** In performing the Work, Principal Project Company shall comply, and require all Contractors to comply, with all applicable federal and State labor, occupational safety and health Laws and orders, including payment of Prevailing Rate of Wages.

**9.8.2** By the 15<sup>th</sup> day of each calendar month during the Term, Principal Project Company shall submit to City certified payroll records for all employees of Principal Project Company and Contractors at all tiers for the preceding calendar month.

**9.8.3** In the event a prevailing wage law violation is discovered, City may withhold from payments owing (including the Milestone Payment and any Availability Payments) Principal Project Company's underpaid back wages plus penalties as required under Section 6.22(e)(8) (Non-Compliance with Wage Provisions – Penalties) of the San Francisco Administrative Code until the violation is resolved.

**9.8.4** All individuals performing the Work shall be qualified, experienced, competent and skilled in the performance of the portion of the Work assigned and related obligations of Principal Project Company in accordance with the Contract Documents and any applicable minimum levels in the Technical Requirements.

**9.8.5** If any individual employed by Principal Project Company or any Contractor lacks required qualifications, skill, competence, experience, licensing, certification, registration, permit, approval, bond or insurance or is not performing the Work in a proper, safe and skillful

manner, then Principal Project Company shall, or shall cause such Contractor to, remove such individual, and such individual shall not be re-employed on the Work.

**9.8.6** If, after Notice and reasonable opportunity to cure, Principal Project Company fails to take action as required by <u>Section 9.8.5</u>, or if Principal Project Company fails to ensure that qualified, skilled, experienced, competent, licensed, certified, registered, permitted and approved personnel are furnished for the proper performance of the Work, then City may suspend the affected portion of the Work by delivering to Principal Project Company Notice of such suspension. Such suspension shall in no way relieve Principal Project Company of any obligation contained in the Contract Documents.

### 9.9 Local Hiring Requirements for Construction Work and Renewal Work

Principal Project Company shall comply with local hiring requirements for Construction Work and Renewal Work, as mandated by the San Francisco Local Hiring Policy for Construction set forth in Chapter 82 of the Administrative Code and related implementing regulations.

#### 9.10 First Source Hiring Program

Principal Project Company shall comply with all applicable provisions of the First Source Hiring Program as set forth in Chapter 83 of the San Francisco Administrative Code, including all enforcement and penalty provisions.

### 9.11 SFMTA Employment Training Program

**9.11.1** As part of the SFMTA Employment Training Program, Principal Project Company shall cause at least 4 professional services trainees to be hired during the period of the D&C Work for professional services performed for the Infrastructure Facility. If a person hired through the First Source Hiring Program also meets the trainee requirements described below, that person may be counted toward these trainee hiring requirements. Trainees may be obtained through the City's One Stop Employment Center, which works with various employment and job training agencies/organizations or other employment referral source.

- **9.11.2** Principal Project Company shall ensure that:
  - (a) Each trainee is hired by either the Principal Project Company or a Contractor providing professional services for the Infrastructure Facility.
  - (b) No trainee is counted towards meeting more than one contract requirement. For example, if City and Principal Project Company enter into this Agreement and another contract, any trainee hired for services under this Agreement would not count toward the trainee hiring requirement for the other contract.
  - (c) Each trainee meets enrollment qualifications established under the City's First Source Hiring Program as follows:
    - (i) "Qualified" with reference to an economically disadvantaged individual shall mean an individual who meets the minimum bona fide occupational qualifications provided by the prospective employer to the

San Francisco Workforce Development System in the job availability notices required by the First Source Hiring Program.

- (ii) "Economically disadvantaged individual" shall mean an individual who is either (i) eligible for services under the Workforce Investment Act of 1988 (29 U.S.C. 2801 et seq.), as determined by the San Francisco Private Industry Council; or (ii) designated "economically disadvantaged" by the FS First Source Hiring Program HP administration, which means an individual who is at risk of relying upon, or returning to, public assistance.
- (iii) "On-the-job training" means the hiring party hire the trainee on a fulltime basis for at least 12 months or on a part-time basis for 24 months (using the full-time or part-time definition of the employer hiring that trainee), with prior approval offering him/her on-the-job training that allows the trainee to progress on a career path.
- (d) Before a trainee is hired by a Contractor or by Principal Project Company, Principal Project Company submits for City's approval a description and summary of training proposed for that trainee, along with the rate of pay for the position.
- (e) A trainee's commitment does not require that he/she is used only on this Project; the trainee may also be used on other Principal Project Company or Contractor projects that may be appropriate for the trainee's skill development.

Principal Project Company acknowledges its obligation to hire or cause to be hired trainees pursuant to the Trainee Plan and shall comply with the Trainee Plan during the Term.

### 9.12 Deductions and Noncompliance Points Relating to Local Hire and SFMTA Training Program Requirements Development

If Principal Project Company fails to meet certain requirements set forth in this <u>Article 9</u> (Contracting and Labor Practices), City may, without limiting City's other rights and remedies under this Agreement, assess Deductions and Noncompliance Points as provided in <u>Exhibit 4</u> (Payment Mechanism).

# 9.13 Ethical Standards

**9.13.1** Principal Project Company or its representatives shall not make, nor cause to be made, any cash payments, commissions, employment, gifts, entertainment, free travel, loans, free work, substantially discounted work, or any other considerations to (a) City representatives, employees, or their relatives, or (b) representatives of subcontractors, or material suppliers or any other individuals, organizations, or businesses receiving funds in connection with this Project.

**9.13.2** Principal Project Company employees (or their relatives), agents, or subcontractors shall not receive any cash payments, commissions, employment, gifts, entertainment, free travel, loans, free work, or substantially discounted work or any other considerations from any other contractors or from any City employee, agent, representative.

**9.13.3** Principal Project Company agrees to provide Notice to a designated City representative within 48 hours of any instance where Principal Project Company becomes aware of a failure to comply with the provisions of this <u>Section 9.13</u> (Ethical Standards).

### ARTICLE 10. INSURANCE; PAYMENT AND PERFORMANCE SECURITY; INDEMNITY

## 10.1 Insurance

## 10.1.1 Insurance Policies and Coverage

Principal Project Company shall procure and maintain, or cause to be procured or maintained, the Insurance Policies identified in this <u>Section 10.1</u> (Insurance) and in <u>Exhibit 7</u> (Insurance Requirements) strictly in accordance with the minimum coverage requirements and terms of coverage as set forth in <u>Exhibit 7</u> (Insurance Requirements) and in this <u>Section 10.1</u> (Insurance). Principal Project Company shall timely pay, or cause to be paid, the premiums for all Insurance Policies and insurance coverages required by this Agreement. There shall be no recourse against City or any of the other City Additional Insureds for payment of premiums or other amounts with respect to the Insurance Policies, except to the extent (a) included within the Availability Payments or (b) an increase in premiums is a compensable Extra Work Cost as part of a Compensable Delay Event or Compensable Relief Event.

# **10.1.2** General Insurance Requirements

# 10.1.2.1 Insurers

All insurance required hereunder shall be procured from insurers that at the time coverage commences are authorized to do business in the State and have a rating of not less than A-:VIII according to A.M. Best's Financial Strength Rating and Financial Size Category, except as otherwise provided in Exhibit 7 (Insurance Requirements) or approved in writing by City in its reasonable discretion.

# 10.1.2.2 Deductibles; Self-Insured Retentions; Claims Exceeding Policy Limits

Except to the extent expressly provided otherwise in the Contract Documents, Principal Project Company shall be responsible for paying all insurance deductibles and self-insured retentions, and City shall have no liability for deductibles, self-insured retentions or claim amounts exceeding the required policy limits. If City is the sole cause of an insured Loss (including Losses caused by the SFMTA's operations) City shall pay the applicable insurance deductibles and self-insured retentions, notwithstanding any subrogation waivers. In the event that any required insurance coverage involves a self-insured retention: (a) the entity responsible for the self-insured retention shall have an authorized representative issue a letter to City, at the same time the Insurance Policy is to be procured, stating that it shall protect and defend City to the same extent as if an insurer provided coverage for City; and (b) Principal Project Company shall ensure that the relevant Insurance Policy expressly permits (but does not obligate) City, or a designee of City, to pay the self-insured retention to satisfy any policy condition requiring payment of the self-insured retention for coverage to apply. If the entity responsible for the selfinsured retention does not pay the self-insured retention amount when due, then City may, but is not obligated to, pay the self-insured retention amount on behalf of such entity, and Principal Project Company shall indemnify City for such amount and any other Losses incurred by City in connection with the entity's failure to pay the self-insured retention amount when due.

Infrastructure Facility DBFOM Agreement

# 10.1.2.3 Primary Coverage

Without any limitation, each policy shall provide that the coverage afforded under the policy is primary and noncontributory with respect to any other insurance available to all named and City Additional Insureds.

### 10.1.2.4 Verification of Coverage

- (a) When Principal Project Company is required by the Contract Documents to initially obtain or cause to be obtained an Insurance Policy, Principal Project Company shall deliver to City a certificate of insurance before policy inception followed by a binder of insurance in a form reasonably acceptable to City no later than 30 days following policy inception. Principal Project Company shall make submit all certificates of insurance including relevant endorsements attached to the certificates of insurance prior to each policy effective date and binders of insurance as soon as they are issued by the insurers but not later than 30 days following policy inception. Each required binder must state the identity of all insurers, list all named and additional insureds, state the type of coverage and limits including any applicable sublimits, list all deductibles and self-insured retentions, provide the policy effective and expiration date, and list all applicable endorsements once available.
- (b) In addition, as soon as they become available, Principal Project Company shall deliver to City (i) a true and complete copy of each such Insurance Policy or modification, or renewal or replacement Insurance Policy obtained by Principal Project Company or a Prime Contractor and all endorsements thereto and (ii) satisfactory evidence of payment of the premium therefor.
- (c) If Principal Project Company has not provided City with the certificates of insurance, including applicable endorsements, prior to policy inception and evidence of premium payment within 30 days after each payment is due, City may, upon written Notice to Principal Project Company, in addition to any other available remedy, without obligation and without further inquiry as to whether such insurance is actually in force, elect to obtain such an Insurance Policy; and Principal Project Company shall reimburse City for any Recoverable Costs thereof upon demand. In addition, City shall have the right, without obligation or liability, to suspend all or any portion of Work, during any time that such proofs of coverage, in compliance with this <u>Section</u> <u>10.1</u> (Insurance), have not been provided.

# 10.1.2.5 Waivers of Subrogation

City and Principal Project Company waive all rights against each other, against each of their respective agents, employees and Project consultants, and against Contractors and their respective members, directors, officers, employees, subcontractors and agents for any claims to the extent covered and paid by insurance obtained pursuant to this <u>Section 10.1</u> (Insurance), with the exception of professional liability insurance and except such rights as they may have to the proceeds of such insurance. Principal Project Company shall require all Contractors to provide similar waivers in writing, each in favor of all other parties specified above. Each policy

for which Principal Project Company or any Contractor is required to provide coverage for the City Additional Insureds shall include a waiver of any right of subrogation against the City Additional Insureds.

### 10.1.2.6 No Recourse

Except as may be inclusive within the MaxAP or as expressly provided otherwise in this Agreement, there shall be no recourse against City for payment of premiums or other amounts with respect to the insurance Principal Project Company is required to provide hereunder.

#### 10.1.2.7 Support of Indemnifications

The insurance coverage Principal Project Company is required to provide hereunder shall support but is not intended to limit Principal Project Company's indemnification obligations under the Contract Documents.

### 10.1.2.8 Additional Terms and Conditions

- (a) Each Insurance Policy shall state or be endorsed to state that coverage cannot be canceled except after 30 days' prior Notice (or ten days in the case of cancellation for non-payment of premium) has been given to City. No policy language or endorsement shall include any limitation of liability of the insurer for failure to provide such Notice.
- (b) No Insurance Policy shall provide coverage on a "claims made" basis, with the exception of any professional liability Insurance Policies, unless specifically approved in writing by City.

#### 10.1.2.9 Requirements Not Limiting

The Parties acknowledge and agree that:

- requirements for specific coverage features or limits contained in this <u>Section</u> <u>10.1</u> (Insurance) and in <u>Exhibit 7</u> (Insurance Requirements) are not intended as a limitation on coverage, limits or other requirements, or a waiver of any coverage normally provided by any Insurance Policy;
- (b) specific reference to a given coverage feature is not intended to be all inclusive, or to the exclusion of other coverage, or a waiver of any type; and
- (c) all insurance coverage and limits provided by Principal Project Company, or by third parties pursuant to obligations of Principal Project Company under this Agreement, and, in each case, available or applicable to this Agreement are intended to apply to the full extent of the Insurance Policies, and nothing contained in this Agreement limits, or shall be deemed to limit, the application of such insurance coverage.

Except as otherwise specifically set forth in the Contract Documents, Principal Project Company may meet its insurance obligations in any manner Principal Project Company deems reasonably

appropriate, so long as, in each case, and with respect to the coverages prescribed for each Insurance Policy, Principal Project Company meets all the requirements therefor.

### 10.1.2.10 Deemed Self-Insurance by Principal Project Company

If, in any instance, Principal Project Company (i) has not performed its obligations respecting insurance coverage set forth in this Agreement or (ii) is unable to enforce and collect any such insurance for failure (x) to assert claims in accordance with the terms of the Insurance Policies or (y) to prosecute claims diligently, then, for purposes of determining Principal Project Company's liability and the limits thereon or determining reductions in compensation due from City to Principal Project Company on account of available insurance, Principal Project Company shall be treated as if it has elected to self-insure up to the full amount of insurance coverage which would have been available had Principal Project Company performed such obligations and not committed such failure.

### 10.1.2.11 Umbrella and Excess Policies

Principal Project Company shall have the right to satisfy the requisite insurance coverage amounts for liability insurance through a combination of primary policies and umbrella or excess policies. Umbrella and excess policies shall comply with all insurance requirements for the applicable type of coverage ("follow form").

### 10.1.2.12 Inadequacy of Required Coverage

City makes no representation that the scope of coverage and limits of liability specified for any Insurance Policy to be carried pursuant to this Agreement or approved variances therefrom are adequate to protect Principal Project Company against its undertakings under this Agreement to City, or its liabilities to any third party. It is the responsibility of Principal Project Company and each Subcontractor to determine if any changes or additional coverages are required to adequately protect their interests. No such limits of liability or approved variances therefrom shall preclude City from taking any actions as are available to it under this Agreement, or otherwise under applicable Law.

#### 10.1.2.13 No Relief from Liabilities and Obligations

Neither compliance nor failure to comply with the insurance provisions of this Agreement will relieve Principal Project Company or City of their respective liabilities and obligations under this Agreement.

#### 10.1.2.14 Adjustments in Coverage Amounts

- (a) At least once every two years during the Term (commencing initially on the Substantial Completion Date), City and Principal Project Company shall review and may, upon mutual agreement, adjust, as appropriate, the per occurrence and aggregate limits for the Insurance Policies that have stated dollar amounts set forth in <u>Exhibit 7</u> (Insurance Requirements).
- (b) In determining adjustments, Principal Project Company and City shall take into account (i) claims and loss experience for the Project, (ii) the physical condition of the Project, (iii) the Safety Compliance and Noncompliance Points

record for the Project, (iv) best practices in the insurance industry and (v) the terms of the Financing Documents.

(c) If a City Change to increase required limits of Insurance Policies results in a net increase in applicable insurance premiums, Principal Project Company shall be entitled to the actual amount of such net increase (without profit or mark-up), provided that to the extent such adjustments are made to reflect Principal Project Company's performance on the Project (including for reasons described in <u>Section 10.1.2.14(b)(i)</u>, (ii), or (iii)), Principal Project Company shall not be entitled to any compensation.

### 10.1.3 Insurance Premium Benchmarking

During the IFM Period, City will allocate the risk or benefit of increases and decreases in insurance premiums through an insurance benchmarking process as set forth in this <u>Section 10.1.3</u> (Insurance Premium Benchmarking). In no event shall City participate in any insurance premium risk associated with additional or extended coverages beyond those required under <u>Exhibit 7</u> (Insurance Requirements), or changes in premiums that are not the result of market-wide factors. The benchmarking process will occur as follows:

Coverage	Agreement Reference	Amount
Worker's Compensation and Employer's Liability	Exhibit 7, Section 2.1	[\$●] [Note: To be included based on Implementation Phase Proposal]
Commercial General Liability	Exhibit 7, Section 2.2	[\$●] [Note: To be included based on Implementation Phase Proposal]
Commercial Automobile Liability	Exhibit 7, Section 2.3	[\$●] [Note: To be included based on Implementation Phase Proposal]
Pollution Legal Liability	Exhibit 7, Section 2.4	[\$●] [Note: To be included based on Implementation Phase Proposal]
Property Insurance	Exhibit 7, Section 2.5	[\$•] [Note: To be included based on Implementation Phase Proposal]

(a) For the purposes of the benchmarking process, the "**Starting Insurance Benchmarking Premiums**" shall equal the following:

- (b) 60 days after the commencement date for the initial policies for the Required Minimum Insurance Policies and every two years thereafter during the IFM Period (each, a "Benchmarking Date"), Principal Project Company shall submit a report ("Insurance Review Report") to City that includes the following elements:
  - (i) Firm current quotes from three established and recognized insurance providers for the Required Minimum Insurance Policies; provided, however, that if three quotes are not available, Principal Project Company may, prior to the due date of Insurance Review Report, request a modification of this requirement, which shall be considered by City, in its good faith discretion. The quotes shall represent the current and fair market cost of providing the Required Minimum Insurance Policies.
  - With the exception of the first Benchmarking Date, copies of the premium invoices for the actual insurance policies obtained by Principal Project Company for the IFM Services for the prior two years ("Actual Insurance Policies").
  - (iii) With the exception of the first Benchmarking Date, a comprehensive written explanation of any effect that a PPC-Related Entity's loss experience has had on the premiums for the Required Minimum Insurance Policies and the Actual Insurance Policies. The explanation shall include: (A) an assessment by Principal Project Company's independent insurance broker addressing industry trends in premiums for the Required Minimum Insurance Policies and analysis (if applicable) of any Project-specific reasons for the increase in premiums; and (B) detailed analysis of any claims (paid or reserved) since the last report period, with claim date(s), description of incident(s), claims amount(s), and the level of deductibles provided.
- (c) City may independently assess the accuracy of the information in the Insurance Review Report and retains the right to perform its own independent insurance review, which may include retaining advisors, obtaining independent quotes for the Required Minimum Insurance Policies or performing its own assessment as to the impact of claims history on renewal costs.
- (d) The Starting Insurance Benchmarking Premiums shall be used in the benchmarking process for the remainder of the Term in accordance with the following procedures:
  - (i) 60 days after each Benchmarking Date, Principal Project Company shall provide the Insurance Review Report, with the information specified in <u>Section 10.1.3(b)</u>. City shall determine the change in premium costs on a coverage-by-coverage basis for the Required Minimum Insurance Policies calculated based on the information obtained from the initial Insurance Review Report or, if City deems

appropriate in its reasonable discretion, from information obtained pursuant to <u>Section 10.1.3(c)</u>.

- (ii) City will use the Starting Insurance Benchmarking Premiums to measure changes in premium costs at each Benchmarking Date for each of the Required Minimum Insurance Policies. The Starting Insurance Benchmarking Premiums shall be adjusted based on the percentage change in CPI ("Escalated Benchmark Insurance Premiums") from the Effective Date. Broker's/agent's fees and/or commissions will not be considered as part of the benchmarking exercise described in this <u>Section 10.1.3</u> (Insurance Premium Benchmarking), shall be identified and excluded from premiums, and are the exclusive responsibility of Principal Project Company.
- (iii) The subsequent Insurance Review Reports shall be used to establish the renewal premiums for the Required Minimum Insurance Policies for purposes of the benchmarking process described in this Section 10.1.3 (Insurance Premium Benchmarking). In no event shall premium increases that are caused by Project-specific losses, changes in deductibles or matters within the control of Principal Project Company or any PPC-Related Entity be subject to the benchmarking exercise or risk sharing described in this Section 10.1.3 (Insurance Premium Benchmarking). Principal Project Company may voluntarily choose to procure an insurance package which exceeds the Required Minimum Insurance Policies (with, for example, higher deductibles or coverage amounts, less exclusions, etc.), in which case Principal Project Company and City recognize that: (A) the actual variations in Principal Project Company's insurance premiums may not necessarily reflect the variations in the minimum insurance requirements and (B) City will disregard the actual insurance package and will rely upon the analysis from the Insurance Review Report and its own independent analysis of the effect on the minimum insurance requirements. However, any insurance beyond the Required Minimum Insurance Policies shall not be subject to the insurance benchmarking process and the MaxAP adjustment described in Section 10.1.3(e).
- (iv) If City elects to retain its own insurance advisor to analyze the extent of eligible premium increases, Principal Project Company shall cooperate in good faith with any reasonable requests for additional information from City's insurance advisor. No later than 30 days after Principal Project Company's submission of the Insurance Review Report, City shall make its determination of the eligible premium increases subject to the risk-allocation described in <u>Section 10.1.3(e)</u>.
- (e) As of a Benchmarking Date:
  - (i) if the aggregate annual insurance premiums for the Actual Insurance Policies, after adjustment for any changes in coverage and deductibles and as such premiums may be adjusted pursuant to

Section 10.1.3(d)(iii) (the "Actual Insurance Premiums") exceed 120% of the aggregate Escalated Benchmark Insurance Premiums, then City shall, within 30 days of the Benchmarking Date and on each 12 calendar month anniversary thereof until the next benchmarking period, make a lump sum payment to Principal Project Company in an amount equal to 85% of the Actual Insurance Premiums that are in excess of 120% of the aggregate Escalated Benchmark Insurance Premiums; and

- (ii) if the Actual Insurance Premiums are less than 80% of the aggregate Escalated Benchmark Insurance Premiums, then City shall, during each 12 calendar month period commencing on the Benchmarking Date until the next benchmarking period, reduce the payments owed to Principal Project Company in an amount equal to 85% of the amount by which Actual Insurance Premiums are lower than 80% of the aggregate Escalated Benchmark Insurance Premiums.
- (f) Any payments or reductions in payments to Principal Project Company pursuant to <u>Section 10.1.3(e)(i)</u> and <u>(ii)</u> shall be subject to a pro-rata adjustment for any period of less than 12 calendar months.
- (g) Principal Project Company shall maintain copies of the Actual Insurance Policies and the Insurance Review Reports and make these documents available upon City's request for the Term plus 10 years.

# 10.1.4 Insurance Unavailability

- **10.1.4.1** If an Insurance Unavailability risk occurs, then:
  - (a) Principal Project Company shall notify City within 10 days of becoming aware that the risk has become an Insurance Unavailability; and
  - (b) City will meet with Principal Project Company within 10 days after receipt of Principal Project Company's Notice to discuss the risk, including whether the risk is in fact Insurance Unavailability.

**10.1.4.2** If Principal Project Company demonstrates to City's reasonable satisfaction that it has used diligent efforts in the global insurance and reinsurance markets to procure the required Insurance Policy coverages for the Insurance Unavailability, and if, despite such diligent efforts and, through no fault of Principal Project Company or any other PPC-Related Entity, any Insurance Unavailability exists or occurs, the Parties shall meet further to discuss how the risk should be managed.

**10.1.4.3** If the Parties cannot agree on how to manage the Insurance Unavailability, then City, in its sole discretion, shall elect one of the following, and City's election shall be final and not subject to **Error! Reference source not found.** (Contract Dispute Procedures):

(a) compensate Principal Project Company for the costs of any Claim or liability incurred in connection with the Insurance Unavailability, up to an amount equal to the Insurance Proceeds that would have been payable had the

relevant Insurance Policy continued to be available on the previous terms of that Insurance Policy and deduct from the Availability Payment or Milestone Payments owing to Principal Project Company 100% of the greater of (i) the amount of insurance premiums Principal Project Company would have been obligated to pay under this Agreement (up to the Commercially Reasonable Insurance Rates) and (ii) the premiums assumed in the Financial Model;

- (b) if the Insurance Policies are available from insurers meeting the requirements in <u>Section 10.1.2.1</u> (Insurers), but not at Commercially Reasonable Insurance Rates, provide Notice to Principal Project Company to obtain the Insurance Policy and that City will be responsible for 100% of the premiums that exceed the Commercially Reasonable Insurance Rates;
- (c) provide Notice to Principal Project Company approving one or more variances from <u>Exhibit 7</u> (Insurance Requirements) such that the risk ceases to be Insurance Unavailability, in which case City will be entitled to a reduction in the Availability Payments equal to 100% of the insurance premiums that Principal Project Company avoids as a result of the variance from <u>Exhibit 7</u> (Insurance Requirements). In determining Principal Project Company's avoided insurance premiums, the Parties shall compare the actual premiums up to the greater of (i) the amount of insurance premiums Principal Project Company would have been obligated to pay for the relevant Insurance Policy had it been available under normal market conditions without variance from <u>Exhibit 7</u> (Insurance Requirements) or (ii) the premiums for the relevant Insurance Policy assumed in the Financial Model; or
- (d) terminate this Agreement by Notice to Principal Project Company, as further set out in <u>Section 17.2.6</u> (Termination for Insurance Unavailability).

# 10.1.5 Review of Insurance Unavailability Risks

**10.1.5.1** Whenever Principal Project Company has received information from its insurance adviser or other credible insurance industry source that Insurance Unavailability will likely exist during the next insurance renewal period, and annually during any period that Insurance Unavailability exists, Principal Project Company shall deliver Notice thereof to City and submit a report to City that includes the following elements:

- (a) evidence of Principal Project Company's efforts to obtain from at least three insurers meeting the requirements in <u>Section 10.1.2.1</u> (Insurers) the relevant Insurance Policy required to be maintained during the Term and an explanation of the reasons such efforts were unavailing;
- (b) a comprehensive assessment by Principal Project Company's independent insurance broker identifying the Insurance Unavailability risk, the reasons for unobtainability of insurance, and trends in insurance market conditions respecting the Insurance Unavailability risk;
- (c) a comprehensive written explanation and analysis of:

- any claims and loss experience (paid or reserved) with respect to any PPC-Related Entity or Affiliate, whether in connection with the Project or Work or in connection with any unrelated work or activity of a PPC-Related Entity or Affiliate, since the last review period, with claim date(s), description of incident(s), claims amount(s), and the level of deductibles or self-insured retentions provided; and
- (ii) the effect thereof on obtainability or unobtainability of the relevant Insurance Policies and coverage.

**10.1.5.2** City retains the right to independently assess the accuracy of the information on insurance markets, Insurance Unavailability and impacts of claims and loss experience, and retains the right to perform its own independent insurance review, which may include retaining advisors and seeking independent quotes for the Insurance Policies, all of which shall be undertaken and performed by City and any of its retained experts in good faith and a commercially reasonable manner.

**10.1.5.3** If City's review conducted in accordance with <u>Section 10.1.5.2</u> determines that the relevant Insurance Unavailability is insurable at Commercially Reasonable Insurance Rates, then Principal Project Company will promptly procure the insurance in connection with that risk in accordance with <u>Exhibit 7</u> (Insurance Requirements).

**10.1.5.4** Principal Project Company shall use commercially reasonable efforts to ensure that the process set forth in this <u>Section 10.1.5</u> (Review of Insurance Unavailability Risks) concludes sufficiently in advance of expiration of such Insurance Policies to ensure continuation of insurance coverage under renewed or replacement Insurance Policies. City will cooperate with Principal Project Company in its discharge of this obligation.

# 10.1.5.5 Defense Costs

Unless otherwise agreed to in writing by City, defense costs shall not erode the limits of coverage of any of the Insurance Policies, except that defense costs may be included within the limits of coverage of professional liability, contractor's pollution liability and pollution legal liability policies.

# 10.1.5.6 Contesting Denial of Coverage

If any Insurer under an Insurance Policy described in <u>Sections 10.1.1</u> (Insurance Policies and Coverage) and <u>10.1.3</u> (Insurance Premium Benchmarking) denies coverage with respect to any claims reported to such Insurer, Principal Project Company and City shall cooperate in good faith to establish whether and to what extent to contest, and how to fund the cost of contesting, the denial of coverage; provided that if the reported claim is a matter covered by an indemnity in favor of City or the denial is the result of Principal Project Company's failure to comply with an insurance requirement, then Principal Project Company shall bear all costs of contesting the denial of coverage.

# **10.1.6** Lender Insurance Requirements

If, under the terms of any Financing Agreement or Security Document, Principal Project Company is obligated to, and does, carry insurance coverage with higher limits, lower deductibles or self-insured retentions, or broader coverage than required under this Agreement, Principal Project Company's provision of such insurance shall satisfy the applicable requirements of this Agreement provided such policy meets all the other applicable requirements of this <u>Section 10.1</u> (Insurance) and <u>Exhibit 7</u> (Insurance Requirements). If Principal Project Company carries insurance coverage in addition to that required under this Agreement, then, except for any directors and officers liability insurance carried by Principal Project Company, Principal Project Company shall include the City Additional Insureds as additional insureds thereunder and shall provide to City the proofs of coverage and certificates, binders and copies of the policy as described in <u>Section 10.1.2.4</u> (Verification of Coverage). If, however, Principal Project Company demonstrates to City that inclusion of such City Additional Insureds as additional insureds will increase the premium, City shall elect either to pay the increase in premium or forego additional insured status.

# 10.1.7 Prosecution of Claims

**10.1.7.1** Unless otherwise directed by City in writing with respect to City's insurance claims, Principal Project Company shall be responsible for reporting and processing all potential claims by City or Principal Project Company against the Insurance Policies required to be provided by Principal Project Company under the Contract Documents. City will make reasonable efforts to report to Principal Project Company incidents that City's Authorized Representative has Actual Knowledge of and that may give rise to an insurance claim against the Insurance Policies required to be provided by Principal Project Company under the Contract Documents. City will report such incidents to Principal Project Company within a reasonable period of time after City's Authorized Representative actually becomes aware of such incidents. Principal Project Company agrees to report timely to the insurer(s) under such policies any and all matters which may give rise to an insurance claim by Principal Project Company or City and to promptly and diligently pursue such insurance claims in accordance with the claims procedures specified in such policies, whether for defense or indemnity or both. Principal Project Company shall enforce all legal rights against the insurer under the applicable Insurance Policies and applicable Laws in order to collect thereon, including pursuing necessary litigation and enforcement of judgments, provided that Principal Project Company shall be deemed to have satisfied this obligation if a judgment is not collectible through the exercise of lawful and diligent means.

**10.1.7.2** Principal Project Company shall immediately provide Notice to City, and thereafter keep City fully informed, of any incident, potential claim, claim or other matter of which Principal Project Company becomes aware that involves or could conceivably involve either City or City Additional Insureds as a defendant. City agrees to promptly provide Notice to Principal Project Company of City's incidents, potential claims, and matters which may give rise to a City insurance claim, to tender to the insurer City's defense of the claim (if applicable) under such Insurance Policies, and to cooperate with Principal Project Company as necessary for Principal Project Company to fulfill its duties hereunder.

**10.1.7.3** If, in any instance, Principal Project Company (i) has not performed its obligations respecting insurance coverage set forth in this Agreement, or (ii) is unable to enforce and collect any such insurance for failure (x) to assert claims in accordance with the terms of the Insurance Policies or (y) to prosecute claims diligently, then for purposes of determining Principal Project Company's liability and the limits thereon or determining reductions in compensation due from City to Principal Project Company on account of available insurance, Principal Project Company

shall be treated as if it has elected to self-insure up to the full amount of insurance coverage which would have been available had Principal Project Company performed such obligations. Nothing in this <u>Section 10.1.7</u> (Prosecution of Claims) or elsewhere in this <u>Section 10.1</u> (Insurance) shall be construed to treat Principal Project Company as electing to self-insure where Principal Project Company is unable to collect due to the bankruptcy or insolvency of any insurer which at the time the Insurance Policy is written meets the rating qualifications set forth in this <u>Section 10.1</u> (Insurance).

**10.1.7.4** In the event that an Insurer providing any of the Insurance Policies required by this Agreement becomes the subject of bankruptcy proceedings, becomes insolvent, or is the subject of an order or directive limiting its business activities given by any Governmental Entity, including the State Department of Insurance, Principal Project Company shall exercise best efforts to promptly and at its own cost and expense secure alternative coverage in compliance with the insurance requirements contained in this <u>Section 10.1</u> (Insurance) so as to avoid any lapse in insurance coverage.

**10.1.7.5** If, in any instance, Principal Project Company has not promptly performed its obligation to report to applicable insurers and process any potential insurance claim tendered by City, then City may report the claim directly to the insurer and thereafter seek coverage under the relevant policy.

# 10.1.8 Application of Insurance Proceeds

All insurance proceeds received for physical property damage to the Project under any Insurance Policies required under <u>Exhibit 7</u> (Insurance Requirements) shall be first applied to repair, reconstruct, rehabilitate, restore, renew, reinstate and replace each part or parts of the Project with respect to which such proceeds were received.

# **10.2** Performance Security

# 10.2.1 Equity Letter of Credit

Principal Project Company represents to City that if there is unfunded equity outstanding as of Financial Close, the Financing Documents will require each Equity Member to provide an Equity Letter of Credit to Principal Project Company or the Collateral Agent, the aggregate amount of which shall be at least equal to the unfunded equity outstanding as of Financial Close that is committed to the Project during the D&C Period.

# 10.2.2 D&C Performance Security

**10.2.2.1** On or before the Financial Close Date, Principal Project Company shall deliver the D&C Performance Security to City. The D&C Performance Security shall be comprised of the following ("**D&C Performance Security**"):

- (a) the D&C Performance Bond; and
- (b) the D&C Payment Bond.

**10.2.2.2** The D&C Performance Bond shall remain in full force and effect until:

- (a) The Final Acceptance Date;
- (b) There exists no PPC Default with respect to the D&C Work; and
- (c) No event has occurred that, with the giving of Notice or passage of time, or both, would constitute a PPC Default with respect to the D&C Work.
- **10.2.2.3** The D&C Payment Bond shall remain in full force and effect until:
  - (a) Receipt of evidence satisfactory to City that all Persons eligible to file a Claim under applicable Law against the D&C Payment Bond have been fully paid;
  - (b) Receipt of unconditional releases of Claims and stop Notices from all Subcontractors who have filed preliminary Notices of Claims against the D&C Payment Bond; and
  - (c) Expiration of the statutory period for Contractors to file a Claim against the D&C Payment Bond, if no claims have been filed.

#### **10.3 General Requirements for D&C Performance Security**

**10.3.1.1** Each D&C Performance Security shall be issued by an Eligible Surety or panel of Eligible Sureties.

**10.3.1.2** The D&C Performance Security will be subject to the rights of Lenders under the Direct Agreement.

**10.3.1.3** If the D&C Contract Amount is increased in connection with a Change Order, City may, in its sole discretion, require a corresponding and proportionate increase in the amount of each D&C Performance Security, or alternative security, as applicable.

**10.3.1.4** Principal Project Company agrees that it may not seek an injunction to restrain City from calling upon any D&C Performance Security.

**10.3.1.5** Unless otherwise specified in this Agreement, a draw on the D&C Performance Security or exercise of any rights under such D&C Performance Security will not be conditioned on prior resort to any other security of, or provided for the benefit of, Principal Project Company.

**10.3.1.6** Principal Project Company will pay all Recoverable Costs imposed in connection with City's exercise of its remedies against any D&C Performance Security or replacements thereof.

**10.3.1.7** Principal Project Company may:

 (a) procure the D&C Performance Security, so that they are security, as applicable, for (i) Principal Project Company's performance obligations under the Contract Documents respecting the D&C Work and (ii) Principal Project Company's payment obligations to the designated Persons supplying labor or materials respecting the D&C Work; or

Infrastructure Facility DBFOM Agreement

- (b) subject to this <u>Section 10.3</u> (General Requirements for D&C Performance Security), deliver D&C Performance Security from the Key Contractors for performance of any portion of the Work, so that each security is security for, as applicable, (i) performance of the Key Contractor's obligations under its Contract for D&C Work or (ii) payment to the designated Persons supplying labor or materials respecting the D&C Work.
- **10.3.1.8** If Principal Project Company makes the election under <u>Section 10.3.1.7(b)</u>, then:
  - (a) the language of the bond forms in <u>Exhibit 6B</u> (Form of D&C Payment Bond), <u>Exhibit 6C</u> (Form of D&C Performance Bond), <u>Exhibit 6D</u> (Form of Multiple Obligee Rider for D&C Payment Bond) and <u>Exhibit 6E</u> (Form of Multiple Obligee Rider for D&C Performance Bond) shall be adjusted to reflect this election, but only as necessary to (i) identify the Key Contract for D&C Work as the bonded contract, and (ii) identify the Key Contractor as the principal; and
  - (b) if there are two or more parties providing the D&C Performance Security, then the aggregate sum of the D&C Performance Security shall equal the required bond amounts under this Agreement and the size of each bond shall be in proportion to the scope and cost of the Work to be provided under each bonded Key Contract. Subject to the terms of this Agreement, City may proceed against any or both of such bonds in the order that City, in its sole discretion, determines.

#### 10.4 Letters of Credit

#### 10.4.1 General Provisions

Wherever in the Contract Documents Principal Project Company has the option or obligation to deliver to City a letter of credit, in addition to any specific requirements relating to a particular letter of credit, the following provisions shall apply.

**10.4.1.1** Except to the extent expressly provided otherwise in the Contract Documents, the letter of credit shall:

- (a) be a direct pay, irrevocable standby letter of credit;
- (b) be issued by a financial institution that is not an Affiliate, has a credit rating for long-term, unsecured debt of at least "A-" (or its equivalent) from one of the Rating Agencies, and has an office in Los Angeles, California, San Francisco, California, Chicago, Illinois, or New York, New York (or within reasonable proximity to these metropolitan areas) at which the letter of credit can be presented for payment. If the issuer's long-term, unsecured debt rating is downgraded such that the issuer no longer meets the ratings standard set forth above, Principal Project Company shall provide a replacement letter of credit issued by a financial institution meeting such standard within 30 days after the downgrade;

- (c) be payable on demand, conditioned only on written presentment from a beneficiary thereof to the issuer of a sight draft drawn on the letter of credit and a certificate stating that the beneficiary has the right to draw under the letter of credit in the amount of the sight draft, up to the amount due to such beneficiary, without requirement to present the original letter of credit;
- (d) be in place for the entire period of time for which the letter of credit is providing security;
- (e) allow for multiple draws;
- (f) name City as sole beneficiary; and
- (g) be consistent with the requirements of this <u>Section 10.4</u> (Letters of Credit).

**10.4.1.2** Except to the extent expressly provided otherwise in the Contract Documents, if Principal Project Company has failed to pay or perform when due the duty, obligation or liability under the Contract Documents for which the letter of credit is held, City has the right, subject to any rights of Lenders under the Direct Agreement, to draw on the letter of credit as and when provided in <u>Section 16.2.7</u> (Performance Bond). If City makes such a draw on the letter of credit, City shall use and apply the proceeds as provided in this Agreement for such letter of credit.

**10.4.1.3** Except to the extent expressly provided otherwise in the Contract Documents and the Direct Agreement, City has the right to draw on the letter of credit, without prior Notice to Principal Project Company, if (a) Principal Project Company has failed to pay or perform when due the duty, obligation, or liability under this Agreement for which the letter of credit is held, (b) for any reason Principal Project Company fails to deliver to City a new or replacement letter of credit, on the same terms, at least 30 days before the expiry of the letter of credit, unless the applicable terms of the Contract Documents expressly provide that no further letter of credit is required with respect to such duty, obligation or liability, or (c) the financial institution issuing the letter of credit fails to meet the requirements in <u>Section 10.4.1.1(b)</u> and Principal Project Company fails to grave a substitute letter of credit issued by a qualified financial institution within 30 days after the downgrade. If City makes such a draw on the letter of credit, City shall be entitled to draw on the full face amount of the letter of credit and shall retain such amount as cash security to secure the obligations under the letter of credit, without payment of interest to Principal Project Company.

**10.4.1.4** Except to the extent expressly provided otherwise in the Contract Documents and the Direct Agreement, a draw on letters of credit shall not be conditioned on prior resort to Principal Project Company or any other security of Principal Project Company. For all draws conditioned on prior Notice from City to Principal Project Company, no such Notice shall be required if it would preclude draw before the expiration date of the letter of credit. City will use and apply draws on letters of credit (or cash security held from draws on letters of credit) toward satisfying the relevant obligation of Principal Project Company (or, if applicable, any other Person for which the letter of credit is performance security). Subject to City's rights under <u>Sections</u> 10.4.1.2 and 10.4.1.3, if City receives proceeds of a draw in excess of the relevant obligation, City will promptly refund the excess to Principal Project Company (or such other Person) after all relevant obligations are satisfied in full.

**10.4.1.5** Except to the extent expressly provided otherwise in the Contract Documents, Principal Project Company's sole remedy in connection with the improper presentment or payment of sight drafts drawn under letters of credit shall be to obtain from City a refund of the proceeds which are misapplied, reimbursement of the reasonable costs Principal Project Company incurs as a result of such misapplication; provided that at the time of such refund Principal Project Company increases the amount of the letter of credit to the amount (if any) then required under applicable provisions of this Agreement. Principal Project Company acknowledges that the presentment of sight drafts drawn upon a letter of credit could not under any circumstances cause Principal Project Company injury that could not be remedied by an award of money damages, and that the recovery of money damages would be an adequate remedy. Accordingly, Principal Project Company covenants (a) not to request or instruct the issuer of any letter of credit to refrain from paying any sight draft drawn under the letter of credit and (b) not to commence or pursue any legal proceeding seeking, and Principal Project Company irrevocably waives and relinquishes any right, to enjoin, restrain, prevent, stop or delay any draw on any letter of credit.

**10.4.1.6** Principal Project Company shall obtain and furnish all letters of credit and replacements thereof at its sole cost and expense, and shall pay all charges imposed in connection with City's presentment of sight drafts and drawing against letters of credit or replacements thereof.

**10.4.1.7** If City makes a permitted assignment of its rights and interests under this Agreement, then Principal Project Company shall cooperate so that concurrently with the effectiveness of such assignment, either replacement letters of credit for, or appropriate amendments to, the outstanding letters of credit shall be delivered to the assignee naming the assignee as beneficiary, at no cost to Principal Project Company.

**10.4.1.8** City acknowledges that if the letter of credit is performance security for a Person other than Principal Project Company (e.g., a Key Contractor), City's draw may only be based on the underlying obligations of such Person.

# 10.5 Guarantees

10.5.1 If Principal Project Company, any Key Contractor, any Affiliate or any Lender receives from any Person a guaranty of payment or performance of any obligation(s) of a Key Contractor, then either Principal Project Company shall cause such Person to expressly include City as a guaranteed party under such guaranty, with the same protections and rights of Notice, enforcement and collection as are available to any other guaranteed party, and deliver to City a duplicate original of such guaranty, which guaranty shall provide that the rights and protections of City shall not be reduced, waived, released or adversely affected by the acts or omissions of any other guaranteed party, other than through the rendering of payment and performance to another guaranteed party; and upon receipt of written Notice from City of the occurrence of the circumstances described in Section 10.5.2, such other documents reasonably satisfactory to City permitting City, subject to the rights of the Collateral Agent under any Direct Agreement, to become the transferee beneficiary under such guaranty and to enforce it, including enforcing the guaranty in favor of City or the Project, or both, which transfer documents shall include a certified copy of the guaranty and an executed transfer and assignment of the beneficiary rights from Principal Project Company or Collateral Agent, as applicable, to City; and the guaranty

shall expressly authorize such transfer without condition and permit draw without presentation of the original guaranty.

**10.5.2** City's rights as a transfer beneficiary are exercisable if, subject to <u>Section 10.5.3</u> and the Direct Agreement, City determines that (a) a Key Contractor has breached or failed to perform any obligations under its Contract and any Notice thereof required under such Contract has been provided and the applicable cure period has expired without full and complete cure, (b) such breach has caused a PPC Default and the applicable cure period has expired without full and complete cure and (c) Principal Project Company or the Collateral Agent has failed to call upon or otherwise enforce such guaranty for the purpose of causing the performance of such obligations by or on behalf of the Contractor within 10 days after City delivers Notice of such breach or expected breach to Principal Project Company and the Collateral Agent and the Cure Period (as defined in the Direct Agreement) has expired.

**10.5.3** So long as Principal Project Company or a Lender is diligently pursuing remedies under a guaranty, City agrees to forbear from (a) exercising remedies under any such guaranty that names City as a direct beneficiary, and (b) exercising its right to become a beneficiary under <u>Section 10.5.1</u>; provided, however, that if the PPC Default giving rise to exercise remedies under any such guaranty remains uncured at the end of the applicable cure period in <u>Section 16.1.2</u> (Default Notice and Cure Periods), City's obligation to forbear from exercising remedies as a guaranteed party shall cease. The foregoing shall not obviate any agreement by City to forbear from exercising its rights and remedies contained in a Direct Agreement.

### 10.6 Indemnities

## 10.6.1 General Indemnity

**10.6.1.1** Subject to <u>Section 10.6.3</u> (Limitations on Indemnification Obligations), Principal Project Company shall defend, indemnify, protect and hold harmless the Indemnitees from and against any and all claims, causes of action, suits, investigations, legal or administrative proceedings, demands and Losses arising out of or in connection with:

- (a) any alleged or actual PPC Fault, if asserted or incurred by or awarded to any Third Party or any PPC-Related Entity;
- (b) Losses to the Infrastructure Facility, and any interference, disruption, or delay to the Project, caused by or related to the work of any JDA-Related Entity; provided, however, that the indemnity set forth in this clause (b) shall not apply to the D&C Work performed by the D&C Contractor to the extent that the City commences construction of a Joint Development Alternative prior to the earlier of (i) Substantial Completion; or (ii) the Substantial Completion Deadline, unless, in either case, the prime contractor undertaking such construction is the D&C Contractor or an affiliate thereof (in which case, the indemnity shall apply);
- damage to public or private property owned by Third Parties (or any PPC-Related Entity), and for injuries to any person or entity, arising out of performance of the Project or Work by any PPC-Related Entity;

- (d) any alleged intellectual property infringement or other allegedly improper appropriation or use of intellectual property by any PPC-Related Entity in performance of the Project or the Work, or in connection with the Infrastructure Facility;
- (e) any and all claims by any governmental or taxing authority claiming taxes based on gross receipts, purchases or sales, or the use of any property or income of any PPC-Related Entity with respect to any payment for the Project or Work made to or earned by any PPC-Related Entity;
- (f) the failure or alleged failure by any PPC-Related Entity to pay sums due for the Work or services of Contractors, laborers, or suppliers;
- (g) any actual or threatened PPC Release;
- (h) the claim or assertion by any Other Contractor or a Utility that any PPC-Related Entity (i) failed to cooperate reasonably with such party, so as to cause interference, disruption, delay or loss; or (ii) interfered with or hindered the progress or completion of work being performed by such Other Contractor or Utility, so as to cause interference, disruption, delay or loss, to the extent such claim arises out of any PPC Fault;
- (i) any PPC-Related Entity's breach of or failure to perform an obligation that City owes to a Third Party, including any Governmental Entity and any Utility, under applicable Law or under any agreement between City and a Third Party, where City has delegated performance of the obligation to Principal Project Company under this Agreement or the acts or omissions of any PPC-Related Entity which render City unable to perform or abide by an obligation that City owes to a Third Party, including any Governmental Entity and any Utility, under any agreement between City and a Third Party, where, in each case, the agreement was expressly disclosed or known to Principal Project Company;
- (j) inverse condemnation, trespass, nuisance or similar taking of or harm to real property by reason of (i) the failure of any PPC-Related Entity to comply with Good Industry Practice, requirements of this Agreement, the Project Management Plan or Regulatory Approvals respecting control and mitigation of construction activities and construction impacts, (ii) the intentional misconduct or negligence of any PPC-Related Entity in connection with the performance of the Project or the Work, or (iii) the actual physical entry onto or encroachment upon another's property by any PPC-Related Entity in connection with the performance of the Project or the Work;
- (k) errors or other Defects in the supply, construction (including installation), operation or maintenance of the Project (except, subject to <u>Section 6.11.1</u>, with respect to operation and maintenance, of the SFMTA O&M Facilities after Substantial Completion or of Utility Adjustments included in the Project or the Work;

- (I) Design Work that fails, in whole or in part, to meet the requirements of this Contract; and
- (m) Any Principal Project Company failure to implement environmental mitigation measures to control environmental impacts, as required by the Governmental Approvals, the CEQA MMRP and the NEPA document.

**10.6.1.2** Principal Project Company's responsibilities pursuant to this <u>Section 10.6.1</u> (General Indemnity) include both the obligation to (a) defend the Indemnitees from and against any and all claims, causes of action, suits, investigations, legal or administrative proceedings, demands and Losses, and (b) indemnify the Indemnitees when liability is sustained pursuant to the Contract Dispute Procedures or through judicial proceedings or is mutually agreed upon by the Parties.

**10.6.1.3** Principal Project Company's indemnification shall include reasonable fees of attorneys, consultants, and experts and related costs and City's costs of investigating any claims against City. In addition to Principal Project Company's obligation to indemnify City, Principal Project Company specifically acknowledges and agrees that it has an immediate and independent obligation to defend City from any claim which actually or potentially falls within this indemnification provision, even if the allegations are or may be groundless, false or fraudulent, which obligation arises at the time such claim is tendered to Principal Project Company by City and continues at all times thereafter.

**10.6.1.4** Principal Project Company's responsibilities pursuant to this <u>Section 10.6.1</u> (General Indemnity) are in addition to any right that City may have under the terms of this Agreement to assess Noncompliance Points and/or Deductions with respect to the same event or circumstance giving rise to a Third Party claim, cause of action, suit, legal or administrative proceeding.

**10.6.1.5** Principal Project Company's defense, indemnity, and hold harmless obligations shall extend to City's consultants (e.g., design professionals and construction managers) providing services covering any portion of the Project under a separate written agreement with City and designated in the Contract Documents as persons or entities to be listed on Principal Project Company's insurance policies as "Additional Insureds." Principal Project Company's defense, indemnity, and hold harmless obligations shall not extend to the liability of a City consultant (including architects and engineers) designated as an Indemnitee or its agents, employees, or subconsultants arising out of, connected with or resulting from such Indemnitee's own active negligence, willful misconduct, bad faith, fraud, errors or omissions or from such Indemnitee's preparation or approval of maps, plans, opinions, reports, surveys, change orders, designs or specifications, or such Indemnitee's issuance of or failure to issue directions or instructions provided that such issuance or failure to issue is the primary cause of the damage or injury.

# 10.6.2 Design Defects

**10.6.2.1** Principal Project Company agrees that, because the Reference Documents are subject to review and modification by Principal Project Company, (a) it is appropriate for Principal Project Company to assume liability for errors, omissions, inconsistencies and other Defects in the completed Project even though they may be related to errors, omissions, inconsistencies and other Defects in the Reference Documents, and (b) such documents shall

not be deemed "design furnished" by City or any of the other Indemnitees, as the term "design furnished" is used in Civil Code section 2782. Principal Project Company hereby waives the benefit (if any) of Civil Code section 2782 and agrees that this <u>Section 10.6.2</u> (Design Defects) constitutes an agreement governed by Civil Code section 2782.5.

**10.6.2.2** Subject to <u>Section 10.6.3</u> (Limitations on Indemnification Obligations), Principal Project Company shall indemnify, defend and hold harmless the Indemnitees from and against any and all claims and Losses arising out of, relating to or resulting from errors, omissions, inconsistencies or other defects in the Design Documents, regardless of whether such errors, omissions, inconsistencies or other defects were also included in the Reference Documents.

# **10.6.3** Limitations on Indemnification Obligations

**10.6.3.1** Subject to <u>Section 23.8</u> (Limitation on Third Party Beneficiaries) and the releases and disclaimers herein, including all the provisions set forth in <u>Section 5.1.3</u> (Limitations on Principal Project Company's Right to Rely), Principal Project Company's indemnity obligations shall not extend to any claims, suits, actions or Losses to the extent directly caused by:

- (a) the active negligence, gross negligence, reckless or willful misconduct, bad faith or fraud of an Indemnitee;
- (b) a City-Caused Delay Event or a City-Caused Relief Event; or
- (c) City's breach of any of its obligations under the Contract Documents.

**10.6.3.2** With respect to Work performed by a design professional as defined in California Civil Code section 2782.8, such indemnities shall apply only to the extent permitted by section 2782.8 as of the Effective Date.

# 10.6.3.3 Claims by Employees

In claims by an employee of a PPC-Related Entity, the indemnification obligation under this <u>Section 10.6</u> (Indemnities) shall not be limited by any limitation on the amount or type of damages, compensation or benefits payable by or for a PPC-Related Entity under workmen's compensation, disability benefit or other employee benefits laws; provided that this provision shall not be construed as a waiver in favor of any employee by Principal Project Company or any Contractor of any limitation of liability afforded by such laws.

# 10.6.3.4 Indemnity as Alternative Cause of Action

The requirement to provide an indemnity as specified in this <u>Section 10.6</u> (Indemnities) is not intended to provide City with an alternative cause of action against Principal Project Company for damages for breach of contract incurred directly by Indemnitees in connection with the event giving rise to the indemnification obligation.

# 10.6.4 Principal Project Company's Defense

In Principal Project Company's defense of Indemnitees under this <u>Section 10.6</u> (Indemnities), negotiation, compromise, and settlement of any action, City shall, without prejudice to the rights of any Indemnitees to be indemnified by Principal Project Company, retain reasonable

discretion in and control of the litigation, negotiation, compromise, settlement, and appeals therefrom.

### 10.7 Indemnities by Contractors

Principal Project Company shall ensure that each Contract includes indemnity provisions appropriate to the scope of the Work to be performed by the Contractor, naming the Indemnitees as indemnitees.

#### 10.8 Notice of Claims by Third Parties

**10.8.1** If the City receives Notice of a claim, cause of action, suit, legal or administrative proceeding covered by the indemnities in <u>Section 10.6</u> (Indemnities), or otherwise has Actual Knowledge of such a claim, cause of action, suit, legal or administrative proceeding that it believes is within the scope of the indemnities under <u>Section 10.6</u> (Indemnities), as soon as practicable after receipt of the claim, cause of action, suit, legal or administrative proceeding, City shall:

- (a) inform Principal Project Company in writing of the claim, cause of action, suit, legal or administrative proceeding, and
- (b) send to Principal Project Company a copy of all relevant written materials City has received asserting such claim, cause of action suit, legal or administrative proceeding.

**10.8.2** As soon as is practicable after Principal Project Company receives Notice of a claim, cause of action, suit, legal or administrative proceeding covered by the indemnities in <u>Section</u> <u>10.6</u> (Indemnities), Principal Project Company shall promptly provide Notice to City in writing and, unless subject to evidentiary privilege, promptly furnish to City copies of all factual reports and factual portions of any other reports given to Principal Project Company's insurance carrier or carriers.

# 10.9 SFMTA O&M Facilities Warranty Bond

**10.9.1** Upon achieving Final Acceptance, Principal Project Company shall provide a warranty bond to City in an amount equal to 20% of the value of the SFMTA O&M Facilities in the form attached hereto as <u>Exhibit 6F</u> (Form of SFMTA O&M Facilities Warranty Bond), with such nonmaterial modifications, if any, as City may approve in its sole discretion (the "SFMTA O&M Facilities Warranty Bond").

**10.9.2** The SFMTA O&M Facilities Warranty Bond shall guarantee performance of Work on the SFMTA O&M Facilities required to be performed during the period following Final Acceptance, including warranty Work, which shall also constitute a payment bond guaranteeing payment to Persons performing such Work.

**10.9.3** City will release the SFMTA O&M Facilities Warranty Bond upon the expiration of the warranty period set forth in <u>Section 6.11.1.1</u>; provided, however, that all of the following conditions have been met: (a) Principal Project Company is not in default under the Contract Documents and no event has occurred which, with the passage of time or the giving of Notice, would constitute a default under the Contract Documents, (b) receipt by City of (i) evidence

satisfactory to City that all Persons eligible to file a Third Party claim against the SFMTA O&M Facilities Warranty Bond have been fully paid and (ii) unconditional releases of liens and Notices from all Contractors who filed preliminary notices of a Third Party claim against the SFMTA O&M Facilities Warranty Bond, using applicable California statutory forms, (c) the statutory period for Subcontractors to file a Third Party claim against the SFMTA O&M Facilities Warranty Bond has expired and no Third Party claims have been filed, and (d) Principal Project Company assigns to City any Contractor and Supplier warranties that may still be in effect as of the effective date of the expiration of the warranty period set forth in <u>Section 6.11.1.1</u>.

# ARTICLE 11. PAYMENTS TO PRINCIPAL PROJECT COMPANY

### 11.1 Milestone Payment

**11.1.1** Subject to any limitations and exceptions expressly provided in this Agreement, City will pay the Milestone Payment to Principal Project Company upon achievement of Substantial Completion, calculated and otherwise in accordance with the process in <u>Exhibit 4A</u> (Milestone Payment Mechanism).

### 11.2 Availability Payments

**11.2.1** Commencing from the Substantial Completion Date and subject to any limitations and exceptions expressly provided in this Agreement, City will make Availability Payments to Principal Project Company as provided in this <u>Section 11.2</u> (Availability Payments) and <u>Exhibit 4B</u> (Availability Payment Mechanism). Principal Project Company is not entitled to earn any Availability Payments before the Substantial Completion Date.

**11.2.2** City will make Availability Payments to Principal Project Company through Quarterly Availability Payments calculated and otherwise in accordance with <u>Exhibit 4B</u> (Availability Payment Mechanism).

**11.2.3** Principal Project Company acknowledges and agrees that any Availability Payment or portion thereof not received by Principal Project Company as a result of a delay in achieving Substantial Completion for which Principal Project Company is not entitled to compensation under this Agreement represents the liquidated amount of delay damages suffered by City due to such delay.

#### 11.3 Pass-Through Costs

**11.3.1** City shall reimburse PPC for Pass-Through Costs during the IFM Period as set out in this Section 11.3.

**11.3.2** PPC shall deliver to the City Authorized Representative copies of each invoice, which must be in a format reasonably approved by City, and receipt of payment of Pass-Through Costs issued to PPC.

**11.3.3** PPC's invoices for Pass-Through Costs may not include any mark-up, profit or overhead on costs paid or payable by PPC to Contractors or utility owners. Such mark-up, profit and overhead together with PPC's reasonable expenses for general administration, insurance premiums, overhead and other business expenses that relate to the provision of services rendered by PPC in connection with Pass-Through Costs shall not be considered as Pass-Through Costs and are to be compensated as part of the Availability Payment.

#### 11.4 Invoice, Other Amounts and Payments

**11.4.1** Prior to issuing any invoice required by this Agreement, Principal Project Company must submit to City a form invoice to be approved by City, and any invoice issued in accordance with this Agreement must be substantially in the form agreed by the City.

**11.4.2** Principal Project Company shall submit an invoice (i) for the Availability Payment with the Quarterly Report delivered to City in accordance with the IFM Specifications no later than the 10th day of the month immediately following the relevant Contract Quarter; (ii) for any other proposed payment adjustments in accordance with this Agreement, including Milestone Payments Compensable Delay Events, Compensable Relief Events or Termination Compensation (collectively, "**Other Amounts**") no later than the 10th day of the month immediately following the applicable event triggering such invoice, unless a different time period is set out in the Agreement for such event. All invoices shall include any supporting mathematical corrections or reconciliations to enhance the accuracy of payments due to Principal Project Company under this Agreement, to the satisfaction of City.

**11.4.3** City shall review each properly submitted invoice within 7 days of receipt. If City, in its sole discretion, determines that the invoice is improper, the invoice will be returned to Developer not later than 7 days after receipt, along with a document setting forth the reasons why the invoice is not proper.

**11.4.4** Unless otherwise specified in this Agreement, City will pay PPC any amount owing under this Agreement within 45 days after receipt of a properly submitted invoice for such payment.

# 11.5 Disputed Amounts

**11.5.1** City may dispute, in good faith, any amount specified in an invoice submitted under <u>Section 11.4</u> (Invoice, Other Amounts and Payments), or any other invoice submitted by Principal Project Company under this Agreement. City shall pay all undisputed amounts for which payment is requested and that are not subject to withholding in accordance with Section <u>11.4</u> (Invoice, Other Amounts and Payments).

**11.5.2** Principal Project Company and City shall use reasonable efforts to resolve any invoice dispute within 30 days after the dispute arises.

# 11.6 Withholding from Payments

**11.6.1** City may deduct from any payment owing to Principal Project Company under this Agreement or make a demand of Principal Project Company for:

- (a) any amount due and payable by Principal Project Company to City (whether in connection with this Agreement or any other Contract Document);
- (b) any sums expended by City in performing any of Principal Project Company's obligations under this Agreement which Principal Project Company has failed to perform; and
- (c) amounts in respect of any other Claim or Losses by City against Principal Project Company in connection with the Work, the IFM Services or the Project.

**11.6.2** The failure by City to deduct any of the sums under this <u>Section 11.6</u> (Withholding from Payments) from a payment must not constitute a waiver of City's right to such sums.

# 11.7 Interest on Late Payments and Overpayments

**11.7.1** If Principal Project Company fails to pay any undisputed amount due and owing from Principal Project Company to the City under this Agreement, Principal Project Company shall pay to City interest on such amount at the Late Payment Rate commencing 90 days after the due date thereof until the date of payment.

**11.7.2** If any properly submitted invoice is disputed and an amount is determined to be due under the Contract Dispute Procedures, payment of the disputed amount shall be made within 30 Business Days following resolution of the dispute, together with interest at the Late Payment Rate on the amount owing from the date that the payment was originally due (based on the agreement of the parties or the decision of the dispute resolver) until the date of payment.

**11.7.3** If as a result of any inaccuracy in an invoice any overpayment is made by City to Principal Project Company then, in addition to the adjustments provided in <u>Section 11.2</u> (Availability Payments), City shall be entitled to deduct or receive as a payment from Principal Project Company interest on such amount at the Late Payment Rate, starting on the date of City's payment of the invoice to the date the overpayment is deducted or paid. City's right to deduct or receive payment of interest is without prejudice to any other rights City may have under this Agreement.

### 11.8 Taxes

Except as provided in <u>Section 2.4.2</u> (Possessory Interest Tax), Principal Project Company shall pay all applicable Taxes on or before the due date (or delinquency date if applicable). Principal Project Company is solely responsible for and has no right to make any Claim due to its misinterpretation of laws respecting Taxes or incorrect assumptions regarding applicability of Taxes. In the event that an exemption from applicable sales or use taxes becomes available for the Project, City shall have no obligation to reimburse Principal Project Company for any such taxes, and City shall be entitled to an upfront payment from Principal Project Company or a reduction in payments made by City, as agreed upon by the Parties, equal to the amount actually saved following the date such exemption becomes available.

#### 11.9 Payment Not Evidence of Approval

No payment by City is or must be construed as:

- (a) evidence of the value of Work or that Work has been satisfactorily carried out in accordance with this Agreement;
- (b) an admission of liability by City;
- (c) approval by City of Principal Project Company's performance or compliance with this Agreement;
- (d) acknowledgement that City has inspected or accepted the Work; or
- (e) waiver of any Claim or right that City may then or thereafter have, including among others, warranty and indemnity rights.

# 11.10 Other Adjustments; Full Compensation

Principal Project Company acknowledges and agrees:

- (a) the Milestone Payment and Availability Payments calculated in accordance with this <u>Article 11</u> (Payments to Principal Project Company) and <u>Exhibit 4</u> (Payment Mechanism) are subject to adjustment to reflect previous overpayments and/or under-payments, any interest payable in respect of any amounts owed, and any other amount due and payable from Principal Project Company to City or from City to Principal Project Company under this Agreement, including credits, deductions or offsets for failure to meet Performance Requirements or pursuant to the Noncompliance and Deduction regime; and
- (b) that the payments provided for in this <u>Article 11</u> (Payments to Principal Project Company) constitute full compensation for performance of all the Work, subject only to Principal Project Company's rights under <u>Articles 12</u> (City Change Process; Unilateral Change Orders; Deviations), <u>13</u> (General Provisions Applying to Delay Events and Relief Events), <u>14</u> (Compensation and Other Relief for Delay Events and Relief Events) and <u>17</u> (Termination).

# 11.11 Appropriation; Certification of Funds

**11.11.1** All payments due from City to Principal Project Company under this Agreement, including any Termination Compensation, shall be paid solely from monies made available to City from an appropriation of funds for the purpose of making all such payments coming due in such fiscal year. SFMTA, or the San Francisco Board of Supervisors, as applicable, shall have the absolute and unconditional right, to be exercised in their discretion, for any reason, not to appropriate such funds.

**11.11.2** This Agreement is subject to the fiscal provisions of the City's Charter and the budget decisions of its Mayor and Board of Supervisors, each acting in its sole discretion. No funds will be available hereunder until prior written authorization certified by the City's Controller. The City's Controller cannot authorize payments unless funds have been certified as available in the budget or in a supplemental appropriation. City shall use commercially reasonable efforts to obtain the certification from the City's Controller if the funds have been appropriated for such purpose. Without prejudice to Principal Project Company's rights and remedies as set forth in this Agreement for unexcused and undisputed non-payment by City of any payment due Principal Project Company when due, City's obligations hereunder shall never exceed the amount certified by the City's Controller for the purpose and period stated in such certification. City, its employees and officers are not authorized to offer or promise any additional funding without City's Controller certification of such additional funding. Without such lawful approval and certification, City shall not be required to provide such additional funding.

# **11.11.3** City shall:

(a) make a timely submission for the fiscal year in which Substantial Completion of the Infrastructure Facility will occur of a budget proposal to the SFMTA

Board of Directors requesting a budget that includes the Milestone Payment and initial Availability Payment;

- (b) for each fiscal year beyond the fiscal year in which the initial Availability Payment is made, make a timely submission of a budget proposal to the SFMTA Board of Directors requesting a budget for each and every Availability Payment, including any subsequent change to the Maximum Availability Payment; and
- (c) request an appropriation of funds for the purpose of paying any Termination Payment payable by City or any other amount due from City under this Agreement other than the Availability Payments.

**11.11.4** City shall respond promptly in writing to any reasonable written request submitted by Principal Project Company for information regarding the status of any request City is obligated to make under <u>Section 11.11.3</u>.

**11.11.5** If City fails to appropriate money to make a Termination Payment, there shall be no contractual obligation of the City to make such payment that may be enforced; such contractual payment obligation shall arise only if and when any such amounts have been appropriated by City.

**11.11.6** City shall provide written Notice to Principal Project Company no later than ten Business Days following the enactment of any City budget with respect to a particular Fiscal Year that does not make an appropriation of funds for the purpose of paying the scheduled Availability Payments or Milestone Payment due for such Fiscal Year. City shall consult with Principal Project Company to discuss the situation and the possible solutions, it being understood that such discussions shall be without prejudice to Principal Project Company's right to termination for unexcused and undisputed non-payment by City of any payment due Principal Project Company when due, as set forth in <u>Section 17.4.1</u> (Termination for City Default).

**11.11.7** The obligation of City to make any payments under this Agreement does not constitute a debt of the City under applicable Law and does not constitute a liability of or a lien or charge upon the funds or property of the City beyond the fiscal year for which there has been an appropriation of funds to make such payments. The obligation of City to make payments hereunder does not constitute an obligation of City for which City is obligated to levy or pledge any form of taxation or for which City has levied or pledged any form of taxation.

# 11.12 Allowances

# [NOTE TO PNC: ALLOWANCE AMOUNTS UNDER DEVELOPMENT; CITY-FURNISHED IT/COMMS ALLOWANCE MAY BE JUST DISBURSED AND HELD FOR CITY IT/COMMS]

# 11.12.1 Allowances, Generally

(a) Each of the Allowances is available to pay for certain portions of the D&C Work, with payments from the Allowance to be made on the basis identified with respect to each such Allowance under this <u>Section 11.12</u> (Allowances).

- (b) Principal Project Company shall be entitled to use Office/Admin and Training Spaces FF&E Allowance as set forth in this <u>Section 11.12</u> (Allowances).
- (c) City shall be entitled to be reimbursed from the City-Furnished IT/Comms Allowance for amounts incurred with respect to procurement, purchase and installation of the IT/Comms Equipment. To the extent that City directs Principal Project Company to procure, purchase or install any City-Furnished IT/Comms FF&E, the provisions in <u>Section 11.11.1(b)</u>, (e) and (g) shall apply to the City-Furnished IT/Comms FF&E Allowance.
- (d) The initial amount of each Allowance is identified in <u>Section 11.12</u> (Allowances). Principal Project Company acknowledges and agrees that the Milestone Payment and the Availability Payment include the Allowances and represent full compensation to Principal Project Company on account of the Allowances.
- (e) Principal Project Company shall keep detailed records of the quantities, units, or other agreed metrics with respect to the Office/Admin and Training Spaces FF&E Allowance and shall submit to City supporting documentation of such quantities with its invoices, and such other information as City may require, in its sole discretion.
- (f) Notwithstanding that the Allowances have been developed for specific elements of the D&C Work, City, in its sole discretion, may elect to use some or all of any Allowance as a source of payment for D&C Work for which Principal Project Company may be entitled under another Allowance or in connection with a Compensable Delay Event.
- (g) No Change Order is required for invoicing amounts remaining (with respect to relevant portions of the D&C Work) within any Allowance. Principal Project Company shall promptly provide Notice to City if it becomes apparent that the amount with respect to Office/Admin and Training Spaces FF&E Allowance will be exceeded, in which event the Parties shall negotiate a Change Order increasing the Office/Admin and Training Spaces FF&E Allowance and/or modifying the scope of the D&C Work to avoid the need to increase the Office/Admin and Training Spaces FF&E Allowance; provided, however, that in lieu of modification to the Office/Admin and Training Spaces FF&E Allowance amount and/or scope of D&C Work, City may, in its sole discretion, issue a Change Order, and such D&C Work otherwise eligible for invoicing against the Office/Admin and Training Spaces FF&E Allowance will not be invoiced against the Office/Admin and Training Spaces FF&E Allowance, but, instead, be invoiced under such Change Order, and such Change Order shall be on the same unit and other pricing terms as if the D&C Work were invoiced against the Office/Admin and Training Spaces FF&E Allowance.
- (h) If the amount with respect to Office/Admin and Training Spaces FF&E Allowance will be exceeded, City will bear the cost of such items.

(i) As part of Final Acceptance, all remaining amounts of any Allowance shall be refunded and paid to City.

#### 11.12.2 Office/Admin and Training Spaces FF&E Allowance

Principal Project Company shall be eligible to invoice against the Office/Admin and Training Spaces FF&E Allowance to address City-approved Office/Admin and Training Spaces FF&E, as set forth herein.

- (a) The Office/Admin and Training Spaces FF&E Allowance shall be in an initial amount of \$6,220,000.
- (b) The Office/Admin and Training Spaces FF&E Allowance shall be used for Office/Admin and Training Spaces FF&E.
- (c) For all Office/Admin and Training Spaces FF&E, Principal Project Company shall obtain at least three arms-length competitive price quotes (or, if Principal Project Company believes it is impractical to obtain such number of competitive price quotes, Principal Project Company may submit a request for waiver or modification of the competitive price quote requirement to City, which shall consider such request in its good faith discretion).
- (d) Invoicing for completed Office/Admin and Training Spaces FF&E Work shall be included in the monthly invoice described in <u>Section 11.3</u> (Invoice, Other Amounts and Payments).
- (e) This <u>Section 11.11.2</u> (Office/Admin and Training Spaces FF&E Allowance) shall not apply to, and the Office/Admin and Training Spaces FF&E Allowance shall not be useable for, repair and replacement of Office/Admin and Training Spaces FF&E that is damaged, defective or missing prior to Substantial Completion.

#### 11.12.3 City-Furnished IT/Comms Allowance

City shall be eligible to (i) invoice against the City-Furnished IT/Comms Allowance for City-Furnished IT/Comms FF&E; or (ii) direct Principal Project Company to procure and install such FF&E in which case Principal Project Company shall be eligible to invoice against the Office/Admin and Training Spaces FF&E Allowance to address City-approved Office/Admin and Training Spaces FF&E, each as set forth herein.

- (a) The City-Furnished IT/Comms Allowance shall be in an initial amount of \$2,850,000.
- (b) The City-Furnished IT/Comms Allowance is to be used for City-Furnished IT/Comms FF&E.
- (c) City shall provide Principal-Project-Company a monthly invoice for the procurement, purchase and installation of City-Furnished IT/Comms FF&E.

(d) Invoicing for completed City-Furnished IT/Comms FF&E Work shall be included in the monthly invoice described in <u>Section 11.3</u> (Invoice, Other Amounts and Payments).

# 11.12.4 Partnering Allowance

Principal Project Company shall be eligible to invoice against the Partnering Allowance for Partnering, as set forth herein.

- (a) The Partnering Allowance shall be in an initial amount of \$200,000.
- (b) The Partnering Allowance is to be used for Partnering costs and

expenses.

(c) City shall provide Principal Project Company a monthly invoice for

Partnering.

(d) Invoicing for completed Partnering shall be included in the monthly invoice described in <u>Section 11.3</u> (Invoice, Other Amounts and Payments).

### ARTICLE 12. CITY CHANGE PROCESS; UNILATERAL CHANGE ORDERS; DEVIATIONS

### 12.1 General

<u>Exhibit 9</u> (Change Procedures) sets out the process with respect to (a) Change Orders issued by City following a Proposed Change Order request by City; (b) Unilateral Change Orders unilaterally issued by City; and (c) Change Orders issued by City following a PPC Change Request.

### 12.2 City Changes

**12.2.1** Subject to <u>Section 12.2.2</u> and in accordance with the procedure set forth in <u>Exhibit 9</u> (Change Procedures), City may at any time make changes to the Work, including additions or reductions in the scope of the D&C Work or IFM Services, or changes to the requirements applicable to the Work, as it may direct in its sole discretion (each, a "**City Change**").

**12.2.2** Principal Project Company shall not be required to implement any City Change to the extent the City Change would:

- result in a breach of Law, a breach of Good Industry Practice or a breach of any conditions of a Regulatory Approval or revocation of any Regulatory Approval;
- (b) require a new Regulatory Approval which would not be reasonably obtainable;
- (c) render any Insurance Policy void or voidable;
- (d) materially and adversely affect the health and safety of any person; or
- (e) materially and adversely affect the nature of the Project as a whole, such that the City Change would constitute a cardinal change under California law.

Principal Project Company acknowledges that a City Change relating to a Joint Development Alternative would not fall under <u>clause (e)</u>.

# ARTICLE 13. GENERAL PROVISIONS APPLYING TO DELAY EVENTS AND RELIEF EVENTS

# 13.1 Interface with Other Portions of the Facility

**13.1.1** Principal Project Company acknowledges and agrees that if any PPC-Related Entity, any JDA-Related-Entity, or any entity that is not a PPC-Related Entity or a JDA-Related-Entity but that executes an interface agreement substantially in the form of <u>Exhibit 8</u> (Form of Interface Agreement), is involved in any aspect of the development, design, construction, operations or maintenance of a Joint Development Alternative, then:

- (a) any interference or delay in the performance of the D&C Work or the IFM Services or damage affecting the Infrastructure Facility, in each case, caused in whole or in part by or in any way related to a Joint Development Alternative, shall be entirely Principal Project Company's responsibility and liability;
- (b) Principal Project Company shall not make any Claim for, and shall not receive or be entitled to, any additional compensation, time extensions, or relief from its obligations under this Agreement for any delay, additional costs, or failure to perform due to Losses, interference, delay, or damage caused in whole or in part by or in any way related to a Joint Development Alternative;
- (c) City shall have no liability or responsibility to Principal Project Company for any Losses, interference, delay, or damage incurred or suffered by Principal Project Company or any PPC-Related Entity with respect to the Infrastructure Facility, including the D&C Work and IFM Services, that is caused, in whole or in part, by a JDA-Related Entity or in any way related to or arising out of a Joint Development Alternative; and
- (d) City shall continue to enforce all of City's rights and remedies under this Agreement in circumstances where Principal Project Company cannot perform its obligations hereunder due to any Losses, interference, delay, or damage caused in whole or in part by or in any way related to a Joint Development Alternative or a JDA Related-Entity, including City's rights to assess and collect Deductions, declare a PPC Default, declare a Persistent PPC Default and terminate this Agreement.

**13.1.2** With respect to the D&C Work performed by the D&C Contractor, the provisions of <u>Section 13.1.1</u> shall not apply to the extent that the City commences construction of a Joint Development Alternative prior to the earlier of (i) Substantial Completion; or (ii) the Substantial Completion Deadline, unless, in either case, the prime contractor undertaking such construction is the D&C Contractor or an affiliate thereof (in which case, the provisions of <u>Section 13.1.1</u> shall apply;

# 13.2 Delay Event and Relief Event Process

**13.2.1** If a Delay Event or Relief Event occurs, subject to the limitations and exclusions provided in this Agreement, Principal Project Company may seek additional compensation, time

extension, and/or other relief, if applicable, in accordance with the entitlements specified in <u>Article 14</u> (Compensation and Other Relief for Delay Events and Relief Events).

**13.2.2** The agreement of the Parties as to the specific compensation, time extension, or other relief to be given Principal Project Company on account of a Delay Event or Relief Event, as applicable, shall be evidenced by a written amendment or change order to this Agreement.

- **13.2.3** Either Party may initiate the Contract Dispute Procedures if:
  - (a) the Parties are unable to agree as to the specific compensation, time extension, or other relief to be given Principal Project Company on account of an alleged Delay Event or Relief Event; or
  - (b) City rejects the Delay Event or Relief Event claim, as applicable.

#### 13.3 Mitigation

**13.3.1** If a Delay Event, Relief Event or any other event occurs as a result of which Principal Project Company considers that it is entitled to claim an extension of time, compensation or relief from performance of its obligations under this Agreement (together "**Relevant Events**"), then Principal Project Company shall, and shall require all PPC-Related Entities to, use and continue to use commercially reasonable efforts to:

- (a) Eliminate or mitigate the liability, Losses, schedule impact and other consequences of such event upon the performance of its obligations under this Agreement, including by re-sequencing, rescheduling, reallocating or redeploying Principal Project Company forces to other work, as appropriate;
- (b) Continue to perform and remain liable and responsible for its obligations under this Agreement notwithstanding the Relevant Event; and
- (c) Resume performance of its obligations under this Agreement affected by the Relevant Event as soon as practicable and in no event later than promptly after the cessation of the Relevant Event.

**13.3.2** To the extent that Principal Project Company does not comply with its obligations under this <u>Section 13.3</u> (Mitigation), then Principal Project Company's entitlement to claim an extension of time, compensation or relief from performance of its obligations under this Agreement with respect to the Relevant Event will be reduced to the extent of such failure.

#### 13.4 Deductions for Relevant Events

The compensation payable to Principal Project Company with respect to any Relevant Event will be reduced by:

(a) any amount which a PPC-Related Entity recovers under any Insurance Policy, or would have recovered if it had complied with the requirements of this Agreement in respect of any Insurance Policy in respect of the Relevant Event, which amount, for greater certainty, will not include any excess or deductibles or any amount over the maximum amount insured under any such Insurance Policy; and

(b) the amount of any Extra Work Costs, other direct costs and margins calculated in accordance with <u>Exhibit 13</u> (Costs Schedule) avoided or otherwise reduced as a result of the Relevant Event.

### 13.5 Acts of a PPC-Related Entity

Principal Project Company's entitlement to claim an extension of time, compensation or relief from performance of its obligations under this Agreement with respect to any Relevant Event will be reduced to the extent the Relevant Event arises out of, relates to or was caused or contributed to by the acts or omissions of any PPC-Related Entity or any PPC Fault.

#### 13.6 Notification; Delay in Notification

**13.6.1** Principal Project Company shall provide Notice regarding Relevant Events:

- (a) as provided in <u>Section 14.1.1</u> (Claim for Delay Event) with respect to Delay Events;
- (b) as provided in <u>Section 14.2.2</u> (Claim for a Relief Event) with respect to Relief Events; and

**13.6.2** as provided in accordance with this Agreement with respect to all other Relevant Events.

### 13.7 Multiple and Overlapping Claims

Principal Project Company may make multiple but not duplicative Claims with respect to a Relevant Event.

#### 13.8 Burden of Proof and Mitigation

Principal Project Company bears the burden of proof in establishing (a) the occurrence of a Relevant Event and (b) the entitlement to, and amount of, relief for such Relevant Event, including demonstrating that Principal Project Company complied with its mitigation obligations under <u>Section 13.3</u> (Mitigation).

#### 13.9 Sole Entitlement

Principal Project Company acknowledges and agrees that:

- (a) Subject to the express terms of this Agreement, the Milestone Payment and Availability Payments constitute full compensation for performance of all of the Work; and
- (b) Principal Project Company's sole right to claim an extension of time, compensation or relief from performance of its obligations under this Agreement or otherwise make any Claim for any liability in connection with a

Delay Event or Relief Event is as set out in <u>Article 13</u> (General Provisions Applying to Delay Events and Relief Events) and <u>Article 14</u> (Compensation and Other Relief for Delay Events and Relief Events).

#### 13.10 Compensation

City will pay the compensation due to Principal Project Company under <u>Article 13</u> (General Provisions Applying to Delay Events and Relief Events) and <u>Article 14</u> (Compensation and Other Relief for Delay Events and Relief Events) in accordance with <u>Exhibit 13</u> (Costs Schedule).

### 13.11 Waiver

As a condition precedent to City's obligation to pay any compensation, grant an extension of time or provide any other relief to fully resolve or address a Delay Event or a Relief Event, Principal Project Company shall execute a full, unconditional, irrevocable waiver and release, in favor of and in a form reasonably acceptable to City, of any other Claims, Losses or rights to relief arising out of such Delay Event or a Relief Event, as applicable, that is not the subject of a Contract Dispute.

### ARTICLE 14. COMPENSATION AND OTHER RELIEF FOR DELAY EVENTS AND RELIEF EVENTS

# 14.1 Relief During the D&C Period

This <u>Section 14.1</u> (Relief During the D&C Period) sets out Principal Project Company's entitlement to an extension of time, Extra Work Costs, Financing Delay Costs and Delay Costs (as applicable) as a result of Delay Events. Principal Project Company's entitlement to relief and compensation under this <u>Section 14.1</u> (Relief During the D&C Period) is subject to the limitations on compensation in <u>Article 13</u> (General Provisions Applying to Delay Events and Relief Events), <u>Exhibit 13</u> (Costs Schedule), and this Agreement.

# 14.1.1 Claim for Delay Event

**14.1.1.1** Principal Project Company shall provide Notice to City's Authorized Representative within 30 days of obtaining Actual Knowledge of the occurrence of Delay Event (or, if earlier, on such date that Principal Project Company should have discovered such Relief Event if Principal Project Company was in full compliance with the terms of the Contract Documents). Failure to do so shall result in a waiver of and forfeiture of any relief, compensation or time extension related to the event or occurrence.

**14.1.1.2** Principal Project Company shall, within 10 Business Days after such initial Notice, provide further details to City's Authorized Representative, which shall, to the extent Principal Project Company has obtained Actual Knowledge, include:

- (a) A summary of the provisions of this Agreement that entitle Principal Project Company to relief. If Principal Project Company seeks relief for City's alleged breach of this Agreement, then Principal Project Company shall identify the provisions of this Agreement which allegedly have been breached and the actions or failures to act constituting such breach;
- (b) Details of the Delay Event, the circumstances from which the Delay Event arises including its nature, the date of its occurrence, its duration (to the extent that the Delay Event and the effects thereof have ceased or estimated duration to the extent that the Delay Event and the effects thereof have not ceased), the portions of the Infrastructure Facility affected;
- (c) Details of the contemporary records which Principal Project Company shall maintain to substantiate its claim for extra time and the substance of any oral communications, if any, relating to the Delay Event and the name of the person or persons making such material oral communications;
- (d) Analysis of consequences (whether direct or indirect, financial or nonfinancial) the Delay Event may have upon achieving the Substantial Completion Date or the Final Acceptance Date, as applicable, including a TIA indicating all activities represented or affected by the change, with activity numbers, durations, predecessor and successor activities, resources and cost, and with a narrative report, in form satisfactory to City, which compares the proposed new schedule to the Project Schedule, as appropriate. Principal

Project Company shall reschedule activities not otherwise affected by the event, in order to take advantage of additional Float available as the result of the time extension. Any such rescheduling shall be reflected in the Project Schedule;

- Where the Delay Event is also a Compensable Delay Event, an itemized estimate of all amounts claimed under <u>Section 14.1.1</u> (Claim for Delay Event). Extra Work Costs and Delay Costs (as applicable) and Financing Delay Costs shall be broken down in accordance with <u>Exhibit 13</u> (Costs Schedule);
- (f) Where the Delay Event is also an Unavoidable Delay Event, an itemized estimate of any Financing Delay Costs claimed under <u>Sections 14.1.6</u> or <u>14.1.7</u> (Additional Limits Relating to Force Majeure Events or Unavoidable Delay Events During D&C Period) broken down in accordance with <u>Exhibit 13</u> (Costs Schedule);
- (g) The type and amount of insurance that may be applicable and amounts that have been or are anticipated to be collected under such insurance; and
- (h) Details of any measures that Principal Project Company has taken to date and proposes to adopt to mitigate the consequences of such Delay Event in accordance with <u>Section 13.3</u> (Mitigation).

**14.1.1.3** Within 7 days of Principal Project Company receiving, or becoming aware of, any supplemental information which may further substantiate or support Principal Project Company's Claim, Principal Project Company shall submit further particulars based on such information to City's Authorized Representative.

**14.1.1.4** City's Authorized Representative shall, after receipt of details under <u>Section 14.1.1.2</u>, or of further particulars under <u>Section 14.1.1.3</u>, be entitled by Notice to Principal Project Company to require that Principal Project Company provide such further supporting particulars as City's Authorized Representative may reasonably consider necessary. Principal Project Company shall provide City's Authorized Representative full access and facilities for investigating and assessing the validity of Principal Project Company's Claim, including, on-site inspection.

**14.1.1.5** Principal Project Company shall provide City with monthly updates, together with further details and supporting documentation, as it receives or develops additional information pertaining to the Delay Event and the matters described in <u>Section 14.1.1.2</u> Without limiting the foregoing, Principal Project Company shall notify City as soon as the Delay Event has ceased and when performance of its affected obligations can be resumed.

# 14.1.2 Extension of Deadlines for Delay Events

**14.1.2.1** Except as provided in <u>Section Error! Reference source not found.</u>, upon the occurrence of a Delay Event, Principal Project Company shall be entitled to an extension of the Contract Deadlines equal to the delay to the Critical Path directly caused by the Delay Event.

**14.1.2.2** City's Authorized Representative shall determine revised Contract Deadlines arising out of a Delay Event, as soon as reasonably practicable and in any event within 30 days of the later of:

- (a) The date of receipt by City's Authorized Representative of Principal Project Company's Notice given in accordance with <u>Section 14.1.1.1</u> and the date of receipt of any further particulars (if such are required under <u>Section 14.1.1.2</u> or <u>Section 14.1.1.3</u>), whichever is later; and
- (b) The date of receipt by City's Authorized Representative of any supplemental information supplied by Principal Project Company in accordance with <u>Section</u> <u>14.1.1.3</u> (if such are required under <u>Section 14.1.1.4</u>), whichever is later.

# 14.1.3 Concurrent Delays

If Principal Project Company has made a Claim for an extension of time in accordance with <u>Section 14.1.2</u> (Extension of Deadlines for General Delay Events) which is caused by City Fault and there is another unrelated delay to a Critical Path for which a PPC-Related Entity is responsible under this Agreement, Principal Project Company shall remain entitled to an extension of time to the Substantial Completion Deadline or Final Acceptance Deadline, as applicable, in accordance with, and subject to, <u>Section 14.1.2</u> (Extension of Deadlines for General Delay Events), but shall not be entitled to any compensation or monetary relief associated with such extension of time or Claim. If Principal Project Company has made a Claim for an extension of time in accordance with <u>Section 14.1.2</u> (Extension of Deadlines for General Delay Events) which is not caused by City Fault and there is another unrelated delay to a Critical Path for which a PPC-Related Entity is responsible under this Agreement, Principal Project Company shall not be entitled to an extension of time to the Substantial Completion <u>14.1.2</u> (Extension of Deadlines for General Delay Events) which is not caused by City Fault and there is another unrelated delay to a Critical Path for which a PPC-Related Entity is responsible under this Agreement, Principal Project Company shall not be entitled to an extension of time to the Substantial Completion Deadline or Final Acceptance Deadline, as applicable, to the extent and for so long as the Delay Event is concurrent with any other unrelated delay to a Critical Path for which any PPC-Related Entity is responsible under this Agreement, Principal Event is concurrent with any other unrelated delay to a Critical Path for which any PPC-Related Entity is responsible under this Agreement.

# 14.1.4 Costs Payable for Compensable Delay Events

Subject to the limitations on compensation in <u>Article 13</u> (General Provisions Applying to Delay Events and Relief Events), this <u>Section 14.1</u> (Relief During the D&C Period) and <u>Exhibit 4</u> (Payment Mechanism), if a Compensable Delay Event occurs, City shall reimburse Principal Project Company for:

- (a) Extra Work Costs calculated in accordance with Exhibit 13 (Costs Schedule) directly attributable to the Compensable Delay Event;
- (b) Delay Costs calculated in accordance with <u>Exhibit 13</u> (Costs Schedule) attributable to the Compensable Delay Event, but only where an extension of time is granted under <u>Section 14.1.2</u> (Extension of Deadlines for General Delay Events) for a City-Caused Delay Event; and
- (c) Financing Delay Costs calculated in accordance with <u>Exhibit 13</u> (Costs Schedule) but only where an extension of time is granted under <u>Section 14.1.2</u> (Extension of Deadlines for General Delay Events) for the Compensable

Delay Event and Principal Project Company is unable to achieve Substantial Completion by the original Substantial Completion Deadline.

# 14.1.5 Additional Limits Relating to Hazardous Materials Event During D&C Period

If a Delay Event is a Hazardous Materials Event, no compensation or time relief shall be provided for ((i) once Principal Project Company obtains Actual Knowledge of the Hazardous Materials (or should have discovered such Hazardous Materials if Principal Project Company was in full compliance with the terms of the Contract Documents), any Hazardous Materials that could have been reasonably avoided by use of available construction techniques or incorporation of anticipated or minor design changes which are consistent with Good Industry Practice; (ii) once Principal Project Company obtains Actual Knowledge of the Hazardous Materials (or should have discovered such Hazardous Materials if Principal Project Company was in full compliance with the terms of the Contract Documents, including by performing a Reasonable Investigation), any Hazardous Materials encountered, to the extent Principal Project Company is required and failed to manage or mitigate against the risk of such Hazardous Materials in accordance with this Agreement; and (iii) once Principal Project Company obtains Actual Knowledge of the Hazardous Materials (or should have discovered such Hazardous Materials if Principal Project Company was in full compliance with the terms of the Contract Documents, including by performing a Reasonable Investigation), any Hazardous Materials encountered, to the extent a contractor, acting in accordance with Good Industry Practice, would have taken preventative measures to prevent or minimize such Hazards Materials, and Principal Project Company has failed to take such preventative measures. With respect to Extra Work Costs, Principal Project Company's entitlement to compensation for a Hazardous Materials Event shall be limited to Extra Work performed under the City-approved plans required to be provided under Section 01 35 44 of Division 10 of the Technical Requirements, and as otherwise limited in accordance with Section 7.7 (Hazardous Materials Management; Risk Allocation).

# 14.1.6 Costs Payable for Force Majeure Events During the D&C Term

**14.1.6.1** If a Delay Event for which an extension of time was granted under <u>Section 14.1.2</u> (Extension of Deadlines for General Delay Events) is:

- (a) a Force Majeure Event; and
- (b) the Force Majeure Event is not insured against and is not required to be insured against in accordance with this Agreement,

City will reimburse Principal Project Company for Financing Delay Costs (if any) and any Extra Work Costs, if any, authorized pursuant to a Change Order under <u>Section 14.6</u> (Loss or Damage Due to Force Majeure Termination Event).

**14.1.6.2** In addition, if (i) Principal Project Company is unable to perform all or substantially all of its obligations under the Contract Documents for a period of 180 consecutive days or more; and (ii) such inability to perform its obligations is not attributable to a concurrent non-Force Majeure Termination Event, City will reimburse Principal Project Company for Delay Costs incurred after such 180 day period until 255 days after the occurrence of such Force Majeure

Termination Event, after which <u>Section 17.2</u> (Termination for Force Majeure Termination Events or Insurance Unavailability) applies.

# 14.1.7 Costs Payable for Unavoidable Delay Events During the D&C Term

If a Delay Event for which an extension of time was granted under <u>Section 14.1.2</u> (Extension of Deadlines for General Delay Events) is an Unavoidable Delay Event, City will reimburse Principal Project Company for Financing Delay Costs (if any), such reimbursement to be Principal Project Company's sole entitlement to compensation for such Unavoidable Delay Event.

# 14.1.8 Impact of Delay Event on Performance of D&C Work

**14.1.8.1** Any failure by Principal Project Company to perform the D&C Work during the D&C Term, to the extent directly arising out of any Delay Event, will:

- (a) not constitute a breach of this Agreement by Principal Project Company;
- (b) not result in accrual of Noncompliance Points or Deductions with respect to any Delay Event ;
- (c) relieve Principal Project Company of its obligations to perform such directly affected D&C Work for the duration and to the extent directly prevented by such Delay Event as described in <u>Section 13</u> of <u>Exhibit 4</u> (Payment Mechanism); and
- (d) not result in a Principal Project Company Default or right of termination or other claim by City, other than either Party's right to terminate this Agreement under <u>Section 17.2</u> (Termination for Force Majeure Events or Insurance Unavailability).

**14.1.8.2** Notwithstanding <u>Section 14.1.7</u> (Impact of Delay Event on Performance of D&C Work), Principal Project Company shall remain fully responsible for performance of all elements of the D&C Work not directly impacted or affected by any Delay Event.

#### 14.1.9 Relief for Adverse Weather Event

**14.1.9.1** The occurrence of Adverse Weather alone shall not be a prima facie reason for an Adverse Weather Event, and Principal Project Company shall make every effort to continue to work under prevailing conditions. Such efforts by Principal Project Company shall include: providing temporary gravel roads; installing a rain dewatering system; protecting interior and exterior areas exposed to rain, wind, and extreme temperatures; and providing temporary heat where required for Work to proceed without delay.

**14.1.9.2** Principal Project Company shall plan the D&C Work to allow for 17 days of Adverse Weather per year during normal working hours. The Project Schedule shall incorporate 17 days per year for the anticipated number of days of Adverse Weather. As used in this <u>Section</u> <u>14.1.8.2</u>, "year" shall mean the period measured between May 1-April 30 of each calendar year

during the D&C Period. If the D&C Period commences or ends during a partial year, the foregoing 17-day period shall be prorated and rounded up to the nearest day.

**14.1.9.3** Principal Project Company shall not be entitled to a time extension or relief from its obligations under this Agreement for Adverse Weather until the 18<sup>th</sup> day of occurrence of Adverse Weather in a year, or such lesser applicable amount for any prorated year, as determined pursuant to <u>Section 14.1.8.2</u>.

**14.1.9.4** If there are years with less days of Adverse Weather than 17 days, or such lesser applicable amount for any prorated year, as determined pursuant to <u>Section 14.1.8.2</u>, the unused days shall be rolled into Float.

### 14.2 Relief During the IFM Period

### 14.2.1 Overview

This <u>Section 14.1.7</u> (Relief During the IFM Period) sets out Principal Project Company's entitlement to Extra Work Costs and relief from performance as a result of Relief Events occurring during the IFM Period. Principal Project Company's right to relief and compensation under this <u>Section 14.1.7</u> (Relief During the IFM Period) is subject to the limitations on compensation in <u>Article 13</u> (General Provisions Applying to Delay Events and Relief Events), <u>Exhibit 13</u> (Costs Schedule), and this Agreement.

### 14.2.2 Claim for a Relief Event

**14.2.2.1** Principal Project Company shall provide Notice to City's Authorized Representative within 30 days of obtaining Actual Knowledge of the occurrence of Relief Event (or, if earlier, on such date that Principal Project Company should have discovered such Relief Event if Principal Project Company was in full compliance with the terms of the Contract Documents). Failure to do so shall result in a waiver and forfeiture of any relief, compensation or time extension related to the event or occurrence.

**14.2.2.2** Principal Project Company shall, within 10 Business Days after its Notice under <u>Section 14.2.2.1</u>, provide a further Notice to City's Authorized Representative, which shall, to the extent Principal Project Company has obtained Actual Knowledge, include:

- (a) A summary of the provisions of this Agreement that entitle Principal Project Company to relief. If Principal Project Company seeks relief for City's alleged breach of this Agreement, then Principal Project Company shall identify the provisions of this Agreement which allegedly have been breached and the actions constituting such breach;
- (b) Details of the Relief Event, circumstances from which the Relief Event arises including its nature, the date of its occurrence, its duration (to the extent that the Relief Event and the effects thereof have ceased or estimated duration to the extent that the Relief Event and the effects thereof have not ceased) and any portions of the Infrastructure Facility affected. Impacts to the IFM Services, if any, shall be stated by Term year;

- (c) Details of the contemporary records which Principal Project Company shall maintain to substantiate its claim for relief or compensation and the substance of any oral communications, if any, relating to the Relief Event and the name of the person or persons making such material oral communications;
- (d) Where the Relief Event is also a Compensable Relief Event, an itemized estimate of all amounts claimed under <u>Section 14.2.3</u> (Costs Payable for Compensable Relief Events) broken down into Extra Work Costs identified in <u>Exhibit 13</u> (Costs Schedule). The estimate shall include, to the extent applicable, Extra Work Costs for additional work for future IFM Services, and a proposal for how to reasonably net present value such amount to current dollars should the City wish to pay such amount in a lump sum payment;
- (e) The type and amount of insurance that may be applicable and amounts that have been or are anticipated to be collected under such insurance; and
- (f) Details of any measures that Principal Project Company has taken to date and proposes to adopt to mitigate the consequences of such Relief Event in accordance with <u>Section 13.3</u> (Mitigation).

**14.2.2.3** As soon as possible but in any event within 7 days of Principal Project Company receiving, or becoming aware of, any supplemental information which may further substantiate or support Principal Project Company's Claim, Principal Project Company shall submit further particulars based on such information to City's Authorized Representative.

**14.2.2.4** City's Authorized Representative shall, after receipt of details under <u>Section 14.2.2.2</u>, or of further particulars under <u>Section 14.2.2.3</u>, be entitled by Notice to require Principal Project Company to provide such further supporting particulars as City's Authorized Representative may reasonably consider necessary. Principal Project Company shall afford City's Authorized Representative full access and facilities for investigating and assessing the validity of Principal Project Company's Claim, including, on-site inspection.

**14.2.2.5** Principal Project Company shall provide City with monthly updates, together with further details and supporting documentation, as it receives or develops additional information pertaining to the Relief Event and the matters described in <u>Section 14.2.2.2</u> Without limiting the foregoing, Principal Project Company shall notify City as soon as the Relief Event has ceased and when performance of its affected obligations can be resumed.

# 14.2.3 Costs Payable for Compensable Relief Events

Subject to the limitations on compensation in <u>Article 13</u> (General Provisions Applying to Delay Events and Relief Events), this Section <u>14.1.7</u> (Relief During the IFM Period) and <u>Exhibit 4</u> (Payment Mechanism), upon the occurrence of a Compensable Relief Event, City will reimburse Principal Project Company for all Extra Work Costs incurred by Principal Project Company as a direct result of the Compensable Relief Event calculated in accordance with <u>Exhibit 13</u> (Costs Schedule). City will not be entitled to make Deductions/assess Noncompliance Points for non-performance that is caused directly as a result of a Compensable Relief Event; <u>provided</u>; <u>however</u>, that the Availability Payment shall be reduced for any avoided costs resulting from the Compensable Relief Event.

# 14.2.4 Costs Payable for Force Majeure During IFM Period

- **14.2.4.1** If a Force Majeure Event:
  - (a) is not insured and is not required to be insured in accordance with this Agreement; and
  - (b) directly and adversely impacts the IFM Services after Substantial Completion,

then,

- (c) City will reimburse Principal Project Company for Extra Work Costs, if any, authorized pursuant to a Change Order under <u>Section 14.6</u> (Loss or Damage Due to Force Majeure Termination Event); and
- (d) City is entitled to deduct from the Availability Payments the amount of avoided and/or reduced costs which are not in fact incurred by Principal Project Company during the period where IFM Services is not required to be performed by Principal Project Company pursuant to the Contract Documents. The deduction shall be in the amount of any costs (including applicable Extra Work Costs, Delay Costs, Financing Delay Costs, profit, overhead and margins calculated in accordance with Exhibit 13 (Costs Schedule) that are avoided or otherwise reduced.

#### 14.2.5 Additional Limits Relating to Hazardous Materials Event During IFM Period

If a Relief Event is a Hazardous Materials Event, no compensation, performance or time relief shall be provided for (i) once Principal Project Company obtains Actual Knowledge of the Hazardous Materials (or should have discovered such Hazardous Materials if Principal Project Company was in full compliance with the terms of the Contract Documents), any Hazardous Materials that could have been reasonably avoided by use of available construction techniques or incorporation of anticipated or minor design changes which are consistent with Good Industry Practice; (ii) once Principal Project Company obtains Actual Knowledge of the Hazardous Materials (or should have discovered such Hazardous Materials if Principal Project Company was in full compliance with the terms of the Contract Documents), any Hazardous Materials encountered, to the extent Principal Project Company is required and failed to manage or mitigate against the risk of such Hazardous Materials in accordance with this Agreement; and (iii) once Principal Project Company obtains Actual Knowledge of the Hazardous Materials (or should have discovered such Hazardous Materials if Principal Project Company was in full compliance with the terms of the Contract Documents), any Hazardous Materials encountered, to the extent a contractor, acting in accordance with Good Industry Practice, would have taken preventative measures to prevent or minimize such Hazards Materials, and Principal Project Company has failed to take such preventative measures. Principal Project Company's entitlement to compensation for a Compensable Relief Event that is a Hazardous Materials Event shall be limited to the Extra Work Costs associated with the Hazardous Materials Event for Extra Work performed under the approved plans required to be provided under Section 01 35 44 of Division 10 of the Technical Requirements, and as otherwise limited in accordance with Section 7.7 (Hazardous Materials Management; Risk Allocation).

Infrastructure Facility DBFOM Agreement

## 14.2.6 Impact of Relief Event on Performance of IFM Services

**14.2.6.1** Any failure by Principal Project Company to perform the IFM Services from the Substantial Completion Date until the expiration of the IFM Period, to the extent directly arising out of any Relief Event, will:

- (a) not constitute a breach of this Agreement by Principal Project Company;
- (b) not result in accrual of Noncompliance Points or Deductions with respect to any Relief Event;
- (c) relieve Principal Project Company of its obligations to perform such directly affected IFM Services for the duration and to the extent directly prevented by such Relief Event as described in <u>Section 13</u> of <u>Exhibit 4</u> (Payment Mechanism); and
- (d) not result in a Principal Project Company Default or right of termination or other claim by City, other than either Party's right to terminate this Agreement under <u>Section 17.2</u> (Termination for Force Majeure Termination Events or Insurance Unavailability).

**14.2.6.2** Notwithstanding <u>Section 14.2.6.1</u>, Principal Project Company shall remain fully responsible for performance of all elements of the IFM Services not directly impacted or affected by any Relief Event.

#### 14.3 Method of Payment of Compensation for Compensable Delay Events and Compensable Relief Events

**14.3.1** Additional compensation due for a Compensable Delay Event or a Compensable Relief Event shall be paid in accordance with <u>Section 7.0</u> (Form and Timing of Compensation) of <u>Exhibit 13</u> (Costs Schedule).

**14.3.2** City shall provide Principal Project Company with prompt written Notice of the method chosen for paying Principal Project Company for the amounts owed under this <u>Article 14</u> (Compensation and Other Relief for Delay Events and Relief Events).

**14.3.3** If City elects to pay for such amounts by periodic payments, progress payments, milestone payments or a lump sum payment, Principal Project Company shall submit an invoice, in a format acceptable to City, for the amount of each such payment, and City will make payment of all undisputed amounts to Principal Project Company within 30 Business Days of receipt of a complete and proper invoice.

#### 14.4 Open Book Basis

Principal Project Company shall share with City all data, documents and information, and shall conduct all discussions and negotiations, pertaining to any claimed Delay Event or Relief Event, as applicable, on an Open Book Basis.

# 14.5 Excavations; Public Contract Code 7104

Information regarding site conditions included in the Technical Requirements and Reference Documents (including any information, reports, or studies about site conditions, geotechnical conditions, Utilities or structure and bridge design, and any interpretations, extrapolations, analyses and recommendations contained therein) shall not be considered "indicated" therein as such term is used in Public Contract Code section 7104. Principal Project Company is responsible for investigating and satisfying itself as to the site conditions affecting the Project Site and the Work. To the maximum extent permitted by Law, Principal Project Company knowingly, unconditionally, irrevocably and specifically waives each and every right and benefit of Public Contract Code section 7104 to the extent that it may be inconsistent with any provision of the Contract Documents. Principal Project Company acknowledges and agrees that this waiver and the risk allocations set forth in the Contract Documents are a material consideration for City to award and enter into this Agreement with Principal Project Company.

### 14.6 Loss or Damage Due to Force Majeure Event

**14.6.1** If the D&C Work or the Infrastructure Facility are wholly or substantially destroyed or damaged by:

- (a) a Force Majeure Event which is not insured against and is not required to be insured against in accordance with this Agreement or Insurance Unavailability; and
- (b) City requires Principal Project Company to repair, replace or rebuild the D&C Work or the Infrastructure Facility,

the Parties will use reasonable efforts to negotiate and agree on how the D&C Work or the Infrastructure Facility will be repaired, replaced or rebuilt in accordance with <u>Exhibit 9</u> (Change Procedure) and the Extra Work Costs of doing so in accordance with <u>Exhibit 13</u> (Costs Schedule).

**14.6.2** The Parties' attempts to negotiate must not limit City's right to issue a Change Order or Unilateral Change Order in accordance with this Agreement or the Parties' right to terminate under <u>Section 17.2</u> (Termination for Force Majeure Termination of Insurance Unavailability).

# ARTICLE 15. DEDUCTIONS AND NONCOMPLIANCE POINTS

#### 15.1 Failure Events

**15.1.1** <u>Appendix A of Exhibit 4</u> (Payment Mechanism) identifies certain D&C Failure Events applicable from the Effective Date until the Final Acceptance Date.

**15.1.2** The Performance Measurements Table identifies certain IFM Failure Events which may become an Availability Failure or Service Failure during the IFM Period as described in Exhibit 4 (Payment Mechanism).

**15.1.3** Each Noncompliance Event will result in either or both:

- (a) the assessment of Noncompliance Points;
- (b) the assessment of Deductions,

in each case in accordance with Exhibit 4 (Payment Mechanism).

#### 15.2 Deductions

- **15.2.1** In accordance with <u>Exhibit 4</u> (Payment Mechanism) and <u>Section 15.4</u>, City may:
  - (a) assess D&C Period Deductions (i) immediately for any D&C Failure Event for which no Rectification Time is specified in the Performance Measurements Table; (ii) at the end of each Rectification Time or any additional Rectification Time for any D&C Failure Event if Principal Project Company has not Rectified the Noncompliance Event; and
  - (b) assess IFM Period Deductions (i) immediately for any IFM Failure Event or D&C Failure Event for which no Rectification Time is specified in the Performance Measurements Table; (ii) at the end of the Response Time (if any) if Principal Project Company has not Responded to the Noncompliance Event; and (iii) the end of each Rectification Time if Principal Project Company has not Rectified the Noncompliance Event.

**15.2.2** Principal Project Company acknowledges that any Deductions, are reasonable liquidated damages under the circumstances existing at the Effective Date, to compensate City for:

- (a) City's increased costs of administering the Contract Documents, including any obligations to pay or reimburse Governmental Entities over the Project for their increased costs of monitoring and enforcing Principal Project Company's compliance with applicable Regulatory Approvals;
- (b) City's potential loss of revenues due to a reduction in its operations at the Infrastructure Facility;
- (c) potential harm to the public; and

(d) potential harm to the credibility and reputation of City with stakeholders, policy makers and with the general public.

**15.2.3** Principal Project Company further acknowledges that such increased costs and loss of revenue, credibility and reputation, would be difficult and impracticable to measure and prove, because, among other things, the costs of administering the Contract Documents prior to increases in the level thereof will be variable and extremely difficult to quantify and the variety of factors that influence use of and demand for the Infrastructure Facility and associated SFMTA services generally make it difficult to sort out causation and quantify the precise revenue loss attributable to the matters that will trigger these liquidated damages.

**15.2.4** Except for other remedies expressly provided in this Agreement, including City's right to assess Noncompliance Points and to be indemnified for Third Party claims, any Deduction assessed in accordance with this Agreement shall constitute City's sole remedy in respect of City's damages arising from the Noncompliance Event, for which such Deduction is assessed.

# 15.3 Noncompliance and Failure Reporting, Notification and Cure Process

### 15.3.1 Noncompliance Database

**15.3.1.1** Principal Project Company shall establish and maintain an electronic database for recording and tracking Failure Events and Noncompliance Events (the "**Noncompliance Database**"). The format and design of the database shall be subject to City's approval, in its reasonable discretion. At a minimum, the database shall:

- (a) include a description of each Failure Event in reasonable detail;
- (b) provide for automatic notification to City of the entry of a Failure Event in the Noncompliance Database;
- (c) identify the location of the Failure Event (if applicable);
- (d) identify the date and time of each Failure Event;
- (e) identify the applicable Response Time and Rectification Time of the IFM Failure Event, if any, stated in the Performance Measurements Table;
- (f) indicate date and time of Response and Rectification for any Failure Event;
- (g) indicate when a Noncompliance Event has developed; and
- (h) record the number of assessed Noncompliance Points, the date of each assessment, and the date when each Noncompliance Event is cured.

**15.3.1.2** Principal Project Company shall provide City with full access to the Noncompliance Database at all times, including the ability to enter Failure Events into the Noncompliance Database as provided in <u>Section 15.3.2</u> (Notification Initiated by City).

# 15.3.1.3 Notification Initiated by Principal Project Company

As an integral part of Principal Project Company's self-monitoring obligations, Principal Project Company shall record in the Noncompliance Database, in real time and upon discovery, the occurrence of any Failure Event specified in the Performance Measurements Table.

### 15.3.2 Notification Initiated by City

If City believes a Failure Event has occurred that Principal Project Company has not recorded in the Noncompliance Database, City may enter the Failure Event into the Noncompliance Database or deliver to Principal Project Company a Notice of Determination as provided in <u>Section 15.3.3.2</u> via the Noncompliance Database, and delivery shall be deemed given upon proper entry of the information into the Noncompliance Database.

#### 15.3.3 Performance Reports

**15.3.3.1** Principal Project Company shall include in each Monthly Report and in each Monthly Progress Status Report required under <u>Section 1.2.4.1</u> of <u>Division 1</u> and <u>Section B.11</u> of the IFM Specifications, a report of all Failure Events and Noncompliance Events that occurred during the preceding calendar month, which reports shall include the same detailed information required to be recorded in the Noncompliance Database. Principal Project Company shall correct any inaccuracies in reporting of Failure Events, Noncompliance Events and Noncompliance Points, within 10 Business Days of City's notification to Principal Project Company of such inaccuracies.

**15.3.3.2** Within a reasonable time after receiving the Monthly Report, City will deliver to Principal Project Company a Notice setting forth for each Failure Event City's determination whether the Failure Event was Responded to or Rectified (as applicable) during the applicable Rectification Time, City's determination whether a Noncompliance Event occurred, and the Noncompliance Points and Deductions to be assessed with respect to such Noncompliance Event (a "**Notice of Determination**").

# 15.3.4 Response and Rectification Times

**15.3.4.1** Principal Project Company shall Respond to each IFM Failure Event by the end of the Response Time (if any) for each such Failure Event.

**15.3.4.2** Principal Project Company shall Rectify each IFM Failure Event by the end of the Rectification Time (if any) for each such Failure Event.

**15.3.4.3** For each IFM Failure Event identified, Principal Project Company's applicable Response Time and Rectification Time with respect to the IFM Failure Event shall be deemed to start on the date and time Principal Project Company first obtained Actual Knowledge or should have first reasonably known of (if Principal Project Company was in full compliance with the terms of the Contract Documents) the IFM Failure Event. For this purpose, if the Notice of the IFM Failure Event is initiated by City, Principal Project Company shall be deemed to first obtain Actual Knowledge of the IFM Failure Event not later than the date of delivery of the Notice to Principal Project Company.

**15.3.4.4** Following the occurrence of an Availability Deduction or Service Failure Deduction, Principal Project Company shall be allowed an additional Rectification Time as set out in

<u>Section 6.3</u> of <u>Exhibit 4</u> (Payment Mechanism). If, before the expiry of this additional period, Principal Project Company Rectifies the Availability Failure or Service Failure, no further Availability Deduction or Service Failure Deduction shall be made. Otherwise, a further Deduction shall be made of the applicable amount and a further Rectification Time of equal duration shall apply. This process and application shall continue until Principal Project Company Rectifies the Availability Failure or Service Failure, subject to any other rights and remedies of City in connection with such Availability Failure or Service Failure.

### 15.3.5 Notification of Response and Rectification

**15.3.5.1** When Principal Project Company determines that its Response or Rectification (as applicable) of an IFM Failure Event has been completed, Principal Project Company shall make an entry in the Noncompliance Database that:

- (a) identifies the IFM Failure Event;
- (b) states that Principal Project Company has completed Response or Rectification; and
- (c) briefly describes the applicable Response or Rectification, including any modifications to the Project Management Plan to prevent future similar IFM Failure Events.

**15.3.5.2** Principal Project Company shall include the same information in the next Monthly Report.

**15.3.5.3** City may, via written Notice of rejection, reject any Principal Project Company Notice of completed Response or Rectification if it determines that Principal Project Company has not rectified the Failure Event and shall, upon making this determination, deliver a written Notice of rejection to Principal Project Company.

#### 15.4 Assessment of Noncompliance Points

**15.4.1** City may assess Noncompliance Points in accordance with this <u>Section 15.4</u> (Assessment of Noncompliance Points), <u>Section 6</u> of <u>Exhibit 4</u> (Payment Mechanism) and the Performance Measurements Table upon the occurrence of Noncompliance Events.

**15.4.2** If, at any time, the Noncompliance Database or a Monthly Report indicates, or City is notified or otherwise obtains Actual Knowledge of, a Failure Event or Noncompliance Event, or City provides Principal Project Company with a Notice of Determination, then, without prejudice to any other right or remedy available to City, City may assess Noncompliance Points, subject to the following terms and conditions:

 (a) except as provided in <u>Section 15.4.2(b)</u>, City may assess Noncompliance Points at each of the following times: (i) immediately for any IFM Failure Event or D&C Failure Event for which no Rectification Time is specified in the Performance Measurements Table; (ii) the end of the Response Time (if any) if Principal Project Company has not Responded to the Noncompliance Event; and (iii) the end of each Rectification Time if Principal Project Company has not Rectified the Noncompliance Event;

- (b) if City initiated Notice of a Failure Event that becomes a Noncompliance Event entitling City to assess Noncompliance Points, and to the extent City has determined to assess Noncompliance Points, City may allocate the applicable Noncompliance Points at the commencement of the applicable Response Time or Rectification Time. If City and Principal Project Company deliver concurrent written Notices of the same Failure Event, or concurrently seek to enter the details of the Failure Event into the Noncompliance Database, Principal Project Company's Notice shall prevail, if complete and compliant;
- (c) the number of points listed in <u>Appendix A and Appendix D</u> to <u>Exhibit 4</u> (Payment Mechanism) or the Performance Measurements Table (as applicable) for any particular Noncompliance Event is the maximum number of Noncompliance Points that may be assessed for each event or circumstance that is a Noncompliance Event. City may, in its sole discretion, assess less than the maximum;
- (d) if a Noncompliance Event continues beyond its relevant Response Time or Rectification Time (as applicable), if any, each subsequent Response Time or Rectification Time (as applicable), shall be treated as a new and separate Noncompliance Event, without necessity for further Notice and Noncompliance Points will continue to accrue for every additional Response Time or Rectification Time (as applicable), shall be treated as a new and separate Noncompliance Event; and
- (e) 100% of the Noncompliance Points assessed during the D&C Period may be carried forward and used by City to calculate, after the Substantial Completion Date, accumulated Noncompliance Points for determining the occurrence of a Persistent PPC Default and for the purposes specified in <u>Sections Error!</u> <u>Reference source not found.</u> (Persistent PPC Default and Increased Oversight), <u>15.8</u> (Cure Plan for Persistent PPC Default), and <u>16.1.1(r)</u>.

**15.4.3** Regardless of the continuing assessment of Noncompliance Points under this <u>Section 15.4</u> (Assessment of Noncompliance Points), City may exercise its step-in rights under <u>Section 16.2.5</u> (City Step-in Rights) and, if applicable, its Work suspension rights under <u>Section 16.2.8</u> (Suspension of Work), after expiration of the applicable Rectification Time.

# 15.5 Performance Notice

**15.5.1** In addition to other remedies available under this Agreement, City may provide Principal Project Company, a Notice, indicating unsatisfactory performance and establishing the matters that give rise to such Notice, if the Performance Notice Threshold is reached ("Performance Notice").

**15.5.2** If a Performance Notice is given by the City, then:

(a) Principal Project Company shall, within 20 days of receiving the Performance Notice, submit to the City a cure plan for review and approval in accordance with Exhibit 11 (Submittals Review Process) ("Performance Cure Plan"). The Performance Cure Plan will include reasons for the unsatisfactory performance and specific actions (including timeframes) to be taken by Principal Project Company to improve its performance.

- (b) The Parties shall consult to develop and agree the Performance Cure Plan within 10 days of Principal Project Company's submittal of such plan; and
- (c) Following agreement or determination of the Performance Cure Plan, Principal Project Company shall implement and comply with the Performance Cure Plan and any failure to do so shall result in a second Performance Notice.

### 15.6 Increased Oversight

**15.6.1** In addition to other remedies available under this Agreement, City may provide Principal Project Company, a Notice for increased Oversight, if the Increased Oversight Threshold is reached.

**15.6.2** In addition to other remedies available under this Agreement, City may, by Notice to Principal Project Company direct Principal Project Company to increase its own oversight or change the type and/or increase the level of City's Oversight of the Project, in such manner and to such level considered fit, if the Increased Oversight Threshold is reached.

**15.6.3** If City changes the type or increases the level of its Oversight due to Principal Project Company reaching the Increased Oversight Threshold, then Principal Project Company shall pay and reimburse City, within 30 days after receipt of written demand and reasonable supporting documentation, all reasonable increased costs and fees City incurs in connection with such action, including City's Recoverable Costs.

**15.6.4** If City directs Principal Project Company to increase its own oversight or changes the type or increases the level of its Oversight, then:

- (a) Principal Project Company shall, within 30 days of Notice submit to the City a cure plan for review and approval in accordance with Exhibit 11 (Submittals Review Process) ("Increased Oversight Cure Plan"). The Increased Oversight Cure Plan shall include specific actions (including timeframes) to be taken by Principal Project Company to improve its performance; and
- (b) the Parties shall consult to develop and agree to the Increased Oversight Cure Plan within 20 days of Principal Project Company's submittal of the Increased Oversight Cure Plan.

**15.6.5** Principal Project Company's obligation to pay and reimburse City for increased Oversight costs shall apply to all changes in the type or increases in the level of City's Oversight occurring until Principal Project Company has:

- (a) implemented and complied with the Increased Oversight Cure Plan;
- (b) fully and completely cured the breaches and failures that gave rise to the Increased Oversight;
- (c) diligently pursued cure of all other Failure Events; and

- (d) during the D&C Period, has accumulated less than 40 Noncompliance Points in the subsequent 90 days following notice of Increased Oversight; or
- (e) during the IFM Period, has accumulated less than 375 Noncompliance Points in the subsequent 90 days following notice of Increased Oversight

#### 15.7 Persistent PPC Default

**15.7.1** In addition to other remedies available under this Agreement including <u>Section 15.6</u>, City may provide PPC with a Notice of Persistent PPC Default ("Persistent PPC Default Notice").

### 15.8 Cure Plan for Persistent PPC Default

- **15.8.1** Upon the occurrence of a Persistent PPC Default:
  - (a) Principal Project Company shall, within 30 days of the Persistent PPC Default, submit to City a cure plan for review and approval in accordance with <u>Exhibit</u> <u>11</u> (Submittals Review Process) ("Persistent PPC Default Cure Plan"). The Persistent PPC Default Cure Plan shall include specific actions (including timeframes) to be taken by Principal Project Company to improve performance
  - (b) City may require, in its good faith discretion, that Principal Project Company's actions under this <u>Section 15.8.1</u> include improving Principal Project Company's quality management practices, plans and procedures, revising and restating the Project Management Plan, changing organizational and management structure, increasing monitoring and inspections, changing Key Personnel and other important personnel, replacement of IFM Provider, replacement of Subcontractors;
  - (c) The Parties shall consult to develop and agree to the Persistent PPC Default Cure Plan within 20 days of Principal Project Company's submittal of the plan; and
  - (d) Following agreement or determination of the Persistent PPC Default Cure Plan, Principal Project Company shall implement and comply with the cure plan and any failure to do so shall result in an additional PPC Default which shall not be subject to any additional Notice or cure period.
- **15.8.2** If:
  - (a) Principal Project Company complies in all material respects with the approved Persistent PPC Default Cure Plan

As of the date it achieves such requirements, there exist no other uncured PPC Defaults for which a Default Notice was given; and

- (b) During the D&C Period, Principal Project Company has accumulated less than 100 points in the subsequent 90 days, following notice of Persistent PPC Default; or
- (c) During the IFM period, Principal Project Company has accumulated less than 750 Noncompliance points in the subsequent 6-month period, following notice of Persistent PPC Default

Upon successfully curing the Persistent PPC Default, during the IFM Period only, City will reduce the number of cumulative Noncompliance Points that would otherwise then be counted toward Persistent PPC Default by 50%. Such reduction will be taken from the earliest assessed Noncompliance Points that would otherwise then be counted toward Persistent PPC Default.

### 15.9 Amendment of IFM Period Performance Measurements Table

**15.9.1** The Performance Measurements Table contains a representational, but not exhaustive, list of IFM Failure Events possible during the IFM Period under the Contract Documents. Subject to <u>Sections 15.9.2</u> and <u>15.9.3</u>, City may:

- (a) replace IFM Failure Events in the Performance Measurements Table by removing an IFM Failure Event and adding in its place an alternate IFM Failure Event identified in accordance with <u>Section 15.9.2(a)</u>; and
- (b) establish the category applicable to, and set a reasonable "Performance Measurement," "Response Time" and "Rectification Time", each alternate Failure Event.

**15.9.2** City's right to revise the Performance Measurements Table is subject to the following restrictions:

- (a) any existing contractual obligation of Principal Project Company may become an alternate IFM Failure Event if:
  - (i) Principal Project Company has failed to comply with that contractual obligation at least twice;
  - (ii) City provided Notice to Principal Project Company within 15 days of Principal Project Company's second failure to comply and such Notice identifies the contractual obligation and the instances of Principal Project Company's failure to comply, and indicates that City will add that contractual obligation to the Performance Measurements Table if Principal Project Company at any time again fails to comply with the specified contractual obligation; and
  - (iii) Principal Project Company fails a third time to comply with the contractual obligation specified in the Notice;
- (b) the total number of Noncompliance Points in the Performance Measurements Table as existing on the Financial Close Date shall not increase by more than

10% for the Term. City may elect to remove contractual obligations and reduce Noncompliance Points allocated to listed contractual obligations in order to comply with the 10% growth limit; and

(c) the number of Noncompliance Points associated with any alternate IFM Failure Event added to the Performance Measurements Table shall not exceed 20 Noncompliance Points.

**15.9.3** Changes to the Performance Measurements Table made in accordance with this <u>Section 15.9</u> (Amendment of IFM Period Performance Measurements Table) will be applied prospectively, starting three days after delivery to Principal Project Company of the revised table or such later date stated in the Notice delivering any revised table.

# ARTICLE 16. DEFAULT; REMEDIES

# 16.1 Default by Principal Project Company; Cure Periods

### 16.1.1 PPC Default

Principal Project Company shall be in breach under this Agreement upon the occurrence of any one or more of the following events or conditions (each a "**PPC Default**"):

- (a) Principal Project Company fails to satisfy the conditions set forth in <u>Section</u> <u>7.4.3</u> (Commencement of Non-Construction Work) within 30 days after City's issuance of NTP 1, to begin the D&C Work within 10 days following City's issuance of NTP 2, or to diligently prosecute the Work to completion in accordance with the Contract Documents;
- (b) Principal Project Company abandons all or a material part of the Project, which abandonment is deemed to occur if (i) Principal Project Company demonstrates through statements, acts or omissions an intent not to continue, for any reason other than a Delay Event or Relief Event that materially impairs Principal Project Company's ability to continue, to design, construct, operate or maintain all or a material part of the Project, or (ii) no significant Work (taking into account the Project Schedule, if applicable, and any Delay Event or Relief Event) on the Project is performed for a continuous period of more than 30 days unless due to Principal Project Company's compliance with a City suspension order issued under this Agreement;
- (c) Principal Project Company fails to achieve Substantial Completion by the Long Stop Date;
- (d) Principal Project Company (i) fails to make any payment owing to City under the Contract Documents when due, or (ii) fails to deposit other funds into any custodial account, trust account or other reserve or account (including the Handback Requirements Reserve Account) in the amount and within the time period required by the Contract Documents;
- (e) Principal Project Company fails to begin the IFM Services within 30 days following the Substantial Completion Date;
- (f) (i) any representation or warranty in the Contract Documents made by Principal Project Company is false in any material respect, materially misleading or inaccurate in any material respect when made or omits material information when made, or (ii) any certificate, schedule, report, instrument or other document delivered by or on behalf of Principal Project Company, any Equity Member, Controlling Affiliate of Principal Project Company, Prime Contractor or Supplier to City as part of the Implementation Proposal or under the Contract Documents is false in any material respect, materially misleading or inaccurate in any material respect when made or omits material information when made;

- (g) Principal Project Company fails to obtain, provide and maintain any insurance, bonds, guarantees, letters of credit or other payment or performance security as required under the Contract Documents for the benefit of relevant parties, or Principal Project Company fails to comply with any requirement of the Contract Documents pertaining to the amount, terms or coverage of the insurance or security or fails to pay the associated premiums, deductibles, retain self-insured retentions, co-insurance or any other such amounts as and when due;
- (h) Principal Project Company ceases performing a substantial portion of its business, or a substantial portion of such business is suspended or is not being performed, whether voluntarily or involuntarily, that has or will have a material adverse effect on Principal Project Company's ability to perform its obligations under this Agreement;
- (i) Principal Project Company makes, attempts to make or suffers a voluntary or involuntary assignment or transfer of all or any portion of the Contract Documents, the Project or PPC's Interest in violation of the limitations on assignment or transfer under this Agreement, (ii) there occurs an Equity Transfer or a Change of Control not permitted under this Agreement, or (iii) any other violation of the limitations on assignment or transfer under this Agreement occurs;
- (j) Principal Project Company fails to timely observe or perform, or cause to be observed or performed any material covenant, agreement, obligation, term or condition required to be observed or performed by Principal Project Company under the Contract Documents, including failure to pay for or perform the Design Work, Construction Work, IFM Services or any portion thereof (except to the extent payment is subject to a good faith payment dispute with a subcontractor) in accordance with the Contract Documents in any material respect, provided that any failure that constitutes a Noncompliance Event or Failure Event is not considered a default under this <u>clause (j)</u> although such failure may become a PPC Default in accordance with <u>clause (r)</u> or (s) below;
- (k) Principal Project Company, any Equity Member, any Controlling Affiliate of Principal Project Company, any Prime Contractor, any Supplier, or any of their respective partners, members, officers, directors, responsible managing officers, or responsible managing employees, has been convicted in a court of competent jurisdiction of any charge of fraud, bribery, collusion, conspiracy, or any other act in violation of any state or federal antitrust law in connection with the bidding upon, award of, or performance of, any public works contract, as defined in California Public Contract Code section 1101, with any public entity, as defined in California Public Contract Code section 1100, provided that, if the conviction relates to an Equity Member, a Controlling Affiliate of Principal Project Company, a Prime Contractor, a Supplier, or any of their or Principal Project Company's respective partners, members, officers, directors, responsible managing officers, or responsible managing employees, (i) such Person is involved in the Project at the time of such conviction and (ii) Principal Project Company fails to remove such Person from the Project;

- Principal Project Company fails to comply with City's order to suspend Work issued in accordance with <u>Sections 9.8.6</u>, <u>10.1.2.4(c)</u>, <u>16.2.4.5(b)</u> or <u>16.2.8</u> (Suspension of Work) within the time reasonably allowed in such order;
- (m) A Bankruptcy Event arises with respect to:
  - (i) Principal Project Company except to the extent such Bankruptcy Event is caused by a failure by City to pay Principal Project Company as required under this Agreement, and/or
  - (ii) Any D&C Contractor or IFM Provider, unless Principal Project Company:
    - (A) enters into a replacement D&C Contract, guarantee or IFM Contract (as relevant) with a reputable counterparty acceptable to City, in its good faith discretion, within 60 days of the relevant Bankruptcy Event; provided, however, that, if Principal Project Company has commenced and is diligently pursuing meaningful steps to secure a replacement D&C Contract, guarantee or IFM Contract (as relevant) immediately after the Bankruptcy Event, Principal Project Company shall have such additional period of time, up to a maximum period of 120 days after the Bankruptcy Event;
    - (B) in the absence of entering into a replacement D&C Contract, Principal Project Company otherwise demonstrates to the satisfaction of City, in its good faith discretion, that Principal Project Company possesses the technical and financial capacity to perform all remaining D&C Work in accordance with this Agreement; or
    - (C) in the absence of entering into a replacement IFM Contract, Principal Project Company otherwise demonstrates to the satisfaction of City, in its good faith discretion, that Principal Project Company possesses the technical and financial capacity to perform all remaining IFM Services in accordance with this Agreement;
- (n) Principal Project Company draws against any custodial account, trust account, allowance or other reserve or account in violation of the Contract Documents or makes a false or materially misleading representation in connection with a draw against any such account, allowance or reserve;
- Principal Project Company fails to comply with applicable Regulatory Approvals or Laws, including the Laws described in <u>Exhibit 16</u> (Federal, State and City Requirements) in any material respect;
- (p) any use of the Project by any PPC-Related Entity that violates requirements of applicable Regulatory Approvals or Laws or otherwise is not permitted under the Contract Documents;

- (q) any D&C Contract or IFM Contract is terminated (other than non-default termination on its scheduled termination date) and Principal Project Company has not either:
  - (i) entered into a replacement D&C Contract or IFM Contract (as relevant) with a reputable counterparty acceptable to City, in its good faith discretion, within 60 days of the termination of the relevant D&C Contract or IFM Contract (as relevant); provided, however, that, if Principal Project Company has commenced and is diligently pursuing meaningful steps to secure a replacement D&C Contract or IFM Contract (as relevant) immediately after the termination, Principal Project Company shall have such additional period of time, up to a maximum period of 120 days after the termination; or
  - (ii) in the absence of entering into a replacement IFM Contract, Principal Project Company otherwise demonstrates to the satisfaction of City, in its good faith discretion, that Principal Project Company possesses the technical and financial capacity to perform all remaining IFM Services in accordance with this Agreement;
- (r) a Persistent PPC Default occurs, City delivers to Principal Project Company a Default Notice, and either (i) Principal Project Company fails to deliver to City, within 30 days after such Notice is delivered, a cure plan meeting the requirements for approval in <u>Section 15.8</u> (Cure Plan for Persistent PPC Default) or (ii) Principal Project Company fails to fully comply with the schedule or specific elements of, or actions required under, the approved cure plan;
- (s) after any rights of appeal have been exhausted, Principal Project Company, any PPC-Related Entity or any Subcontractor (i) is determined to be disqualified, suspended or debarred, or otherwise excluded from bidding, proposing or contracting with a federal, State, or City department or agency, or (ii) has not dismissed any Subcontractor whose work is not substantially complete and who is determined to be disqualified, suspended or debarred, or otherwise excluded from bidding, or proposing or contracting with a federal, State, or City department or agency; or
- (t) In the event that LEED BD+C Gold Rating, including the mandatory 12x EAc2 Optimize Energy Performance credits, is not obtained within 24 months after the Final Acceptance Date, other than as a sole and direct result of any affirmative act or omission of City or unless unreasonably withheld by USGBC or an equivalent body.

# 16.1.2 Default Notice and Cure Periods

- **16.1.2.1** Principal Project Company shall promptly:
  - (a) provide Notice to City upon the occurrence of a PPC Default; and

(b) take steps to commence the cure of and mitigate the effects of any PPC Default.

**16.1.2.2** If Principal Project Company notifies City of a PPC Default in accordance with <u>Section</u> <u>16.1.2.1</u> or City considers a PPC Default has occurred, City may give PPC a Notice ("**Default Notice**") which contains:

- (a) details of the PPC Default;
- (b) the cure period (if any) by which Principal Project Company shall cure the PPC Default in accordance with <u>Section 16.1.2.3</u>; and
- (c) if the PPC Default is not capable of being cured or no cure period is applicable, a date by which Principal Project Company shall comply with any requirements of City in connection with that PPC Default.

**16.1.2.3** The following list identifies Principal Project Company's rights to receive Notice and opportunity to cure before City may exercise its right to terminate this Agreement or enforce its Performance Bond in accordance with <u>Section 16.2.7</u> (Performance Bond), and identifies other PPC Defaults that are not subject to cure:

- (a) respecting any PPC Default under <u>Sections 16.1.1(h)</u> or <u>16.1.1(m)</u>, a period of 15 days after delivery by City to Principal Project Company of a Default Notice;
- (b) respecting a PPC Default under <u>Sections 16.1.1(a)</u>, <u>16.1.1(b)</u>, <u>16.1.1(d)</u>, <u>16.1.1(i)</u>, <u>16.1.1(n)</u>, <u>16.1.1(o)</u>, or <u>16.1.1(p)</u>, a cure period of 30 days after City delivers to Principal Project Company a Default Notice; provided that City may effect cure, at Principal Project Company's expense, if a PPC Default under <u>Section 16.1.1(g)</u> continues beyond five days after such Notice is delivered;
- (c) respecting a PPC Default under <u>Sections 16.1.1(f)</u>, <u>16.1.1(j)</u> or <u>16.1.1(k)</u>, a cure period of 30 days after City delivers to Principal Project Company a Default Notice; provided that:
  - (i) if the nature of such PPC Default is such that the cure cannot with diligence be completed within such time period and Principal Project Company has commenced and is diligently pursuing meaningful steps to cure immediately after receiving the Default Notice, Principal Project Company shall have such additional period of time, up to a maximum cure period of 60 days after City delivers the Default Notice, as is reasonably necessary to diligently effect cure; and
  - (ii) as to <u>Section 16.1.1(f)</u>, cure will be regarded as complete when the adverse effects of the breach are cured;
- (d) respecting a PPC Default under <u>Sections 16.1.1(c)</u>, <u>16.1.1(e)</u>, <u>16.1.1(l)</u>, <u>16.1.1(q)</u>, <u>16.1.1(r)</u>, or <u>16.1.1(s)</u>, no cure period is allowed; and

(e) respecting a PPC Default under <u>Section 16.1.1(t)</u>, a cure period of 30 days after City delivers to Principal Project Company a Default Notice, provided that if the nature of such PPC Default is such that the cure cannot with diligence be completed within such time period and Principal Project Company has commenced and is diligently pursuing meaningful steps to cure immediately after receiving the Default Notice, Principal Project Company shall have such additional period of time, up to a maximum cure period of 365 days after City delivers the Default Notice, as is reasonably necessary to diligently effect cure.

## 16.1.3 Warning Notices

**16.1.3.1** Without prejudice to any other right or remedy available to City, City may deliver a Notice (a "**Warning Notice**") to Principal Project Company, with a copy to the Collateral Agent, stating explicitly that it is a "Warning Notice" of a material PPC Default and stating in reasonable detail the matter or matters giving rise to the Warning Notice and, if applicable, amounts due from Principal Project Company, whenever there occurs a PPC Default.

**16.1.3.2** If City issues a Warning Notice for any PPC Default after it issues a Default Notice, then the remaining cure period available to Principal Project Company, if any, for such PPC Default before City may terminate this Agreement on account of such PPC Default will be extended by the time period between the date the Default Notice was issued and the date the Warning Notice is issued. However, this shall not affect the time when City may exercise any remedy other than termination respecting such PPC Default.

# 16.1.4 Principal Project Company to Comply with Default Notice and Provide Cure Plan

- **16.1.4.1** If City gives a Default Notice to Principal Project Company, then:
  - (a) Principal Project Company shall comply with the Default Notice;
  - (b) Except for a PPC Default under Sections <u>16.1.1(b)</u>, <u>16.1.1(e)</u>, <u>16.1.1(i)</u>, <u>16.1.1(m)</u>, <u>16.1.1(q)</u>, and <u>16.1.1(r)</u>, and PPC Defaults for which no cure period is applicable, Principal Project Company shall, as soon as possible, give City a plan for review and approval in accordance with <u>Exhibit 11</u> (Submittal Review Process) to cure the PPC Default and comply with any requirements of City in accordance with the terms of City's Default Notice (which plan shall also specify steps to address the underlying cause of the PPC Default and to avoid similar PPC Defaults occurring in the future);
  - (c) The Parties shall consult to develop and agree to the cure plan; and
  - (d) Following agreement or determination of the cure plan, Principal Project Company shall implement and comply with the cure plan. Any failure to implement the cure plan or comply with the agreed cure plan will result in a PPC Default which is not subject to any cure period.

**16.1.4.2** In the case of a Persistent PPC Default, Principal Project Company shall comply with <u>Section 15.8</u> (Cure Plan for Persistent PPC Default).

## 16.2 City Remedies for PPC Default

**16.2.1** Subject to the rights of Lenders under any Direct Agreement, upon occurrence of a PPC Default that has not been cured within the applicable cure period, if any, City may:

- Terminate this Agreement as provided in <u>Section 17.3</u> (Termination for PPC Default);
- (b) Exercise its step-in rights in <u>Section 16.2.5</u> (City Step-In Rights);
- (c) Recover any Losses on account of the occurrence of a PPC Default, regardless of when the Default Notice is given, whether the Losses accrue after the occurrence of the PPC Default or whether the PPC Default is subsequently cured;
- (d) Where such PPC Default is not cured within the applicable cure period, if any, specified in <u>Section 16.1.2.3</u> make demand upon and enforce any performance security, and make demand upon, draw on and enforce and collect any letter of credit, guaranty or other payment or performance security (including the Handback Requirements Reserve Account and City's rights to withhold payment) available to City under this Agreement with respect to the PPC Default in any order, in City's sole discretion, without Notice to Principal Project Company. City will apply the proceeds of any such action to the satisfaction of Principal Project Company's obligations under this Agreement, including payment of amounts due City;
- (e) Suspend the Work in whole or part in accordance with <u>Section 16.2.8</u> (Suspension of Work); or
- (f) Exercise any other remedies available under this Agreement or at law or in equity.

**16.2.2** Each right and remedy of City upon the occurrence of a PPC Default is cumulative as set out in <u>Section 16.2.10</u> (Cumulative, Non-Exclusive Remedies).

## 16.2.3 Immediate City Entry and Cure of Wrongful Use

Without Notice and without awaiting lapse of the period to cure, in the event of any PPC Default under <u>Section 16.1.1(p)</u>, City may enter and take control of the relevant portion of the Project to restore the permitted uses and reopen and continue operations for the benefit of the public, until such breach is cured or City terminates this Agreement; <u>provided</u>, <u>however</u>, that the foregoing shall be subject to the rights of Lenders under any Direct Agreement except in the event of an Emergency, an imminent safety risk, an unauthorized use of the Project by Principal Project Company or a PPC Default that materially precludes the use of the Project by City in the manner contemplated under this Agreement. Principal Project Company shall pay to City on demand City's Recoverable Costs in connection with such action, which payment will be reimbursed by City if a determination is ultimately made that no PPC Default occurred, promptly following such a determination. So long as City undertakes such action in good faith, even if under a mistaken belief in the occurrence of such a PPC Default, such action shall not be deemed unlawful or a breach of this Agreement, shall not expose City to any liability to Principal Project Company, other than the reimbursement obligation described above, and shall not entitle Principal Project Company to any other remedy except if City's action constitutes gross negligence, recklessness or willful misconduct. Principal Project Company acknowledges that City has a high priority, paramount public interest in maintaining the authorized uses of the Project and continuous access to the Project. City's good faith determination that such action is needed shall be deemed conclusive in the absence of clear and convincing evidence to the contrary. City will promptly relinquish control and possession of the relevant portion of the Project to Principal Project Company once City determines that such PPC Default has been cured.

## 16.2.4 Remedies for Failure to Meet Safety Standards or Perform Safety Compliance

**16.2.4.1** If at any time Principal Project Company or its Surety fails to meet any Safety Standard or timely perform Safety Compliance or if City and Principal Project Company cannot reach an agreement regarding the interpretation or application of a Safety Standard or the valid issuance of a Safety Compliance Order within a period of time acceptable to City, City may undertake or direct Principal Project Company to undertake any work required to ensure implementation of and compliance with Safety Standards as interpreted or applied by City or with the Safety Compliance Order. If at any time a condition or deficiency of the Project violates any Law respecting health, safety or right of use and access, including the Americans with Disabilities Act of 1990, 42 U.S.C. Sections 12101 et seq., and regulations of the United States Occupational Safety and Health Administration (OSHA), City may take any immediate corrective actions required.

**16.2.4.2** Subject to <u>Section 16.2.4.3</u>, to the extent that any work done under <u>Section 16.2.4.1</u> is undertaken by City and is reasonably necessary to comply with Safety Standards, perform valid previously issued Safety Compliance Orders or correct a violation of Law or a Regulatory Approval respecting health, safety or right of use and access, Principal Project Company shall pay to City on demand the costs of such work including City's Recoverable Costs in connection with such work. In such event, City (whether it undertakes the work or has directed Principal Project Company to undertake the work) shall have no obligation or liability to compensate Principal Project Company for any Losses Principal Project Company suffers or incurs arising from, resulting to or resulting from such work.

**16.2.4.3** To the extent that City requires Principal Project Company to perform Work under <u>Section 16.2.4.1</u> that is not reasonably necessary to comply with Safety Standards, perform valid previously issued Safety Compliance Orders or correct a violation of Law or Regulatory Approval respecting health, safety or right of use and access, such requirement shall be considered a City Change, and Principal Project Company's obligation to pay City's Recoverable Costs shall not include City's costs relating to the City Change.

**16.2.4.4** Notwithstanding anything to the contrary contained in this Agreement and without limiting City's other rights and remedies under this Agreement, City shall have the rights and remedies specified in <u>Section 16.2.4.5</u> if, in the good faith judgment of City:

 Principal Project Company has failed to meet any Safety Standards or perform Safety Compliance and the failure results in an Emergency or danger to persons or property; or (b) Principal Project Company is not then diligently taking all necessary steps to rectify or deal with such Emergency or danger.

**16.2.4.5** Upon the occurrence of any event described in <u>Section 16.2.4.4</u>, City may without Notice and without awaiting lapse of the period to cure any breach:

- (a) immediately take such action as may be reasonably necessary to rectify the Emergency or danger, in which event Principal Project Company shall pay to City on demand the cost of such action, including City's Recoverable Costs, which payment will be reimbursed by City if a determination is ultimately made that Principal Project Company did not fail to meet applicable Safety Standards or to perform Safety Compliance; or
- (b) suspend Work and/or close or cause to be closed any and all portions of Project affected by the Emergency or danger.

**16.2.4.6** So long as City undertakes any action under <u>Section 16.2.4.5</u> in good faith, even if under a mistaken belief in the occurrence of such failure or existence of an Emergency or danger, such action shall not be deemed unlawful or a breach of this Agreement, shall not expose City to any liability to Principal Project Company, other than the reimbursement obligation described in <u>Section 16.2.4.5</u>, and shall not entitle Principal Project Company to any other remedy, except if City's action was undertaken in bad faith or constitutes gross negligence, recklessness or willful misconduct.

**16.2.4.7** Principal Project Company acknowledges that City has a high priority, paramount public interest in protecting public and worker safety at the Project and adjacent areas. City's good faith determination of the existence of such a failure, Emergency or danger shall be deemed conclusive in the absence of clear and convincing evidence to the contrary. Immediately following rectification of such Emergency or danger, as determined by City, City will allow the Work to continue or such portions of the Project to reopen, as applicable.

# 16.2.5 City Step-in Rights

**16.2.5.1** If Principal Project Company has not fully and completely cured a PPC Default by the expiration of the applicable cure period, if any, City may, subject to the terms of any Direct Agreement and in accordance with the terms and conditions of this <u>Section 16.2.5</u> (City Step-in Rights), pay and perform all or any portion of Principal Project Company's obligations under the Contract Documents that are the subject of such PPC Default.

**16.2.5.2** Upon exercising its rights pursuant to <u>Section 16.2.5.1</u>, City may, to the extent necessary to cure a PPC Default:

- (a) perform or attempt to perform, or cause to be performed, such Work;
- (b) employ security guards and other safeguards to protect the Project;
- (c) spend such sums as City deems reasonably necessary to employ and pay such architects, engineers, consultants and contractors and obtain materials and equipment as may be required to perform such Work, without obligation or liability to Principal Project Company or any Contractors for loss of

opportunity to perform the same Work or supply the same materials and equipment;

- (d) in accordance with Section 16.2.7 (Performance Bond), draw on and use proceeds from the Performance Bond and any other available security or source of funds available to Principal Project Company for the purposes set forth in this Section 16.2.5.2 including amounts held in an operating account, to the extent such instruments provide recourse to pay such sums, provided City's right to access amounts held in an operating account shall not include a security interest in such funds nor shall the exercise of such right by City interfere with the right of the Lenders, if any, under the Security Documents and the Direct Agreement to access such funds;
- (e) execute all applications, certificates and other documents as may be required;
- (f) make decisions respecting, assume control over and continue Work as may be reasonably required;
- (g) modify or terminate any contractual arrangements in Principal Project Company's Contracts in City's sole discretion, without liability for termination fees, costs or other charges in accordance with the terms of those Contracts, including the requirements for each Contract listed in <u>Section 9.3.2</u> (Key Contract Provisions);
- (h) meet with, coordinate with, direct and instruct Contractors and Suppliers, process invoices and applications for payment from Contractors and Suppliers, pay Contractors and Suppliers, and resolve claims of Contractors and Suppliers;
- (i) take any and all other actions it may consider necessary to effect cure and perform the Work; and
- (j) prosecute and defend any action or proceeding incident to the Work.

**16.2.5.3** Principal Project Company shall reimburse City on demand for its Recoverable Costs incurred in connection with the performance of any act or work authorized by this <u>Section 16.2.5</u> (City Step-in Rights).

**16.2.5.4** In addition to its continuing ownership of the Project Site and rights to have access to the Project Site throughout the Term, including the rights described in <u>Section 6.7</u> (Coordination, Cooperation and Access), City and its Authorized Representatives, contractors, subcontractors, vendors and employees shall have the right to enter onto all Temporary Areas, exercisable at any time or times without Notice, for the purpose of carrying out City's step-in rights under <u>Section 16.2.3</u> (Immediate City Entry and Cure of Wrongful Use) and this <u>Section 16.2.5</u> (City Step-in Rights). Principal Project Company grants City a perpetual, non-rescindable right of entry onto the Temporary Areas for such purpose.

**16.2.5.5** If City exercises any right to pay or perform under this <u>Section 16.2.5</u> (City Step-in Rights), it nevertheless shall have no liability to Principal Project Company for the sufficiency, adequacy or quality of any such payment or performance, or for any effect of such payment or

performance on the Work or the Project, unless caused by the gross negligence, recklessness or willful misconduct of City. City and its Authorized Representatives, contractors, subcontractors, vendors and employees shall have no liability to Principal Project Company for any inconvenience or disturbance arising out of any entry onto the Project Site or Temporary Areas as contemplated by <u>Section 16.2.3</u> (Immediate City Entry and Cure of Wrongful Use) or this <u>Section 16.2.5</u> (City Step-in Rights), provided that the foregoing shall not absolve any Person of liability as a matter of law for its gross negligence, recklessness or willful misconduct.

**16.2.5.6** City's rights under this <u>Section 16.2.5</u> (City Step-in Rights) (but not City's rights under <u>Section 16.2.3</u> (Immediate City Entry and Cure of Wrongful Use) regarding City's immediate right of entry and right to cure wrongful use of the Project) are subject to the right of any Surety to assume performance and completion of all bonded work under a Performance Bond.

**16.2.5.7** If City takes action described in this <u>Section 16.2.5</u> (City Step-in Rights) and it is later finally determined that City lacked the right to do so because there did not occur a PPC Default or because Principal Project Company had previously fully cured the default in accordance with this Agreement, then City's action shall be treated as a City Change.

## 16.2.6 Damages; Offset

**16.2.6.1** Except as limited by City's agreement to liquidate certain damages as specified in this Agreement, and subject to the limitations in <u>Section 16.6</u> (Limitation on Consequential Damages), City shall be entitled to recover any and all damages available at law on account of the occurrence of a PPC Default. Principal Project Company shall owe any such damages that accrue after the occurrence of PPC Default regardless of when Default Notice is given or whether PPC Default is subsequently cured.

**16.2.6.2** In the case of a termination for PPC Default, City may deduct and offset any damages owing to it under the Contract Documents from and against any amounts City may owe to Principal Project Company. If the amount of damages owing to City is not liquidated or known with certainty at the time a payment is due from City to Principal Project Company with respect to such termination for a PPC Default, City may deduct and offset the amount it reasonably estimates will be due, subject to City's obligation to adjust such deduction or offset when the amount of damages owing to City is liquidated or becomes known with certainty.

**16.2.6.3** Damages owed to City under this <u>Section 16.2.6</u> (Damages; Offset) shall bear interest at the Late Payment Rate from and after the date any amount becomes due to City until the date paid.

## **16.2.6.4** Without limiting <u>Section 16.2.6.2</u>:

- (a) the amount of any damages attributable to Principal Project Company may be deducted or offset from periodic payments owing by City under <u>Section 11.1</u> (Milestone Payment) (including damages attributable to any PPC Default that concerns the D&C Work and is the subject of a Notice delivered to Principal Project Company before the date such payments are owing), subject to <u>Section 2.4</u> of <u>Exhibit 4A</u> (Milestone Payment Mechanism); and
- (b) the amount of any damages attributable to Principal Project Company may be deducted or offset from the Availability Payments (including damages

attributable to any PPC Default), provided that in certain cases damages are liquidated by the Deductions under <u>Exhibit 4B</u> (Availability Payment Mechanism).

**16.2.6.5** If the amount of damages owing to City from Principal Project Company is not liquidated or known with certainty at the time the payment is due, City may deduct and offset up to 105% of the amount it reasonably estimates will be due, in which case, the Parties shall adjust such deduction or offset when the amount of damages owing to City becomes known with certainty, with interest at the Late Payment Rate payable by City on excess amounts withheld and interest at the same rate payable by Principal Project Company on any shortfall.

**16.2.6.6** City's right of offset includes all amounts paid by City to satisfy, discharge and defend against any claim of lien, stop Notice, equitable lien or any other demand for payment or security made or filed with City, City's property or the Project by any Person claiming that any PPC-Related Entity has failed to perform its contractual obligations or to make payment for any obligation incurred for or in connection with the Work, provided that no such offset shall be made if Principal Project Company has filed surety bonds fully releasing City and City's property from such claim or lien under applicable Law.

## 16.2.7 Performance Bond

**16.2.7.1** Subject to <u>Sections 10.5.2</u> and <u>16.2.6.2</u>, upon the occurrence of a PPC Default and expiration, without full and complete cure, of the applicable cure period, if any, under <u>Section</u> <u>16.1.2</u> (Default Notice and Cure Periods), without waiving or releasing Principal Project Company from any obligations or limiting other remedies that may be available to City, City may make demand upon and enforce any Performance Bond, and make demand upon, draw on and enforce and collect any letter of credit, guaranty or other payment or performance security available to City under this Agreement with respect to such PPC Default in any order. City will apply the proceeds of any such action to the satisfaction of Principal Project Company's obligations under this Agreement, including payment of amounts due to City.

**16.2.7.2** If City is an additional obligee under a Performance Bond, or is a transferee beneficiary under any letter of credit or guaranty, then City will forbear from exercising remedies as additional obligee or transferee beneficiary so long as Principal Project Company or the Collateral Agent commences the good faith, diligent exercise of remedies within 10 days after City delivers Notice to Principal Project Company and the Collateral Agent of City's intent to make a claim under such security, letter of credit or guaranty and subsequently continues such good faith, diligent exercise of remedies until the default is cured.

**16.2.7.3** Section 16.2.7.2 shall not apply where access to a bond, letter of credit, guaranty or other payment or performance security is for the purpose of satisfying damages owing to City, in which case City shall be entitled to make demand, draw, enforce and collect regardless of whether PPC Default is subsequently cured.

**16.2.7.4** City will provide Notice to Principal Project Company at the same time or promptly after City takes any action to make demand upon, draw on, enforce or collect any Performance Bond.

## 16.2.8 Suspension of Work

**16.2.8.1** Subject to the rights of the Lenders as provided in the Direct Agreement, in addition to any other right to suspend the Work under the Contract Documents, City may suspend all or part of the Work by Notice to Principal Project Company if Principal Project Company fails to cure and correct, within the applicable cure period available to Principal Project Company (if any), the following:

- failure to perform the Work in compliance with the Contract Documents, where such failure is not substantially cured within 15 days after City delivers Notice thereof to Principal Project Company;
- (b) failure to comply with any Law or Regulatory Approval (including failure to implement protective measures for Threatened or Endangered Species, failure to handle, preserve and protect paleontological and cultural (including archaeological and historic) resources, or failure to handle Hazardous Materials, in accordance with applicable Laws and Regulatory Approvals);
- (c) performance of Design Work before achieving all conditions in <u>Section 7.4.3</u> (Commencement of Non-Construction Work), or performance of Construction Work before all conditions precedent to commencement of Construction Work have been met;
- (d) discovery of Nonconforming Work not corrected within 15 days after City delivers Notice of such Nonconforming Work to Principal Project Company or, if correction will require more than 15 days, failure of Principal Project Company to deliver to City, within 15 days of said Notice, a plan acceptable to City for correction of the Nonconforming Work or to diligently execute and complete such plan;
- (e) failure to pay in full when due sums owing to any Contractor or Supplier for services, materials or equipment, except only for retainage provided in the relevant Contract and amounts in dispute, which amounts in dispute may include amounts withheld on account of damage caused by such Contractor or Supplier or arising out of such Contractor or Supplier's failure to comply with the relevant Contract;
- (f) failure to provide proof of required insurance coverage under <u>Section 10.1.2.4</u> (Verification of Coverage);
- (g) failure to deliver or maintain Payment Bonds or Performance Bonds;
- (h) the existence of conditions unsafe for workers, other Project personnel or the general public, including failures to comply with Safety Standards or perform Safety Compliance in accordance with <u>Section 16.2.4</u> (Remedies for Failure to Meet Safety Standards or Perform Safety Compliance) (and in any such case the order of suspension may be issued without awaiting any cure period);

- (i) failure to carry out and comply with Unilateral Change Orders, where such failure is not substantially cured within 15 days after City delivers Notice thereof to Principal Project Company; and
- (j) for the reasons specified in <u>Sections 9.8.6, 10.1.2.4(c)</u> and <u>16.2.4.4</u>.

**16.2.8.2** In addition to the protections from liability under <u>Section 16.2.4.6</u>, City shall have no liability to Principal Project Company, and Principal Project Company shall have no right to make any Claim against City, due to any suspension under <u>Section 16.2.8.1</u>

**16.2.8.3** City shall have the right and authority to order suspension of Work, in whole or in part, for reasons other than those in <u>Section 16.2.8.1</u> If City purports to order suspension of Work under <u>Section 16.2.8.1</u> but it is finally determined under the Contract Dispute Procedures that no grounds under <u>Section 16.2.8.1</u> exist for such suspension, then it shall be deemed a suspension under this <u>Section 16.2.8.3</u>

**16.2.8.4** Principal Project Company shall promptly comply with any such suspension order, even if Principal Project Company disputes the grounds for suspension. Principal Project Company shall promptly recommence the Work upon receipt of Notice from City directing Principal Project Company to resume work. City will lift the suspension order promptly after Principal Project Company fully cures and corrects the applicable breach or failure to perform or any other reason for the suspension order ceases to apply.

**16.2.8.5** Without limiting Principal Project Company's rights with respect to a Compensable Delay Event under <u>clause (q)</u> of the definition thereof or a Compensable Relief Event under <u>clause (p)</u> of the definition thereof, in case of suspension of work for any cause, Principal Project Company shall be responsible for the Project and shall take such precautions as may be necessary to prevent loss or damage to the materials, equipment and Work, provide for normal drainage and shall erect any necessary temporary structures, signs, or other facilities at Principal Project Company's expense.

## 16.2.9 Other Rights and Remedies

City shall also be entitled to exercise any other rights and remedies available under this Agreement, or available at law.

## 16.2.10 Cumulative, Non-Exclusive Remedies

Subject to the exception specified in <u>Section 16.1.1(j)</u> regarding failures that constitute a Noncompliance Event or Failure Event not being considered defaults under <u>Section 16.1.1(j)</u>, each right of City under this Agreement shall be cumulative and shall be in addition to every other right provided under this Agreement or at law, and the exercise or beginning of the exercise by City of any one or more of any of such rights or remedies shall not preclude the simultaneous or later exercise by City of any or all other such rights or remedies.

## 16.3 Default by City; Cure Periods

#### 16.3.1 City Default

City shall be in breach under this Agreement upon the occurrence of any one or more of the following events or circumstances (each a "**City Default**"):

- (a) City fails to make any payment due to Principal Project Company under this Agreement when due, provided that such payment is not subject to a Contract Dispute;
- (b) City ceases to be legally authorized to make any undisputed payment to the Principal Project Company under this Agreement; provided, however, that a failure to appropriate or an injunction or other legal or administrative constraint on making such payment shall not be considered a City Default; or
- (c) any representation or warranty made by City to Principal Project Company in the Agreement is false in any material respect, materially misleading or inaccurate in any material respect when made or omits material information when made.

#### 16.3.2 Cure Periods

City shall have the following cure periods to cure City Defaults:

**16.3.2.1** Respecting a City Default under <u>Section 16.3.1(a)</u>, a period of 45 days after Principal Project Company delivers to City Notice of such City Default; and

**16.3.2.2** Respecting a City Default under <u>Section 16.3.1(c)</u>, a period of 90 days after Principal Project Company delivers to City Notice of such City Default; provided that:

- (a) if the City Default is of such a nature that the cure cannot with diligence be completed within such time period and City has commenced meaningful steps to cure promptly after receiving the default Notice, City shall have such additional period of time, up to a maximum cure period of 180 days, as is reasonably necessary to diligently effect cure; and
- (b) cure will be regarded as complete when the adverse effects of the breach are cured.

#### 16.4 Principal Project Company Remedies for City Breach

#### 16.4.1 Termination

Refer to <u>Section 17.4</u> (Principal Project Company Rights to Terminate) for provisions regarding Principal Project Company's right to terminate for City Default.

## 16.4.2 Interest on Late Payment

If City fails to make payments of undisputed Availability Payments or the Milestone Payment that are due and owing to Principal Project Company under this Agreement, then Principal Project Company shall be entitled to interest at the Late Payment Rate commencing 90 days after the due date thereof in accordance with <u>Section 11.7</u> (Interest on Late Payments and Overpayments).

## 16.4.3 Limitations on Remedies

**16.4.3.1** Principal Project Company shall have no right to seek, and irrevocably waives and relinquishes any right to, non-monetary relief against City, except for (a) any sustainable action for relief available in mandamus, (b) any sustainable action to stop, restrain or enjoin use, reproduction, duplication, modification, adaptation or disclosure of PPC Intellectual Property in violation of the licenses granted, or to specifically enforce City's duty of confidentiality, under <u>Section 21.4</u> (Intellectual Property), (c) declaratory relief under the Contract Dispute Procedures declaring the rights and obligations of the Parties under the Contract Documents, or (d) declaratory relief under the Contract Dispute that shall bind the Parties.

**16.4.3.2** If City wrongfully withholds an approval or consent required under this Agreement, or wrongfully issues an objection to or disapproval of a Submittal or other matter under this Agreement, Principal Project Company's sole remedies against City shall be compensation and extensions of time, each to the extent provided in <u>Articles 13</u> (General Provisions Applying to Delay Events and Relief Events) and <u>14</u> (Compensation and Other Relief for Delay Events and Relief Events). Principal Project Company shall have no right to suspend Work.

## 16.4.4 Remedies at Law and in Equity

Subject to <u>Section 16.4.3</u> (Limitations on Remedies) and except as specifically provided otherwise in this Agreement, upon breach of this Agreement by City, Principal Project Company may exercise any remedies available at law or in equity.

# 16.4.5 Principal Project Company Right to Suspend

Principal Project Company may suspend Work based on City's failure, after required Notice and the expiration of the applicable cure period, to pay undisputed amounts owing to Principal Project Company with a cumulative value of \$1,000,000 or more, subject to the following:

- (a) Principal Project Company shall provide City with Notice regarding its intent to suspend at least 30 days before implementing the suspension, and may implement the suspension only if the breach remains uncured as of the suspension date; and
- (b) a suspension by Principal Project Company under this <u>Section 16.4.5</u> (Principal Project Company Right to Suspend) shall be deemed to be a suspension of Work order issued by City for its convenience under <u>Section 16.2.8.2</u>, and the suspension order shall be deemed lifted upon Principal Project Company's receipt of payment in full of all undisputed amounts owing.

## 16.5 No Duplicative Payment

Notwithstanding any other provisions of this Agreement to the contrary, neither Party shall be entitled to recover compensation or make a claim under this Agreement with respect to any loss that it has incurred to the extent that it has already been compensated with respect to that loss under this Agreement or otherwise.

### 16.6 Limitation on Consequential Damages

**16.6.1** Except as stated here, Principal Project Company shall have no liability to City for any type of special, consequential, or incidental damages arising out of or connected with Principal Project Company's performance of the Work. This limit of liability applies under all circumstances including the breach, completion, termination, suspension or cancellation of the services under this Agreement, and negligence or strict liability of Principal Project Company. This limit of liability, however, shall not apply to, limit or preclude:

- Principal Project Company's obligation to pay liquidated damages as set forth in the Contract Documents, including any credits, deductions or offsets for failure to meet performance requirements or pursuant to the noncompliance/deduction regime;
- (b) damages caused by a PPC Fault (other a PPC Fault that is a breach by a PPC-Related Entity of any obligations under the Contract Documents);
- (c) Principal Project Company's obligations to indemnify, defend and hold City and other Indemnities harmless for Third Party Claims as set forth in the Contract Documents;
- (d) Principal Project Company's liability damages that fall within the insurance coverages required under this Agreement;
- (e) Principal Project Company's liability for statutory damages imposed by City as a Governmental Entity under applicable Law;
- (f) statutory fines, penalties, and statutory damages, including punitive damages, treble damages, and statutory attorney fees and costs, including those due as a result of violations of environmental regulations and laws;
- (g) amounts that Principal Project Company is obligated to reimburse City under the express provisions of this Agreement, including any Recoverable Costs and any express interest thereon;
- (h) damages and other liability arising under claims by Third Parties for loss or damage to property or personal injuries, including wrongful death; or
- (i) damages and other liability for infringement of any Intellectual Property right.

# ARTICLE 17. TERMINATION

### 17.1 Termination for Convenience; Condemnation

**17.1.1** City may terminate this Agreement in whole, but not in part, if City determines in its sole discretion that termination is in City's best interest (a "**Termination for Convenience**").

**17.1.2** City may exercise a Termination for Convenience by delivering to Principal Project Company a Notice of Termination for Convenience specifying the election to terminate and its effective date, which shall not be earlier than 30 days after the date of delivery of such Notice.

**17.1.3** In the event of a Termination for Convenience, City shall pay compensation to Principal Project Company (or to the Collateral Agent as provided in the Direct Agreement) in an amount (without double counting) equal to:

- (i) all amounts shown in the Financial Model as payable by Principal Project Company to Equity Members from the Early Termination Date, either as Distributions on Committed Investments or as payments of interest or repayments of principal made by Principal Project Company in respect of Equity Members Funding Agreements, each amount discounted back at the Equity IRR from the date on which it is shown to be payable in the Financial Model to the date on which the Termination Compensation is paid, minus (ii) Deferred Equity Amounts, discounted back at the Equity IRR from the date on which it is shown to be payable in the Financial Model to the date on which the Termination Compensation is paid; plus
- (b) Lenders' Liabilities; plus
- (c) PPC Employee and Contractor Breakage Costs; minus
- (d) Account Balances; minus
- (e) Insurance Proceeds; minus
- (f) any Deductions to the extent not deducted in full from the Milestone Payment or Quarterly Availability Payments.
- **17.1.4** If:
  - (a) City confiscates, sequesters, condemns or appropriates the PPC's Interest or any material part thereof; or
  - (b) as a direct result of a City ordinance Principal Project Company is unable to perform all or substantially all of its obligations under the Contract Documents for a period of 180 consecutive days or more,

then this Agreement shall be deemed terminated for convenience and City shall pay compensation to Principal Project Company in the amount described in <u>Section 17.1.3</u>.

## 17.2 Termination for Force Majeure Termination Events or Insurance Unavailability

## 17.2.1 Notice of Conditional Election to Terminate – Force Majeure Termination Events

Subject to <u>Section 17.2.2</u>, either Party may deliver to the other Party Notice of its conditional election to terminate this Agreement ("**Notice of Conditional Termination**") if a Force Majeure Event (excluding a Force Majeure Event under <u>clauses (i)</u> or <u>(i)</u> of the definition of Force Majeure Event) ("**Force Majeure Termination Event**") has occurred and:

- (i) the Notice of Conditional Termination is delivered before the Substantial Completion Date; (ii) as a direct result of the Force Majeure Termination Event, Principal Project Company is unable to perform all or substantially all of its obligations under the Contract Documents for a period of 255 consecutive days or more; and (iii) such inability to perform its obligations is not attributable to a concurrent non-Force Majeure Termination Event; or
- (b) (i) the Notice of Conditional Termination is delivered on or after the Substantial Completion Date; (ii) as a direct result of the Force Majeure Termination Event, all or substantially all of the Project has become and remains inoperable for a period of 255 consecutive days or more; and (iii) such suspension of operations is not attributable to another concurrent non-Force Majeure Termination Event; and
- (c) Principal Project Company could not have mitigated or cured such result through the exercise of diligent efforts; and
- (d) such result is continuing at the time of delivery of the written Notice; and
- (e) the written Notice sets forth in reasonable detail a description of the Force Majeure Termination Event, a description of the direct result and its duration, and a statement that the notifying Party's intends to terminate this Agreement.

#### 17.2.2 No Right to Termination Election

Notwithstanding the foregoing, if, following the occurrence of any Force Majeure Termination Event that results in damage or partial destruction of the Project:

- (a) the conditions listed in <u>Section 17.2.1</u> (Notice of Conditional Election to Terminate Force Majeure Termination Events) are satisfied;
- (b) insurance proceeds are available to fund work required to remedy the effects of the Force Majeure Termination Event; and
- (c) the Parties agree to a restoration plan in respect of such work required to remedy the effect of the Force Majeure Termination Event,

then neither Party shall have the right to elect to terminate this Agreement pursuant to <u>Section 17.2.1</u> (Notice of Conditional Election to Terminate – Force Majeure Termination Events).

## 17.2.3 Principal Project Company Options Upon City Notice

**17.2.3.1** If City delivers a Notice of Conditional Termination, Principal Project Company shall have the option either to accept such Notice or to continue this Agreement in effect by delivering to City Notice of Principal Project Company's choice not later than 30 days after City delivers its Notice. If Principal Project Company does not deliver such Notice within such 30 day period, then Principal Project Company shall be deemed to have accepted City's election to terminate this Agreement.

**17.2.3.2** If Principal Project Company delivers timely Notice under <u>Section 17.2.3.1</u> choosing to continue this Agreement in effect, then:

- (a) City shall have no obligation to compensate Principal Project Company for any Extra Work Costs, Financing Delay Costs, Delay Costs, costs of restoration, repair or replacement arising out of the Force Majeure Termination Event and incurred after the date on which City gives written Notice of conditional election to terminate under this <u>Section 17.2</u> (Termination for Force Majeure Events or Insurance Unavailability), except as provided in <u>Section 17.2.3.2(b)</u>;
- (b) if the Force Majeure Termination Event occurred before the Substantial Completion Date and resulted in a Relief Event Delay, Principal Project Company shall be entitled to an extension of the applicable deadlines in accordance with <u>Section 14.1.2</u> (Extension of Deadlines for General Delay Events); and
- (c) this Agreement shall continue in full force and effect and City's election to terminate shall not take effect.

## 17.2.4 City Options Upon Principal Project Company Notice

**17.2.4.1** If Principal Project Company delivers a Notice of Conditional Termination, including an estimate (with supporting documentation) of the compensation that would be paid or reimbursed to Principal Project Company under <u>Section 17.2.4.2</u>, City shall have the option either: (a) to accept such Notice; or (b) to continue this Agreement in effect if City in its reasonable discretion determines that the Project can be completed or re-opened, as applicable, on a commercially reasonable basis. City shall exercise such option by delivering to Principal Project Company written Notice of City's choice not later than 30 days after Principal Project Company delivers its Notice of Conditional Termination. If City does not deliver such written Notice within such 30-day period, then it shall be conclusively deemed to have accepted Principal Project Company's election to terminate this Agreement.

**17.2.4.2** If City delivers timely written Notice under <u>Section 17.2.4.1</u> choosing to continue this Agreement in effect, then:

- (a) subject to <u>Section 13.8</u> (Burden of Proof and Mitigation), City shall be obligated to pay or reimburse Principal Project Company an amount equal to (without double-counting):
  - (i) where the Force Majeure Termination Event occurs prior to Substantial Completion, Financing Delay Costs and the Extra Work Costs directly

caused by the Force Majeure Termination Event which are incurred after the date Principal Project Company delivers its written Notice of conditional election to terminate; plus

- where the Force Majeure Termination Event occurs prior to Substantial Completion, Delay Costs which are incurred after the date Principal Project Company delivers its written Notice of conditional election to terminate; plus
- (iii) such other amounts so as to result in Principal Project Company achieving the same Equity IRR (with reference to the Base Case Equity IRR) from and after the date Principal Project Company delivers its written Notice of conditional election to terminate as if such Force Majeure Termination Event had not occurred; and
- (b) From and after the date which is 255 days after the occurrence of the Force Majeure Termination Event, Principal Project Company shall not be subject to Deductions or Noncompliance Points with respect to such Force Majeure Termination Event; and
- (c) this Agreement shall continue in full force and effect and Principal Project Company's election to terminate shall not take effect for the period specified in City's written Notice under this <u>Section 17.2.4</u> (City Options Upon Principal Project Company Notice) or such longer period as may be mutually agreed to in writing by the Parties.

# 17.2.5 No Waiver

No election by Principal Project Company under <u>Section 17.2.3</u> (Principal Project Company Options Upon City Notice) or by City under <u>Section 17.2.4</u> (City Options Upon Principal Project Company Notice) to continue this Agreement in effect shall prejudice or waive such Party's right to thereafter, at any time, give a Notice of conditional election to terminate with respect to the same or any other Force Majeure Termination Event.

# 17.2.6 Termination for Insurance Unavailability

If it becomes apparent that insurance required under the Contract Documents is not available as a result of Insurance Unavailability, City may deliver to Principal Project Company Notice of its election to terminate this Agreement for Insurance Unavailability and the effective date of the termination, which shall not be earlier than 30 days after the date of delivery of such Notice. Such Notice shall include reasonable details regarding the affected coverages and associated risks, as well as the estimated cost of premiums if Commercially Reasonable Insurance Rates are not available. In the event of a termination for Insurance Unavailability, City shall pay compensation to Principal Project Company (or to the Collateral Agent as provided in the Direct Agreement) calculated in accordance with <u>Section 17.2.8</u> (Termination Compensation for Force Majeure Termination Events and Insurance Unavailability).

## 17.2.7 Concurrent Notices

If City and Principal Project Company deliver concurrent Notices of Conditional Termination under this <u>Section 17.2</u> (Termination for Force Majeure Termination Events or Insurance Unavailability), Principal Project Company's Notice shall prevail. Notices shall be deemed to be concurrent if each Party sends its Notice before actually receiving the Notice from the other Party. Knowledge of the other Party's Notice obtained before actual receipt of the Notice shall have no effect on determining whether concurrent Notice has occurred.

# 17.2.8 Termination Compensation for Force Majeure Termination Events and Insurance Unavailability

If either Party accepts, or is deemed to accept, the other Party's conditional election to terminate under <u>Section 17.2.3</u> (Principal Project Company Options Upon City Notice) or <u>Section 17.2.4</u> (City Options Upon Principal Project Company Notice), as applicable, then this Agreement shall be deemed terminated on an Early Termination Date that is 30 days after the date of acceptance or deemed acceptance of the conditional election to terminate; and Principal Project Company will be entitled to compensation calculated as follows (calculated at the Early Termination Date and without double-counting):

- (a) an amount equal to its Equity Investments less Distributions paid to the Equity Members, which shall never be a negative number; plus
- (b) Lenders' Liabilities; plus
- (c) PPC Employee and Contractor Breakage Costs; minus
- (d) Account Balances; minus
- (e) Insurance Proceeds; minus
- (f) any Deductions to the extent not deducted in full from Milestone Payments or Quarterly Availability Payments.

## 17.3 Termination for PPC Default

# 17.3.1 PPC Defaults Triggering City Termination Rights

**17.3.1.1** Subject to the rights of the Lenders pursuant to any Direct Agreement, in the event that:

- (a) any PPC Default occurs and has not been cured within the applicable cure period, if any, set out in <u>Section 16.1.2</u> (Default Notice and Cure Periods) or (if relevant) in accordance with any cure plan accepted by City under <u>Section</u> <u>15.8</u> (Cure Plan for Persistent PPC Default); or
- (b) any PPC Default occurs for which there is no cure period under <u>Section</u> <u>16.1.2</u> (Default Notice and Cure Periods);

# (each a "Default Termination Event")

City may terminate this Agreement with immediate effect upon written Notice to Principal Project Company.

**17.3.1.2** Termination of this Agreement for a Default Termination Event in accordance with this <u>Section 17.3</u> (Termination for PPC Default) will take effect on the date stated in the Notice given by City to Principal Project Company under <u>Section 17.3.1.1</u>.

## 17.3.2 Compensation to Principal Project Company

**17.3.2.1** Subject to <u>Section 17.3.2.2</u>, if City issues a Notice of termination of this Agreement due to a PPC Default:

- (a) if termination occurs before the Substantial Completion Date, Principal Project Company will be entitled to compensation in an amount equal to the lesser of:
  - (i) the D&C Work Value; and
  - (ii) the amount equal to:
    - (A) Lenders' Liabilities; minus
    - (B) Account Balances; minus
    - Insurance Proceeds (excluding proceeds of personal injury, property damage or other Third Party liability insurance payable to or for the account of a Third Party); minus
    - (D) any D&C Period Deductions to the extent not deducted in full from Milestone Payments;
- (b) if termination occurs on or after the Substantial Completion Date, Principal Project Company will be entitled to compensation equal to the amount calculated at the Early Termination Date (without double-counting) as follows:
  - (i) eighty percent (80%) of Lenders' Liabilities; minus
  - (ii) Maintenance Rectification Costs; minus
  - (iii) Account Balances; minus
  - (iv) Deferred Equity Amounts; minus
  - (v) Insurance Proceeds; minus;
  - (vi) any Deductions to the extent not deducted in full from the Milestone Payment or Quarterly Availability Payments; plus
  - (vii) the balance standing to the credit of the Handback Requirements Reserve Account on the Early Termination Date.

**17.3.2.2** If the calculation described in <u>Section 17.3.2.1</u> results in a negative number, the negative value shall represent damages recoverable by City in accordance with <u>Section 16.2.6.2</u>.

# 17.3.3 Finality

If City issues Notice of termination of this Agreement due to a PPC Default, termination shall be effective and final regardless of whether City is correct in determining that it has the right to terminate for PPC Default. If it is determined that City lacked such right, then such termination shall be treated as a Termination for Convenience as provided in <u>Section 17.1</u> (Termination for Convenience; Condemnation) for the purpose of determining the Termination Compensation due.

# 17.4 Principal Project Company Rights to Terminate

## 17.4.1 Termination for City Default

**17.4.1.1** If a City Default under <u>Section 16.3.1</u> (City Default) remains uncured following (a) Notice and expiration of the applicable cure period under <u>Section 16.3.2</u> (Cure Periods), and (b) Principal Project Company's compliance with the warning requirements set forth in <u>Section 17.4.1.2</u>, Principal Project Company shall have the right to terminate this Agreement, effective immediately upon delivery of written Notice of termination to City. In the event of such termination, City shall pay compensation to Principal Project Company equal to the amount described in <u>Section 17.1.3</u>.

**17.4.1.2** Principal Project Company shall provide a warning Notice to City at least 15 Business Days before terminating, which Notice may not be delivered until 30 Business Days after delivery of the Notice under <u>Section 16.3.2.1</u> Principal Project Company shall provide a second warning Notice to City at least five Business Days before terminating, which Notice may not be delivered until 10 Business Days after delivery of the first warning Notice. If City fails to effect cure within five Business Days after the date of delivery of the second warning Notice, Principal Project Company shall have the right to terminate this Agreement in accordance with <u>Section 17.4.1.1</u>

## 17.4.2 Termination for Suspension of Work

**17.4.2.1** If City issues a suspension order under <u>Section 16.2.8.3</u> that suspends the Work for a period of 270 days or more, and provided that (a) such suspension is not the result of any PPC Fault; and (b) Principal Project Company has delivered a warning Notice to City at least 15 days before terminating, Principal Project Company shall have the right to terminate this Agreement, effective immediately upon delivery of written Notice of termination to City. In the event of such termination, Principal Project Company will be entitled to compensation equal to the amount described in <u>Section 17.1.3</u>.

**17.4.2.2** Principal Project Company may not terminate under this <u>Section 17.4.2</u> (Termination for Suspension of Work) if, at the time Principal Project Company's right to terminate would arise, circumstances exist entitling either Party to terminate under <u>Sections 17.2</u> (Termination for Force Majeure Termination Events or Insurance Unavailability), <u>17.3</u> (Termination for PPC Default), <u>17.4.3</u> (Termination Due to Court Ruling) or <u>17.5</u> (Termination if Financial Close Fails to Occur).

# 17.4.3 Termination Due to Court Ruling

Termination Due to Court Ruling means, and becomes effective upon: (a) issuance of a final order by a court of competent jurisdiction after exhaustion of all appeals to the effect that this Agreement is void, voidable, and/or unenforceable or impossible to perform in its entirety for reasons beyond the reasonable control of Principal Project Company; or (b) issuance of a final order by a court of competent jurisdiction after exhaustion of all appeals upholding the binding effect on Principal Project Company and/or City of a Change in Law that causes impossibility of performance of a fundamental obligation by Principal Project Company or City under the Contract Documents or impossibility of exercising a fundamental right of Principal Project Company will be entitled to compensation in an amount equal to the amount described in <u>Section 17.2.8</u> (Termination Compensation for Force Majeure Termination Events and Insurance Unavailability); provided that if the Termination Due to Court Ruling is caused solely and directly by a City Default or a City Fault, Principal Project Company will be entitled to compensation in the amount described in <u>Section 17.1.3</u>.

## 17.5 Termination if Financial Close Fails to Occur

<u>Sections 3.6</u> (No-Fault Termination) and <u>3.7</u> (Principal Project Company's Failure to Achieve Financial Close) set forth the terms applicable to termination before Financial Close.

## 17.6 Termination Procedures and Duties

The provisions of this <u>Section 17.6</u> (Termination Procedures and Duties) shall apply upon expiration of the Term or earlier termination of this Agreement. Principal Project Company shall timely comply with such provisions independently of, and without regard to, the timing for determining, adjusting, settling and paying any amounts due Principal Project Company or City on account of termination. If City determines that Principal Project Company has failed to comply with the provisions of this <u>Section 17.6</u> (Termination Procedures and Duties), then upon Notice from City to Principal Project Company making reference to this <u>Section 17.6</u> (Termination Procedures and Duties), Principal Project Company acknowledges and agrees it shall be deemed to have surrendered its access rights in accordance with the Contract Documents.

# 17.6.1 Performance of Work Pending Early Termination Date

In any case where Notice of termination precedes the effective Early Termination Date, Principal Project Company shall continue performing the Work in accordance with all the standards, requirements and terms of the Contract Documents.

## 17.6.2 Transition Plan

**17.6.2.1** Within 90 days before expiration of the Term, or, if applicable, within three days after Principal Project Company receives or delivers a Notice of termination, Principal Project Company shall meet and confer with City for the purpose of developing a transition plan for the orderly transition of Work, demobilization and transfer of Project management, maintenance, operation, care, custody and control to City. The Parties shall use diligent efforts to complete preparation of the transition plan within 30 days before expiration of the Term or, if applicable,

within 15 days after the date Principal Project Company receives or delivers the Notice of termination.

**17.6.2.2** The transition plan shall include a plan to promptly deliver to City or its designee possession of all the property, data and documents described in <u>Sections 17.6.5.1</u> and <u>17.6.5.2</u>.

**17.6.2.3** The transition plan shall include an estimate of costs and expenses to be incurred by both Parties in connection with implementation of the transition plan. Neither Party shall be liable for the other Party's transition costs and expenses, regardless of the reason for termination.

**17.6.2.4** The transition plan shall:

- be in form and substance acceptable to City and shall include and be consistent with the other provisions and procedures in this <u>Section 17.6</u> (Termination Procedures and Duties);
- (b) reserve the last calendar month for City's operation of the system with technical support and other necessary management support from Principal Project Company similar to the demonstration test performed by Principal Project Company before Substantial Completion; and
- (c) if required by City, provide for Principal Project Company to continue to perform specified Work during the period between the Termination Date and the effective date of the release and discharge, following payment of Termination Compensation.

## 17.6.3 Relinquishment and Possession of Project

**17.6.3.1** On or as soon as possible after the Termination Date as provided in the approved transition plan, Principal Project Company shall relinquish and surrender care, custody and control of the Project, to City, and shall cause all persons and entities claiming under or through Principal Project Company to do likewise (except as may be set forth in any Direct Agreement), in at least the condition required by <u>Section 8.5.1</u> (Handback Condition).

**17.6.3.2** On the later of the Termination Date or the date Principal Project Company relinquishes all care, custody and control as provided in the transition plan, City shall have the exclusive right to, and shall assume responsibility at its expense for, care, custody and control of the Project and the Project Site, subject to any rights to damages against Principal Project Company where the termination is due to a Termination for PPC Default.

**17.6.3.3** If the transition plan developed under <u>Section 17.6.2</u> (Transition Plan) requires Principal Project Company to perform any obligations under this Agreement after the Termination Date, this Agreement will remain in full force and effect only to the extent necessary for Principal Project Company to perform such obligations. On the Termination Date, or such later date provided in the approved transition plan, City grants to Principal Project Company a right to access the Project Site for the limited purpose of carrying out Principal Project Company's obligations contemplated by this <u>Section 17.6</u> (Termination Procedures and Duties), including execution of the transition plan contemplated in <u>Section 17.6.2</u> (Transition Plan). This right of access is subject to rescission by City for Principal Project Company's failure to perform any of its obligations under this Agreement after the Termination Date, and shall automatically expire upon Principal Project Company's fulfillment of such obligations.

## 17.6.4 Continuance or Termination of Key Contracts Before Work Completion

**17.6.4.1** If, as of the Termination Date, Principal Project Company has not completed the Work, in whole or in part, City may elect, by written Notice to Principal Project Company to continue in effect some or all of the Key Contracts or to require their termination. If City elects to continue any Key Contracts, then Principal Project Company shall execute and deliver (or shall cause the relevant Key Contractor(s) to execute and deliver) to City a written assignment, in form and substance acceptable to City, acting reasonably, of all Principal Project Company's (or Key Contractor's) right, title and interest in and to such Key Contracts, and City shall assume in writing Principal Project Company's (or such Key Contractor's) obligations thereunder that arise from and after the Termination Date.

**17.6.4.2** If City elects (or is deemed to elect) to require termination of any Key Contracts, then Principal Project Company shall:

- (a) take such steps as are necessary to terminate the relevant Key Contracts, including providing Notice to each Key Contractor that its Key Contract is being terminated and that each of them is to stop Work on the date and to the extent specified in the Notice of termination and stop and cancel orders for materials, services or facilities, unless otherwise approved by City;
- (b) promptly and orderly demobilize and secure in a safe manner the Project Site in a manner satisfactory to City, remove all debris and waste materials and complete any Hazardous Materials Management Work already in process, except as otherwise approved by City;
- (c) take such other actions as are necessary or appropriate to mitigate further costs;
- (d) subject to City's reasonable prior approval, settle all outstanding liabilities and all claims arising out of, relating to or resulting from such Key Contracts;
- (e) as a condition to City's obligation to make payments to Principal Project Company under this <u>Article 17</u> (Termination) and under the requirements of the transition plan, cause each of the Key Contractors to execute and deliver to City a written assignment, in form and substance acceptable to City, of all of their interest in (i) all Regulatory Approvals, Third Party agreements and permits pertaining to the Project or the Work (excluding Subcontracts), provided that City assumes in writing all of the Key Contractor's obligations under said approvals, agreements and permits that arise after the Termination Date; and (ii) all warranties, to the extent assignable, claims and causes of action held by each of them against Subcontractors and other Third Parties pertaining to the Project or the Work, provided that Principal Project Company shall retain the right to pursue any cause of action against the Subcontractor or other Third Parties for damages incurred by Principal Project Company; and

(f) as a condition to City's obligation to make payments to Principal Project Company under this <u>Article 17</u> (Termination) and under the requirements of the transition plan, carry out such other reasonable directions as City may give for termination of the Work in accordance with the transition plan.

# 17.6.5 Other Close-Out Activities

**17.6.5.1** Within 90 days before expiration of the Term, or within 30 days after any earlier Notice of termination is delivered, Principal Project Company shall deliver to City a true and complete list of all materials, goods, machinery, equipment, hardware, parts, supplies and other tangible property in inventory or storage (whether then held by Principal Project Company or any Person on behalf of or for the account of Principal Project Company) for use in or respecting the Work or the Project, or on order or previously completed but not yet delivered from Suppliers for use in or respecting the Work or the Project. In addition, on or as soon after the Termination Date as is possible or as is provided in the approved transition plan, Principal Project Company shall transfer title through bills of sale or other documents of title, as directed by City, and deliver to City's Authorized Representative, all such materials, goods, machinery, equipment, hardware, parts, supplies and other property, provided City assumes in writing all of Principal Project Company's obligations under any contracts relating to the foregoing that arise after the later of the Termination Date or the effective date of the transfer.

**17.6.5.2** Within 90 days before the expiration of the Term, or within 30 days after any earlier Notice of termination is delivered, Principal Project Company shall provide City with a true and complete list of all the data and documents identified in this <u>Section 17.6.5.2</u>. Subject to <u>Sections 21.4</u> (Intellectual Property) and <u>21.5</u> (Intellectual Property Escrows), Principal Project Company shall execute and deliver to City an executed bill of sale or other written instrument, in form and substance reasonably acceptable to City, assigning and transferring to City the following:

- (a) all completed or partially completed drawings (including plans, elevations, sections, details and diagrams), specifications, designs, Design Documents, Record Documents, plans, surveys, and other documents and information pertaining to the design or construction of the Project or the Utility Adjustments;
- (b) all samples, borings, boring logs, geotechnical data and similar data and information relating to the Project or the Project Site;
- (c) all Books and Records, reports, test reports, studies and other documents of a similar nature relating to the Work, the Project or the Project Site; and
- (d) all Intellectual Property, and IP Materials, documents evidencing licenses of PPC Intellectual Property and Third Party IP to City, other work product and other materials relating to all such Intellectual Property and PPC Intellectual Property.

Such bill of sale or other instrument shall be delivered to City as provided in the approved transition plan (or if not specified in the transition plan, shall be delivered on the Termination Date or as soon thereafter as possible), and shall be accompanied by originals or copies, as appropriate, of all of the materials described therein.

**17.6.5.3** Principal Project Company shall take all action that may be necessary, or that City may direct, for the protection and preservation of the Project, the Work and such materials, goods, machinery, equipment, hardware, parts, supplies, data, documentation and other property.

**17.6.5.4** On or as soon as possible after the Termination Date or as provided in the approved transition plan, Principal Project Company shall execute and deliver to City a written assignment, in form and substance acceptable to City, all of Principal Project Company's interest in any IP Escrows or similar arrangements for the protection of Source Code and Source Code Documentation of others used for or relating to the Project or the Work.

**17.6.5.5** On or as soon as possible after the Termination Date or as provided in the approved transition plan, Principal Project Company shall execute and deliver to City a written assignment, in form and substance acceptable to City, of all Principal Project Company's interest in all warranties, claims and causes of action held by Principal Project Company against Third Parties in connection with the Project or the Work, including claims under casualty and business interruption insurance, except to the extent that City has already received credit for such matters in calculating Termination Compensation amounts, in which case they may be pursued by Principal Project Company for its own account.

**17.6.5.6** Principal Project Company shall otherwise assist City in such manner as City may require before and for a reasonable period following the Termination Date to ensure the orderly transition of management, maintenance, operation, care, custody and control of the Project, to City, and shall, if appropriate and if requested by City, take all steps as may be necessary to enforce the provisions of the Key Contracts pertaining to the surrender of Project management, maintenance, operation, care, custody and control.

**17.6.5.7** For a period of four years following the Termination Date, Principal Project Company shall maintain a secure archive copy of all electronic data transferred to City.

# 17.6.6 Calculation of Compensation

**17.6.6.1** Within 30 days after the Early Termination Date for termination pursuant to <u>Section 17.3</u> (Termination for PPC Default), and within 15 days after the Early Termination Date for termination pursuant to <u>Section 17.1</u> (Termination for Convenience; Condemnation), <u>Section 17.2</u> (Termination for Force Majeure Termination Events or Insurance Unavailability) or <u>Section 17.4</u> (Principal Project Company Rights to Terminate), Principal Project Company shall:

- (a) provide City with a written statement prepared by the Collateral Agent as to (i) the Lenders' Liabilities and (ii) the Account Balances, to the extent such accounts are controlled by the Collateral Agent, with documentation reasonably required by City to support such statement and the Collateral Agent's certification that such amounts are true and correct;
- (b) provide a written statement as to the following amounts, (without duplication of any Account Balances verified by the Collateral Agent under <u>Section 17.6.6.1(a)(ii)</u>), together with documentation reasonably required by City to support such statement and a certification that such amounts are true and correct:

- (i) for termination pursuant to <u>Section 17.1</u> (Termination for Convenience; Condemnation) or <u>Section 17.4</u> (Principal Project Company Rights to Terminate), amounts described in <u>clauses (i)</u> and <u>(ii)</u> of <u>Section</u> <u>17.1.3(a)</u> and <u>Sections 17.1.3(c)</u> through <u>17.1.3(f)</u>;
- (ii) for termination pursuant to <u>Section 17.2</u> (Termination for Force Majeure Termination Events or Insurance Unavailability), amounts described in <u>Sections 17.2.8(a)</u> and <u>17.2.8(c)</u> through <u>17.2.8(f)</u>; or
- (iii) for termination pursuant to <u>Section 17.3</u> (Termination for PPC Default),
   (A) the amounts described in <u>clauses (ii)</u> through <u>(vii)</u> of <u>Section 17.3.2.1(b)</u>, and (B) the Insurance Proceeds (excluding proceeds of personal injury, property damage or other Third Party liability insurance payable to or for the account of a Third Party; and
- (c) for termination pursuant to <u>Section 17.1</u> (Termination for Convenience Condemnation), <u>Section 17.2</u> (Termination for Force Majeure Termination Events or Insurance Unavailability), or <u>Section 17.4</u> (Principal Project Company Rights to Terminate), an estimate of any interest and fees that will accrue (on a daily basis) on the outstanding principal due to Lenders at the rate due (excluding default interest), under the Financing Documents over the period between the Early Termination Date and the anticipated date that the Termination Compensation will be paid by City.

**17.6.6.2** From and after the Termination Date, except as otherwise stated in this <u>Article 17</u> (Termination), Principal Project Company shall cease to have any right to (a) Availability Payments except for those accrued and owing before the Early Termination Date and (b) any other compensation, except as provided under <u>Section 17.8.2</u>.

# 17.7 Payment of Termination Compensation

**17.7.1** For termination under <u>Section 17.1</u> (Termination for Convenience; Condemnation) or <u>Section 17.4</u> (Principal Project Company Rights to Terminate), provided Principal Project Company has timely provided to City the statements and information required under <u>Section 17.6.6.1</u>, City shall, within 120 days of the Early Termination Date, pay to Principal Project Company an amount equal to the:

- (a) Termination Compensation; plus
- (b) interest and fees that accrued on the outstanding principal due to Lenders (excluding default interest) under the Financing Documents over the period between the Early Termination Date and such payment date.

**17.7.2** For termination under <u>Section 17.2</u> (Termination for Force Majeure Termination Events or Insurance Unavailability), provided Principal Project Company has timely provided to City the statements and information required under <u>Section 17.6.6.1</u>, City shall, within 120 days of the Early Termination Date, pay to Principal Project Company an amount equal to the:

(a) Termination Compensation; plus

(b) interest and fees that accrued on the outstanding principal due to Lenders (excluding default interest) under the Financing Documents over the period between the Early Termination Date and such payment date.

**17.7.3** For termination pursuant to <u>Section 17.3</u> (Termination for PPC Default), provided Principal Project Company has timely provided to City the statements and information required under <u>Section 17.6.6.1</u>, City shall, within 120 days of the Early Termination Date, pay to Principal Project Company an amount equal to the Termination Compensation.

**17.7.4** If as of the date that City is required to tender payment of Termination Compensation under <u>Sections 17.7.1</u>, <u>17.7.2</u> or <u>17.7.3</u>, as applicable, the Parties have not agreed upon the amount of Termination Compensation due, then:

- (a) City shall proceed to make payment to Principal Project Company of the undisputed portion of the Termination Compensation;
- (b) Within 30 days after receiving such payment Principal Project Company shall deliver to City written Notice of the additional amount of Termination Compensation that Principal Project Company in good faith determines is still owing (the "**disputed portion**"); and
- (c) City shall pay the disputed portion of the Termination Compensation to Principal Project Company in immediately available funds after the disputed portion is agreed to by the Parties or otherwise determined to be payable pursuant to <u>Error! Reference source not found.</u> (Contract Dispute Procedures), as the case may be.

## 17.8 Effect of Termination

**17.8.1** Except to the extent set out in <u>Section 17.6.3.3</u>, and regardless of City's prior Actual Knowledge or constructive knowledge thereof:

- no Contract or agreement to which Principal Project Company is a party as of the Termination Date shall bind City, unless City elects to assume such Contract or agreement; and
- (b) except in the case of City's express written assumption, no such Contract or agreement shall entitle the contracting party to continue performance of work or services respecting the Project following Principal Project Company's relinquishment to City of Project care, custody and control, or to any claim, legal or equitable, against City.
- **17.8.2** Termination of this Agreement shall not:
  - (a) relieve Principal Project Company or any Surety of its obligation for any claims arising before the Termination Date;
  - (b) excuse either Party from any liability arising out of, related to or resulting from any default as provided in this Agreement that occurred before the Termination Date; and

(c) relieve City of claims of Principal Project Company to payment of Compensation Amounts for adverse cost and revenue impacts accruing before the Termination Date due to Compensable Delay Events or Compensable Relief Events that occurred before the Termination Date.

# 17.9 Liability after Termination; Final Release

Notwithstanding the foregoing, any termination of this Agreement under Sections 17.2 (Termination for Force Majeure Termination Events or Insurance Unavailability), 17.3 (Termination for PPC Default), 17.4.3 (Termination Due to Court Ruling) or 17.5 (Termination if Financial Close Fails to Occur) shall automatically extinguish any claim of Principal Project Company to payment of Compensation Amounts for adverse cost and revenue impacts accruing after the Early Termination Date due to Relief Events that occurred before termination. Except as otherwise expressly provided in this Agreement (other than Section 16.2.10 (Cumulative, Non-Exclusive Remedies)), if this Agreement is earlier terminated for any reason, then City's payment to Principal Project Company of the amounts required under this Agreement (if any) shall constitute full and final satisfaction of, and upon payment City shall be forever released and discharged from, any and all claims, causes of action, suits, demands and Losses, known or unknown, suspected or unsuspected, that Principal Project Company may have against City arising out of, relating to or resulting from this Agreement or termination thereof, the other Contract Documents, or the Project, excluding any proceedings that are pending as of 30 days after the Termination Date, remain unresolved at the time of such payment, and are not related to termination or Termination Compensation. Upon such payment, Principal Project Company shall execute and deliver to City all such releases and discharges, including a release pursuant to Civil Code Section 1542, as City may reasonably require to confirm the foregoing, but no such written release and discharge shall be necessary to give effect to the foregoing satisfaction and release.

# 17.10 Exclusive Termination Rights

This <u>Article 17</u> (Termination) contain the entire and exclusive provisions and rights of City and Principal Project Company regarding termination of this Agreement, and each Party waives, to the maximum extent permitted by Law, any and all other rights to terminate at law.

## 17.11 Access to Information

Principal Project Company shall conduct all discussions and negotiations to determine any Termination Compensation, and shall share with City all data, documents and information pertaining to the termination, on an Open Book Basis.

## ARTICLE 18. PARTNERING; CONTRACT DISPUTE PROCEDURES

### 18.1 Introduction

**18.1.1** The Parties acknowledge fostering a collaborative working relationship during the entire Term is critical to early identification and resolution of issues, claims, and Contract Disputes. This <u>Article 18</u> provides a structured approach for the Parties to address issues, claims, and Contract Disputes within the framework of informal project-level partnering and formal Contract Dispute procedures.

**18.1.2** The procedures in this <u>Article 18</u> (Partnering; Contract Dispute Procedures) are necessary for the City to address potential and actual Contract Disputes. Prior knowledge of potential Contract Disputes before Principal Project Company starts disputed Work, and proper documentation from Principal Project Company during that Work, are critical for the City to make informed decisions affecting the Project's budget and schedule.

#### 18.2 Partnering

#### 18.2.1 Collaborative Partnering

The Parties will engage in a partnering process to foster effective communication, enhance cooperation, and proactively address issues and claims to prevent them from escalating into Contract Disputes. The requirements for the partnering process are set forth in <u>Part 3</u> of <u>Section 01 31 33</u> (Partnering Procedures) of <u>Division 10</u> of the Technical Requirements.

#### 18.2.2 Confidentiality

In accordance with California Evidence Code sections 1119 and 1152, and Federal Rule of Evidence 408, and similar prohibitions, statements made during the partnering process, including the workshops and informal negotiations within the framework of the Issue Resolution Ladder, are confidential and not admissible or discoverable in any proceeding to resolve a Contract Dispute.

#### **18.3 Contract Dispute Procedures**

## 18.3.1 Disputes Generally

#### 18.3.1.1 Outline of Dispute Process

The following is an outline of the sequential process for submitting and resolving Contract Disputes, with each step serving as a condition precedent for Principal Project Company to proceed to subsequent steps, beginning with the Notice of Potential Dispute. These steps must be completed prior to submission by Principal Project Company of any Contract Dispute to litigation in San Francisco Superior Court.

(a) Principal Project Company must submit a timely Notice of Contract Dispute in compliance with <u>Section 18.3.2</u> before Principal Project Company can file a Contract Dispute.

- (b) Principal Project Company must submit a timely Contract Dispute in compliance with <u>Section 18.3.3</u> before Principal Project Company can participate in the Issue Resolution Ladder.
- (c) Principal Project Company must timely comply with the Issue Resolution Ladder provisions in compliance with <u>Section 18.3.3.5</u> before Principal Project Company can participate in mediation.
- (d) Principal Project Company must timely comply with the mediation provisions in <u>Section 18.3.3.6</u> in order to exhaust its administrative remedies under this Agreement.
- (e) Principal Project Company must complete all steps outlined in this <u>Section</u> <u>18.3.1.1</u> above prior to commencing litigation against the City for disputes arising out of or related to this Agreement that are not expressly excluded from the Contract Dispute process by <u>Section 18.3.1.3</u>. The Principal Project Company must also submit a timely Government Code Claim in order to commence litigation against the City for disputes arising out of or related to this Agreement pursuant to <u>Section 18.3.3.7</u>.

**18.3.1.2** Disputed issues not timely raised and properly documented by Principal Project Company in conformance with this <u>Section 18.3</u> will be deemed waived by Principal Project Company and may not be asserted in any subsequent litigation, or legal action. Furthermore, by executing this Agreement, Principal Project Company waives any and all claims or defenses of waiver, estoppel, or any other type of excuse of non-compliance with the Contract Dispute submission requirements.

**18.3.1.3** The Contract Dispute procedures specified in this <u>Section 18.3</u> (Contract Dispute Procedures) do not apply to the following:

- claims respecting penalties for forfeitures prescribed by statute or regulation that a government agency is specifically authorized to administer, settle, or determine;
- (b) claims respecting personal injury, death, reimbursement, or other compensation arising out of or resulting from personal injury of death;
- (c) claims by the City;
- (d) stop notices.
- (e) any matters that this Contract expressly states are final, binding, or not subject to Dispute resolution (including any Person's exercise of sole discretion);
- (f) disputes regarding compliance with applicable Law, the rights of City to terminate this Contract, or indemnification;
- (g) any claim for injunctive relief;
- (h) claims arising solely in tort;

- (i) any claim against an insurance company, including any Contractor Dispute that is covered by insurance;
- (j) Disputes regarding matters under the jurisdiction of OSHA;
- (k) issues regarding DBE participation;
- (I) any claim for, or Dispute based on, remedies expressly created by statute;
- (m) any Dispute that is actionable only against a Surety;
- (n) any claim arising out of or relating to the Work where a third party is a necessary or appropriate party (excluding any PPC-Related Entity); and
- (o) any claim or dispute that does not arise under the Contract.

## 18.3.2 Notice of Contract Dispute

#### 18.3.2.1 Timely Submission

Principal Project Company must submit a Notice of Contract Dispute to the City within seven-Days of the event, activity, occurrence, or other cause that gives rise to the Contract Dispute. If the Contract Dispute pertains to a Change or the rejection of a Relief Event by the City, the seven-Day period starts when the City's Authorized Representative issues a final written decision denying, in whole or in part, the corresponding PPC Change Request or Relevant Event claim.

#### 18.3.2.2 Content Requirements

Notices of Contract Dispute must include:

- (a) description of the Contract Dispute's nature and circumstances;
- (b) reasons Principal Project Company believes additional compensation or time may be due, including Contract Document references and citations supporting its position;
- (c) good faith estimates of the potential cost and/or time impacts; and
- (d) for Contract Disputes related to a disputed Relevant Event, identification of items in the Implementation Proposal that form the basis for Principal Project Company's performance the affected Work.

## 18.3.3 Contract Dispute

#### 18.3.3.1 Timely Submission

(a) Principal Project Company must submit a Contract Dispute to the City no later than 30 Days after the City's receipt of the corresponding Notice of Contract Dispute.

#### 18.3.3.2 Content Requirements of all Contract Disputes

Contract Disputes must include:

- (a) cover letter;
- (b) certification, signed by Principal Project Company's Authorized Representative, stating, under penalty of perjury, that:
  - (i) the Contract Dispute is made in good faith;
  - (ii) supporting data are accurate and complete, to the best of their knowledge and belief; and
  - (iii) the amount requested accurately reflects adjustment in compensation or time for which Principal Project Company believes the City is liable;
- (c) narrative summary of Contract Dispute's merits and amount, and the specific provisions of Contract Documents under which the Contract Dispute is made;
- (d) list and copy of documents relating to Contract Dispute:
  - (i) Contract Documents;
  - (ii) correspondence; and
  - (iii) Agreement Exhibits;
- (e) detailed chronology of events and correspondence;
- (f) analysis of Contract Dispute merit; and
- (g) analysis of Contract Dispute cost (money and time).

# 18.3.3.3 Additional Requirements for Contract Disputes Relating to Delay Events Impacting the Critical Path

- (a) In addition to the contents required under <u>Section 18.3.3.3</u> (Content Requirements of all Contract Disputes), Contract Disputes related to Delay Events prior to Substantial Completion and seeking time extensions or challenging the assessment of delay or liquidated damages shall include a written analysis of all changes and all delays impacting the Critical Path, including a TIA.
- (b) The TIA shall be updated to include the most recent information about the Project Schedule and the TIA shall include all activities represented or affected by the change, with activity numbers, durations, predecessor and successor activities, resources and cost, and with a narrative report, in form satisfactory to City, which compares the proposed new schedule to the Project Schedule, as appropriate. Principal Project Company shall reschedule

activities not otherwise affected by the event, in order to take advantage of additional Float available as the result of the time extension. Any such rescheduling shall be reflected in the Project Schedule.

(c) The TIA shall account for all Delay Events in the relevant time frame with actual logic ties.

#### 18.3.3.4 **Procedure for City Review and Determination**

- (a) The City will review only a timely, certified, and properly documented Contract Dispute.
- (b) The City will respond to a Contract Dispute in writing, within 45 days of receipt of such Contract Dispute. In its response, City will either grant or deny the Contract Dispute in whole or in part. If City does not respond to a Contract Dispute within the 45-day period, the Contract Dispute is deemed denied in its entirety.

#### 18.3.3.5 Issue Resolution Ladder

After the City issues its written decision under <u>Section 18.3.3.4</u> or the Contract Dispute is deemed denied in its entirety, the Parties shall utilize the Issue Resolution Ladder to negotiate and attempt to resolve the Contract Dispute in a timely and efficient manner before proceeding to mediation in accordance with <u>Section 18.3.3.6</u>. The requirements for the Issue Resolution Ladder are set forth in <u>Part 3</u> of <u>Section 01 31 33</u> (Partnering Procedures) of <u>Division 10</u> of the Technical Requirements.

## 18.3.3.6 Non-binding Mediation

If the Parties are unable to resolve a Contract Dispute during the Issue Resolution Ladder process, either Party may submit the dispute to mediation, by written notice to the other, that a mutually acceptable third-party mediator shall be selected for the purpose of facilitating a negotiated resolution of the Contract Dispute. The Contract Dispute shall be submitted to mediation within 10 Days of the conclusion of the Issue Resolution Ladder process. The Parties will share the costs of the mediation equally. If the mediation is unsuccessful, the Principal Project Company shall submit a Government Claims Act claim pursuant to <u>Section 18.3.3.7</u>, Government Code Claim Requirement.

#### 18.3.3.7 Government Code Claim

The timely submittal of a complete and proper PPC Change Request or Relevant Event claim and compliance with the procedures specified in this <u>Article 18</u> (Partnering; Contract Dispute Procedures) shall operate to toll Principal Project Company's compliance with the Government Code dispute requirements under California Government Code section 900, et seq., and Administrative Code Chapter 10 until the Parties complete mediation. The Principal Project Company must comply with California Government Code section 900, et seq. and Administrative Code Chapter 10 prior to submitting a Contract Dispute to litigation in San Francisco Superior Court.

# **ARTICLE 19. REPRESENTATIONS AND WARRANTIES**

## 19.1 Principal Project Company Representations and Warranties

Principal Project Company represents and warrants to City, as of the Effective Date, as follows:

The Financial Model (a) was prepared by or on behalf of Principal Project Company in 19.1.1 good faith, (b) uses financial formulas that, as of the Effective Date are mathematically and formulaically correct and suitable for making reasonable projections, (c) was audited and verified by an independent recognized model auditor immediately before the Effective Date, (d) fully discloses all cost, revenue and other financial assumptions and projections that Principal Project Company has used or is using in making its decision to enter into this Agreement and in making disclosures to potential equity investors and Lenders under the Initial Financing Agreements and (e) as of the Effective Date represents the projections that Principal Project Company believes in good faith are the most realistic and reasonable for the Project; subject to the understanding that such projections are based upon a number of estimates and assumptions and are subject to significant business, economic and competitive uncertainties and contingencies, and that Principal Project Company's stated belief regarding the projections does not constitute a representation that any of the assumptions are correct, that such projections will be achieved or that the forward-looking statements expressed in such projections will correspond to actual results.

**19.1.2** Principal Project Company has evaluated the feasibility of performing the Work by the Contract Deadlines and for the Milestone Payment and Availability Payments, and has reasonable grounds for believing and does believe, subject to the express terms of this Agreement, that such performance is feasible and practicable.

**19.1.3** Principal Project Company has reviewed all applicable Laws relating to Taxes, and has taken into account all requirements imposed by such Laws in preparing the Financial Model.

**19.1.4** Based upon the work undertaken by PPC-Related Entities pursuant to the Predevelopment Agreement, review of information provided by City, and other Reasonable Investigations undertaken by PPC-Related Entities, Principal Project Company has evaluated the constraints affecting the development, design, and construction of the Project, including the Project Site, surface and subsurface conditions discoverable through such investigation, the terms of the CEQA Approval, the terms of the NEPA Approval and requirements of applicable Laws, and, based on the foregoing, Principal Project Company has reasonable grounds for believing and does believe that the Project can be developed, designed, and constructed.

**19.1.5** Principal Project Company conducted a Reasonable Investigation of property to which it had access and other available information regarding conditions at the Project Site, and as a result of such investigation, Principal Project Company is familiar with and accepts the physical requirements of the Work, subject to Principal Project Company's rights regarding Relief Events. Principal Project Company shall undertake and complete the Work and the Project at its sole cost and without any additional compensation, any extension of time, excuse from compliance or other relief on account of such compliance, regardless of whether such compliance would require additional time for performance or additional design, construction operations, maintenance, financing, labor, equipment, supplies and/or materials not expressly

provided for in the Contract Documents or would have an adverse effect on costs, subject only to Principal Project Company's rights regarding Delay Events and Relief Events.

**19.1.6** Principal Project Company familiarized itself with the requirements of all applicable Laws and the conditions of any required Regulatory Approvals.

**19.1.7** Principal Project Company has no reason to believe that any Regulatory Approval required to be obtained by Principal Project Company will not be granted in due course and thereafter remain in effect so as to enable the Work to proceed in accordance with the Contract Documents.

**19.1.8** All Work furnished by Principal Project Company will be performed by or under the supervision of Persons who hold all necessary licenses, certifications, registrations, permits or approvals to practice in the State, by personnel who are experienced, competent and skilled in their respective trades or professions, who are professionally qualified to perform the Work in accordance with the Contract Documents and who shall assume professional responsibility for the accuracy and completeness of the Design Documents, Construction Documents and other documents prepared or checked by them.

**19.1.9** Principal Project Company is a [\_\_\_\_\_], duly organized and validly existing under the laws of the [\_\_\_\_\_], has the requisite power and all required licenses to carry on its present and proposed activities, and has full power, right and authority to execute, and deliver this Agreement and to perform each and all of the obligations of Principal Project Company provided for under this Agreement. Principal Project Company is duly qualified to do business and is in good standing in the State as of the Effective Date, and will remain duly qualified and in good standing throughout the Term and for as long as any obligations remain outstanding under the Contract Documents.

**19.1.10** The execution, delivery and performance of this Agreement has been duly authorized by all necessary action of Principal Project Company's governing body, each person executing this Agreement has been duly authorized to execute and deliver each such document on behalf of Principal Project Company and this Agreement has been duly executed and delivered by Principal Project Company.

**19.1.11** No default under, violation of, or conflict with the governing instruments of Principal Project Company or any agreement, judgment or decree to which Principal Project Company is a party or is bound will result from (a) the execution and delivery by Principal Project Company of this Agreement, or (b) performance by Principal Project Company of its obligations under this Agreement.

**19.1.12** The execution and delivery by Principal Project Company of this Agreement, and the performance by Principal Project Company of its obligations under this Agreement will not conflict with any Laws applicable to Principal Project Company that are valid and in effect on the date of execution and delivery. As of the Effective Date, Principal Project Company is not in breach of any applicable Law that would have a material adverse effect on the Work or the performance of any of its obligations under the Contract Documents.

**19.1.13** This Agreement constitutes the legal, valid and binding obligation of Principal Project Company, enforceable against Principal Project Company in accordance with its terms, subject

only to applicable bankruptcy, insolvency and similar laws affecting the enforceability of the rights of creditors generally and the general principles of equity.

**19.1.14** No proceeds of or commitments to provide the Committed Investment, as applicable, is or shall be from a Prohibited Person.

**19.1.15** There is no action, suit, proceeding, investigation or litigation pending and served on Principal Project Company which challenges Principal Project Company's authority to execute, deliver or perform, or the validity or enforceability of, this Agreement or which challenges the authority of the representative of Principal Project Company executing this Agreement; and Principal Project Company has disclosed to City any pending and un-served or threatened action, suit, proceeding, investigation or litigation with respect to such matters of which Principal Project Company is aware. No current, pending or outstanding criminal, civil, or enforcement actions have been initiated against Principal Project Company by City or the State, and Principal Project Company agrees that it will immediately provide Notice to City if any such action is initiated during the Term.

**19.1.16** Principal Project Company has disclosed to City in writing all organizational conflicts of interest of Principal Project Company and its Contractors of which Principal Project Company was actually aware, which have not been approved in writing by City.

**19.1.17** As of the effective date of the relevant Key Contract. (a) each Key Contractor is duly organized, validly existing and in good standing under the laws of the state of its organization and is duly qualified to do business, and is in good standing, in the State, (b) the ownership interests (including options, warrants and other rights to acquire ownership interests) of each Key Contractor that is a single purpose entity formed for the Project are held by those Persons identified in a written certification delivered by Principal Project Company to City before such effective date; (c) each Key Contractor has the power and authority to do all acts and things and execute and deliver all other documents as are required to be done, observed or performed by it in connection with its engagement by Principal Project Company; (d) each Key Contractor has (i) obtained and will maintain all required registrations, licenses, certifications, permits and approvals required under applicable Law as of such date and (ii) expertise, gualifications, experience, competence, and skills and is gualified to perform the Work for which it is responsible; (e) each Key Contractor will be required by the applicable Key Contract to comply with all health, safety and Environmental Laws in the performance of any work activities for, or on behalf of, Principal Project Company for the benefit of City; and (f) no Key Contractor is in breach of any applicable Law that would have a material adverse effect on any aspect of the Work.

**19.1.18** Principal Project Company has not employed or retained, and Principal Project Company shall not employ or retain, any Person other than employees, agents, attorneys, consultants and advisors of a PPC-Related Entity, to solicit or secure this Agreement, and that it has not paid or agreed to pay any Person any fee or any other consideration contingent on the making of this Agreement.

**19.1.19** Principal Project Company warrants that it owns, or will own, and has, or will have, good and marketable title and sufficient rights to all materials, Intellectual Property, equipment, tools and supplies furnished, or to be furnished, by any PPC-Related Entity that become part of the Project or are purchased for City for the development, operation, maintenance or repair of

the Project, free and clear of all liens, royalties, fees or other charges of any kind or nature. All such materials, Intellectual Property, equipment, devices, or processes shall be delivered free of any claim of any Third Party for infringement of any Intellectual Property rights or ownership. Refer to <u>Section 2.4.3</u> (Passage of Title) for provisions regarding transfer of title to City.

**19.1.20** Principal Project Company warrants that the individual signing this Agreement on behalf of Principal Project Company is the properly authorized representative, agent, member or officer of Principal Project Company, that he/she has not, nor has any other member, employee, representative, agent or officer of Principal Project Company, directly or indirectly, to the best of the undersigned's knowledge, entered into or offered to enter into any combination, collusion or agreement to receive or pay, and that he/she has not received or paid, any sum of money or other consideration for the execution of this Agreement other than that which appears upon the face of this Agreement.

**19.1.21** [\_\_\_\_\_] is a [\_\_\_\_\_], duly organized and validly existing under the laws of [\_\_\_\_\_], has the requisite power and all required licenses to carry on its present and proposed activities, and has full power, right and authority to perform each and all of the obligations of an Equity Member provided for under this Agreement. [Note to PNC: Include for each Equity Member of PPC.]

## **19.2** City Representations and Warranties

City represents and warrants to Principal Project Company as follows:

**19.2.1** The City and County of San Francisco is a charter city and municipal corporation duly organized and validly existing under the Constitution of the State of California. As of the Effective Date, City has full power, right and authority to execute, deliver this Agreement and City has full power, right and authority to perform its obligations under this Agreement.

**19.2.2** Each person executing this Agreement on behalf of City is duly authorized to execute and deliver this Agreement, and this Agreement has been duly executed and delivered by City.

**19.2.3** This Agreement has each been duly authorized by City, and constitutes a legal, valid and binding obligation of City enforceable against City in accordance with its terms.

**19.2.4** As of the Effective Date, there is no action, suit, proceeding, investigation or litigation pending and served on City which challenges City 's authority to execute, deliver and perform this Agreement, or which challenges the validity or enforceability of, this Agreement or which challenges the authority of City officials executing this Agreement; and City has disclosed any pending action, suit, proceeding, investigation or litigation against City (including filed but unserved complaints of which City has Actual Knowledge) relating to this Agreement or the Project.

**19.2.5** Neither the execution and delivery by City of this Agreement, nor the consummation of the transactions contemplated under this Agreement, is in conflict with or has resulted or will result in a default under or a violation of any agreement, judgment or decree to which City is a party or is bound.

**19.2.6** The execution and delivery by City of this Agreement, and the performance by City of its obligations under this Agreement, will not conflict with any Laws applicable to City that are

Infrastructure Facility DBFOM Agreement

valid and in effect on the date of execution and delivery. City is not in breach of any applicable Law that would have a material adverse effect on the performance of any of its obligations under the Contract Documents.

**19.2.7** City's execution and delivery of this Agreement is not subject to any requirement to obtain consent or approval of any other Person (including Governmental Entities), other than consents and approvals already obtained.

#### **19.3** Special Remedies for Mutual Breach of Warranty

Notwithstanding any other provision of this Agreement, if any circumstance or event occurs that constitutes or results in a concurrent breach by both Principal Project Company and City of similar warranties referenced in Section 16.1.1(f) or 16.3.1(c) but does not also constitute or result in any other breach or default by either Party, then the only remedies shall be for the Parties to take action to rectify or mitigate the effects of such circumstance or event, to pursue severance and reformation of the Contract Documents in accordance with Section 23.11 (Severability), or Termination Due to Court Ruling in accordance with Section 17.4.3 (Termination Due to Court Ruling).

# ARTICLE 20. ASSIGNMENT AND TRANSFER

### 20.1 Restrictions on Equity Transfers and Change of Control

- **20.1.1** Except as provided in <u>Section 20.1.3</u>, any:
  - (a) Change of Control of Principal Project Company; or
  - (b) Equity Transfer that results in any Equity Member ceasing to own (directly or indirectly) the same percentage of the issued share capital, partnership or membership interests, as applicable, in Principal Project Company that it owned (directly or indirectly) as of the Financial Close Date,

shall be subject to City's prior written approval in accordance with <u>Section 20.2</u> (Standards and Procedures for City Approval).

**20.1.2** Neither an Equity Transfer to a Prohibited Person, nor a Change of Control that would involve the provision of any amount of Committed Investment directly or indirectly from a Prohibited Person, is permitted at any time. Further, none of the events described in <u>clauses (b)</u> through (<u>a</u>) of the definition of Change of Control are permitted at any time if it would result in the direct or indirect ownership by a Prohibited Person of any interest in Principal Project Company.

**20.1.3** Transfers and transactions within any of the exceptions described in <u>clauses (a)</u> through (<u>g)</u> of the definition of Change of Control are allowed at any time without necessity for City's approval, provided that:

- (a) for an exception described in <u>clause (a)</u> (with respect to any initial public offering), or <u>clause (b)</u>, (c) or (d), Principal Project Company shall deliver to City, on or before 10 Business Days before the effectiveness of the transfer or transaction, written Notice describing the transfer or transaction and (if applicable) the names of the transferor and transferee, together with documentation demonstrating that the transfer or transaction is within such an exception; and
- (b) for an exception described in <u>clause (a)</u> (other than with respect to an initial public offering), Principal Project Company shall deliver to City, within five days after the effectiveness of the transfer or transaction, written Notice describing the transfer or transaction and (if applicable) the names of the transferor and transferee, together with documentation demonstrating that the transfer or transaction is within such an exception.

### 20.2 Standards and Procedures for City Approval

**20.2.1** Where City's prior written approval is required for a proposed sale, assignment, conveyance, transfer, pledge, mortgage, encumbrance, grant of right of entry, or grant of other special use, management or control, or for any proposed Equity Transfer or Change of Control (each, a "**Transaction**"), and such Transaction is proposed at any time during the period ending two years after the Substantial Completion Date, City may withhold or condition its approval in

its sole discretion. Any such decision of City to withhold consent shall be final, binding and not subject to the Contract Dispute Procedures.

**20.2.2** After the second anniversary of the Substantial Completion Date, City shall not unreasonably withhold its approval of a Transaction.

**20.2.3** Among other factors and considerations, it shall be reasonable for City to withhold its approval if:

- (a) Principal Project Company fails to demonstrate to City's reasonable satisfaction that: (i) the proposed Transaction that would amount to a Change of Control of Principal Project Company will not have any adverse effect on Principal Project Company's ability to timely perform its obligations under the Contract Documents and the Principal Project Documents, taking into account the financial resources, qualifications and experience of the proposed assignee, grantee or transferee; and (ii) the proposed assignee, grantee or transferee is in compliance with City's rules, regulations, and adopted written policies regarding organizational conflicts of interest; or
- (b) at the time of the proposed Transaction, there exists any uncured PPC Default or any event or circumstance that with the lapse of time, the giving of Notice or both would constitute a PPC Default, unless City receives from the proposed transferee assurances of cure and performance acceptable to City in its reasonable discretion.

**20.2.4** For Transactions subject to City's prior reasonable approval, City will approve or disapprove in writing within 60 days after it receives from Principal Project Company:

- (a) a request for approval;
- (b) a reasonably detailed description of the proposed Transaction;
- (c) such information, evidence, and supporting documentation as City may request concerning the identity, financial resources, qualifications, experience, and potential conflicts of interest of the proposed transferee and its proposed contractors; and
- (d) such evidence of organization and authority, and such incumbency certificates, certificates regarding debarment or suspension, and other certificates, representations, and warranties as City may reasonably request.

**20.2.5** For Transactions subject to City's prior reasonable approval, City will evaluate the identity, financial resources, qualifications, experience, and potential conflicts of interest using the same standards and criteria that it is then currently applying, or if there is no current application, then the same standards and criteria it most recently applied, to the evaluation of Persons responding to City's requests for qualifications for similar agreements for comparable projects and facilities.

**20.2.6** If for any reason City does not act within such 60 day period, or any extension thereof by mutual agreement of the Parties, then the proposed Transaction shall not be permitted,

subject to Principal Project Company's right, in the case of a proposed Transaction governed by <u>Section 20.2.2</u>, to submit a Contract Dispute for resolution according to the Contract Dispute Procedures.

#### 20.3 Restrictions on Assignment, Subletting and Other Transfers of PPC's Interest or the Project

**20.3.1** Principal Project Company shall not voluntarily or involuntarily sell, assign, convey, transfer, pledge, mortgage or otherwise encumber PPC's Interest or any portion thereof without City's prior written approval, in its sole discretion, except:

- (a) to Lenders for security as permitted by this Agreement, provided Principal Project Company retains responsibility for the performance of Principal Project Company's obligations under the Contract Documents; or
- (b) to any Substituted Entity approved by City in accordance with the Direct Agreement, provided that such Substituted Entity assumes in writing full responsibility for performance of the obligations of Principal Project Company under this Agreement, the other Contract Documents, and the Key Contracts and Financing Documents arising from and after the date of assignment.

**20.3.2** Subject to <u>Section 20.3.1</u>, assignments and transfers of the PPC's Interest permitted under this <u>Section 20.3</u> (Restrictions on Assignment, Subletting and Other Transfers of PPC's Interest or the Project), or otherwise approved in writing by City, shall be effective only upon City's receipt of a written instrument executed by the transferee of PPC's Interest, including any Person who acquires PPC's Interest through foreclosure, transfer in lieu of foreclosure or similar proceeding, in form and substance acceptable to City, in which the transferee, without condition or reservation, (i) assumes all of PPC's Interest then in effect, (ii) agrees to perform and observe all of PPC's Interest, and (iii) agrees to be bound by, the Project Management Plan, the Key Contracts, the Regulatory Approvals, all agreements between the transferor and Third Parties related to or arising out of the Work and/or the Project, and all agreements between the transferor and Governmental Entities with jurisdiction over the Project or the Work, except to the extent otherwise approved by City, in its sole discretion.

**20.3.3** Principal Project Company shall not grant any right of entry onto, use of, or right to manage and control the Project to any other Person except as expressly contemplated in this Agreement without City's prior written approval, in its sole discretion. Any purported voluntary or involuntary sale, assignment, subletting, conveyance, transfer, pledge, mortgage, encumbrance or grant of other special use, management or control of the Project in violation of this provision shall be null and void ab initio and City, at its option, may declare any such attempted action to be a material PPC Default.

# 20.4 Assignment by City

City may assign all or any portion of its right, title and interest in the Contract Documents, Payment Bonds and Performance Bonds, guarantees, letters of credit and other security for payment or performance:

(a) in its sole discretion and without Principal Project Company's consent, to any other Person that succeeds to the powers and authority of SFMTA under the

San Francisco City Charter and San Francisco Administrative Code, has the legal authority to perform its obligations and has sources of funds to perform the payment obligations of the City that are at least as adequate and secure as the City's as of the time of the assignment; and

(b) to others with the prior written consent of Principal Project Company, which consent cannot be unreasonably withheld if City's assignee has a credit rating equal to or better than City's senior lien rating at the time of the assignment as measured by a Rating Agency.

### 20.5 Change of Organization or Name

**20.5.1** Principal Project Company shall not change the legal form of its organization in a manner that adversely affects City's rights, protections and remedies under the Contract Documents without the prior written approval of City.

**20.5.2** If either Party changes its name, such Party agrees to promptly furnish the other Party with Notice of change of name and appropriate supporting documentation.

# ARTICLE 21. RECORDS AND AUDITS; INTELLECTUAL PROPERTY

### 21.1 Maintenance and Inspection of Records

**21.1.1** Principal Project Company shall undertake the following with respect to its Books and Records:

- (a) keep and maintain Books and Records, including copies of all original documents delivered to City, in San Francisco, California, or in another location approved by City in writing, and provide Notice to City where the Books and Records are kept;
- (b) keep and maintain Books and Records in accordance with applicable provisions of the Contract Documents, including the Technical Requirements, applicable provisions of the Project Management Plan, and in accordance with Good Industry Practice;
- (c) make all Books and Records available for inspection by City and its representatives in Principal Project Company's principal offices in San Francisco, California, or in accordance with each IP Escrow, at all times during normal business hours, or at other reasonable times during the Term;
- (d) provide to City, or make available to City for review in accordance with each IP Escrow, copies of any Books and Records as and when reasonably requested by City. City may inspect upon 48 hours' prior Notice or without prior Notice where there is good faith suspicion of fraud. City's right of inspection includes the right to make extracts and take notes and shall not be construed as a waiver by Principal Project Company of the attorney-client privilege;
- (e) retain all Books and Records related to the D&C Work until five years after the Substantial Completion Date and retain all Books and Records related to the IFM Services until five years after the date of final payment under the Contract Documents, provided that all records which are being audited or which relate to Claims and Contract Disputes being processed or actions brought under the Contract Dispute Procedures shall be retained and made available until any later date that such audits, Claims, Contract Disputes and actions are finally resolved; and
- (f) permit City, upon 10 days' prior Notice to Principal Project Company (which Notice shall identify the persons City requests to be present for an interview and describe with reasonable specificity the subject matter to be raised in the interview), to discuss the obligations of Principal Project Company under this Agreement with any of the directors, chief executive officer and chief financial officer of Principal Project Company or its representatives, for the purpose of enabling City to determine whether Principal Project Company is in compliance with this Agreement and applicable Law. The interviewees and their employers may have counsel present at the interviews.

**21.1.2** Principal Project Company shall cause each Key Contract to include the provisions of <u>Section 21.1.1</u>, to the extent applicable, modified as appropriate to apply to the Contractor's Books and Records.

**21.1.3** <u>Exhibit 16</u> (Federal, State and City Requirements) includes additional requirements regarding maintenance and inspection of Books and Records.

### 21.2 Audits

**21.2.1** City may review and audit Principal Project Company, its Contractors and their respective Books and Records as and when City deems necessary for purposes of verifying compliance with the Contract Documents and applicable Law and verifying Claims.

- 21.2.2 Without limiting <u>Section 21.2.1</u>:
  - (a) City may audit the Project Management Plan and compliance therewith, including the right to inspect Work and/or activities and to verify the accuracy and adequacy of the Project Management Plan and its component parts, plans and other documentation;
  - (b) the audits may be performed by employees or consultants of City, City or the City's Controller, or by an auditor under contract with City, City or the City's Controller;
  - (c) Principal Project Company shall provide adequate and appropriate work space for City or its representative(s) to conduct audits;
  - (d) Principal Project Company shall: allow auditor(s) access to such Books and Records during normal business hours; provide to City copies thereof, in any physical and/or digital medium, as and when reasonably requested by City; allow interviews of any employee who might have information related to such Books and Records; and otherwise cooperate with the auditors including furnishing a management representation letter upon request of the auditor; and
  - (e) Principal Project Company shall cause each Contract to include a similar right of City to audit records and interview staff of the Contractor, and a similar covenant to cooperate with the auditors.

The foregoing shall not be deemed to waive the right of Principal Project Company or Contractor to have counsel or other appropriate representatives present at the interview.

**21.2.3** If any City audit results in a material correction to the Books and Records, as determined by City in its reasonable discretion, Principal Project Company shall pay the reasonable costs of City in conducting the audit, but if not, City will bear the costs of the audit.

**21.2.4** Failure of Principal Project Company, Contractors or their agents to maintain and retain sufficient Books and Records to allow the auditors to verify all or a portion of a Claim or to permit the auditor access to its Books and Records to verify a Claim shall be sufficient basis for City to deny recovery by Principal Project Company of the Claim to the extent of such failure.

**21.2.5** Full compliance by Principal Project Company with the provisions of <u>Section 21.2</u> (Audits) is a contractual condition precedent to Principal Project Company's right to seek relief on a Claim.

**21.2.6** City's rights of audit include the right to observe the business operations of Principal Project Company and its Contractors to confirm the accuracy of Books and Records.

**21.2.7** Principal Project Company shall include in the Project Management Plan internal procedures to facilitate review and audit by City and, if applicable, City representatives and the City's Controller or their employees and consultants.

**21.2.8** Principal Project Company represents and warrants the completeness and accuracy of all information it or its agents provides in connection with City audits, and shall cause all Contractors to represent and warrant the completeness and accuracy of all information such Contractors provide in connection with City audits.

**21.2.9** Principal Project Company's internal and Third Party quality and compliance auditing responsibilities shall be identified in the Project Management Plan, in accordance with <u>Section</u> <u>1.4</u> of <u>Division 1</u> of the Technical Requirements and other related provisions concerning QA and compliance auditing.

**21.2.10** Nothing in the Contract Documents shall in any way limit the constitutional and statutory powers, duties and rights of elected officials, including the independent rights of the City's Controller, in carrying out his or her legal authority.

# 21.3 Public Records Act and San Francisco Sunshine Ordinance

**21.3.1** Principal Project Company acknowledges and agrees that all Submittals, records, documents, drawings, plans, specifications and other materials in City's possession, including any Books and Records submitted by Principal Project Company to City, may be considered public information subject to disclosure under the California Public Records Act (the "**PRA**") and San Francisco Sunshine Ordinance.

**21.3.2** If Principal Project Company believes any Books and Records submitted to City constitute trade secrets, proprietary information or other information that is not subject to or excepted from disclosure under the PRA, Principal Project Company shall be solely responsible for specifically and conspicuously designating that information by placing "CONFIDENTIAL" in the center header of each such page affected, as it determines to be appropriate. Any such designation of trade secret or other basis for exemption shall be accompanied by a concise statement of reasons supporting the claim including the specific Law that authorizes the exemption from disclosure under the PRA.

**21.3.3** If City receives a request for public disclosure of information or materials that have been designated by Principal Project Company as "CONFIDENTIAL," City will use reasonable efforts to provide Notice to Principal Project Company of the request and may request advice from City's counsel before disclosing any such documents in accordance with applicable Law. Principal Project Company shall then have the opportunity to either consent to the disclosure or assert its basis for non-disclosure and claimed exception under the PRA or other applicable Law to City within the time period specified in the Notice issued by City (if any) and before the deadlines for release in the PRA and other applicable Law.

Principal Project Company to monitor requests for disclosure and proceedings and make timely filings. City may make filings of its own concerning possible disclosure; however, City is under no obligation to support Principal Project Company's positions. By entering into this Agreement, Principal Project Company consents to, and expressly waives any right to contest, provision by City to City's counsel of all, or representative samples of, information or materials designated as "CONFIDENTIAL" by Principal Project Company, in accordance with the PRA. City shall have no responsibility or obligation for a failure of Principal Project Company to respond or to respond timely to any request for disclosure of information or materials designated as "CONFIDENTIAL" by Principal a disclosure or otherwise taking action under the PRA or other applicable Law. Under no circumstances will City be responsible or liable to Principal Project Company or any other party as a result of disclosure is deemed required by Law or by an order of court or City's general counsel or occurs through inadvertence, mistake or negligence on the part of City or its officers, employees, contractors or consultants.

**21.3.4** Nothing contained in this <u>Section 21.3</u> (Public Records Act and San Francisco Sunshine Ordinance) shall modify or amend requirements and obligations imposed on City by the PRA or other applicable Law, and the provisions of the PRA or other Laws shall control to the extent of a conflict between the procedures under this Agreement and applicable Law. City will not advise a submitting party or Principal Project Company as to the nature or content of documents entitled to protection from disclosure under the PRA or other applicable Laws, as to the interpretation of such Laws, or as to definition of trade secret. Principal Project Company is advised to contact its own legal counsel concerning the effect of applicable Laws to Principal Project Company's Books and Records and actions to be taken to preserve confidentiality.

**21.3.5** In the event of any proceeding or litigation concerning the disclosure of any Books and Records to Third Parties, City's sole involvement will be as a stakeholder retaining the material until otherwise ordered by a court or other authority having jurisdiction. Principal Project Company shall be responsible for prosecuting or defending any action, acting on its own behalf, concerning such materials at its sole expense and risk; provided, however, that City may intervene or participate in the litigation in such manner as it deems necessary or desirable. Principal Project Company shall indemnify and hold harmless Indemnitees from and against any and all claims, causes of action, suits, legal or administrative proceedings, damages, losses, liabilities, response costs, costs and expenses, including any injury to or death of persons or damage to or loss of property (including damage to utility facilities), and including attorneys' and expert witness fees and costs, arising out of, relating to or resulting from City's refusal to disclose any material that Principal Project Company has designated as a trade secret.

# 21.4 Intellectual Property

# 21.4.1 Developed IP

**21.4.1.1** Principal Project Company acknowledges and agrees that all Developed IP, in any medium, is either owned by City or specially ordered or commissioned by City, including works made for hire pursuant to 17 U.S.C. § 101 (the U.S. Copyright Act of 1976), and shall be owned by City upon authorship, creation, development or invention. Principal Project Company hereby assigns to City all rights, title and interest in and to the Developed IP including any and all Software, Work Product and designs. If any Developed IP is not the proper subject matter or is determined not to be a work-made-for-hire pursuant to the U.S. Copyright Act, Principal Project

Company hereby assigns, and shall cause all PPC-Related Entities to assign, to City all rights, title and interest in and to the Developed IP including any deliverable. Principal Project Company agrees to execute, and shall cause all PPC-Related Entities to execute, such further documents and to do such further acts as may be necessary to perfect, register, or enforce City's ownership of such rights, in whole or in part. If any PPC-Related Entity fails or refuses to execute any such documents, Principal Project Company, for itself and on behalf of any PPC-Related Entity, hereby appoints City as the necessary PPC-Related Entity's attorney-in-fact (this appointment is irrevocable and is coupled with an interest) to act on PPC-Related Entity's behalf and to execute such documents. Principal Project Company hereby forever waives and agrees never to assert, and shall cause any PPC-Related Entity to waive and never to assert, against City, its successors or licensees any and all "moral rights" (including claims based on 17 U.S.C. §§ 101-810 (the Copyright Act of 1976, as modified), specifically including 17 U.S.C. § 106A(a) (the Visual Artists Rights Act of 1990, "VARA")) that such PPC-Related Entity may have in any Intellectual Property or deliverable even after expiration or termination of this Agreement.

**21.4.1.2** All deliverables authored, created or developed under or for the purpose of this Agreement, the Work or the Project shall be owned by City immediately upon creation or generation, physically or digitally, and whether or not such deliverable and/or work product have been delivered to City under the terms of this Agreement.

**21.4.1.3** Principal Project Company shall deliver to City all deliverables and/or other work product authored, created or developed under or for the purpose of this Agreement (i) at time(s)/dates(s) pursuant to this Agreement, or (ii) as soon as reasonably practicable after such creation or generation, but in no event later than the effective date of termination of this Agreement.

# 21.4.2 Principal Project Company IP

**21.4.2.1** Principal Project Company hereby grants to City an irrevocable, perpetual, nonexclusive, transferable (solely to a permitted City's assignee), fully paid-up right and license to use, execute, perform, sublicense, exploit, manufacture, distribute, reproduce, adapt, display, and prepare derivative works ("**Base License Rights**") of Principal Project Company IP in connection with the Work or the Project. The rights granted herein shall survive the termination, expiration or cancellation of this Agreement or any rights related thereto.

**21.4.2.2** Principal Project Company shall identify and disclose to City all Principal Project Company IP required by, incorporated in, or combined with the Work or the Project.

# 21.4.3 Third Party IP

**21.4.3.1** Principal Project Company shall use commercially reasonable efforts to secure Base License Rights in the name of City to license Third Party IP in connection with the Project or Work, and shall pay any and all royalties and license fees required to be paid for any Intellectual Property required by, incorporated in, or combined with the Project IP.

**21.4.3.2** Subject to <u>Section 21.4.3.3</u>, if the owner of Third Party IP refuses to grant Base License Rights pursuant to <u>Section 21.4.3.1</u>, Principal Project Company shall:

(a) obtain City's prior written approval, which shall not to be unreasonably withheld, of the terms and conditions of Third Party IP licenses;

- (b) identify and disclose to City all Third Party IP required by, incorporated in, or combined with the Project IP; and
- (c) obtain from each owner of the Third Party IP prior consent to have the relevant Third Party IP deposited into an IP Escrow in accordance with <u>Section 21.5</u> (Intellectual Property Escrows), or, to the extent the owner of the relevant Third Party IP has not provided such consent, obtain City's prior written approval for a waiver of this requirement, not to be unreasonably withheld.

**21.4.3.3** <u>COTS</u>. Only if the owner of Third Party IP refuses to grant Base License Rights pursuant to <u>Section 21.4.3.1</u> and the subject Third Party IP is COTS, Principal Project Company shall secure license(s) in the name of City, based on commercially available terms for the COTS, including any standard end user license Contract. Principal Project Company shall provide (i) an outline of license deficiencies vs. Base License Rights and (ii) the identification of at least one (1) other COTS available for the same purpose, function or design. Principal Project Company shall identify and disclose to City all COTS required by, incorporated in, or combined with the Work and/or the Project.

# 21.4.4 City IP and City Data

**21.4.4.1** City hereby grants to PPC-Related Entities a limited, non-exclusive license to use, exploit, manufacture, distribute, reproduce, adapt and display the Project IP, City IP, and City Data, and any deliverable and/or other work product incorporating such Intellectual Property, solely in connection with and limited to the Allowed Uses. All rights not specifically granted in this <u>Section 21.4.4.1</u> are reserved to City. For the avoidance of doubt, no rights to City trademarks, whether or not the subject of a trademark state or US application or registration, ("City Marks") are granted to Principal Project Company and Principal Project Company may not incorporate, refer to, or otherwise use City Marks for any marketing, promotional or advertising purposes.

**21.4.4.2** In addition to Principal Project Company's obligations and restrictions related to City Data in this Agreement, Principal Project Company acknowledges and agrees that all City Data, including the results or creation of any anonymization, de-identification, aggregation or other analysis of such City Data, whether physical or digital, is owned by City. Except as specifically provided in this Agreement, no PPC-Related Entity shall make use of City Data, including any anonymized, de-identified, or aggregated versions thereof, even if such use is for such PPC-Related Entity's internal use or analysis, whether or not commercial value is available or received, and/or such information or data is available in other, separate or cumulative sources.

**21.4.4.3** Notwithstanding any other term or condition of this Agreement, the rights and permissions granted under this <u>Section 21.4.4</u> (City IP and City Data) shall terminate (i) upon the effective date of termination of this Agreement or (ii) upon 24-hour written Notice by City to Principal Project Company, whichever is earlier.

**21.4.4.4** Except as, and to the limited extent, required by applicable Laws, Principal Project Company shall keep and maintain, and shall cause all PPC-Related Entities to keep and maintain, all City IP and City Data strictly confidential. Before any release of any City IP or CTIP Data pursuant to applicable Laws, Principal Project Company must consult with City and the

City Attorney's Office regarding such release and obtain consent to such release. Any release shall be limited to the minimum required to satisfy the applicable Law.

### 21.4.5 Delivery of IP Materials

Principal Project Company shall deliver to City all IP Materials related to Principal Project Company IP and Third Party IP, or deposit such IP Materials into IP Escrow(s) in accordance with <u>Section 21.5</u> (Intellectual Property Escrows), as soon as reasonably practicable following incorporation of the relevant Intellectual Property into the Project or Work.

#### 21.4.6 Payments Inclusive

Principal Project Company acknowledges and agrees that the payments provided for in <u>Article</u> <u>11</u> (Payments to Principal Project Company) include all royalties, fees, costs and expenses arising from or related to the Project IP, including any fees pursuant to <u>Section 21.5</u> (Intellectual Property Escrows).

#### 21.5 Intellectual Property Escrows

**21.5.1** City and Principal Project Company acknowledge that Principal Project Company or other PPC-Related Entities may deliver IP Materials pursuant to <u>Section 21.4.5</u> (Delivery of IP Materials) that include Software, Source Code and Documentation or other Intellectual Property and may not wish to deliver the applicable IP Materials directly to City as public disclosure could deprive such Person of commercial value. Principal Project Company further acknowledges that City nevertheless must be ensured access to such IP Materials at any time, and must be assured that the IP Materials are delivered to City pursuant to <u>Section 21.4.5</u> (Delivery of IP Materials).

**21.5.2** In lieu of delivering the IP Materials directly to City, Principal Project Company may elect to deposit the IP Materials with a neutral depository. In such event, City and Principal Project Company shall: (a) mutually select one or more escrow companies or other neutral depositories (each an "**IP Escrow Agent**") engaged in the business of receiving and maintaining escrows of software source code and/or other Intellectual Property; (b) establish one or more escrows (each an "**IP Escrow**") with the IP Escrow Agent on terms and conditions reasonably acceptable to City and Principal Project Company for the deposit, retention, upkeep, authentication, confirmation and release of IP Materials to City pursuant to this Agreement; (c) determine the date(s) for Principal Project Company's deposit of the IP Materials into the IP Escrow; and (d) determine a process for releasing from escrow the IP Materials to be delivered to City pursuant to this Agreement. IP Escrows also may include Affiliates as parties and may include deposit of their Intellectual Property. Principal Project Company shall be responsible for the fees and costs of the IP Escrow Agent(s).

**21.5.3** The IP Escrows shall survive expiration or earlier termination of this Agreement regardless of the reason.

**21.5.4** The IP Materials shall be released and delivered to City in any of the following circumstances:

(a) this Agreement expires or is terminated prior to expiration for any reason;

- (b) voluntary or involuntary bankruptcy of Principal Project Company, PPC-Related Entity or the owner of Third Party IP, as to Principal Project Company IP or Third Party IP respectively; or
- (c) Principal Project Company, PPC-Related Entity or the owner of Third Party IP is dissolved or liquidated or otherwise ceases to engage in the ordinary course of the business of manufacturing, supplying, maintaining, and servicing the Software, product, part, or other item containing the relevant Intellectual Property.

# 21.6 City's Use of IP Materials

**21.6.1** City may exercise all rights, including the Base License Rights, granted to City pursuant to <u>Section 21.4</u> (Intellectual Property) for the purposes of the Project, including any subsequent expansion or additions, except that City's ownership or assigned rights pursuant to <u>Section 21.4.1</u> (Developed IP) shall not be limited in any way, for any purpose. For the avoidance of doubt, City's rights include the right to sublicense any City rights to a future vendor. City's rights under this clause shall survive the termination, expiration or cancellation of this Agreement.

**21.6.2** City shall maintain the confidentiality of any IP Materials released pursuant to <u>Section</u> 21.5 (Intellectual Property Escrows) pursuant to <u>Section 21.4</u> (Intellectual Property) and shall enter into a non-disclosure agreement with any third party to whom City, in its sole discretion, grants access to such IP Materials to the extent that such IP Materials contain Confidential Information.

# ARTICLE 22. ADVERTISING AND OTHER BUSINESS OPPORTUNITIES

#### 22.1 Rights and Interests in the Project and Project Site

Principal Project Company's rights and interests in the Project and Project Site under this Agreement are limited to such rights and interests that are required for performing the Work and Principal Project Company's timely fulfillment of its obligations under the Contract Documents. Principal Project Company's rights and interests exclude any Airspace or other real property interest.

#### 22.2 Advertising and Business Opportunities

- **22.2.1** City reserves all rights and opportunities concerning:
  - (a) advertising on the Infrastructure Facility and, as between Principal Project Company and City, within the Project Site, including use of Infrastructure Facility physical assets for advertising purposes; and
  - (b) entrepreneurial, commercial and business activities that are ancillary or collateral to the use and operation of the Infrastructure Facility and Project Site, whether developed or pursued by City or through others worldwide. The rights and opportunities reserved to City under this <u>Section 22.2.1(b)</u> include the rights described in <u>Section 22.2.1(a)</u> and any sponsorships, naming rights, etc. (collectively, "**Business Opportunities**").

**22.2.2** Principal Project Company shall cooperate and, during the D&C Period, grant all necessary access to the Project Site to City and any Third Party designees, including tenants and vendors, authorized by City in connection with City's exercise of its rights relating to the Project Site and any advertising and Business Opportunities. During the IFM Period, Principal Project Company shall not interfere with access to the Infrastructure Facility by City and any Third Party designees, including tenants and vendors, authorized by City in connection with City's exercise of its rights relating to its operations at the Project Site and any advertising and Business Opportunities. Unless otherwise agreed to by the Parties, City shall be entitled to all revenues generated by business opportunities arising out of, relating to or resulting from the Infrastructure Facility or in the Project Site's Airspace, except rents paid by any Housing and Commercial Component tenants. Principal Project Company shall be compensated for reasonable costs and expenses incurred directly by Principal Project Company in installing and maintaining facilities for advertising or Business Opportunities (other than routine maintenance), through a Change Order.

**22.2.3** Except as authorized by City, Principal Project Company shall not engage in, and shall not permit:

- (a) any advertising within the Project Site or within or on the exterior of the Infrastructure Facility;
- (b) use or occupation of the Project for any Business Opportunities; and

(c) operation of any business at the Project Site or the Infrastructure Facility, including (i) the sale of products or services (including any newsstand or concession stand for the sale of food, beverages or gifts or other retail or rental services); or (ii) the sale or rental of any wire, cable, transmission or receiving device or any other utility on, or transmission or receipt of any electronic communication to or from, any part of the Project.

**22.2.4** Principal Project Company may request City to consider Business Opportunities. If City, in its sole discretion, consents, the Parties shall execute an amendment to the Contract Documents memorializing the agreement reached, including any agreement as to any revenue and cost attribution. Notwithstanding the foregoing, Principal Project Company shall be compensated pursuant to a City Change for Principal Project Company's reasonable costs and expenses that are directly attributable to implementation of such Business Opportunities (other than routine maintenance) as well as any support efforts the City Change requires Principal Project Company to provide.

**22.2.5** Unless expressly approved by City, Principal Project Company may not permit any Person to use or occupy the Project for any ancillary or collateral purpose.

# 22.3 Remedies

If a PPC Default concerns a breach of the provisions of <u>Section 22.2</u> (Advertising and Business Opportunities), then, in addition to any other remedies available to City under this Agreement or applicable Law, City shall be entitled to receive from Principal Project Company an amount equal to all profits from the prohibited activity, together with interest thereon at the Late Payment Rate from the date of collection until the date payment is made. In addition, City may require Principal Project Company to restore the Project Site and the Infrastructure Facility to its original condition or to transfer to City all of Principal Project Company's interest in the prohibited assets and improvements and revenues derived therefrom, or any combination of the foregoing.

# ARTICLE 23. MISCELLANEOUS

# 23.1 Standard for Approvals

In all cases where approvals, acceptances or consents are required to be provided by City or Principal Project Company hereunder, such approvals, acceptances or consents shall not be withheld unreasonably except in cases where a different standard (such as sole discretion) is specified. Any authorization by City shall be in writing. Any approval required by the SFMTA Board or Board of Supervisors will be at their respective sole discretion. In cases where sole discretion is specified, the decision shall not be subject to Contract Dispute Procedures hereunder.

#### 23.2 Amendments

The Contract Documents may be amended only by a written instrument duly executed by or on behalf of the Parties, except to the extent expressly provided otherwise in this Agreement.

#### 23.3 Waiver

**23.3.1** The failure of a Party to exercise or delay in exercising any right under this Agreement shall not:

- (a) constitute a waiver of such right or any other right under the Contract Documents; or
- (b) relieve the other Party from performance of its obligations under the Contract Documents except as otherwise provided in the Contract Documents.

**23.3.2** No waiver of any right under this Agreement shall be effective unless made in a writing duly executed by a duly authorized representative of the Party charged with the waiver.

**23.3.3** Any waiver under <u>Section 23.3.2</u> shall be limited to the specific instance and shall not constitute a waiver of such right in the future or of any other right under this Agreement.

**23.3.4** If the Parties make and implement any interpretation of the Contract Documents without documenting such interpretation by an instrument in writing signed by both Parties, such interpretation and implementation thereof will not be binding in the event of any future Contract Disputes.

**23.3.5** No waiver of any right under this Agreement shall be deemed to have occurred as the result of any acceptance by City, any payment for or acceptance of the whole or any part of the Work, any extension of time, or any possession taken by City.

#### 23.4 Independent Contractor; No Joint Venture or Partnership

**23.4.1** Principal Project Company is an independent contractor. Neither Principal Project Company nor any of its employees or agents is or shall be deemed to be an employee or agent of City, and in no event shall the relationship between City and Principal Project Company be construed as creating any relationship whatsoever between City and Principal Project Company's employees or agents. Except as otherwise provided in the Contract Documents,

Principal Project Company has sole authority and responsibility to employ, discharge and otherwise control its employees and has complete and sole responsibility as a principal for its agents, for all Contractors and for all other Persons that Principal Project Company or any Contractor hires to perform or assist in performing the Work.

**23.4.2** Nothing in the Contract Documents is intended or shall be construed to create any partnership, joint venture or similar relationship between City and Principal Project Company; and in no event shall either Party take a position in any tax return or other writing of any kind that a partnership, joint venture or similar relationship exists.

**23.4.3** Principal Project Company shall not have, or be deemed to have, power or authority to make any commitments on City's behalf or to execute agreements in the name of or on behalf of City. Principal Project Company shall not enter into any agreement with any Governmental Entity, Utility Owner, property owner or other Third Party having regulatory jurisdiction over any aspect of the Project or Work or having any property interest affected by the Project or the Work that in any way purports to obligate City, or states or implies that City has an obligation to the Third Party, to undertake any activity, unless City otherwise approves.

### 23.5 Successors and Assigns

The Contract Documents shall be binding upon and inure to the benefit of City and Principal Project Company and each of their permitted successors, assigns and legal representatives.

# 23.6 Designation of Representatives; Cooperation with Representatives

**23.6.1** City and Principal Project Company shall each designate an individual or individuals who shall be authorized to make decisions and bind the Parties on matters relating to the Contract Documents. <u>Exhibit 10</u> (Initial Designation of Authorized Representatives) provides the initial Authorized Representative designations. A Party may change such designations by written Notice in accordance with <u>Section 23.10</u> (Notices and Communications).

**23.6.2** Principal Project Company shall cooperate with City and all representatives of City designated as described above in performance of their obligations under the Contract Documents.

### 23.7 Survival

The following provisions shall survive the expiration or earlier termination of this Agreement and/or the completion of the Work:

- (a) Principal Project Company's and City's representations and warranties;
- (b) the Contract Dispute Procedures;
- (c) the indemnifications, limitations and releases contained in <u>Section 10.6</u> (Indemnities);
- (d) the limitations on remedies contained in <u>Section 16.6</u> (Limitation on Consequential Damages);

- (e) the express obligations of the Parties following termination (including those in <u>Articles 17</u> (Termination) and <u>20</u> (Assignment and Transfer);
- (f) the Direct Agreement); and
- (g) all other provisions which by their inherent character should survive expiration or earlier termination of this Agreement and/or completion of the Work.

### 23.8 Limitation on Third Party Beneficiaries

Except to the extent that specific provisions (such as the warranty and indemnity provisions, and the provisions for the protection of certain Lenders under any Direct Agreement) identify Third Parties and state that they are entitled to benefits, (a) it is not intended by any of the provisions of the Contract Documents to create any third party beneficiary to this Agreement or to authorize anyone not a Party to maintain a suit for personal injury or property damage under this Agreement, and (b) the duties, obligations and responsibilities of the Parties with respect to third parties shall remain as imposed by Law. The Contract Documents shall not be construed to create a contractual relationship of any kind between City and a Contractor or any Person other than Principal Project Company.

### 23.9 Governing Law; Venue

The Contract Documents shall be governed by and construed in accordance with the laws of the State, any applicable federal law, the San Francisco City Charter and Municipal Code, and the ordinances, regulations, codes, and Executive Orders enacted and/or promulgated pursuant thereto. The venue for any litigation arising from a Contract Dispute shall be in San Francisco, California to the extent that a court located in San Francisco has subject matter jurisdiction.

### 23.10 Notices and Communications

**23.10.1** All notices, requests, demands, instructions, certificates, consents, explanations, agreements, approvals and other communications (each being a "**Notice**") required or permitted under this Agreement must be in writing (whether or not "written notice" or "notice in writing" is specifically required by the applicable provision of this Agreement) and (a) delivered in person, (b) sent by commercial courier, next business day delivery requested, or (c) by registered, certified mail or express mail, return receipt requested, with postage prepaid, to the mailing addresses below. All Notices under this Agreement will be deemed given, received, made or communicated on the date personal receipt actually occurs or, if mailed, on the delivery date or attempted delivery date shown on the return receipt. For the convenience of the Parties, copies of Notices may also be given by email to the email address given below but the emailed copy will not be binding on either Party.

**23.10.2** The effective time of a Notice will not be affected by the receipt of the email copy of the Notice.

**23.10.3** Any mailing address, or email address, may be changed at any time by giving written Notice of the change in the manner provided above at least ten (10) days before the effective date of the change.

(a) All Notices to Principal Project Company shall be delivered to the following address or as otherwise directed by Principal Project Company's Authorized Representative:

c/o	
Attention:	
Telephone: _	

(b) All Notices to City shall be marked as regarding the Project and shall be delivered to the following address or as otherwise directed by City's Authorized Representative:

	San Francisco Municipal Transportation Agency 1 South Van Ness, 8th Floor San Francisco, CA 94103 Attn: Chris Lazaro Email: <u>Chris.Lazaro@sfmta.com</u>
With a copy to:	Office of the City Attorney City Hall, Room 234 1 Dr. Carlton B. Goodlett Place San Francisco, California 94102-4682 Attn: Real Estate & Finance Group Re: Potrero Yard Modernization Project Email: isidro.jimenez@sfcityatty.org

**23.10.4** Any technical or other communications pertaining to the Work shall be conducted by Principal Project Company's Authorized Representative and City's Authorized Representative.

**23.10.5** Principal Project Company shall promptly provide to City a copy of each communication received from any Lender relating to any default or event of default under any Financing Agreement or Security Document.

### 23.11 Severability

**23.11.1** If any provision or part of the Contract Documents is ruled invalid (including invalidity due to any statutory change or other change in Law) by a court having proper jurisdiction, then the Parties shall: (a) promptly meet and negotiate a substitute for such provision or part, which shall, to the greatest extent legally permissible, effect the original intent of the Parties, including an equitable adjustment to the Financial Model Update (or, if there has been no Update, the original Financial Model) and Principal Project Company's compensation to account for any change in the Work resulting from such invalidated portion; and (b) if necessary or desirable, apply to the court or other decision maker (as applicable) which declared such invalidity for an interpretation of the invalidated portion to guide the negotiations. The invalidity or

unenforceability of any such provision or part shall not affect the validity or enforceability of the balance of the Contract Documents, which shall be construed and enforced as if the Contract Documents did not contain such invalid or unenforceable provision or part.

**23.11.2** If after the efforts required by <u>Section 23.11.1</u>, no interpretation or reformation of the Contract Documents can reasonably be adopted that will return the Parties to the benefits of their original bargain, then the court order shall be treated as a Termination Due to Court Ruling under <u>Section 17.4.3</u> (Termination Due to Court Ruling).

# 23.12 Construction and Interpretation of Agreement

**23.12.1** The Contract Documents shall be construed simply, as a whole and in accordance with the fair meaning of the language used and not strictly for or against any Party.

**23.12.2** The Parties acknowledge and agree that: (a) the Contract Documents are the product of an extensive and thorough, arm's-length exchange of ideas, questions, answers, information and drafts during the Implementation Proposal preparation process; (b) each Party has been given the opportunity to independently review the Contract Documents with legal counsel; and (c) each Party has the requisite experience and sophistication to negotiate, understand, interpret and agree to the particular language of the provisions of the Contract Documents. Accordingly, in the event of a conflict, ambiguity or inconsistency in or Contract Dispute regarding the interpretation of the Contract Documents, the Contract Documents shall not be interpreted or construed against the Party preparing it, and instead the Contract Dispute resolver shall consult other customary rules of interpretation and construction.

**23.12.3** City's final answers to the questions posed during the Implementation Proposal preparation process for the Contract Documents shall in no event be deemed part of the Contract Documents and shall not be relevant in interpreting the Contract Documents except as they may clarify provisions otherwise considered ambiguous.

**23.12.4** The captions of the articles, sections and subsections in the Contract Documents are for convenience only and are not to be treated or construed as part of this Agreement.

**23.12.5** Unless otherwise expressly stated in the Contract Documents, words that have well-known technical or construction industry meanings are used in the Contract Documents in accordance with such recognized meaning.

**23.12.6** Wherever the word "including", "includes" or "include" is used in the Contract Documents, it is deemed to be followed by the words "without limitation." Wherever reference is made in the Contract Documents to a particular Governmental Entity, it includes any public agency succeeding to the powers and authority of such Governmental Entity.

**23.12.7** References to "days" contained in the Contract Documents shall mean calendar days unless otherwise stated.

**23.12.8** Subject to <u>Section 23.12.9</u>, if the day on or by which any thing is to be done in accordance with this Agreement is not a Business Day, that thing must be done on the next Business Day.

**23.12.9** If the Contract Documents require action to be taken in the event of an emergency and otherwise where it is clear that performance is intended to occur on a non-Business Day, Principal Project Company shall be required to perform such obligations, even though the date in question may fall on a non-Business Day.

**23.12.10** As used in this Agreement and as the context may require, the singular includes the plural and vice versa, and the masculine gender includes the feminine and vice versa.

**23.12.11** All monetary amounts and obligations in the Contract Documents are expressed and payable in U.S. dollars.

**23.12.12** Each party must perform its obligations in accordance with this Agreement at its own cost, unless expressly provided otherwise.

**23.12.13** If this Agreement requires calculation of an amount payable to a party there must be no double counting in calculating that amount.

#### 23.13 Further Assurances

Each Party shall promptly execute and deliver to the other Party all such instruments and other documents and assurances as are reasonably requested by the second Party to further evidence its obligations hereunder, including, specifically with respect to Principal Project Company, assurances regarding the validity of (a) the assignments of Contracts contained herein, and (b) any instruments securing performance hereof.

#### 23.14 Entire Agreement

City and Principal Project Company agree and expressly intend for the Contract Documents to constitute a single, non-severable, integrated agreement whose terms are interdependent and non-divisible. The Contract Documents contain the entire understanding of the Parties with respect to the subject matter of this Agreement and supersede all prior agreements, understandings, statements, representations and negotiations, in each case oral or written, between the Parties with respect to the subject to the subject to the subject matter of this Agreement.

#### 23.15 Counterparts

This instrument may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. IN WITNESS WHEREOF, the Parties have executed this Agreement as of the date first written above.

# APPROVED AS TO FORM

David Chiu, City Attorney

CITY AND COUNTY OF SAN FRANCISCO, a municipal corporation, operating by and through the San Francisco Municipal Transportation Agency

By:		
	[•]	

Deputy City Attorney

By: Jeffrey Tumlin Director of Transportation

Date:\_\_\_\_\_

# Date:\_\_\_\_\_

APPROVED BY:

San Francisco Municipal Transportation Agency	/	
Board of Directors		
Resolution No:		
Adopted:		
Attest:		

Secretary, SFMTA Board of Directors

#### [PRINCIPAL PROJECT COMPANY]

By:

Signature

Print Name

Print Title

Ву:\_\_\_\_\_

Signature

\_\_\_\_\_

Print Name

Print Title

### EXHIBIT 1 ABBREVIATIONS AND DEFINITIONS

# EXHIBIT 1

### ABBREVIATIONS AND DEFINITIONS

Unless otherwise specified, whenever the following abbreviations or terms are used in this Agreement and the other Contract Documents, they have the meanings given below. References in this <u>Exhibit 1</u> (Abbreviations and Definitions) to Sections and Exhibits shall mean sections of and exhibits to this Agreement.

ACI	American Concrete Institute
ACM	Asbestos Containing Material
ADA	Americans with Disabilities Act
ADAAG	Americans with Disabilities Act Accessibility Guidelines
AED	Automatic Electronic Defibrillator
AHA	Activity Hazard Analysis
AHJ	Authority(ies) Having Jurisdiction
AISC	American Institute of Steel Construction
ANSI	American National Standards Institute
APTA	American Public Transportation Association
AQMD	Air Quality Management District
ARI	Air Conditioning and Refrigerator Institute
ASCE	American Society of Civil Engineers
ASHRAE	American Society of Heating, Refrigerating and Air-Conditioning Engineers
ASME	American Society of Mechanical Engineers
ASTM	ASTM International (formerly known as the American Society for Testing and
	Materials)
AW	Assigned Weight
AWS	American Welding Society
BCI	Buildings Cost Index
BEB	Battery Electric Bus
BAS	Building Automation System
BIL	Basic Insulation Level
BIM	Building Information Modeling
BOCC	Building Operations Control Center
BODR	Basis of Design Report
BTU	British Thermal Unit
CADD	Computer Aided Design and Drafting
CALM	Coordination and Logistics Management Program
CAFM	Computer Aided Facility Management
CAP	Compliance Action Plan
CARPP	Capital Asset Replacement Program Plan
CBC	California Building Code
CCTV	Closed-Circuit Television
CDRL	Contract Data Requirements List
CEC	California Electrical Code
CEI	Construction Engineering and Inspection
CEL	Certifiable Elements List
CEQA	California Environmental Quality Act, California Public Resources Code § 21000
	et seq., as it may be amended
CFA	Certificate of Final Acceptance
CFR	Code of Federal Regulations

CIB	Communications Infrastructure Backbone
CIDH	Cast In Drilled Hole
CIH	Central Instrument House
CIL	Certifiable Items List
CIR	Committed Information Rate
CIS	Customer Information System
CL	Checklist
CM	Construction Management
CO	Carbon Monoxide
CPESEC	Customer Premises Equipment
CPI	Consumer Price Index
СРМ	Critical Path Method
CPT	Cone Penetrometer Test
CPTED	Crime Prevention through Environmental Design
CPU	Central Processing Unit
CPUC	California Public Utilities Commission
CQCM	Construction Quality Control Manager
CS	Construction/Installation Supervisor
CSEMS	Construction Site Environmental Management Supervisor
CSP	Construction Security Plan
CWA	Clean Water Act
D&C	Design and Construction
D/CID	Design Construction Integration Documents
DC	Direct Current
DTS	Data Transmission System
DVD	Digital Video Disc
DVMS	Digital Video Management System
EA	Environmental Assessment
ECI	Environmental Compliance Inspector
ECP	Environmental Compliance Plan
EEO	Equal Employment Opportunity
EIA	Electronic Industries Association
EMP	Emergency Management Panel
EMS	Emergency Medical Services
EP	Extraction Procedure (toxicity)
EPA	Environmental Protection Agency
EPS	Electrical Power System
ER	Equipment Room
ERRS	Electricity Rate Risk Share
ESC	Erosion and Sediment Control
ESD	Environmental Site Design
ET	Environmental Team
ETEL	Emergency Telephone
EVP	Emergency Vehicle Preemption
FCI	Facility Condition Index
FA	Forced Air
FAS	Fire Alarm System
FC	Footcandle
FDC	Fire Department Connections
FEIR	Final Environmental Impact Report
FF&E	Furniture, Fixtures and Equipment
FMP	Fire Management Panel

FMS       Fire Management System         FOD       Foreign Object Debris         FONSI       Finding of No Significant Impact         FOV       Field of View         FTA       Federal Transit Administration         GDR       Geotechnical Engineering Reports         GIS       Geotechnical Engineering Reports         GSR       Geotechnical Planning Report         GRM       Galions per Minute         GSD       Geotechnical Planning Report         GRS       Galvanized Rigid Steel         GSD       General System Display         HCF       Hundred Cubic Feet         HVAC       Heating, Ventilation, and Air Conditioning         ICEA       Insulated Cable Engineers Association         ID       Identifier         IDO       Intermediate Distribution Frame         IDO       Interdepartmental Order         IEEE       Institute of Electrical and Electronics Engineers         IEEE       Institute of Electrical and Electronics Engineers         IESNA       Illuminating Engineer Society of North America (also known as IES)         IF       Infrastructure Facility Maintenance         IMP       Incident Management Plan         IPMVP       International Performance Measurement and Verification Protoco
FONSI       Finding of No Significant Impact         FOV       Field of View         FTA       Federal Transit Administration         GDR       Geotechnical Data Report         GER       Geotechnical Engineering Reports         GIS       Geographical Information System         GPM       Gallons per Minute         GPR       Geotechnical Planning Report         GRS       Galvanized Rigid Steel         GSD       General System Display         HCF       Hundred Cubic Feet         HVAC       Heating, Ventilation, and Air Conditioning         IDE       Intermediate Distribution Frame         IDD       Identifier         IDF       Interdepartmental Order         IEC       Interdepartmental Order         IEE       Institute of Electrical and Electronics Engineers         IESNA       Illuminating Engineer Society of North America (also known as IES)         IF       Infrastructure Facility         IFM       Infrastructure Facility         IFM       Infrastructure Facility Maintenance         IMP       Incident Management Plan         IPMVP       International Performance Measurement and Verification Protocol         IAQ       Indoor Air Quality         ISO
FOV       Field of View         FTA       Federal Transit Administration         GDR       Geotechnical Data Report         GER       Geotechnical Engineering Reports         GIS       Geographical Information System         GPM       Gallons per Minute         GPR       Geotechnical Planning Report         GRS       Galvanized Rigid Steel         GSD       General System Display         HCF       Hundred Cubic Feet         HVAC       Heating, Ventilation, and Air Conditioning         IDE       Identifier         IDF       Intermediate Distribution Frame         IDO       Interrdepartmental Order         IEEE       Institute of Electrical and Electronics Engineers         IEEE       Institute of Electrical and Electronics Engineers         IESNA       Illuminating Engineer Society of North America (also known as IES)         IF       Infrastructure Facility Maintenance         IMP       Incident Management Plan         IPMVP       International Performance Measurement and Verification Protocol         IAQ       Indoor Air Quality         ISO       International Organization for Standardization         LAN       Local Area Network         LCD       Liguid Crystal Display
FTA       Federal Transit Administration         GDR       Geotechnical Data Report         GER       Geotechnical Engineering Reports         GIS       Geographical Information System         GPM       Galions per Minute         GPR       Geotechnical Planning Report         GRS       Galvanized Rigid Steel         GSD       General System Display         HCF       Hundred Cubic Feet         HVAC       Heating, Ventilation, and Air Conditioning         ID       Identifier         IDD       Insulated Cable Engineers Association         ID       Identifier         IDO       Intermediate Distribution Frame         IDO       Intermediate Distribution Frame         IDO       Interrational Electrotechnical Commission         IEEE       Institute of Electrical and Electronics Engineers         IESNA       Illuminating Engineer Society of North America (also known as IES)         IF       Infrastructure Facility Maintenance         IMP       Incident Management Plan         IPMVP       International Performance Measurement and Verification Protocol         IAQ       Indoor Air Quality         ISO       International Organization for Standardization         LAN       Local Area Network
GDR       Geotechnical Data Report         GER       Geotechnical Engineering Reports         GIS       Geographical Information System         GPM       Gallons per Minute         GPR       Geotechnical Planning Report         GRS       Galvanized Rigid Steel         GSD       General System Display         HCF       Hundred Cubic Feet         HVAC       Heating, Ventilation, and Air Conditioning         ICEA       Insulated Cable Engineers Association         ID       Identifier         IDF       Intermediate Distribution Frame         IEC       Intermediate Distribution Frame         IDO       Interdepartmental Order         IEEE       Institute of Electrotechnical Commission         IEEE       Institute of Electrotechnical Commission         IEEE       Institute of Electrical and Electronics Engineers         IESNA       Illuminating Engineer Society of North America (also known as IES)         IF       Infrastructure Facility         IFM       Infrastructure Facility         IFM       Infrastructure Facility Maintenance         IMP       Incident Management Plan         IPMVP       International Organization for Standardization         LAN       Local Area Network
GER       Geotechnical Engineering Reports         GIS       Geographical Information System         GPM       Gallons per Minute         GPR       Geotechnical Planning Report         GRS       Galvanized Rigid Steel         GSD       General System Display         HCF       Hundred Cubic Feet         HVAC       Heating, Ventilation, and Air Conditioning         ICEA       Insulated Cable Engineers Association         ID       Identifier         IDF       Intermediate Distribution Frame         IDO       Interdepartmental Order         IEE       Institute of Electrical and Electronics Engineers         IESNA       Illuminating Engineer Society of North America (also known as IES)         IF       Infrastructure Facility         IFM       Infrastructure Facility Maintenance         IMP       Incident Management Plan         IPMVP       International Performance Measurement and Verification Protocol         IAQ       Indoor Air Quality         ISO       International Organization for Standardization         LAN       Local Area Network         LCD       Liquid Crystal Display         LED       Leadership in Energy & Environmental Design         LEED NC       LEED New Construction
GIS       Geographical Information System         GPM       Gallons per Minute         GPR       Geotechnical Planning Report         GRS       Galvanized Rigid Steel         GSD       General System Display         HCF       Hundred Cubic Feet         HVAC       Heating, Ventilation, and Air Conditioning         ICEA       Insulated Cable Engineers Association         ID       Identifier         IDF       Intermediate Distribution Frame         IDO       Interrediate Distribution Frame         IDO       International Electrotechnical Commission         IEEE       Institute of Electrical and Electronics Engineers         IESNA       Illuminating Engineer Society of North America (also known as IES)         IF       Infrastructure Facility         IFM       Infrastructure Facility Maintenance         IMP       Incident Management Plan         IPMVP       International Organization for Stan
GPM       Gallons per Minute         GPR       Geotechnical Planning Report         GRS       Galvanized Rigid Steel         GSD       General System Display         HCF       Hundred Cubic Feet         HVAC       Heating, Ventilation, and Air Conditioning         ICEA       Insulated Cable Engineers Association         ID       Identifier         IDF       Intermediate Distribution Frame         IDO       International Electrotechnical Commission         IEEE       Institute of Electrical and Electronics Engineers         IESNA       Illuminating Engineer Society of North America (also known as IES)         IF       Infrastructure Facility         IFM       Infrastructure Facility Maintenance         IMP       Incident Management Plan         IPMVP       International Performance Measurement and Verification Protocol         IAQ       Indoor Air Quality         ISO       International Organization for Standardization         LAN       Local Area Network         LCD       Liquid Crystal Display         LED       Leght Emitting Diode         LEED NC       LEED New Construction         LEED NC       LEED New Construction         LEED NC       Liquiditight Flexible Metallic Conduit
GPR       Geotechnical Planning Report         GRS       Galvanized Rigid Steel         GSD       General System Display         HCF       Hundred Cubic Feet         HVAC       Heating, Ventilation, and Air Conditioning         ICEA       Insulated Cable Engineers Association         ID       Identifier         IDF       Intermediate Distribution Frame         IDO       Interdepartmental Order         IECE       Institute of Electrical and Electronics Engineers         IEEE       Institute of Electrical and Electronics Engineers         IF       Infrastructure Facility         IFM       Infrastructure Facility Maintenance         IMP       Incident Management Plan         IPMVP       International Performance Measurement and Verification Protocol         IAQ       Indoor Air Quality         ISO       International Organization for Standardization         LAN       Local Area Network         LCD       Liquid Crystal Display         LEED       Leadership in Energy & Environmental Design         LEED NC       LEED New Construction         LFMC       Liquidight Flexible Metallic Conduit         LiDAR       Light Detecting and Ranging
GRS       Galvanized Rigid Steel         GSD       General System Display         HCF       Hundred Cubic Feet         HVAC       Heating, Ventilation, and Air Conditioning         ICEA       Insulated Cable Engineers Association         ID       Identifier         IDF       Intermediate Distribution Frame         IDO       International Electrotechnical Commission         IEEE       Institute of Electrical and Electronics Engineers         IESNA       Illuminating Engineer Society of North America (also known as IES)         IF       Infrastructure Facility         IFM       Infrastructure Facility Maintenance         IMP       Incident Management Plan         IPMVP       International Performance Measurement and Verification Protocol         IAQ       Indoor Air Quality         ISO       International Organization for Standardization         LAN<
GSD       General System Display         HCF       Hundred Cubic Feet         HVAC       Heating, Ventilation, and Air Conditioning         ICEA       Insulated Cable Engineers Association         ID       Identifier         IDF       Intermediate Distribution Frame         IDO       Interdepartmental Order         IEC       International Electrotechnical Commission         IEEE       Institute of Electrical and Electronics Engineers         IESNA       Illuminating Engineer Society of North America (also known as IES)         IF       Infrastructure Facility         IFM       Infrastructure Facility Maintenance         IMP       Incident Management Plan         IPMVP       International Performance Measurement and Verification Protocol         IAQ       Indoor Air Quality         ISO       International Organization for Standardization         LAN       Local Area Network         LCD       Liquid Crystal Display         LED       Light Emitting Diode         LEED NC       LEED New Construction         LEED NC       LEED New Construction         LFMC       Liquidtight Flexible Metallic Conduit         LiDAR       Light Detecting and Ranging
HCF       Hundred Cubic Feet         HVAC       Heating, Ventilation, and Air Conditioning         ICEA       Insulated Cable Engineers Association         ID       Identifier         IDF       Intermediate Distribution Frame         IDO       Interdepartmental Order         IEC       International Electrotechnical Commission         IEEE       Institute of Electrical and Electronics Engineers         IESNA       Illuminating Engineer Society of North America (also known as IES)         IF       Infrastructure Facility         IFM       Infrastructure Facility         IPMVP       International Performance Measurement and Verification Protocol         IAQ       Indoor Air Quality         ISO       International Organization for Standardization         LAN       Local Area Network         LCD       Liquid Crystal Display         LEED       Leadership in Energy & Environmental Design         LEED NC       LEED New Construction         LEED NC       LEED New Construction         LFMC       Liquidtight Flexible Metallic Conduit         LiDAR       Light Detecting and Ranging
HVAC       Heating, Ventilation, and Air Conditioning         ICEA       Insulated Cable Engineers Association         ID       Identifier         IDF       Intermediate Distribution Frame         IDO       Interdepartmental Order         IEC       International Electrotechnical Commission         IEEE       Institute of Electrical and Electronics Engineers         IESNA       Illuminating Engineer Society of North America (also known as IES)         IF       Infrastructure Facility         IFM       Infrastructure Facility Maintenance         IMP       Incident Management Plan         IPMVP       International Performance Measurement and Verification Protocol         IAQ       Indoor Air Quality         ISO       International Organization for Standardization         LAN       Local Area Network         LCD       Liquid Crystal Display         LED       Leadership in Energy & Environmental Design         LEED NC       LEED New Construction         LEED NC       LEED New Construction         LFMC       Liquidtight Flexible Metallic Conduit         LiDAR       Light Detecting and Ranging
ICEA       Insulated Cable Engineers Association         ID       Identifier         IDF       Intermediate Distribution Frame         IDO       Interdepartmental Order         IEC       International Electrotechnical Commission         IEEE       Institute of Electrical and Electronics Engineers         IESNA       Illuminating Engineer Society of North America (also known as IES)         IF       Infrastructure Facility         IFM       Infrastructure Facility Maintenance         IMP       Incident Management Plan         IPMVP       International Performance Measurement and Verification Protocol         IAQ       Indoor Air Quality         ISO       International Organization for Standardization         LAN       Local Area Network         LCD       Liquid Crystal Display         LEED       Leadership in Energy & Environmental Design         LEED NC       LEED New Construction         LFMC       Liquidtight Flexible Metallic Conduit         LiDAR       Light Detecting and Ranging
ID       Identifier         IDF       Intermediate Distribution Frame         IDO       Interdepartmental Order         IEC       International Electrotechnical Commission         IEEE       Institute of Electrical and Electronics Engineers         IESNA       Illuminating Engineer Society of North America (also known as IES)         IF       Infrastructure Facility         IFM       Infrastructure Facility Maintenance         IMP       Incident Management Plan         IPMVP       International Performance Measurement and Verification Protocol         IAQ       Indoor Air Quality         ISO       International Organization for Standardization         LAN       Local Area Network         LCD       Liquid Crystal Display         LED       Light Emitting Diode         LEED       Leadership in Energy & Environmental Design         LEED NC       LEED New Construction         LFMC       Liquidtight Flexible Metallic Conduit         LiDAR       Light Detecting and Ranging
IDFIntermediate Distribution FrameIDOInterdepartmental OrderIECInternational Electrotechnical CommissionIEEEInstitute of Electrical and Electronics EngineersIESNAIlluminating Engineer Society of North America (also known as IES)IFInfrastructure FacilityIFMInfrastructure Facility MaintenanceIMPIncident Management PlanIPMVPInternational Performance Measurement and Verification ProtocolIAQIndoor Air QualityISOInternational Organization for StandardizationLANLocal Area NetworkLCDLiquid Crystal DisplayLEEDLeadership in Energy & Environmental DesignLEED NCLEED New ConstructionLFMCLiquidtight Flexible Metallic ConduitLiDARLight Detecting and Ranging
IDOInterdepartmental OrderIECInternational Electrotechnical CommissionIEEEInstitute of Electrical and Electronics EngineersIESNAIlluminating Engineer Society of North America (also known as IES)IFInfrastructure FacilityIFMInfrastructure Facility MaintenanceIMPIncident Management PlanIPMVPInternational Performance Measurement and Verification ProtocolIAQIndoor Air QualityISOInternational Organization for StandardizationLANLocal Area NetworkLCDLiquid Crystal DisplayLEDLeadership in Energy & Environmental DesignLEED NCLEED New ConstructionLFMCLiquidtight Flexible Metallic ConduitLiDARLight Detecting and Ranging
IECInternational Electrotechnical CommissionIEEEInstitute of Electrical and Electronics EngineersIESNAIlluminating Engineer Society of North America (also known as IES)IFInfrastructure FacilityIFMInfrastructure Facility MaintenanceIMPIncident Management PlanIPMVPInternational Performance Measurement and Verification ProtocolIAQIndoor Air QualityISOInternational Organization for StandardizationLANLocal Area NetworkLCDLiquid Crystal DisplayLEDLeadership in Energy & Environmental DesignLEED NCLEED New ConstructionLFMCLiquidtight Flexible Metallic ConduitLiDARLight Detecting and Ranging
IEEEInstitute of Electrical and Electronics EngineersIESNAIlluminating Engineer Society of North America (also known as IES)IFInfrastructure FacilityIFMInfrastructure Facility MaintenanceIMPIncident Management PlanIPMVPInternational Performance Measurement and Verification ProtocolIAQIndoor Air QualityISOInternational Organization for StandardizationLANLocal Area NetworkLCDLiquid Crystal DisplayLEEDLeadership in Energy & Environmental DesignLEED NCLEED New ConstructionLFMCLiquidtight Flexible Metallic ConduitLiDARLight Detecting and Ranging
IESNAIlluminating Engineer Society of North America (also known as IES)IFInfrastructure FacilityIFMInfrastructure Facility MaintenanceIMPIncident Management PlanIPMVPInternational Performance Measurement and Verification ProtocolIAQIndoor Air QualityISOInternational Organization for StandardizationLANLocal Area NetworkLCDLiquid Crystal DisplayLEDLight Emitting DiodeLEEDLeadership in Energy & Environmental DesignLEED NCLEED New ConstructionLFMCLiquidtight Flexible Metallic ConduitLiDARLight Detecting and Ranging
IF       Infrastructure Facility         IFM       Infrastructure Facility Maintenance         IMP       Incident Management Plan         IPMVP       International Performance Measurement and Verification Protocol         IAQ       Indoor Air Quality         ISO       International Organization for Standardization         LAN       Local Area Network         LCD       Liquid Crystal Display         LED       Light Emitting Diode         LEED       Leadership in Energy & Environmental Design         LEED NC       LEED New Construction         LFMC       Liquidtight Flexible Metallic Conduit         LiDAR       Light Detecting and Ranging
IFMInfrastructure Facility MaintenanceIMPIncident Management PlanIPMVPInternational Performance Measurement and Verification ProtocolIAQIndoor Air QualityISOInternational Organization for StandardizationLANLocal Area NetworkLCDLiquid Crystal DisplayLEDLight Emitting DiodeLEEDLeadership in Energy & Environmental DesignLEED NCLEED New ConstructionLFMCLiquidtight Flexible Metallic ConduitLiDARLight Detecting and Ranging
IMP       Incident Management Plan         IPMVP       International Performance Measurement and Verification Protocol         IAQ       Indoor Air Quality         ISO       International Organization for Standardization         LAN       Local Area Network         LCD       Liquid Crystal Display         LED       Light Emitting Diode         LEED       Leadership in Energy & Environmental Design         LEED NC       LEED New Construction         LFMC       Liquidtight Flexible Metallic Conduit         LiDAR       Light Detecting and Ranging
IPMVP       International Performance Measurement and Verification Protocol         IAQ       Indoor Air Quality         ISO       International Organization for Standardization         LAN       Local Area Network         LCD       Liquid Crystal Display         LED       Light Emitting Diode         LEED       Leadership in Energy & Environmental Design         LEED NC       LEED New Construction         LFMC       Liquidtight Flexible Metallic Conduit         LiDAR       Light Detecting and Ranging
IAQ       Indoor Air Quality         ISO       International Organization for Standardization         LAN       Local Area Network         LCD       Liquid Crystal Display         LED       Light Emitting Diode         LEED       Leadership in Energy & Environmental Design         LEED NC       LEED New Construction         LFMC       Liquidtight Flexible Metallic Conduit         LiDAR       Light Detecting and Ranging
ISO       International Organization for Standardization         LAN       Local Area Network         LCD       Liquid Crystal Display         LED       Light Emitting Diode         LEED       Leadership in Energy & Environmental Design         LEED NC       LEED New Construction         LFMC       Liquidtight Flexible Metallic Conduit         LiDAR       Light Detecting and Ranging
LANLocal Area NetworkLCDLiquid Crystal DisplayLEDLight Emitting DiodeLEEDLeadership in Energy & Environmental DesignLEED NCLEED New ConstructionLFMCLiquidtight Flexible Metallic ConduitLiDARLight Detecting and Ranging
LED       Light Emitting Diode         LEED       Leadership in Energy & Environmental Design         LEED NC       LEED New Construction         LFMC       Liquidtight Flexible Metallic Conduit         LiDAR       Light Detecting and Ranging
LED       Light Emitting Diode         LEED       Leadership in Energy & Environmental Design         LEED NC       LEED New Construction         LFMC       Liquidtight Flexible Metallic Conduit         LiDAR       Light Detecting and Ranging
LEED         Leadership in Energy & Environmental Design           LEED NC         LEED New Construction           LFMC         Liquidtight Flexible Metallic Conduit           LiDAR         Light Detecting and Ranging
LEED NC         LEED New Construction           LFMC         Liquidtight Flexible Metallic Conduit           LiDAR         Light Detecting and Ranging
LiDAR Light Detecting and Ranging
LRFD Load and Resistance Factor Design
LRU Lowest Replaceable Unit
LTA Lenders' Technical Advisor
MAC Media Access Control
MaxAP Maximum Availability Payment
MDE Maximum Design Earthquake
MDF Main Distribution Frame
MEP Mechanical, Electrical and Plumbing
MMIS Maintenance Management Information System
MMRP Mitigation Monitoring and Reporting Program
MOT Maintenance of Traffic
MOU Memorandum of Understanding
MPH Miles Per Hour
MSD Major Service Degradation
MSDS Material Safety Data Sheets
MSI Master Systems Integrator
MTBHE Mean Time Between Hazard Events

MUTCD	Manual on Uniform Traffic Control Devices
MW	Megawatt
NACE	National Association of Corrosion Engineers
NCE	Noncompliance Event
NCEER	National Center for Earthquake Engineering Research
NCHRP	National Cooperative Highway Research Program
NCR	Non-Conformance Report
NEC	National Electrical Code
NEMA	National Electrical Manufacturers Association
NEPA	National Environmental Policy Act of 1969, 42 U.S.C. §§ 4321 et seq., as it may
	be amended
NESC	National Electrical Safety Code
NETA	International Electrical Testing Association
NFPA	National Fire Protection Association
NIST	National Institute of Standards and Technology
NOI	Notice of Intent
NPDES	National Pollutant Discharge Elimination System
NRC	Noise Reduction Coefficient
NRCS	Natural Resource Conservation Service
NRHP	National Register of Historic Places
NSC	National Safety Council
NVR	Network Video Recorder
OA	Other Adjustments
O&SHA	Operating and Support Hazards Analysis
O/S-SSPP	Operating System Safety Program Plan
000	Operations Control Center
OCR	Optical Character Recognition
OCS	Overhead Catenary System
ODE	Operating Design Earthquake
OEM	Original Equipment Manufacturer
OH	Overhead
OSHA	Occupational Safety and Health Administration
PA	Public Address
PAB	Private Activity Bond
PABX	Private Automatic Branch Exchange
PCB	Polychlorinated Biphenyl
PCI	Precast/Prestressed Concrete Institute
PDF	Portable Document Format
PDM	Precedence Diagram Method
PDS	Power Distribution System
PHA	Preliminary Hazards Analysis
PICO	Post-Installation Checkout
PM	Project Management
PMP	Project Management Plan
PNC	Potrero Neighborhood Collective LLC
PoE	Power over Ethernet
POH	Point of Handoff
PPC	Principal Project Company
PCQP	Principal Project Company's Construction Quality Plan
PDQP	Principal Project Company's Design Quality Plan
PQP	Principal Project Company's Quality Program
PPCECP	Principal Project Company's Environmental Compliance Plan

	Dringing Draiget Companyle Quality Dreaman Dian
PPCQPP	Principal Project Company's Quality Program Plan
PPCSE	Principal Project Company's Safety Engineer
PPCSHP	Principal Project Company Safety & Health Plan
PPCSS	Principal Project Company's Safety Supervisor
PPCSSSP	Principal Project Company Site Specific Safety Plan
PQM	Program Quality Manager
PRA	California Public Records Act
P-S	P-S Suspension Loggings
PSTN	Public Switched Telephone Network
PTZ	Pan, Tilt, Zoom
PVC	Polyvinyl Chloride
QA	Quality Assurance
QC	Quality Control
QM	Quality Management
QMP	Quality Management Plan
QMS	Quality Management System
QoS	Quality of Service
QPM	Quality Program Manager
QSD	Qualified SWPPP Developer
QT	Qualification Test
RAMS	Reliability, Availability, Maintainability, and Safety/Security/Service
RDE	Restorable Design Earthquake
RFC	Request for Comments
RFCCD	Release for Conformed Construction Documents
RFCD	Release for Construction Documents
RFP	Request for Proposals
RGS	Rigid Galvanized Steel
RTLS	Real-Time Location System
SBE	Small Business Enterprise
SC	Safety Certification
SC Plan	Safety Certification Plan
SCADA	Supervisory Control and Data Acquisition
SCI	System Condition Index
SCS	Structured Cabling System
SCVM	System Compliance Verification Matrix
SCVR	Safety Certification Verification Report
SF	Square Feet
SF6	Sulfur Hexafluoride
SFMTA	San Francisco Municipal Transportation Agency
SHA	System Hazards Analysis
SHPO	State Historic Preservation Officer
SLP	State Histolic Preservation Onicel Service Loss Percentage,
SMARTS	Service Loss Percentage, Stormwater Multi-Application and Report Tracking System
SNMP	Simple Network Management Protocol
SOP	
	Standard Operating Procedure Principal Project Company's Statement of Qualification
SOQ	Principal Project Company's Statement of Qualification
SPCCP	Spill Prevention Control and Countermeasures Plan
SPFMA	System Performance and Failure Management Analysis
SQAP	Software Quality Assurance Plan
SSEPP	System Security Emergency Preparedness Plan
SSHA	System Subsystem Hazards Analysis
SSI	Sensitive Security Information

SSOState Safety OversightSSPSystem Security PlanSSPPSystem Safety Program PlanSSPWCStandard Specifications for Public WorSSRCSafety and Security Review Committee	
SSPPSystem Safety Program PlanSSPWCStandard Specifications for Public Wor	
SSPWC Standard Specifications for Public Wor	
SSRC Safety and Security Review Committee	
	9
SUE Subsurface Utility Engineering	
SUSMP/LID BMP Standard Urban Stormwater Mitigation	Plan/Low Impact Development Best
Management Practices	
SWP Safe Work Plan	
SWPPP Storm Water Pollution Prevention Plan	
SWRCB         State Water Resources Control Board	
TIA Time Impact Analysis	
TMP Transportation Management Plan	
TR Technical Requirements	
TRB Transportation Research Board	
TRO Temporary Restraining Order	
TSC Traffic Signal Coordination	
TVC Terminal Vertical Core	
U.S.C. United States Code	
UL Underwriters Laboratory	
UPS Uninterruptable Power Supply	
USDOT United States Department of Transport	ation, or its successor entity
USO United Service Organizations Inc.	
VAR Volt-Amp Resistance	
VLAN Virtual Local Area Network	
VMS Variable Message Sign	
VoIP Voice over Internet Protocol	
VRLA Valve-Regulated, Lead Acid	
VSS Video Surveillance System	
VT Verification Test	
WAN Wide Area Network	
WBS Work Breakdown Structure	

<u>1 Year Scheduled Maintenance Plan</u> means detailed schedule, which highlights IFM Services activities that may impact SFMTA O&M Services, identifies implications and provides plans on how the potential impact will be mitigated or eliminated, prepared and submitted by Principal Project Company in accordance with <u>Section C.6.7.1</u> of the IFM Specifications, as updated and approved by the City in accordance with this Agreement.

24/7/365(6) means 24 hours per day 7 days per week, 365(6) days per year.

<u>5 Year Scheduled Maintenance Plan</u> means detailed schedule, which identifies all major IFM Services activities that may impact SFMTA O&M Services, prepared and submitted by Principal Project Company in accordance with <u>Section C.6.7.2</u> of the IFM Specifications, as updated and approved by the City in accordance with this Agreement.

Access Date has the meaning set forth in Section 2.2 (Right of Entry).

Accessibility Condition means a state or condition of the relevant Functional Unit or the means of access to it which allows all persons who are entitled to enter, occupy or use the

relevant Functional Unit to enter and leave the Functional Unit safely and conveniently and using normal access routes.

<u>Account Balances</u> means all amounts standing to the credit of any bank account held by or on behalf of Principal Project Company (excluding the Handback Requirements Reserve Account), or the value of any letter of credit issued in lieu of any bank account held or required to be held by or on behalf of Principal Project Company (excluding the Handback Requirements Reserve Account), at the Early Termination Date.

Actual Insurance Policy(ies) has the meaning set forth in Section 10.1.3(b)(ii).

Actual Insurance Premiums has the meaning set forth in Section 10.1.3(e)(i).

<u>Actual Knowledge</u> means (i) as to Principal Project Company, facts and information actually known to Principal Project Company or Principal Project Company's Authorized Representative (in each case, as applicable), after reasonable consultation with other personnel of each PPC-Related Entity that are involved with the performance of the Work or this Agreement, as applicable; and (ii) as to City, City's Authorized Representative, facts and information actually known to City's Authorized Representative.

Additional Property(ies) has the meaning set forth in Section 7.5.1.2.

<u>Adhoc Services</u> means services which, in accordance with the IFM Specifications, City is entitled to require Principal Project Company to provide but where Principal Project Company's obligation to provide those services does not arise unless and until it is requested to do so by City.

Adjusted Annual Energy Target or AAET means the Annual Energy Target in any Contract Year after adjusting pursuit to Section 3.2.1 of Exhibit 12.

# Adverse Event means:

(a) an Affordability Event, with the understanding that an Affordability Event occurring before the Effective Date shall be deemed to occur the day after the Effective Date;

(b) either or both of the CEQA Approval or the NEPA Approval is invalidated for a reason other than PPC Fault;

(c) a temporary restraining order, injunction or other form of legal order by a court prohibiting City or Principal Project Company from performing any of their respective material obligations under this Agreement or materially delaying the critical path;

(d) the occurrence of exceptional circumstances in the financial markets in one or more of Europe, the United States of America, Japan/Asia Pacific and Canada that, in City's opinion determined in City's reasonable discretion, (i) results in material and substantial cessation of lending activity in national or relevant international capital or interbank markets and (ii) adversely affects access by Principal Project Company to such markets preventing Financial Close by the Scheduled Financial Close Date; (e) any action, litigation or proceeding pending against City or affecting the Project, which, in each case, has a material likelihood of success and if determined adversely would have the effect of (i) preventing Financial Close; or (ii) prohibiting or materially impairing City from performing any of its material obligations under the Principal Project Documents;

(f) a downgrade to City's credit rating by a Rating Agency that has a material negative impact on the credit rating of any bond financing included in the Finance Plan when compared to the indicative investment grade rating(s) of such bonds included in the Finance Plan as of the Effective Date;

(g) the approval of this Agreement by the SFMTA Board or the Board of Supervisors is invalidated for any reason other than PPC Fault;

(h) failure of City to timely provide the deliverables set forth in <u>Section 3.2.2</u> (City Deliverables);

(i) any event set forth in <u>clauses (b)</u>, <u>(e)-(f)</u>, <u>(h)</u>, <u>(l)-(m)</u>, <u>(o)</u>, and <u>(s)-(t)</u> of the definition of Compensable Delay Event; or

(j) any event set forth in <u>clauses (a)-(h)</u> of the definition of "Force Majeure Event".

Adverse Event Notice has the meaning set forth in Section 3.4.2.

<u>Adverse Weather</u> means heavy rain, windstorm, flood, or any severe atmospheric condition that: (i) occurs at the Project Site during the D&C Period; and (ii) prevents Principal Project Company from proceeding with at least 75 percent of the scheduled labor, material, and equipment resources for at least five hours per Day on activities shown in the Critical Path on the Project Schedule).

<u>Adverse Weather Event</u> means Adverse Weather that exceeds the anticipated number of Days of Adverse Weather specified per year in <u>Section 14.1.8</u> (Relief for Adverse Weather Event).

### Affiliate means:

- (a) any Equity Member;
- (b) any Person which directly or indirectly through one or more intermediaries controls, or is controlled by, or is under common control with, Principal Project Company or any Equity Member; and
- (c) any Person owned in whole or in part by (i) Principal Project Company, (ii) any Equity Member or (iii) any Affiliate of Principal Project Company under <u>clause (b)</u> of this definition, whether the ownership interest is direct or indirect, beneficial or of record, provided that ownership of less than 10% of the equity interest in a Person shall not give rise to Affiliate status.

For purposes of this definition, the term "control" means the possession, directly or indirectly, of the power to cause the direction of the management of a Person, whether through voting rights or securities, by contract, family relationship or otherwise.

<u>Affordability Event</u> means the occurrence of a fluctuation in the Base Interest Rates and/or Baseline Credit Spreads during the Bank Debt Rate Protection Period or Bond Rate Protection Period that results, or in the good faith opinion of City is likely to result, in an increase to the Base Capital MaxAP of 10% or greater.

<u>Aggregate Actual Consumption</u> means the actual consumption of all Energy at the Infrastructure Facility, as invoiced by the relevant Utility companies for each Contract Year.

**<u>Agreement</u>** means the Design-Build-Finance-Operate-Maintain Agreement to which this <u>Exhibit 1</u> (Abbreviations and Definitions) is attached, including all exhibits, appendices and attachments, as such agreement may be modified from time to time.

**<u>Airspace</u>** means any and all real property, including the surface of the ground, within the vertical column extending above and below the surface boundaries of the Project Site and not necessary or required for the Infrastructure Facility (including Upgrades) or developing, permitting, designing, financing, constructing, installing, equipping, operating, maintaining, repairing, reconstructing, restoring, rehabilitating, renewing or replacing the Infrastructure Facility (including Upgrades) or Principal Project Company's timely fulfillment of its obligations under the Contract Documents. If the Project Site is a separate legal parcel with specific vertical limits, then the Airspace shall only mean the area within those vertical limits.

<u>Allowance</u> means the Office/Admin and Training Spaces FF&E Allowance, the City-Furnished IT/Comms Allowance and the Partnering Allowance, and <u>Allowances</u> means all of them as the context requires.

<u>Annual Energy Target</u> or <u>AET</u> means the targeted consumption of Energy in any Contract Year in respect of the Regulated Load, and, for clarity, does not include Non-Regulated Load consumption. Metered, Non-Regulated Load consumption shall be included for reference only.

Annual Energy Review Meeting has the meaning set forth in Section 4.3 of Exhibit 12.

<u>Annual Service Plan</u> means the plan prepared and submitted by Principal Project Company that provides, at a minimum, the required items listed in <u>Section B.5.1</u> of the IFM Specifications, as updated and approved by the City in accordance with this Agreement.

Annual Vandalism Deductible has the meaning set forth in Section 8.3.4.

<u>Applicable Law and Standards</u> means applicable Law, all applicable Standards and Specifications, manufacturers' recommended maintenance and operations activities and maintenance and operations activities that would normally be undertaken in accordance with Good Industry Practices.

<u>As-Built Documents</u> means, collectively, the documents referred to in <u>Section 1.6.3</u> of <u>Division</u> <u>1</u> of the Technical Requirements, as well as the following items referred to in <u>Division 1</u> of the Technical Requirements: As-Built Schedule, As-Built Drawings, and LOD 500 As-Built Model (<u>Section 1.10.1.2</u>).

<u>As-Built Drawings</u> means, revised sets of drawings submitted to reflect any changes made during the construction process, depicting the actual conditions, dimensions, and locations of installed elements as opposed to the originally planned design.

<u>As-Built Schedule means has the meaning set forth in Section 1.2.5</u> of <u>Division 1</u> of the technical Requirements.

<u>As-Built Schedule Analysis</u> has the meaning set forth in <u>Section 18.3.5</u> (Additional Requirements for Contract Disputes Relating to Delay Events Impacting the Critical Path).

<u>Authority(ies) Having Jurisdiction</u> means an organization, including City acting in its regulatory capacity, office or individual responsible for enforcing the requirements of a code or standard, or for approving equipment, materials, an installation, or a procedure. The term "Authorities Having Jurisdiction" refers to more than one such organization, office or individual.

<u>Authorized Representative</u> means the authorized representative for either Party identified in <u>Exhibit 10</u> (Initial Designation of Authorized Representatives) or otherwise designated in accordance with this Agreement.

<u>Availability Condition</u> means any of (i) the Accessibility Condition, (ii) the Safety Condition or (iii) the Use Condition.

<u>Availability Deduction</u> means a Deduction from an Availability Payment as a result of Functional Unit or Functional Component being deemed Unavailable.

<u>Availability Failure</u> means an IFM Failure Event which has not been Rectified within the relevant Rectification Time and which causes a Functional Unit to be Unavailable. For the avoidance of doubt, an Availability Failure may be applied to one or more Functional Components.

<u>Availability Failure Deduction</u> means a deduction from the Availability Payment as a result of an Availability Failure.

<u>Availability Payment</u> means the payments to be made by City to Principal Project Company under <u>Section 11.2</u> (Availability Payments), determined on an annual basis in accordance with <u>Exhibit 4B</u> (Availability Payment Mechanism) and payable quarterly.

**Bank Debt** means any debt financing, other than Bond Financing, provided by a bank or similar financial institution.

Bank Debt Pricing Date means with respect to any Bank Debt proposed in the Finance Plan, the earliest to occur of:

- (a) the date of Financial Close;
- (b) the date at which the Bank Debt is fixed or hedged by Principal Project Company; and
- (c) such other date as is mutually agreed to by Principal Project Company and City.

**Bank Debt Rate Protection Period** means the period beginning at 3:00 p.m. Pacific Time on [•], 2025 and ending on, with respect to any Bank Debt, the Bank Debt Pricing Date.

**Bankruptcy Event** means any of the following events:

(a) a receiver, receiver manager or other encumbrance holder taking possession of or being appointed over, or any distress, execution or other process being levied or enforced upon, the whole or any material part of the assets of an entity;

(b) any proceedings with respect to the entity being commenced under the Bankruptcy Law and if such proceedings are commenced against and are disputed by the entity, such proceedings are not discontinued, withdrawn, dismissed or otherwise remedied within 60 days of such proceedings being instituted;

(c) the entity making an assignment for the benefit of its creditors, being declared bankrupt or committing an act of bankruptcy, becoming insolvent, making a proposal or otherwise taking advantage of provisions for relief under the Bankruptcy Law or similar legislation in any jurisdiction, or any other type of insolvency proceedings being commenced by or against the entity under the Bankruptcy Law or otherwise and, if proceedings are commenced against the entity and are disputed by the entity, such proceedings are not stayed, dismissed or otherwise remedied within 60 days of such proceedings being instituted;

(d) in any voluntary or involuntary case seeking liquidation, reorganization or other relief with respect to the entity or its debts under any bankruptcy Law or foreign bankruptcy, insolvency or other similar law now or hereafter in effect, this Agreement or any of the other Contract Documents is rejected, including a rejection under 11 U.S.C. § 365 or any successor statute; or

(e) the entity ceasing to carry on business.

**Bankruptcy Law** means the United States Bankruptcy Code, 11 U.S.C. §§ 101 *et seq.* as amended from time to time and any successor statute thereto. "Bankruptcy Law" also includes any similar state law relating to bankruptcy, insolvency, the rights and remedies of creditors, the appointment of receivers or the liquidation of companies and estates that are unable to pay their debts when due.

**Base Capital MaxAP** means, initially, **[\$•]**, which is the Base Capital MaxAP reflected in the Financial Model for the first Contract Year, as adjusted from time to time in accordance with this Agreement.

**Base Case Financial Model** means the Financial Model as approved by the Parties as of the Effective Date.

**Base IFM MaxAP** means, initially, **[\$•]**, which is the Base IFM MaxAP reflected in the Financial Model for the first Contract Year, as adjusted from time to time in accordance with this Agreement.

**<u>Base Interest Rate</u>** means the publicly-documented interest rates of each maturity included in the following indices:

- (a) [●];
- (b) [●];
- (C) [●].

The Base Interest Rates do not include any additional credit spread, margin or fee components.

**Baseline Credit Spreads** means the set of credit spreads issued by City that assumes different coupons for the range of maturities, ratings and types of Bond Financings that will serve as the basis for the Credit Spread Risk Mitigation.

**Basis of Design Report (BODR)** means the report described in <u>Section 1.8.5.2.3</u> of <u>Division 1</u> of the Technical Requirements, concerning design methodology and approach for the D&C Work, key assumptions and operations and maintenance design methodology.

**<u>Bedding-In Period</u>** means the three-month period following the Substantial Completion Date.

**Benchmarking Date** has the meaning set forth in <u>Section 10.1.3(b)</u>.

**Best Management Practices (BMP)** has the meaning set forth in Storm Water Management For Construction Activities: Developing Pollution Prevention Plans and Best Management Practices (EPA Document 832 R 92-005).

**<u>BIM Project Execution Plan</u>** means Principal Project Company's plan describing BIM-enabled workflows and systems to successfully deliver, design, construct, operate, and maintain the Project, prepared and submitted by Principal Project Company in accordance with <u>Section</u> <u>1.10.2</u> of <u>Division 1</u> of the Technical Requirements, as updated and approved by the City in accordance with this Agreement.

**Board of Supervisors** means the Board of Supervisors of the City and County of San Francisco.

**Bond Financing** means any debt financing comprising of bonds, which includes tax-exempt bonds issued by a Conduit Issuer or taxable capital market instruments.

**Bond Pricing Date** means with respect to any Bond Financing proposed in the Finance Plan, the earliest to occur of (a) the date of Financial Close, (b) the date at which any Bond Financing is priced, and (c) the date at which any Bond Financing is fixed or hedged.

**Bond Rate Protection Period** means the period beginning at 3:00 p.m. Pacific Time on [•], 2025 and ending on, with respect to any Bond Financing, the Bond Pricing Date.

**Books and Records** means any and all documents, books, records, papers, or other information of any PPC-Related Entity or Affiliate relating to the Project, the Work, the PPC-Furnished FF&E and the Project Site, including (a) all design and construction documents, and all operations and maintenance documents (including drawings, specifications, Submittals, Contracts, invoices, schedules, meeting minutes, budgets, forecasts, requests for change proposals, change responses, change requests, plans (including the Project Management Plan), reports and manuals), (b) daily time sheets and supervisor's daily reports, union agreements, insurance, welfare and benefits records, payroll registers, earning records, payroll tax forms, invoices and requisitions, equipment records, payment certificates, cancelled checks, job cost reports, job payroll ledges, general ledger, cash disbursement journal; (c) income statements, balance sheets, statements of cash flow and changes in financial position, details regarding operating income, expenses, capital expenditures and budgeted operating results; (d) all budgets, certificates, claims, correspondence, data (including test data), documents, expert

analyses, facts, files, information, investigations, materials, notices, plans, projections, proposals, records, reports, requests, samples, schedules, settlements, statements, studies, surveys, tests, test results, vehicular traffic information, operational information analyzed, categorized, characterized, created, collected, generated, maintained, processed, produced, prepared, provided, recorded, stored or used by Principal Project Company or its Contractors or any of their representatives in connection with the Project, (e) the Base Case Financial Model, Financial Model, subsequent Financial Model Updates and Financial Modeling Data, (f) records of the expenditure and investment of the Milestone Payment including records identifying expenditures for D&C Work and debt service payments, and (q) with respect to all of the above. any information that is stored electronically or on computer-related media (in its original source and not converted to PDF or other format). For purposes of the requirements of the Contract Documents to maintain Books and Records, the term "Books and Records" includes documents or information that are subject to the attorney-client privilege, but for purposes of requirements of the Contract Documents to provide access to Books and Records, the term specifically excludes the disclosure by any Party of Books and Records that are protected by the attorneyclient or other legal privilege based upon an opinion of counsel reasonably satisfactory to the other Party.

**Breakage Costs** means any commercially reasonable prepayment premiums or penalties (including documented SOFR breakage fees, customary and reasonable trustee, Collateral Agent and Lender fees but excluding any fees related to legal or other consulting costs), make-whole payments or other prepayment amounts, including costs of early termination of interest rate and inflation rate hedging, swap, collar or cap arrangements, payable by or on behalf of, or credited against payments owing to, Principal Project Company, under any Financing Agreement or Security Document or otherwise as a result of the payment (including pre-payment), redemption or acceleration of all or any portion of the principal amount of Project Debt prior to its scheduled payment date (less any breakage benefits), excluding, however, any such amounts included in the principal amount of any Refinancing. The term "Breakage Costs" excludes any such premiums, penalties, payments or other amounts relating to Equity Member Debt.

**Building Automation System** or **BAS** means the computer-based control system installed in buildings that controls, monitors and integrates the buildings' systems, including HVAC, lighting, security, fire alarm and elevator control. The BAS controls building environmental conditions including temperature, humidity, CO2, illumination, heating and cooling and air flow distribution.

**Building Occupant(s)** means City Personnel, City Parties, Visitors, PPC-Related Entities and third party members of the public lawfully present on or using the Infrastructure Facility including, or each of them as the context requires.

**Buildings Cost Index (BCI)** means the "Buildings Cost Index" in San Francisco, as published by Engineering News-Record. If the BCI is discontinued or substantially altered, a suitable replacement will be determined by the Parties in accordance with general market practice at the time.

**Bus Yard Component** means the Facility's transit component, which (a) will include the spaces needed for City's operation and maintenance activities at the Facility after Substantial Completion of the Infrastructure Facility, (b) must meet the Bus Yard Component criteria in the Technical Requirements except as otherwise approved by City in writing, which approval shall be at its sole discretion, and (c) is not Common Infrastructure. The Bus Yard Component is

generally described in the "Project Description" included in <u>Division 1</u> of the Technical Requirements.

**Business Day** means any weekday (i.e., Monday through Friday) except for those weekdays on which (a) City is officially closed for business or (b) banks are not required or authorized by Law to be open in the State.

Business Opportunity(ies) has the meaning set forth in Section 22.2.1(b).

<u>CAFM (Computerized Aided Facility Management) System</u> means, the computerized system used to track, record, manage and communicate day-to-day Infrastructure Facility operations and to support long-term planning. Modules may include asset, location, inventory, work order/request, Scheduled Maintenance, personnel and reports.

<u>Capacity Improvement</u> means any Project expansion, improvement, measure or procedure that both (a) maintains or increases the throughput capacity of the Project or any portion thereof and (b) improves the level of service of the Project.

**CEQA Approval** means the Final Environmental Impact Report for the Potrero Yard Modernization Project at 2500 Mariposa Street (Planning Department Case No. 2019-021884ENV), certified by the Planning Commission on January 11, 2024 and available at https://sfplanning.org/environmental-reviewdocuments?title=Potrero+Yard+Modernization+Project.

#### CEQA Event means:

- (a) any new or modified CEQA Approval necessitated solely by a City Change, a Delay Event or Relief Event;
- (b) legal action being taken in respect of the CEQA Approval that results in a temporary restraining order, preliminary injunction or other form of interlocutory relief by a court of competent jurisdiction that prohibits prosecution of, or by complying with such temporary restraining order, preliminary injunction or other form of interlocutory relief by a court of competent jurisdiction results in prohibiting the prosecution of, a material portion of the Work;
- (c) review or revocation or material change to, the CEQA Approval; or
- (d) any review or revocation of, or change to, a CEQA Approval directly resulting from the circumstances specified in <u>clauses (b)</u> and <u>(c)</u>,

except, in each case, to the extent resulting, in whole or in part, from Principal Project Company's design, Work or from any PPC Fault.

<u>Certificate of Final Acceptance</u> means the certificate issued by City as contemplated in <u>Section 7.9</u> (Final Acceptance).

<u>Certificate of Substantial Completion</u> means the certificate issued by City as contemplated in <u>Section 7.8</u> (Substantial Completion).

<u>Change</u> means any acceleration, addition, decrease, omission, deletion, removal or modification from or to the Work.

#### Change in Law means:

- (a) any repeal (in whole or in part) of, or amendment or modification to, any applicable Law by, any Governmental Entity or any written change in interpretation or application of, any applicable Law, in each case, after the Setting Date; and
- (b) the adoption or enactment of any new applicable Law by any Governmental Entity after the Setting Date,

which, in either case, is materially inconsistent with any existing applicable Law or any existing interpretation or application of, any such applicable Law in effect prior to the Setting Date; but excluding, (i) any repeal of, or amendment or modification to, a written change in interpretation or application of, and applicable Law, or any new applicable Law, in each case, that is pending, passed or adopted but not yet effective as of the Setting Date, (ii) any repeal of, or amendment or modification to, or written change in interpretation or application of, or the adoption or enactment of, state or federal tax laws of general application, and (iii) any repeal of, or amendment or modification to, or written change in interpretation or application of, or adoption or enactment of, state labor laws.

**Change of Control** means any Equity Transfer, transfer of an interest, direct or indirect, in an Equity Member, or other assignment, sale, financing, grant of security interest, hypothecation, conveyance, transfer of interest or transaction of any type or description, including by or through voting securities, asset transfer, contract, merger, acquisition, succession, dissolution, liquidation, bankruptcy or otherwise, that results, directly or indirectly, in a change in possession of the power to direct or control or cause the direction or control of the management of Principal Project Company or a material aspect of its business. A change in possession of the power to direct or control of Principal Project Company if such Equity Member may constitute a Change of Control of Principal Project Company if such Equity Member possesses, immediately prior to such Change of Control, the power to direct or control or cause the direction or control of the management of an Equity Member possesses, immediately prior to such Change of Control, the power to direct or control or cause the direction or control of the management of Principal Project Company. Notwithstanding the foregoing, the following shall not constitute a Change of Control:

- (a) a change in possession of the power to direct or control the management of Principal Project Company or a material aspect of its business due solely to bona fide open market transactions in securities effected on a recognized public stock exchange, including such transactions involving an initial public offering;
- (b) an upstream reorganization or transfer of indirect interests in Principal Project Company (including, for greater certainty, in connection with the issuance or redemption of interests pursuant to any employee ownership program) so long as no change occurs in the entity with ultimate power to direct or control or cause the direction or control of the management of Principal Project Company;
- (c) a change in possession of the power to direct or control the management of Principal Project Company or a material aspect of its business due solely to a bona fide transaction involving a beneficial interest in the ultimate parent

organization of an Equity Member (but not if the Equity Member is the ultimate parent organization), provided, however, that this exception shall not apply if at the time of the transaction the transferee is suspended or debarred from bidding, proposing or contracting with the City or any federal or State department or agency, or is subject to a suspension or debarment proceeding;

- (d) an Equity Transfer, where the transferring Equity Member and the transferee are under the same ultimate parent organization ownership, management and control before and after the transfer;
- (e) a transfer of interests (i) between managed funds that are under common ownership, management or control or (ii) by an Equity Member to a fund, investment vehicle or other entity managed by or under common control of such Equity Member, except, in each case, a change in the management or control of a fund, investment vehicle or other entity, as applicable, that manages or controls;
- (f) the exercise of minority veto or voting rights (whether pursuant to applicable Law, by Principal Project Company's organizational documents or by related member or shareholder agreements or similar agreements) over major business decisions of Principal Project Company, provided that if such minority veto or voting rights are exercised pursuant to shareholder or similar agreements, City received copies of such agreements on or before the date of this Agreement; and
- (g) the grant of Security Documents, including the Initial Security Documents, in compliance with the Direct Agreement or the exercise of lender remedies under such Security Documents, including foreclosure.

<u>Change Order</u> has the meaning set forth in <u>Section 1.4.3</u> of <u>Exhibit 9</u> (Change Procedures).

<u>Change Proposal</u> has the meaning set forth in <u>Section 1.3.1</u> of <u>Exhibit 9</u> (Change Procedures).

<u>City</u> has the meaning set forth in the <u>Preamble</u>.

<u>City Access Period</u> has the meaning set forth in <u>Section 7.13.5.1</u>.

<u>City Additional Insured(s)</u> means any (and all) of the City and County of San Francisco, and its respective successors, assigns, officeholders, officers, directors, agents, representatives, consultants and employees.

<u>**City-Caused Delay Event**</u> means any event falling under <u>clauses (a)-(e)</u>, (k), (n), (q), (r) and (u)-(v) of the definition of Compensable Delay Event.

<u>City-Caused Relief Event</u> means any event falling under <u>clauses (a)-(c)</u>, (g), (j), (l) and (r) of the definition of Compensable Relief Event.

<u>City Change</u> has the meaning set forth in <u>Section 12.2.1</u>.

<u>City Data</u> means any information, data, or document, whether or not protectable Intellectual Property, which is created, developed, or collected by, or on behalf of, City related to

transportation operations, national infrastructure planning and personal information of the City employees, vendors and consumers. For the avoidance of doubt, City Data shall include, but not be limited to, (a) all "nonpublic information," as defined by the Gramm-Leach-Bliley Act (15 USC § 6801 et seq.), (b) personal information as defined by California Civil Code §§ 1798.29, 1798.82, and 1798.140 (California Consumer Privacy Act of 2018, effective January 1, 2020), as amended and supplemented by the California Privacy Rights Act of 2020 (effective December 16, 2020; operative January 1, 2023), (c) protected health information or individually identifiable health information as defined by the Health Insurance Portability and Accountability Act of 1996 (HIPAA) and the Health Information Technology for Economic and Clinical Health (HiTECH) Act or as defined by the Code of Federal Regulations (45 CFR § 160.103), and/or (d) personal data as defined by the EU General Data Protection Regulation (Regulation (EU) 2016/679). For the further avoidance of doubt, City Data is not limited to proprietary or confidential information, and need not constitute trade secret information.

<u>**City Default**</u> has the meaning set forth in <u>Section 16.3.1</u> (City Default).

#### City Fault means:

- (a) a breach by City of any of its obligations or any representation or warranty under this Agreement;
- (b) City's violation of any applicable Law; or
- (c) fraud, criminal conduct, intentional misconduct, recklessness, bad faith or gross negligence of the City.

<u>**City-Furnished Equipment</u>** means all FF&E classified as City-Furnished Equipment (or Cityfurnished) in the Technical Requirements (including items marked OP in the Equipment List, <u>Table 5</u> and <u>Table 6</u> of <u>Division 1</u> of the Technical Requirements) or otherwise required to be included as such under <u>Section 7.13.2</u> (Selection of City-Furnished Equipment).</u>

<u>**City-Furnished Equipment List**</u> means the list of equipment identified as being City-Furnished Equipment in the Equipment List and <u>Tables 5</u> and <u>6</u> of <u>Division 1</u> of the Technical Requirements, as updated or amended in accordance with <u>Section 7.13.2</u> (Selection of City-Furnished Equipment).

<u>City-Furnished IT/Comms Allowance</u> means the Allowance to be used pursuant to <u>Section</u> <u>11.11.3</u> (City-Furnished IT/Comms Allowance) by City for City-Furnished IT/Comms Equipment. The initial City-Furnished IT/Comms Allowance amount is identified in <u>Section 11.11.3</u>.

<u>City-Furnished IT/Comms FF&E</u> means all FF&E classified as City-Furnished IT/Comms FF&E in the Technical Requirements, all additional FF&E identified as City-furnished "IT/Comms FF&E" in <u>Table 5</u> and <u>Table 6</u> of <u>Division 1</u> of the Technical Requirements and all additional IT/Comms Equipment selected by City under <u>Section 7.13.1</u> (Selection and Procurement of PPC-Furnished FF&E).

<u>**City IP**</u> means all Intellectual Property owned by, or sufficiently licensed to, City including, without limitation, all rights, grants and interests pursuant to the Predevelopment Agreement.

<u>City Personnel</u> means every person employed, engaged, or hired by City or every person

employed, engaged, or hired by City Contractors to carry out any of the responsibilities or any of the services provided by City with respect to the Infrastructure Facility.

<u>City Ready for Move Condition</u> means the following state of condition of the Project Site:

- With respect to Utilities, all Utility infrastructure will remain in place and will not be sealed, capped, disconnected, turned off and or controlled prior to vacation of the Project Site by City;
- (b) The buildings, grounds and existing structures will remain in their current state as of the Access Date, including all furniture, equipment, materials and supplies; and
- (c) All furniture, equipment, materials and supplies that are to be moved by Principal Project Company shall be boxed or otherwise prepared so they may be moved.

<u>City Relocation Plan</u> means the plan developed by Principal Project Company pursuant to the Predevelopment Agreement, and approved by City, that describes the scope of Work to be undertaken by Principal Project Company to fully relocate the applicable City equipment, supplies and materials from the Project Site.

<u>**City Relocation Scope**</u> means the scope of Work undertaken by Principal Project Company to fully relocate the applicable City equipment, furniture, supplies and materials from the Project Site, as more particularly described in the approved City Relocation Plan.

<u>City Policies</u> means policies developed, administered and maintained by City on a specific topic that are directly applicable to the IFM Services and are required to be complied with by the Principal Project Company.

<u>Claim</u> means any claim, proceeding, action, cause of action, investigation, demand or suit (including by way of contribution or indemnity) made:

- (a) in connection with this Agreement or the Infrastructure Facility; and
- (b) at law or for specific performance, in equity, restitution, payment of money (including damages), increase in the Milestone Payment or Availability Payments and extension of time or other form of relief.

<u>Cleaning Services</u> has the meaning given in Section C.5.2 of the IFM Specifications.

<u>Climate Baseline</u> means, for each year, the CAEC will be adjusted based on the Weather Data from the Energy Performance Commencement Date to the Current Contract Year.

<u>Collateral Agent</u> means (a) the Institutional Lender listed or otherwise designated in the Security Documents as trustee or agent and authorized to act on behalf of or at the direction of the other Lenders, or (b) the Institutional Lender designated to act as trustee or agent on behalf of or at the direction of the other Lenders in an intercreditor agreement or other document executed by all Lenders to whom Security Documents are outstanding at the time of execution of such document, a copy of which shall be delivered by Principal Project Company to City. For

any Project Debt issued and held by a single Lender, Collateral Agent means such Lender. The bond trustee, if an Institutional Lender, may also be the Collateral Agent.

<u>Commercial Close</u> means the execution and delivery of this Agreement by Principal Project Company and City on the Effective Date.

<u>Commercially Reasonable Insurance Rates</u> means insurance premiums that are less than or equal to the greater of (a) rates that a reasonable and prudent risk manager for a Person seeking to insure comparable risks would conclude are justified by the risk protection afforded, and (b) 200% of the rates indicated for the period in question in the Base Case Financial Model and related Financial Modeling Data.

**<u>Committed Investment</u>** means the sum of the Equity Investments and Deferred Equity Amounts.

<u>Common Infrastructure</u> means the physical infrastructure component of the Facility that is shared by the Bus Yard Component and the Joint Development Alternative, as further described in the Technical Requirements, including <u>Division 1</u> of the Technical Requirements.

**<u>Communication Protocol</u>** means the protocol prepared and submitted by Principal Project Company and approved by the City in accordance with <u>Section B.9.1.1</u> of the IFM Specifications, as updated in accordance with this Agreement.

<u>Compensable Delay Event</u> means any of the following events or circumstances to the extent, in each case, that it directly and adversely impacts (i) the D&C Work during the D&C Period or (ii) Principal Project Company's completion of Punch List items between the Substantial Completion Date and the Final Acceptance Date:

- (a) the implementation of a City Change, excluding any Change following a Change Request;
- (b) City Fault;
- (c) failure by City to (i) provide Principal Project Company with access to the Project Site on the Access Date; or (ii) leave the Project Site in the City Ready for Move Condition as of 90 days after the Access Date;
- (d) failure by City to issue NTP 1 or NTP 2, as applicable, within 10 Business Days after full and complete satisfaction of the conditions precedent to the relevant NTP under this Agreement;
- (e) a Hazardous Materials Event caused by City;
- (f) a Relevant Change in Law;
- (g) discovery of an Unidentified Utility within the Project Site;
- (h) a CEQA Event or a NEPA Event;
- (i) compliance by Principal Project Company with an order or direction of an

Emergency service provider in an Emergency (except for Emergencies that are or arise out of Force Majeure Termination Events);

- (j) any condemnation or other taking by eminent domain of any material portion of the Project Site or the D&C Work;
- (k) except with respect to Regulatory Approvals under the jurisdiction of the City acting in its regulatory capacity, failure of City to provide responses to proposed schedules, plans, design documents, and other Submittals and matters submitted to City after the Commercial Close for which response is required under this Agreement as an express prerequisite to Principal Project Company's right to proceed or act, within the time periods indicated in this Agreement, or if no time period is indicated, within a reasonable time, taking into consideration the nature, importance and complexity of the Submittal or matter; provided, however, that the foregoing shall apply only following delivery of Notice after the expiration of the applicable time period from Principal Project Company requesting such action in accordance with the terms and requirements of this Agreement;
- discovery at, near or on the Project Site of any archeological, paleontological or cultural resources (including historic properties), excluding any such substance or resources known or disclosed to Principal Project Company as of the Setting Date (or which should have been known to Principal Project Company pursuant to a Reasonable Investigation);
- (m) discovery at, near or on the Project Site of any threatened or endangered species, excluding any such presence of species known or disclosed to Principal Project Company as of the Setting Date (or which should have been known to Principal Project Company pursuant to a Reasonable Investigation);
- (n) loss or damage to the Work directly caused by City Fault;
- (o) issuance of a temporary restraining order, preliminary injunction or other form of interlocutory relief by a court of competent jurisdiction that prohibits prosecution of, or by complying with such temporary restraining order, preliminary injunction or other form of interlocutory relief by a court of competent jurisdiction results in prohibiting the prosecution of, any material portion of the D&C Work;
- (p) discovery of:
  - actual subsurface or latent physical conditions that differ materially from the baseline subsurface conditions indicated in the Geotechnical Baseline Report, excluding any such conditions (w) that are not defined in the Geotechnical Baseline Report; (x) known or disclosed to Principal Project Company prior to the Setting Date; or (y) that could have been reasonably anticipated as potentially present by an experienced global civil works contractor based on the information contained in such Geotechnical Baseline Report; or
  - (ii) discovery of actual subsurface physical conditions within the Project Site of an unusual nature, differing materially from those ordinarily

encountered in the area and generally recognized as inherent in the type of work provided for in this Agreement, excluding any such conditions known or disclosed to Principal Project Company prior to the Setting Date;

- (q) any order of City to suspend for convenience under <u>Section 16.2.8.3</u> exceeding 24 hours in total for a single suspension or 144 cumulative hours in total across multiple suspensions;
- (r) subject to Principal Project Company complying with its obligations for coordination set forth in this Agreement, actions of the City (including its employees) or Other Contractors (other than any PPC-Related Entities, JDA-Related Entities and any entity that is not a JDA-Related-Entity but that executes an interface agreement substantially in the form of <u>Exhibit 8</u> (Form of Interface Agreement)), that materially and directly disrupts, damages or interferes with the D&C Work;
- (s) a change in Standards and Specifications materially impacting the D&C Work or the Project with which City directs Principal Project Company to comply under this Agreement;
- (t) a Hazardous Materials Event caused by Third Parties or which exists prior to the Access Date; and
- (u) issuance by City of a Safety Compliance Order for a reason other than to comply with Safety Standards, perform valid previously issued Safety Compliance Orders or correct a violation of Law or Regulatory Approval respecting health, safety or right of use and access.

<u>Compensable Relief Event</u> means any of the following events or circumstances to the extent, in each case, that it interferes directly and adversely with, or causes a failure of, the performance of the IFM after the Substantial Completion Date:

- (a) the implementation of a City Change, excluding any Change following a Change Request;
- (b) City Fault;
- (c) a Hazardous Materials Event caused by the City;
- (d) compliance by Principal Project Company with an order or direction of an Emergency service provider in an Emergency (except for Emergencies that are or arise out of Force Majeure Termination Events);
- (e) a change in Standards and Specifications during the IFM Period materially impacting the IFM or the Project with which City directs Principal Project Company to comply under this Agreement;
- (f) a Relevant Change in Law;

- (g) failure of City to provide responses to proposed schedules, plans, design documents, and other Submittals and matters submitted to City after the Effective Date for which response is required under this Agreement as an express prerequisite to Principal Project Company's right to proceed or act, within the time periods (if any) indicated in this Agreement, or if no time period is indicated, within a reasonable time, taking into consideration the nature, importance and complexity of the Submittal or matter; provided, however, that the foregoing shall apply only following delivery of Notice after the expiration of the applicable time period from the Principal Project Company requesting such action in accordance with the terms and requirements of this Agreement;
- (h) discovery at, near or on the Project Site of any archeological, paleontological or cultural resources (including historic properties), excluding any such substance or resources known or disclosed to the Principal Project Company as of the Setting Date (or which should have been known to Principal Project Company pursuant to a Reasonable Investigation);
- discovery at, near or on the Project Site of any Threatened or Endangered Species (regardless of when the species was listed as threatened or endangered), excluding any such presence of species known to the Principal Project Company as of the Setting Date (or which should have been known or disclosed to Principal Project Company pursuant to a Reasonable Investigation);
- (j) loss or damage to the Infrastructure Facility directly caused by City Fault;
- (k) issuance of a temporary restraining order, preliminary injunction or other form of interlocutory relief by a court of competent jurisdiction that prohibits prosecution of a material portion of the IFM;
- (I) subject to Principal Project Company complying with its coordination obligations under this Agreement, actions of the City (including its employees) or Other Contractors (other than any PPC-Related Entities)), except JDA-Related Entities and any entity that is not a JDA-Related-Entity but that executes an interface agreement substantially in the form of <u>Exhibit 8</u> (Form of Interface Agreement), that materially and directly disrupts, damages or interferes with the IFM Services;
- (m) Vandalism, excluding any Vandalism affecting the Infrastructure Facility building envelope or exterior grounds;
- any condemnation or other taking by eminent domain of any material portion of the Project Site or the Project;
- (o) a Hazardous Materials Event caused by Third Parties or which exists prior to the Access Date;
- (p) issuance by City of a Safety Compliance Order for a reason other than to comply with Safety Standards, perform valid previously issued Safety Compliance Orders or correct a violation of Law or Regulatory Approval respecting health, safety or right of use and access;

- (q) a permanent change in voltage by a Utility Owner supplying electricity to the Project that is greater than 10%, has a material adverse effect on the IFM Services and requires the purchase and installation of one or more new transformers by Principal Project Company; and
- (r) any order of City to suspend IFM Services for convenience under <u>Section</u> <u>16.2.8.3</u>.

**Computerized Aided Facility Management System** or **CAFM** means the computerized system used to track, record, manage and communicate day-to-day Infrastructure Facility operations and to support long-term planning. Modules may include: asset, location, inventory, work order/request, scheduled maintenance, personnel and reports.

**Condition Report** has the meaning set forth in <u>Section 8.5.4.2</u>.

**Conditions to Assistance** has the meaning set forth in <u>Section 7.6.12.2</u>.

<u>Conduit Issuer</u> means an existing agency that will issue Private Activity Bonds in connection with the Project.

**Confidential Information means**, subject to <u>Section 21</u> and without limiting applicable Law, all confidential and proprietary information that a party has designated as confidential and which is supplied, or to which access is granted, to or on behalf of the other party (whether before or after the date of this Agreement), either in writing, or in any other form, directly or indirectly pursuant to discussions with the other party and includes all analyses, compilations, studies and other documents whether prepared by or on behalf of a party which contain or otherwise reflect or are derived from such designated information.

<u>Construction Documents</u> means all shop drawings, working drawings, fabrication plans, material and hardware descriptions, specifications, construction quality control reports, construction quality assurance reports and samples necessary or desirable for construction of the Project included in the Construction Work, in accordance with the Contract Documents.

**Construction Equity Ratio** means, at any time, the ratio between:

- (a) the Committed Investments at the time; and
- (b) the sum of (i) Committed Investments at the time and (ii) the amount of Project Debt scheduled to be outstanding at the time as set forth in the Financial Model.

**Construction Management Plan** means Principal Project Company's plan describing its approach to undertake and achieve the requirements in <u>Section 1.11</u> of <u>Division</u> 1 of the Technical Requirements, prepared and submitted by Principal Project Company in accordance with <u>Section 1.1.7.2</u> of <u>Division 1</u> of the Technical Requirements, as updated and approved by the City in accordance with this Agreement.

**Construction Work** means all Work to build or construct, make, form, manufacture, furnish, install, supply, deliver, landscape, equip, test and commission or demolish any structure, building, or other improvement to real property included in the Project, but excluding:

- (a) Design Work, professional environmental services and similar services;
- (b) preparing and processing applications for Regulatory Approvals;
- (c) coordinating with adjacent property owners and Utility Owners; and
- (d) IFM Services.

**Consumer Price Index (CPI)** means the Consumer Price Index for the City of San Francisco, as published by US Bureau of Labor Statistics from time to time, or failing such publication, such other index as the Parties (acting reasonably) may agree, or as may be determined in accordance with **Error! Reference source not found.** (Contract Dispute Procedures), which most closely resembles such index.

<u>Contract</u> means any agreement, and any modification of such agreement, by Principal Project Company with any Person to perform any part of the Work or provide any materials, equipment or supplies for any part of the Work, or any such agreement, supplement or amendment at a lower tier, between a Contractor and its lower tier Contractor or a Supplier and its lower tier Supplier, at all tiers. The term "Contract" does not include agreements with Utility Owners.

<u>Contract Deadline</u> means the Substantial Completion Deadline, the Final Acceptance Deadline or the Long Stop Date, as the context may require, and "Contract Deadlines" means the Substantial Completion Deadline, the Final Acceptance Deadline and the Long Stop Date, collectively.

**Contract Dispute(s)** means any dispute, disagreement or controversy between City and Principal Project Company concerning their respective rights and obligations under the Contract Documents, including concerning any Claim, alleged breach or failure to perform and remedies.

<u>Contract Dispute Procedures</u> means the procedures for resolving Contract Disputes set forth in <u>Error! Reference source not found.</u> (Contract Dispute Procedures).

<u>Contract Documents</u> means this Agreement and all its exhibits including the Technical Requirements and other documents identified in <u>Section 1.2</u> (Contract Documents; Rules to Reconcile Conflicting Provisions).

<u>Contract Month</u> means a calendar month, except with respect to the first Contract Month, which runs from the Payment Commencement Date until the end of the calendar month in which the Payment Commencement Date falls, and the last Contract Month, which runs from the first day of the calendar month in which the Termination Date falls until the Termination Date.

<u>Contract Quarter</u> means a Quarter, except with respect to the first Contract Quarter, which runs from the Payment Commencement Date until the end of the calendar quarter in which the Payment Commencement Date falls, and the last Contract Quarter, which runs from the first day of the Quarter in which the Termination Date falls until the Termination Date.

<u>Contract Year</u> means a calendar year, provided that:

(i) the first Contract Year shall be such period that commences on the Payment Commencement Date and ends on the next ensuing December 31<sup>st</sup>; and

(ii) the final Contract Year shall be such period that commences on the January 1st that precedes the date on which this Agreement expires or is terminated, for whatever reason, and ends on the expiry or termination of this Agreement.

Any computation made on the basis of a Contract Year shall be adjusted on a pro rata basis to take into account any Contract Year of less than 365 or 366 days, whichever is applicable.

<u>Contractor</u> means any Person or entity with whom Principal Project Company has entered into any Contract to perform any part of the Work or provide any materials, equipment or supplies for the Project on behalf of Principal Project Company, and any other Person or entity with whom any Contractor has further subcontracted any part of the Work, at all tiers.

<u>Controlling Affiliate</u> means any Person which directly, or indirectly through one or more intermediaries, controls a majority of the voting shares of Principal Project Company, or controls the election of a majority of the board of directors, trustees or other persons exercising similar functions for Principal Project Company. For purposes of this definition, the term "control" means the possession, directly or indirectly, of the power to cause the direction of the management of a Person, whether through voting rights or securities, by contract, family relationship or otherwise.

<u>Controlling Work Item</u> means the activity or work item on the Critical Path of the D&C Work having the least amount of total float.

Core Hours means the period:

(a) for Maintenance and Transit Spaces, commencing at 4am and ending at 10pm, every day of every week.

(b) for Office/Admin and Training Spaces, commencing at 8am and ending at 5pm, Monday to Friday of every week.

<u>Corrected Aggregate Energy Consumption</u> or <u>CAEC</u> means, for a Contract Year, the Aggregate Actual Consumption less the Non-Regulated Load consumption with appropriate adjustments for the Climate Baseline.

<u>Cost and Pricing Data</u> means the data (including calculations, formulas, unit and material prices, and other cost and fee information) acknowledged and accepted by Principal Project Company and delivered by Lead Developer as specified in the Predevelopment Agreement, which data supports and explains the basis of Principal Project Company's cost estimates for development, design, construction, operations, and maintenance of the Project and provides all cost assumptions for human resources, including salary and benefits for non-management personnel performing the Work.

<u>Cost to Complete</u> means (without double-counting):

(a) those costs (internal and external) that City reasonably and properly projects that it will incur in carrying out any process to request tenders from any parties interested in entering into a contract with City to achieve Final Acceptance, including all costs related to the preparation of tender documentation, evaluation of tenders and negotiation and execution of relevant contracts; plus

- (b) costs that City reasonably and properly projects that it will incur in achieving Final Acceptance; plus
- (c) any other Losses that, but for the termination of this Agreement, City would not have incurred prior to Final Acceptance; minus
- (d) any insurance proceeds available to City for the purposes of achieving Final Acceptance.

**<u>COTS</u>** (or, <u>**Commercially Available Off-the-Shelf Software**</u>) means Software (i) sold in substantial quantities, (ii) readily available to City without Principal Project Company or third party participation, (iii) provided without modification in the same form in which it is sold in the commercial marketplace, and (iv) for which there are at least two (2) readily available alternative solutions or items with the same or substantially similar design, use or function as the proposed COTS. For the avoidance of doubt, COTS does not include so-called open source software or sole-source software.

# Credit Spread Risk Mitigation has the meaning set forth in Section 3.2.3.2.

<u>Critical Path</u> means each critical path on the Project Schedule, which ends on the Substantial Completion Deadline or the Final Acceptance Deadline, as applicable (i.e., the term shall apply only following consumption of all available Float in the schedule for Substantial Completion or Final Acceptance, as applicable). The lower case term "critical path" means the activities and durations associated with the longest chain(s) of logically connected activities through the Project Schedule with the least amount of total float for all chains.

<u>Critical Path Method (CPM)</u> is a method of planning and scheduling a construction project where activities are arranged based on activity relationships.

<u>**Current Status Documents</u>** has the meaning set forth in <u>Section 1.6.3(C)</u> of <u>Division 1</u> of the Technical Requirements.</u>

D&C Contract means:

- (a) this Agreement, where Principal Project Company will self-perform the D&C Work; or
- (b) the contract for the D&C Work entered into between Principal Project Company and D&C Contractor dated on or about the date of this Agreement, where Principal Project Company will not self-perform the D&C Work; and
- (c) any other contract between Principal Project Company and a Subcontractor for the undertaking of the D&C Work.

## **D&C Contract Amount** means [\$•].

**D&C Contractor** means the Contractor primarily responsible for the D&C Work.

<u>**D&C Noncompliance**</u> means and where a D&C Failure Event occurs which, where a Rectification Time applies, has not been Rectified within the relevant time. For the avoidance of

doubt, where no Rectification Time applies (for example, in respect of scheduled activities) there shall be a D&C Failure at the point at which the D&C Failure Event occurred (for example, non-performance of the scheduled activity by the scheduled time).

**<u>D&C Failure Event</u>** means each event identified in <u>Appendix A</u> (D&C Noncompliance Points and Deductions) of <u>Exhibit 4</u> (Payment Mechanism) which occurs prior to the Final Acceptance Date.

<u>D&C Noncompliance Points</u> means points accrued in association with each D&C Noncompliance as set out in <u>Section 2</u> (Escalation Factor) of <u>Exhibit 4B</u> (Availability Payment Mechanism).

**D&C Management Plan** means the City-approved document attached at <u>Exhibit 2</u> (Project Management Plan) setting forth Principal Project Company's prescribed approaches to, and plan for, its scope of Work, as it may be modified and updated from time to time, following written approval thereof by City. The preliminary D&C Management Plan shall be replaced by the final approved D&C Management Plan, which shall be developed by Principal Project Company and approved by City pursuant to <u>Division 1</u> of the Technical Requirements, as a condition precedent to issuance of NTP 2, as updated in accordance with this Agreement.

**<u>D&C Payment Bond</u>** means the payment bond(s) to secure payment for labor and materials for the D&C Work, as required under this Agreement.

**D&C Percentage** means the value of Design Work, Construction Work, and Work relating to the manufacturing and supply of equipment, completed at a given point in time divided by the value of the total Design Work, Construction Work, and Work relating to the manufacturing and supply of equipment required to be completed to achieve Final Acceptance.

**D&C Performance Bond** means the performance bond(s) securing performance of the D&C Work required under this Agreement.

**D&C Performance Security** has the meaning set forth in <u>Section Error! Reference source</u> not found.

**D&C Period** means that portion of the Term that commences on the Effective Date and ends at 11:59 p.m. on the day immediately preceding the Substantial Completion Date.

**<u>D&C Period Deduction</u>** means a deduction from the Milestone Payment made in accordance with <u>Section 2</u> (D&C Period Deductions) of <u>Exhibit 4A</u> (Milestone Payment Mechanism).

**D&C Site** means the real property owned by City, under the jurisdiction of SFMTA, commonly known as 2500 Mariposa Street in San Francisco, California, which is a 4.4-acre site comprised of Assessor's Block No. 3971-001, bounded by Bryant S<sup>tr</sup>eets, 17th Street, Hampshire Street, and Mariposa Street, any Additional Properties acquired by City in accordance with <u>Section</u> 7.5.1 (Additional Acquisitions), and such additional areas as may, from time to time, be designated in writing by City for Principal Project Company's use in performance of the Work. If the parcel is later subdivided to accommodate a Joint Development Alternative, this definition may be amended upon mutual agreement. For purposes of insurance (subject to any notification and other requirements imposed by the insurer(s) for approval), indemnification, safety and security requirements, the prevailing wage requirements, and payment for use of

equipment, the term "Project Site" shall also include (a) the field office sites, (b) any property used for bonded storage of material for the Project approved by City, (c) staging areas dedicated to the Project, and (d) areas where activities incidental to the Project are being performed by Principal Project Company and Contractors covered by the worker's compensation policy but excluding any permanent locations of Principal Project Company or such covered Contractors.

**D&C Quality Management Plan** means Principal Project Company's plan to ensure that the Project deliverables and completed Work meet the minimum required standards of quality, prepared and submitted by Principal Project Company in accordance with <u>Section 1.4</u> of <u>Division 1</u> of the Technical Requirements, as updated and approved by the City in accordance with this Agreement.

**<u>D&C Work</u>** means all Work to be performed to achieve Final Acceptance, including Design Work and Construction Work.

**D&C Work Value** means an amount equal to the D&C Contract Amount minus the aggregate of (i) the Cost to Complete, (ii) the amount of the Milestone Payment if paid to Principal Project Company prior to the Early Termination Date, (and (iii) any Deduction or other reduction to the Milestone Payment or any Availability Payments under <u>Exhibit 4</u> (Payment Mechanism) accrued prior to the Early Termination Date that has, in each case, not been deducted from the Milestone Payment or any Availability Payment (as applicable).

Day means calendar day, unless otherwise expressly specified.

# DBE Compliance Manual means [Note: City to provide definition]

## DBE Contractor means [Note: City to provide definition]

<u>Deduction</u> means a D&C Period Deduction or an IFM Period Deduction, or both as the context requires.

<u>Deduction Amounts</u> means the amount in dollars of each Deduction associated with a D&C Period Deduction, and/or an IFM Period Deduction in accordance with the applicable Functional Component Rank category in <u>Appendix B</u> (Response and Rectification Times) to <u>Exhibit 4B</u> (Availability Payment Mechanism).

## Default Notice has the meaning set forth in Section 16.1.2.2.

## Default Termination Event has the meaning set forth in Section 17.3.1.1.

<u>Defect</u> means a defect, whether due to design, construction, installation, damage or wear, affecting the condition, use, functionality or operation of any element of the Project, which would cause or have the potential to cause one or more of the following:

- (a) a hazard, nuisance or other risk to public or worker health or safety, including the health and safety of Building Occupants;
- (b) a structural deterioration of the affected element or any other part of the Project;

- (c) damage to a Third Party's property or equipment;
- (d) damage to the Environment; and
- (e) failure of the affected element to meet the requirements of the Contract Documents.

**Deferred Equity Amounts** means, on any date, any amount of unfunded cash equity that has been committed to Principal Project Company (including commitments to provide an Equity Investment or Equity Member Debt) and is shown to be available for use in the Financial Model prior to the Final Acceptance Date, but only to the extent that the commitment to provide such amount is supported by an Equity Letter of Credit.

**Delay Costs** means direct costs incurred by Principal Project Company relating to Controlling Work Items that result solely and directly from a Delay Event, which are limited to those costs identified in and calculated in accordance with <u>Exhibit 13</u> (Costs Schedule). The term "Delay Costs" does not include any Extra Work Costs, including those incurred by Principal Project Company that relate to non-Controlling Work Items incident or collateral to a Delay Event or Controlling Work Items that do not result solely and directly from a Delay Event. In any event, Principal Project Company shall not be entitled to Delay Costs to the extent Principal Project Company is responsible for the delay or could have reasonably mitigated.

**Delay Event** means any of the following events or circumstances that occurs prior to the Final Acceptance Date and directly impacts the D&C Work:

- (a) an Unavoidable Delay Event; or
- (b) a Compensable Delay Event.

Delayed Financial Close Date has the meaning set forth in Section 3.4.2(a).

**Demand Maintenance** means all unscheduled, corrective, or first response maintenance (other than Scheduled Maintenance) to the Maintained Elements, which includes the response to malfunctions and provision of minor repairs, adjustments and general maintenance as follows:

- (a) first response to equipment malfunctions and assessment of the problem (e.g., operator error, Utility problem, minor or major failure), and required response; and
- (b) performance of minor repairs and general maintenance, including topping-up fluids, adjustments, resets, clearing blockages, minor carpentry and replacing minor parts such as rollers, wheels, pulley and hoses.

**Demand Requisition** means a requisition for Demand Maintenance.

<u>Design Deliverables</u> has the meaning set forth in <u>Section 1.8.5</u> of <u>Division 1</u> of the Technical Requirements.

<u>Design Documents</u> means all studies, quantitative assessments, evaluations, reports, documents, calculations, plans, drawings, diagrams, specifications, and other documentation

that manifest the design for the Project, at all stages, as developed by Principal Project Company or any portion, component or element thereof, including design required in connection with the operation and maintenance of the Project and Renewal Work, in each case regardless of whether such documents are required by the Contract Documents or are prepared or used by Principal Project Company in the Design Work. Design Documents include the Final Design Documents.

**Design Management Plan** means Principal Project Company's plan describing its approach to undertake and achieve the requirements in <u>Section 1.8</u> of <u>Division 1</u> of the Technical Requirements, prepared and submitted by Principal Project Company in accordance with <u>Section 1.1.7.1</u> of <u>Division 1</u> of the Technical Requirements, as updated and approved by the City in accordance with this Agreement.

**Design Work** means all Work related to the design, engineering, architecture and other professional services for the Project, including all such Work relating to the Infrastructure Facility. Design Work includes any Early Works related to the design, engineering, architecture and other professional services for the Project.

**Developed IP** means Intellectual Property that is authored, created, invented or reduced to practice under or for the purposes of this Agreement, the Work or the Project, whether or not such Intellectual Property is incorporated into the Project IP but excluding any adaptation, continuation or derivative work that constitutes PPC Intellectual Property.

**Deviation(s)** means any change, deviation, modification or alteration from the Technical Requirements (other than Change Orders and amendments to this Agreement). Deviations are not intended to amend or modify portions of this Agreement other than the Technical Requirements.

**Direct Agreement** means the agreement substantially in the form of <u>Exhibit 5B</u> (Form of Direct Agreement), by and among City, Principal Project Company and the Lender (or if there is more than one Lender, the Collateral Agent on behalf of the Lenders) in connection with Lenders' rights in relation to the Contract Documents.

**Dispatch Function** means the Functional Components corresponding to Functional Unit references L2-011 as specified in <u>Table B6</u> (Functional Components and Functional Units for Infrastructure Facility) of <u>Appendix B</u> (Functional Areas, Functional Units, Functional Unit Rankings and Deduction Amounts for Availability Failures and Service Failures) of <u>Exhibit 4B</u> (Availability Payment Mechanism).

**Distributions** means the proceeds of any Refinancing to the extent such proceeds are distributed to any Equity Member or to any Affiliate of Principal Project Company, and any of the following, whether in cash or in kind, and whether made or projected to be made to any Equity Member, an Affiliate of Principal Project Company. Any:

- (a) Dividend or other distribution in respect of any equity interest of any class;
- (b) Payments or other distributions in reduction of capital, redemption or purchase of shares or any other reorganization or variation to share capital;
- (c) Payments in respect of Equity Member Debt (whether of fees, principal, interest

including capitalized interest and interest on overdue interest, breakage costs, or otherwise and whether or not such items are included or excluded from the definition of Equity Member Debt);

- (d) Payment, Ioan, contractual arrangement, including any management agreement or payment in respect thereof, or transfer of assets or rights, in each case to the extent made or entered into after the date of this Agreement and not in the ordinary course of business nor on reasonable commercial terms including to any Equity Member, or any Affiliate of any Equity Member; and
- (e) Conferral of any other benefit which is not received in the ordinary course of business and not on commercially reasonable terms, including to any Equity Member, any Affiliate of any Equity Member or to Principal Project Company;
- (f) Other payment to any Equity Member, any current or former Affiliate of any Equity Member or Principal Project Company howsoever arising and whether made under the terms of an agreement or otherwise or in respect of any equity interest of any class in the capital of Principal Project Company if, in any such case, such payment would not have been made were it not for the occurrence of any Refinancing or Change in Control; or
- (g) Early release of any contingent funding liabilities, the amount of such release being deemed to be a gain for the purposes of any calculation of Refinancing Gain.

**Early Termination Date** means the effective date of termination of this Agreement prior to the stated expiration of the Term set forth in <u>Section 2.3</u> (Term).

Early Works has the meaning set forth in the Early Works Agreement.

**Early Works Agreement** means the agreement entered into by City and Principal Project Company or an Affiliate with respect to the performance of certain Work by Principal Project Company or such Affiliate during the PDA Term and prior to Financial Close.

**Earthquake** means all land movement due to seismic activity, including shocks, tremors, volcanic action, tsunami and earth rising or shifting, including any aftershocks or other Earthquakes for a period of 168 hours after the initial event, which proximately causes damage to the physical improvements of the Project or interrupts the Work.

**Earthquake Inspection Plan** means the plan detailing an approach to minimize delay in reoccupying the Infrastructure Facility following a significant seismic event as part of the City of San Francisco's Building Occupancy Resumption Program (BORP), prepared and submitted by Principal Project Company in accordance with <u>Section B.23.1(5)</u> of the IFM Specifications, as updated and approved by the City in accordance with this Agreement.

Effective Date means the date of Commercial Close.

**<u>Eligible Investments</u>** has the meaning set forth in <u>Section 8.6.2.3</u>.

**Eligible LC Issuer** means a financial institution with long term unsecured debt ratings of at least the following, from at least two of the listed major rating agencies: (a) A- by Standard & Poor's Ratings Services; (b) A3 by Moody's Investor Service, Inc.; or (c) A- by Fitch Ratings.

**<u>Eligible Surety(ies)</u>** means a surety or insurance company, as applicable, meeting the requirements of applicable Law, licensed or authorized to do business in the State and rated at least "A" (excellent or above) according to A.M. Best's Financial Strength Rating and "VIII" or better according to A.M. Best's Financial Size Rating.

**Emergency** means any unplanned event within or adjacent to the Project Site that (a) causes or has the potential to cause disruption to operation of the Infrastructure Facility, (b) presents an immediate or imminent threat to the long term integrity of any part of the Facility, to the Environment, to property immediately adjacent to the Facility or to the safety of Building Occupants or the traveling public, or (c) is recognized or declared to be an emergency by the Mayor, Board of Supervisors, any other City official authorized under law, the Governor of the State, the Federal Emergency Management Administration (FEMA), the U.S. Department of Homeland Security or other Governmental Entity with authority to declare an emergency.

**Energy** means energy/power including electricity and natural gas.

Energy Analysis Report has the meaning set forth in Section 4.7 of Exhibit 12.

<u>Energy Pain Share/Gain Share Mechanism</u> means the energy risk share amount calculated pursuant to <u>Appendix B</u> (Functional Areas, Functional Units, Functional Unit Rankings and Deduction Amounts for Availability Failures) to <u>Exhibit 4B</u> (Availability Payment Mechanism).

**Energy Performance Commencement Date** means the date on which the Independent Energy Auditor issues the AET that is agreed to by Principal Project Company and City.

<u>Energy Service</u> means any calculated or metered provision of Energy in respect of the Regulated Load and Non-Regulated Load at the Utility connection.

**Engineer of Record (EOR)** has the meaning set forth in <u>Section 1.1.4.2</u> of <u>Division 10</u> of the Technical Requirements.

**Environment** means air, soils, submerged lands, surface waters, groundwaters, land, stream sediments, surface or subsurface strata, biological resources, including endangered, threatened and sensitive species, natural systems, including ecosystems, cultural (including historic and archaeological) resources and paleontological resources.

**Environmental Approvals** means all Regulatory Approvals arising from or required by any Environmental Law in connection with the Project.

**Environmental Compliance Plan** means the procedures and plans described in <u>Sections 01</u> <u>35 49</u>, <u>01 35 50</u> and <u>01 35 51</u> of <u>Division 10</u> of the Technical Requirements.

**Environmental Law(s)** means (a) any Law applicable to the Project or the Work regulating or imposing liability or standards of conduct that pertains to the Environment, Hazardous Materials, contamination of any type whatsoever, or health and safety matters, and (b) any requirements and standards that pertain to the protection of the Environment, or to the management or

Release of Hazardous Materials, contamination of any type whatsoever, or health and safety matters with respect to Hazardous Materials, set forth in any agreements, permits, licenses, approvals, plans, rules, regulations or ordinances adopted, or other criteria and guidelines promulgated, pursuant to Laws applicable to the Project or the Work, as each of the foregoing have been or are amended, modified, or supplemented from time to time (including any present and future amendments thereto and reauthorizations thereof), including those relating to:

- (a) the manufacture, processing, use, distribution, existence, treatment, storage, disposal, generation and transportation of Hazardous Materials;
- (b) air, soil, surface and subsurface strata, stream sediments, surface water, and groundwater;
- (c) releases of Hazardous Materials;
- (d) protection of wildlife, endangered, threatened, and sensitive species, wetlands, water courses and water bodies, paleontological, cultural, archaeological and historical resources and natural resources;
- (e) the operation and closure of underground or aboveground storage tanks;
- (f) health and safety of employees and other persons with respect to Hazardous Materials; and
- (g) notification, documentation and record keeping requirements relating to the foregoing.

Without limiting the above, the term "Environmental Laws" shall also include the following (each as may be amended from time to time):

- (i) the National Environmental Policy Act (42 U.S.C. § 4321 et seq.);
- (ii) the Comprehensive Environmental Response, Compensation, and Liability Act (42 U.S.C. § 9601 et seq.) and associated Superfund Amendments and Reauthorization Act (42 U.S.C. § 9601 et seq.);
- (iii) the Solid Waste Disposal Act (42 U.S.C. § 6901 et seq.);
- (iv) the Emergency Planning and Community Right to Know Act of 1986 (42 U.S.C. § 11001 et seq.);
- (v) the Clean Air Act (42 U.S.C. § 7401 et seq.);
- (vi) the Federal Water Pollution Control Act, as amended by the Clean Water Act (33 U.S.C. § 1251 et seq.);
- (vii) the Resource Conservation and Recovery Act (42 U.S.C. § 6901 et seq.);
- (viii) the Hazardous and Solid Waste Amendments of 1984 (42 U.S.C. § 6924 et seq.);

- (ix) the Toxic Substances Control Act (15 U.S.C. § 2601 et seq.);
- (x) the Hazardous Materials Transportation Act (49 U.S.C. § 5101 et seq.);
- (xi) section 404 of the Clean Water Act (33 U.S.C. § 1344);
- (xii) the Oil Pollution Act (33 U.S.C. § 2701 et. seq.);
- (xiii) the Federal Insecticide, Fungicide and Rodenticide Act (7 U.S.C. § 136 et seq.);
- (xiv) the Federal Safe Drinking Water Act (42 U.S.C. § 300 et seq.);
- (xv) the Federal Radon and Indoor Air Quality Research Act (42 U.S.C. § 7401 et seq.);
- (xvi) the Occupational Safety and Health Act (29 U.S.C. § 651 et seq.);
- (xvii) the Endangered Species Act (16 U.S.C. § 1531 et seq.);
- (xviii) the Fish and Wildlife Coordination Act (16 U.S.C. § 661 et seq.);
- (xix) the California Environmental Quality Act (§ 21000 et seq. of the California Public Resources Code);
- (xx) the California Clean Air Act of 1988 (§ 39000 et seq. of the California Health and Safety Code);
- (xxi) the California Occupational Safety and Health Act of 1973 (§ 6300 et seq. of the California Labor Code);
- (xxii) the Porter-Cologne Water Quality Act (§ 13000 et seq. of the California Water Code);
- (xxiii) the Integrated Waste Management Act of 1989 (§ 40000 et seq. of the California Public Resources Code);
- (xxiv) the Safe Drinking Water and Toxic Enforcement Act of 1986 (§ 25249.5 et seq. of the California Health and Safety Code);
- (xxv) Hazardous Waste Control (§ 25100 et seq. of the California Health and Safety Code); and
- (xxvi) Fish and Wildlife Protection and Conservation (§ 1600 et seq. of the California Fish and Game Code).

**Environmental Management System** or **EMS** means the system prepared and submitted by Principal Project Company and approved by the City in accordance with <u>Section B.22.2</u> of the IFM Specifications, as updated in accordance with this Agreement.

<u>Environmental Protection Program</u> means the overarching system by which Principal Project Company shall ensure that commitments made during the environmental approval and permitting processes, and other environmental requirements, be carried forward and reflected, as appropriate, in the design and implemented throughout the Work, as provided in <u>Sections 01</u> <u>35 49</u>, <u>01 35 50</u> and <u>01 35 51</u> of <u>Division 10</u> of the Technical Requirements.

**Equipment List** means the list of all FF&E for the Infrastructure Facility prepared and maintained in accordance with this Agreement which identifies PPC-Furnished FF&E, City-Furnished Equipment and Existing FF&E. The initial form of the Equipment List is included in Appendix 1 of Division 3 of the Technical Requirements.

**Equity Investment(s)** means (a) any form of direct cash investment by Equity Members, including the purchase of newly issued equity shares in Principal Project Company and/or subordinated loans to Principal Project Company, and (b) any cash draws by or on behalf of Principal Project Company under the letter(s) of credit described in the definition of Deferred Equity Amount.

**Equity IRR** means the nominal post-tax internal rate of return to the Committed Investment over the full Term calculated, using the Financial Model, at the discount rate that, when applied to Committed Investment cash flows, gives a zero net present value. Equity IRR is initially equal to the Original Equity IRR and can change when and if the Financial Model is updated pursuant to <u>Section 4.7.2</u> (Updates to the Financial Model). For purposes of this definition the phrase "post-tax" refers only to U.S. federal income tax, state income tax, and San Francisco gross receipts, homelessness gross receipts, and overpaid executive gross receipts taxes liability of Principal Project Company or its Equity Members (or, if an Equity Member is a single-member limited liability company disregarded for federal and state income tax purposes, the owner of such Equity Member) and specifically excludes (a) any foreign income tax and other tax of any kind, and (b) any withholding tax, including any tax that Principal Project Company or an Equity Member is obligated to withhold on Distributions (whether actual or constructive) or other payments or allocations to Equity Members or holders of debt of or equity interests in an Equity Member under 26 U.S.C. §§ 1441 – 1446, notwithstanding 26 U.S.C. § 1461.

**Equity Letter of Credit** means an irrevocable and unconditional on-demand letter of credit (which may be annually renewable) issued by an Eligible LC Issuer for the account of an Equity Member naming Principal Project Company and/or the Collateral Agent as beneficiary and securing the provision of any Deferred Equity Amounts by a date that is not later than the Final Acceptance Date.

**Equity Member(s)** means (a) [\_\_\_\_\_] and any Person that holds a direct ownership interest (legal and beneficial) in [\_\_\_\_\_] either based on Committed Investments or as a result of an Equity Transfer; or (b) each Person that will hold a 10% or greater indirect interest in Principal Project Company.

**Equity Member(s) Debt** means bona fide indebtedness for funds borrowed that: (a) is held by any Equity Member or Affiliate of any Equity Member; and (b) is subordinated in priority of payment and security to all Project Debt held by Persons who are not Equity Members or Affiliates of any Equity Member.

**Equity Members Funding Agreement** means any loan agreement, credit agreement or other similar financing agreement or subordination agreement providing for or evidencing Equity Member Debt.

**Equity Transfer** means any assignment, mortgage, encumbrance, hypothecation, conveyance, sale, or other transfer of equity interest in Principal Project Company.

### Escalated Benchmark Insurance Premiums has the meaning set forth in Section 10.1.3(d)(ii).

**Escalation Factor** means the escalation factor calculated in accordance with <u>Section 2</u> (Escalation Factor) of <u>Exhibit 4B</u> (Availability Payment Mechanism).

#### Exempt Refinancing means:

- (a) any Refinancing that was fully and specifically identified and taken into account in the Base Case Financial Model;
- (b) any of the following, provided that Principal Project Company does not receive a direct or indirect financial benefit therefrom, individually or in the aggregate:
   (i) amendments, modifications, supplements or consents to Financing Agreements and Security Documents; or (ii) the exercise by a Lender of rights, waivers, consents and similar actions, in the ordinary course of day-to-day loan administration and supervision;
- (c) movement of monies between the Project accounts in accordance with the terms of Financing Agreements and Security Documents;
- (d) any of the following acts by a Lender of senior lien priority Project Debt: (i) the syndication of any of such Lender's rights and interests in the senior Financing Agreements; (ii) the grant by such Lender of any rights of participation, or the disposition by such Lender of any of its rights or interests, with respect to the senior Financing Agreements in favor of any other Lender of senior lien Project Debt or any other investor; or (iii) the grant by such Lender of any other form of benefit or interest in either the senior Financing Agreements or the revenues or assets of Principal Project Company, whether by way of security or otherwise, in favor of any other Lender of senior lien Project Debt or any other Lender of senior lien Project Debt or any other Lender of senior lien Project Company, whether by way of security or otherwise, in favor of any other Lender of senior lien Project Debt or any investor; and
- (e) periodic resetting and remarketing of tax-exempt or taxable bonds that bear interest at a variable or floating rate and are money market eligible under SEC Rule 2a-7.

**Existing FF&E** means all FF&E identified by City in the updated Equipment List under <u>Section</u> 7.13.2 (Selection of City-Furnished Equipment and Existing FF&E).

**External Energy Auditor** means a company licensed in California, specializing in assessing the energy efficiency of commercial and industrial buildings ('audit'), including the review of specifications, analysis of consumption data, energy modeling, and provision of recommendations.

**Extra Work** means any Work in the nature of additional work, altered work or deleted work that is directly attributable to the occurrence of a Compensable Delay Event, a Compensable Relief Event or uninsured Force Majeure Event (to the extent permitted under <u>Section 14.6</u> (Loss or Damage Due to Force Majeure Termination Event)) and, absent the Compensable Delay Event, Compensable Relief Event or uninsured Force Majeure Event (to the extent permitted under <u>Section 14.6</u> (Loss or Damage Due to Force Majeure Termination Event)) and, absent the Compensable Delay Event, Compensable Relief Event or uninsured Force Majeure Event (to the extent permitted under <u>Section 14.6</u> (Loss or Damage Due to Force Majeure Termination Event)), would not be required by the Contract Documents. The term "Extra Work" does not include "Delay".

<u>Extra Work Costs</u> means the incremental increase in Principal Project Company's cost of labor, material, equipment and other direct and indirect costs directly attributable to Extra Work, calculated in accordance with <u>Section 1</u> of <u>Exhibit 13</u> (Costs Schedule).

**Facility** means the Bus Yard Component, Joint Development Alternative (if built), and Common Infrastructure, collectively.

**Facility Condition Index** or **FCI** means the total maintenance, repair and replacement deficiencies of the Infrastructure Facility divided by the current replacement value of the Infrastructure Facility.

**Failure Event** means a D&C Failure Event or an IFM Failure Event, or both as the context requires.

**Failure Type** means the designation of either "SF" (Service Failure), in the Performance Measurements Table or "AF" (Availability Failure) in the case of an IFM Failure Event which has not been rectified within the relevant Rectification Time and which causes a Functional Part to be Unavailable.

**Final Acceptance** means that all D&C Work is complete and all other prerequisites for Final Acceptance have been met. Final Acceptance is deemed to have occurred upon satisfaction of all the conditions in <u>Section 7.9</u> (Final Acceptance), as confirmed by City's issuance of a certificate in accordance with the procedures and within the time frame established in <u>Section 7.9</u> (Final Acceptance).

Final Acceptance Date means the date that Final Acceptance is achieved.

**Final Acceptance Deadline** means 180 days after the Substantial Completion Date, as such date may be extended from time to time pursuant to <u>Section 14.1.2</u> (Extension of Deadlines for General Delay Events).

**Final Design** means the general design stage, consistent of all elements, collections of elements or areas of the Project at 100% design completion, as more fully set forth in the Technical Requirements, and depending on the context: the term "Final Design" may refer to (a) the Final Design Documents, (b) the design concepts set forth in the Final Design Documents or (c) the process of development of the Final Design Documents.

**Final Design Documents** means the complete final construction plans (including plan sheets, specifications, technical memoranda, reports, studies, calculations, drawings, elevations, sections, details and diagrams) and specifications needed for performance of Construction Work, which includes all Submittals specified as required to be part of the Final Design or Final Design Documents under the Technical Requirements.

**Final Environmental Impact Report (FEIR)** means the Final Environmental Impact Report for the Potrero Yard Modernization Project at 2500 Mariposa Street (Planning Department Case No. [\_\_\_\_]), dated [\_\_\_\_] and available at <a href="https://sfplanning.org/environmental-review-documents?title=Potrero+Yard+Modernization+Project">https://sfplanning.org/environmental-review-documents?title=Potrero+Yard+Modernization+Project</a>.

**<u>Final Schedule Report</u>** means has the meaning set forth in <u>Section 1.2.5</u> of <u>Division 1</u> of the Technical Requirements.

**Finance Plan** means Principal Project Company's plan for financing the Project submitted to City at Performance Milestone 32 of the Predevelopment Agreement.

**Finance Plan Due Date** means the date on which Principal Project Company submitted its final Finance Plan to City at Performance Milestone 32 of the Predevelopment Agreement.

**Finance Plan Validity Period** means the period of 180 consecutive days commencing on the Finance Plan Due Date, as such period may be extended under the terms of the Predevelopment Agreement. If the period of 180 consecutive days commencing on the Finance Plan Due Date ends on a weekend or national holiday, then the Final Finance Plan Validity Period shall end on the next Business Day.

**Financial Close** means satisfaction by Principal Project Company of (a) all conditions precedent to the effectiveness of commitments of the Lenders under the Financing Documents; and (b) the conditions set forth in <u>Section 3.2.4</u> (Conditions Precedent to Financial Close).

Financial Close Date means the date on which Financial Close occurs.

**Financial Close Deadline** means [\_\_\_\_\_\_, 2025], as such date may be extended by agreement of the Parties or as a result of an Adverse Event or other excuse set forth in this Agreement.

**Financial Close Security** means the bond or letter of credit in the amount of \$15,000,000 provided by Principal Project Company to City as a condition precedent to execution and delivery of this Agreement by both City and Principal Project Company.

**Financial Model** means the Microsoft Excel-based financial model that includes financial forecasts, projections and calculations with respect to revenues, expenses, the repayment of Project Debt and Distributions to Equity Members that result in achievement of the Equity IRR, which shall be in macro-enabled excel format, on electronic storage media, or in such other format and medium as the Parties may agree in writing.

Financial Model Update(s) has the meaning set forth in Section 4.7.2.1.

**Financial Modeling Data** means all books, documents and back-up information in any media or format setting forth all assumptions, estimates, projections, calculations and methodology used in the preparation of the Implementation Proposal, the Financial Model, and in Financial Model Updates of revenues, pricing, costs, expenses, repayment of Project Debt, Distributions and internal rate of return, including:

(a) the logical layout and structure of the Financial Model, including the names of all worksheets and a description of the color coding and/or labeling scheme(s);

- (b) an assumptions book, fully describing all assumptions and their sources underlying the estimates, projections and calculations in the Financial Model, revisions to the Base Case Financial Model in accordance with <u>Sections 3.2.4</u> (Conditions Precedent to Financial Close) and <u>3.3.1</u>, and updates to such assumptions book related to Financial Model Updates;
- a detailed description of the function and intended use of all macros (each of which must be logically structured and well documented with comments included within the programming code);
- (d) the step-by-step instructions on the procedure to run and to optimize the Financial Model submitted with the Finance Plan and each Financial Model Update, such that City will be able to read, use and modify the data contained therein, operate the Financial Model and conduct detailed sensitivity analyses;
- (e) copies of all offers, and all data and information within this definition, received from all Contractors (at all tiers) identified in the Implementation Proposal;
- (f) copies of all offers, and all data and information within this definition, received from all Contractors (at all tiers) related to any Relief Event; and
- (g) all other supporting data, technical memoranda, calculations, formulas, unit and materials prices (if applicable) and such other cost, charge, fee, financing, equity return, and revenue information used by Principal Project Company in the creation and derivation of the Implementation Proposal, Finance Plan, the Financial Model, or of any Financial Model Update, or related to any Delay Event, Relief Event or Change Order.

**Financial Proposal** means the Financial Proposal provided at <u>Exhibit 3A</u> (Financial Proposal). [Note to PNC: This will include the Finance Plan, including the Base Case Financial Model, with its audited report, and respective commitment letters for Bank Debt, Equity Investments, Security Documents and Insurance Policies, among other documents, and including the Final Price as determined under the PDA, PPC's financing costs, along with a schedule including the proposed Milestone Payment amount and the proposed Availability Payments payable from Substantial Completion to the end of the PDA Term.]

## Financing Agreement means:

- (a) any loan agreement, funding agreement, account maintenance or control agreement, insurance or reimbursement agreement, intercreditor agreement, subordination agreement, trust indenture, agreement from any Equity Member in favor of any Lender, hedging agreement, interest rate swap agreement, guaranty, indemnity agreement, agreement between any Contractor and any Lender, or other agreement by, with or in favor of any Lender pertaining to Project Debt (including any Refinancing), other than Security Documents;
- (b) any note, bond or other negotiable or non-negotiable instrument evidencing the indebtedness of Principal Project Company for Project Debt (including any Refinancing); and

(c) any amendment, supplement, variation or waiver of any of the foregoing agreements or instruments.

## Financing Delay Costs means:

- (a) in the case of Delay Events, an amount equal to principal payments, interest payments, hedging costs and other financing costs accrued and/or paid, or which became payable, by Principal Project Company to the Senior Lenders in accordance with the Financing Agreements that accrued during the period of the schedule extension granted pursuant to <u>Section 14.1.2</u> (Extension of Deadlines for General Delay Events); and
- (b) in the case of City-Caused Delay Event only, such other amounts so as to result in Principal Project Company achieving the same Equity IRR (with reference to the Base Case Equity IRR) as if such Compensable Delay Event had not occurred.

Financing Documents means Financing Agreements and Security Documents.

Financing Document Terms means Exhibit 5G (Financing Document Terms).

**Fiscal Year** means the consecutive 12 calendar month period starting on July 1 and ending on June 30.

**Fixed IFM Costs** means the sum of those fixed costs:

(a) incurred by the IFM Provider with respect to insurance premiums, office rental and office expenses, equipment hire costs, depreciation on items of equipment used by Principal Project Company in the performance of the IFM Services excluding:

- (ii) equipment incorporated into the Infrastructure Facility and not capable of being demobilized and remobilized to a different site or portion of the Work, the Project, the Infrastructure Facility, the Project Site or to another project;
- (iii) third party accounting, audit and legal costs; and
- (iv) any costs payable by Principal Project Company to the IFM Provider for the provision of labor,

(a) incurred directly by the IFM Provider during the IFM Period in meeting its obligations in accordance with this Agreement and solely and directly attributable to the Project and which are not reasonably capable of being deferred or avoided by the IFM Provider; and

(b) evidenced in writing to the reasonable satisfaction of the City's Authorized Representative.

<u>Float</u> means the amount of time that any given activity or logically connected sequence of activities shown on the Project Schedule may be delayed before it will affect the Substantial

Completion Deadline or Final Acceptance Deadline, as applicable. Float is generally identified on the Project Schedule as the difference between the early start date and late start date, or early completion date and late completion date for activities shown on the Project Schedule, and shall include any float contained within an activity as well as any period containing an artificial activity (that is, an activity that is not encompassed within the meaning of the definition of "D&C Work").

**Force Account Work** means Change Order work that City authorizes and will pay for on the basis of direct costs plus markup on direct costs for overhead and profit as provided in <u>Exhibit</u> <u>13</u> (Costs Schedule).

**Force Majeure Event** means the occurrence of any of the following events or circumstances which directly causes either Party to be unable to perform all or a material part of its obligations under this Agreement:

- (a) war (including civil war or revolution), invasion, violent act of foreign enemy or armed conflict occurring within the continental United States or military or armed blockade, or military or armed takeover of the Project;
- (b) any act of terrorism, riot, insurrection, civil commotion or sabotage that causes direct physical damage to, or otherwise directly causes interruption to construction or direct losses during maintenance of, the Project;
- (c) national or state-wide strikes not specific to Principal Project Company, or embargoes, that, in each case, directly cause interruption to construction or direct losses during maintenance of the Project;
- (d) nuclear, radioactive, or biological contamination of the Project unless the source or cause of the contamination is brought to or near the Project Site by PPC-Related Entities, or is the result of any PPC Fault;
- (e) flood, fire, explosion, tidal wave, sinkhole caused by natural events, or landslide caused by natural events, in each case, occurring within the City and directly impacting the Project or performance of Work;
- (f) SFMTA staff strikes and work stoppages not caused by PPC Fault that directly cause interruption to construction or direct losses during maintenance of the Project;
- (g) any City, or governor-declared, Emergency within the limits or precluding access to the Project Site;
- (i) any epidemic in the State; or (ii) Pandemic (defined as the worldwide spread of a disease due to lack of immunity in most people, excluding COVID-19);
- (i) a Change in Law (other than a Relevant Change in Law);
- (j) an electricity outage upstream from the connection to the Project Site that affects prosecution of the IFM Services, provided that Principal Project Company was not, in whole or in part, the cause of such outage and was in compliance with

<u>Section 2.10.5</u> of <u>Division 3</u> of the Technical Requirements and all other requirements in the Contract Documents regarding back-up power for the Project; and

(k) a seismic event in excess of a magnitude 3.5 on the Richter Scale, where (i) such Earthquakes, ground shaking, liquefaction, settlement, or ground movements directly impact, and cause damage to, temporary or permanent works of the Project; and (ii) such event is not insured or required to be insured under the terms of the Contract Documents.

## Force Majeure Termination Event has the meaning given in Section 17.2.1.

**Functional Component** means a room, space or system within a Functional Unit in the Infrastructure Facility, which is specified as such in <u>Appendix C</u> (Functional Areas, Functional Units, Functional Unit Rankings and Deduction Amounts for Availability Failures) of <u>Exhibit 4</u> (Payment Mechanism).

**Functional Component Rank** means the rank on a scale of 1 to 5 ascribed to each Functional Component in the Infrastructure Facility for the purpose of establishing Deduction Amounts associated with each Functional Component and calculating Deductions for Availability Failures based on the Rectification Time as set out in <u>Appendix B</u> (Response and Rectification Times) of <u>Exhibit 4</u> (Payment Mechanism).

**Functional Part** means either a Functional Unit or a Functional Component.

**Functional Unit** means an area of the Infrastructure Facility specified as such in <u>Appendix C</u> (Functional Areas, Functional Units, Functional Unit Rankings and Deduction Amounts for Availability Failures) of <u>Exhibit 4</u> (Payment Mechanism) comprising one or more Functional Components.

<u>Geotechnical Baseline Report (GBR)</u> means the report entitled Geotechnical Baseline Report, set forth in <u>Appendix A</u> of <u>Division 3</u> of the Technical Requirements.

<u>Good Faith Efforts</u> means (a) with respect to DBE, the efforts to meet the DBE goal required under 49 CFR Part 26, Appendix A, (B) with respect to SBE, the efforts to meet the SBE Goal required by the Project Agreement.

## Good Industry Practice means:

(a) in the case of Work (excluding the Design Work), the exercise of the degree of skill, diligence, prudence and foresight which would reasonably and ordinarily be expected from a skilled and experienced constructor, supplier, operator or maintenance provider, as applicable, operating in the United States under the same or similar circumstances and conditions, seeking in good faith to comply with its contractual obligations, the Contract Documents and all applicable Laws and Regulatory Approvals in conformance with all professional engineering principles, construction, operations and maintenance practices generally accepted as standards of the industry in the State. With respect to storm water management for construction activities, Good Industry Practice means Best Management Practices; in the case of Design Work, the exercise of the degree of skill, diligence, prudence, and foresight which would reasonably and ordinarily be expected from a professional designer or engineer, as applicable, operating in the United States under the same or similar circumstances and conditions, seeking in good faith to comply with its contractual obligations, the Contract Documents and all applicable Laws and Regulatory Approvals and engaged in the same type of undertaking under circumstances similar to the Project and conditions similar to those within the same geographic area as the Project Site.

<u>Government Code Dispute</u> has the meaning set forth in <u>Section 18.4</u> (Government Code Dispute).

<u>Governmental Entity(ies)</u> means City in its regulatory capacity, any federal, state, local or foreign government and any political subdivision or any governmental, quasi-governmental, judicial, public or statutory instrumentality, administrative agency, authority, body or entity other.

**Governmental Utility(ies)** has the meaning set forth in <u>Section 00 73 20</u> of <u>Division 10</u> of the Technical Requirements.

**Handback** means the stage when Principal Project Company has done everything that this Agreement requires to enable Principal Project Company to handover the Project to meet the Handback Requirements at the end of the Term or Early Termination Date.

Handback Inspection has the meaning set forth in Section 8.5.4.1(c).

Handback Renewal Work means the Renewal Work required in order for the Project to meet the Handback Requirements.

<u>Handback Renewal Work Plan</u> means the plan prepared and submitted by Principal Project Company in accordance with <u>Section F</u> of the IFM Specifications, as updated and approved by the City in accordance with this Agreement.

<u>Handback Requirements</u> means the terms, conditions, requirements and procedures governing the condition in which Principal Project Company is to deliver the Infrastructure Facility and the Project Site upon Handback, as stated in <u>Section F</u> of the IFM Specifications.

Handback Requirements Letter(s) of Credit has the meaning set forth in Section 8.6.5.1.

Handback Requirements Recovery Plan means the plan to perform additional work needed to meet the Handback Requirements at the end of the IFM Period prepared and submitted by Principal Project Company in accordance with <u>Section F.2</u> of the IFM Specifications, as updated and approved by the City in accordance with this Agreement.

Handback Requirement Recovery Work means the additional work as identified in the Handback Requirements Recovery Plan needed to bring any element of the Infrastructure Facility to meet the Handback Requirements.

Handback Requirements Reserve Account means the account established pursuant to Section 8.6.1.1.

Handback Requirements Reserve Balance means: (a) if no Handback Requirements Letter of Credit is delivered pursuant to <u>Section 8.6.5.1</u>, all amounts standing to the credit of the Handback Requirements Reserve Account on the Termination Date; or (b) if any Handback Requirements Letter of Credit is delivered pursuant to <u>Section 8.6.5.1</u>, all amounts that would have been standing to the credit of the Handback Requirements Reserve Account on the Termination Date if no Handback Requirements Letter of Credit of the Handback Requirements Reserve Account on the <u>Section 8.6.5.1</u>.

Handback Reserve Amount has the meaning set forth in Section 8.5.4.2(h).

<u>Handback Year</u> means each consecutive 12 calendar month period commencing seven full years before the expected end of the Term.

**Hazardous Material(s)** means any (a) substance, product, waste, pollutant, contaminant or other material of any nature whatsoever that exceeds maximum allowable concentrations for that material, as defined by any Environmental Law; (b) substance, product, waste, pollutant, contaminant or other material of any nature whatsoever that is or becomes listed, regulated, or addressed under any Environmental Law; (c) substance, product, waste, pollutant, contaminant or other material of any nature whatsoever which may give rise to liability under <u>clause (a)</u> or (b) or under any statutory or common law theory based on negligence, trespass, intentional tort, nuisance, or strict liability or under any reported decisions of a state or federal court; (d) petroleum hydrocarbons excluding de minimis amounts and excluding petroleum hydrocarbon products contained within regularly operated motor vehicles; and (e) hazardous building materials including asbestos or asbestos-containing materials, lead or PCBs in structures and/or other improvements on or in the Project Site or in subsurface artifacts. The term "Hazardous Materials" includes Hazardous Waste and contaminated materials.

<u>Hazardous Materials Event</u> means Hazardous Materials in, over, under or emanating from the Project Site or which has migrated onto the Project Site from land or premises adjoining the Project Site that (1) render use of the Project unsafe or potentially unsafe absent assessment, containment and/or remediation, or (2) are required by applicable Law to be recycled, treated or stored, excluding Hazardous Materials:

- (a) that are Known or Suspected Hazardous Materials; and
- (b) to the extent caused, contributed, or, once discovered, exacerbated, by any PPC-Related Entity.

Hazardous Materials Management means procedures, practices and activities to address and comply with Environmental Laws and Environmental Approvals with respect to Hazardous Materials encountered, impacted, caused by or occurring in connection with the Project, Project Site or the Work (including demolition Work), including investigation and remediation of such Hazardous Materials. Hazardous Materials Management may include sampling, characterization, stock-piling, storage, backfilling in place, asphalt batching, recycling, treatment, clean-up, remediation, transportation and/or off-site disposal of Hazardous Materials, whichever approach is effective, most cost-efficient and permitted under applicable Law.

<u>Hazardous Waste</u> means waste as defined in 40 CFR Part 261 and/or 22 California Code of Regulations 66260.10.

**<u>HCC Agreement</u>** means an agreement between City and the Housing Project Company (or between City and the Principal Project Company and assigned by Principal Project Company to a Housing Project Company), pursuant to which City will grant a long-term real property interest in certain premises for the development of the Housing and Commercial Component.

<u>Health and Safety Plan</u> means Principal Project Company's plan describing the policies and procedures to be followed by all PPC personnel at the Site under CFR Title 29, CCR Title 8 and other applicable regulations, prepared and submitted by Principal Project Company in accordance with <u>Section 01 35 45</u> of <u>Division 10</u>, as updated and approved by the City in accordance with this Agreement.

**Help Desk** means the contact point to be established by Principal Project Company pursuant to these IFM Specification for the notification of Failure Events and other day to day matters arising in relation to the provision of IFM Services.

<u>Help Desk Services</u> means services described in <u>Section B.17</u> of the IFM Specifications, including methods and systems to receive, track and dispatch work requests, and track IFM Failure Events.

**Housing and Commercial Component** or <u>HCC</u> means the Facility's housing and commercial component, which, if implemented, would include the commercial space, the housing units and their associated support spaces (e.g., lobbies, vertical and horizontal circulation, storage, open space, rooms for building systems, offices for property management and residential services, and resident amenities such as laundry and community rooms) that are not used for the Bus Yard Component or the Common Infrastructure.

**IFM Contract** means each Contract between Principal Project Company and an IFM Provider. There may be more than one IFM Contract concurrently in effect.

**IFM Failure Event** means an incident or state of affairs which does not meet or comply with the IFM Specifications and/or results in an Availability Condition not being met in a Functional Part. An IFM Failure Event will become:

- (a) an Availability Failure, if it causes a Functional Unit or Functional Component (as the case made be) to be Unavailable and the IFM Failure Event is not Rectified within the Rectification Time;
- (b) separate and distinct Service Failures, if the IFM Failure Event is not Responded to within the Response Time or Rectified within the Rectification Time (in the case of IFM Failure Events requiring Rectification which do not constitute a breach of the Availability Conditions); or
- (c) if, in accordance with Performance Measurements Table, no Rectification Time or Response Time applies, the IFM Failure Event shall be a Service Failure.

**IFM Management Plan** means the document titled "IFM Management Plan" that is included in the approved PMP, which is attached as part of <u>Exhibit 2</u> (Project Management Plan). The preliminary "IFM Management Plan" will be replaced with a final approved "IFM Management Plan", which shall be developed and submitted by Principal Project Company and approved by the City pursuant to <u>Section 1.1.8</u> of <u>Division 1</u> of the Technical Requirements and applicable

sections of the IFM Specifications, as updated in accordance with this Agreement.

**IFM Noncompliance Point(s)** means an accrual of points in accordance with <u>Section 6</u> (IFM Noncompliance Points System) of <u>Exhibit 4B</u> (Availability Payment Mechanism).

**IFM Operations Committee** has the meaning set forth in <u>Section B.9.1.2</u> (IFM Operations Committee) of the IFM Specifications.

**IFM Period** means that portion of the Term commencing on the later of the Substantial Completion Date and the Substantial Completion Deadline and ending on the Termination Date.

**IFM Period Deduction(s)** means an Availability Deduction and/or a Service Failure Deduction from the Availability Payment in accordance with <u>Section 3</u> (Deductions from Availability Payments) of <u>Exhibit 4B</u> (Availability Payment Mechanism).

**IFM Project Site** means the real property owned by City, under the jurisdiction of SFMTA, on which the Infrastructure Facility is located as described and shown on the [IFM Site Plan]t. For purposes of insurance (subject to any notification and other requirements imposed by the insurer(s) for approval), indemnification, safety and security requirements, the prevailing wage requirements, and payment for use of equipment, the term "Project Site" shall also include (a) the field office sites, (b) any property used for bonded storage of material for the Project approved by City, (c) staging areas dedicated to the Project, and (d) areas where activities incidental to the Project are being performed by Principal Project Company and Contractors covered by the worker's compensation policy but excluding any permanent locations of Principal Project Company or such covered Contractors.

**IFM Provider** means the Contractor primarily responsible for the IFM Services.

**IFM Quality Management Plan** means the plan prepared and submitted by Principal Project Company and approved by the City pursuant to <u>Section B.14</u> of the IFM Specifications, as updated in accordance with this Agreement.

**IFM Records** means all data in connection with maintenance, operation, renewals and expansion of the Project including (a) all inspection and inventory records, whether generated by Principal Project Company or a Third Party, (b) any communication to and/or from City or a Third Party, and (c) any information system (as may be introduced or amended by City from time to time) in connection with operation, maintenance, renewal or handback of the Project that City requires Principal Project Company to use or operate.

**IFM Services** means all management, engineering, repairs and maintenance, renewals and replacement, and other ancillary services required at all times for the Infrastructure Facility to allow for the ongoing operations and maintenance activities needed at the Infrastructure Facility and to meet the service requirements specified in the IFM Management Plan and the Technical Requirements. IFM Services includes Maintenance Services, Renewal Work, and Technology Enhancements, but excludes D&C Work and SFMTA O&M Services.

**IFM Services Procedures** means the procedures prepared and submitted by Principal Project Company and approved by the City in accordance with <u>Section B.6(3)</u> of the IFM Specifications, as updated in accordance with this Agreement.

**IFM Services Submittals** means all Submittals required to be submitted under this Agreement with respect to the IFM Services.

**IFM Specifications** means <u>Division 7</u> of the Technical Requirements.

**Implementation Proposal** means the Financial Proposal, provided at <u>Exhibit 3A</u> (Financial Proposal); and the specific elements of the following documents included at <u>Exhibit 3B</u> (Technical Proposal Elements): (i) the Lead Developer's design submittals under the Predevelopment Agreement, (ii) the D&C Contractor's design-build proposal for the D&C Contract submitted to Lead Developer in accordance with the Predevelopment Agreement, and (c) the IFM Provider's proposal for the IFM Contract submitted to Lead Developer in accordance with the Predevelopment Agreement.

**Incidental Utility Work** means, for all Utilities, all of the following work necessary for construction of the Project, including any necessary coordination with Utility Owners and property owners, furnishing design, performing construction, and obtaining and complying with all required Regulatory Approvals:

- (a) Service Line relocations;
- (b) The adjustment of Utility appurtenances (e.g., manholes, valve boxes and vaults) for line and grade upon completion of roadway work;
- (c) Protection in Place of Utilities;
- (d) All work necessary to remove and dispose of any Utilities (whether or not in use as of the Effective Date) in situations for which leaving the Utilities in place is not feasible or not permitted, or is required in order to accommodate or permit construction of the Project, regardless of whether replacements for such Utilities are being or have been installed in other locations;
- (e) All work necessary to abandon in place any Utility in accordance with applicable Law and proper Utility Owner and/or industry procedures (e.g., flushing, capping, slurry backfill, etc.) regardless of whether replacements for such Utilities are being or have been installed in other locations;
- (f) Traffic control for Utility Adjustment work;
- Resurfacing and re-striping of streets; reconstruction of curbs, gutters and sidewalks; reinstallation of signage; and reinstallation or replacement of traffic signals;
- Supplemental investigation, potholing, electronic detection, surveying and any other methods used to determine Utility locations, preparation of Principal Project Company's Utility conflict matrix and other material information concerning Utilities;
- (i) Temporary Relocations; and
- (j) Earthwork trenching requested by a Utility Owner.

**Increased Oversight Cure Plan** means the plan including specific actions (including timeframes) to be taken by Principal Project Company, to improve its performance, prepared and submitted by Principal Project Company in accordance with <u>Section 15.6.4(a)</u> of the Agreement, as updated and approved by the City in accordance with this Agreement.

### Increased Oversight Threshold means:

During the D&C Phase;

- Principal Project Company receives a total of 200 or more Noncompliance Points over the course of 12 consecutive calendar months (determined on a rolling basis); or
- Principal Project Company receives a total of 350 or more Noncompliance Points over the course of 36 consecutive calendar months (determined on a rolling basis);

During the IFM Period;

- (c) Principal Project Company receives a total of 1406 or more Noncompliance Points over the course of 12 consecutive calendar months (determined on a rolling basis); or
- (d) Principal Project Company receives a total of 3687 or more Noncompliance Points over the course of 36 consecutive calendar months (determined on a rolling basis); or
- (e) Principal Project Company receives a second Performance Notice, either on a successive basis or for matters related to a prior Performance Notice.

**Indemnitees** means City, including its boards and commissions, other persons and entities designated as "Indemnitees" in the Contract Documents, and all of their respective officeholders, officers, directors, agents, members, consultants, employees, authorized representatives, successors, and assigns.

**Independent Energy Auditor** or **IEA** means a company specializing in energy modelling and reporting that is a licensed, professional engineer in the state of California.

**Independent Engineer** means the independent engineering consultant retained by the Parties in accordance with <u>Section 8.6.2</u> (Independent Engineer).

**Information Management Plan** means Principal Project Company's plan describing procedures to produce and control all documentation and relevant information, prepared and submitted by Principal Project Company in accordance with <u>Section 1.6</u> of <u>Division 1</u> of the Technical Requirements, as updated and approved by the City in accordance with this Agreement.

Infrastructure Facility means the Bus Yard Component and the Common Infrastructure.

**Initial Financing Agreements** means the documents identified in <u>Part A</u> (Initial Financing Agreements) of <u>Exhibit 5A</u> (List of Initial Financing Documents).

Initial Financing Documents means Initial Financing Agreements and Initial Security Documents.

**Initial Project Debt** means the Project Debt originally incurred to finance the Project and Work, in the total face amount at each lien priority, and with the particular Lenders, set forth in the Contract Documents, which Project Debt is evidenced by the Initial Financing Agreements and secured by the Initial Security Documents.

**Initial Schedule** means the schedule attached as <u>Exhibit 19</u> (Initial Schedule), setting out all the relevant D&C Work activities from Commercial Close through Final Acceptance, with key interim milestones during that period of work, and at a minimum at Level 3 defined in AACE 91R-16 "Schedule Development".

**Initial Security Documents** means the documents identified in <u>Part B</u> (Initial Security Documents) of <u>Exhibit 5A</u> (List of Initial Financing Documents).

## Institutional Lender means:

- (a) The United States of America, any state thereof or any agency or instrumentality of either of them, any municipal agency, public benefit corporation or public authority, advancing or insuring mortgage loans or making payments which, in any manner, assist in the financing, development, operation and maintenance of projects;
- (b) Any (i) bank, trust company (whether acting individually or in a fiduciary capacity), savings and loan organization or insurance company organized and existing under the laws of the United States of America or any state thereof, (ii) foreign insurance company or bank qualified to do business as such, as applicable under the laws of the United States of America or any state thereof, or (iii) pension fund, foundation or university or college endowment fund; provided that an entity described in this <u>clause (b)</u> only qualifies as an Institutional Lender if it is subject to the jurisdiction of state and federal courts in the State in any actions;
- (c) Any "qualified institutional buyer" under Rule 144(a) under the Securities Act of 1933, 15 U.S.C. § 77a et seq., or any other similar Law hereinafter enacted that defines a similar category of investors by substantially similar terms;
- (d) Any purchaser of debt securities the proceeds of which are used to finance the Project that are not publicly offered pursuant to the exception to registration provided in Section 4(2) of the U.S. Securities Act of 1933; or
- (e) Any other financial institution or entity designated by Principal Project Company and approved in writing by City; provided that such institution or entity, in its activity under this Agreement, is acceptable under then current guidelines and practices of City;

<u>provided</u>, <u>however</u>, that each such entity (other than entities described in <u>clause (c)</u> and <u>(d)</u> of this definition), or combination of such entities if the Institutional Lender is a combination of such entities, shall have individual or combined assets, as applicable, of not less than \$1 billion. The foregoing dollar minimums shall automatically increase at the beginning of each calendar year by the percentage increase, if any, in the CPI during the immediately preceding calendar year, determined by comparing the CPI most recently published for the immediately preceding year with the CPI most recently published for the second preceding year.

**Insurance Policy(ies)** means all of the insurance policies Principal Project Company is required to carry under Exhibit 7 (Insurance Requirements).

**Insurance Proceeds** means all proceeds from insurance payable to Principal Project Company (or that would have been payable to Principal Project Company but for Principal Project Company's breach of any obligation under this Agreement to procure or maintain said insurance) on or after the Early Termination Date.

Insurance Review Report means the report furnished in accordance with Section 10.1.3(b).

Insurance Unavailability means either:

- (a) Any Insurance Policy coverage required under <u>Section 10.1</u> (Insurance) or <u>Exhibit 7</u> (Insurance Requirements) is deemed unavailable under <u>Section 10.1.4</u> (Insurance Unavailability); or
- (b) Provision of all such Insurance Policy coverages has become unavailable at Commercially Reasonable Insurance Rates from insurers meeting the requirements in <u>Section 10.1.2.1</u> (Insurers).

For the purpose of <u>clause (b)</u>, the only increases in premiums that may be considered are those caused by changes in general market conditions in the insurance industry affecting insurance for project-financed transit facilities, and Principal Project Company shall bear the burden of proving that premium increases are the result of such changes in general market conditions. No increase in insurance premiums attributable to particular conditions of the Project or Project Site, or to claims or loss experience of any PPC-Related Entity or Affiliate, whether under an Insurance Policy required to be placed under this Agreement or in connection with any unrelated work or activity of PPC-Related Entities or Affiliates, shall be considered in determining whether Insurance Unavailability exists or has occurred.

**Intellectual Property** means all current and future legal and/or equitable rights and interests, anywhere in the world, in know-how, patents (including applications), copyrights (including moral rights), trademarks (registered and unregistered), service marks, trade secrets (as defined by the Defend Trade Secrets Act § 2(b)(1) (18 USC § 1839(3)), and pursuant to US state and federal laws), designs (registered and unregistered), utility models, circuit layouts, mask works, business and domain names, inventions, solutions embodied in technology, and other intellectual activity, and applications of or for any of the foregoing, subsisting in or relating to the Work or IP Materials. Without limiting the foregoing, Intellectual Property includes Software and City Data. For the avoidance of doubt, Intellectual Property is distinguished from the physical, electronic, and/or mechanical embodiments of such Intellectual Property (see IP Materials).

**IP Escrow(s)** has the meaning set forth in <u>Section 21.5.2</u> of this Agreement.

**IP Escrow Agent** has the meaning set forth in <u>Section 21.5.2</u> of this Agreement.

**IP Materials** means all physical, electronic and/or mechanical embodiments of, and documents disclosing, Intellectual Property. Without limiting the generality of the foregoing, IP Materials include embodiments, documents, deliverables and/or Work incorporating concepts, inventions (whether or not protected under patent laws), works of authorship, information, new or useful art, combinations, discoveries, formulae, algorithms, specifications, manufacturing techniques, technical developments, systems, computer architecture, artwork, Software, Source Code, decompilation instructions, programming, applets, scripts, designs, procedures, processes, and methods of doing business, and any other media, materials, plans, reports, project plans, work plans, documentation, training materials, and other tangible objects produced under the Contract or required by, incorporated into or combined with the Work or the Project.

**IT / Comms Site** means those areas related to City-Furnished IT/Comms FF&E, including locations of each MDF, IDF, wireless access point, RTLS point, a 15-foot radius immediately around each of these locations, and other areas the City, acting reasonably, deems necessary and identifies for PPC at least 14 days prior to the beginning of the City Access Period.

<u>JDA Project Company(ies)</u> means an entity, including a special-purpose entity(ies), that will undertake a Joint Development Alternative pursuant to an amendment to this Agreement or a separate agreement, including an HCC Agreement (with respect to the HCC).

#### JDA-Related Entity(ies) means:

- (a) each JDA Project Company;
- (b) any other persons or entities performing any work relating to a JDA;
- (c) any Affiliate of the Principal Project Company and any affiliate of any other PPC-Related Entity, any JDA Project Company and any other JDA-Related Entity participating in or responsible for any aspect of a JDA;
- (d) any other persons or entities for which a JDA Project Company may be legally or contractually responsible;
- (e) any non-City party to any separate agreement relating to a JDA, including an HCC Agreement, including all subcontractors, subconsultants and suppliers of any tier; and
- (f) the employees, personnel, officers, directors, agents, representatives, successors and assigns of any of the foregoing.

**Joint Development Alternative** or **JDA** means the design, construction, operation and/or maintenance of (i) an HCC; (ii) the Paratransit Facility or (iii) any other project or portion of the Facility for an SFMTA use, if and to the extent implemented.

Joint Technical Review has the meaning set forth in Section 8.5.4.1.

Joint Technical Review Plan has the meaning set forth in Section Error! Reference source not found.

Joint Technical Review Report has the meaning set forth in <u>Section Error! Reference source</u> not found.

Key Contract means any one of the following:

- (a) all Prime Contracts for Construction Work and Design Work;
- (b) all Prime Contracts for IFM Services;
- (c) all Prime Contracts for project or program management services; and
- (d) all other Prime Contracts with a single Contractor (or a single Contractor and its affiliates) that individually or in the aggregate total in excess of \$25 million on a term (not annual) basis.

The term "Key Contracts" means all such Contracts in the aggregate or more than one of such Contracts.

Key Contractor means each Contractor under any Key Contract.

**<u>Key Personnel</u>** means, collectively, those individuals: (a) in <u>Exhibit 20</u> (List of Key Personnel); and (b) appointed by Principal Project Company and approved by City from time to time to fill the "Key Personnel" positions identified in <u>Section 1.1.4</u> of <u>Division 1</u> of the Technical Requirements (each is a "**Key Person**").

<u>Key Ratios</u> means ratios contained in the Financing Documents that have financial covenants attached to them or ratios that are condition precedents to Financial Close under the Financing Agreements.

Known or Suspected Hazardous Materials means Hazardous Materials (i) that are known or reasonably suspected to exist as of the Setting Date based on information, data or analysis contained or referenced in the Reference Documents as of the Setting Date; (ii) excluding Hazardous Materials that are entirely subsurface, which should have been known or reasonably suspected pursuant to a Reasonable Investigation prior to the Setting Date, including those listed in the Technical Requirements; (iii) are identified or referenced in any of the information, data, analyses or reports undertaken by Principal Project Company pursuant to the Predevelopment Agreement; (iv) constituting asbestos, lead, PCBs, any byproducts of the foregoing and/or any other material or substance indicated as present or potentially present at, \_ [PNC to list/identify the haz mat in, on or under the Project Site in that \_ report(s) it has prepared during PDA Term]; (v) anticipated or likely to be present in a building or structure that was constructed in 1915 and in continuous operation as a bus maintenance facility and heavy industrial use since that time; and/or (vi) excluding Hazardous Materials that are entirely subsurface, which should have been identified or referenced pursuant to a Reasonable Investigation undertaken pursuant to the Predevelopment Agreement. Known or Suspected Hazardous Materials include Hazardous Materials arising in or from any of the Hazardous Materials sites listed in the CEQA documents, the NEPA documents or in the Technical Requirements.

<u>Late Payment Rate</u> means the legal rate, as set forth in Section 685.010(a) of the California Code of Civil Procedure, as may be amended from time to time, up to a maximum amount equal to 10% per annum.

**Law(s)** means (a) any statute, law, code, regulation, ordinance, rule or common law, (b) any binding judgment (other than regarding a Claim), (c) any binding judicial or administrative writ, order, judgment, injunction, award or decree (other than regarding a Claim), (d) any written directive, guideline, policy requirement or other governmental restriction (including those resulting from the initiative or referendum process, but excluding those by City within the scope of its administration of the Contract Documents) or (e) any similar form of decision of or determination by, or any written interpretation or administration of any of the foregoing by, any Governmental Entity, in each case which is applicable to or has an impact on the Project or the Work, whether taking effect before or after the Effective Date. The term "Laws", however, excludes Regulatory Approvals.

**Lead Developer** means and Potrero Neighborhood Collective LLC, a limited liability company organized under the laws of the State of Delaware.

**Lender(s)** means each of the holders and beneficiaries of Security Documents, including any financial guarantor providing Project Debt or any guaranty or credit enhancement in respect thereof, and their respective successors, assigns, participating parties, trustees and agents, including the Collateral Agent.

Lenders' Liabilities means, at the relevant time, the aggregate of (without double counting) all principal, interest (including capitalized and default interest under the Financing Documents, but with respect to default interest, only to the extent that it arises as a result of City making any payment later than the date that it is due under this Agreement or any other default by City under this Agreement), Breakage Costs, banking fees, premiums or reimbursement obligations with respect to financial insurance policies, agent and trustee fees, costs and expenses properly incurred owing or outstanding to the Lenders by Principal Project Company under or pursuant to the Financing Documents on the Early Termination Date, including any prepayment premiums or penalties, make-whole payments or other prepayment amounts, including costs of early termination of interest rate and inflation rate hedging, swap, collar or cap arrangements, that Principal Project Company must pay, or that may be payable or credited to Principal Project Company, under any Financing Agreement or Security Document or otherwise as a result of the payment, redemption or acceleration of all or any portion of the principal amount of Project Debt prior to its scheduled payment date that were determined to be reasonable by City at the time City reviewed and approved the Financing Agreements in connection with Financial Close, as evidenced by City's approval of such Financing Agreements.

**Lenders' Technical Advisor (LTA)** means the Lenders' technical advisor, [\_\_\_\_\_], and any replacement technical advisor engaged by Principal Project Company and the Lenders with respect to the Project.

**Long Stop Date** means 12 calendar months after the Substantial Completion Deadline, as such date may be extended from time to time pursuant to <u>Section 14.1.2</u> (Extension of Deadlines for General Delay Events).

**Loss(es)** means, whether asserted, suffered or incurred by a Party or a Third Party, any debt, assessment, claim, action, loss, proceeding, damage, injury, liability, obligation, cost, response

cost, expense (including attorneys', accountants' and expert witnesses' fees and expenses (including those incurred in connection with the enforcement of any indemnity or other provision of this Agreement), loss (whether direct or indirect), proceedings, demands and charges whether arising under statute, contract or at common law), compensation, charge or liability of any kind (including fees, judgments, or penalties), and whether or not arising under or for breach of contract, in tort (including negligence), restitution, under statute or otherwise at law. Losses include injury to or death of persons, damage or loss of property, and harm or damage to natural resources, utility facilities or Intellectual Property.

**LTA Agreement** means the technical advisory agreement dated as of the Financial Close Date between Principal Project Company, the Lenders (or the Collateral Agent) and Lenders' Technical Advisor.

<u>Maintenance and Transit Spaces</u> means the areas described as Maintenance and Transit Spaces and FF&E, as set forth in in Section A.2.1.2 of the IFM Specifications, and identified as Parking, Bays and Shops, Fare Box & Clipper Card Reader Repair Shop, Service & Clean, Parts, Maintenance, Operations, Transit Services (MRO) and enclosed circulation between in the Division 3 of the Technical Requirements.

<u>Major Approval</u> means the site permit issued by the City and County of San Francisco Department of Building Inspection, acting in its regulatory capacity.

## Major Approval Conditions to Assistance has the meaning set forth in Section 6.3.8.2.

<u>Major Approval Deadline</u> means the following time period and deadline for Principal Project Company to secure each Major Approval:

# [Note: PNC to propose time period applicable to the Major Approval for City review and approval, in its sole discretion]

Each such time period:

- (a) begins on the date the first Major Approval request is submitted to the applicable Governmental Entity after the last to occur of (x) Principal Project Company has submitted a fully compliant and complete request and application to the applicable Governmental Entity in accordance with applicable Law and the policies and procedures of such Governmental Entity; (y) Principal Project Company has paid in full all fees and charges required for the review, consultation and issuance of the Major Approval by the applicable Governmental Entity; and (z) Principal Project Company has satisfied all other conditions to the commencement of review by the applicable Governmental Entity in accordance with applicable Law and the policies and procedures of such Governmental Entity; and ends on the date that Governmental Entity's approval is received; and
- (b) includes any Governmental Entity comment requirements and Principal Project Company resubmissions.

<u>Major Approval Delay</u> means a delay to the Work resulting from the failure to obtain a Major Approval by the applicable Major Approval Deadline where that delay is not due, in whole or in part, to a change to the Principal Project Company's design between the Effective Date and Final Design Documents (unless such design change is as a direct and sole result of a City

Change, Compensable Delay Event or Unavoidable Delay Event (excluding Major Approval Delay)) or due to any act or omission of any PPC-Related Entity or any PPC Fault (including any failure to provide City with an updated Project Schedule or to perform the Work in accordance with the Project Schedule).

Major Utility Adjustment means the following Utility Adjustments:

## [Note: PNC to propose <u>limited</u> list of Utility Adjustments that City would provide time / Financing Delay Costs for City review and approval, in its sole discretion.]

<u>Major Utility Adjustment Deadline</u> means the deadline to complete each Major Utility Adjustments specified below:

[Note: PNC to propose time period for the above limited list of Utility Adjustments for City review and approval, in its sole discretion. Time period is to commence after submittal of complete, compliant Utility Adjustment package, design, etc. to Utility. With respect to any Utility Adjustment proposed, if included, this needs to be based on schedule that PNC has worked out with the affected Utility plus a reasonable contingency/buffer]

**Major Utility Adjustment Delay** means a delay to the Utility Adjustment Work resulting from a failure by a Utility Owner to complete a Major Utility Adjustment by the applicable Major Utility Adjustment Deadline where that delay is not due, in whole or in part, to a change to the Principal Project Company's design between the Effective Date and Final Design Documents (unless such design change is as a direct and sole result of a City Change, Compensable Delay Event or Unavoidable Delay Event (excluding Major Utility Adjustment Delay)) or due to any act or omission by any PPC-Related Entity or any PPC Fault or failure of the Principle Project Company to act diligently in accordance with the requirements of the Contract Documents and Good Industry Practice (including any failure to provide City with an updated Project Schedule or to perform the Work in accordance with the Project Schedule).

**Main or Trunkline Utility** means an underground Utility, which is not a Service Line, and which relative to the particular system of which it is a part, (a) is a larger line serving as a main line to connecting tributary lines and (b) serves a larger area, all as determined by City, in its sole discretion. In determining whether a facility should be considered a Main or Trunkline Utility, City may refer to definitions in the relevant manual or code, if any, of the Utility Owner.

<u>Maintained Elements</u> means the Infrastructure Facility and the Project Site, including all equipment, component, finishes and any other elements required to satisfy the requirements of the Agreement, including all constructed maintained elements, building systems, cabling material, equipment, finishes, furniture and fixtures, excluding the SFMTA O&M Facilities.

<u>Maintenance and Transit Spaces</u> means the areas described as Maintenance and Transit Spaces and FF&E, as set forth in in <u>Section A.2</u> of the IFM Specifications, and identified as Parking, Bays and Shops, Fare Box & Clipper Card Reader Repair Shop, Service & Clean, Parts, Maintenance, Operations, and Transit Services (MRO) in <u>Division 3</u> of the Technical Requirements.

Maintenance of Traffic (MOT) means the comprehensive effort to maintain traffic.

<u>Maintenance Plan</u> means the plan to properly monitor and maintain the landscaped areas, prepared and submitted by Principal Project Company and approved by the City in accordance with <u>Section C.4.3</u> of the IFM Specifications, as updated in accordance with this Agreement.

<u>Maintenance Rectification Costs</u> means, in respect of any termination of this Agreement that occurs after Substantial Completion, all Losses that City determines it is reasonably likely to incur as a direct result of the termination of this Agreement, including (without double counting):

- (a) those costs (internal and external) that City is reasonably likely to incur as a direct result of carrying out any process to request tenders from any parties interested in entering into a contract with City to carry out the IFM Services, including all costs related to the preparation of tender documentation, evaluation of tenders and negotiation, and execution of relevant contracts; and
- (b) those costs reasonably projected to be incurred by City in relation to:
  - (i) remediation or, if remediation is not possible or would cost more than renewal, renewal of any defective D&C Work or IFM Services;
  - (ii) rectification or cure of any breach of this Agreement by Principal Project Company; and
  - (iii) carrying out of all other matters necessary in order to ensure that within a reasonable period of the Early Termination Date, the Project complies with the requirements of the Contract Documents, and has a reasonable prospect of continuing to perform to the same standard and cost that it would have continued to perform at had this Agreement not been terminated and the Project been in compliance with all of the requirements of the Contract Documents,

including, for the avoidance of doubt, any amount which, but for the termination of this Agreement, either should have been deposited with or paid to City in accordance with the terms of this Agreement.

<u>Maintenance Services</u> means all Work during the IFM Period to maintain, repair, preserve, replace, refurbish, test, commission and modify the Project, including the supply of machinery, equipment, materials, hardware, software, systems or any other items related to such Work, and including Demand Maintenance, Scheduled Maintenance and Renewal Work, but excluding (a) D&C Work remaining to be performed following Substantial Completion, and (b) SFMTA O&M Services.

<u>Major Service Failure</u> means a Service Failure which has been designated in the Technical Requirements or <u>Exhibit 4</u> (Payment Mechanism) as a Major Service Failure.

<u>Manual</u> or <u>Manuals</u> means policies, procedures, practices, guidelines, plans, checklists, deliverables, etc. that are developed, implemented and updated by Principal Project Company.

<u>Maximum Availability Payment (MaxAP)</u> means the maximum annual Availability Payment that Principal Project Company can earn in a given Contract Year during the IFM Period, as

calculated in accordance with <u>Exhibit 4B</u> (Availability Payment Mechanism), and as may be further adjusted in accordance with the Contract Documents.

<u>Maximum Quarterly Availability Payment (MaxQAP)</u> means the maximum Quarterly Availability Payment that Principal Project Company can earn in a given Contract Quarter during the IFM Period, as calculated in accordance with <u>Exhibit 4B</u> (Availability Payment Mechanism), and as may be further adjusted in accordance with the Contract Documents.

<u>Medium Service Failure</u> means a Service Failure which has been designated in the Technical Requirements or <u>Exhibit 4</u> (Payment Mechanism) as a Medium Service Failure.

<u>Meet and Greet</u> means the Functional Components corresponding to Functional Unit references GL-025 as specified in <u>Table B5</u> (Functional Components and Functional Units for Infrastructure Facility) of <u>Appendix B</u> (Functional Areas, Functional Units, Functional Unit Rankings and Deduction Amounts for Availability Failures and Service Failures) of <u>Exhibit 4B</u> (Availability Payment Mechanism).

<u>Milestone Payment</u> has the meaning set forth in <u>Section 1(a)</u> of <u>Exhibit 4A</u> (Milestone Payment Mechanism).

<u>Minimum Agreed Availability Condition</u> means all of the Accessibility Conditions, the Safety Conditions and the Use Conditions, as temporarily modified as permitted in accordance with <u>Exhibit 4B</u> (Availability Payment Mechanism) for the purposes of a Temporary Repair.

<u>Minor Service Failure</u> means a Service Failure which has been designated in the Technical Requirements or <u>Exhibit 4</u> (Payment Mechanism) as a Minor Service Failure.

Model Update Event has the meaning set forth in Section 4.7.1.1.

<u>Monthly Progress Meeting</u> has the meaning set forth in <u>Section 1.1.3.7</u> of <u>Division 1</u> of the Technical Requirements.

<u>Monthly Progress Status Report</u> has the meaning set forth in <u>Section 1.2.4.1</u> of <u>Division 1</u> of the Technical Requirements.

<u>Monthly Report</u> means each of the reports described in <u>Section 1.2.4.1</u> of <u>Division 1</u> and <u>Section B.11</u> of the IFM Specifications.

**Move-In** has the meaning set forth in <u>Section 7.14</u> (Move-In).

<u>Move-In Plan</u> means the plan prepared and submitted by Principal Project Company and approved by the City in accordance with <u>Section B.25</u> of the IFM Specifications, as updated in accordance with this Agreement.

Move-In Resource Candidate has the meaning set forth in Section 7.14.3.1.

<u>Move-In Resource</u> means the advisor selected in accordance with Section <u>Error! Reference</u> <u>source not found.</u>

Move-In Subcommittee has the meaning set forth in Section 7.14.2.1.

NEPA Approval means the	dated [	] and available at
-------------------------	---------	--------------------

#### **NEPA Event** means:

- (a) any new or modified NEPA Approval necessitated solely by a City Change, a Delay Event or Relief Event;
- (b) legal action being taken in respect of the NEPA Approval that results in a temporary restraining order, preliminary injunction or other form of interlocutory relief by a court of competent jurisdiction that prohibits prosecution of, or by complying with such temporary restraining order, preliminary injunction or other form of interlocutory relief by a court of competent jurisdiction results in prohibiting the prosecution of, a material portion of the Work;
- (c) review or revocation or material change to, the NEPA Approval;
- (d) any review or revocation of, or change to, a NEPA Approval directly resulting from the circumstances specified in <u>clauses (b)</u> and <u>(c)</u>; or

except, in each case, to the extent resulting, in whole or in part, from Principal Project Company's design, Work or from any PPC Fault.

**Noncompliance Database** has the meaning set forth in <u>Section 15.3.1.1</u>.

**Noncompliance Event** means a D&C Noncompliance Event, Service Failure or Unavailability Failure.

#### Non-Core Hours means:

- (a) for Maintenance and Transit Spaces, commencing at 10:01 pm until 3:59 am every day of every week; and
- (b) for Office/Admin and Training Spaces, commencing at 5:01 pm until 7:59 am, Monday to Friday of every week.

**Non-Dig Alert Utilities** means Utilities that are not documented by or recorded with USA Dig Alert.

<u>Noncompliance Points</u> means the points that may be assessed in accordance with <u>Section 15.4</u> (Assessment of Noncompliance Points).

**Nonconforming Work** means any D&C Work (including any product of the D&C Work) that does not conform to the requirements of the Contract Documents, the Regulatory Approvals,

applicable Law, the Design Documents or the Construction Documents, including any Work required to be repaired or replaced under <u>Section 7.11</u> (Nonconforming Work) of this Agreement.

**Non-Digalert Utility** means a Utility owned or operated by a Utility Owner that is not a member of the California Digalert organization/notification process. "Non-Digalert Utilities" are not "Unidentified Utilities."

**Non-Regulated Load** means the metered provision of Energy other than Regulated Load. Non-Regulated Loads are to be aggregated in a logical fashion through a submetering system.

**Non-Revenue Vehicle Parking Space** means the Functional Components corresponding to Functional Unit references GL-025 as specified in <u>Table B5</u> (Functional Components and Functional Units for Infrastructure Facility) of <u>Appendix B</u> (Functional Areas, Functional Units, Functional Unit Rankings and Deduction Amounts for Availability Failures and Service Failures) of <u>Exhibit 4B</u> (Availability Payment Mechanism).

**Notice** has the meaning given in <u>Section 23.10</u> (Notices and Communications).

**Notice(s) of Conditional Termination** has the meaning set forth in <u>Section 17.2.1</u> (Notice of Conditional Election to Terminate – Force Majeure Termination Events).

Notice of Determination has the meaning set forth in Section 15.3.3.2.

<u>Notice of Potential Claim</u> means a written Notice submitted by Principal Project Company in accordance with <u>Section 18.2</u> (Notice of Potential Claim).

**Notice of PPC Default** means a written Notice provided by City concerning City's determination that a PPC Default has occurred, with respect to any default for which a cure period is allowed under <u>Section 16.3.2</u> (Cure Periods).

<u>NTP 1</u> means the written Notice issued by City to Principal Project Company in accordance with <u>Section 7.4.3</u> (Commencement of Non-Construction Work) authorizing Principal Project Company to proceed with non-Construction Work.

<u>NTP 2</u> means the written Notice issued by City to Principal Project Company in accordance with <u>Section 7.4.4</u> (Commencement of Construction Work) authorizing Principal Project Company to proceed with Construction Work.

<u>Office/Admin and Training Spaces FF&E</u> means all FF&E to be included in Office/Admin and Training Spaces and all additional FF&E identified as "Office/Admin and Training Spaces FF&E" in the Equipment List or otherwise required to be included in the Office/ Admin and Training Spaces under <u>Section 7.13.1</u> (Selection and Procurement of PPC-Furnished FF&E).

<u>Office/ Admin and Training Spaces</u> means administration and training facility spaces including Office/Admin and Training Spaces FF&E, as set forth in <u>Section A.2</u> of the IFM Specifications and identified as "Office Modules", "Training" and "Operations" and enclosed circulation between in <u>Division 3</u> of the Technical Requirements.

**Open Book Basis** means review by City of documentation showing Principal Project Company's underlying assumptions, documents, information and data associated with the issue in question, including assumptions as to the Financial Model and each updated Financial Model, pricing or compensation or adjustments thereto, costs of the Work, costs or compensation claimed with respect to Model Update Events, schedule, composition of equipment spreads, equipment rates (including rental rates), labor rates and benefits, productivity, estimating factors, design and productivity allowance, contingency and indirect costs, Extra Work Costs, risk pricing, discount rates, interest rates, inflation and deflation rates, swap and hedge rates, insurance rates, bonding rates, letter of credit fees, overhead, profit, traffic volumes and other items reasonably required by City to satisfy itself as to validity or reasonableness.

Ordinance has the meaning set forth in Recital D.

**<u>Original Equity IRR</u>** means [•]%, which is the Equity IRR listed in the Base Case Financial Model.

Other Adjustments (OA) means the aggregate of any other adjustments for a Contract Quarter to reflect previous over-payments and/or under-payments, any interest payable in respect of any amounts owed, and any other amount due and payable from Principal Project Company to the City or from the City to Principal Project Company under this Agreement, as set out in <u>Section</u> <u>11.9</u> (Other Adjustments; Full Compensation).

Other Amounts has the meaning given in Section 11.4.2.

<u>Other Contractor</u> means any contractor, tradesperson or other person engaged by City to do work other than Principal Project Company and its Subcontractors.

**Oversight** means monitoring, inspecting, sampling, measuring, spot-checking, reviewing, attending, observing, testing, investigating, auditing and conducting any other ongoing oversight respecting any part or aspect of the Project or the Work, by any Person so entitled, including all the activities described in <u>Section 5.4.2</u>.

**Paratransit Facility** means the paratransit component which will (a) include the spaces needed for paratransit operation and maintenance activities at the Paratransit Facility, (b) reside on the roof of the bus yard component with access and egress from vehicular ramps extending from the Bus Yard Component onto the Bus Yard Component podium roof, (c) must meet the paratransit criteria in the Technical Requirements, (c) delivered pursuant to a Change Order or by Principal Project Company or another entity under a separate agreement. Paratransit Facility would be developed as Joint Development Alternative 2 Paratransit Component, as described in the "Project Description" included in <u>Division 1</u> of the Technical Requirements.

**Partnering** means the partnering process as set forth in <u>Article 18</u> of this Agreement.

<u>**Party</u>** means Principal Project Company or City, as the context may require, and "Parties" means Principal Project Company and City, collectively.</u>

**Pass-Through Costs** means the cost of diesel incurred by PPC directly and paid to utility owners with respect to PPC's performance of the IFM Services consistent with Section A. 2.1.1 of the IFM Specifications for operations and management of back-up power systems and Section 11.3.

**Payment Bond** means the payment bond(s) to secure payment for labor and materials, as required under this Agreement.

**Payment Commencement Date** means the date that is one day after the Substantial Completion Date.

**PDA-Related Costs** means the following costs incurred by or accruing to City under the PDA:

- (a) The reasonably required costs of any assistance, action, activity or work undertaken by City which Principal Project Company is liable for or is obligated to reimburse City for under the terms of the PDA, including the charges of third party contractors and reasonably allocated wages, salaries, compensation and overhead of City staff and employees performing such action, activity or work; plus
- (b) Reasonably required out-of-pocket costs City incurs to publicly procure any such third party contractors; plus
- (c) Reasonable fees and costs of attorneys (including the reasonably allocable fees and costs of the Office of the City Attorney), financial advisors, engineers, architects, insurance brokers and advisors, investigators, traffic and revenue consultants, risk management consultants, other consultants, and expert witnesses, as well as court costs and other litigation costs, in connection with any such assistance, action, activity or work, including in connection with defending claims by and resolving disputes with third party contractors; plus
- (d) Interest on all the foregoing sums at the Late Payment Rate, commencing on the date due under the applicable terms of the PDA and continuing until paid; plus
- (e) An administrative fee of 10% of the actual aggregate costs incurred by City as described in <u>clauses (a)</u> through <u>(d)</u> of this definition.

**PDA Term** means the period commencing on November 2, 2022 and expiring on the Effective Date.

**Performance Action Plan** or **PAP** means the plan, at minimum, containing a summary of the issues raised by City, an analysis of the root causes of the issues, the steps and timeframes for such steps to be undertaken by Principal Project Company to address the issues, prepared and submitted by Principal Project Company in accordance with <u>Section B.18.2</u> of the IFM Specifications, as updated and approved by the City in accordance with this Agreement.

<u>**Performance Bond**</u> means the performance bond(s) securing performance of the Work required under this Agreement.

**Performance Cure Plan** means the plan prepared and submitted by Principal Project Company and approved by the City in accordance with <u>Section 15.5.2(a)</u> of the Agreement, as updated in accordance with this Agreement.

<u>Performance Measure</u> has the meaning given in the Performance Measurements Table.

#### **Performance Measurements Table** means the table in <u>Section G</u> of the IFM Specifications.

<u>Performance Monitoring</u> means a detailed inspection and/or monitoring of the Project by Principal Project Company during the IFM Period in accordance with the Technical Requirements, including the IFM Specifications, to verify compliance with the Contract Documents.

<u>Performance Monitoring Report</u> means a report documenting all Failure Events, associated Noncompliance Points and Deductions, and showing space temperatures during period, including identification of rooms that have not met the requirements of the Technical Requirements, including the IFM Specifications.

Performance Notice means a written Notice provided by City, as set forth in Section 15.5.1.

#### Performance Notice Threshold means

- (a) if Principal Project Company receives a total of 375 or more Noncompliance Points over the course of 3 consecutive calendar months (determined on a rolling basis); or
- (b) A material Noncompliance Event occurs, as determined by the City

**<u>Permanent Repair</u>** means Rectification where a Temporary Repair has been permitted and carried out pursuant to <u>Section 9</u> (Amendment of IFM Period Performance Measures Table) of <u>Exhibit 4B</u> (Availability Payment Mechanism).

**Permanent Repair Deadline** has the meaning given in <u>Section 8(b)</u> of <u>Exhibit 4B</u> (Availability Payment).

#### Persistent PPC Default means if:

During the D&C Phase;

- Principal Project Company receives a total of 500 or more Noncompliance Points over the course of 12 consecutive calendar months (determined on a rolling basis); or
- Principal Project Company receives a total of 875 or more Noncompliance Points over the course of 36 consecutive calendar months (determined on a rolling basis);

During the IFM Period;

- (a) Principal Project Company receives a total of 3867 or more Noncompliance Points over the course of 12 consecutive calendar months (determined on a rolling basis); or
- (b) Principal Project Company receives a total of 10,635 or more Noncompliance Points over the course of 36 consecutive calendar months (determined on a rolling basis); or

(c) 58,005 or more cumulative Noncompliance Points or have occurred over the course of the IFM Period.

<u>Persistent PPC Default Cure Plan</u> means the plan including specific actions (including timeframes) to be taken by Principal Project Company to improve performance, prepared and submitted by Principal Project Company in accordance with <u>Section 15.8.1</u> of the Agreement, as updated and approved by the City in accordance with this Agreement.

**Person** means any individual, corporation, joint venture, limited liability company, company, voluntary association, partnership, trust, unincorporated organization or Governmental Entity.

Planning Department means the City's Planning Department.

<u>Post-Refinancing Financial Model</u> has the meaning set forth in <u>Section 1.2</u> (Create a Post-Refinancing Financial Model) of <u>Exhibit 5C</u> (Calculation of Refinancing Gain).

Potential Contract Dispute has the meaning set forth in Section 18.2.1.

**PNC** has the meaning set forth in <u>Recital E</u>.

**PPC Change Request** has the meaning set forth in <u>Section 3.1</u> of <u>Exhibit 9</u> (Change Procedures).

**PPC Default** has the meaning set forth in <u>Section 16.1.1</u> (PPC Default).

**PPC Employee(s)** means employee(s) of Principal Project Company or any other PPC-Related Entity at the Project Site.

#### PPC Employee and Contractor Breakage Costs means:

- the payment of all wages earned, accrued unused vacation time, and any other payments required to be made by Principal Project Company to its employees under law, or under the terms and conditions of Principal Project Company's employment agreements with its employees as a direct result of the termination of this Agreement;
- (b) Losses that have been or will be reasonably and properly incurred by Principal Project Company under a Key Contract as a direct result of the termination of this Agreement (and which shall not include lost profit or lost opportunity), but only to the extent that:
  - (i) the Losses are incurred in connection with the Project and with respect to the Work required to be provided or carried out, including:
    - (1) any materials or goods ordered or subcontracts placed that cannot be cancelled without such Losses being incurred;
    - (2) any expenditure incurred in anticipation of the provision of services or the completion of Work in the future; and

- (3) the cost of demobilization including the cost of any relocation of equipment used in connection with the Project; and
- (ii) the Losses are reasonably and properly incurred under arrangements and/or agreements that are consistent with terms that have been entered into in the ordinary course of business and on an arm's length basis; and
- (c) Principal Project Company and the relevant Key Contractor have each used their reasonable efforts to mitigate such Losses.

#### PPC Fault means:

- (a) the breach by any PPC-Related Entity of any of its obligations or any representation, warranty or covenant under the Contract Documents;
- (b) a breach of applicable Law or any Regulatory Approval by any PPC-Related Entity; and
- (c) fraud, criminal conduct, intentional misconduct, recklessness, bad faith, gross negligence, negligence or other culpable act or omission of any PPC-Related Entity.

**PPC Intellectual Property** means Intellectual Property developed by Principal Project Company, Contractors, or any of their respective Affiliates either (a) prior to the Effective Date, (b) independently of the Contract Documents or (c) any adaptation, continuation or derivative work which requires the incorporation, exercise or practice of Intellectual Property that is the subject of either subsection (a) or (b).

**PPC Release** means, with respect to Hazardous Materials, (a) any Release of Hazardous Material, or the exacerbation of any such Release, attributable to any PPC Fault; (b) any Release of Hazardous Materials arranged to be brought onto the Project Site by any PPC-Related Entity; regardless of cause; (c) any migration of Hazardous Materials into, onto, under or from the Project Site where the source of such Hazardous Materials is a PPC-Related Entity; or (d) any use, containment, storage, management, handling, transport or disposal of any Hazardous Materials, by any PPC-Related Entity in violation of the requirements of the Contract Documents, Good Industry Practice or any applicable Law or Regulatory Approval.

<u>PPC Resources</u> means the employees of Principal Project Company or any Principal Project Company that are not at the Project Site.

**PPC's Construction Quality Plan (PCQP)** means the Principal Project Company deliverable described in <u>Section 1.4.3</u> of <u>Division 1</u> of the Technical Requirements, following approval thereof by City, as updated in accordance with this Agreement.

**PPC's Design Quality Plan (PDQP)** means the Principal Project Company deliverable described in <u>Section 1.4.2</u> of <u>Division 1</u> of the Technical Requirements, following approval by City, as updated in accordance with this Agreement.

**PPC-Furnished FF&E** means all FF&E required by the Technical Requirements or the Equipment List (including items marked CP) or otherwise required to be included on the

Equipment List under <u>Section 7.13.1</u> (Selection and Procurement of PPC-Furnished FF&E) including the Office/Admin and Training Spaces FF&E and the IT/Comms Equipment, but excluding City-Furnished Equipment and Existing FF&E.

<u>PPC's Interest</u> means all right, title and interest of Principal Project Company in, to, under or derived from the Contract Documents, including the Infrastructure Facility, Project Management Plan, Subcontracts, Submittals, Claims and Intellectual Property.

**PPC-Maintained FF&E** means the PPC-Furnished FF&E, excluding the Office/Admin and Training Spaces FF&E and the IT/Comms Equipment.

**PPC's Quality Program (PQP)** means the program described in <u>Section 1.4.1</u> of <u>Division 1</u> of the Technical Requirements, including all sub-plans.

**PPC's Quality Program Plan (PQPP)** means Principal Project Company's plan describing its approach to undertake and achieve the requirements in <u>Section 1.4.1</u> of <u>Division 1</u> of the Technical Requirements, prepared and submitted by Principal Project Company in accordance with <u>Section 1.4</u> of <u>Division 1</u> of the Technical Requirements, as updated and approved by the City in accordance with this Agreement.

## PPC-Related Entity(ies) means:

- (a) Principal Project Company;
- (b) Principal Project Company's Equity Members, including Lead Developer;
- (c) Contractors;
- (d) any other persons or entities performing any of the Work;
- (e) any other persons or entities for whom Principal Project Company may be legally or contractually responsible;
- (f) any Affiliate of the Principal Project Company, or any other PPC-Related Entity (excluding such Affiliates to the extent that they are participating in or responsible for any aspect of a Joint Development Alternative); and
- (g) the employees, agents, officers, directors, representatives, consultants, successors and assigns of any of the foregoing.

**PPC Spaces** has the meaning given in Section 2.1.4 of the IFM Specifications.

**Predevelopment Agreement** has the meaning set forth in <u>Recital F</u> of this Agreement.

Pre-Existing Hazardous Materials means Hazardous Materials that:

(a) are located in, on or under, or are emanating from, any parcel within the boundaries of the Project Site as of the date access to such parcel is provided to Principal Project Company; or

(b) existed in another location as of the date access to such a parcel was provided and thereafter migrated to such parcel,

excluding any Hazardous Materials that are located in, on or under, or are emanating from any Additional Property or Temporary Areas or which arise as a result of any act or omission of any PPC-Related Entity in connection with any Additional Property or Temporary Areas.

<u>Pre-Refinancing Financial Model</u> has the meaning set forth in <u>Section 1.1</u> (Create a Pre-Refinancing Financial Model) of <u>Exhibit 5C</u> (Calculation of Refinancing Gain).

<u>Prevailing Rate of Wages</u> has the meaning given in Section 6.22(e) and 21C of the San Francisco Administrative Code.

Prime Contract means a direct Contract between Principal Project Company and a Contractor.

<u>Prime Contractor</u> means any Contractor that has a direct contract with Principal Project Company.

Principal Project Company (PPC) means [\_\_\_\_\_], a [\_\_\_\_\_] organized under the laws of the State of [\_\_\_\_\_], and its permitted successors and assigns.

**Principal Project Company IP** means Intellectual Property that is (i) owned by any PPC-Related Entity prior to the Effective Date, (ii) developed or acquired by any PPC-Related Entity independently of this Agreement or not for the purposes of performing the Work (iii) any adaptation, continuation or derivative work which requires the incorporation, exercise or practice of Intellectual Property that is the subject of either subsection (i) or (ii).

<u>Principal Project Document(s)</u> means the Contract Documents, the Key Contracts and the escrow agreements relating to the escrow of the Financial Model and Cost and Pricing Data and the IP Escrows.

**Priority 1** has the meaning given in <u>Section B.18.1.1 of the IFM Specifications</u>.

**Priority 2** has the meaning given in <u>Section B.18.1.1 of the IFM Specifications</u>.

Priority 3 has the meaning given in Section B.18.1.1 of the IFM Specifications.

<u>**Private Placement**</u> means the issuance of debt securities that are exempt from registration under U.S. securities law and are sold directly in the private market.

<u>Proceeds Account</u> means a deposit account with the Collateral Agent, bearing [Insert name of bank] Account No. [To be completed at Financial Close].

Prohibited Person means any Person who is:

(a) debarred, suspended, proposed for debarment with a final determination still pending, declared ineligible or voluntarily excluded from participating in procurement or nonprocurement transaction or determined to be a non-responsible bidder or contractor (as such terms are defined or used in City's Contractor Responsibility Program);

- (b) indicted or convicted of a crime, including misdemeanors, or had a civil suit or administrative judgment rendered against such Person involving the bidding, awarding, or performance of a government contract, a false claim or material misrepresentation to any private or governmental entity, or the crime of theft, fraud, embezzlement, perjury, or bribery (as such terms are defined or used in City's Contractor Responsibility Program);
- (c) identified on the list entitled "Entities Prohibited from Contracting with Public Entities in California per the Iran Contracting Act of 2010" maintained by the Department of General Services;
- (d) a financial institution against which, or a banking institution chartered or licensed in a jurisdiction against which, the United States Secretary of the Treasury has imposed special measures under Section 311 of the USA PATRIOT Act;
- (e) located within or is operating from a jurisdiction that has been designated as a "high-risk and non-cooperative jurisdiction" by the Financial Action Task Force;
- (f) a "senior foreign political figure" or a prohibited "foreign shell bank" within the meaning of 31 CFR Section 1010.605; or
- (g) engaged in litigation with the City of San Francisco relating to performance of a contract or business practices (unless the City has first waived (in the City's sole discretion) by written Notice to the transferring equity holder, with a copy to Principal Project Company, the prohibition on a transfer to such Person during the continuance of the relevant litigation).

**<u>Project</u>** means the design, construction, financing, operation and maintenance of the Infrastructure Facility as described in the Contract Documents.

Project Debt means bona fide indebtedness (including subordinated indebtedness) for or with respect to funds borrowed or obligations incurred (including bona fide indebtedness with respect to any financial insurance issued for funds borrowed) or for the value of goods or services rendered or received, the repayment of which has specified payment dates and is secured by one or more Security Documents. Project Debt includes principal, capitalized interest, accrued interest, customary and reasonable lender, financial insurer, agent and trustee fees, costs, expenses and premiums with respect thereto, payment obligations under interest rate and inflation rate hedging agreements or other derivative facilities with respect thereto, reimbursement obligations with respect thereto and lease financing obligations. Project Debt excludes (a) any indebtedness of Principal Project Company or any Equity Member of Principal Project Company that is secured by anything less than the entire Principal Project Company's Interest, such as indebtedness secured only by an assignment of economic interest in Principal Project Company or of rights to cash flow or dividends from Principal Project Company, (b) debt that constitutes consideration paid for the sale of the economic rights in Principal Project Company or Principal Project Company's Equity Members, and (c) any increase in indebtedness to the extent resulting from an agreement or other arrangement Principal Project Company enters into after it was aware (or should have been aware, using reasonable due diligence) of the occurrence or prospective occurrence of an event of termination giving rise to an obligation of City to pay Termination Compensation, including Principal Project Company's receipt of a Notice of Termination for Convenience and occurrence of a City Default of the type

entitling Principal Project Company to terminate this Agreement. In addition, no debt shall constitute Project Debt unless and until the Collateral Agent provides City with Notice thereof and the related Financing Agreements and Security Documents in accordance with the relevant Direct Agreement.

Project Debt Competition has the meaning set forth in Section 3.4.2(d)(ii).

**Project Financial Performance** means the cashflows from Principal Project Company's operations, financing and investments, including: (a) cash revenues and expenses, (b) Project Debt drawdowns, fees, interest expense and repayments, (c) Committed Investment drawdowns and Distributions; and (d) funding and release of reserves.

**Project IP** means all PPC Intellectual Property, Third Party IP and City IP incorporated into the Project, Implementation Proposal or Work.

**Project Management Plan (PMP)** means the D&C Management Plan and the IFM Management Plan (or either as the context requires).

**<u>Project Manager</u>** means the Key Person described in <u>Section •</u> of <u>Exhibit 3</u> (Implementation Proposal).

**Project Schedule** means (i) during the period from the Effective date until NTP2, the Initial Schedule; and (ii) from NTP2 until Final Acceptance, the fully cost loaded CPM schedule for all D&C Work meeting the requirements of the Technical Requirements, including the requirements in <u>Section 1.2.1</u> of <u>Division 1</u> thereof, as approved and updated in accordance with this Agreement.

Project Site means the D&C Site and the IFM Project Site, or either, as applicable.

**Proposed Change Order** means a Change Order proposed by City in accordance with <u>Section</u> <u>1.2</u> of <u>Exhibit 9</u> (Change Procedures).

**<u>Proprietary Design Review</u>** has the meaning set forth by <u>Section 1.8.5</u> of <u>Division 1</u> of the Technical Requirements.

**Protection(s) in Place** means any action taken to avoid damaging a utility facility which does not involve removing or relocating that facility, including staking the location of a facility, exposing the facility, locating construction equipment so as to avoid impacts to facilities, installing steel plating or concrete slabs, encasement in concrete, temporarily de-energizing power lines, and installing physical barriers. For example, temporarily lifting power lines without cutting them would be considered a Protection in Place; whereas temporarily moving power lines to another location after cutting them would be considered a phased Utility Adjustment and not a Protection in Place. The term "Protection in Place" includes both temporary measures and permanent installations meeting the foregoing definition.

**<u>Provision Document(s)</u>** has the meaning set forth in <u>Section 1.8.5.2.4</u> of <u>Division 1</u> of the Technical Requirements.

**Public Outreach Plan** has the meaning set forth in <u>Section 1.15.2</u> of <u>Division 1</u> of the Technical Requirements.

<u>Public Records Act (PRA)</u> means California Public Records Act, Cal. Gov't Code § 6250 et seq., as amended from time to time.

**Punch List** means an itemized list of D&C Work as agreed upon by Principal Project Company and City which remains to be completed after the Substantial Completion Date with respect to the Infrastructure Facility as a condition to Final Acceptance, and which is limited to minor incidental items of Work necessary to correct imperfections which have no adverse effect on the safety, use or operability of the Project.

**Quality Assurance (QA)** means all planned and systematic actions by Principal Project Company necessary to provide confidence that QC is performed in accordance with Principal Project Company's Design Quality Plan and Principal Project Company's Construction Quality Plan, that all Work complies with the Contract Documents and that all materials incorporated in the Work, and that all equipment and all elements of the Work will perform satisfactorily for the purpose intended. QA actions include: monitoring and verification of design through auditing, spot-checking and participation in the review of the Design Documents and work plans; and monitoring and verification of construction, manufacturing/process facilities and equipment, onsite equipment and QC documentation through auditing, spot inspections and reconciliation of material acceptance and rejection based on QC testing and Verification Sampling and Testing at production sites and the Project Site. Quality Assurance also includes documentation of all QA efforts.

<u>Quality Audit</u> means the audits performed by Principal Project Company in connection with its quality audit program, as described in further detail in <u>Section 1.4.1</u> of <u>Division 1</u> of the Technical Requirements.

**Quality Control (QC)** means the total of all activities performed by Principal Project Company to ensure that the Work meets the requirements of the Contract Documents. For design, this includes procedures for design quality, checking, and design review including reviews for constructability, and review and approval of work plans. For construction, this includes: procedures for materials handling and construction quality; inspection, sampling, testing and acceptance/rejection of materials, plants, production and construction; material certifications; calibration and maintenance of equipment; production process control; and monitoring of environmental compliance. Quality Control also includes documentation of all QC design and construction efforts.

<u>Quality Management Plan</u> means the plan to manage and measure the quality of IFM Services, prepared and submitted by Principal Project Company and approved by the City in accordance with <u>Section B.16.2</u> of the IFM Specifications, as updated and approved by City in accordance with this Agreement.

**Quality Program Manager** has the meaning set forth in <u>Section 1.1.4.9</u> of <u>Division 1</u> of the Technical Requirements.

<u>Quality Records</u> means the documentation required to be produced and maintained by PPC-Related Entities in accordance with Principal Project Company's Quality Program.

Quarter means a time period comprised of three calendar consecutive calendar months.

<u>Quarterly Availability Payment (QAP)</u> means the sum in dollars payable by City to Principal Project Company in each Contract Quarter of the IFM Period in accordance with this Agreement, as calculated in <u>Exhibit 4B</u> (Availability Payment Mechanism).

**Quarterly Deduction Cap** has the meaning given in <u>Section 5.1.1(a)</u> of <u>Exhibit 4B</u> (Availability Payment Mechanism).

**<u>Rating Agency(ies)</u>** means any credit rating agency registered with the Securities and Exchange Commission as a nationally recognized statistical rating organization ("**NRSRO**").

**<u>Reasonable Accuracy</u>** means with respect to the description or identification of a Utility in the Utility Information:

- (a) The Utility's actual centerline location is located at or less than five feet distant from the horizontal centerline location indicated therefor in the Utility Information (without regard to vertical location);
- (b) The Utility Information shows an active and existent Utility as not abandoned;
- (c) The Utility Information shows a non-existent or inactive Utility as abandoned; or
- (d) The Utility has an actual nominal diameter (excluding casings and any other appurtenances) greater than 12 inches, and its actual nominal diameter is either greater than or less than the diameter shown in the Utility Information by 25% or less of the diameter shown in the Utility Information.

Any other inaccuracies in the Utility Information (e.g., as to type of material or encasement status) shall have no impact on "reasonable accuracy" of its identification and shall not result in a determination that the Utility was not identified with "reasonable accuracy." If there is any discrepancy between any of the components of the Utility Information, only the most accurate information shall be relevant for purposes of determination of "reasonable accuracy."

**<u>Reasonable Investigation</u>** means the following activities by appropriate, qualified professionals:

- (a) Visit and inspection of the Project Site and adjacent locations, including inspection to identify the presence of other facilities, such as barriers, railing, structures, manholes or identifying markers;
- (b) Undertaking the geotechnical, Utility, Hazardous Material and other intrusive site conditions investigations, sampling, testing, analyses and assessments as required to identify conditions at a specific point or to identify where conditions change between points, and otherwise contemplated by the Predevelopment Agreement;
- (c) Review and analysis of all Reference Documents and online map tools;
- (d) Review and analysis of the CEQA Approval, NEPA Approval and other Regulatory Approvals;

- (e) Without limiting <u>clause (b)</u>, reasonable inquiry with Utility Owners, including request for and review of Utility plans provided by Utility Owners;
- (f) Review and analysis of material applicable Law applicable to the Project or the Work as of the Setting Date; and
- (g) Other activities consistent with Good Industry Practice sufficient to familiarize Principal Project Company with surface and subsurface conditions, including the presence of Utilities, Hazardous Materials, archeological, paleontological and cultural resources, and Threatened or Endangered Species, affecting the Project Site or surrounding locations.

**Reasonable Wear and Tear** means wear and tear that is reasonable given the use and age of the Infrastructure Facility and consistent with wear and tear that could reasonably be expected to exist at a similar facility, operating in a similar environment and similar circumstances but not including any degradation in the functionality or operability of the IFM Facility that, subject to the exceptions specified in the Technical Requirements, will result in the Infrastructure Facility failing to meet the Technical Requirements or failing to comply with Applicable Law and Standards.

**Record Documents** means construction drawings, specifications and related documentation furnished by Principal Project Company to reflect the actual conditions and location in detail of Work as constructed and installed, which may be generated initially as marked-up Release for Construction Documents, revised subsequently as-finished revised drawings and documents, and updated thereafter as required by the Technical Requirements.

## Recoverable Costs means:

- (a) The reasonably required costs of any assistance, action, activity or work undertaken by City which Principal Project Company is liable for or is obligated to reimburse City for under the terms of the Contract Documents, including the charges of third party contractors and reasonably allocated wages, salaries, compensation and overhead of City staff and employees performing such action, activity or work; plus
- (b) Reasonably required out-of-pocket costs City incurs to publicly procure any such third party contractors; plus
- (c) Reasonable fees and costs of attorneys (including the reasonably allocable fees and costs of the Office of the City Attorney), financial advisors, engineers, architects, insurance brokers and advisors, investigators, traffic and revenue consultants, risk management consultants, other consultants, and expert witnesses, as well as court costs and other litigation costs, in connection with any such assistance, action, activity or work, including in connection with defending claims by and resolving disputes with third party contractors; plus
- (d) Interest on all the foregoing sums at the Late Payment Rate, commencing on the date due under the applicable terms of the Contract Documents and continuing until paid; plus

(e) An administrative fee of 10% of the actual aggregate costs incurred by City as described in <u>clauses (a)</u> through <u>(d)</u> of this definition.

**<u>Recovery Schedule</u>** has the meaning set forth in <u>Section 1.2.1.4</u> of <u>Division 1</u> of the Technical Requirements.

<u>Rectification</u> means resolving and curing the Service Failure or Availability Failure (as applicable) in a way that Principal Project Company is in full compliance with this Agreement, with respect to that Service Failure or Availability Failure.

<u>Rectification Time</u> means, with respect to each Service Failure or Availability Failure (as applicable), the relevant period within which Principal Project Company shall Rectify such failure as specified in <u>Appendix B</u> (Response and Rectification Time) of <u>Exhibit 4</u> (Payment Mechanism).

**<u>Reference Documents</u>** means the documents provided with and so designated by City in the RFP or in writing during the PDA Term, which are provided for disclosure purposes only and without any warranty as to their accuracy, completeness or fitness for any particular purpose.

## Refinancing means:

- (a) any amendment, variation, novation, extension, renewal, supplement, refunding, defeasance or replacement of any Project Debt, Financing Agreement or Security Document (other than any Equity Member Debt);
- (b) any Project Debt incurred by Principal Project Company in addition to the Initial Project Debt, secured or unsecured;
- (c) the disposition of any rights or interests in, or the creation of any rights of participation with respect to, Project Debt, Financing Agreements and Security Documents or the creation or granting by Principal Project Company or any Lender of any other form of benefit or interest in either Project Debt, Financing Agreements and Security Documents or Principal Project Company's Interest whether by way of security or otherwise; or
- (d) any other arrangement put in place by Principal Project Company or another Person which has an effect similar to any of <u>clauses (a)</u> through <u>(c)</u> above.

<u>Refinancing Data</u> means the Pre-Refinancing Financial Model, the Post-Refinancing Financial Model, any interim Financial Models, and all assumptions, calculations and other information supporting the calculation of the Refinancing Gain, including any debt term sheets or other similar documentation relating to the proposed Refinancing.

<u>**Refinancing Gain**</u> means an amount equal to the greater of (a) zero and (b) (A - B), as such variables are calculated in accordance with <u>Exhibit 5C</u> (Calculation of Refinancing Gain).

**<u>Regulated Load</u>** means the metered provision of Energy for the following end-uses: (i) space heating; (ii) humidification; (iii) space cooling; (iv) dehumidification; (v) heat rejection; (vi) all fans; (vii) lighting; (viii) domestic hot water; and (ix) all pumps except for any domestic water booster pumps and any process-related pumps.

**Regulatory Approvals** means all permits, licenses, consents, concessions, grants, franchises, authorizations, waivers, variances or other approvals, guidance, protocol(s), mitigation agreement(s), or memoranda of agreement/understanding, and any amendment(s) or modification(s) of any of them, required or provided by Governmental Entities, including State, local, or federal regulatory agencies, agents, or employees, which authorize or pertain to the Project or the Work or take actions required to complete obligations in connection with this Agreement. Regulatory Approvals include encroachment permits and other access rights or right of entries for Work to be performed in areas under the jurisdiction of Governmental Entities, Major Approvals and Environmental Approvals.

<u>Regulatory Approvals Plan</u> means Principal Project Company's plan defining its approach to obtain all Regulatory Approvals, prepared and submitted by Principal Project Company in accordance with <u>Section 1.1.6</u> of <u>Division 1</u> of the Technical Requirements, as updated and approved by the City in accordance with this Agreement.

**<u>Related Project(s)</u>** has the meaning set forth in <u>Section 1.14.2</u> of <u>Division 1</u> of the Technical Requirements.

**<u>Release</u>** or <u>**Release of Hazardous Materials**</u> means, with respect to Hazardous Materials, any spill, leak, emission, release, discharge, injection, escape, leaching, dumping or disposal of Hazardous Materials into the soil, air, water, groundwater or environment, including any exacerbation of an existing release or condition of Hazardous Materials contamination.

<u>Release for Construction Documents (RFCD)</u> means Design Documents that have been authorized to be used as the basis for Construction Work or Renewal Work, in accordance with the design management portion of the approved Project Management Plan, and as set forth in <u>Section 1.8.6</u> of <u>Division 1</u> of the Technical Requirements.

Relevant Event has the meaning set forth in Section 13.3.1.

**Relevant Change in Law** means a discriminatory Change in Law which is principally borne by Principal Project Company (including PPC-Related Entities) and principally directed at, affects or relates only to the design, supply, construction (including installation), operation and maintenance of the Infrastructure Facility:

- (a) Except where such change (i) is in response, in whole or in part, to any PPC Fault, or (ii) is otherwise expressly permitted under the Contract Documents;
- (b) Where such Change in Law is a Change in Law for which compliance requires material capital expenditures by Principal Project Company; or
- (c) Where such Change in Law is a Change in Law for which compliance requires specific and material adverse changes in Principal Project Company's normal and compliant operation or maintenance procedures.

As used in this definition of "Relevant Change in Law", material capital expenditures are capital expenditures in excess of \$25,000 per event, such threshold to be index-linked and increased by the Escalation Factor calculated in accordance with <u>Section 2</u> of <u>Exhibit 4B</u> (Availability Payment Mechanism).

<u>Relief Event</u> means any of the following events or circumstances to the extent, in each case, that the event interferes directly and adversely with, or causes a failure of, the performance of the IFM Services after the Substantial Completion Date:

- (a) a Force Majeure Event; and
- (b) any Compensable Relief Event.

**<u>Renewal Work</u>** means all work related to the capital replacement, reconstruction, overhaul, refurbishment and reinstatement of the Infrastructure Facility, including the supply of machinery, equipment, materials, hardware, software, systems or any other items related to such Work, carried by Principal Project Company during the Term to maintain compliance with the Contract Documents. Renewal Work excludes D&C Work and SFMTA O&M Services.

**<u>Renewal Work Plan</u>** means the plan which defines design life, specific replacement and/or refurbishment strategies, prepared and submitted by Principal Project Company in accordance with <u>Section E</u> of the IFM Specifications, as updated and approved by the City in accordance with this Agreement.

**<u>Repeat Failure Ratchet</u>** has the meaning given in <u>Section 4.3</u> (Repeat Service Failures and Repeat Failure Ratchet) of <u>Exhibit 4B</u> (Availability Payment Mechanism).

**<u>Request for Change Proposal</u>** means a written Notice issued by City to Principal Project Company setting forth a proposed City Change and requesting Principal Project Company's assessment of cost and schedule impacts thereof, as described in <u>Section 1.1</u> of <u>Exhibit 9</u> (Change Procedures).

**<u>Request for Proposals (RFP)</u>** means the request for proposals to design, build, finance, and maintain the Infrastructure Facility and design, build, finance, operate and maintain the Housing and Commercial Component at the Project Site, issued by City on April 9, 2021, as amended.

<u>Request for Qualifications (RFQ)</u> means the request for qualifications to invite interested parties to submit a statement of qualifications to design, build, finance and maintain the Infrastructure Facility at the Project Site and design, build, finance, operate and maintain the Housing and Commercial Component at the Project Site, issued by City on August 21, 2020 as amended.

**Required Minimum Insurance Policy(ies)** means the Insurance Policies required in Exhibit 7 (Insurance Requirements) for performing the IFM Services, subject to Section 10.1.2.12 (Inadequacy of Required Coverage).

Rescue Refinancing means any Refinancing that:

- (a) occurs due to the failure or imminent failure of Principal Project Company to comply with any material financial obligation under any Financing Document;
- (b) results in the cure of such failure or imminent failure;
- (c) does not result in an increase in the Equity IRR beyond the Original Equity IRR; and

(d) does not result, singly or in the aggregate, in an actual or potential increase of the Lenders' Liabilities (determined without including any Exempt Refinancings) by more than 10%.

**Residual Life** means, for an element of the Work, the period remaining until the element will next require reconstruction, rehabilitation, restoration, renewal or replacement. The Residual Life of an element would be equal to its originally calculated Useful Life less its age if (a) the element has performed in service in the manner and with the levels of Reasonable Wear and Tear, and (b) Principal Project Company has performed the type of routine maintenance of the element which is normally included as an annually recurring cost in transit facility maintenance and repair budgets, and as a result thereof the element complies throughout its originally calculated Useful Life with all applicable requirements of the Contract Documents. The Residual Life of an element would be different from its originally calculated Useful Life minus its age if any of the foregoing conditions is not true.

#### Response means:

- (a) the surrounding area to a state or condition such that in the good faith opinion of City's Authorized Representative, the area is free of conditions which might otherwise create a condition that is hazardous or unsafe for maintenance and operating personnel; and
- (b) sufficiently reducing the risk during the remainder of the Rectification Time that further damage, nonperformance, safety hazards or adverse consequences caused by the Service Failure or Availability Failure (as applicable), might occur.

**<u>Response Time</u>** means with respect to each Service Failure or Availability Failure (as applicable), the relevant period within which Principal Project Company shall Respond to such failure as specified in <u>Appendix B</u> (Response and Rectification Time) of <u>Exhibit 4</u> (Payment Mechanism).

**<u>Retaining Wall</u>** means a wall structure which retains fill with a minimum exposed face height of 4 feet and a minimum overall height from top of footing to top of wall of 5 feet.

**Revenue Vehicle Parking Space** means the Functional Units corresponding to Functional Unit references AL-010 and AL-011, as specified in <u>Table C2</u> (Functional Components and Functional Units for Infrastructure Facility) of <u>Appendix C</u> (Functional Areas, Functional Units, Functional Unit Rankings and Deduction Amounts for Availability Failures and Service Failures) of <u>Exhibit 4B</u> (Availability Payment Mechanism).

**Review Date** has the meaning set forth in <u>Section 8.5.4.1</u>.

**<u>Revised Project Schedule</u>** has the meaning set forth in <u>Section 1.2.1.3</u> of <u>Division 1</u> of the Technical Requirements.

**Room Data Sheet** means a sheet or document that summarizes the requirements for each room in the Infrastructure Facility. It should include the room name, number, location, and the relevant prescriptive criteria and performance criteria from the Technical Requirements including for finishes, fixtures and fittings such as furniture and storage, mechanical and

electrical requirements, acoustic and lighting performance, and IT Comms and audio-visual requirements.

<u>Safety Compliance</u> means any and all improvements, repair, reconstruction, rehabilitation, restoration, renewal, replacement, changes in configuration, or procedures implemented to correct a specific safety condition or risk that City has reasonably determined to exist by investigation or analysis (excluding a safety condition or risk that exists by reason of Principal Project Company's failure to comply with the requirements of the Contract Documents).

<u>Safety Compliance Order</u> means a written order or directive from City to Principal Project Company to implement Safety Compliance.

<u>Safety Condition</u> means a state or condition of the relevant Functional Part which allows those persons who it can reasonably be expected may from time to time require to enter, leave, occupy, and use such Functional Part to do so safely, including compliance with applicable Law, relevant policies of the City and the City's requirements related to fire safety or health or workplace safety.

<u>Safety Orientation Program</u> means Principal Project Company's detailed plan for the safety orientation of employees and visitors, as more particularly described in <u>Section 01 35 45</u> of <u>Division 10</u> of the Technical Requirements.

<u>Safety Standards</u> means those provisions of the Technical Requirements that relate to Project safety, including <u>Section B.8.1.6</u> and <u>Section 01 35 45</u> of <u>Division 10</u> of the Technical Requirements. Safety Standards are considered to be important measures to protect public safety, worker safety or the safety of property.

<u>Scheduled Financial Close Date</u> has the meaning set forth in <u>Section 3.1.2</u> (Date of Financial Close).

<u>Scheduled Maintenance</u> means regular, routine and planned maintenance scheduled and performed according to a usage, time or other factors to determine frequency and timing of the maintenance.

## Scheduled Milestone Payment Date means [Note to PNC: Date to be inserted].

<u>Security Document</u> means any mortgage, deed of trust, pledge, lien, indenture, trust agreement, hypothecation, assignment, collateral assignment, account control agreement, financing statement under the enacted Uniform Commercial Code of any jurisdiction, security instrument or other charge or encumbrance of any kind, including any lease in the nature of a security instrument, given to any Lender as security for Project Debt or Principal Project Company's obligations pertaining to Project Debt and encumbering Principal Project Company's Interest.

<u>Senior Debt Service Amount</u> means the debt service payments that are senior to all other debt obligations in the cash flow waterfall.

<u>Service Category(ies)</u> means the discreet services as outlined in <u>Section B</u> and <u>Section C</u> of the IFM Specifications.

<u>Service Failure</u> means any failure by Principal Project Company to provide the IFM Services in accordance with Performance Measures designated Failure Type "SF" in the Performance Measurements Table and which, where a Response Time or Rectification Time applies, has not been Responded to or Rectified (as the case may be) within the relevant time. For the avoidance of doubt, where no Response Time and/or Rectification Time applies (for example, in respect of scheduled activities) there shall be a Service Failure at the point at which the IFM Failure Event occurred (for example, non-performance of the scheduled activity by the scheduled time).

<u>Service Failure Deduction</u> means a deduction from the Availability Payment as a result of a Service Failure.

<u>Service Failure Performance Measure</u> means a Performance Measure designated as SF in the Performance Measurements Table.

<u>Service Line(s)</u> means a utility line, the function of which is to connect directly the improvements on an individual property (e.g., a commercial building or an industrial warehouse) to another utility line located off such property, which other utility line connects more than one such individual line to a larger system, as well as any cable or conduit that supplies an active feed from a Utility Owner's facilities to activate or energize a Governmental Entity's local lighting and electrical systems, traffic control systems, street lights, communications systems and/or irrigation systems.

<u>Service Requests</u> means Demand Maintenance, Demand Requisitions, any request for service, report of a Service Failure, report of an Incident or IFM Failure Event or any other report, inquiry, complaint or comment, made by a Building Occupant or automatically generated, to the Help Desk or to Principal Project Company, whether by phone or electronically.

<u>Setting Date means</u>, 2025 [30 days prior to the proposal due date for submittal of proposals in connection with the D&C Contractor RFP].

**<u>SFMTA Board</u>** means the San Francisco Municipal Transportation Agency Board of Directors.

**SFMTA O&M Facilities** means the elements of the Project generally described in <u>Section</u> <u>A.2.1.2</u> of the IFM Specifications for which SFMTA will provide SFMTA O&M Services.

**<u>SFMTA O&M Facilities Warranty Bond</u>** has the meaning set forth in <u>Section 10.9</u> (SFMTA O&M Facilities Warranty Bond).

**SFMTA O&M Services** means the transit operations and maintenance services specified in Section A.2.1.2 of the IFM Specifications [and Retail Space maintenance services specified in Section A.2.1.2 of the IFM Specifications] that SFMTA will perform with respect to the SFMTA O&M Facilities. [Note: Further consideration required with respect to retail construction and services]

<u>Site Security Plan (SSP)</u> Principal Project Company's plan defining its oversight management program, team organization, and operating strategy to provide and maintain work site security, prepared and submitted by Principal Project Company in accordance with <u>Section 1.11.2.2</u> of <u>Division 1</u> of the Technical Requirements, as updated and approved by the City in accordance with this Agreement.

**SOFR** with respect to any US banking day means a rate per annum equal to the secured overnight financing rate published for such US banking day by the Federal Reserve Bank of New York, on the Federal Reserve Bank of New York's Website. The calculation of SOFR with respect to a loan that bears interest at a rate based on SOFR shall be the rate for the one/three/six calendar months period on the date of determination published by the SOFR Administrator; provided, however, that if as of 5:00 p.m. (New York City time) on any date of determination the SOFR rate for the applicable tenor has not been published by the SOFR Administrator, then the SOFR rate will be the SOFR rate for such tenor as published by the SOFR Administrator on the first preceding day.

<u>SOFR Administrator</u> means the CME Group Benchmark Administration Limited (CBA) (or a successor administrator of the SOFR rate as determined by the City, in its reasonable discretion.

<u>Software</u> means individually each, and collectively all, of the computer programs developed or provided by Principal Project Company, and any PPC-Related Entity, under this Agreement (including Developed IP, Principal Project Company IP and/or Third-Party IP), including as to each such program, the processes, and routines used in the processing of data, the object code, interfaces to be provided hereunder by PPC-Related Entity, updates, upgrades, and any and all programs otherwise provided by PPC-Related Entity under this Agreement.

<u>Source Code</u> means the version of a Software computer program in which the programmer's original programming statements are expressed in any programming language.

**Source Code Documentation** means (a) software written in programming languages, such as C and Fortran, including all comments and procedural code, such as job control language statements, which shall be in a form intelligible to trained programmers and capable of being translated into object or machine readable code for operation on computer equipment through assembly or compiling, and (b) documentation, including flow charts, schematics, statements of principles of operations, architectural standards, and commentary, explanations and instructions for compiling, describing the data flows, data structures, and control logic of the software in sufficient detail to enable a trained programmer through study of such documentation to maintain and/or modify the software without undue experimentation. Source Code and Source Code Documentation also include all modifications, revisions, additions, substitutions, replacements, updates, upgrades and corrections made to the foregoing items.

<u>Source of Supply Certificate of Compliance</u> has the meaning set forth in <u>Section 1.4.3.3</u> of <u>Division 1</u> of the Technical Requirements.

**Special Tools** means (a) specialized tools necessary for maintenance or repair of Project elements or equipment (including vehicles), and (b) other tools obtained or developed by Principal Project Company or any Contractor for use in performance of the IFM Services. The term excludes tools used in performance of the Work that were not procured for the Project or developed for use on the Project and that were acquired by Principal Project Company or Contractor at its own expense for use on multiple projects. The term "tool" as used in the Contract Documents includes "special tools."

<u>Standards and Specifications</u> means the standards, specifications and other documents referenced in the Technical Requirements.

#### Starting Insurance Benchmarking Premiums has the meaning set forth in Section 10.1.3(a).

<u>State</u> means the State of California.

<u>Statement of Qualification (SOQ)</u> means Principal Project Company's Statement of Qualification, provided in response to the Request for Qualifications.

<u>Step-in Party</u> means (a) the Collateral Agent, a Lender or any entity that is wholly owned by a Lender or group of Lenders, or (b) any Person approved by City as a Substituted Entity; in each case where such Person is not a Prohibited Person.

**<u>Structure</u>** means, as the context may require, bridges, Culverts, catch basins, drop inlets, retaining walls, cribbing, manholes, end walls, buildings, sewers, service pipes, under drains, foundation drains, steps, fences and other features which may be encountered in the Work and not otherwise classified.

<u>Subcontract</u> means each Contract with a Subcontractor.

<u>Subcontractor</u> means each Contractor that is not a Prime Contractor.

<u>Submittal(s)</u> means, generally, any document, work product or other written or electronic endproduct, report or item required to be delivered or submitted to City, Third Parties, or Governmental Entities in connection with this Agreement or the Project.

<u>Submittal Schedule</u> has the meaning set forth in <u>Section 1.2.1.6</u> of <u>Division 1</u> and <u>Section</u> <u>B.6.1</u> of the IFM Specifications.

<u>Substantial Completion</u> means that all D&C Work is complete (except for Punch List items that do not affect normal and safe use and operation of the Infrastructure Facility and any D&C Work that, by its nature, is to be performed after the Substantial Completion Date), and all other prerequisites for Substantial Completion have been met. Substantial Completion is deemed to have occurred upon satisfaction of all the conditions for the Project in <u>Exhibit 15C</u> (Conditions to Substantial Completion), as confirmed by City's issuance of a Certificate of Substantial Completion in accordance with the procedures and within the time frame established in <u>Section 7.8</u> (Substantial Completion).

<u>Substantial Completion Date</u> means the date City issues a Certificate of Substantial Completion for the Project.

<u>Substantial Completion Deadline</u> means [April 30, 2029 [IF D&C CONTRACTOR PROPOSES EARLIER, THIS DAY WILL BE THE EARLIER DATE], as such date may be extended from time to time pursuant to <u>Section 14.1.2</u> (Extension of Deadlines for General Delay Events).

<u>Substituted Entity</u> means a Third Party proposed by the Collateral Agent and approved by City under a Direct Agreement to act in Principal Project Company's stead and not merely as a Stepin Party, in each case where such Person is (a) a Suitable Substitute and (b) not a Prohibited Person.

Suitable Substitute means a Person, approved in writing by City in accordance with the Direct

Agreement that:

- (a) has the legal capacity, power and authority to become the party to and perform the obligations of Principal Project Company under this Agreement;
- (b) is in compliance with City's rules and regulations, and has adopted written policies regarding organizational conflicts of interest consistent with City's conflicts of interest policy;
- (c) has ensured that all of its subcontractors are in compliance with City's rules and regulations;
- (d) has adopted written policies regarding organizational conflicts of interest consistent with City's conflicts of interest policy;
- (e) employs individuals having the appropriate qualifications, experience and technical competence to timely perform Principal Project Company's obligations under the Contract Documents and the Principal Project Documents; and
- (f) otherwise has available resources (including committed financial resources and subcontracts) sufficient to enable it to perform the obligations of Principal Project Company under the Contract Documents and the Principal Project Documents.

<u>Superfund Site</u> means a site listed on either the National Priorities List or the Proposed National Priorities List at <u>https://www.epa.gov/superfund/superfund-national-priorities-list-npl</u>.

**Supplier** means any Person not performing work at or on the Project Site that supplies machinery, equipment, materials, hardware, software, systems or any other appurtenance to the Project to Principal Project Company or to any Contractor in connection with the performance of the Work. Persons that merely transport, pick up, deliver or carry materials, personnel, parts or equipment or any other items or persons to or from the Project Site shall not be deemed to be performing Work at the Project Site.

<u>Surety(ies)</u> means each properly licensed surety company, insurance company or other Person approved by City, which has issued a Payment Bond or a Performance Bond.

<u>Sustainability Management Plan</u> means Principal Project Company's plan describing its approaches to ensure achievement of the sustainability requirements of the Contract Documents, prepared and submitted by Principal Project Company in accordance with <u>Section</u> <u>1.9.1</u> of <u>Division 1</u> of the Technical Requirements, as updated and approved by the City in accordance with this Agreement.

<u>System Condition Index or SCI</u> means the total maintenance, repair and replacement deficiencies of the Infrastructure Facility divided by the current replacement value of the Infrastructure Facility.

**Systems Manual** has the meaning set forth in <u>Section 6.6.4.1</u> (Systems Manual) of <u>Division 6</u> of the Technical Requirements.

**Taxes** means federal, State, local or foreign income, margin, gross receipts, sales, use, excise, transfer, consumer, license, payroll, employment, severance, stamp, business, occupation, premium, windfall profits, environmental (including taxes under Section 59A of the Internal Revenue Code of 1986, as amended), customs, permit, capital stock, franchise, profits, withholding, social security (or similar), unemployment, disability, real property, personal property, registration, value added, alternative or add-on minimum, estimated or other taxes, levies, imposts, duties, fees or charges imposed, levied, collected, withheld or assessed at any time, whether direct or indirect, relating to, or incurred in connection with, the Project, the performance of the Work, revenues or act, business, status or transaction of Principal Project Company, including any interest, penalty or addition to such amounts, and including utility rates or rents, in all cases whether disputed or undisputed.

Technical Requirements means Exhibit 18 (Technical Requirements).

<u>Technology Enhancements</u> means modifications, additions, refinements, substitutions, revisions, replacements and upgrades made to Intellectual Property, equipment, mechanism, operational technology, or to any related documentation, that accomplish incidental, performance, structural, or functional improvements. Technology Enhancements specifically includes modifications, updates, or revisions made to software or any related documentation that correct errors or support new models of input-output devices with which the software is designed to operate.

<u>Temporary Areas</u> means areas outside of the Project Site where activities incidental to construction of the Project are being performed by Contractors, including field office sites, storage sites, staging areas dedicated to the Project, temporary work areas and parking areas, but excluding any permanent locations of Principal Project Company or any Contractor.

<u>Temporary Relocation</u> means any (a) interim relocation of any Utility (i.e., the installation, removal and disposal of an interim facility) pending installation of a permanent facility in the same or a new location, and (b) removal and reinstallation of the Utility in the same place without an interim relocation.

**Temporary Repair** means, in respect of the occurrence of an IFM Failure Event which results in an Availability Condition not being met in Functional Unit or Functional Component, works of a temporary nature that do not constitute Rectification but satisfy the Minimum Agreed Availability Conditions and substantially make good the relevant IFM Failure Event for the period until a Permanent Repair can be undertaken.

<u>**Temporary Work**</u> means any temporary works and structures necessary for the construction of permanent improvements. This includes falsework, formwork, scaffolding, shoring, temporary earthworks, sheeting, cofferdams, special erection equipment and the like.

**<u>Term</u>** means the period commencing on the Effective Date and ending on the date specified in <u>Section 2.3</u> (Term).

**Termination Compensation** means the measure of compensation owing from City to Principal Project Company upon termination of this Agreement prior to the stated expiration of the Term, as determined in accordance with this Agreement.

<u>Termination Date</u> means (a) the date of expiration of the Term, or (b) if applicable, the Early Termination Date.

**Termination Due to Court Ruling** has the meaning set forth in <u>Section 17.4.3</u> (Termination Due to Court Ruling).

**Termination for Convenience** has the meaning set forth in <u>Section 17.1.1</u> (Termination for Convenience).

**Test Procedures** means a description of a test to be performed. Test Procedures shall describe the test configuration, the test equipment required, the personnel required to perform the test, all construction and testing pre-requisites that must be completed before the test can be performed, lists each individual step to be followed in the test, expected result of each step, and the pass/fail criteria for each step, and shall include a Test Report form on which the results of the test shall be recorded.

<u>Test Report</u> means a form that lists each step to be performed in a test. Each Test Report shall identify the expected results of each step, a location to enter the actual results of each step, a place for the person responsible for performing the test to sign and date the form, and a place for any witnesses of the test to sign and date the form.

<u>Three-week Look-ahead Activity Reports</u> has the meaning set forth in <u>Section 1.2.1.7</u> of <u>Division 1</u> of the Technical Requirements.

<u>Third Party(ies)</u> means any person or entity unrelated to City or Principal Project Company. Third Parties expressly excludes any PPC-Related Entities.

<u>Third Party IP</u> means Intellectual Property owned by any Person unrelated to the Principal Project Company, any Contractor, or any of their respective Affiliates.

<u>Third Party and Utility Owner Coordination Work Plans</u> has the meaning set forth in <u>Section</u> <u>1.13</u> of <u>Division 1</u> of the Technical Requirements.

**Threatened or Endangered Species** means any species listed by the United States Fish and Wildlife Service as threatened or endangered under the Endangered Species Act, as amended, 16 U.S.C. § 1531 et seq., or any species listed as threatened or endangered under the California Endangered Species Act, Fish and Game Code § 2050 et seq.

<u>Total Unavailability</u> has the meaning set forth in <u>Section 5.4</u> (Total Unavailability) of <u>Exhibit 4B</u> (Availability Payment Mechanism).

<u>Traffic Control Plan(s) (TCP)</u> means the site-specific design plan that provides necessary details to control traffic through and around the work area.

<u>Trainee Plan</u> means the plan prepared and submitted by Principal Project Company and approved by the City to implement the SFMTA Trainee Program.

Transaction has the meaning set forth in Section 20.2.1.

<u>Transaction Survey(s)</u> means surveys covering IFM Services and people making enquiries or Service Requests by telephone, in writing, electronically or otherwise to Principal Project Company.

<u>Transportation Management Plan</u> means Principal Project Company's plan describing how safe traffic operations will be managed and maintained during each phase of construction and in every work zone of the Project, prepared and submitted by Principal Project Company in accordance with <u>Section 1.11.3.1</u> of <u>Division 1</u> of the Technical Requirements, as updated and approved by the City in accordance with this Agreement.

<u>Utility Easement</u> means a permanent replacement easement and/or other interest in real property (excluding a franchise) located outside of the Project Site that is necessary for a Utility Adjustment.

<u>Unavailable</u> means, in relation to a Functional Unit or Functional Component, that such Functional Unit or Functional Component is in a state or condition which does not comply with any one or more of the Availability Conditions and "Unavailability" shall be construed accordingly.

<u>Unavailable Component Threshold</u> means, the number of Functional Components in a Functional Unit that if exceeded by Unavailable Components at any point triggers deemed Unavailability of all Functional Components in the relevant Functional Unit. <u>Table 2</u> (Unavailable Component Thresholds) in <u>Section 5.6</u> (Unavailable Component Thresholds) of <u>Exhibit 4B</u> (Availability Payment Mechanism) lists the threshold value for each affected Functional Unit.

<u>Unavoidable Delay Event</u> means any of the following events or circumstances that occurs prior to the Final Acceptance Date and directly impacts the D&C Work:

- (a) a Force Majeure Event;
- (b) a Major Approval Delay;
- (c) a Utility Delay; or
- (d) an Adverse Weather Event.

<u>Unidentified Utility</u> means a subsurface Main or Trunkline Utility, (i) where the Utility Information incorrectly indicates that the subject Main or Trunkline Utility does not exist anywhere within the boundary lines of the Project Site and which Main or Trunkline Utility could not have been reasonably inferred as of the Setting Date based on a Reasonable Investigation; (ii) where the Utility Information does not identify the subject Main or Trunkline Utility within the boundary lines of the Project Site with Reasonable Accuracy and such information could not have been reasonably inferred as of the Setting Date based on a Reasonable Investigation; (iii) where such Main or Trunkline Utility is not identified or referenced in any of the information, data, analyses or reports undertaken by Principal Project Company pursuant to the Predevelopment Agreement; and (iv) where such Main or Trunkline Utility should have been identified or referenced pursuant to a Reasonable Investigation undertaken pursuant to the Predevelopment Agreement. Unidentified Utilities specifically exclude any Utility that serves the Project Site as of the Setting Date and which is to be de-energized or de-activated and removed as part of the Project. <u>Unilateral Change Order</u> has the meaning set forth in <u>Section 2</u> (Unilateral Change Orders) of <u>Exhibit 9</u> (Change Procedures).

<u>Uninterruptible Power Supply (UPS)</u> has the meaning set forth in <u>Section 3.12.14</u> and <u>Appendix G, Section 2.10.8</u> of <u>Division 3</u> of the Technical Requirements.

<u>Unsafe Condition</u> means a condition that (a) gives rise to the imminent possibility of serious injury to workers or the public or of serious damage to property or the environment, or (b) affects safe movement.

**Upgrades** means alterations, improvements, modifications or changes, including Capacity Improvements, that Principal Project Company makes to any portion of the Project, as originally designed and constructed, at any time after the Substantial Completion Date, except as part of ordinary maintenance or Renewal Work. Upgrades may include alterations, improvements, modifications or changes that require an amendment or supplement to the final environmental impact documents for the Project or that are to be located outside the boundaries of the original Project Site. Upgrades exclude Technology Enhancements and any alterations, improvements, modifications or changes undertaken in the use or development of a Business Opportunity.

<u>Use Condition</u> means a state or condition of the relevant Functional Unit or Functional Component which satisfies the Use Parameters for that Functional Part.

<u>Use Parameters</u> means the range of functional requirements for the proper use and enjoyment of a Functional Unit or Functional Component, including the use of, in the case of the Infrastructure Facility, the ancillary equipment for its particular purpose as set out in the Technical Requirements.

<u>Useful Life</u> means, for an element, the period following its first installation, or following its last reconstruction, rehabilitation, restoration, renewal or replacement, until the element will next require reconstruction, rehabilitation, restoration, renewal or replacement.

<u>Utility(ies) or utility(ies)</u> means a privately, publicly or cooperatively owned facility (which term includes lines, systems and other facilities, and includes municipal and/or government facilities) for transmitting or distributing communications, cable television, power, electricity, gas, oil, crude products, water, steam, waste, or any other similar commodity, including any fire or police signal system as well as streetlights associated with roadways owned by local agencies. However, when used in the context of Utility Adjustments of facilities to accommodate the Project, the term "Utility" or "utility" excludes traffic signals, ramp metering systems, flashing beacon systems, and lighting systems for roads adjacent to the Project. Necessary appurtenances to each utility facility (including the utility source, guide poles, Service Lines, supports, etc.) shall be considered part of the facility. Without limitation, any service lateral connecting directly to a utility shall be considered an appurtenance to that utility, regardless of the ownership of such service lateral.

<u>Utility Adjustment</u> means each relocation (temporary or permanent), abandonment, Protection in Place, removal (of previously abandoned utility facilities as well as newly-abandoned facilities), replacement, reinstallation, and/or modification of existing Utilities necessary to accommodate the Project or the Work. For any utility crossing the Project Site or public right of way, the Utility Work for each crossing of the Project Site or public right of way, as applicable, by that utility shall be considered a separate Utility Adjustment. For any utility installed

longitudinally within the Project Site or the public right of way, the Utility Work for each continuous segment of that utility located within the Project Site or public right of way, as applicable, shall be considered a separate Utility Adjustment.

<u>Utility Betterment</u> means any upgrading of a Utility that is not attributable to the construction of the Project and is made during the course of a Utility Adjustment solely for the benefit of and at the election of the Utility Owner, including an increase in the capacity, capability, efficiency or function of the facility over that which was provided by the existing Utility. The following are not considered Utility Betterments:

- (a) any upgrading required for accommodation of the Project;
- (b) any upgrading required under the Contract Documents;
- (c) replacement devices or materials that are of equivalent standards although not identical;
- (d) replacement of devices or materials no longer regularly manufactured with an equivalent or next higher grade or size;
- (e) any upgrading required by applicable Law;
- (f) replacement devices or materials that are used for reasons of economy (e.g., non-stocked items may be uneconomical to purchase); and
- (g) any upgrading required by the Utility Owner's applicable Utility Standards.

Utility Delay has the meaning set forth in Section 7.6.13.1

<u>Utility Information</u> means as-built drawings, existing facility drawings, or other documentation of existing underground Utilities within the Project Site available to Principal Project Company or generated by any PPC-Related Entity. Utility Information includes the Utility conflict matrix and subsurface utility engineering investigations undertaken by or on behalf of Principal Project Company during the PDA Term.

<u>Utility Owner</u> means the owner or operator of any Utility (including both privately held and publicly held entities, cooperative utilities, and municipalities and other governmental agencies), including City acting by and through the San Francisco Public Utilities Commission (it being understood that the San Francisco Public Utilities Commission is not the "City" for other purposes under this Agreement)).

<u>Utility Standards</u> means the standard specifications, standards of practice, and construction methods that a Utility Owner customarily applies to facilities (comparable to those being adjusted on account of the Project) constructed by the Utility Owner (or for the Utility Owner by its contractors), at its own expense. Except as may be specifically identified in the Technical Requirements, "Utility Standards" are not "Standards and Specifications."

<u>Utility Work</u> means the design and construction necessary for a Utility Adjustment. Any Utility Work furnished or performed by Principal Project Company is part of the D&C Work; and any

Utility Work furnished or performed by a Utility Owner is not part of the D&C Work. Utility Work expressly includes Utility Adjustments of PG&E's power lines and poles that impact the Project.

Vandalism means willful or malicious damage or defacement (including graffiti) that:

- (a) was not the result of Principal Project Company's failure to comply with its obligations under the Contract Documents and its IFM Management Plan with respect to Vandalism; and
- (b) does not arise from, or was not contributed to, directly or indirectly, by any act or omission of Principal Project Company or any PPC-Related Entity.

<u>Verification Sampling and Testing</u> means sampling and testing performed to validate the quality of the product. City, or its designee, will perform Verification Sampling and Testing as part of its QA Oversight efforts.

Warning Notice has the meaning set forth in Section 16.1.3.1.

<u>Weather Data</u> means a record of annual hourly data for dry and wet bulb temperature, dew point, relative humidity, total horizontal solar radiation, wind speed and direction and atmospheric pressure at the Weather Monitoring Station.

<u>Weather Monitoring Station</u> means the closest National Weather Service weather monitoring station to the Infrastructure Facility.

**Work** means all of the work, services and obligations required to be furnished, performed and provided by Principal Project Company under the Contract Documents, including activities to obtain financing as well as all administrative, design, engineering, construction, demolition, Utility Adjustments, payment to Third Parties, support services, financing services, operations, maintenance and other work of renewal, reconstruction, repair or reinstatement of Project improvements and equipment, and management services. The term does not include any efforts which the Contract Documents expressly specify will be performed by Persons other than PPC-Related Entities.

# EXHIBIT 2

# PROJECT MANAGEMENT PLAN

#### [NOTE TO PNC: FULL PRELIMINARY DRAFT TO BE INCLUDED AT COMMERCIAL CLOSE. FINAL APPROVED DRAFT TO BE INSERTED AND REPLACE PRELIMINARY AS A CONDITION TO NTP 2]

## EXHIBIT 3

### **IMPLEMENTATION PROPOSAL**

#### Exhibit 3A Financial Proposal

[NOTE TO PNC: This will include the Finance Plan, including the Base Case Financial Model, with its audited report, and respective commitment letters for Bank Debt, Equity Investments, Security Documents and Insurance Policies, among other documents, and including the Final Price as determined under the PDA, PPC's financing costs, along with a schedule including the proposed Milestone Payment amount and the proposed Availability Payments payable from Substantial Completion to the end of the PDA Term. To be included at Commercial Close.]

#### Exhibit 3B Technical Proposal Elements

[NOTE TO PNC: This will include the following elements developed pursuant to the PDA, as selected by City in its discretion:

- aspects PNC/PPC's design and other submittals
- aspects of D&C Contractor's design-build proposal
- aspects of IFM Contractor's proposal

This may include listed subs, identified Key Personnel, designs, enhancements to the Technical Requirements, etc. that City chooses to include as contractual requirements. To be included at Commercial Close.]

# <u>EXHIBIT 4</u>

## PAYMENT MECHANISM<sup>1</sup>

## PART A: DEFINITIONS

#### 1 **DEFINITIONS**

Unless otherwise specified, capitalized terms used in this <u>Exhibit 4</u> (Payment Mechanism) have the meanings given in <u>Exhibit 1</u> (Abbreviations and Definitions) to the Agreement.

<sup>&</sup>lt;sup>1</sup> This Exhibit remains subject to continued review in light of ongoing refinements to the IFM Specifications and Technical Requirements

#### EXHIBIT 4A

#### MILESTONE PAYMENT MECHANISM

#### 1. MILESTONE PAYMENT

- (a) Subject to the terms and conditions of this <u>Section 1</u> (Milestone Payment) and the Contract Documents, Principal Project Company shall be entitled to receive a single milestone payment from the City as partial compensation for Principal Project Company's performance of the D&C Work and reaching Substantial Completion (the "Milestone Payment").
- (b) The Milestone Payment shall be calculated using the following formula:

MP = MMP - D

Where:

MP = Milestone Payment;

MMP = the amount  $[\$\bullet];$ 

D = D&C Deductions incurred for the period commencing on the Effective Date and ending on the Substantial Completion Date as set out in <u>Section 2</u> (D&C Period Deductions) of <u>Exhibit 4A</u> (Milestone Payment Mechanism).

- (c) For the Milestone Payment, Principal Project Company shall submit an invoice no later than the 10th day of the month immediately following issuance of a Certificate of Substantial Completion in accordance with the process in <u>Section 11.3</u> (Invoice, Other Amounts and Payments) of the Agreement.
- (d) The City will pay Principal Project Company the Milestone Payment accordance with the process in <u>Section 11.3</u> (Invoice, Other Amounts and Payments) of the Agreement.
- (e) The Milestone Payment shall not be used to pay the principal of any outstanding obligations, the interest on which is excludable from gross income for federal income tax purposes as set forth under the Internal Revenue Code of 1986, as amended.

### 2. D&C PERIOD DEDUCTIONS

- **2.1** Subject to the provisions of this Agreement, if at any time prior to the Substantial Completion Date, a D&C Failure Event occurs, the City may make a Deduction from the relevant Milestone Payment in respect of that D&C Failure Event.
- 2.2 Principal Project Company shall submit to City, 30 days before the anticipated date of Substantial Completion and Final Acceptance, a report providing a preliminary summary of all D&C Noncompliance Points and D&C Period Deductions calculated as of 30 days prior to the anticipated Substantial Completion Date and Final Acceptance Date (as applicable). Within 30 days following the Substantial Completion Date, Principal Project Company shall submit to City a final summary of all D&C Period Noncompliance Points and D&C Period Deductions calculated as of the Substantial Completion Date or Final Acceptance Date (as applicable).
- 2.3 Principal Project Company will accrue D&C Period Noncompliance Points for every D&C Failure Event in accordance with this <u>Section 2</u>, <u>Section 15.4</u> of the Agreement and as set forth in <u>Appendix A</u> (D&C Noncompliance Points and Deductions) in this <u>Exhibit 4A</u> (Milestone Payment Mechanism) up until the Substantial Completion Date for the D&C Period.
- **2.4** Principal Project Company will include in its Performance Monitoring Report a summary of all D&C Period Noncompliance Points accrued in the month and cumulative number of points accrued up to the month being reported from the Effective Date.

#### [Note: D&C Thresholds under development/review]

## APPENDIX A

## **D&C PERIOD NONCOMPLIANCE POINTS AND DEDUCTIONS**

## Table A1 – D&C Failure Event Table

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
1	Division 01 - General							
1. 1	Project Management	Project Management Plan	Prepare, submit and update Project Management Plan in accordance with the Contract Documents	5 days	5 per D&C Failure Event	-	2 days	Technical Requirements, Division 01, Section 1.1.1
1. 2	Project Management	Initiation Meetings	Conduct Initiation Meetings and submit meeting agendas and minutes in accordance with the Contract Documents	1 day	3 per D&C Failure Event	\$150	1 day	Technical Requirements, Division 01, Section 1.1.3.1-5
1. 3	Project Management	Weekly Project Coordination Meetings	Conduct Weekly Project Coordination Meetings in accordance with the Contract Documents	1 day	3 per D&C Failure Event	\$150	1 day	Technical Requirements, Division 01, Section 1.1.3.6
1. 4	Project Management	Monthly Progress Meetings	Conduct Monthly Progress Meetings and submission of meeting agenda and minutes in accordance with the Contract Documents	1 day	3 per D&C Failure Event	\$150	1 day	Technical Requirements, Division 01, Section 1.1.3.7
1. 5	Project Management	Ongoing Coordination Meeting	Conduct Ongoing Coordination Meetings and submission of meeting agenda and minutes in	3 days	1 per D&C Failure Event	\$50	1 day	Technical Requirements, Division 01, Section 1.1.3.8

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
			accordance with the Contract Documents					
1. 6	Project Management	Special Meetings	Conduct Special Meetings and submission of meeting agenda and minutes in accordance with the Contract Documents	3 days	1 per D&C Failure Event	\$50	1 day	Technical Requirements, Division 01, Section 1.1.3.9
1. 7	Intentionally Omitted							
1. 8	Project Management	Regulatory Approvals Plan	Prepare, submit and update a Regulatory Approvals Plan in accordance with the Contract Documents	5 days	3 per D&C Failure Event	\$150	3 days	Technical Requirements, Division 01, Section 1.1.7
1. 9	Project Management	Design and Construction Management Plans	Prepare, submit and update a complete Design Management Plan to achieve the requirements of Design Management in accordance with the Contract Documents	5 days	5 per D&C Failure Event	-	2 days	Technical Requirements, Division 01, Section 1.1.7.1
1. 10	Project Management	Design and Construction Management Plans	Prepare, submit and update a complete Construction Management Plan to achieve the requirements of Construction Management in	5 days	5 per D&C Failure Event	-	2 days	Technical Requirements, Division 01, Section 1.1.7.2

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
			accordance with the Contract Documents					
1. 11	Project Management	IFM Management Plan	Prepare, submit and update a complete preliminary IFM Management Plan in accordance with the Contract Documents	5 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 01, Section 1.1.8
1. 12	Project Controls and Performance Management	Project Schedules	Prepare, submit and update Initial Schedule in accordance with the Contract Documents	5 days	5 per D&C Failure Event	\$1,000	2 days	Technical Requirements, Division 01, Section 1.2.1.2
1. 13	Project Controls and Performance Management	Project Schedules	Prepare, submit and update Project Schedule (excluding the Initial Schedule) in accordance with the Contract Documents	3 days	5 per D&C Failure Event	\$300	1 day	Technical Requirements, Division 01, Section 1.2.1.3 and Section 1.2.2
1. 14	Project Controls and Performance Management	Project Schedules	Prepare, submit and update Recovery Schedule in accordance with the Contract Documents	5 days	5 per D&C Failure Event	\$300	2 days	Technical Requirements, Division 01, Section 1.2.1.4 and Section 1.2.2
1. 15	Project Controls and Performance Management	Project Schedules	Prepare, submit and update complete Project Schedule Monthly Updates in accordance with the Contract Documents	2 days	3 per D&C Failure Event	\$150	1 day	Technical Requirements, Division 01, Section 1.2.1.5 and Section 1.2.2

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
1. 16	Project Controls and Performance Management	Project Schedules	Prepare, submit and update complete preliminary and final Submittal Schedule for D&C Submittals in accordance with the Contract Documents	5 days	3 per D&C Failure Event	\$150	2 days	Technical Requirements, Division 01, Section 1.2.1.6 and Section 1.2.2
1. 17	Project Controls and Performance Management	Project Schedules	Prepare, submit and update complete Look-Ahead Activity Reports in accordance with the Contract Documents	1 day	1 per D&C Failure Event	\$50	1 day	Technical Requirements, Division 01, Section 1.2.1.7 and Section 1.2.2
1. 18	Project Controls and Performance Management	Schedule System	Comply with requirements of the Scheduling System	5 days	1 per D&C Failure Event	\$50	1 day	Technical Requirements, Division 01, Section 1.2.3
1. 19	Project Controls and Performance Management	Performance Reporting	Prepare, submit and update Monthly Progress Status Report in accordance with the Contract Documents, including City acceptance of format	5 days	3 per D&C Failure Event	\$150	3 days	Technical Requirements, Division 01, Section 1.2.4.1
1. 20	Project Controls and Performance Management	As-Built Schedules	Prepare, submit and update complete As- Built Schedule and Final Schedule Report in accordance with the Contract Documents	5 days	3 per D&C Failure Event	\$150	3 days	Technical Requirements, Division 01, Section 1.2.5

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
1. 21	Project Controls and Performance Management	Time Impact Analysis for Proposed Extensions of Time	Prepare, submit and update Time Impact Analysis for proposed extensions of time in accordance with the Contract Documents	2 days	3 per D&C Failure Event	\$150	1 day	Technical Requirements, Division 01, Section 1.2.6
1. 22	Submittal Management	Submittal Management	Prepare, submit and update Initial List of Submittals in accordance with the Contract Documents	5 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 01, Section 1.3
1. 23	Submittal Management	Submittal Management	Meet requirements and standards of each Submittal provided to the City and other entities in accordance with the Contract Documents	5 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 01, Section 1.3
1. 24	Quality Management	Principal Project Company Quality Program	Prepare, submit and update complete Initial PPC Quality Program Plan in accordance with the Contract Documents	5 days	5 per D&C Failure Event	-	3 days	Technical Requirements, Division 01, Section 1.4.1
1. 25	Quality Management	Principal Project Company Quality Program	Prepare, submit and update complete subsequent updates to PPC Quality Program Plan in accordance with the Contract Documents	5 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 01, Section 1.4.1
1. 26	Quality Management	Principal Project Company Quality Program	Establish Quality Management System and submit necessary documentation to the City	5 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 01, Section 1.4.1.2

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
1. 27	Quality Management	Principal Project Company Quality Program	Prepare, submit and update Design Control Procedures in accordance with the Contract Documents	10 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 01, Section 1.4.1.3
1. 28	Quality Management	Principal Project Company Quality Program	Prepare, submit and update Product Identification and Traceability Procedures in accordance with the Contract Documents	10 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 01, Section 1.4.1.4
1. 29	Quality Management	Principal Project Company Quality Program	Prepare, submit and update Process Control Procedures in accordance with the Contract Documents	10 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 01, Section 1.4.1.5
1. 30	Quality Management	Principal Project Company Quality Program	Prepare, submit and update Schedule of Testing Equipment in accordance with the Contract Documents	10 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 01, Section 1.4.1.7
1. 31	Quality Management	Principal Project Company Quality Program	Prepare, submit and update Control of Nonconforming Work Procedures in accordance with the Contract Documents	10 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 01, Section 1.4.1.9
1. 32	Quality Management	Principal Project Company Quality Program	Prepare, submit and update Nonconforming Work Report in accordance with the Contract Documents	24 hours	1 per D&C Failure Event	\$50	24 hours	Technical Requirements, Division 01, Section 1.4.1.9

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
1. 33	Quality Management	Principal Project Company Quality Program	Prepare, submit and update Corrective and Preventive Action Procedures in accordance with the Contract Documents	5 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 01, Section 1.4.1.10
1. 34	Quality Management	Principal Project Company Quality Program	Prepare, submit and update Control of Quality Records Procedures in accordance with the Contract Documents	5 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 01, Section 1.4.1.11
1. 35	Quality Management	Principal Project Company Quality Program	Prepare, submit and update Quality Audit Procedures in accordance with the Contract Documents	5 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 01, Section 1.4.1.12
1. 36	Quality Management	Principal Project Company Quality Program	Prepare, submit and update Quality Audit Reports in accordance with the Contract Documents	3 days	1 per D&C Failure Event	\$50	1 day	Technical Requirements, Division 01, Section 1.4.1.12
1. 37	Quality Management	Principal Project Company Quality Program	Prepare, submit and update Training Procedures in accordance with the Contract Documents	5 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 01, Section 1.4.1.13
1. 38	Quality Management	Principal Project Company Quality Program	Prepare, submit and update Training Records and Certifications in accordance with the Contract Documents	3 days	1 per D&C Failure Event	\$50	1 day	Technical Requirements, Division 01, Section 1.4.1.13
1. 39	Quality Management	Principal Project Company Design Quality	Prepare, submit and update complete Initial Principal Project Company Design Quality Plan	5 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 01, Section 1.4.2

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
			in accordance with the Contract Documents					
1. 40	Quality Management	Principal Project Company Design Quality	Prepare, submit and update updates to Principal Project Company Design Quality Plan in accordance with the Contract Documents	5 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 01, Section 1.4.2
1. 41	Quality Management	Principal Project Company Design Quality	Prepare, submit and update Design Progress Tracking Report in accordance with the Contract Documents	1 day	1 per D&C Failure Event	\$50	1 day	Technical Requirements, Division 01, Section 1.4.2.1.A
1. 42	Quality Management	Principal Project Company Design Quality	Prepare, submit and update Monthly PDQP Report in accordance with the Contract Documents	1 day	1 per D&C Failure Event	\$50	1 day	Technical Requirements, Division 01, Section 1.4.2.1.B
1. 43	Quality Management	Principal Project Company Design Quality	Prepare, submit and update Final Design Report in accordance with the Contract Documents	3 days	1 per D&C Failure Event	\$50	2 days	Technical Requirements, Division 01, Section 1.4.2.1.C
1. 44	Quality Management	Principal Project Company Design Quality	Prepare, submit and update Quantity Estimates in accordance with the Contract Documents	3 days	1 per D&C Failure Event	\$50	2 days	Technical Requirements, Division 01, Section 1.4.2.1.D
1. 45	Quality Management	Principal Project Company Design Quality	Prepare, submit and update Quality Surveillance and Mitigation Monitoring Reports in accordance with the Contract Documents	3 days	1 per D&C Failure Event	\$50	2 days	Technical Requirements, Division 01, Section 1.4.2.2.A

Infrastructure Facility DBFOM Agreement

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
1. 46	Quality Management	Principal Project Company Design Quality	Prepare, submit and update Log of Nonconforming Design Work Reports in accordance with the Contract Documents	1 day	1 per D&C Failure Event	\$50	1 day	Technical Requirements, Division 01, Section 1.4.2.2.C
1. 47	Quality Management	Principal Project Company Design Quality	Assign a Design Quality Assurance Manager (DQAM) to assess and evaluate the efficacy and effectiveness of Principal Project Company's quality in design activities.	5 days	5 per D&C Failure Event	\$500	3 days	Technical Requirements, Division 01, Section 1.4.2.3
1. 48	Quality Management	Principal Project Company Construction Quality	Prepare, submit and update Principal Project Company Construction Quality Plan (PCQP) in accordance with the Contract Documents	5 days	3 per D&C Failure Event	\$150	3 days	Technical Requirements, Division 01, Section 1.4.3
1. 49	Quality Management	Principal Project Company Construction Quality	Prepare, submit and update Candidate Independent Testing Laboratory Credentials in accordance with the Contract Documents	5 days	3 per D&C Failure Event	\$150	3 days	Technical Requirements, Division 01, Section 1.4.3.1
1. 50	Quality Management	Principal Project Company Construction Quality	Prepare, submit and update ITL Portable and Satellite Policies and Procedures in accordance with the Contract Documents	2 days	3 per D&C Failure Event	\$150	0.5 day	Technical Requirements, Division 01, Section 1.4.3.1

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
1. 51	Quality Management	Principal Project Company Construction Quality	Prepare, submit and update Material and Equipment Conformance Certifications in accordance with the Contract Documents	2 days	1 per D&C Failure Event	\$50	0.5 day	Technical Requirements, Division 01, Section 1.4.3.3
1. 52	Quality Management	Principal Project Company Construction Quality	Prepare, submit and update Source of Supply Compliance Certifications in accordance with the Contract Documents	2 days	1 per D&C Failure Event	\$50	0.5 day	Technical Requirements, Division 01, Section 1.4.3.3
1. 53	Quality Management	Principal Project Company Construction Quality	Prepare, submit and update Construction Quality Work Plans in accordance with the Contract Documents	3 days	1 per D&C Failure Event	\$50	1 day	Technical Requirements, Division 01, Section 1.4.3.4
1. 54	Quality Management	The City Quality Oversight	Prepare, submit and update Software Quality Assurance Plan in accordance with the Contract Documents	10 days	3 per D&C Failure Event	\$150	3 days	Technical Requirements, Division 01, Section 1.4.5
1. 55	Existing Conditions	Pre-Construction Information, Pre- and Post- Construction Surveys	Prepare, submit and update Pre- construction Survey Reports in accordance with the Contract Documents	5 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 01, Section 1.5.1
1. 56	Existing Conditions	Pre-Construction Information, Pre- and Post- Construction Surveys	Prepare, submit and update Post- construction Survey Reports in accordance with the Contract Documents	10 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 01, Section 1.5.1

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
1. 57	Information Management	Information Management	Prepare, submit and update Information Management Plan (IMP) in accordance with the Contract Documents	10 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 01, Section 1.6
1. 58	Information Management	Information Management	Prepare, submit and update Document Control Procedures in accordance with the Contract Documents	5 days	1 per D&C Failure Event	\$50	2 days	Technical Requirements, Division 01, Section 1.6.1.A
1. 59	Information Management	As-Built Documents	Prepare, submit and update complete As- Built Documents for individual items in accordance with the Contract Documents	10 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 01, Section 1.6.3.A
1. 60	Information Management	As-Built Documents	Prepare, submit and update complete aggregated As-Built Documentation prior to Final Acceptance in accordance with the Contract Documents	10 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 01, Section 1.6.3.A
1. 61	Training	Training	Prepare, submit and update Training Procedures, Materials and Records in accordance with the Contract Documents	10 days	3 per D&C Failure Event	\$150	3 days	Technical Requirements, Division 01, Section 1.7
1. 62	Design Management	Proprietary Design Reviews	Prepare, submit and update Proprietary Design Reviews in accordance with the Contract Documents	5 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 01, Section 1.8.5

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
1. 63	Design Management	Proprietary Design Reviews	Prepare, submit and update to the City, and maintain a comprehensive Basis of Design Report (BODR) for the Project in accordance with the Contract Documents	5 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 01, Section 1.8.5.2.3
1. 64	Design Management	Release for Construction Documents	Prepare, submit and update RFCDs for approval by City in accordance with the Contract Documents	None	5 per D&C Failure Event	-	1 day	Technical Requirements, Division 01, Section 1.8.6
1. 65	Design Management	Release for Construction Documents	Prepare, submit and update Design Review Report in accordance with the Contract Documents	5 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 01, Section 1.8.6.1
1. 66	Sustainability	Sustainability	Prepare, submit and update Sustainability Management Plan in accordance with the Contract Documents	10 days	3 per D&C Failure Event	\$150	3 days	Technical Requirements, Division 01, Section 1.9.1
1. 67	Building Information Modelling	Building Information Modelling	Prepare, submit and update BIM Project Execution Plan in accordance with the Contract Documents	10 days	3 per D&C Failure Event	\$150	3 days	Technical Requirements, Division 01, Section 1.10.1.1
1. 68	Building Information Modelling	Building Information Modelling	Appoint a full-time, project-dedicated BIM Manager in accordance with the Contract Documents	5 days	3 per D&C Failure Event	\$500	3 days	Technical Requirements, Division 01, Section 1.10.1.3
1. 69	Construction Management	Construction Security	Prepare, submit and update Project Site Security Plan in	10 days	3 per D&C Failure Event	\$150	3 days	Technical Requirements, Division 01, Section 1.11.2

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
			accordance with the Contract Documents					
1. 70	Construction Management	Maintenance of Traffic and Work Restrictions	Comply with requirements for maintenance of traffic and work restrictions in accordance with the Contract Documents	5 days	3 per D&C Failure Event	\$150	3 days	Technical Requirements, Division 01, Section 1.11.3
1. 71	Construction Management	Temporary Facilities and Utilities	Prepare, submit and update City Temporary Utility Designs and Engineered Drawings in accordance with the Contract Documents	5 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 01, Section 1.11.4.1
1. 72	Construction Management	Temporary Facilities and Utilities	Provide temporary Construction Management Office in accordance with the Contract Documents	5 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 01, Section 1.11.4.3
1. 73	Coordination with Third Parties	Third Party Coordination	Engage a Third Party and Utility Coordination Manager in accordance with the Contract Documents	5 days	3 per D&C Failure Event	\$500	3 days	Technical Requirements, Division 01, Section 1.12.1
1. 74	Coordination with Utility Owners	Utility Coordination Work Plan	Prepare, submit and update Coordination Work Plan (UCWP) in accordance with the Contract Documents	5 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 01, Section 1.13.1

<b>ID</b> 1.	Main Heading Coordination with Utility	Subheading Utility Project	Failure To Prepare, submit and update Utility Project Execution Plan in	Rectification Time 5 days	D&C Noncompliance Points 3 per D&C	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events) \$150	Recurrence Period 3 days	Reference Section(s) of Contract Documents
75	Owners	Execution Plan	accordance with the Contract Documents	5 uays	Failure Event	\$150 	5 uays	Division 01, Section 1.13.2,
1. 76	Coordination with Utility Owners	Utility Tracking Report	Prepare, submit and update Utility Tracking Report in accordance with the Contract Documents	5 days	3 per D&C Failure Event	\$150	3 days	Technical Requirements, Division 01, Section 1.13.3
1. 77	Communicati on and Public Information	Public Outreach and Engagement	Meet requirements to support City Public Outreach and Engagement Program	3 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 01, Section 1.15.2.I1-2
1. 78	Communicati on and Public Information	Public Outreach and Engagement	Prepare, submit and update Public Outreach and Engagement Plan in accordance with Contract Documents	5 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 01, Section 1.15.2.3
2	Division 04 - Supplement al Design Criteria							
2. 1	Supplemental Noise and Vibration Requirements	Proprietary Design Review Deliverables	Comply with submittal requirements for Supplemental Noise and Vibration for Proprietary Design Review Deliverables in accordance with the Contract Documents	5 days	5 per D&C Failure Event	\$300	2 days	Technical Requirements, Division 04, Section 1.5
2. 2	Supplemental Noise and	Proprietary Design Review Deliverables	Prepare, submit and update Testing Report in	5 days	5 per D&C Failure Event	\$300	2 days	Technical Requirements, Division 04, Section 1.5

						D&C Noncompliance		
ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
	Vibration Requirements		accordance with the Contract Documents					
2. 3	Seismic Resilience Performance Requirements	Deliverables	Comply with submittal requirements for seismic resilience performance for Proprietary Design Review Deliverables in accordance with the Contract Documents	5 days	5 per D&C Failure Event	\$300	2 days	Technical Requirements, Division 04, Section 2
3	Division 07 - Infrastructur e Facility Management (IFM) Specification s							
3. 1	IFM Service Procedures	Submission Process	Prepare, submit and update a Schedule of Submittals required under the IFM Specifications before Substantial Completion	10 days	3 per D&C Failure Event	\$150	5 days	Technical Requirements, Division 07, Section B.6.1 (1)
3. 2	IFM Service Procedures	IFM Service Procedures	Prepare, submit and update IFM Service Procedures prior to Substantial Completion as required in accordance with the Contract Documents	10 days	1 per D&C Failure Event	\$50	5 days	Technical Requirements, Division 07, Section B.6.1 (1)

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
3. 3	Move-In Services	Move-In Services	Prepare, submit and update a Move- In Plan no later than six months prior to Substantial Completion	10 days	3 per D&C Failure Event	\$150	5 days	Technical Requirements, Division 07, Section B.25
3. 4	Renewal Work Services	Renewal Work Plans and Reports	Prepare, submit and update Initial 30- Year Renewal Work Plan prior to Substantial Completion	10 days	5 per D&C Failure Event	\$150	5 days	Technical Requirements, Division 07, Section E.2
4	Division 10 - SFPW Div 01 General Requirement s for Construction							
4. 1	Existing Utility Facilities	Governmental Facilities	Comply with requirements related to Governmental Facilities within excavations in accordance with Contract Documents	3 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 10, Section 00 73 20 (1.2)
4. 2	Existing Utility Facilities	Non- governmental Facilities	Comply with requirements related to Non- governmental Facilities in accordance with Contract Documents	3 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 10, Section 00 73 20 (1.3)
4. 3	Utility Crossings	Existing Utility Company	Comply with requirements for supporting working and protecting existing utility company facilities in	3 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 10, Section 00 73 21

ID	Main Heading	Subheading	Failure To accordance with Contract Documents	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
4. 4	Special Instructions	General	Comply with requirements associated with Special Instruction in accordance with Contract Documents	3 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 10, Section 01 12 00
4. 5	Overhead Contact System (OS) Isolation Support	General	Comply with requirements associated with overhead contract system isolation support	3 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 10, Section 01 13 00
4. 6	Artwork Coordination	Artwork Coordination	Comply with requirements for schedules for Pre- Art Installation Conference, Artwork installation and Artwork acceptance in accordance with the Contract Documents	5 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 10, Section 01 14 00 (1.3, 1.4)
4. 7	Artwork Coordination	Artwork Coordination	Comply with requirements by the City and artwork acceptance by Principal Project Company in accordance with the Contract Documents	5 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 10, Section 01 14 00 (1.5, 1.11)

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
4. 8	Partnering Procedures	Partnering Training	Attend, complete, and comply with City Partnering Fundamentals Training in accordance with Training Documents	10 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 10, Section 01 31 33 (1.6)
4. 9	Partnering Procedures	Partnering Elements	Comply with requirements for partnering in accordance with the Contract Documents	10 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 10, Section 01 31 33 (3.2)
4. 10	Photographic Documentatio n	Photographic Documentation	Comply with requirements for administrative and procedural requirements for construction photographs in accordance with the Contract Documents	3 days	3 per D&C Failure Event	\$150	1 day	Technical Requirements, Division 10, Section 01 32 33 (3.1A)
4. 11	Environmenta I Procedures	Submittals	Prepare, submit and update all Submittals to City's Authorized Representative prior to mobilization in accordance with the Contract Documents except those specifically listed in this Table A1 – D&C Failure Event Table with separate Deductions and Noncompliance Points	3 days	3 per D&C Failure Event	-	2 days	Technical Requirements, Division 10, Section 01 35 43 (1.4.A.1)

								Drait November 0, 2024
ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
4. 12	Environmenta I Procedures	Qualifications	Comply with various qualified personnel	3 days	3 per D&C Failure Event	\$1,000	2 days	Technical Requirements, Division 10, Section 01 35 43 (1.4.A.2)
4. 13	Environmenta I Procedures	Submittals	Prepare, submit and update all Submittals to City's Authorized Representative during construction in accordance with the Contract Documents except those specifically listed in this Table A1 – D&C Failure Event Table with separate Deductions and noncompliance Points	3 days	3 per D&C Failure Event	\$1,000	2 days	Technical Requirements, Division 10, Section 01 35 43 (1.4.B)
4. 14	Environmenta I Procedures	Submittals	Prepare, submit and update all submittals to ERO prior to receiving a final certificate of occupancy	3 days	3 per D&C Failure Event	\$1,000	2 days	Technical Requirements, Division 10, Section 01 35 43 (1.4.B)
4. 15	Environmenta I Procedures	Inspection and Monitoring	Cooperate with inspection and monitoring activities, including providing access and making facilities/ records available	3 days	3 per D&C Failure Event	\$300	2 days	Technical Requirements, Division 10, Section 01 35 43 (1.5.C)
4. 16	Environmenta I Procedures	Off-road equipment and engines	Comply with requirements associated with Administrative Code Section 6.25 and	None	3 per D&C Failure Event	\$100	1 day	Technical Requirements, Division 10, Section 01 35 43 (1.5.D)

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
			Chapter 25 of the Environment Code					
4. 17	Environmenta I Procedures	Tree Protection	Provide adequate protection or selective pruning to trees	None	3 per D&C Failure Event	\$250	1 day	Technical Requirements, Division 10, Section 01 35 43 (1.5.E.1)
4. 18	Environmenta I Procedures	Tree Protection	Appropriate presence of Specialty Environmental Monitor - Archaeologist	None	3 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (1.5.F)
4. 19	Environmenta I Procedures	General Construction	Maintain the Project Site and Construction Work areas in a clean and safe condition	None	3 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.1.A)
4. 20	Environmenta I Procedures	Stockpile Management	Implement stockpile management best practices	1 day	3 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.2)
4. 21	Environmenta I Procedures	Dust Control	Ensure dust mitigation measures are in place during construction	None	3 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.3)
4. 22	Environmenta I Procedures	Dust Control	Comply with grading/ excavation including complying with Site-Specific Dust Control Plan (DCP)	None	3 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.3)
4. 23	Environmenta I Procedures	Stormwater Management	Maintain effective project-specific sediments controls	None	3 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.4)

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
4. 24	Environmenta I Procedures	Spills and Leaks	Comply with spills and leaks best management practices	None	3 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.5)
4. 25	Environmenta I Procedures	Emissions- Control	Comply with Emissions- Control best management practices	None	3 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.5)
4. 26	Environmenta I Procedures	Noise Control	Comply with Noise Control best management practices including preparation of Noise- Control Plan	None	3 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.6)
4. 27	Environmenta I Procedures	Asbestos	Comply with regulations for construction impacted by Naturally Occurring Asbestos	None	5 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.8)
4. 28	Environmenta I Procedures	Night Work	Comply with requirements for Night Works in accordance with Contract Documents	None	3 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.10)
4. 29	Environmenta I Procedures	Bird Protection	Comply with environmental regulations for bird protection	None	3 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.12)
4. 30	Environmenta I Procedures	Tree Protection	Arrange meeting with City, General Contractor and others prior to commencement of work	5 days	1 per D&C Failure Event	\$1,000	3 days	Technical Requirements, Division 10, Section 01 35 43 (3.14.A)

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
4. 31	Environmenta I Procedures	Tree Protection	Comply with environmental requirements associated with tree management and protection	None	3 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.14)
4. 32	Environmenta I Procedures	Site Restoration	Comply with environmental requirements associated with site restoration	5 days	3 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.15)
4. 33	Environmenta I Procedures	Paleontological Resources	Comply with environmental requirements associated with unanticipated paleontological resources	None	5 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.16)
4. 34	Environmenta I Procedures	Human Remains	Comply with environmental requirements associated encounter of human remains	None	5 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.17)
4. 35	Environmenta I Procedures	Archaeological Resource Protection	Comply with environmental requirements associated with archaeology protection	None	3 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.18)
4. 36	Environmenta I Procedures	Historical Cultural Resource Protection	Comply with environmental requirements associated with historical cultural resource protection	None	3 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.19)

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
4. 37	Environmenta I Procedures	Vibration Control Plan	Prepare, submit and update Vibration Control Plan in accordance with contract documents	5 days	3 per D&C Failure Event	\$1,000	3 days	Technical Requirements, Division 10, Section 01 35 43 (3.19)
4. 38	Environmenta I Procedures	Work in Air Pollutant Exposure Zone	Comply with environmental requirements associated with SF Environment Code Chapter 25	None	3 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.20)
4. 39	Environmenta I Procedures	Construction Site Runoff Control Permit	Prepare, submit and update Construction Site Runoff Control Permit in accordance with the Contract Documents	5 days	3 per D&C Failure Event	\$1,000	3 days	Technical Requirements, Division 10, Section 01 35 43 (3.21)
4. 40	Environmenta I Procedures	Construction Site Runoff Control Permit	Comply with requirements and conditions associated with the Construction Site Run off Control Permit	None	3 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.21)
4. 41	Environmenta I Procedures	City Water- Quality Permitting	Comply with requirements associated with geotechnical well or soil boring	None	3 per D&C Failure Event	\$1,000	1 day	Technical Requirements, Division 10, Section 01 35 43 (3.23)
4. 42	Hazardous Building Materials - Scope of Work	Hazardous Building Materials - Scope of Work	Prepare, submit and update to City certifications or proof of the environmental trainings in accordance with the Contract Documents	5 days	3 per D&C Failure Event	-	3 days	Technical Requirements, Division 10, Section 01 35 44 (1.1-E)

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
4. 43	Hazardous Building Materials - Scope of Work	Abatement Contractor's Qualifications	Prepare, submit and update Hazardous Materials Management Plan (HMMP) in accordance with the Contract Documents	5 days	3 per D&C Failure Event	-	3 days	Technical Requirements, Division 10, Section 01 35 44 (1.4-D)
4. 44	Hazardous Building Materials - Scope of Work	Waste Handling and Characterization	Prepare, submit and update Waste Management Plan (WMP) as specified under Section 02 80 13 Building Related Hazardous Materials Remediation	5 days	3 per D&C Failure Event	-	3 days	Technical Requirements, Division 10, Section 01 35 44 (1.8-A)
4. 45	Hazardous Building Materials - Scope of Work	Waste Handling and Characterization	Use a bill of lading or non-hazardous waste form in accordance with the Contract Documents when shipping fluorescent lamps to a recycler	5 days	3 per D&C Failure Event	-	3 days	Technical Requirements, Division 10, Section 01 35 44 (1.8-O)
4. 46	Hazardous Building Materials - Scope of Work	Waste Handling and Characterization	Provide, Prepare, submit and update Uniform Hazardous Waste Manifest Form for asbestos hazardous waste shipments in accordance with the Contract Documents	5 days	3 per D&C Failure Event	-	3 days	Technical Requirements, Division 10, Section 01 35 44 (1.8-Q)

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
4. 47	Hazardous Building Materials - Scope of Work	Use of Non- Hazardous Waste Manifest for Class II Material or Lesser	Prepare, submit and update the Non- Hazardous Waste Manifest form for the Generator's signature at least 72 hours in advance in accordance with the Contract Documents	1 day	5 per D&C Failure Event	\$300	0.5 day	Technical Requirements, Division 10, Section 01 35 44 (1.9)
4. 48	Health and Safety Criteria	Health and Safety Plan	Prepare, submit and update a complete Health and Safety Plan (HASP) in accordance with the Contract Documents	3 days	5 per D&C Failure Event	-	1 day	Technical Requirements, Division 10, Section 01 35 45 (1.5)
4. 49	Health and Safety Criteria	Injury and Illness Prevention Programs (IIPP) and Code of Safe Practices (CSP)	Prepare, submit and update Injury and Illness Prevention Programs (IIPP) and Code of Safe Practices (CSP) in accordance with the Contract Documents	3 days	3 per D&C Failure Event	\$150	1 day	Technical Requirements, Division 10, Section 01 35 45 (1.6)
4. 50	Health and Safety Criteria	Submittals	Prepare, submit and update all other health and safety Submittals apart from HASP, IIPP, and CSP in accordance with the Contract Documents	3 days	3 per D&C Failure Event	\$150	1 day	Technical Requirements, Division 10, Section 01 35 45 (1.3A)

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
4. 51	Health and Safety Criteria	Submittals	Prepare, submit and update sufficient information, to demonstrate the qualifications and experience of the it's Project Safety Representative (PSR) in accordance with the Contract Documents	2 days	3 per D&C Failure Event	\$150	1 day	Technical Requirements, Division 10, Section 01 35 45 (1.3B)
4. 52	Health and Safety Criteria	Submittals	Make submissions as required throughout the course of construction - daily inspection reports and records of toolbox meetings in accordance with the Contract Documents	2 days	3 per D&C Failure Event	\$150	1 day	Technical Requirements, Division 10, Section 01 35 45 (1.3C)
4. 53	Health and Safety Criteria	Submittals	Make submissions as required throughout the course of construction - initial and final incident or Near-miss investigation reports in accordance with the Contract Documents	1 day	3 per D&C Failure Event	\$150	0.5 day	Technical Requirements, Division 10, Section 01 35 45 (1.3C)
4. 54	Health and Safety Criteria	Submittals	Make submissions as required throughout the course of construction - HASP modification	2 days	3 per D&C Failure Event	\$150	1 day	Technical Requirements, Division 10, Section 01 35 45 (1.3C)

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
			requests, and others in accordance with the Contract Documents			Events)		
4. 55	Health and Safety Criteria	Project Safety Representative (PSR)	Comply with Project Safety Representative (PSR)	10 days	3 per D&C Failure Event	\$150	5 days	Technical Requirements, Division 10, Section 01 35 45 (1.7, 1.21)
4. 56	Health and Safety Criteria	Contractor's Asbestos Competent Person	Comply with Asbestos Competent Person (ACP)	5 days	3 per D&C Failure Event	\$150	2 days	Technical Requirements, Division 10, Section 01 35 45 (1.8)
4. 57	Health and Safety Criteria	Accident Documentation and Reporting	Comply with accident documentation and reporting in accordance with the Contract Documents	1 day	3 per D&C Failure Event	\$150	0.5 days	Technical Requirements, Division 10, Section 01 35 45 (1.22)
4. 58	Health and Safety Criteria	San Francisco Municipal Transportation Agency (SFMTA) Health and Safety Requirements	Comply with San Francisco Municipal Transportation Agency (SFMTA) Health and Safety Requirements	5 days	3 per D&C Failure Event	\$150	2 days	Technical Requirements, Division 10, Section 01 35 45 (1.24)
4. 59	Additional Environmenta I Procedures	Submittals	Prepare, submit and update all submittals in accordance with the Contract Documents	5 days	3 per D&C Failure Event	-	3 days	Technical Requirements, Division 10, Section 01 35 50 (Part 2)
4. 60	Additional Environmenta I Procedures	Execution	Comply with requirements associated with the execution of the additional environmental procedures	None	3 per D&C Failure Event	-	1 day	Technical Requirements, Division 10, Section 01 35 50 (Part 3)

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
4. 61	Temporary Facilities and Controls	Temporary Project Sign	Comply with Temporary Project Sign in accordance with the Contract Documents	10 days	1 per D&C Failure Event	\$50	5 days	Technical Requirements, Division 10, Section 01 50 00
4. 62	Material Reduction and Recovery Plan	Material Reduction and Recovery Plan	Prepare, submit and update a Demolition Debris Recovery Plan (DDRP) in accordance with the Contract Documents	10 days	5 per D&C Failure Event	\$300	5 days	Technical Requirements, Division 10, Section 01 74 50 (1.4)
4. 63	Material Reduction and Recovery Plan	Material Reduction and Recovery Plan	Comply with all provisions for Material Reduction and Recovery Plan including monthly updates and Final Recovery Report	5 days	3 per D&C Failure Event	\$150	2 days	Technical Requirements, Division 10, Section 01 74 50, (1.5-1.7)
4. 64	Closeout Procedures	Substantial Completion	Prepare, submit and update all Closeout Procedures in accordance with the Contract Documents	5 days	5 per D&C Failure Event	-	3 days	Technical Requirements, Division 10, Section 01 77 00, (1.3)
4. 65	Closeout Procedures	Project Record Documents	Prepare, submit and update the final approved Project Record Drawings to the City Representative prior to Final Acceptance. Refer to Section 01 78 39 - Project Record Documents	5 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 10, Section 01 77 00, (1.6)
4. 66	Operation and Maintenance Data	Operation and Maintenance Data Requirements	Comply with submittal and scheduling requirements for	5 days	5 per D&C Failure Event	-	3 days	Technical Requirements, Division 10, Section 01 78 23, 1.4

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
			Operations and Maintenance Data					
4. 67	Operation and Maintenance Data	Electronic and Hard Copy Formats for O&M Manual	Comply with requirements for submittals of Electronic and Hard Copy Formats for O&M Manual	5 days	5 per D&C Failure Event	-	2 days	Technical Requirements, Division 10, Section 01 78 23, 1.5, 1.6
4. 68	Operation and Maintenance Data	Instruction of City Personnel	Comply with requirements with respect to training schedules and procedures	5 days	3 per D&C Failure Event	-	3 days	Technical Requirements, Division 10, Section 01 78 23, Section 1.8
4. 69	Warranties	Warranties	Comply with all warranty requirements in accordance with the Contract Documents	5 days	5 per D&C Failure Event	-	3 days	Technical Requirements, Division 10, Section 01 78 36, (1.2, 1.3, 1.4)
4. 70	Structure Demolitions	Submittals	Prepare, submit and update all submittals in accordance with the Contract Documents	3 days	1 per D&C Failure Event	-	2 days	Technical Requirements, Division 10, Section 02 41 16 (1.06)
4. 71	Structure Demolitions	Execution	Comply with requirements associated with the execution of structure demolition	None	3 per D&C Failure Event	-	1 day	Technical Requirements, Division 10, Section 02 41 16 (Part 3)
4. 72	Hazardous Building Materials Remediation	Submittals	Perform all Hazardous Materials remediation work as per this specification	5 days	3 per D&C Failure Event	-	2 days	Technical Requirements, Division 10, Section 02 80 13

<b>ID</b> 4. 73	Main Heading Environmenta I Management of Excavated Materials	Subheading Environmental Management of Excavated Materials	Failure To Comply with requirements for Environmental Management of Excavated Materials	Rectification Time 5 days	D&C Noncompliance Points 3 per D&C Failure Event	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period 2 days	Reference Section(s) of Contract Documents         Technical Requirements, Division 10, Section 02 81 10
5	Contracting and Labor Practices		Note that DBE Requirements are still being developed					
5. 1	Contracting and Labor Practices	Labor Standards	Comply and require all Contractors to comply, with all applicable federal and State labor, occupational safety and health Laws and orders, including payment of prevailing wages	N/A	3 per D&C Failure Event	-	N/A	DBFOM Agreement, Section 9.8 & Exhibit 16B
5. 2	Contracting and Labor Practices	Labor Standards	Submit to City certified payroll records for all employees of Principal Project Company and Contractors at all tiers for the preceding calendar month	5 days	5 per D&C Failure Event	-	N/A	DBFOM Agreement, Section 9.8
5. 3	Contracting and Labor Practices	Labor Standards	Require that individuals performing the Work be qualified, experienced, competent, and skilled in the performance of the	N/A	5 per D&C Failure Event	-	N/A	DBFOM Agreement, Section 9.8

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
			Work and related obligations as per requirement					
5. 4	Contracting and Labor Practices	Local Hiring Requirements for Construction Work and Renewal Work	Comply with local hiring requirements for Construction Work and Renewal Work as per requirements	5 days	5 per D&C Failure Event	-	N/A	DBFOM Agreement, Section 9.9
5. 5	Contracting and Labor Practices	First Source Hiring Program	Comply with all applicable provisions of the First Source Hiring Program as per requirement	5 days	5 per D&C Failure Event	-	N/A	DBFOM Agreement, Section 9.10
5. 6	Contracting and Labor Practices	SFMTA Employment Training Program	Comply with all applicable provisions of the SFMTA Employment Training Program	5 days	5 per D&C Failure Event	-	N/A	DBFOM Agreement, Section 9.11
6	Commission ing and Testing Requirement s							
6. 1	Commissioni ng Provider (CxP)	Commissioning Provider (CxP) appointment	Prepare, submit and update details on the appointment of the CxP entity for review and acceptance by City	10 days	3 per D&C Failure Event	\$150	5 days	Technical Requirements, Division 6, Section6.5
6. 2	Pre- Construction Requirements	Commissioning Plan	Prepare, submit and update Final Commissioning Plan in accordance with the Contract Documents	5 days	3 per D&C Failure Event	\$150	3 days	Technical Requirements, Division 6, Section 6.6.2.1

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
6. 3	Pre- Construction Requirements	Commissioning Issues and Resolution Log	Maintain Commissioning Issues and Resolution Log in accordance with the Contract Documents	5 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 6, Section 6.6.2.3 and 6.6.3.2
6. 4	Construction Requirements	Functional Performance Testing (FPT)	Provide notice and results of Functional Performance Testing in accordance with the Contract Documents	5 days	3 per D&C Failure Event	\$150	3 days	Technical Requirements, Division 6, Section 6.6.3.6
6. 5	Construction Requirements	Commissioning Report	Prepare, submit and update Commissioning Report in accordance with the Contract Documents	5 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 6, Section 6.6.3.7
6. 6	Post Substantial Completion Requirements	Deliver Systems Manual	Prepare, submit and update the necessary operating documents and reports to City in accordance with the Contract Documents	10 days	1 per D&C Failure Event	\$50	10 days	Technical Requirements, Division 6, Section 6.6.4.1
6. 7	Post Substantial Completion Requirements	Post Occupancy Review of Building Operations	Meet requirements associated with the Post Occupancy Review of Building Operations in accordance with Contract Documents	10 days	3 per D&C Failure Event	\$150	3 days	Technical Requirements, Division 6, Section 6.6.4.2
6. 8	Post Substantial Completion Requirements	Near Warranty End Post Occupancy Review	Meet requirements associated with the Warranty End Post Occupancy Review in accordance with Contract Documents	10 days	3 per D&C Failure Event	\$150	3 days	Technical Requirements, Division 6, Section 6.6.4.3

ID	Main Heading	Subheading	Failure To	Rectification Time	D&C Noncompliance Points	D&C Noncompliance Event Deductions in \$ (for New or Recurred Events)	Recurrence Period	Reference Section(s) of Contract Documents
6. 9	Post Substantial Completion Requirements	On-going Commissioning Plan	Prepare, submit and update the Ongoing Commissioning Plan in accordance with the Contract Documents	5 days	1 per D&C Failure Event	\$50	3 days	Technical Requirements, Division 6, Section 6.6.4.4
6. 10	Operational Readiness	Operational Readiness	Prepare, submit and update an integrated Operational Readiness Plan in accordance with the Contract Documents	5 days	5 per D&C Failure Event	\$300	3 days	Technical Requirements, Division 6, Section 6.8.

# EXHIBIT 4B

# **AVAILABILITY PAYMENT MECHANISM**

# 3 Availability Payments

# 3.1 Maximum Availability Payment

The Maximum Availability Payment for each Contract Year (y) ("MaxAPy") shall be calculated for any period commencing on or after the Payment Commencement Date as follows:

$$MaxAP_y = MaxAPC_y + (MAPIFM_y \times ESC_y)$$

Where:

- *MaxAPC*<sub>y</sub> means the capital portion of the MaxAP for Contract Year y as set out in <u>Appendix D</u> (Payment Schedule) to this <u>Exhibit 4B</u> (Availability Payment Mechanism)
- *MaxAPIFMy* means the IFM Payment MaxAP for Contract Year y, as set out in <u>Appendix D</u> (Payment Schedule) to this <u>Exhibit 4B</u> (Availability Payment Mechanism)
- *ESC<sub>y</sub>* is the escalation factor calculated in accordance with <u>Section 4</u> (Escalation Factor) of this <u>Exhibit 4B</u> (Availability Payment Mechanism)

# 3.2 Maximum Quarterly Availability Payment

Subject to any limitations and exceptions expressly provided in the Agreement, Availability Payments shall be paid to the Principal Project Company through Quarterly Availability Payments. The Maximum Quarterly Availability Payment ("MaxQAP") shall be calculated as the quarterly amount of the MaxAP in accordance with the following formula.

$$MaxQAP_{q,y} = \frac{1}{4} \times MaxAP_{y}$$

Where:

*MaxAP<sub>y</sub>* is the Maximum Availability Payment for Contract Year **y** 

MaxQAP<sub>q,y</sub> is the MaxQAP for Contract Quarter q in Contract Year **y** 

# 3.3 Availability Payments

**3.3.1** The Maximum Availability Payment ("MaxAP") shall be subject to Deductions, if any, with respect to the availability and performance of the Infrastructure Facility. Each Quarterly Availability Payment paid to the Principal Project Company in Contract Year **y** shall be calculated as follows:

$$QAP_{q,y} = MaxQAP_{q,y} - \sum D_{q,y} + ENC_{q,y} + \sum OA_{q,y}$$

where:

**QAP**<sub>q,y</sub> is the Quarterly Availability Payment for Contract Quarter **q** in Contract Year **y**;

**MaxQAP**<sub>q,y</sub> is the Maximum Quarterly Availability Payment for Contract Quarter **q** in Contract Year **y**;

 $\sum D_{q,y}$  is the aggregate sum of all Deductions for Contract Quarter q in Contract Year y in relation to D&C Failure Events (to the extent not deducted from the Milestone Payment), Service Failures and Availability Failures calculated in accordance with the provisions set out in Section 5 (Deductions from Availability Payments) of this Exhibit 4B (Availability Payment Mechanism);

 $ENC_{q,y}$  is Energy Net Compensation (determined pursuant to the Energy Pain Share/Gain Share Mechanism) for Contract Quarter **q** in Contract Year **y**;

 $\sum OA_{q,y}$  is the aggregate of any other adjustments for Contract Quarter q in Contract Year y to reflect previous over-payments and/or under-payments, any interest payable in respect of any amounts owed, and, except as set out in Section 11.3.2(ii) of the Agreement any Other Amounts due and payable from Principal Project Company to the City or from the City to Principal Project Company under the Agreement.

**3.3.2** If in the Contract Quarter in which the Payment Commencement Date falls or in the last Contract Quarter of the Term, a pro rata adjustment shall be made to reflect the actual number of days in the Contract Quarter from and including the Payment Commencement Date (for the first Contract Quarter) and up to and including the last day of the IFM Period (for the last Contract Quarter) for purposes of calculating the Quarterly Availability Payments and Deductions.

# 3.4 Invoicing and Payment of Quarterly Availability Payments

For each Quarterly Availability Payment, Principal Project Company shall submit an invoice no later than the 10<sup>th</sup> day of the month immediately following the relevant Contract Quarter in accordance with the process in <u>Section 11.3</u> (Invoice, Other Amounts and Payment) of the Agreement. The City will pay Principal Project Company the Quarterly Availability Payment accordance with the process in <u>Section 11.3</u> (Invoice, Other Amounts and Payments) of the Agreement.

# 4 ESCALATION FACTOR

**4.1** The Escalation Factor shall be calculated in accordance with the following formula:

$$ESC_y = \frac{CPI_y}{CPI_0}$$

where:

**ESC**<sub>y</sub> is the escalation factor applicable to the relevant Contract Year y;

 $CPI_y$  Consumer Price Index as of the reference month that is available in the year that is immediately prior to the commencement of Contract Year y, as published by US Bureau of Labor Statistics, for Urban Consumers, for the San Francisco-Oakland-Hayward, CA areas (All Items).  $ESC_y$  shall apply to all calculations relating to Contract Year y; and

**CPI**<sub>0</sub> is the prevailing value of Consumer Price Index as of the Effective Date, published by US Bureau of Labor Statistics, for Urban Consumers for the San Francisco-Oakland-Hayward, CA areas (All Items).

# 5 DEDUCTIONS FROM AVAILABILITY PAYMENTS

# 5.1 Entitlement to Make Deductions

- **5.1.1** Subject to the provisions of this Agreement, if at any time during the Term, a D&C Failure Event (to the extent not deducted from the Milestone Payment), Service Failure or Availability Failure occurs, the City may make a Deduction from the relevant Maximum Availability Payment in respect of that Failure Event, except that:
  - the maximum aggregate Deduction amount that the City can make from a Quarterly Availability Payment shall not exceed 35% of the Maximum Quarterly Availability Payment ("MaxQAPIFMq") for that Contract Quarter ("Quarterly Deduction Cap"); and
  - (b) if the aggregate Deduction accrued for a Contract Quarter exceed the Quarterly Deduction Cap, the Deduction amount in excess of the Quarterly Deduction Cap shall be applied to the next Quarterly Availability Payment where the Quarterly Deduction Cap has not been exceeded.

# 5.2 Classification of IFM Failure Events

5.2.1 The classification of an IFM Failure Event as a potential Availability Failure or a Service Failure, shall be made at the time at which the occurrence of the IFM Failure Event is reported to the Help Desk. For clarity, this classification is for the identification of applicable Response Time and Rectification Time, however the IFM Failure Event shall not be considered (a) an Availability Failure unless it causes a Functional Unit or Functional Component (as the case may be) to be Unavailable and the IFM Failure Event is not Rectified within the Rectification Time; or (b) a Service Failure, if the IFM Failure Event is not Responded to within the Response Time or Rectified within the Rectification Time; or the or Rectified within the Rectification Time or Rectification which do not constitute a breach of the Availability Conditions). If, in accordance with Performance Measurements Table, no Rectification Time or Response Time applies, the IFM Failure Event shall be a

Service Failure An IFM Failure Event which is incorrectly classified may be re-classified with the approval of Authorized Representatives, acting reasonably, in which case the applicable Performance Monitoring Report will be revised accordingly.

- **5.2.2** For the purposes of calculating Deductions and Noncompliance Points, and noting that classification of IFM Failure Events shall proceed in accordance with <u>Section 5.2.1</u>,
  - the identification of Availability Failures will be made in accordance with the Functional Units and Components Table included in <u>Appendix B</u> (Functional Areas, Functional Units, Functional Unit Rankings and Deduction Amounts for Availability Failures and Services Failures) to this <u>Exhibit 4B</u> (Availability Payment Mechanism); and
  - (b) the identification of Service Failures shall be made in accordance with the Performance Measurements included in the Performance Measurements Table and with reference to <u>Appendix A</u> (Space Tolerances) of <u>Division 7</u> (IFM Specifications) of <u>Exhibit 18</u> (Technical Requirements).

# 5.3 Bedding-In Period and Seasonal Bedding-In Period

- **5.3.1** During the Bedding-In Period, the number of Noncompliance Points and the amount of any Deductions in respect of Service Failures occurring in the provision of any IFM Service shall be reduced by 25%.
- **5.3.2** For the avoidance of doubt, there shall be no relief from Noncompliance Points or Deductions relating to Availability Failures during the Bedding-In Period.

# 6 SERVICE FAILURE DEDUCTIONS

### 6.1 Service Failure Deductions

- **6.1.1** Service Failure Deductions shall be applied in accordance with <u>Section 15.2</u> of the Agreement and this <u>Exhibit 4B</u>.
- **6.1.2** Subject to this <u>Section 6</u> (Deductions for Service Failures), the amount of the Service Failure Deduction shall be as specified below.
  - in the case of a Minor Service Failure, the sum of \$75 per Service Failure, indexlinked using the Escalation Factor calculated in accordance with <u>Section 4</u> (Escalation Factor) of this <u>Exhibit 4B</u> (Availability Payment Mechanism);
  - (b) in the case of a Medium Service Failure, the sum of \$300 per Service Failure, index-linked using the Escalation Factor calculated in accordance with <u>Section 4</u> (Escalation Factor) of this Exhibit 4B (Availability Payment Mechanism); and
  - in the case of a Major Service Failure, the sum of \$650 per Service Failure, indexlinked using the Escalation Factor calculated in accordance with <u>Section 4</u> (Escalation Factor) of this <u>Exhibit 4B</u> (Availability Payment Mechanism).

# 6.2 Performance Action Plans and Energy Performance Action Plans

- **6.2.1** Missed submittals of Performance Action Plans or failure to implement particular actions within the Performance Action Plan are to be considered separate and distinct Service Failure Deductions as described below;
  - (a) where a complete Performance Action Plan is not submitted within the due date as identified in [Section B.16.2 of Division 7 (IFM Specifications) of Exhibit 18 (Technical Requirements)], a Performance Action Plan deduction of \$300 as of the first missed due date, index-linked using the Escalation Factor calculated in accordance with Section 4 (Escalation Factor) of this Exhibit 4B (Availability Payment Mechanism);
  - (b) for every 7 calendar days following the first due date where that Performance Action Plan is not submitted, a further Deduction of \$300 applies, index-linked using the Escalation Factor calculated in accordance with <u>Section 4</u> (Escalation Factor) of this <u>Exhibit 4B</u> (Availability Payment Mechanism); and
  - (c) where each key activity listed in the Performance Action Plan is not implemented according to the milestone date included in the Performance Action Plan, a Deduction of \$300 applies. A further \$300 Deduction applies for each key activity milestone still not implemented every 14 calendar days following the first milestone date, index-linked using the Escalation Factor calculated in accordance with <u>Section 4</u> (Escalation Factor) of this <u>Exhibit 4B</u> (Availability Payment Mechanism).
- **6.2.2** Missed submittals of Energy Performance Action Plans or failure to implement particular actions within the Energy Performance Action Plan are to be considered separate and distinct Service Failure Deductions as described below;
  - (a) where a complete Energy Performance Action Plan is not submitted within the due date as identified in <u>Section 4.1.1</u> of <u>Exhibit 12</u> Energy Management], an Energy Performance Action Plan deduction of \$500 applies as of the first missed due date, index-linked using the Escalation Factor calculated in accordance with <u>Section 4</u> (Escalation Factor) of this <u>Exhibit 4B</u> (Availability Payment Mechanism);
  - (b) for every 7 days following the first due date where that Energy Performance Action Plan is not submitted, a further Deduction of \$500 applies, index-linked using the Escalation Factor calculated in accordance with <u>Section 4</u> (Escalation Factor) of this <u>Exhibit 4B</u> (Availability Payment Mechanism); and
  - (c) where each key activity listed in the Energy Performance Action Plan is not implemented according to the milestone date included in the Energy Performance Action Plan, a Deduction of \$2,000 applies. A further \$2,000 Deduction applies for each key activity milestone still not implemented every 14 days following the first milestone date, index-linked using the Escalation Factor calculated in accordance with <u>Section 4</u> (Escalation Factor) of this <u>Exhibit 4B</u> (Availability Payment Mechanism).

### 6.3 Repeat Service Failures and Repeat Failure Ratchet

- **6.3.1** Following the occurrence of a Service Failure Deduction, Principal Project Company shall be allowed an additional Rectification Time equivalent to the original Rectification Time, provided that:
  - there shall be no Rectification Time (where applicable) for a second Service Failure of the same Performance Measurement which occurs within the same Day; and
  - (b) where three or more Service Failures occur with respect to the same Performance Measurement in any rolling period of 10 days, then a repeat failure ratchet of 150% of the applicable <u>Deduction</u> ("Repeat Failure Rachet") shall be applied to the third and every subsequent Service Failure Deduction for each Service Failure related to the same Performance Measure in such 10-day period.
- **6.3.2** The provisions of this <u>Section 4.3</u> (Repeat Service Failures and Repeat Failure Ratchet) of <u>Exhibit 4B</u> (Availability Payment Mechanism) shall not apply to Service Failures in cases where, if the Response or Rectification is not carried out within the Response Time or the Rectification Time, as applicable, the City's Authorized Representative notifies Principal Project Company's Authorized Representative that the City no longer requires the relevant IFM Services, in which cases no Deduction shall be applied.
- **6.3.3** Where a Service Failure has no Response Time or Rectification Time, a Service Failure shall occur upon the occurrence of the IFM Failure Event in question.
- **6.3.4** Following the occurrence of a Service Failure Deduction, Principal Project Company shall be allowed additional Rectification Time and a Service Failure Deduction shall apply in accordance with this <u>Section 6</u> (Service Failure Deductions) of <u>Exhibit 4B</u> (Availability Payment Mechanism) and shall repeat as follows:
  - (a) every 24 hours until rectified for Major Service Failures;
  - (b) every 72 hours for Medium Service Failures; or
  - (c) every 7 calendar days for Minor Service Failures.

# 7 DEDUCTIONS FOR AVAILABILITY FAILURES

### 7.1 Unavailability Deductions

- **7.1.1** Unavailability Deductions shall be applied in accordance with <u>Section 15.2</u> of the Agreement and this <u>Exhibit 4B</u>.
- **7.1.2** Subject to this <u>Exhibit 4B</u> (Availability Payment Mechanism), the amount to be deducted from the Maximum Availability Payment in respect of any Availability Failure shall be the aggregate of the Availability Failure Deduction amounts for all Functional Components

made Unavailable as a result of the Availability Failure, calculated based on the duration the Functional Component is Unavailable.

- 7.1.3 Availability Failure Deductions will be calculated in accordance with the conditions and deduction thresholds included in <u>Section 7</u> (Deductions for Availability Failures) of <u>Exhibit 4B</u> (Availability Payment Mechanisms) and <u>Appendix A</u> (Response and Rectification Times) to this <u>Exhibit 4B</u> (Availability Payment Mechanism) and will be indexed using the Escalation Factor calculated in accordance with <u>Section 4</u> (Escalation Factor) of this <u>Exhibit 4B</u> (Availability Payment Mechanism).
- **7.1.4** If any of the Functional Unit are rendered Unavailable but the City continues to use it (or any part thereof), the Availability Failure Deduction for that Functional Unit will be reduced by 50% and the IFM Noncompliance Points will be reduced by 50%. Principal Project Company shall be obliged to continue to provide IFM Services in respect of that Functional Part such that Performance Requirements are normally provided as part of day to day functioning, acting reasonably, and any failure to provide the same may, subject to the provisions of this <u>Exhibit 4B</u> (Availability Payment Mechanism), give rise to a Service Failure.

# 7.2 Service Failure Becoming an Availability Failure

A Service Failure may become or lead to an Availability Failure if circumstances change, or the Service Failure continues. In such a circumstance, if a Functional Part becomes Unavailable, the Service Failure will have ended (without prejudice to the Service Failure Deductions that have accrued to that point) and an Availability Failure will have occurred.

### 7.3 Effect of Availability Failures on Service Failures

**7.3.1** Until an Availability Failure has been Rectified, the Deduction in respect of the Availability Failure shall be the only Deduction available to be made in respect of any Functional Part in which the Availability Failure has occurred. No Service Failure Deduction shall be made for any Service Failure that occurs subsequent to the Availability Failure which may occur in the relevant Functional Part during the period until Rectification has been completed.

### 7.4 Total Unavailability

- **7.4.1** Total Unavailability means, that all of the following events have occurred at any given time:
  - (a) 40% or greater of all Revenue Vehicle Parking Spaces are Unavailable;
  - (b) 100% of the Functional Components associated with the Dispatch Function are Unavailable: and
  - (c) the Meet and Greet is Unavailable.
- **7.4.2** In the event of Total Unavailability, all Functional Units in the Infrastructure Facility shall be deemed to be Unavailable for purposes of calculating deductions associated with Availability Failures.

# 7.5 Repeat Availability Failures and Repeat Failure Rachet

- **7.5.1** Following the occurrence of an Availability Failure and the relevant Deduction has been applied, Principal Project Company shall be allowed an additional Rectification Time equivalent to the initial Rectification Time, provided that;
  - (a) there shall be no Rectification Time for a second Availability Failure which occurs within the same Day, of the same type of Availability Failure and which relates to the same Functional Component or to a different Functional Component within the same Functional Unit and there is reason to believe that the root cause of each Availability Failure is the same; and
  - (b) where three or more Availability Failures occur with respect to the same type of Availability Failure related to the same Functional Part in any rolling period of 20 days, then a Repeat Failure Ratchet of 150% of the applicable Availability Failure shall be applied to the third and every subsequent Availability Failure Deduction for each Availability Failure related to the same Availability Failures and Functional Part in such 20-day period.

### 7.6 Unavailable Component Tiers and Thresholds

**7.6.1** For Revenue Vehicle Parking Spaces and Non-Revenue Vehicle Parking Spaces, City shall be entitled to make Deductions with respect to the number of Unavailable Functional Components as a percentage of the Functional Unit, in a tiering system as per <u>Table 2</u> (Unavailable Tiers for Revenue / Non-Revenue Parking) below.

Table 2 – Unavailable Tiers for Revenue/ Non-Revenue Vehicle Parking	
Spaces	

Unavailable % of Functional Unit	Deductions
0-9% of Functional Unit	Deductions to be applied based on number of Unavailable Functional Components
10- 19% of Functional Unit	20% of Functional Unit
20 – 29% of Functional Unit	40% of Functional Unit
30 – 39% of Functional Unit	60% of Functional Unit
40 – 49% of Functional Unit	80% of Functional Unit
50- 100% of Functional Unit	100% of Functional Unit

**7.6.2** For the Functional Units included in <u>Table 3</u> (Unavailable Component Thresholds) below, if any combination of Availability Failures occur that render a number of Functional Components Unavailable that exceeds the Unavailable Component Threshold for that Functional Unit, all of the Functional Components in that Functional Unit will be deemed Unavailable until the number of Unavailable Functional Components falls below the Unavailable Component Threshold.

# Table 3– Unavailable Component Thresholds

Item	Functional Unit Reference	Unavailable Unit Threshold
Elevators at grid A7-B8 (Levels B1, Ground, Mezzanine)	AL-016A	1
60' Bus Repair Bays	GL-011	4
60' Preventative Maintenance Bays	GL-012	2

# 7.7 Rectification

- **7.7.1** No Availability Failure shall occur if Principal Project Company successfully carries out the Rectification within the specified Rectification Time and in such circumstances no Deduction shall be made and no Noncompliance Points shall be assigned.
- **7.7.2** When carrying out a Rectification, or a Temporary Repair pursuant to <u>Section 9</u> (Temporary Repairs) of this <u>Exhibit 4B</u> (Availability Payment Mechanism), Principal Project Company shall act in accordance with Applicable Law and Standards and Good Industry Practice. Failure to do so shall be deemed to be a new Minor Service Failure, unless the failure constitutes a breach of applicable Law, in which case a Major Service Failure will be deemed to have occurred.

# 8 IFM NONCOMPLIANCE POINTS SYSTEM

# 8.1 IFM Noncompliance Points Application

- 8.1.1 The provisions included in <u>Sections 6</u> (Deductions for Availability Failures) of this <u>Exhibit 4B</u> (Availability Payment Mechanism) notwithstanding, IFM Noncompliance Points shall accrue to relevant IFM Failure Event in accordance with this <u>Section 8.1</u>, <u>Section 15.4</u> of the Agreement and <u>Appendix C</u> (IFM Noncompliance Points) to this <u>Exhibit 4B</u> (Payment Mechanism).
- **8.1.2** In the case of failure in more than one Functional Component, the number of IFM Noncompliance Points to be awarded in respect of that the IFM Failure Event shall be determined by the number of Functional Components affected in accordance with conditions and thresholds included in <u>Appendix C</u> (IFM Noncompliance Points) to this <u>Exhibit 4B</u> (Payment Mechanism).
- **8.1.3** The maximum aggregate IFM Noncompliance Points that the City can award in any one Response Time or Rectification Time shall be 250 IFM Noncompliance Points, regardless of the number of Functional Components impacted by such IFM Failure Event.
- **8.1.4** Principal Project Company will report on all accumulated IFM Noncompliance Points in the Performance Monitoring Report, including a report on the accumulated IFM Noncompliance Points per Functional Part, an explanation of the reasons for the accumulation, total IFM Noncompliance Points for the previous quarters and Contract Year and the total cumulative IFM Noncompliance Points for the period since the commencement of the Payment Commencement Date.

8.1.5 The cumulative points record over the IFM Period will be used for purposes of tracking thresholds leading to Performance Notices, Increased Oversight, cure plans and assessing Persistent PPC Default and other related actions as defined in Article 15 (Deductions and Noncompliance Points) of the Agreement.

#### 9 **TEMPORARY REPAIRS**

- 9.1 If Principal Project Company informs the City that it is unable to Rectify an IFM Failure Event within the specified Rectification Time due to the need for specialized materials or personnel that are not, and cannot reasonably be expected to be, immediately available at the Infrastructure Facility but that a Temporary Repair can be effected:
  - the Citv shall permit Principal Project Company to carry out the Temporary Repair (a) proposed by Principal Project Company unless the City, acting reasonably, considers that, if the Temporary Repair proposed by Principal Project Company is carried out, the use of the relevant Functional Part will not be in accordance with Good Industry Practice or applicable Laws; and
  - where a Temporary Repair is permitted pursuant to Section 8.1(a), a deadline by (b) which a Permanent Repair must be made shall be agreed to by the Parties, each acting reasonably, giving Principal Project Company a reasonable period within which to carry out the Permanent Repair (the "Permanent Repair Deadline").
- 9.2 During any period beginning at the time when a Temporary Repair is permitted and ending at the earlier of:
  - the time at which a Permanent Repair is successfully completed; and (a)
  - (b) the Permanent Repair Deadline.

the Availability Conditions shall be replaced by the Minimum Agreed Availability Conditions, to be agreed between the City's Authorized Representative and the Principal Project Company's Authorized Representative, each acting reasonably, for the purposes of assessing if the relevant Functional Part is Unavailable.

- 9.3 If the Response Time (where applicable) is met, and Temporary Repair is effected within the specified Rectification Time and the Permanent Repair is effected by no later than the Permanent Repair Deadline, no Availability Failure or Service Failure occur, and no Deduction may be made, in respect of the IFM Failure Event.
- 9.4 If the Temporary Repair is not effected within the specified Rectification Time, a Service Failure or, as the case may be, Availability Failure, shall be deemed to occur and the following provisions shall apply:
  - (a) there shall be a further period in which to effect the Temporary Repair beginning at the expiry of the Rectification Time and of a duration equal to that of the original Rectification Time;
  - Principal Project Company shall ensure that the Temporary Repair is successfully (b) carried out prior to the expiry of the additional period referred to in Section 9.4(a):

- if the Temporary Repair is not successfully carried out prior to the expiry of the (c) additional period referred to in Section 8.4(a), a further Service Failure or, as the case may be, Availability Failure shall occur and a further additional period shall commence: and
- (d) if the Temporary Repair is not successfully carried out prior to the Permanent Repair Deadline, and no Permanent Repair has been successfully carried out, the right for Principal Project Company to carry out a Temporary Repair pursuant to this Section 9 (Temporary Repairs) of Exhibit 4B (Availability Payment Mechanism) shall cease and Sections 6 (Deductions for Service Failures) and 7 (Deductions for Availability Failures) of this Exhibit 4B (Availability Payment Mechanism) shall apply.

#### 10 AMENDMENT OF IFM PERIOD PERFORMANCE MEASURES TABLE

Section 15.9 (Amendment of IFM Period Performance Measurement Table) of the Agreement shall apply to any amendment of the Performance Measures Table.

#### 11 FAILURE BY PRINCIPAL PROJECT COMPANY TO MONITOR OR REPORT

- 11.1 The Performance Monitoring Reports produced by Principal Project Company for any month shall be the source of factual information regarding the performance of the IFM Services for the relevant Contract Quarter for the purposes of reporting the occurrence of all Availability Failures and Service Failures, and the calculation of resulting Deductions and Noncompliance Points in accordance with the provisions of this Exhibit 4B (Availability Payment Mechanism).
- 11.2 Principal Project Company shall also report the aggregate Deductions arising as a result of Availability Failures and Service Failures in accordance with the categories and rolling Contract Quarter periods identified for the purposes of measuring the performance of the IFM Services in terms of the amount of Deductions incurred. If there is any error or omission in the Performance Monitoring Report for any Contract Month, Principal Project Company and the City shall agree to an amendment to the Performance Monitoring Report or, failing agreement within 10 days of notification of the error or omission, which shall be made not more than 2 calendar months following the submission of the applicable Performance Monitoring Report to the City, except in the "relevant circumstances" referred to in this Section 10 (Failure by Principal Project Company to Monitor or Report) either Party may refer the matter to the Contract Dispute Procedures.
- 11.3 If Principal Project Company fails to monitor or accurately report an IFM Failure Event (Service Failure or Availability Failure) then, without prejudice to the Deduction to be made in respect of the relevant Service Failure or Availability Failure (if any), the failure to monitor or report the IFM Failure Event shall be deemed to be a new Minor Service Failure, unless the "relevant circumstances" set out below apply, in which case it shall be deemed to be a new Major Service Failure.
- 11.4 In the event that any inspection or investigation by the City of records made available pursuant to the Agreement reveals any further matters of the type referred to in this Section 10 (Failure by Principal Project Company to Monitor or Report), those matters

shall be dealt with in accordance with Section 9 (Amendment of IFM Period Performance Measures Table) of this <u>Exhibit 4B</u> (Availability Payment Mechanism), as appropriate, and the City shall, in addition, be entitled to make Deductions in respect of any Service Failures or Availability Failures in the manner prescribed in <u>Sections 6</u> (Service Failure Deductions), <u>7</u> (Deductions for Availability Failures) and <u>8</u> (IFM Noncompliance Points System) of this <u>Exhibit 4B</u> (Availability Payment Mechanism). Any such Deductions shall be made from the Maximum Quarterly Availability Payment payable in respect of the Contract Quarter in which the relevant matters were revealed by the City's investigations.

- **11.5** For the purposes of this <u>Section 11</u> (Failure by Principal Project Company to Monitor or Report) of <u>Exhibit 4B</u> (Availability Payment Mechanism), the "relevant circumstances" are:
  - (a) fraudulent action or inaction;
  - (b) deliberate misrepresentation; or
  - (c) gross misconduct or incompetence in each case on the part of PPC-Related Entities
- **11.6** The provisions of this <u>Section 11</u> (Failure by Principal Project Company to Monitor or Report) of <u>Exhibit 4B</u> (Availability Payment Mechanism)shall be without prejudice to any rights of the City pursuant to the relevant sections of the Agreement.

# 12 EXTENSION OF RESPONSE TIME OR RECTIFICATION TIME

- **12.1** Principal Project Company will be entitled to an extension of the duration of the Response Time or Rectification Time (as applicable) if Principal Project Company is unable to Respond to or Rectify a Service Failure or Availability Failure due to:
  - (a) a direction of the City's Authorized Representative to delay or reschedule the Response, or Rectification;
  - (b) Principal Project Company performing Scheduled Maintenance and Renewal Work;
  - (c) Principal Project Company not being able to access the applicable portion of the Project Site due to work performed by City or Other Contractors, but only to the extent and for the period that Principal Project Company's ability to Respond or Rectify the Noncompliance Event was directly and adversely affected; or
  - (d) a Relief Event, but only to the extent and for the period that Principal Project Company's ability to Respond to or Rectify the Service Failure or Availability Failure was directly and adversely affected, as determined by the City's Authorized Representative, acting in good faith.

### APPENDIX A RESPONSE AND RECTIFICATION TIMES

Service Failures and/or Availability Failures (unless specifically identified otherwise), will only occur upon the expiry of the following Response and Rectification Times in accordance with this <u>Appendix A</u> (Response and Rectification Times) of this <u>Exhibit 4B</u> (Availability Payment Mechanism) and the Performance Measurement Table.

# **RESPONSE AND RECTIFICATION TIMES**

Category	Definition	Response Time	Rectification Time
Priority 1	Situations requiring immediate action to return the Infrastructure Facility to normal operations, stop accelerated deterioration, or correct a safety hazard that imminently threatens life or serious injury to public and/or City employees.	15 minutes	4 hours (for all Functional Component Rank categories)
Priority 2	Situations that will imminently become critical, if not corrected expeditiously, includes intermittent interruptions and/or potential safety hazards.	The longer of 30 minutes or prior to the resumption of core work hours	The longer of 8 hours or prior to the resumption of core work hours (for all FunctionalComponent Rank categories)
Priority 3	Conditions requiring appropriate attention to preclude deterioration or potential downtime and associated damage or higher costs if deferred further. Items representing a practical improvement to existing conditions. These items are not required for the most basic functions of the Infrastructure Facility but will improve the overall usability and accessibility and/or reduce long- term maintenance.	The longer of 2 hours or prior to the resumption of core work hours	7 calendar days

### Table A1 – Service Failure Response and Rectification Times by Event Category

Table A2 – Availability Failure Response and Rectification Times per Functional
Unit Rank Category

Functional Component Rank Category	Response Time	Rectification Time
5	15 minutes	4 hours
4	15 minutes	8 hours
3	The longer of 30 minutes or prior to the resumption of core work hours	12 hours
2	The longer of 30 minutes or prior to the resumption of core work hours	24 hours
1	The longer of 60 minutes or prior to the resumption of core work hours	48 hours

# APPENDIX B

### FUNCTIONAL AREAS, FUNCTIONAL UNITS, FUNCTIONAL UNIT RANKINGS, CORE HOURS, AND DEDUCTION AMOUNTS FOR AVAILABILITY FAILURES

### Table B1: Deduction Amounts for Availability Failures – Facility

Functional Unit Rank Category	Deduction Amount (per Rectification Time period)
5	\$2,500, index-linked using the Escalation Factor calculated in accordance with <u>Section 2</u> (Escalation Factor) of this <u>Exhibit</u> <u>4B</u> (Availability Payment Mechanism)
4	\$1,000, index-linked using the Escalation Factor calculated in accordance with <u>Section 2 (Escalation Factor)of this Exhibit</u> <u>4B</u> (Availability Payment Mechanism)
3	\$550, index-linked using the Escalation Factor calculated in accordance with <u>Section 2</u> (Escalation Factor) of this <u>Exhibit</u> <u>4B</u> (Availability Payment Mechanism)
2	\$200, index-linked using the Escalation Factor calculated in accordance with <u>Section 2</u> (Escalation Factor) of this <u>Exhibit</u> <u>4B</u> (Availability Payment Mechanism)
1	\$90, index-linked using the Escalation Factor calculated in accordance with <u>Section 2</u> (Escalation Factor) of this <u>Exhibit</u> <u>4B</u> (Availability Payment Mechanism)

# Table B2: Space Types and Core Hours

Space Type	Core Hours
Maintenance and Transit Spaces ("Transit/ Maintenance")	4am – 10pm, Monday to Sunday
Office/ Admin and Training Spaces	8am – 5pm, Monday to Friday
("Training/ Administration")	

# Functional Units in Facility

The following tables presented in this <u>Appendix B</u> identify the Functional Components and Functional Units in the Infrastructure Facility, and Functional Component Rank Category. The Functional Components and Functional Units have been identified by floor level in the Infrastructure Facility and the functions carried out in that part of the Infrastructure Facility.

<b>Table B3: Functional Components and Functional</b>	Units for facilities on multiple levels
---	---

Level	Space	Space Type	Functional Unit	FU Referen ce	Functional Components	No. of Functional Components	Component Ref.	Rank
			Bus Parking (40' Buses)	AL-010	40' Bus Parking Bays	53	AL-010-1	4
			Bus Parking (60' Buses)	AL-011	60' Bus Parking Bays	160	AL-011-1	4
	Parking	Transit/ Maintenance	Parking for Standard Non-Revenue Vehicles (staff)	AL-012	Std N-Rev Parking Bays	151	AL-012-1	2
All Levels			Parking for Large Non-Revenue Vehicles	AL-013	Large N-Rev Parking Bays	12	AL-013-1	2
(Basement, Ground, Levels 2-4)	Bus Washing	Transit/ Maintenance	Bus Washer Function	AL-014	Bus Washer (Levels 3 & 4)	2	AL-014-1	4
					Stairs at grid A7-B8 (B1, Ground, Level 1)	1	AL-015-1	5
	Building	Transit/			Stair at grid A11-B12 (Ground and Level 1)	12	AL-015-2	5
	Spaces	Maintenance	Stairs	AL-015	Stairs at grid D12-E13 (Ground, L1-4) - common stairs	1	AL-015-3	5
					Stairs at grid B1-B2 (L1)	1	AL-015-4	5

Infrastructure Facility DBFOM Agreement

EXHIBIT 4B – Appendix B Page 2 of 18

		Stairs at grid H11-J12 (Ground, L1-4)	1	AL-015-5	5
		Stairs at grid G1-H2 (Ground, L1)	1	AL-015-6	5
Elevators at grid A7- B8 (Levels B1, Ground, Mezzanine)	AL-016	Elevators	3	AL-016-1	4
Other Elevators for		Elevator at grid C12-D13 (Ground, Mezzanine)	1	AL-017-1	4
BYC	AL-017	Elevator at grid H11 (Ground, Mezzanine, L2 and 3)	1	AL-017-2	4
Distribution Chases, Shafts, or Raceways, vertical or horizontal.	AL-018	Distribution Chases, Shafts, or Raceways, vertical or horizontal.	1	AL-018-1	3
Common- Use/Exterior Spaces	AL-019	All public spaces such as sidewalks, landscaping, etc.	1	AL-019-1	1

### Table B4: Functional Components and Functional Units for Basement Level

Level	Space	Space Type	Functional Unit	FU Referen ce	Functional Components	No. of Functional Components	Component Ref.	Rank
					Parking for Car Share	5	B1-010-1	2
	Parking	Transit/ Maintenance	Basement Parking Spaces	B1-010	FMO Parking	3	B1-010-2	2
ВҮС					Bicycle parking	10	B1-010-3	2
Basement	Service & Clean	Transit/ Maintenance	Water Reclamation	B1-011	Water Reclamation	2	B1-011-1	2
	FMO		Office spaces	B1-012	Site Manager	1	B1-012-1	2

Infrastructure Facility DBFOM Agreement

EXHIBIT 4B – Appendix B Page 3 of 18

				Admin Open Office Area	1	B1-012-2	2
				Janitor Closet 1	1	B1-013-1	2
		Janitor	B1-013	Janitor Closet 2	1	B1-013-2	2
				IT Room	1	B1-014-1	3
		IT & Storage	B1-014	Tech Shop	1	B1-014-2	2
	Transit/ Maintenance	IT & Storage	B1-014	Office Storage	1	B1-014-3	3
	Wantenance			Spare Parts Storage	1	B1-014-4	3
				Lunch and Break Room	1	B1-015-1	3
				Male Washroom and Showers	1	B1-015-2	3
		Shared Staff spaces	B1-015	Female Washroom and Showers	1	B1-015-3	3
				FM Locker Room	1	B1-015-4	3
				Unisex Washroom	1	B1-015-5	3
			B1-016	Main Point of Entry (MPOE)	1	B1-016-1	3
		Electrical-mechanical		BYC Fire Pump	1	B1-016-2	3
		Facilities		Electrical Room	1	B1-016-3	3
Shared	Transit/			Mechanical Room	1	B1-016-4	3
Facilities	Maintenance Wast Stora Transit/ HCC/	Waste Facilities	B1-017	Trash/Recycling/Compost Compactor	1	B1-017-1	3
				Hazardous Waste	1	B1-017-2	2
		Storage	B1-018	Diesel Storage Room	1	B1-018-1	1
HCC/BYC		HCC/BYC MEP	B1-019	HCC Thermal Storage Pump Room	1	B1-019-1	3
MEP	Maintenance	Facilities	51-019	AFF Electrical Room	1	B1-019-2	3

				AFF Emergency Electrical Room	1	B1-019-3	5
				WRK Electrical Room	1	B1-019-4	3
				WRK Emergency Electrical Room	1	B1-019-5	5
				DCW Booster Room	1	B1-019-6	3
				Grey Water Tank	1	B1-019-7	3
				Rain Water Tank	1	B1-019-8	3
				Processing Plant	1	B1-019-9	5
Unassigned	Training/ Administrati on	Unassigned Spaces	B1-020	Storage	3	B1-020-1	3

# Table B5: Functional Components and Units for Ground Level

Level	Space	Space Type	Functional Unit	FU Referen ce	Functional Components	No. of Functional Components	Component Ref.	Rank (1- 5)
					Running Repair Supervisor	3	LG-010-1	2
					Control Room Clerk	2	LG-010-2	2
DVC Laval		Tropoit	Office Space	GL-010	Floor Supervisor	2	LG-010-3	2
BYC Level Ground	Bays & Shops	Transit/ Maintenance			Preventative Maintenance Supervisor	2	LG-010-4	2
					Electronic Supervisor	1	LG-010-5	2
			Repair Bays	GL-011	60' Bus Repair Bays	10	LG-011-1	4

Infrastructure Facility DBFOM Agreement

EXHIBIT 4B – Appendix B Page 5 of 18

		Preventive Maintenance	GL-012	60' Preventative Maintenance Bays	5	LG-012-1	4
		Lube and Compressor	01 01 0	Lube Room	1	LG-013-1	4
		Rooms	GL-013	Compressor Room	1	LG-013-2	4
				60' Bus Tire Bay	1	LG-014-1	4
		Tire Bay, Shop and Storage	GL-014	Tire Shop	1	LG-014-2	4
		otorago		Tire Storage	1	LG-014-3	4
				60' Bus Minor Body Repair	1	LG-015-1	4
		Body Repair	GL-015	60' Bus Chassis Wash	1	LG-015-2	4
				Minor Body Shop	1	LG-015-3	4
				Electronic Shop Workstations	1	LG-016-1	4
		Electronics Shop	GL-016	Electronic Bench Shop	1	LG-016-2	4
				Data/Comm Room	1	LG-016-3	5
				Common Work Area	2	LG-017-1	3
		Common Work Area	GL-017	Portable Equipment Storage	2	LG-017-2	3
		& Tools Storage	GL-017	Tool Box Storage	3	LG-017-3	3
				Tool Storage	1	LG-017-4	3
Service & Clean	Transit/ Maintenance	Service & Clean	GL-018	Cleaning Equipment Storage	1	LG-018-1	1
				Parts Supervisor	1	LG-019-1	2
Ground Level	Transit/	Staff Spaces	GL-019	Parts Lockers	1	LG-019-2	3
Parts Spaces	Maintenance	Staff Spaces C	GL-019	Break Room	1	LG-019-3	3
				Gender Neutral Restroom	1	LG-019-4	3

EXHIBIT 4B – Appendix B Page 6 of 18

		Γ					
				Parts Storage	1	LG-020-1	3
		Parts Shop and	GL-020	Parts Shopkeeper	5	LG-020-2	2
		Storage	GL-020	Parts Window	1	LG-020-3	3
				Receiving Office	1	LG-020-4	3
				Superintendent	1	LG-021-1	2
				Assistant Superintendent	1	LG-021-2	2
		Staff Work Spaces	GL-021	Senior Controller	1	LG-021-3	2
				Administrative Assistant	2	LG-021-4	2
				Hoteling - Workstation	4	LG-021-5	1
		Shop, Records and		Support Shop	1	LG-022-1	2
			GL-022	Copy/Supply	1	LG-022-2	2
				Records Storage	1	LG-022-3	2
	Transit/	Resources		Archive Record Storage	1	LG-022-4	2
Maintenance	Maintenance			Library/Online Resources	1	LG-022-5	2
				Data/Comm Room	1	LG-022-6	5
				Kitchenette/Vending	1	LG-023-1	3
				Break Room	1	LG-023-2	3
				Training Room	1	LG-023-3	3
		Staff Common Spaces	GL-023	Uniform Alcove	1	LG-023-4	3
				Men's Restroom/Shower	1	LG-023-5	3
				Men's Locker	1	LG-023-6	3
				Women's Restroom/Shower	1	LG-023-7	3

EXHIBIT 4B – Appendix B Page 7 of 18

Functional Unit Rankings and Deduction Amounts for Availability Failures and Service Failures

	1		1	1		1	
				Women's Locker	1	LG-023-8	3
				Gender Neutral Accessible Locker/Shower/Restroom	1	LG-023-9	3
				Custodial	1	LG-023-10	3
Operations	Transit/ Maintenance	Yard Starter Office	GL-024	Yard Starter Office	1	LG-024-1	5
Shared	Transit/	Charad Offices		Revenue Office	1	LG-025-1	5
Offices	Maintenance	Shared Offices	GL-025	Meet and Greet	1	LG-025-2	5
Transit Services (MRO)	Transit/ Maintenance	Transit Operations/ Equipment Storage/ Component Rebuilding Assembly	GL-026	Transit Operations/ Equipment Storage/ Component Rebuilding Assembly	1	LG-026-1	3
Training	Training/ Administrati on	Training	GL-027	Data/Comm Room	1	LG-027-1	5
				Mechanical Room	1	LG-028-1	3
HCC/BYC MEP	Transit/ Maintenance	HCC/BYC MEP Facilities	GL-028	Main Electrical Room	1	LG-028-2	3
	Maintenance			Electrical Room for Chargers	1	LG-028-3	3
Unassigned	Training/ Administrati	Unassigned Spaces	GL-029	General Shop Processing Equipment	1	LG-029-1	2
5	on	<b>5</b> 1		Storage	2	LG-029-2	2

# Table B6: Functional Components and Functional Units for Level 2

Level	Space	Space Type	Functional Unit	FU Referen ce	Functional Components	No. of Functional Components	Component Ref.	Rank
-------	-------	------------	-----------------	---------------------	-----------------------	------------------------------------	-------------------	------

Page 8 of 18

Infrastructure Facility DBFOM Agreement	EXHIBIT 4B – App	endix B	Functional Areas, Functional Ur
Initiastructure Facility DBFOM Agreement	$E \wedge I I B I 4 B - A P P$		Functional Areas, Functional Ur

					<u>Cumprintendent</u>	1	12.010.1	2
					Superintendent	1	L2-010-1	2
					Assistant Superintendent	2	L2-010-2	2
					Operations Supervisor	1	L2-010-3	2
		Transit/ Maintenance	Staff Work Spaces	L2-010	Trainer	1	L2-010-4	2
					Administrative Assistant	2	L2-010-5	2
					Hoteling - Workstation	4	L2-010-6	1
				Union Office	1	L2-010-7	2	
					Receiver	1	L2-011-1	5
N	Transit/ Maintenance Dispat	Dispatch Function	L2-011	Dispatch	2	L2-011-2	5	
				Operator Check- in/Dispatch/Receiver	1	L2-011-3	5	
BYC Level 2	BYC Level 2 Operations		Records and Comms	L2-012	Copy/Supply	1	L2-012-1	2
					Records Storage	1	L2-012-2	2
					Data/Comm Room	1	L2-012-3	5
					Uniform Storage	1	L2-013-1	3
					Kitchenette/Vending	1	L2-013-2	3
		Transit/ Maintenance			Break Room	1	L2-013-3	3
			Staff Common	L2-013	Lockers	1	L2-013-4	3
			Spaces	LZ-U13	Lockers Changing Area	2	L2-013-5	3
					Recreation Area	1	L2-013-6	3
					TV Room	1	L2-013-7	3
					Quiet Room	1	L2-013-8	2

					Men's Restroom/Shower	1	L2-013-9	3
						•		
					Women's Restroom/Shower	1	L2-013-10	3
					Gender Neutral Accessible Locker/Shower/Restroom	1	L2-013-11	3
					Custodial	2	L2-013-12	2
					SFMTA Open Space	1	L2-013-13	2
					Operations Manager	2	L2-014-1	2
	Transit/ Maintenance Transit Services (MRO)				Transit Manager II	1	L2-014-2	2
			Staff Work Spaces	L2-014	Operations Specialist	1	L2-014-3	2
					MRO, Street Operations	10	L2-014-4	2
					Junior Management Assistant	4	L2-014-5	2
		Co	Conference, Training, Comms	L2-015	Conference Room	1	L2-015-1	4
					Training Room	1	L2-015-2	4
					Data/Comm Room	1	L2-015-3	5
					Break Room	1	L2-016-1	3
					Lockers	1	L2-016-2	3
		Training/ Administrati			Locker Changing Area	5	L2-016-3	3
		on	Staff Common	L2-016	Men's Restroom/Shower	1	L2-016-4	3
		Spaces	Spaces	L2-016	Women's Restroom/Shower	1	L2-016-5	3
					Gender Neutral Accessible Locker/Shower/Restroom	1	L2-016-6	3
					Custodial	1	L2-016-7	2

				I			
	Training/		L2-017	Lobby	2	L2-017-1	3
	Administrati	Lobby, Conference, Comms		Medium Conference Room	2	L2-017-2	3
	on	Commis		Large Conference/Small Training	2	L2-017-3	3
				Facilities Stationary Engineer	2	L2-018-1	2
	Transit/ Maintenance	Staff Work Spaces, Storage	L2-018	Transit Maintenance Engineer	2	L2-018-2	2
				Security Office	1	L2-018-3	2
		Storage, Comm, MEP	L2-019	Building Maintenance Storage	1	L2-019-1	2
				Data/Comm Room	1	L2-019-2	5
Shared Facilities	Transit/ Maintenance Common			Main Telecommunication Room	1	L2-019-3	3
				Mechanical Room Allowance	4	L2-019-4	2
				Chiller Room	1	L2-019-5	4
				Boiler Room	1	L2-019-6	4
				HVAC Room	1	L2-019-7	4
			+	Fitness	1	L2-020-1	3
		L2-020	Gender Neutral Accessible Restroom	1	L2-020-2	3	
		Rest/Leisure Spaces	12 020	Community Room	1	L2-020-3	3
				Lactation Room	1	L2-020-4	3
				Reception	1	L2-021-1	2
Training	Training/		10.001	Manager	1	L2-021-2	2
Training	Administrati on	Staff Work Spaces	L2-021	Superintendent	1	L2-021-3	2
				Assistant Superintendent	4	L2-021-4	2

EXHIBIT 4B – Appendix B Page 11 of 18

				Supervisors	2	L2-021-5	2
				Clerical Staff	3	L2-021-6	2
				Team Leader	6	L2-021-0	2
					2	L2-021-7 L2-021-8	2
				CAT Training			
				Instructors	15	L2-021-9	2
				IT Office	1	L2-022-1	3
				Classrooms (A-B)	2	L2-022-2	3
				Classrooms (C-D)	2	L2-022-3	3
		Training Conserve	L2-022	Conference Rooms (A&B)	2	L2-022-4	3
		Training Spaces		Simulator Rooms	3	L2-022-5	3
				Computer Lab	1	L2-022-6	3
				Handouts Storage	1	L2-022-7	3
				Training Aid Storage	1	L2-022-8	3
			L2-023	Records Storage	1	L2-023-1	2
				Records Archive Storage	1	L2-023-2	2
		Storage		Copy/Supply	1	L2-023-3	2
				Uniform Storage	1	L2-023-4	2
				Kitchenette/Vending	1	L2-024-1	2
				Break Room	1	L2-024-2	3
	Staff Common	Staff Common Spaces	L2-024	Operator Locker	1	L2-024-3	3
		-1		Instructor Locker	1	L2-024-4	3
				Lactation Room	1	L2-024-5	3

				Men's Restroom/Shower	1	L2-024-6	3
				Women's Restroom/Shower	1	L2-024-7	3
				Gender Neutral Accessible Locker/Shower/Restroom	1	L2-024-8	3
				Custodial	2	L2-024-9	2
HCC/BYC MEP	Transit/ Maintenance	Electrical Room for Chargers	L2-025	Electrical Room for Chargers	2	L2-025-1	3
Unassigned	Training/ Administrati on	Storage	L2-026	Storage	1	L2-025-2	3

# Table B7: Functional Components and Functional Units for Levels 3 and 4

Level	Space	Space Type	Functional Unit	FU Referen ce	Functional Components	No. of Functional Components	Component Ref.	Rank
	Service &	Transit/			Service Supervisor Office	1	L3-010-1	3
	Clean	Maintenance	Service & Clean	Service & Clean L3-010	Service Position	2	L3-010-2	3
	Function Cleaning Equipment St	Cleaning Equipment Storage	3	L3-010-3	3			
BYC Level 3	Maintenance	Transit/ Maintenance	Maintenance L3-011 C	CTRL RM	1	L3-011-1	3	
BIC Level 3	Shared Admin	Training/ Administrati on	Restrooms	L3-012	Gender Neutral Accessible Restroom	2	L3-012-1	1
		Training/	Storage, Janitor L3-013	Wash Equipment Room	1	L3-013-1	1	
	Unassigned	d Administrati on		L3-013	Janitor Closet	1	L3-013-2	1
	Service &	Maintenance	Service & Clean L4-		Service Supervisor Office	1	L4-010-1	3
BYC Level 4	Clean			L4-010	Service Position	1	L4-010-2	3
	Function				Cleaning Equipment Storage	3	L4-010-3	3

Infrastructure Facility DBFOM Agreement

EXHIBIT 4B – Appendix B Page 13 of 18

Shared	Training/ Administrati on	Restrooms	L4-011	Gender Neutral Accessible Restroom	2	L4-011-1	1
	Training/	Storage, Janitor L4-012	Wash Equipment Room	1	L4-012-1	1	
Unassigned	Administrati on		Janitor Closet	1	L4-012-2	1	

EXHIBIT 4B – Appendix B Page 14 of 18

# APPENDIX C

# IFM NONCOMPLIANCE POINTS

IFM Noncompliance Points will be assessed based on the following categorization for IFM Failure Events in accordance with this <u>Exhibit 4</u> and the Performance Measurement Table.

Table C1: IFM Noncompliance Points Table for Service Failures
---

Service Failure Level	Noncompliance Points	Recurrence Period
MAJOR	10	As per 'FREQ' column in Performance Measurement Table in accordance with relevant Performance Measure
MED	5	As per 'FREQ' column in Performance Measurement Table in accordance with relevant Performance Measure
MINOR	1	As per 'FREQ' column in Performance Measurement Table in accordance with relevant Performance Measure

# Table C2: IFM Noncompliance Points Table for Availability Failures

Functional Component Rank Category	Noncompliance Points	Recurrence Period
5	10	4 hours
4	10	8 hours
3	5	12 hours
2	5	24 hours
1	1	48 hours

# APPENDIX D

# PAYMENT SCHEDULE

[To be inserted based on Financial Proosal]

# EXHIBIT 5

# **FINANCE DOCUMENTS**

- Exhibit 5A: List of Initial Financing Documents
- Exhibit 5B: Form of Direct Agreement
- Exhibit 5C: Calculation of Refinancing Gain
- Exhibit 5D: Form of Opinion from City's Legal Counsel
- Exhibit 5E: Form of Opinion from Principal Project Company's Legal Counsel
- Exhibit 5F: Base Capital MaxAP Adjustment for Base Interest Rate Fluctuation and Credit Spread Risk Mitigation
- Exhibit 5G: Financing Document Terms

# EXHIBIT 5A

# LIST OF INITIAL FINANCING DOCUMENTS

# PART A: INITIAL FINANCING AGREEMENTS

# PART B: INITIAL SECURITY DOCUMENTS

# EXHIBIT 5B

## FORM OF DIRECT AGREEMENT

THIS DIRECT AGREEMENT dated as of [\_\_\_\_], 2025 ("Direct Agreement") among the City and County of San Francisco ("City"), a municipal corporation, acting by and through the San Francisco Municipal Transportation Agency ("SFMTA"), [\_\_\_\_\_] ("Principal Project Company or PPC"), a [\_\_\_\_\_] and [\_\_\_\_], as trustee or collateral agent (in such capacity, together with its successors in such capacity, the "Collateral Agent") for the Lenders (as defined in the IF DBFOM Agreement).

### WHEREAS

(A) On November 2, 2022, City and Potrero Neighborhood Collective LLC, a limited liability company organized under the laws of the State of Delaware, entered into a Predevelopment Agreement for the Potrero Yard Modernization Project (the "**Predevelopment Agreement**"), which is comprised of a transit operations component ("**Bus Yard Component**"), a mixed-income housing and commercial component ("**Housing and Commercial Component**"), and the common infrastructure shared by the two components ("**Common Infrastructure**"),

(B) Following the processes described in the Predevelopment Agreement, City and Principal Project Company entered into an Infrastructure Facility Design-Build-Finance-Operate-Maintain Agreement dated as of [\_\_\_\_\_], 2025 (the "IF DBFOM Agreement") for the Bus Yard Component and Common Infrastructure (collectively, the "Infrastructure Facility"), and the integration of the Infrastructure Facility, and its interface with, the Housing and Commercial Component (together, the "Project"). The IF DBFOM Agreement contemplates Principal Project Company obtaining financing or Refinancing for the Project from third parties. The Housing and Commercial Commercial Component will be delivered by [\_\_\_\_\_], under separate agreements,

(C) In order to enable Principal Project Company to finance certain activities and certain obligations with respect to the Project, the Lenders have agreed to make available debt facilities, on the terms set out in the Financing Agreements and Security Documents, for the purpose of financing the Project, subject to provision of certain assurances from City regarding Lender's and Collateral Agent's rights in the event of an Event of Default or PPC Default,

(D) In reliance on such assurances, and on this Direct Agreement, Lenders have agreed to make available such financing or Refinancing facilities for the purpose of financing or Refinancing all or part of the Project, and

(E) The execution of this Direct Agreement by City in favor of the Collateral Agent is a condition precedent to such financing or Refinancing facilities being made available to Principal Project Company by Lender.

**NOW, THEREFORE**, in consideration of the foregoing and the mutual terms and covenants contained herein, and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged by the parties, each of City, Principal Project Company and Collateral Agent hereby agree as follows:

## 1. DEFINITIONS AND INTERPRETATION

## 1.1 Definitions

Capitalized terms used but not otherwise defined in this Direct Agreement and references used but not construed in this Direct Agreement have the respective meanings and constructions assigned to such terms in the IF DBFOM Agreement. In addition, the following terms have the meanings specified below:

<u>**City**</u> has the meaning given to it in the Preamble.

<u>City Notice</u> has the meaning given to it in <u>Section 5.1</u>.

**<u>Collateral Agent</u>** has the meaning given to it in the Preamble.

<u>Control Agreement</u> means the Control Agreement, dated as of the date hereof, by and among Principal Project Company, City and Custodian, with respect to the Handback Requirements Reserve Account.

<u>Cure Period</u> means the period starting on the date of the receipt of the City Notice and ending on the earlier of:

- (a) the Step-in Date; and
- (b) 90 days after the expiration of any cure periods provided to Principal Project Company under the IF DBFOM Agreement, provided that:
  - (i) if the Collateral Agent provides a Notice of Intent to City within 30 days of the date of the receipt of the City Notice indicating that the Lenders' intend to proceed to cure the PPC Default(s) or to exercise step-in rights, then the Cure Period will end on the earlier of the Step-in Date and 120 days after the expiration of any cure periods provided to Principal Project Company under the IF DBFOM Agreement; and
  - (ii) if the Collateral Agent is prohibited from curing any non-monetary default or from substituting Principal Project Company with the Substituted Entity by any process, stay or injunction issued by any Governmental Entity or pursuant to any bankruptcy or insolvency proceeding or other similar proceeding involving Principal Project Company, then the time period specified herein for curing a default shall be extended for the period of such prohibition.

**<u>Custodian</u>** means [\_\_\_\_\_], as custodian under the Control Agreement.

**Default** means an Event of Default as defined in any Financing Agreement or any event or circumstance specified in any Financing Agreement which would (with the expiration of a grace period, the giving of notice, the lapse of time, the making of any determination under the Financing Documents or any combination of any of the foregoing) be an Event of Default.

**Direct Agreement** has the meaning given to it in the Preamble.

**Discharge Date** means the date on which all of the obligations of Principal Project Company under the Financing Documents have been irrevocably discharged in full to the satisfaction of the Collateral Agent.

**Event of Default** means an Event of Default as defined in any Financing Agreement.

**IF DBFOM Agreement** has the meaning given to it in the Recitals.

**Lender Notice** has the meaning given to it in <u>Section 7.1</u>.

**Notice of Intent** has the meaning given to it in <u>Section 5.4</u>.

**Predevelopment Agreement** has the meaning given to it in the Recitals.

**Principal Project Company** or **PPC** has the meaning given to it in the Preamble.

**Project** has the meaning given to it in the Recitals.

**<u>Property</u>** means any right or interest in or to property of any kind whatsoever, whether real, personal or mixed and whether tangible or intangible.

**<u>Revival Date</u>** has the meaning given to it in <u>Section 14.1</u>.

**<u>Step-in Date</u>** has the meaning given to it in <u>Section 10.1</u>.

**<u>Step-in Notice</u>** has the meaning given to it in <u>Section 9.1</u>.

**<u>Step-in Party</u>** has the meaning given to it in <u>Section 9.2</u>.

**<u>Step-in Period</u>** means the period from and including the Step-in Date until the earliest of:

- (a) the Substitution Effective Date;
- (b) the Step-out Date;
- (c) the date of termination of the IF DBFOM Agreement by City in accordance with this Direct Agreement and the IF DBFOM Agreement;
- (d) the date of the expiration or early termination of the Term under the IF DBFOM Agreement;
- (e) 12 calendar months after the Step-in Date; and
- (f) the Long Stop Date,

<u>provided</u>, <u>however</u>, that if the Collateral Agent is prohibited from curing any nonmonetary default after the Step-in Date or from substituting Principal Project Company with the Substituted Entity by any process, stay or injunction issued by any Governmental Entity or pursuant to any bankruptcy or insolvency proceeding or other similar proceeding involving Principal Project Company, then the time periods specified herein for curing a default shall be extended for the period of such prohibition. <u>Step-out Date</u> means the date upon which the notice period set forth in any Step-out Notice expires.

**<u>Step-out Notice</u>** has the meaning given to it in <u>Section 11</u> (Step-out).

<u>Substitute Accession Agreement</u> means the agreement to be entered into by a Substituted Entity pursuant to <u>Section 13.1</u>.

<u>Substituted Entity</u> means any Person selected by Lenders and approved by City in accordance with <u>Section 12</u> (Substitution Entities and Substitution Proposals) to perform all or a portion of Principal Project Company's obligations and succeed to all or a portion of Principal Project Company's rights under the Contract Documents.

**Substitution Effective Date** has the meaning given to it in <u>Section 13.1</u>.

**Substitution Notice** has the meaning given to it in <u>Section 12.2</u>.

#### 1.2 Interpretation

Unless the context otherwise clearly requires:

- (a) The definitions of terms herein shall apply equally to the singular and plural forms of the terms defined;
- (b) Whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms;
- (c) The words "include", "includes" and "including" shall be deemed to be followed by the phrase "without limitation";
- (d) The word "will" shall be construed to have the same meaning and effect as the word "shall";
- (e) Any definition of or reference to any agreement, instrument or other document herein shall be construed as referring to such agreement, instrument or other document as from time to time amended, supplemented or otherwise modified (subject to any restrictions on such amendments, supplements or modifications set forth herein);
- (f) Any reference herein to any Person, or to any Person in a specified capacity, shall be construed to include such Person's successors and assigns or such Person's successors in such capacity, as the case may be;
- (g) The words "herein", "hereof" and "hereunder", and words of similar import, shall be construed to refer to this Direct Agreement in its entirety and not to any particular provision hereof;
- (h) All references herein to Sections and Schedules shall be construed to refer to Sections of and Schedules to this Direct Agreement. Any Schedules to this Direct Agreement are an integral part hereof. The provisions of this Direct Agreement

shall prevail over the provisions of any Schedules to the extent of any inconsistency;

- (i) The headings used in this Direct Agreement are for convenience of reference only and are not to affect the construction of or to be taken into consideration in interpreting this Direct Agreement; and
- (j) "Winding-up", "liquidation", "dissolution", "insolvency", "adjustment" or "reorganization" of a Person and references to the "liquidator", "assignee", "administrator", "receiver", "custodian", "conservator", "sequestrator" or "trustee" of a Person shall be construed so as to include any equivalent or analogous proceedings or, as the case may be, insolvency representatives or officers under the law of the jurisdiction in which such Person is incorporated, organized or constituted or any jurisdiction in which such Person or, as the case may be, insolvency representative or officer carries on business including the seeking of winding up, liquidation, dissolution, reorganization, administration, arrangement, adjustment or relief of debtors.

# 2. **REPRESENTATIONS AND WARRANTIES**

- **2.1** City represents and warrants that:
  - (a) Organization; Power and Authority. City is a charter city and municipal corporation duly organized and validly existing under the Constitution of the State of California. City has full power, right and authority to execute and deliver this Direct Agreement and the IF DBFOM Agreement; and City has full power, right and authority to perform its obligations under the provisions hereof and thereof.
  - (b) Authorizations, Enforceability. This Direct Agreement and the IF DBFOM Agreement have been duly authorized by City, and this Direct Agreement and the IF DBFOM Agreement constitute legal, valid and binding obligations of City, enforceable against City in accordance with their terms, except as such enforceability may be limited by (i) applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors' rights generally and (ii) general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).
  - (c) No Default. As of the date of the execution of this Direct Agreement, there is no City Default. City is not aware of any PPC Default, and there exists no event or condition of which City has Actual Knowledge that would, with the giving of notice or passage of time or both, constitute such a PPC Default or City Default.
- **2.2** Principal Project Company represents and warrants that:
  - (a) Organization; Power and Authority. Principal Project Company is a [\_\_\_\_\_], duly organized and validly existing under the laws of [\_\_\_\_\_] and has the requisite power and all required licenses to carry on its present and proposed activities, and has full power, right and authority to execute and deliver this Direct Agreement and the IF DBFOM Agreement; and

Principal Project Company has the power and authority to perform the provisions hereof and thereof.

- (b) Authorizations, Enforceability. This Direct Agreement and the IF DBFOM Agreement have been duly authorized by Principal Project Company, and this Direct Agreement and the IF DBFOM Agreement constitute legal, valid, and binding obligations of Principal Project Company, enforceable against Principal Project Company in accordance with their terms, except as such enforceability may be limited by (i) applicable bankruptcy, insolvency, reorganization, moratorium, or other similar laws affecting the enforcement of creditors' rights generally; and (ii) general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).
- **2.3** The Collateral Agent represents and warrants that:
  - (a) **Organization; Power and Authority**. The Collateral Agent is a [\_\_\_\_\_], duly organized and validly existing under the laws of [\_\_\_\_\_] and has the requisite power and all required licenses to carry on its present and proposed activities, and has full power, right and authority to execute and deliver this Direct Agreement and to perform each and all of the obligations of the Collateral Agent provided for under this Direct Agreement.
  - (b) Authorizations, Enforceability. This Direct Agreement has been duly authorized by the Collateral Agent, and this Direct Agreement constitutes a legal, valid and binding obligation of the Collateral Agent, enforceable against the Collateral Agent in accordance with its terms, except as such enforceability may be limited by (i) applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors' rights generally and (ii) general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law). The Collateral Agent is duly authorized by the Lenders to enter into this Direct Agreement on behalf of the Lenders and to validly bind the Lenders to the terms and conditions hereof.

# 3. CONDITIONS AND LIMITATIONS RESPECTING LENDERS' RIGHTS

**3.1** No Lender shall be entitled to the rights, benefits and protections of this Direct Agreement unless and until City has received a copy (certified as true and correct by the Collateral Agent) of the Financing Agreements and Security Documents bearing, if applicable, the date and instrument number or book and page of recordation or filing thereof, including a copy of a specimen bond, note or other obligation (certified as true and correct by the Collateral Agent) secured by such Security Document, together with written notice of the address of the Collateral Agent to which notices may be sent. In the event of an assignment of any such Financing Agreement (excluding any instrument described under subsection (b) of the definition of Financing Agreement) or Security Document, such assignment shall not be subject to the terms of this Direct Agreement unless and until City has received prior written notice and a certified copy thereof, which copy shall, if required to be recorded, bear the date and instrument number or book and page of recordation thereof, together with written notice of the assignee thereof to which notices may be sent. In the event of any change in the identity of the Collateral Agent, such change shall not be binding upon City unless and until City has received a written notice thereof signed by the replaced and substitute Collateral Agent and setting forth the address of the substitute Collateral Agent to which notices may be sent.

**3.2** Neither the Collateral Agent nor the Lenders shall exercise any right it may have pursuant to any Financing Documents to assign, transfer or otherwise dispose of any right, title or interest it may have in, or obligations it may have pursuant to, the Financing Documents to the extent the exercise of such rights would constitute a Refinancing and Principal Project Company has failed to comply with the requirements of <u>Section 4.4</u> (Refinancing) of the IF DBFOM Agreement.

**3.3** No Security Document or other instrument purporting to mortgage, pledge, encumber, or create a lien, charge or security interest on or against PPC's Interest shall extend to or affect the fee simple interest of City in the Project or the Site or improvements thereto or City's rights or interests under the Contract Documents.

**3.4** City shall not have any obligation to any Lender pursuant to the IF DBFOM Agreement, except for the express obligations to Lenders set forth in this Direct Agreement or any other instrument or agreement signed by City in favor of such Lender or Collateral Agent, provided that the Collateral Agent has notified City of the existence of its Security Documents.

**3.5** Each Financing Agreement and Security Document shall require that the Collateral Agent deliver to City, concurrently with delivery to Principal Project Company or any other Person, any notice of default or notice of election or enforcement of remedies, including an election to sell or foreclose, notice of sale or foreclosure or other notice required by Law or by the Security Document in connection with the exercise of remedies under the Financing Agreement or Security Document.

**3.6** No Financing Documents shall grant to the Lender any right to apply funds in the Handback Requirements Reserve Account or to apply proceeds from any Handback Requirements Letter of Credit to the repayment of Project Debt, to any other obligation owing the Lender or to any other use except the uses set forth in <u>Section 8.7.3</u> (Use of Handback Requirements Reserve Account) of the IF DBFOM Agreement, and any provision purporting to grant such right shall be null and void; provided, however, that the foregoing shall not preclude any Lender or Substituted Entity from, following foreclosure or transfer in lieu of foreclosure, automatically succeeding to all rights, claims and interests of Principal Project Company in and to the Handback Requirements Reserve Account.

# 4. AGREEMENTS, CONSENT TO SECURITY AND SUBORDINATION OF SECURITY

**4.1** City acknowledges notice and receipt of copies of the Initial Financing Documents and Initial Security Documents. Notwithstanding anything in the IF DBFOM Agreement to the contrary, but subject to <u>Section 3</u> (Conditions and Limitations Respecting Lenders' Rights), City:

(a) consents to (i) the assignment by Principal Project Company to the Collateral Agent of all of PPC's Interest, and (ii) the granting by each Equity Member to the Collateral Agent of a security interest in such Equity Member's equity interest in Principal Project Company, in each case pursuant to the terms and provisions of the applicable Initial Security Documents; and (b) agrees that such collateral assignment and grant of security interests in and first lien over all of the PPC's Interest pursuant to the Initial Security Documents and the grant of the security interest by each Equity Member in its equity interests in Principal Project Company pursuant to such Security Documents, and the execution by Principal Project Company and City of this Direct Agreement and the performance of their respective obligations hereunder, in each case, does not (i) constitute a PPC Default or any other breach by Principal Project Company of the Contract Documents, (ii) with the giving of notice or lapse of time, or both, constitute a PPC Default or any other breach by Principal Project Company of the IF DBFOM Agreement, or (iii) require the consent of City except as provided herein.

**4.2** Except as expressly contemplated in the IF DBFOM Agreement as of the Effective Date thereof, while any Security Document is in effect, no agreement between City and Principal Project Company for the modification or amendment of the IF DBFOM Agreement that in any way could reasonably be expected to have a material adverse effect on the rights or interests of the Lender(s) shall be binding on the Lender(s) under such Security Document without the Collateral Agent's consent.

**4.3** As long as any Project Debt secured by any Security Document shall remain outstanding, City shall promptly provide the Collateral Agent with a copy of any notice it sends to Principal Project Company concerning an actual or potential PPC Default.

**4.4** Except as set forth in this Direct Agreement, City shall not be precluded from or delayed in exercising any remedies, including termination of the IF DBFOM Agreement due to the accumulation of Noncompliance Points during the Step-in Period and City's rights to cure PPC Default at Principal Project Company's expense; provided, however, City shall not be entitled to exercise its right of termination due to Noncompliance Points accumulated prior to such step in.

4.5 Neither City nor any officer, employee, agent or representative of City shall have any liability whatsoever for payment of the principal sum of any Project Debt, any other obligations issued or incurred by Principal Project Company or a PPC-Related Entity in connection with the IF DBFOM Agreement or the Project, or any interest accrued thereon or any other sum secured by or accruing under any Financing Document. Except for a violation by City of its express obligations to Lenders under this Direct Agreement, no Lender is entitled to seek any damages or other amounts from City, whether for Project Debt or any other amount. City's review of any Financing Documents or other Project financing documents is not a guarantee or endorsement of the Project Debt, any other obligations issued or incurred by Principal Project Company or a PPC-Related Entity in connection with the IF DBFOM Agreement or the Project, and is not a representation, warranty or other assurance as to the ability of Principal Project Company or a PPC-Related Entity to perform its obligations with respect to the Project Debt or any other obligations issued or incurred by Principal Project Company or a PPC-Related Entity in connection with the IF DBFOM Agreement or the Project, or as to the adequacy of the Payments to provide for payment of the Project Debt or any other obligations issued or incurred by Principal Project Company in connection with the IF DBFOM Agreement or the Project. The foregoing does not affect City's liability to Principal Project Company under Article 17 (Termination) of the IF DBFOM Agreement for Termination Compensation that is measured in whole or in part by outstanding Project Debt.

**4.6** The Collateral Agent consents to the grant of security by Principal Project Company to City of a first priority security interest in the Handback Requirements Reserve Account (or in any Handback Requirements Letter of Credit delivered by Principal Project Company in accordance with <u>Section 8.7.5</u> (Handback Requirements Letters of Credit) of the IF DBFOM Agreement in lieu of establishing a Handback Requirements Reserve Account).

# 5. CITY NOTICE OF TERMINATION AND EXERCISE OF REMEDIES; LENDER NOTICE OF INTENT

- 5.1 City shall give the Collateral Agent written notice (a "City Notice"):
  - (a) promptly upon obtaining Actual Knowledge of the occurrence of any event giving rise to a PPC Default, City's right to terminate or give notice terminating the IF DBFOM Agreement pursuant to <u>Section 17.3.1</u> (PPC Defaults Triggering City Termination Rights) of the IF DBFOM Agreement, or exercise any rights under <u>Sections 16.2.4</u> (Remedies for Failure to Meet Safety Standards or Perform Safety Compliance), <u>16.2.5</u> (City Step-in Rights) or <u>16.2.7.1</u> of the IF DBFOM Agreement; or
  - (b) promptly upon obtaining Actual Knowledge of the occurrence of any event giving rise to City's right to suspend its performance (including in connection with any insolvency or bankruptcy proceeding in relation to Principal Project Company) under the IF DBFOM Agreement.
- **5.2** A City Notice shall specify:
  - (a) the unperformed obligations of Principal Project Company under the IF DBFOM Agreement and grounds for termination of, or suspension of performance or the other rights all as referred to in <u>Sections 17.3.1</u> (PPC Defaults Triggering City Termination Rights), <u>16.2.4</u> (Remedies for Failure to Meet Safety Standards or Perform Safety Compliance), <u>16.2.5</u> (City Step-in Rights) or <u>16.2.7.1</u> under the IF DBFOM Agreement, in detail sufficient to enable the Collateral Agent to assess the scope and amount of any liability of Principal Project Company resulting therefrom;
  - (b) any other unperformed obligations of Principal Project Company of which City obtains Actual Knowledge as of the date of such City Notice;
  - (c) all amounts due and payable by Principal Project Company to City under the IF DBFOM Agreement on or before the date of such City Notice and which remain unpaid at such date and the nature of Principal Project Company's obligation to pay such amounts; and
  - (d) the amount of Principal Project Company's payment obligation to City that City reasonably foresees will arise during the applicable Cure Period.

**5.3** City shall update its City Notice to reflect unperformed obligations of Principal Project Company under the IF DBFOM Agreement that have been identified or that have arisen and amounts payable by Principal Project Company to City that become due, in each case, after the date of the City Notice but prior to the proposed Step-in Date.

**5.4** Within 30 days of the date of receipt of the City Notice, the Collateral Agent shall provide City with notice, via written correspondence and email, of the intent of the Lenders' and shall indicate in such notice whether the Lenders' intend to proceed to cure the PPC Default(s) or to exercise step-in rights (a "**Notice of Intent**"). The hardcopy Notice of Intent shall be sent to City via certified or registered mail, return receipt requested.

# 6. PROCEEDS ACCOUNT

Subject to the Collateral Agent directing otherwise in accordance with this Direct Agreement, Principal Project Company irrevocably directs City to remit all amounts due and owing to Principal Project Company under the IF DBFOM Agreement directly to the Proceeds Account.

# 7. LENDER NOTICE

**7.1** The Collateral Agent shall give City notice (a "**Lender Notice**") via written correspondence and email, with a hardcopy and electronic copy to Principal Project Company, promptly upon becoming aware of the occurrence of any Default or Event of Default (whether or not a City Notice has been served relating to the same event). The hardcopy Lender Notice shall be sent to City via certified or registered mail, return receipt requested.

**7.2** The Collateral Agent shall specify in any Lender Notice the circumstances and nature of the Default or Event of Default to which Lender Notice relates.

**7.3** Unless directed otherwise by the Collateral Agent pursuant to <u>Section 7.4</u>, City shall, following receipt of a Lender Notice of the occurrence of any Event of Default and until further notice from the Collateral Agent pursuant to <u>Section 7.8</u>, continue to make any payments required to be made by City to Principal Project Company under the IF DBFOM Agreement, including any payment of any termination sum calculated in accordance with <u>Article 17</u> (Termination) of the IF DBFOM Agreement, to the Proceeds Account.

**7.4** The Collateral Agent may direct City to make the payments described in <u>Section 7.3</u> to an alternate account designated by the Collateral Agent by providing written notice to City specifying the following information with respect to the account: (a) account details and wiring instructions; and (b) the name, title, mailing address, telephone number, fax number, and email address of the individual responsible for administering the account. City shall have no liability, whatsoever, for any delay in processing any payment request pursuant to this <u>Section 7.4</u>, provided that such delay does not extend 20 days beyond the date of City's certified, return-receipt or registered mail receipt of the Lender Notice.

**7.5** All sums paid as provided in <u>Sections 7.3</u> and <u>7.4</u> shall be deemed paid to Principal Project Company under the IF DBFOM Agreement and shall constitute a complete discharge of City's relevant payment obligations to Principal Project Company.

**7.6** The Collateral Agent shall promptly notify City via written correspondence and email of any decision to accelerate amounts outstanding under the Financing Documents or to exercise any enforcement remedies under the Financing Documents.

**7.7** Neither the Collateral Agent nor the Lender shall exercise any right it may have pursuant to the Security Documents to assign, transfer or otherwise dispose of any right, title or interest it may have in, or obligations it may have pursuant to, the Security Documents to the extent the

exercise of such rights would constitute a Refinancing and Principal Project Company has failed to comply with the requirements of <u>Section 4.4</u> (Refinancing) of the IF DBFOM Agreement.

**7.8** The Collateral Agent shall promptly notify City via written correspondence and email, with a hardcopy and electronic copy to Principal Project Company, of a full cure of an Event of Default that is the subject of a Lender Notice.

**7.9** Following receipt of a Lender Notice of the occurrence of an Event of Default until delivery of a further notice under <u>Section 7.8</u> with respect to any full cure of such Event of Default, the Collateral Agent shall have the right to deliver to City a Step-in Notice as provided in <u>Section 9</u> (Step-in Notice).

# 8. NO TERMINATION DURING CURE PERIOD

**8.1** City agrees not to take any of the following actions prior to the expiration of any applicable Cure Period:

- terminate or give notice terminating the IF DBFOM Agreement or exercise any rights under <u>Sections 10.4</u> (Letters of Credit), <u>10.5</u> (Guarantees), <u>16.2.4</u> (Remedies for Failure to Meet Safety Standards or Perform Safety Compliance), <u>16.2.5</u> (City Step-in Rights), <u>16.2.7</u> (Performance Bond), <u>16.2.8</u> (Suspension of Work), <u>16.2.9</u> (Other Rights and Remedies) or <u>17.3.1</u> (PPC Defaults Triggering City Termination Rights) of the IF DBFOM Agreement or in respect of the IFM Contract or D&C Contract;
- (b) suspend its performance (including with respect to (i) payments required to be made to Principal Project Company to the Proceeds Account or to an alternate account designated by the Collateral Agent in accordance with <u>Section 7.4</u>, or (ii) any insolvency or bankruptcy proceeding in relation to Principal Project Company) under the IF DBFOM Agreement; and
- (c) take or support any action for the liquidation, bankruptcy, administration, receivership, reorganization, dissolution or winding up of Principal Project Company or for the composition or readjustment of Principal Project Company's debts, or any similar insolvency procedure in relation to Principal Project Company, or for the appointment of a receiver, trustee, custodian, sequestrator, conservator, liquidator, administrator or similar official for Principal Project Company or for any part of Principal Project Company's property;

provided that such agreement of City shall not prevent City from taking actions which are permitted under this Direct Agreement on a Revival Date in respect of any other prior PPC Default or other breach by Principal Project Company of the IF DBFOM Agreement which has occurred and has not been remedied or waived.

**8.2** During any Cure Period, without giving a Step-in Notice, the Collateral Agent shall have the right (but shall have no obligation), at its sole option and discretion, to perform or arrange for the performance of any act, duty, or obligation required of Principal Project Company under the IF DBFOM Agreement, or to cure any default of Principal Project Company thereunder, which performance by the Collateral Agent shall be accepted by City in lieu of performance by Principal Project Company and in satisfaction of Principal Project Company's obligations under

the IF DBFOM Agreement. To the extent that any default of Principal Project Company under the IF DBFOM Agreement is cured and/or any payment liabilities or performance obligations of Principal Project Company are performed by the Collateral Agent during the Cure Period, such action shall discharge the relevant liabilities or obligations of Principal Project Company to City. Subject to the terms of this Direct Agreement, the Collateral Agent's right to cure any default of Principal Project Company as provided in this <u>Section 8.2</u> may be exercised after the expiration of relevant cure period granted to Principal Project Company in <u>Section 16.1.2</u> (Default Notice and Cure Periods) of the IF DBFOM Agreement, provided that the cure occurs within the Cure Period. Curing of any PPC Default by the Collateral Agent shall not be construed as an assumption by the Collateral Agent of any obligations, covenants or agreements of Principal Project Company under the Contract Documents, except to the extent the Collateral Agent has exercised step-in rights or proposed a Substitute Entity and otherwise as set forth in this Direct Agreement.

# 9. STEP-IN NOTICE

**9.1** Upon the issuance of a City Notice or a Lender Notice of the occurrence of any Event of Default, the Collateral Agent may give a written notice (a "**Step-in Notice**") under this <u>Section 9</u> (Step-in Notice) to City at any time during the Cure Period in the case of the issuance of a City Notice or at any time following the receipt by City of a Lender Notice, provided that the Event of Default to which Lender Notice relates is continuing.

**9.2** The Collateral Agent shall nominate, in the Step-in Notice: (a) the Collateral Agent, a Lender or any of their respective Affiliates; or (b) any Person, subject to approval by City in accordance with <u>Section 12</u> (Substitution Entities and Substitution Proposals), and the person so nominated being referred to as the "**Step-in Party**."

# 10. RIGHTS AND OBLIGATIONS ON STEP-IN

**10.1** On and from the date of the receipt of the Step-in Notice and the approval of City to the appointment of the Step-in Party if required by <u>Section 9.2</u> (the "**Step-in Date**") and during the Step-in Period, the Step-in Party shall be:

- entitled to exercise and enjoy the rights and powers expressed to be assumed by or granted to Principal Project Company under the IF DBFOM Agreement and this Direct Agreement;
- (b) entitled to exercise and enjoy the rights and powers expressed to be assumed by or granted to a Step-in Party under this Direct Agreement; and
- (c) liable for the performance of all of Principal Project Company's obligations under the IF DBFOM Agreement and this Direct Agreement arising on or after the Step-in Date.

**10.2** Without prejudice to <u>Section 14</u> (Revival of Remedies), during the Step-in Period, City shall:

 (a) not terminate or give notice terminating the IF DBFOM Agreement or exercise any rights under <u>Sections 10.4</u> (Letters of Credit), <u>10.5</u> (Guarantees), <u>16.2.4</u> (Remedies for Failure to Meet Safety Standards or Perform Safety Compliance), <u>16.2.5</u> (City Step-in Rights), <u>16.2.7</u> (Performance Bond), <u>16.2.8</u> (Suspension of Work), <u>16.2.9</u> (Other Rights and Remedies), or <u>17.3.1</u> (PPC Defaults Triggering City Termination Rights) of the IF DBFOM Agreement or in respect of the IFM Contract or D&C Contract, unless the grounds for termination or giving notice of termination pursuant to <u>Section 17.3.1</u> (PPC Defaults Triggering City Termination Rights) of the IF DBFOM Agreement or exercising its rights under the above listed sections of the IF DBFOM Agreement are failure by the Step-in Party to perform Principal Project Company's obligations under the IF DBFOM Agreement;

- (b) not suspend its performance (including in connection with any insolvency or bankruptcy proceeding in relation to Principal Project Company) under the IF DBFOM Agreement, unless the grounds for suspension of performance are failure by the Step-in Party to perform Principal Project Company's obligations under the IF DBFOM Agreement;
- (c) not take or support any action for the liquidation, bankruptcy, administration, receivership, reorganization, dissolution or winding up of Principal Project Company or for the composition or readjustment of Principal Project Company's debts, or any similar insolvency procedure in relation to Principal Project Company, or for the appointment of a receiver, trustee, custodian, sequestrator, conservator, liquidator, administrator or similar official for Principal Project Company or for any part of Principal Project Company's property;
- (d) continue to make payments required to be made to Principal Project Company under the IF DBFOM Agreement to the Proceeds Account or to an alternate account designated by the Collateral Agent in accordance with <u>Section 7.4</u>; and
- (e) endorse or pay over, as directed by the Collateral Agent, any checks received by City with respect to, or funds drawn by City under, the Performance Bond; provided that the Collateral Agent reimburses City for any Losses incurred by City in attempting to cure the PPC Default as and to the extent: (i) City is entitled to such reimbursement pursuant to the IF DBFOM Agreement; (ii) City has promptly notified the Collateral Agent of such Losses at or prior to the time of endorsement or payment and (iii) the Collateral Agent's obligation to reimburse City for such Losses does not exceed the proceeds from any such security.

**10.3** City shall owe its obligations under the IF DBFOM Agreement and this Direct Agreement to Principal Project Company and the Step-in Party jointly; provided that:

- the receipt of, or performance by City in favor of, either such Step-in Party or Principal Project Company shall be a good and effective discharge of City's obligations under this Direct Agreement and the IF DBFOM Agreement;
- (b) the Collateral Agent shall be entitled at any time by notice in writing to City to direct (such direction being binding on the Collateral Agent, City and Principal Project Company) that, at all times during the Step-in Period, the Step-in Party shall be solely entitled to make any decisions, to give any directions, approvals or consents, to receive any payments or otherwise to deal with City under the IF DBFOM Agreement and this Direct Agreement; and

(c) any amount due from Principal Project Company to City under the IF DBFOM Agreement or this Direct Agreement as of the Step-in Date and notified to such Step-in Party prior to the Step-in Date shall be paid to City on the Step-in Date, failing which City shall be entitled to exercise its rights under the IF DBFOM Agreement in respect of the amount so due and unpaid.

**10.4** Principal Project Company shall not be relieved from any of its obligations under the IF DBFOM Agreement or this Direct Agreement, whether arising before or after the Step-in Date, by reason of the Step-in Party exercising the rights provided herein, except to the extent provided in <u>Section 8.2</u> and <u>Section 11</u> (Step-out).

# 11. STEP-OUT

A Step-in Party may, at any time, by giving not less than 30 days' prior written notice ("**Step-out Notice**") to City terminate its obligations to City under this Direct Agreement, in which event such Step-in Party shall be released from all obligations under this Direct Agreement, except for any obligation or liability of the Step-in Party arising during the Step-in Period. The obligations of City to the Step-in Party under this Direct Agreement shall also terminate on the Step-Out Date. Notwithstanding the foregoing, this Direct Agreement shall continue to remain effective according to its terms after the Step-Out Date if the Step-in Party is the Collateral Agent or a Lender.

# 12. SUBSTITUTION ENTITIES AND SUBSTITUTION PROPOSALS

**12.1** Any payment to be made or action to be taken by the Collateral Agent as a prerequisite to keeping the IF DBFOM Agreement in effect shall be deemed properly to have been made or taken by the Collateral Agent if such payment is made or action is taken by a Substituted Entity proposed by the Collateral Agent and reasonably approved by City. City shall have no obligation to recognize any claim to PPC's Interest by any person or entity that has acquired PPC's Interest by, through, or under any Security Document or whose acquisition shall have been derived immediately from any holder thereof, unless such person or entity is a Substituted Entity reasonably approved by City in accordance with this <u>Section 12</u> (Substitution Entities and Substitution Proposals).

**12.2** The Collateral Agent may give a notice ("**Substitution Notice**") under this <u>Section 12</u> (Substitution Entities and Substitution Proposals) in writing to City at any time:

- (a) during any Cure Period;
- (b) during any Step-in Period; or
- (c) after delivery of a Lender Notice of the occurrence of any Event of Default and prior to delivery by the Collateral Agent to City of a further notice under <u>Section 7.9</u>.

**12.3** In any Substitution Notice, the Collateral Agent shall notify City that it intends to designate a Substituted Entity.

**12.4** The Collateral Agent shall, as soon as practicable, provide to City the information regarding the proposed Substituted Entity and any third party entering into a material

subcontract with such Substituted Entity as required by this <u>Section 12</u> (Substitution Entities and Substitution Proposals), including:

- (a) the name and address of the proposed Substituted Entity;
- (b) the names of the proposed Substituted Entity's shareholders or members and the share capital or partnership or membership interests, as the case may be, held by each of them;
- (c) the manner in which the proposed Substituted Entity will be financed and the extent to which such financing is committed;
- (d) copies of the proposed Substituted Entity's most recent financial statements (and if available such financial statements shall be for the last three financial years) or in the case of a newly-formed special purpose company its opening balance sheet;
- (e) a copy of the proposed Substituted Entity's formation documents;
- (f) details of the resources available to the proposed Substituted Entity and the proposed Substituted Entity's appropriate qualifications, experience and technical competence available to the proposed Substituted Entity to enable it to perform the obligations of Principal Project Company under the IF DBFOM Agreement;
- (g) the names of the proposed Substituted Entity's directors and any key personnel who will have responsibility for the day-to-day management of its participation in the Project;
- a rectification plan providing details of the plan to rectify Principal Project Company's breaches with respect to the breaches which are capable of being rectified by the Substituted Entity;
- such other information, evidence and supporting documentation concerning the identity, financial resources, pre-qualifications, experience and potential conflicts of interest of the proposed Substituted Entity and its contractors as City may reasonably request; and
- (j) such evidence of organization, authority, incumbency certificates, certificates regarding debarment or suspension, and other certificates, representations and warranties as City may reasonably request.

**12.5** City will approve or disapprove a proposed Substituted Entity within 45 days after it confirms receipt from the Collateral Agent of a request for approval together with the information required under <u>Section 12.4</u>. City will evaluate the financial resources, qualifications, experience and potential conflicts of interest of the proposed Substituted Entity and its contractors using the same standards and criteria that it is then currently applying, or if there is no current application, then the same standards and criteria it most recently applied, to the evaluation of Persons responding to City's requests for qualifications for concession or similar agreements for comparable projects and facilities.

- **12.6** City shall have no obligation to approve the proposed Substituted Entity:
  - (a) unless the Collateral Agent demonstrates to City's reasonable satisfaction that: (i) the proposed Substituted Entity and its contractors collectively have the financial resources, qualifications and experience to timely perform Principal Project Company's obligations under the Contract Documents and Key Contracts to which Principal Project Company is a party; (ii) the proposed Substituted Entity and its contractors, each of their respective direct and indirect beneficial owners, any proposed key personnel, each of their respective officers and directors and each of their respective affiliates have a good and sound background and reputation (including the absence of criminal, civil or regulatory claims or actions against any such Person, and each such Person's adherence to Good Industry Practice, contract terms and applicable standards regarding past or present performance on comparable projects); and (iii) the proposed Substituted Entity and its contractors are in compliance with City's rules, regulations and adopted written policies regarding pre-qualification and organizational conflicts of interest;
  - (b) if there are unremedied breaches under the IF DBFOM Agreement and there is no rectification plan reasonably acceptable to City with respect to the breaches which are capable of being rectified by the Substituted Entity; and
  - (c) if any proposed security interests to be granted by the proposed Substituted Entity to the Collateral Agent and/or the Lender in addition to (or substantially different from) the security interests granted to the Collateral Agent and/or the Lender under the Initial Financing Documents materially and adversely affect the ability of the Substituted Entity to perform Principal Project Company's obligations under the Contract Documents or have the effect of increasing any liability of City, whether actual or potential (unless a Rescue Refinancing is concurrently proposed, in which case the Lenders' Liabilities may increase by up to 10%).

**12.7** The Collateral Agent may request approval of more than one Substituted Entity. The Collateral Agent may request approval at any time or times. Any approval by City of a Substituted Entity shall expire (unless otherwise agreed in writing by City) one year after the approval is issued if the Substituted Entity has not succeeded to PPC's Interest within that period of time. City may revoke an approval if at any time prior to succeeding to PPC's Interest the Substituted Entity ceases to be in compliance with City's rules and regulations regarding organizational conflicts of interest. If the Substituted Entity succeeds to PPC's Interest, then City shall not be entitled to terminate due to Noncompliance Points accumulated by Principal Project Company prior to its replacement by the Substituted Entity, provided the Noncompliance Event that resulted in such Noncompliance Points are being cured by the Substituted Entity as quickly as practicable using commercially reasonable efforts. Once all Noncompliance Events have been cured, City shall cancel any Noncompliance Points accured prior to succession.

**12.8** Notwithstanding the foregoing, any entity that is wholly owned by a Lender or group of Lenders shall be deemed a Substituted Entity, without necessity for City approval, upon delivery to City of documentation proving that the entity is duly formed, validly existing and wholly owned by the Lender, including a certificate signed by a duly authorized officer of each Lender in favor of City certifying, representing and warranting such ownership.

# 13. SUBSTITUTION

**13.1** If City approves (or is deemed to have approved) a Substitution Notice pursuant to <u>Section 12</u> (Substitution Entities and Substitution Proposals), the Substituted Entity named therein shall execute a duly completed Substitute Accession Agreement substantially in the form attached to this Direct Agreement as Schedule A and submit it to City (with a copy thereof to the other parties to this Direct Agreement) and such assignment shall become effective on and from the date on which City countersigns the Substitute Accession Agreement (the "**Substitution Effective Date**") or the date that is 10 days after the date City receives the completed Substitute Accession Agreement.

- **13.2** As of the Substitution Effective Date:
  - such Substituted Entity shall become a party to the IF DBFOM Agreement and this Direct Agreement in place of Principal Project Company who shall be immediately released from its obligations arising under, and cease to be a party to, the IF DBFOM Agreement and this Direct Agreement from and after Substitution Effective Date;
  - (b) all of Principal Project Company's obligations and liabilities under the IF DBFOM Agreement and under this Direct Agreement arising from and after the Substitution Effective Date shall be immediately and automatically transferred to the Substituted Entity;
  - (c) such Substituted Entity shall exercise and enjoy the rights and perform the obligations of Principal Project Company under the IF DBFOM Agreement and this Direct Agreement; and
  - (d) City shall owe its obligations (including any undischarged liability with respect to any loss or damage suffered or incurred by Principal Project Company prior to the Substitution Effective Date) under the IF DBFOM Agreement and this Direct Agreement to such Substituted Entity in place of Principal Project Company, subject to City's right to offset any losses or damages suffered or incurred by City as provided under the IF DBFOM Agreement and this Direct Agreement.

**13.3** City shall use its reasonable efforts to facilitate the transfer to the Substituted Entity of Principal Project Company's obligations under the IF DBFOM Agreement and this Direct Agreement.

**13.4** The Substituted Entity shall pay to City on the Substitution Effective Date any amount due to City under the IF DBFOM Agreement and this Direct Agreement, including City's reasonable costs and expenses incurred in connection with (a) Principal Project Company's default and termination, (b) City's activities with respect to the Project during any period City was in possession of the Project, and (c) the approval of the Substituted Entity, all as of the Substitution Effective Date and notified to such Substituted Entity prior to the Substitution Effective Date. City's receipt of the payment pursuant to this <u>Section 13.4</u> shall be a condition precedent to the Substitution Effective Date.

**13.5** As of the Substitution Effective Date:

- (a) any right of termination suspended by virtue of <u>Section 8.1</u> shall be of no further effect and City shall not be entitled to terminate or suspend performance of the IF DBFOM Agreement and this Direct Agreement by virtue of any act, omission or circumstance that occurred prior to such Substitution Effective Date; and
- (b) City shall enter into an equivalent direct agreement on substantially the same terms as this Direct Agreement, save that Principal Project Company shall be replaced as a party by the Substituted Entity.

## 14. **REVIVAL OF REMEDIES**

**14.1** If a City Notice has been given, the grounds for that notice are continuing and have not been remedied or waived and:

- (a) as of the end of the Cure Period, no Step-in Notice has been given and no Substituted Entity becomes a party to the IF DBFOM Agreement and this Direct Agreement; or
- (b) the Step-in Period ends without a Substituted Entity becoming a party thereto,

then, from and after the date such Cure Period or such Step-in Period, as the case may be, expires (the "**Revival Date**"), City shall be entitled to:

- act upon any and all grounds for termination or suspension available to it in relation to the IF DBFOM Agreement in respect of defaults under the IF DBFOM Agreement not remedied or waived;
- (ii) pursue any and all claims and exercise any and all remedies against Principal Project Company; and
- (iii) if and to the extent that it is then entitled to do so under the IF DBFOM Agreement, take or support any action of the type referred to in <u>Section 16.2</u> (City Remedies for PPC Default) of the IF DBFOM Agreement.

## 15. NEW PROJECT AGREEMENT

- 15.1 lf:
  - (a) the IF DBFOM Agreement is rejected by a trustee or debtor-in-possession in, or terminated as a result of, any bankruptcy or insolvency proceeding involving Principal Project Company, or
  - (b) a PPC Default under Section 16.1.1(m) of the IF DBFOM Agreement occurs with respect to any Equity Member with a material financial obligation owing to Principal Project Company for a Committed Investment, and Equity Members' obligations relating to Principal Project Company or the Project are rejected by a trustee or debtor-in-possession in, or terminated as a result of any bankruptcy or insolvency proceeding involving such Equity Member and, within 90 days after such rejection or termination, the Collateral Agent shall so request and shall

certify in writing to City that it intends to perform the obligations of Principal Project Company as and to the extent required under the IF DBFOM Agreement,

then City will execute and deliver to the Collateral Agent (or any Substituted Entity satisfying the requirements of this Direct Agreement if directed to do so by the Collateral Agent) a new project agreement. Such new project agreement shall contain conditions, agreements, terms, provisions and limitations which are the same as those of the IF DBFOM Agreement, except for any obligations that have been fulfilled by Principal Project Company, any party acting on behalf of or stepping-in for Principal Project Company or City prior to such rejection or termination. References in this Direct Agreement to the "Agreement" shall be deemed also to refer to any such new project agreement.

## 16. **RECEIVERS**

**16.1** The appointment of a receiver at the behest of Principal Project Company shall be subject to City's prior written approval in its sole discretion. The appointment of a receiver at the behest of any Lender shall be subject to the following terms and conditions:

- (a) City's prior approval shall not be required for the appointment of the receiver or the selection of the Person to serve as receiver;
- (b) whenever any Lender commences any proceeding for the appointment of a receiver, the Collateral Agent shall serve on City not less than ten (10) days' prior written notice of the hearing for appointment and of the Lender's pleadings and briefs in the proceeding;
- (c) City may appear in any such proceeding to challenge the selection of the Person to serve as receiver, but waives any other right to oppose the appointment of the receiver; and
- (d) City may at any time seek an order for replacement of the receiver by a different receiver.

**16.2** No receiver appointed at the behest of Principal Project Company or any Lender shall have any power or authority to replace the IFM Contractor or D&C Contractor except by reason of default or unless the replacement is a Substituted Entity reasonably approved or deemed approved by City.

### 17. ESTOPPEL CERTIFICATES

**17.1** At any time and from time to time, within 30 days after written request of any Lender or proposed Lender, City, without charge, shall certify by written instrument duly executed and acknowledged, to any Lender or proposed Lender as follows:

- (a) as to whether the IF DBFOM Agreement has been supplemented or amended, and if so, attaching a copy of such supplement or amendment to such certificate;
- (b) as to the validity and force and effect of the IF DBFOM Agreement against City, in accordance with its terms, subject to applicable bankruptcy, insolvency and

similar laws affecting the enforceability of creditors' rights generally and the general principles of equity;

- (c) as to the existence of any PPC Default of which City has Actual Knowledge;
- (d) as to the existence of events which, by the passage of time or notice or both, would constitute a PPC Default, to City's Actual Knowledge;
- (e) as to the then accumulated amount of Noncompliance Points;
- (f) as to the existence of any Claims by City regarding the IF DBFOM Agreement; and
- (g) as to the Effective Date and the expiration date of the Term.

**17.2** City shall deliver the same certified, written instrument to a Substituted Entity or proposed Substituted Entity within fifteen (15) days after receiving its written request, provided that the request is delivered to City either before the proposed Substituted Entity succeeds to PPC's Interest or within sixty (60) days after the Substituted Entity has succeeded to PPC's Interest.

**17.3** Any such certificate may be relied upon by, and only by, the Lender, proposed Lender, Substituted Entity or proposed Substituted Entity to whom the same may be delivered, and the contents of such certificate shall be binding on City.

### 18. GENERAL

**18.1** Neither the Lender nor the Collateral Agent shall have any obligation hereunder to extend credit to City or any contractor to City at any time, for any purpose.

**18.2** For so long as any amount under the Financing Documents is outstanding, City shall not, without the prior written consent of the Collateral Agent, consent to any assignment, transfer, pledge or hypothecation of the IF DBFOM Agreement or any interest therein by Principal Project Company, other than as specified in the IF DBFOM Agreement or this Direct Agreement.

**18.3** No Lender holding Equity Member Debt, whether secured or unsecured, shall have any rights, benefits or protections under this Direct Agreement.

### 19. TERMINATION

This Direct Agreement shall remain in effect until the earlier to occur of (a) the Discharge Date; (b) the time at which all of City's obligations and liabilities have expired or have been satisfied in accordance with the terms of the IF DBFOM Agreement and this Direct Agreement; and (c) any assignment to a Substituted Entity has occurred under <u>Section 13</u> (Substitution) and City shall have entered into an equivalent direct agreement on substantially the same terms as this Direct Agreement, save that Principal Project Company has been replaced as a party by the Substituted Entity.

# 20. EFFECT OF BREACH

Without prejudice to any rights a party may otherwise have, a breach of this Direct Agreement shall not of itself give rise to a right to terminate the IF DBFOM Agreement.

## 21. NO PARTNERSHIP

Nothing contained in this Direct Agreement shall be deemed to constitute a partnership between the parties to this Direct Agreement. None of the parties shall hold itself out contrary to the terms of this <u>Section 21</u> (No Partnership).

## 22. REMEDIES CUMULATIVE; NO WAIVER

No failure or delay by City, the Lenders or the Collateral Agent (or their designee) in exercising any right or power hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any such right or power, or any abandonment or discontinuance of steps to enforce such a right or power, preclude any other or further exercise thereof or the exercise of any other right or power. The remedies provided herein are cumulative and not exclusive of any remedies provided by law and may be exercised by the Lenders, the Collateral Agent or any designee, transferee or assignee thereof from time to time. In no event shall any provision of this Direct Agreement or any consent to any departure by any party therefrom be effective unless such waiver is permitted by <u>Section 23</u> (Amendment), and then such waiver or consent shall be effective only in the specific instance and for the purpose for which given.

## 23. AMENDMENT

No amendment, modification or waiver of any provision of this Direct Agreement, or consent to any departure herefrom by any party to this Direct Agreement, shall be effective against any party to this Direct Agreement unless the same shall be in writing and signed by the party against whom enforcement is sought, and then such amendment or waiver shall be effective only in the specific instance and for the specific purpose for which it was given.

# 24. SUCCESSORS AND ASSIGNS

**24.1** No party to this Direct Agreement may assign or transfer any part of its rights or obligations hereunder without the consent of the other parties, save that the Collateral Agent may assign or transfer its rights and obligations hereunder to a successor Collateral Agent in accordance with the Financing Documents. In connection with any such assignment or transfer, City agrees to enter into a new Direct Agreement with the successor Collateral Agent on terms that are substantially the same as those of this Direct Agreement.

**24.2** This Direct Agreement shall be binding upon and inure to the benefit of the parties to this Direct Agreement and their respective successors and permitted assigns.

# 25. COUNTERPARTS

This Direct Agreement may be executed in any number of counterparts, each of which shall be identical and all of which, taken together, shall constitute one and the same instrument, and the parties may execute this Direct Agreement by signing any such counterpart. Transmission by facsimile or electronic mail of an executed counterpart of this Direct Agreement shall be deemed

to constitute due and sufficient delivery of such counterpart, to be followed thereafter by an original of such counterpart. The Parties, in the manner specified by City, may sign this Direct Agreement electronically.

## 26. SEVERABILITY

If, at any time, any provision of this Direct Agreement is or becomes illegal, invalid or unenforceable in any respect under any law of any jurisdiction, neither the legality, validity or enforceability of the remaining provisions nor the legality, validity or enforceability of such provision under the law of any other jurisdiction will in any way be affected or impaired.

## 27. NOTICES

**27.1** Any notice, approval, election, demand, direction, consent, designation, request, agreement, instrument, certificate, report or other communication required or permitted to be given or made under this Direct Agreement (each, a notice) to a party must be given via written correspondence and email. All notices will be validly given if given on a Business Day to each party at the following address:

To City:	San Francisco Municipal Transportation Agency 1 South Van Ness, 8th Floor San Francisco, CA 94103 Attn: Chris Lazaro Email: <u>Chris.Lazaro@sfmta.com</u>
with copies to:	Office of the City Attorney City Hall, Room 234 1 Dr. Carlton B. Goodlett Place San Francisco, California 94102-4682 Attn: Real Estate & Finance Group Re: Potrero Yard Modernization Project Email: isidro.jimenez@sfcityatty.org
To Principal Project Company:	[Address] Attention: [] E-Mail: []
with copies to:	[Address] Attention: [] E-Mail: []
To the Collateral Agent:	[] [Address] Attention: [] Email: []

- **27.2** A notice shall be deemed to have been given on the earliest of:
  - (a) date of receipt, if delivered in person;

- (b) date of receipt (confirmed by automatic answer back or equivalent evidence of receipt), if validly transmitted electronically before 3:00 p.m. (local time at the place of receipt) on a Business Day;
- (c) one Business Day after delivery to the courier properly addressed, if delivered by overnight courier; and
- (d) four Business Days after deposit with postage prepaid and properly addressed, if delivered by United States certified or registered mail.

**27.3** Each of the parties will notify each other via written correspondence and email of any change of address, such notification to become effective 15 days after notification.

## 28. GOVERNING LAW AND JURISDICTION

**28.1** The venue for any litigation arising out of, relating to or resulting from any matter relating to this Direct Agreement shall be in San Francisco, California to the extent that a court located in San Francisco has subject matter jurisdiction.

**28.2** This Direct Agreement shall be governed by and construed in accordance with the laws of the State of California, any applicable federal law, the San Francisco City Charter and Municipal Code, and the ordinances, regulations, codes, and Executive Orders enacted and/or promulgated pursuant thereto.

**28.3** Each of Principal Project Company, City and the Collateral Agent irrevocably consents to service of process by personal delivery, certified mail, postage prepaid or overnight courier. Nothing in this Direct Agreement will affect the right of any party to serve process in any other manner permitted by law.

**28.4** Each of City, Principal Project Company and the Collateral Agent (a) certifies that no representative, agent or attorney of another party has represented, expressly or otherwise, that such party would not, in the event of a proceeding, seek to enforce the mutual waivers in this <u>Section 28</u> (Governing Law and Jurisdiction) and (b) acknowledges that it has been induced to sign, or change its position in reliance upon the benefits of, this Direct Agreement by, among other things, the mutual waivers and certifications in this <u>Section 28</u> (Governing Law and Jurisdiction).

# 29. CONFLICT WITH PROJECT AGREEMENT

In the event of any conflict or inconsistency between the provisions of this Direct Agreement and the IF DBFOM Agreement, the provisions of this Direct Agreement shall prevail.

**IN WITNESS WHEREOF**, each of the parties to this Direct Agreement has caused this Direct Agreement to be duly executed by its duly authorized officer as of the date first written above.

# [REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

## [INSERT COLLATERAL AGENT'S NAME]

By:\_\_\_\_\_

Name:\_\_\_\_\_

Title:\_\_\_\_

# [INSERT PRINCIPAL PROJECT COMPANY'S NAME]

CA Contractors License No.:\_\_\_\_\_

By:\_\_\_\_\_

Name:\_\_\_\_\_

Title:\_\_\_\_\_

# CITY AND COUNTY OF SAN FRANCISCO

By:\_\_

Jeffrey Tumlin Director of Transportation

## Approved as to Form David Chiu, City Attorney

Date:\_\_\_\_\_

By:\_\_\_\_\_] Deputy City Attorney

Attest:

By:\_\_

Secretary (Signature)

Print Name:\_\_\_\_\_

[SEAL]

## SCHEDULE A Form of Substitute Accession Agreement

[Date]

To: City and County of San Francisco For the attention of: Contracting Officer and Chief Counsel [Lender and other parties to Financing Agreements to be listed] [insert address] For the attention of: [\_\_\_\_]

From: [Substituted Entity]

## POTRERO YARD MODERNIZATION PROJECT: INFRASTRUCTURE FACILITY SUBSTITUTE ACCESSION AGREEMENT

Ladies and Gentlemen:

Reference is made to the Design-Build-Finance-Operate-Maintain Agreement, dated as of [\_\_\_\_\_], 2025 (as amended, amended and restated, supplemented or otherwise modified from time to time, the "*IF DBFOM Agreement*") between the City and County of San Francisco ("*City*"), a municipal corporation, acting by order of and through the San Francisco Municipal Transportation Agency, and [\_\_\_\_\_\_] ("*Principal Project Company*") and the Direct Agreement, dated as of [\_\_\_\_\_], 2025 (as amended, amended and restated, supplemented or otherwise modified from time to time, the "*Direct Agreement*") among City, Principal Project Company and [\_\_\_\_\_], as Collateral Agent. Terms defined in the Direct Agreement and not otherwise defined herein have the respective meanings set forth in or incorporated into in the Direct Agreement.

1. We confirm that we are a Substituted Entity pursuant to <u>Section 13</u> (Substitution) of the Direct Agreement.

2. We acknowledge and agree that, upon and by reason of our execution of this Substitute Accession Agreement, we will become a party to the IF DBFOM Agreement and the Direct Agreement as a Substituted Entity and, accordingly, shall have the rights, powers and obligations of Principal Project Company under the IF DBFOM Agreement and the Direct Agreement.

3. Our address, telephone number and address for electronic mail for the purpose of receiving notices are as follows:

### [Contact details of Substituted Entity]

4. This Substitute Accession Agreement is subject to and shall be construed and interpreted in accordance with the laws of the State of California, any applicable federal law, the San Francisco City Charter and Municipal Code, and the ordinances, regulations, codes, and Executive Orders enacted and/or promulgated pursuant thereto.

The terms set forth herein are agreed to:

[Substituted Entity]

By:\_\_\_\_\_ Name: \_\_\_\_\_ Title: \_\_\_\_\_

Agreed for and on behalf of:

# CITY AND COUNTY OF SAN FRANCISCO

By:\_\_\_

Jeffrey Tumlin Director of Transportation

Approved as to Form David Chiu, City Attorney

Date:\_\_\_\_\_

By:\_\_\_\_\_] Deputy City Attorney

### Attest:

By:\_

Secretary (Signature)

Print Name:\_\_\_\_\_

[SEAL]

# EXHIBIT 5C

# **CALCULATION OF REFINANCING GAIN**

## 1. Calculation of the Refinancing Gain

The total amount of Refinancing Gain equals (A - B), as calculated in accordance with this <u>Exhibit 5C</u> (Calculation of the Refinancing Gain), provided that such amount is greater than zero. Each of the variables used to calculate Refinancing Gain shall be determined as follows:

# 1.1 Create a Pre-Refinancing Financial Model

Principal Project Company shall create a pre-Refinancing financial model (the "**Pre-Refinancing Financial Model**") by updating the Financial Model then in effect for actual and projected Project Financial Performance so as to be current immediately prior to the Refinancing, but excluding the impact of the current Refinancing. The Pre-Refinancing Financial Model shall take into account any Refinancing that qualifies as an Exempt Refinancing (under clause (a) of the definition of Exempt Refinancing) exactly as it was reflected in the Base Case Financial Model. In the Pre-Refinancing Financial Model, the resulting net present value of Distributions projected from the anticipated date of the Refinancing through to the end of the Term, discounted at the Original Equity IRR, shall be the value of "B".

# 1.2. Create a Post-Refinancing Financial Model

Principal Project Company shall create a post-Refinancing financial model (the "**Post-Refinancing Financial Model**") by updating the Pre-Refinancing Financial Model for the impact of the Refinancing, including costs reasonably incurred to achieve the Refinancing. In the Post-Refinancing Financial Model, the resulting net present value of Distributions projected from the anticipated date of the Refinancing through to the end of the Term, discounted at the Original Equity IRR, shall be the value of "A".

# 2. Calculating City's 50% Entitlement

# 2.1 Create a Final Refinancing Financial Model

**2.1.1** Principal Project Company shall create a final refinancing financial model by updating the Post-Refinancing Financial Model to reflect the impact of the amounts and timing of the payment(s) (or credits) to be made to City under <u>Section 4.5</u> (Refinancing Gain) of the Agreement, such that the net present value of such payments, discounted at the Original Equity IRR, is equal to 50% of the amount equal to (A - B). The resulting nominal post-tax equity internal rate of return shall be deemed to be the "**Refinancing Equity IRR**."

**2.1.2** If the Refinancing Equity IRR is lower than the Original Equity IRR, then Principal Project Company shall adjust the amounts and timing of the payment(s) (or credits) to be made to City under <u>Section 4.5</u> (Refinancing Gain) of the Agreement, such that the Refinancing Equity IRR would be increased to a level up to, but not in excess of, the Original Equity IRR.

## 3. Mutual Agreement and Financial Model Update

Subject to the mutual agreement by Principal Project Company and City of the Refinancing Gain calculations and entitlement amount, Principal Project Company shall prepare a Financial Model Update to the Financial Model then in effect in accordance with <u>Section 4.7.2</u> (Updates to the Financial Model) of the Agreement that incorporates actual and projected Project Financial Performance, the impact of the Refinancing and the sharing of the Refinancing Gain.

# EXHIBIT 5D

## FORM OF OPINION FROM CITY'S LEGAL COUNSEL

## [City Attorney's Office Letterhead]

## 1. [Insert Address of Principal Project Company and Lender(s)]

. 2025

Ladies and Gentlemen:

I am a Deputy City Attorney assigned to provide counsel to the City and County of San Francisco ("**City**"), a municipal corporation acting by and through the San Francisco Municipal Transportation Agency ("**SFMTA**"), and as such have advised City in connection with its execution of:

(i) the Infrastructure Facility Design-Build-Finance-Operate-Maintain Agreement for the Potrero Yard Modernization Project dated as of [\_\_\_], 2025 by and between the City and [\_\_\_], a [\_\_\_] (the "**Principal Project Company**"); and

(ii) the Direct Agreement dated as of [\_\_\_], 2025 by and between the City, the Principal Project Company and [\_\_\_], the collateral agent for the Lenders.

The agreements in <u>clauses (i) – (ii)</u> are collectively referred to hereinafter as the "**City Agreements**".

We have examined executed originals or copies identified to our satisfaction of each of the City Agreements. We have further examined such other documents, certificates and other materials as we have deemed necessary in order to render the opinions expressed herein.

With respect to the various factual matters material to my opinion, we have relied upon certificates and representations of the City and other public officials. We have assumed the due execution and delivery, pursuant to due authorization of the City Agreements by parties other than the City, the validity and binding effect thereof as to such other parties and the genuineness of all signatures (other than those of the City) on all documents seen or reviewed by us, the authenticity of all documents submitted to us as originals and the conformity with the original documents of all documents submitted to us as copies.

Whenever an opinion expressed herein is stated to be to my knowledge, or known to me, it means that, during the course of my representation of the City, I have not acquired information giving me actual knowledge of the existence or absence of the facts forming the basis for such opinion, and that, except to the extent expressly set forth herein I have not conducted an independent investigation to determine the existence or absence of such facts.

Based on the foregoing, I am of the opinion that:

1. Ordinance No. 38-21, adopted by the Board of Supervisors on March 16, 2021, has been duly enacted and is in full force and effect.

- 2. The City and County of San Francisco is a charter city and municipal corporation duly organized and validly existing under the Constitution of the State of California.
- 3. The City is duly organized and operating pursuant to the San Francisco Charter and has the full legal capacity, right, power and authority to enter into and deliver the City Agreements. The City has the full legal capacity, right, power and authority to carry out and perform its obligations under the City Agreements.
- 4. Each of the City Agreements has been duly authorized, executed and delivered by the City, and such documents are legal, valid and binding instruments of the City and are enforceable against the City in accordance with their terms, except as enforceability may be limited or otherwise affected by (a) bankruptcy, insolvency, reorganization, moratorium and other laws affecting the rights of creditors generally, (b) principles of equity, whether considered at law or in equity, (c) the sovereign immunity of the City; provided that sovereign immunity shall not bar the enforcement of claims presented in accordance with the provisions of the applicable City Agreement, City Charter and Administrative Codes and California law, so long as such claims are based on contractual rights, and (d) the limitations and conditions set forth herein.
- 5. The execution and delivery by the City of the City Agreements and the performance of its obligations under the City Agreements are within its powers and do not and will not conflict with, or constitute a breach or result in a violation of (a) any existing constitutional or statutory provision of the State of California, (b) to my knowledge, any existing agreement or other instrument to which the City is a party or by which it is bound, or (c) to my knowledge, any existing order, rule, regulation, judgment, decree ordinance of any court, government or governmental authority having jurisdiction over the City as applicable or its properties.
- 6. All official action required to be taken by the City and all consents, approvals, authorizations or orders of or filings with or notice to, any governmental or regulatory authority necessary to authorize and enable the City to execute and deliver the City Agreements or to authorize and enable City to perform its obligations thereunder (other than acts under the City Agreements that may be taken by City in the future) have been taken and obtained.

The opinions expressed in this letter are limited to the opinions expressly stated, and no other opinions should be inferred. The opinions are expressed only as of the date hereof.

Sincerely,

[Insert Signature]

# <u>EXHIBIT 5E</u>

# FORM OF OPINION FROM PRINCIPAL PROJECT COMPANY'S LEGAL COUNSEL

[Insert Firm Letterhead]

[Insert City Address]

[Insert Lender Addressees]

\_\_\_\_, 2025

Ladies and Gentlemen:

We [\_\_\_] ("Firm") have acted as special counsel to [\_\_\_], a [\_\_\_] (the "Principal Project Company") in connection with the Infrastructure Facility Design-Build-Finance-Operate-Maintain Agreement for the Potrero Yard Modernization Project dated as of [\_\_\_], 2025 (the "Agreement") between the City and County of San Francisco ("City"), a municipal corporation acting by and through the San Francisco Municipal Transportation Agency ("SFMTA"), and the Principal Project Company (Contract # \_\_\_\_). This opinion is being furnished pursuant to Section 3.2.4 (Conditions Precedent to Financial Close) of the Agreement. Capitalized terms used herein without definition have the meanings ascribed to such terms in the Agreement.

In connection with the opinions contained herein, we have examined executed counterparts, or copies of such executed counterparts certified or otherwise identified to our satisfaction, of:

- (i) the Agreement;
- (ii) the Direct Agreement, by and between the City, the Principal Project Company and the Collateral Agent;
- (iii) the Equity Members Funding Agreements;
- (iv) the D&C Contract;
- (v) the IFM Contract; and
- (vi) the Financing Documents.

### [add other agreements, if necessary]

Collectively, (i) through (iv) above shall be hereinafter referred to as the **"Transaction Documents**".

We have further examined and relied upon the accuracy of original, certified, conformed, photocopied or telecopied copies of such records, agreements, certificates and other

documents as we have deemed necessary or appropriate to enable us to render the opinions expressed herein. In all such examinations we have assumed the genuineness of signatures on original documents and the conformity to such original documents of all copies submitted to us as certified, conformed, photocopied or telecopied copies. We have further assumed that none of such documents has been subsequently rescinded, revoked, restated, modified or amended in any way other than by documents that have been submitted to us. We have made no independent inquiry or investigation of any factual matters or circumstances relevant to the opinions herein set forth, but instead have relied solely upon the accuracy of oral or written statements and representations of officers and other representatives of the Principal Project Company, statements, representations and warranties made in the Transaction Documents.

In our examination, we have assumed, without independent investigation, the following:

- (i) the Transaction Documents have been duly authorized, executed and delivered by each party thereto (other than the Principal Project Company);
- that each party (other than the Principal Project Company) to each Transaction Document is duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization, and has full power and authority to enter into and to carry out its obligations under such Transaction Document;
- (iii) the authorization, execution and delivery by each party (other than the Principal Project Company) of each Transaction Document to which it is a party does not, and such party's performance thereunder will not, breach, conflict with, or constitute a violation of, (A) the organizational documents, bylaws or similar governing documents of such party or (B) any laws or any writ, order, injunction or decree of any court or governmental authority or any provision of any agreement or instrument to which such party or its properties may be bound;
- (iv) that each Transaction Document is the legal, valid and binding obligation of each party thereto (other than the Principal Project Company) enforceable against such party in accordance with the terms of such Transaction Document;
- (v) the legal capacity and competency of all Persons signing the Transaction Documents on behalf of the parties thereto (other than the Principal Project Company); and
- (vi) that each party to the Transaction Documents (other than the Principal Project Company) has all necessary governmental consents, approvals, licenses or permits required in order to execute and deliver the Transaction Documents to which it is a party.

The opinions set forth below that make reference to, or are stated to be qualified by, the expression "to our knowledge" or any expression of similar import are limited to the current actual knowledge of the individual attorneys at this Firm who have devoted substantial attention to the representation of the Principal Project Company in connection with the preparation, negotiation, execution and delivery of the Transaction Documents (but not the knowledge of any other attorney

of this Firm or any constructive or imputed knowledge of any information, whether by reason of our representation of the Principal Project Company or otherwise). We have not undertaken an independent investigation to determine the accuracy of any such statement, and any limited inquiry undertaken by us during the preparation of this letter should not be regarded as such an investigation.

Based upon the foregoing and our examination of such questions of law as we have deemed necessary or appropriate, and subject to limitations and qualifications set forth below, it is our opinion that:

- 1. The Principal Project Company is a [\_\_\_] validly existing and in good standing under the laws of [\_\_\_] and has all requisite limited liability company power to execute and deliver, and to perform its obligations under, the Transaction Documents.
- 2. The Principal Project Company is qualified to transact interstate business as a foreign limited liability company in the State of California.
- 3. The execution and delivery of, and the performance of obligations under, the Transaction Documents by the Principal Project Company, have been duly authorized by all necessary action on the part of the Principal Project Company. Each person executing such Transaction Documents on the Principal Project Company's behalf has been duly authorized to execute and deliver each such document on Principal Project Company's behalf, and such Transaction Documents have been duly executed and delivered by the Principal Project Company.
- 4. Each of the Transaction Documents constitutes the valid and binding obligation of the Principal Project Company, enforceable against the Principal Project Company in accordance with the terms thereof.
- 5. The execution and delivery by the Principal Project Company of the Transaction Documents does not, and the consummation by the Principal Project Company of any of the transactions contemplated thereunder will not, result in the violation by Principal Project Company of any United States federal statute, rule or regulation, which in our experience is normally applicable with respect to transactions of the type contemplated by the Transaction Documents (without taking into account the particular nature of the business conducted by the parties to the Transaction Documents). The execution and delivery by the Principal Project Company of the Transaction Documents does not, and the consummation by it of the transactions contemplated thereunder will not, result in a violation by Principal Project Company of any provision of the LLC Agreement (as defined in Annex I) or the Certificate of Formation (as defined in Annex I).
- 6. The execution and delivery by the Principal Project Company of the Transaction Documents and the performance of its obligations under the Transaction Documents are within its powers and do not and will not conflict with, or constitute a breach or result in a violation of (a) any existing constitutional or statutory provision of the State of California, (b) to my knowledge, any existing agreement or other instrument to which Principal Project Company is a party or by which it is

bound, or (c) to my knowledge, any existing order, rule, regulation, judgment, decree ordinance of any court, government or governmental authority having jurisdiction over Principal Project Company as applicable or its properties.

Our opinions are subject to the following additional assumptions, limitations and qualifications:

A. Our opinions are subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws of general applicability relating to or affecting creditors' rights and to general principles of equity. Without limiting the foregoing qualifications, the opinions expressed herein do not purport to cover, and we express no opinion with respect to, the applicability of Section 548 of the federal Bankruptcy Code or any comparable provision of state law, including the provisions relating to fraudulent conveyances.

B. Indemnities, rights of contribution, exculpatory provisions, provisions in respect of rights to liquidated damages, penalties or punitive damages, waivers of rights and provisions requiring arbitration of disputes may be limited on public policy grounds or may be prohibited by law.

C. Certain of the remedies in the Transaction Documents may be limited or rendered unenforceable by applicable law. In our opinion, however, applicable law does not render the remedies afforded by the Financing Documents inadequate for the practical realization by the Lenders of the principal benefits intended to be provided by the Financing Documents.

D. We express no opinion with respect to any provision of the Transaction Documents that purports to require a prevailing party in a dispute to pay attorney's fees and expenses, or other costs, to a non-prevailing party.

E. We express no opinion as to the enforceability of any agreements in any Transaction Document by the parties thereto to agree in the future upon any matter.

This letter speaks only as of the date hereof, and we disclaim any undertaking to update this letter to take into account any future changes of fact or law. The opinions expressed herein are solely for your benefit, and may not be relied on in any manner or for any purposes by any other Person without our written consent, and are not to be used, circulated, quoted, published or otherwise referred to or disseminated for any other purpose.

Very truly yours,

[Insert Signature]

# EXHIBIT 5F

# BASE CAPITAL MaxAP ADJUSTMENT FOR BASE INTEREST RATE FLUCTUATION AND CREDIT SPREAD RISK MITIGATION

To facilitate the adjustment process described in this Exhibit, Principal Project Company and City will collaboratively prepare a Financial Model closing protocols document that will set forth additional details for how the calculations set forth in this Exhibit will be applied to the Base Case Financial Model and the Key Ratios therein.

## 1. Base Interest Rate Fluctuations

**1.1** Promptly following Financial Close, Principal Project Company shall update the Base Case Financial Model in effect as of the Effective Date to:

- (a) reflect 100% of the changes to the Base Interest Rates that have occurred during the Bond Rate Protection Period or Bank Debt Rate Protection Period (as applicable), without changing the coupon structure of any Bond Financing;
- (b) solve for the lowest possible MaxAP;
- (c) maintain the Equity IRR to be equal to the Original Equity IRR; and
- (d) satisfy all Key Ratios in the Base Case Financial Model.

**1.2** The capital portion of the MaxAP resulting from the update to the Base Case Financial Model in accordance with <u>Section 1.1</u> will be the Base Interest Rate adjusted Base Capital MaxAP ("**MaxAP**<sub>BIRBase</sub>") and the resulting Financial Model will be referred to herein as "**BIR Financial Model**").

# 2. Credit Spread Fluctuations

- **2.1** If Principal Project Company is not eligible for Credit Spread Risk Mitigation:
  - Principal Project Company shall bear 100% of the impact (either positive or negative) on its Original Equity IRR for changes in credit spreads over the Bond Rate Protection Period;
  - (b) the remainder of this <u>Section 2</u> (Credit Spread Fluctuations) and <u>Section 3</u> (Adjustments to the Base Capital MaxAP for Base Interest Rate fluctuation and Credit Spread Risk Mitigation) of this <u>Exhibit 5F</u> (Base Capital MaxAP Adjustment for Base Interest Rate Fluctuation and Credit Spread Risk Mitigation) do not apply to Principal Project Company; and
  - (c) MaxAP<sub>BIRBase</sub> calculated pursuant to <u>Section 1</u> (Base Interest Rate Fluctuations) will become the interim Base Capital MaxAP ("MaxAP<sub>INTERIM</sub>") used in the final Financial Model solve required under <u>Section 4</u> (Financial Model Update, Base Capital MaxAP and Base Operating MaxAP Revision).

**2.2** If Principal Project Company is eligible for Credit Spread Risk Mitigation, Principal Project Company shall calculate the following, in each case, to achieve the lowest possible applicable MaxAP and maintain the Original Equity IRR at the same levels reflected in Principal Project Company's Base Case Financial Model while its Key Ratios remain in compliance with the Financing Agreements and consistent with the Base Case Financial Model:

- (a) "MaxAP<sub>BASELINE</sub>" representing the Base Capital MaxAP calculated by updating the BIR Financial Model (which was prepared in accordance with <u>Section 1</u> (Base Interest Rate Fluctuations) of this <u>Exhibit 5F(Base Capital MaxAP</u> Adjustment for Base Interest Rate Fluctuation and Credit Spread Risk Mitigation)) with the applicable Baseline Credit Spreads (if different from those already in BIR Financial Model);
- (b) **"MaxAP**<sub>PRICING</sub>" representing the Base Capital MaxAP calculated by updating the Base Case Financial Model with the actual Bond Financing pricing and the actual net proceeds, coupon and term structure as of the Bond Pricing Date; and
- (c) the upper boundary ("**MaxAP**UPPER") and lower boundary ("**MaxAP**LOWER") between which Principal Project Company shall bear the full impact of changes in credit spreads (the "**No-Mitigation Range**"), calculated as follows:
  - (i) MaxAP<sub>UPPER</sub> shall be the greater of MaxAP<sub>BIRBase</sub> and MaxAP<sub>BASELINE</sub>, MaxAP<sub>UPPER</sub> = Maximum (MaxAP<sub>BIRBase</sub>, MaxAP<sub>BASELINE</sub>)

and

(ii)  $MaxAP_{LOWER}$  shall be the lessor of  $MaxAP_{BIRBase}$  and  $MaxAP_{BASELINE}$  $MaxAP_{LOWER}$  = Minimum (MaxAP\_{BIRBase}, MaxAP\_{BASELINE}).

# 3. Adjustments to the Base Capital MaxAP for Base Interest Rate fluctuation and Credit Spread Risk Mitigation

Principal Project Company shall, subject to City's review and approval, calculate the Base Capital MaxAP ("**MaxAP**<sub>INTERIM</sub>") to be used in <u>Section 4</u> (Financial Model Update, Base Capital MaxAP and Base Operating MaxAP Revision), in accordance with one of the following scenarios:

(a) if MaxAP<sub>PRICING</sub> is above the No-Mitigation Range, then MaxAP<sub>INTERIM</sub> shall be MaxAP<sub>BIRBase</sub> <u>plus</u> 50% of the difference between the MaxAP<sub>PRICING</sub> and the higher of the MaxAP<sub>BASELINE</sub> and MaxAP<sub>BIRBase</sub>:

if  $MaxAP_{PRICING} > MaxAP_{UPPER}$ , then  $MaxAP_{INTERIM} = MaxAP_{BIRBase} + 50\% \times (MaxAP_{PRICING} - MaxAP_{UPPER})$ ;

(b) if MaxAP<sub>PRICING</sub> is below the No-Mitigation Range, then MaxAP<sub>INTERIM</sub> shall be MaxAP<sub>BIRBase</sub> reduced by 50% of the difference between the MaxAP<sub>PRICING</sub> and the lower of the MaxAP<sub>BASELINE</sub> and MaxAP<sub>BIRBase</sub>:

if  $MaxAP_{PRICING} < MaxAP_{LOWER}$ , then  $MaxAP_{INTERIM} = MaxAP_{BIRBase} + 50\% \times (MaxAP_{PRICING} - MaxAP_{LOWER})$ ; and (c) if MaxAP<sub>PRICING</sub> falls on or within the No-Mitigation Range, then there will be no further adjustment to MaxAP<sub>BIRBase</sub> and MaxAP<sub>INTERIM</sub> will equal MaxAP<sub>BIRBase</sub>.

if (  $MaxAP_{PRICING} \leq MaxAP_{UPPER} AND MaxAP_{PRICING} \geq MaxAP_{LOWER}$ ), then  $MaxAP_{INTERIM} = MaxAP_{BIRBASE}$ .

<u>Tables 3A and 3B</u> below illustrate the scenarios described in <u>Section 3</u> (Adjustments to the Base Capital MaxAP for Base Interest Rate fluctuation and Credit Spread Risk Mitigation).

## Table 3A - Scenario A

After adjusting for movements in Base Interest Rates, the Base Capital MaxAP is equal to or lower than the Base Capital MaxAP calculated using the Baseline Credit Spreads (also adjusted for movements in Base Interest Rates).

	<b>(i)</b> MaxAP <sub>PRICING</sub> is above the No-Mitigation Range	Pricing Outcomes: (ii) MaxAP <sub>PRICING</sub> is below the No-Mitigation Range	<b>(iii)</b> MaxAP <sub>PRICING</sub> falls on or within the No-Mitigation Range
	MaxAP <sub>PRICING</sub> ↑		
MaxAPBASELINE			¥
			MaxAP <sub>PRICING</sub>
MaxAP <sub>BIRBase</sub>			<b>†</b>
		¥	
		MaxAP <sub>PRICING</sub>	
	MaxAP <sub>BIRBase</sub> is increased by 50% of the difference between MaxAP <sub>BASELINE</sub> and MaxAP <sub>PRICING</sub>	MaxAP <sub>BIRBase</sub> is decreased by 50% of the difference between MaxAP <sub>BIRBase</sub> and MaxAP <sub>PRICING</sub>	No adjustment made to MaxAP <sub>BIRBase</sub>

## Table 3B - Scenario B

After adjusting for movements in Base Interest Rates, the Base Capital MaxAP is higher than the Base Capital MaxAP calculated using the Baseline Credit Spreads (also adjusted for

movements in Base Interest Rates).

	(i) MaxAP <sub>PRICING</sub> is above the No-Mitigation Range	Pricing Outcomes: (ii) MaxAP <sub>PRICING</sub> is below the No-Mitigation Range	<b>(iii)</b> MaxAP <sub>PRICING</sub> falls on or within the No-Mitigation Range
	MaxAP <sub>PRICING</sub> ♠		
MaxAP <sub>BIRBase</sub>			+
			MaxAPPRICING
MaxAPBASELINE			♠
		↓ MaxAP <sub>PRICING</sub>	
	MaxAP <sub>BIRBase</sub> is increased by 50% of the difference between MaxAP <sub>BIRBase</sub> and MaxAP <sub>PRICING</sub>	MaxAP <sub>BIRBase</sub> is decreased by 50% of the difference between MaxAP <sub>BASELINE</sub> and MaxAP <sub>PRICING</sub>	No adjustment made to MaxAP <sub>BIRBase</sub>

#### 4. Financial Model Update, Base Capital MaxAP and Base Operating MaxAP Revision

**4.1** On the date of Financial Close, Principal Project Company shall update and solve the Base Case Financial Model then in effect with the following:

- (a) Base Capital MaxAP set equal to the MaxAP<sub>INTERIM</sub> (as calculated in accordance with <u>Section 2.1</u> or <u>Section 3</u> (Adjustments to the Base Capital MaxAP for Base Interest Rate fluctuation and Credit Spread Risk Mitigation), as applicable);
- (b) actual Project Debt pricing, including net proceeds, Base Interest Rates, credit spreads, coupons and term structure as of the Bank Debt Pricing Date or Bond Pricing Date, as applicable;
- (c) all other changes in terms of financing assumed between those indicated in the Bid Financial Model submitted with Principal Project Company's Financial Proposal and in Principal Project Company's financial plan as set forth in the Financing Documents as of the Financial Close Date; and
- (d) all other changes required to be reflected in the Financial Model as result of any steps taken under <u>Section 3.4.2</u> of the Agreement.

**4.2** The financial model resulting from the update to the Base Case Financial Model in accordance with <u>Section 4.1</u> shall be proposed by Principal Project Company to City as the Financial Model Update described in <u>Section 3.3.1</u> of the Agreement and, upon execution of the amendment agreement referred to in <u>Section 3.3.2</u> of the Agreement: (a) the resulting financial model shall become the "**Financial Model**"; (b) the projected Base Capital MaxAP for the first Contract Year (annualized to be a 12-calendar month period) therein will become the new "**Base Capital MaxAP**"; (c) the Base Operating MaxAP shall become the new "**Base Operating MaxAP**"; and (d) the internal rate of return on equity therein shall become the "**Equity IRR**."

## EXHIBIT 5G

## FINANCING DOCUMENT TERMS

(a) The Security Documents may only secure Project Debt, the proceeds of which are obligated to be used exclusively for the purposes of:

(i) performing the Work;

(ii) paying interest and principal on other existing Project Debt and any costs and fees in connection with the development and award of the Project, achieving commercial close and closing and administering of any permitted Project Debt;

(iii) paying reasonable development fees to PPC-Related Entities or to a D&C Contractor or its affiliates for services related to the Project;

(iv) paying fees and premiums to any Lender of the Project Debt or such Lender's agents;

(v) paying Taxes;

(vi) funding reserves required under this Agreement, Financing Documents, applicable securities laws, or Environmental Laws;

- (vii) making Distributions; and
- (viii) Refinancing any Project Debt in accordance with this Agreement.

(b) The Security Documents may only secure Project Debt and Financing Documents issued and executed by (a) Principal Project Company, (b) its permitted successors and permitted assigns, (c) a special purpose entity that owns Principal Project Company but no other assets and has purposes and powers limited to the Project and the Work or (d) any special purpose entity or subsidiary wholly owned by Principal Project Company or such entity.

(c) Project Debt under a Financing Agreement and secured by a Security Document must be issued and held only by Institutional Lenders who qualify as such at the date the Security Document is executed and delivered (or, if later, at the date any such Institutional Lender becomes a party to the Security Document), except that (i) qualified investors, including PABs holders that may purchase the PABs in a public offering, other than Institutional Lenders may acquire and hold interests in Project Debt, but only if an Institutional Lender acts as Collateral Agent for such Project Debt and (ii) Equity Member Debt is not subject to this provision.

(d) The Security Documents, as a whole, securing each separate issuance of debt shall encumber the entire PPC's Interest, provided that the foregoing does not preclude entry by Principal Project Company into subordinate Security Documents (such subordination to be in accordance with the terms set forth in the Financing Documents) or into equipment lease financing.

(e) No Security Document or other instrument purporting to mortgage, pledge, encumber, or create a lien, charge or security interest on or against PPC's Interest shall extend to or affect the

right, title and interest of the City in the Project or the Facility or City's rights or interests under this Agreement.

(f) No Security Document or other instrument purporting to mortgage, pledge, encumber, or create a lien, charge or security interest on or against PPC's Interest shall extend to or affect City's right to access Principal Project Company's funds held in an operating account in the event of a PPC Default and the exercise of remedies by City under <u>Section 16.2</u> (City Remedies for PPC Default) of the Agreement; provided such right shall not include a security interest in such funds nor shall the exercise of such right by City interfere with the right of the Lenders, if any, under the Security Documents and the Direct Agreement to access such funds. Specifically, with respect to any such operating account (and associated account agreement and account control agreement):

(i) City shall be named as an authorized representative of Principal Project Company for purposes of the account, including specifically under any associated account control agreement, in the event of City's exercise of its Step-In Rights, and

(ii) Any account control agreement shall expressly state the terms, prohibitions, City rights and other provisions in this <u>Exhibit 5G</u> (Financing Document Terms), including specifically those set forth in <u>Sections (g)</u> through (m) of this <u>Exhibit 5G</u> (Financing Document Terms).

(g) Each note, bond or other negotiable or non-negotiable instrument evidencing Project Debt, or evidencing any other obligations issued or incurred by any Person described in this <u>Exhibit 5G</u> (Financing Document Terms) in connection with this Agreement or the Project shall include or refer to a document controlling or relating to the foregoing that includes a conspicuous recital on its face to the effect that payment of the principal thereof and interest thereon is a valid claim only as against the obligor and the security pledged by Principal Project Company or the obligor therefor, is not an obligation, moral or otherwise, of the City, any other agency, instrumentality or political subdivision of the City, or any elected official, member, director, officer, employee, agent or representative of any of them, and neither the full faith and credit nor the taxing power, and no assets, of the City, or any other agency, instrumentality or political subdivision of the principal thereof and interest thereon.

(h) Each Financing Document containing provisions regarding default by Principal Project Company shall require that if Principal Project Company is in default thereunder and the Collateral Agent gives notice of such default to Principal Project Company, then the Collateral Agent shall also give concurrent notice of such default to City. Each Financing Document and Security Document that provides Lender remedies for default by Principal Project Company or the borrower shall require that the Collateral Agent deliver to City, concurrently with delivery to Principal Project Company or any other Person, every notice of election or enforcement of remedies, including an election to sell or foreclose, notice of sale or foreclosure or other notice required by Applicable Law or by the Security Document in connection with the exercise of remedies under the Financing Document or Security Document.

(i) No Financing Document that may be in effect during any part of the period that the Handback Requirements apply shall grant to the Lender any right to apply funds in the Handback Requirements Reserve Account or to apply proceeds from any Handback Requirements Letter of Credit in accordance with <u>Section 8.7.5</u> (Handback Requirements Letters of Credit) of the Agreement to the repayment of Project Debt, to any other obligation owing the Lender or to any

other use except the uses set forth in <u>Section 8.6</u> (Handback) of the Agreement, and any provision purporting to grant such right shall be null and void, <u>provided</u>, <u>however</u>, that (a) any Lender or Substituted Entity shall, following foreclosure or transfer in lieu of foreclosure, automatically succeed to all rights, claims and interests of Principal Project Company in and to the Handback Requirements Reserve Account and (b) a Financing Document or Security Document may create such rights regarding excess funds described in <u>Section 8.7.2.5</u> of the Agreement.

Each relevant Financing Document and Security Document that may be in effect during any part of the period that the Handback Requirements apply shall expressly permit, without condition or qualification, Principal Project Company to (a) use and apply funds in the Handback Requirements Reserve Account in the manner contemplated by this Agreement, (b) issue additional Project Debt, secured by the PPC's Interest, for the added limited purposes of funding work pursuant to Handback Requirements in accordance with <u>Section 8.6</u> (Handback) of the Agreement and (c) otherwise comply with its obligations in this Agreement regarding Handback Renewal Work, the Handback Requirements and the Handback Requirements Reserve Account. Subject to the foregoing, any protocols, procedures, limitations and conditions concerning draws from the Handback Requirements Reserve Account set forth in any Financing Document or the issuance of additional Project Debt as described in clause (b) above shall be consistent with the permitted uses of the Handback Requirements Reserve Account, and shall not constrain Principal Project Company's or City's access thereto for such permitted uses, even during the pendency of a default under the Financing Document.

For the avoidance of doubt:

(i) The Lenders then holding Project Debt may limit additional Project Debt if other funds are then, in City's determination, readily available to Principal Project Company for the purpose of funding the Work;

(ii) No Lender then holding Project Debt is required hereby to grant pari passu lien or payment status to any such additional Project Debt; and

(iii) The Lenders then holding Project Debt may impose reasonable and customary requirements as to performance and supervision of the Work that are no more onerous than those set forth in their respective existing Financing Documents or Security Documents.

(j) Each Financing Document and Security Document shall expressly state that the Lender shall not name or join City, any other agency, instrumentality or political subdivision of the City, or any elected official, member, director, officer, employee, agent or representative of any of them in any legal proceeding seeking collection of the Project Debt or other obligations secured thereby or the foreclosure or other enforcement of the Financing Documents, except as such language may be modified in the Financing Documents approved by City as provided in any Direct Agreement.

(k) Each Financing Document shall expressly state that the Lender shall not seek any damages or other amounts from City, any other agency, instrumentality or political subdivision of the City, or any elected official, member, director, officer, employee, agent or representative of any of them, whether for Project Debt or any other amount, except (a) damages from City only for a violation by City of its express obligations to Lenders set forth in any Direct Agreement, if applicable, and (b) amounts due from City under this Agreement where the Lender has succeeded

to the rights and interests of Principal Project Company under this Agreement, whether by way of assignment or subrogation.

(I) Each Financing Document shall expressly state that the Lender and the Collateral Agent shall respond to any request from City or Principal Project Company for consent to a modification or amendment of any of this Agreement within a reasonable period of time.

(m) Each Financing Document that addresses the use of insurance proceeds shall expressly state that the Lender and the Collateral Agent shall (a) apply proceeds from Insurance Policies as specified in Exhibit 7 (Insurance Requirements) and (b) remit any amounts due to City from proceeds from Insurance Policies, as set forth in Exhibit 7 (Insurance Requirements), in each case, if any such Lender or Collateral Agent is named as the loss payee for the Insurance Policy.

# EXHIBIT 6

#### FORMS OF PAYMENT AND PERFORMANCE SECURITY

- Exhibit 6A: Financial Close Security
- Exhibit 6B: Form of D&C Payment Bond<sup>1</sup>
- Exhibit 6C: Form of D&C Performance Bond<sup>1</sup>
- Exhibit 6D: Form of Multiple Obligee Rider for D&C Payment Bond<sup>1</sup>
- Exhibit 6E: Form of Multiple Obligee Rider for D&C Performance Bond<sup>1</sup>

#### Exhibit 6F: Form of SFMTA O&M Facilities Warranty Bond

<sup>1</sup> If the bond is to secure the performance or payment obligations of the D&C Contractor or other Key Contractor, rather than Principal Project Company, then the form of the bond shall be revised in accordance with <u>Section 10.3.1.8</u> of this Agreement.

## EXHIBIT 6A

#### FORM OF FINANCIAL CLOSE SECURITY

## FORM 6A-1 - FORM OF FINANCIAL CLOSE BOND<sup>1</sup>

Bond No. \_\_\_\_\_

KNOW ALL PERSONS BY THESE PRESENTS, that the \_\_\_\_\_\_, as Principal and \_\_\_\_\_\_, as Surety, each authorized and licensed to transact business in the State of California ("State"), and each a corporation duly organized under the laws of the State indicated on the attached page, having its principal place of business at the address listed on the attached page, in the State indicated on the attached page, and each Surety authorized as a surety in the State, are jointly and severally held and firmly bound unto the City and County of San Francisco (the "City"), a municipal corporation acting by and through the San Francisco Municipal Transportation Authority ("SFMTA"), in the sum of [\_\_\_\_\_\_ United States Dollars (US \$\_\_\_\_\_)] (the "Bonded Sum"), the payment of which we each bind ourselves, and our heirs, executors, administrators, representatives, successors, and assigns, jointly and severally, firmly by these presents.

**WHEREAS**, the Principal has entered into an Infrastructure Facility Design-Build-Finance-Operate-Maintain Agreement with the City dated as of \_\_\_\_\_ 2025 (the "**Agreement**") to design, construct, finance, operate and maintain the Infrastructure Facility of the Potrero Yard Modernization Project (the "**Project**");

**NOW, THEREFORE**, the condition of this bond is such that this obligation shall be null and void upon (a) occurrence of Financial Close for the Project by the Financial Close Deadline set forth in the Agreement; or (b) Principal's receipt of written Notice from City that the Agreement is terminated pursuant to <u>Section 17.6</u> of the Agreement; otherwise it shall remain in full force and effect, and the Bonded Sum will be forfeited to City as liquidated damages and not as a penalty, upon receipt by Principal and Surety of Notice of such forfeiture from City:

The Surety hereby agrees to pay to City the full Bonded Sum hereinabove set forth, as liquidated damages and not as a penalty, within ten days after Notice from City that Principal has failed to achieve Financial Close by the Financial Close Deadline set forth in the Agreement, and such failure is not excused in accordance with the Agreement.

<sup>&</sup>lt;sup>1</sup> The amount of a single bond may be less than \$[\_\_\_\_\_] on the condition that Proposer provides Financial Close Security that totals \$[\_\_\_\_\_] in the aggregate.

The following terms and conditions shall apply with respect to this bond:

1. If suit is brought on this bond by City and judgment is recovered, Surety shall pay all costs incurred by City in bringing such suit, including, without limitation, reasonable attorneys' fees and costs as determined by the court.

2. Any extension(s) of the time for award of the Agreement that Principal may grant in accordance with the Agreement or otherwise, shall be subject to the reasonable approval of Surety.

3. Correspondence, Notices or claims relating to this bond should be sent to Surety at the following address:

\_\_\_\_\_

This agreement shall be binding on the Principal and Surety executing the same, their legal representatives, successors and assigns.

Capitalized terms used but not defined herein shall have the meaning given to them in the Agreement.

SIGNED and SEALED this	day of	, 2025
Principal		
Ву:		
Surety		
By: Attorney in Fact		

[add appropriate Surety acknowledgments]

#### FORM 6A-2 - FORM OF FINANCIAL CLOSE LETTER OF CREDIT<sup>b</sup>

TTER OF CREDIT - FINANCIAL CLOSE
T: (Name and Address of Bank/Branch <u>MUST</u> be San Francisco, California, Los Angeles, California, Chicago, Illinois, or New York, New York address)
Y OF SAN FRANCISCO
llars (US \$[])]

The above named Issuer issues this Irrevocable Standby Letter of Credit in favor of the City and County of San Francisco ("**City**"), for any sum or sums up to the aggregate amount of [[\_\_\_\_] **United States Dollars (US \$**[\_\_\_\_])], available by draft at sight drawn on the Issuer. Any draft under this Irrevocable Standby Letter of Credit shall be signed by an authorized representative of the above named Beneficiary and shall:

- 1. Identify this Irrevocable Standby Letter of Credit by the name of the Issuer, and the Letter of Credit number, amount, and place and date of issue, stated above; and
- 2. State:

"This drawing is due to <u>(Applicant's name)</u>'s failure to achieve Financial Close by the Financial Close Deadline set forth in the Infrastructure Facility Design-Build-Finance-Operate-Maintain Agreement with the City and County of San Francisco dated as of <u>2025</u> (the "**Agreement**") without excuse under the Agreement."

All drafts presented in compliance with the terms of this Irrevocable Standby Letter of Credit will be honored if presented by physical delivery of such original documents or via overnight courier

<sup>&</sup>lt;sup>b</sup> The amount of a single letter of credit may be less, on the condition that Proposer provides Financial Close Security that totals \$[\_\_\_\_] in the aggregate.

of such original documents to <u>(Name & San Francisco, California, Los Angeles, California, Chicago, Illinois, or New York, New York address)</u> on or before the stated expiration date described above or any extended expiration date.

This Irrevocable Standby Letter of Credit is subject to the rules of the "International Standby Practices" ISP98. If a conflict between ISP98 and New York law should arise, New York law shall prevail, without regard to principles of conflicts of law. If legal proceedings are initiated by any party with respect to payment of the Irrevocable Standby Letter of Credit, we agree that such proceeding shall be governed by New York law and may be brought in the State Courts of New York, venue in New York City, provided that the obligations of City shall be governed and construed in accordance with California law. Venue for any proceeding involving issues related to the obligations of City shall be in San Francisco, California.

Any failure by you to draw upon this Irrevocable Standby Letter of Credit as permitted hereunder shall not cause this Irrevocable Standby Letter of Credit to be unavailable for any future drawing, provided that this Irrevocable Standby Letter of Credit has not expired prior to such future drawing and that all requirements of this Irrevocable Standby Letter of Credit are independently satisfied with respect to any such future drawing.

Capitalized terms used but not defined herein shall have the meaning given to them in the Agreement.

Issuer:

By: \_\_\_\_\_(Authorized signature of Issuer)

## EXHIBIT 6B

#### FORM OF D&C PAYMENT BOND

(Bond No. )

**[INSERT PRINCIPAL PROJECT COMPANY NAME]** (the "**Principal Project Company**") has entered into a contract with the City and County of San Francisco (the "**City**"), a municipal corporation acting by and through the San Francisco Municipal Transportation Agency ("**SFMTA**"), bearing the date of **[INSERT DATE]**, 2025, to design, build, finance, operate and maintain the Infrastructure Facility of the Potrero Yard Modernization Project (the "**Agreement**"); and

That we **[INSERT CONTRACTOR NAME]**, an entity duly authorized to do business in the State of California (the "**State**") and having its principal place of business at **[INSERT STREET ADDRESS, CITY, STATE, ZIP AND PHONE #]** (the "**Principal**" or "**Contractor**"), have entered into a contract (the "**Contract**") with Principal Project Company bearing the date of **[INSERT DATE]**, for the performance of design and construction work in connection with the Infrastructure Facility of the Potrero Yard Modernization Project; and

That the Principal and **[INSERT SURETY NAME]** (the "**Surety**") (or the "**Co-Sureties**"), duly authorized to do business in the State, having its principal place of business at **[INSERT HOME OFFICE ADDRESS]** are held and firmly bound unto Principal Project Company, in the full and just sum of **[INSERT D&C PAYMENT BOND AMOUNT]** lawful money of the United States of America **(US\$\_\_\_)**, to whom payment well and truly will be made, we bind ourselves, our heirs, executors, administrators, successors and assigns, jointly and severally;

CONDITIONS OF THIS BOND ARE AS FOLLOWS:

- A. Principal shall promptly make all payments owing when due to all persons who furnish labor, services, or materials for the prosecution of the work provided for in the Contract.
- B. Each said claimant shall have a right of action against the Principal and Surety (or Co-Sureties) for the amount due him or her, including unpaid finance charges due under the claimant's contract.
- C. A claimant, except a laborer, who is not in privity with the Principal shall, before commencing or not later than 20 days after commencing to furnish labor, materials, or supplies for the prosecution of the work, furnish the Principal with a preliminary Notice that he or she intends to look to this bond for protection. No action for the labor, materials, or supplies may be instituted against the Principal or the Surety (or Co-Sureties) unless the preliminary Notice has been given. Notices required or permitted shall be given in compliance with the requirements of Sections 8100 et seq. and 9300 et seq. of the California Civil Code.
- D. The amount of this bond is a fixed amount and shall not be changed.

E. Neither any change in or under the Contract, nor any compliance or noncompliance with any formalities provided in the Contract nor the change shall relieve the Surety (or Co-Sureties) of its obligations under this bond.

Correspondence, Notices or claims relating to this bond should be sent to the following Surety's authorized representative and address:

\_\_\_\_\_

If there are Co-Sureties, the Co-Sureties agree to empower the above authorized representative with the authority to act on behalf of all of the Co-Sureties with respect to this bond, so that City will have no obligation to deal with multiple sureties hereunder. The above authorized representative may be changed only by delivery of written Notice (by personal delivery or by certified mail, return receipt requested) to City designating a single new authorized representative, signed by all of the Co-Sureties.

IN WITNESS WHEREOF, Principal and Surety have executed these presents and the Surety (or Co-Sureties) has affixed its seal, this \_\_\_\_\_ day of \_\_\_\_\_, 2025.

PRINCIPAL:		Date:	, 2025
Authorized Signature:			
Print Name & Title:			
NAME OF SURETY:			
Date:	, 2025		
Signature (Attorney-in-Fact):			
(Affix Seal)			
Print Full Name, Address and	d Telephone No.		

## [If more than one Surety, then add appropriate number of lines to signature block]

## NOTES CONCERNING SURETY AND EXECUTION:

A. <u>SURETY COMPANY REQUIREMENTS</u>

To be acceptable to the City, each Surety shall meet all of the requirements of the Agreement and laws of California.

#### B. <u>EXECUTION OF BOND</u>

- 1. Enter each Surety's name and address on each copy of this bond in the space provided.
- 2. Have each copy of this bond signed on behalf of the Principal by the same person that signed the Contract on behalf of the Principal (affix Corporate Seal, if appropriate).
- 3. Have each copy of this bond signed by the person authorized to sign on behalf of the Surety (or Co-Sureties). Put the date the signature was affixed in the space provided. Print that person's name in the place provided on each copy of this bond. Also, have each Surety's Corporate Seal affixed to each copy of this bond beside that person's signature (no copies are acceptable).
- 4. Each copy of this bond must have a Power of Attorney attached indicating that the person in item B.3 above is authorized to sign on behalf of the relevant Surety.
- 5. Each copy of the Power of Attorney must have the Surety's Corporate Seal manually affixed.
- 6. The date of execution of the Power of Attorney is the same as the date shown on the signature line for the Surety Attorney-In-Fact.

#### CALIFORNIA ALL PURPOSE ACKNOWLEDGEMENT

A notary public or other officer completing this certificate verifies only the identity of the individual who signed the document to which this certificate is attached, and not the truthfulness, accuracy or validity of that document.

State of \_\_\_\_\_

County of \_\_\_\_\_

On this \_\_\_\_ day of \_\_\_\_\_ in the year of \_\_\_\_\_ before me, a notary public in and for the county and state aforesaid, personally appeared \_\_\_\_\_\_

\_\_\_\_\_\_who proved to me on the basis of satisfactory evidence to be the person(s) whose name(s) is/are subscribed to within the instrument and acknowledged to me that he/she executed the same in his/her authorized capacity(ies), and that by his/her signature(s) on the instrument, the person(s) or the entity upon behalf of which the person(s) acted, executed the instrument.

Witness my hand and official seal:

\_\_\_\_\_(SEAL)\_\_\_\_\_

Signature of Notary Public

## EXHIBIT 6C

## FORM OF D&C PERFORMANCE BOND

(Bond No. )

**[INSERT PRINCIPAL PROJECT COMPANY NAME]** (the "**Principal Project Company**") has entered into a contract with the City and County of San Francisco (the "**City**"), a municipal corporation acting by and through the San Francisco Municipal Transportation Agency ("**SFMTA**"), bearing the date of **[INSERT DATE]**, 2025, to design, build, finance, operate and maintain the Infrastructure Facility of the Potrero Yard Modernization Project (the "**Agreement**"); and

That we **[INSERT CONTRACTOR NAME**], an entity duly authorized to do business in the State of California (the "**State**") and having its principal place of business at **[INSERT STREET ADDRESS, CITY, STATE, ZIP AND PHONE #)**] (the "**Principal**" or "**Contractor**"), have entered into a contract (the "**Contract**") with Principal Project Company bearing the date of **[INSERT DATE]** related to the performance of design and construction work for the Infrastructure Facility of the Potrero Yard Modernization Project; and

That the Principal and **[INSERT SURETY NAME]** (the "**Surety**") (or the "**Co-Sureties**"), duly authorized to do business in the State, having its principal place of business at **[INSERT HOME OFFICE ADDRESS]** are held and firmly bound unto Principal Project Company and the Indemnitees (as defined in the Agreement) (Principal Project Company and the Indemnitees are collectively referred to herein as the "**Obligees**") in the full and just sum of **[INSERT D&C PERFORMANCE BOND AMOUNT]** lawful money of the United States of America **(US\$\_\_\_\_)**, to whom payment well and truly will be made, we bind ourselves, our heirs, executors, administrators, successors and assigns, jointly and severally and firmly by these presents;

**WHEREAS**, it was one of the conditions of the Contract and the Agreement that these presents shall be executed;

**NOW, THEREFORE,** the conditions of this obligation are such that if Principal shall faithfully, promptly, efficiently and fully perform in accordance with the obligations of the Contract and shall indemnify and save harmless the Obligees from all cost and damage by reason of Principal's failure to do so, then this obligation shall be void; otherwise, it shall remain in full force and effect.

The Surety's obligations under this bond shall arise after:

(a) Either: (i) Principal Project Company provides Notice to Principal and the Surety that Principal Project Company is considering declaring a default by Principal under the Contract; or (ii) the City provides Notice to the Principal Project Company, with a copy to the Surety, that it is considering declaring a default of Principal Project Company under the Agreement as the result of a material breach by Principal of its obligations to Principal Project Company under the Contract. The relevant Notice shall indicate whether the Principal Project Company or the City, as the case may be, is requesting a conference among Principal Project Company, Principal, City and the Surety to discuss Principal's performance. If a conference is not requested, the Surety may, within five (5) business days after receipt of the relevant Notice, request such a conference. If the Surety timely requests a conference, Principal Project Company and City shall attend. Unless Principal Project Company and City agrees otherwise, any such conference shall be held within ten (10) business days of the Surety's receipt of the relevant Notice; and

(b) Principal Project Company declares a default by Principal under the Contract, terminates the Contract and provides Notice to the Surety.

Failure on the part of Principal Project Company or the City to comply with the Notice requirement in <u>Section (a)</u> shall not constitute a failure to comply with a condition precedent to the Surety's obligations, or release the Surety from its obligations.

Should the conditions set forth in (a)-(b) above be satisfied, the Surety (or Co-Sureties) shall pay Principal Project Company (or any one or more of the Obligees) all costs assessed against the Principal because of the default which were not withheld from Contract proceeds, and upon an Obligee's demand, the Surety (or Co-Sureties) shall (i) take over performance of the Contract obligations; provided, however, that in the event Principal Project Company (or any one or more of the Obligees) elects to have the Surety (or Co-Sureties) take over performance of the Contract obligations, the Surety (or Co-Sureties) may not select the Principal or any affiliate of the Principal to perform the Contract obligations for and on behalf of the Surety (or Co-Sureties) without the express written consent of the Obligees; or (ii) waive the right to take over performance of the Contract obligations and tender payment of an amount necessary to compensate the Obligees for the cost to perform the Principal's obligations under the Contract, up to the full penal sum of this bond.

The Surety (or Co-Sureties) shall be fully liable under this bond up to the full penal sum hereof, regardless of any modifications (of whatever amount) to the Contract amount, provided that the Surety's obligations under this bond shall not be greater than those of Principal under the Contract.

No alteration, modification or supplement to the Contract or the nature of the work to be performed thereunder, including without limitation any extension of time for performance, shall in any way affect the obligations of Surety (or Co-Sureties) under this bond. Surety hereby waives Notice of any such alteration, modification or supplement, including changes in time, to the Contract.

Correspondence, Notices or claims relating to this bond should be sent to the following Surety's authorized representative and address:

\_\_\_\_\_

If there are Co-Sureties, the Co-Sureties agree to empower the above authorized representative with the authority to act on behalf of all of the Co-Sureties with respect to this bond, so that City will have no obligation to deal with multiple sureties hereunder. The above authorized representative may be changed only by delivery of written Notice (by personal delivery or by certified mail, return receipt requested) to City designating a single new authorized representative, signed by all of the Co-Sureties.

If any provision of this bond is found to be unenforceable as a matter of law, all other provisions shall remain in full force and effect.

This bond shall be governed by and construed in accordance with the laws of the State of California, without regard for conflict of laws principles, and any action seeking enforcement of the bond will be litigated exclusively in the courts of the State of California.

WITNESS the signature of the Principal and the signature of the Surety (or Co-Sureties) by \_\_\_\_\_\_ its \_\_\_\_\_ (Agent or Attorney-in-Fact) with the seals of said Principal and Surety (or Co-Sureties) hereunto affixed this \_\_\_\_\_ day of \_\_\_\_\_, 2025.

Complete the following as appropriate
Principal (Entity Name)*
Authorized Signature:
*Signature:
Printed Name:
*
Title:
(Seal)
*Include the signature and printed name of each partner required to be affixed per partnership agreement.

Principal shall record this bond in the official records of the Clerk of Court of the county where the improvement is located prior to commencing the work.

Organized and existing under the laws of the State of _	and authorized to do business in the	
State of California, pursuant to the laws of the State of	California.	
Countersigned: California Licensed Insurance Agent Print information below (California Licensed Insurance Agent; whether in Attorney-in-Fact or Countersignature role): Name: Business Address:	Surety Company Name (Print) (Seal) By:	
	California Licensed Insurance Agent or Attorney-in-Fact (Surety)	
	Above Signatory is also a California Licensed Insurance Agent (check if applicable and complete business name, address and telephone number block;	
Telephone:	if not, have such an agent countersign and complete block). NOTE: Power of Attorney showing authority of Surety's Agent or Attorney-in-Fact is to be attached.	

[If more than one Surety, then add appropriate number of lines to signature block]

Send "Notices to City" to:

#### CALIFORNIA ALL PURPOSE ACKNOWLEDGEMENT

A notary public or other officer completing this certificate verifies only the identity of the individual who signed the document to which this certificate is attached, and not the truthfulness, accuracy or validity of that document.

State of \_\_\_\_\_

County of \_\_\_\_\_

On this \_\_\_\_ day of \_\_\_\_\_ in the year of 2025 before me, a notary public in and for the county and state aforesaid, personally appeared \_\_\_\_\_\_

who proved to me on the basis of satisfactory evidence to be the person(s) whose name(s) is/are subscribed to within the instrument and acknowledged to me that he/she executed the same in his/her authorized capacity(ies), and that by his/her signature(s) on the instrument, the person(s) or the entity upon behalf of which the person(s) acted, executed the instrument.

Witness my hand and official seal:

\_\_\_\_\_(SEAL)\_\_\_\_\_

Signature of Notary Public

## EXHIBIT 6D

## FORM OF MULTIPLE OBLIGEE RIDER FOR D&C PAYMENT BOND

#### MULTIPLE OBLIGEE RIDER

This Rider is executed concurrently with and shall be attached to and form a part of Payment Bond \_\_\_\_\_\_(hereinafter referred to as the "**Payment Bond**").

WHEREAS, **[INSERT PRINCIPAL PROJECT COMPANY NAME]** (the "**Principal Project Company**") has entered into a contract with the City and County of San Francisco (the "**City**"), a municipal corporation acting by and through the San Francisco Municipal Transportation Agency ("**SFMTA**"), bearing the date of **[INSERT DATE]**, 2025, to design, build, finance, operate and maintain the Infrastructure Facility of the Potrero Yard Modernization Project (the "**Agreement**"); and

WHEREAS, on or about the	day of	, 2025_,	
[INSERT (	CONTRACTOR	NAME], (hereinafter ca	alled the
"Principal"), entered into a written agr	reement bearing	the date of	, 2025
(hereinafter called the "Contract") with	h Principal Proje	ct Company, (hereinaf	ter called the
"Primary Obligee") for the performance	ce of design and	construction work for	the Infrastructure
Facility of the Potrero Yard Moderniza	ition Project; and	1	

WHEREAS, the Primary Obligee requires that Principal provide a payment bond and that the City and County of San Francisco, a municipal corporation acting by and through the San Francisco Municipal Transportation Agency, the other Indemnitees (as defined in the Agreement) [and \_\_\_\_\_ [INSERT COLLATERAL AGENT, if appropriate]] ("Additional Obligee(s)") be named as additional obligee(s) under the payment bond; and

WHEREAS, Principal and the Surety (or Co-Sureties) identified below have agreed to execute and deliver this Rider concurrently with the issuance of the Payment Bond, upon the conditions herein stated.

NOW, THEREFORE, the undersigned hereby agree and stipulate as follows:

1. The Additional Obligee(s) is/are hereby added to the Payment Bond as named obligee(s).

2. The aggregate liability of the Surety (or Co-Sureties) to the Primary Obligee and the Additional Obligee(s) is limited to the penal sum of the Payment Bond.

3. The Additional Obligee(s)'s rights under the Payment Bond are subject to the same defenses that the Principal and/or the Surety (or Co-Sureties) have against the Primary Obligee.

# Potrero Yard Modernization Project Draft November 8, 2024

Signed, sealed and dated this _	day of	, <u>2025</u> .
Attest:	(SEAL)	(Name of Principal
Secretary		By: Title:
Attest:	(SEAL)	(Surety / Co-Surety)
Signature Bonding Agent's Name:		By: Title:
Agent's Address:		(Business Address of Surety / Co-Surety)
Attest:	(SEAL)	(Co-Surety)
Signature Bonding Agent's Name:		By: Title:
Agent's Address:		(Business Address of Co-Surety)
Attest:	(SEAL)	(Co-Surety)
Signature Bonding Agent's Name:		By: Title:
Agent's Address:		(Business Address of Co-Surety)

Attest: (SEAL)	(Co-Surety)
Signature Bonding Agent's Name:	By: Title:
Agent's Address:	(Business Address of Co-Surety)
Approved as to legal form and sufficiency this	
day of 2025	
Deputy City Attorney	

[Note: Add lines to signature block if needed, and strike signature lines not used.]

[Note: The bond shall be signed by authorized persons. Where such persons are signing in a representative capacity (e.g., an attorney-in-fact), but is not a member of the firm, partnership, or joint venture, or an officer of the legal entity involved, evidence of authority must be furnished.]

# EXHIBIT 6E

## FORM OF MULTIPLE OBLIGEE RIDER FOR D&C PERFORMANCE BOND

#### MULTIPLE OBLIGEE RIDER

This Rider is executed concurrently with and shall be attached to and form a part of Performance Bond \_\_\_\_\_\_(hereinafter referred to as the "**Performance Bond**").

WHEREAS, **[INSERT PRINCIPAL PROJECT COMPANY NAME]** (the "**Principal Project Company**") has entered into a contract with the City and County of San Francisco (the "**City**"), a municipal corporation acting by and through the San Francisco Municipal Transportation Agency ("**SFMTA**"), bearing the date of **[INSERT DATE]**, 2025, to design, build, finance, operate and maintain the Infrastructure Facility of the Potrero Yard Modernization Project (the "**Agreement**"); and

WHEREAS, on or about the	day of	, 2025,	
[INSERT (	CONTRACTOR	NAME], (hereinafter c	alled the
"Principal"), entered into a written agr	reement bearing	the date of	, 2025
(hereinafter called the "Contract") with	h Principal Proje	ect Company, (hereinaf	iter called the
"Primary Obligee") for the performan	ce of design and	construction work for	the Infrastructure
Facility of the Potrero Yard Moderniza	ation Project; and	k	

WHEREAS, the Primary Obligee requires that Principal provide a performance bond and that the City and County of San Francisco, a municipal corporation acting by and through the San Francisco Municipal Transportation Agency, the Indemnitees (as defined in the Agreement) [and \_\_\_\_\_ [INSERT COLLATERAL AGENT, if appropriate]] ("Additional Obligee(s)") be named as additional obligee(s) under the performance bond; and

WHEREAS, Principal and the Surety (or Co-Sureties) identified below have agreed to execute and deliver this Rider concurrently with the issuance of the Performance Bond, upon the conditions herein stated.

NOW, THEREFORE, the undersigned hereby agree and stipulate as follows:

1. The Additional Obligee(s) is/are hereby added to the Performance Bond as named obligee(s).

2. The Surety (or Co-Sureties) shall not be liable under the Performance Bond to the Primary Obligee, the Additional Obligee(s), or any of them, unless the Primary Obligee, the Additional Obligee(s), or any of them, shall make payments to the Principal (or in the case the Surety (or Co-Sureties) arrange for completion of the Contract, to the Surety (or Co-Sureties)) in accordance with the terms of the Contract as to payments and shall perform all other obligations to be performed under the Contract in all material respects at the time and in the manner therein set forth such that no material breach by the Primary Obligee shall have occurred and be continuing under the Contract.

3. The aggregate liability of the Surety (or Co-Sureties) under the Performance Bond, to any or all of the obligees, as their interests may appear, is limited to the penal sum of the Performance Bond. The Additional Obligee(s)'s rights hereunder are subject to the same defenses Principal and/or Surety (or Co-Sureties) have against the Primary Obligee. The total liability of the Surety (or Co-Sureties) under the Contract shall in no event exceed the amount recoverable from the Principal by the Primary Obligee.

4. The Surety (or Co-Sureties) may, at their option, make any payments under the Performance Bond by check issued jointly to all of the obligees.

5. In the event of a conflict between the Performance Bond and this Rider, this Rider shall govern and control. All references to the Performance Bond, either in the Performance Bond or in this Rider, shall include and refer to the Performance Bond as supplemented and amended by this Rider. Except as herein modified, the Performance Bond shall be and remains in full force and effect.

Signed, sealed and dated this _	day of	. <u>2025</u> .
Attest:	(SEAL)	(Name of Principal
Secretary		By: Title:
Attest:	(SEAL)	(Surety / Co-Surety)
Signature Bonding Agent's Name: Agent's Address:		By: Title: (Business Address of Surety / Co-Surety)
Attest:	(SEAL)	(Co-Surety)
Signature Bonding Agent's Name: Agent's Address:		By: Title: (Business Address of Co-Surety)

Attest: (SEAL)	(Co-Surety)
Signature Bonding Agent's Name: Agent's Address:	By: Title: (Business Address of Co-Surety)
Attest: (SEAL)	(Co-Surety)
Signature Bonding Agent's Name: Agent's Address:	By: Title: (Business Address of Co-Surety)
Approved as to legal form and sufficiency this day of 2025	

[Note: Add lines to signature block if needed, and strike signature lines not used.]

[Note: The bond shall be signed by authorized persons. Where such persons are signing in a representative capacity (e.g., an attorney-in-fact), but is not a member of the firm, partnership, or joint venture, or an officer of the legal entity involved, evidence of authority must be furnished.]

# EXHIBIT 6F

## FORM OF SFMTA O&M FACILITIES WARRANTY BOND

Bond No. \_\_\_\_\_

For

#### POTRERO YARD MODERNIZATION PROJECT

#### KNOW ALL WHO SHALL SEE THESE PRESENTS:

**THAT WHEREAS,** the City and County of San Francisco ("City" or "Obligee"), a municipal corporation acting in its proprietary capacity by and through the San Francisco Municipal Transportation Agency, has awarded to [\_\_\_\_\_] (the "Principal Project Company" or "Principal"), an Infrastructure Facility Design-Build-Finance-Operate-Maintain Agreement dated as of [\_\_\_\_\_], 2025 (as amended from time to time, the "Agreement"), which Agreement is specifically incorporated by reference in this Bond, for the design, build, finance, operation and maintenance of the Potrero Yard Modernization Project (the "Project");

**AND WHEREAS**, initially capitalized terms not otherwise defined in this Bond have the meaning given in the Agreement;

**AND WHEREAS**, pursuant to <u>Section 10.9</u> of the Agreement, upon achieving Final Acceptance Principal Project Company must provide a warranty bond to City for the SFMTA O&M Facilities warranty Work (this "Bond");

**NOW THEREFORE**, We the undersigned Principal and \_\_\_\_\_\_ (the "Surety" or "Co-Sureties") are firmly bound and held unto the Obligee, in the penal sum of [\_\_\_\_\_] (\$[\_\_\_\_]) (the "Bonded Sum"), good and lawful money of the United States of America for the payment whereof, well and truly to be paid to the Obligee, we bind ourselves, our heirs, successors, executors, administrators, and assigns, jointly and severally, firmly by these presents.

**THE CONDITION OF THE FOREGOING OBLIGATION IS SUCH THAT**, if Principal shall promptly and faithfully perform all of its obligations under the Agreement, as they may be amended or supplemented, including, without limitation, the performance of all SFMTA O&M Facilities warranty Work and payment of claims as described in <u>paragraph 5</u> below, then the obligations under this Bond shall be null and void; otherwise this Bond shall remain in full force and effect, it being expressly understood and agreed that the liability of Surety for any and all claims hereunder shall in no event exceed the bonded sum.

The following terms and conditions shall apply with respect to this Bond:

**1.** The Agreement is incorporated by reference into this Bond.

2. If the Principal shall promptly and faithfully perform all of its obligations under the Agreement, as they may be amended or supplemented, related to the SFMTA O&M Facilities warranty Work, enforcement of Contractor warranties, and payment of claims as described in <u>paragraph 6</u> below, then the obligations under this Bond shall be null and void; otherwise this Bond shall remain in full force and effect, it being expressly understood and agreed that the liability of Surety (or Co-Sureties) for any and all claims hereunder shall in no event exceed the bonded sum.

3. If the above bound Principal, or its heirs, executors, administrators, successors or assigns, shall in all things stand to and abide by and well and truly keep and perform the covenants, conditions, obligations and agreements in the Agreement related to the SFMTA O&M Facilities warranty Work, including any and all amendments, supplements, and alterations made to the Agreement as therein provided, on the Principal's part to be kept and performed at the time and in the manner therein specified, and shall indemnify, defend and save harmless the Obligee and all other Indemnitees (as defined in the Agreement), as therein stipulated, then this obligation shall become and be null and void; otherwise, it shall be and remain in full force and virtue.

**4.** This Bond shall cover the cost to perform all the obligations of the Principal related to the SFMTA O&M Facilities warranty Work. The obligations covered by this Bond specifically include all payment obligations, liability for damages and warranties related to the SFMTA O&M Facilities warranty Work as specified in the Agreement, but not to exceed the bonded sum.

**5.** Whenever the Principal shall be, and is declared by the Obligee to be, in default under the Agreement with respect to the related to the SFMTA O&M Facilities warranty Work, the Surety (or Co-Sureties) shall promptly:

(a) remedy such default;

(b) complete the SFMTA O&M Facilities warranty Work and perform the obligations covered by this Bond in accordance with the terms and conditions of the Agreement then in effect; or

(c) select a contractor or contractors to complete the SFMTA O&M Facilities warranty Work and perform the obligations covered by this Bond in accordance with the terms and conditions of the Agreement then in effect, using a contractor or contractors approved by the Obligee in its sole discretion, arrange for a contract that contains substantially the same terms and conditions of the Agreement between such contractor or contractors and the Obligee, and make available as work progresses (even though there should be a default or a succession of defaults under such contract or contracts of completion arranged under this paragraph) sufficient funds to pay the cost of completion of the SFMTA O&M Facilities warranty Work and obligations covered by this Bond; but not exceeding, including other costs and damages for which the Surety (or Co-Sureties) is (are) liable hereunder, the bonded sum.

**6.** This Bond shall inure to the benefit of anyone required to be paid by law under the Agreement so as to give a right of action to such persons or their assigns in any suit brought upon this Bond. The obligations covered by this Bond specifically include:

(a) payments owing to any of the persons involved in prosecution of the SFMTA O&M Facilities warranty Work, as provided for in the Agreement;

(b) any amounts required to be deducted, withheld, and paid over to the Department of Revenue from the wages of employees of the Principal and its Contractors with respect to such work and labor, and

(d) any other payments owing to anyone required to be paid by law.

In case suit is brought to enforce the provisions of this paragraph 6, the Surety (or Co-Sureties) will pay reasonable attorneys' fees, to be fixed by the court.

7. The Surety (or Co-Sureties) agree(s) that no change, extension of time, alterations, additions, omissions or other modifications of the terms of the Agreement, or in the work to be performed with respect to the Project, or in the specifications or plans, or any change or modification of any terms of payment or extension of time for any payment pertaining or relating to the Agreement, or any rescission of this Bond, solely due to acts of Principal, or any fraud practiced by any other person other than the claimant seeking to recover this Bond, shall in any way affect its obligations on this Bond, and it does hereby waive notice of such changes, extension of time, alterations, additions, omissions or other modifications.

8. [*NTD: Use in case of multiple or co-sureties*] The Co-Sureties agree to empower a single representative with authority to act on behalf of all of the Co-Sureties with respect to this Bond, so that the Obligee and claimants will have no obligation to deal with multiple sureties hereunder. All correspondence from the Obligee or claimants to the Co-Sureties and all claims under this Bond shall be sent to such designated representative. The designated representative may be changed only by delivery of written notice (by personal delivery or by certified mail, return receipt requested) to the Obligee designating a single new representative, signed by all of the Co-Sureties. The initial representative shall be

	F, we have hereunto set our hands and seals on this atday of, 2025.
	Principal (full legal name):
	Address:
	B <sub>V</sub> c
	By: Contact Name:
	Phone: ( )
Surety (full legal name	ə):
Address:	
By:	
Contact Name:	
Phone:	
[Note: if more than one s	urety, then add appropriate phone: ()]

[Note: If more than one surety, then add appropriate number of lines to signature block.]

[Note: The bond shall be signed by authorized persons. Where such persons are signing in a representative capacity (e.g., an attorney-in-fact), but is not a member of the firm, partnership, or joint venture, or an officer of the legal entity involved, evidence of authority must be furnished.]

# <u>EXHIBIT 7</u>

## **INSURANCE REQUIREMENTS**

Principal Project Company shall obtain and keep in force, or cause to be obtained and kept in force, the following policies of insurance, in accordance with the terms of this <u>Exhibit 7</u>. Each policy shall be obtained and be effective as set forth below. Each policy shall contain, or be endorsed to contain, a provision that coverage cannot be canceled, voided, suspended, lapsed or modified or reduced in coverage except for 60 days' (or for non-payment of premium, 10 days') prior written notice has been given to City.

#### 1. Insurance during the Design and Construction Period

As a condition precedent to Financial Close and issuance of NTP 1, Principal Project Company shall obtain and keep in force, or cause to be obtained and kept in force, from and after the Financial Close Date and throughout the D&C Period, the following insurance coverage.

1.1. <u>Worker's Compensation and Employer's Liability</u>. Worker's compensation and employer's liability insurance for Principal Project Company and Contractors as required by applicable Law, and employer's liability insurance having coverage limits of \$1,000,000 for each accident, \$1,000,000 for disease (each employee), and \$1,000,000 for disease (policy limit). Policies shall contain a voluntary compensation endorsement; an alternative employer endorsement; and on an "if any" basis, endorsements providing coverage for all states, U.S. Longshore and Harbor Workers' Act, Jones Act and Federal Employer's Liabilities Act.

1.2. <u>Commercial General Liability.</u> A commercial general liability insurance policy, written on an occurrence basis and covering liabilities arising out of the construction of the Project. Coverage shall be at least as broad as the broadest available version of Insurance Services Office form CG 00 01. This insurance policy shall:

- have a limit for any one occurrence of not less than \$2,000,000 per occurrence, a \$4,000,000 general aggregate with one reinstatement during the D&C Period, and a \$4,000,000 completed operations aggregate, applicable solely to the construction of the Project;
- (b) have no "contractor's limitation" endorsements, as that term is defined, as of the date of this Agreement, in the Glossary of Insurance and Risk Management Terms published by the International Risk Management Institute, that have not been reviewed and approved by the City. There shall be no endorsement or modification of the CGL limiting the scope of coverage for liability assumed under an insured contract;
- (c) have no exclusion for professional services except the latest ISO form CG 22 79 or CG 22 80 or both;
- (d) include products and completed operations liability coverage for a period of not less than the California Statute of Repose which is currently 10 years following Final Completion or the Termination Date, whichever occurs later; and
- (e) be maintained throughout the Term until Final Completion.

The City, SFMTA, and San Francisco Board of Supervisors and their employees, officers, and officials shall be added to the primary policy as insureds using Insurance Services Office forms CG 20 10 10 01 and CG 20 37 10 01 or, in City's sole discretion, forms providing the same scope of coverage, including coverage for completed operations. The policy shall provide for separation of insureds. The policy shall contain no insured vs. insured exclusion. Principal Project Company shall require its Contractors to have these additional insureds included as insureds in the same manner as specified for Principal Project Company in this <u>Section 1.2</u> (Commercial General Liability), and shall prohibit use of any endorsement forms that require, to effect additional insured status, the execution or existence of any contract directly between such Contractors and any of the additional insureds. Exclusions are prohibited for work within 50 feet of a railroad. For use of any unmanned aircraft vehicles (UAV), any Principal Project Company-Related Entity may provide insurance either through an aircraft liability insurance policy, or by endorsement to the entity's commercial general liability insurance policy and excess liability policies, which shall be not less than \$10,000,000 per occurrence.

1.3. <u>Commercial Excess Liability.</u> A policy or policies of commercial umbrella/excess liability insurance covering bodily injury, property damage, personal injury and advertising injury with an annual reinstatement of limits of not less than \$100,000,000 per occurrence and general aggregate. There shall also be a project-specific products and completed operations aggregate of not less than \$100,000,000. Coverage may be arranged in any combination or structure so that total required limits of liability are met. Coverage must apply as excess over commercial general liability insurance as required in <u>Section 1.2</u> (Commercial General Liability), and may apply as excess over commercial automobile liability insurance and employer's liability insurance. Umbrella/excess liability policies shall follow form to all underlying policies, including coverage in the excess liability policies for insureds covered under the primary policies.

1.4. <u>Commercial Automobile Liability.</u> Commercial automobile liability insurance with limits of liability of not less than \$20,000,000 combined single limit for Principal Project Company and Key Contractors. The required limits can be met in any combination of primary and excess/umbrella liability policies. Coverage shall be at least as broad coverage provided in Insurance Services Office form CA 00 01. The insurance must cover "any auto" ("**Symbol 1**"). If Principal Project Company or any Contractor's activities involve transportation of materials (including Hazardous Materials) that require endorsement MCS 90 (as described below), the automobile liability Insurance Policy for Principal Project Company or such Contractor shall be endorsed to include for private, non-commercial vehicles Motor Carrier Act Endorsement-Hazardous Materials Clean up (MCS-90) and shall be endorsed to provide coverage for liability arising from release of pollutants (CA 99 48 – Pollution Liability – Broadened Coverage for Coverage Autos – Business Auto, Motor Carrier and Truckers Coverage Form).

1.5. <u>Contractor's Pollution Liability</u>. Contractor's pollution liability insurance written on an occurrence form with a limit for any one occurrence of not less than \$20,000,000 and a policy aggregate limit of \$20,000,000. The policy must be written on an occurrence basis and not on a "claims made" form. The policy must include coverage to extend to the full California statue of repose, which is currently 10 years.

The policy shall cover sums that the insured becomes legally obligated to pay to a third party or for the investigation, removal, remediation (including associated monitoring) or disposal of soil, surface water, groundwater or other contamination to the extent required by environmental laws (together "**clean-up costs**") caused by pollution conditions resulting from covered operations, subject to the policy terms and conditions, including bodily injury, property

damage (including natural resource damages), clean-up costs, and legal defense costs. Such policy shall cover claims related to pollution conditions to the extent such are caused (a) by the performance of Work, (b) by transportation, including loading and unloading, by owned and non-owned vehicles and/or (c) by other activities performed by or on behalf of Principal Project Company that occur on the Project. The policy shall have no exclusions or limitations for loss occurring over water including but not limited to a navigable waterway. Coverage shall apply to sudden and non-sudden pollution conditions resulting from the escape or release of smoke, vapors, fumes, acids, alkalis, toxic chemicals, liquids, or gases, waste materials, or other irritants, contaminants, or pollutants. The policy shall contain a severability provision.

1.6. <u>Professional Liability Insurance</u>. Professional liability insurance, in one or more policies, at Principal Project Company's discretion, which shall:

- (a) cover claims for liability for providing Design Work for the Project;
- (b) cover claims for liability for providing Design Work by any Contractor;
- (c) be in an amount not less than \$20,000,000 per claim and in the aggregate;
- (d) have a deductible or self-insured retention no greater than \$1,000,000 unless approved by the City;
- (e) cover the performance of Design Work or other professional services in connection with the Project (except Design Work that results in the provision of a product) and shall be fully retroactive to the first date any such Design Work was performed, with no exclusion for prior acts applying to any pre-award professional services provided by any insured; and
- (f) have an extended reporting period, or be renewed to be continuous for a period, of not less than ten years after the Effective Date.

1.7. <u>Builder's Risk</u>. A builder's risk insurance policy covering all real property at the Project Site, during testing and commissioning, while in transit and at any temporary off-site location; including all materials, supplies, machinery, fixtures and equipment intended to become a permanent part of the Project or for permanent use in the Project. Property that is incidental to the construction; foundations, including pilings, but excluding normal settling, shrinkage, or expansion. All temporary structures at the Project Site that are to be used in or incidental to the fabrication, erection, testing, or completion of the Project shall be insured and declared to and approved by the insurers and to the extent the cost thereof is included in the Work, included in the coverage while on or about the Project Site awaiting or during construction. The builder's risk policy:

- (a) shall be obtained prior to the start of construction and maintained until the end of the D&C Period;
- (b) shall be in an amount not less than the estimated completed value of the Project or other such amount as may be agreed up on by Principal Project Company and the City;

- (c) shall be written on an "all risk," replacement cost basis with no coinsurance clauses or penalties;
- (d) during any period of exposure to loss of property in transit, shall cover transit, including ocean marine (unless insured by the Supplier or through a separate marine cargo policy), with sub-limits sufficient to insure the maximum value of any property in transit at any one time;
- (e) shall cover physical damage arising because of faulty workmanship or materials;
- (f) shall cover ensuing loss from design error not otherwise excluded (LEG 3);
- (g) shall cover water damage and flood, (including the overflow of inland or tidal waters, the unusual accumulation or runoff of surface waters from any source, or mudslides or mudflows which are caused by flooding) with a sublimit of no less than \$10,000,000;
- (h) shall cover perils of earthquake and earth movement with a sublimit of no less than \$10,000,000;
- (i) shall cover physical damage resulting from machinery accidents but excluding normal and natural wear and tear, corrosion, erosion, inherent vice or latent defect in the machinery;
- shall cover demolition and debris removal coverage, with a sublimit of 20% of the loss or no less than \$10,000,000 insuring the buildings, structures, machinery, equipment, materials, facilities, fixtures and all other properties constituting a part of the Project;
- (k) shall cover increased cost for repair, rebuilding or reconstruction of damaged property due to enforcement of any law or ordinance with a sublimit of no less than \$20,000,000, including professional fees with a sublimit of no less than \$10,000,000;
- (I) shall include a sublimit for soft costs to include recurring costs should there be physical damage to the project, delay costs, loss of revenues during a delayed opening, including architects and engineers fees, and owner's extended project costs including administration and overhead during a restoration period of 12 months,
- (m) shall include civil authority ingress and egress;
- (n) shall cover plans, blueprints and specifications;
- (o) shall cover full collapse, including collapse resulting from design error, as set forth in LEG3 coverage;
- (p) shall include as named insureds City, SFMTA, Principal Project Company, and Contractors;
- (q) shall not include "as their interests may appear" language pertaining to insureds;

- (r) shall include a blanket waiver of subrogation as required by contract; and
- (s) may include deductibles, but such deductible shall not exceed \$1,000,000 per occurrence unless prior approval from the City.

1.8. <u>Railroad Protective Liability.</u> If required by any railroad, a railroad protective liability insurance policy or policies as required by one or more railroads operating near or adjacent to the Project location. The terms, limits, and specifications of such policy(ies) are set forth by each railroad.

1.9. <u>Other.</u> Any other form of insurance and with such limits, in such form, in amounts and for risks as the City or SFMTA, acting reasonably, may require from time to time. Principal Project Company's compensation shall be adjusted to reflect the cost of any such additionally required insurance.

## 2. Insurance During the Infrastructure Facilities Maintenance (IFM) Period

Principal Project Company shall obtain and keep in force, or cause to be obtained and kept in force, throughout the IFM Period the following insurance coverage:

2.1. <u>Worker's Compensation and Employer's Liability</u>. Worker's compensation and employer's liability insurance as required by applicable Law, and employer's liability insurance having coverage limits of \$1,000,000 for each accident, \$1,000,000 for disease (each employee), and \$1,000,000 for disease (policy limit).

Commercial General Liability. Commercial general liability insurance insuring 2.2. against liability of Principal Project Company and Lead IFM Contractor with respect to the Project or arising out of the Work, written on an occurrence basis. Coverage shall be at least as broad as the broadest available version of Insurance Services Office form CG 00 01. City, SFMTA, and the Board of Supervisors and their employees, officers, and officials shall be added to the primary policy as insureds using Insurance Services Office form CG 20 10 10 01 or, in City's sole discretion, forms providing the same scope of coverage. Principal Project Company shall require its Contractors to have these additional insureds included as insureds in the same manner as specified for Principal Project Company in this Section 2.2 (Commercial General Liability), and shall prohibit use of any endorsement forms that require, to effect the insured status, the execution or existence of any contract directly between such Contractors and any of the additional insureds. The insurance shall (a) apply separately for each insured against whom a claim is made or a lawsuit is brought, subject only to the insurance policy limits of liability and (b) have coverage for any one occurrence of not less than \$50,000,000, \$50,000,000 general aggregate, and \$50,000,000 products/completed operations which requirement may be met by any combination of primary and excess coverage so long as the excess coverage is written on a "follow form" basis. Exclusions are prohibited for work within 50 feet of a railroad.

2.3. <u>Commercial Automobile Liability</u>. Commercial automobile liability insurance with limits of liability of not less than \$5,000,000 per accident. The insurance must cover liability arising from any motor vehicle (i.e., "**Symbol 1**"), including owned, hired or non-owned vehicles, assigned to or used in connection with the operation and maintenance of the Project.

2.4. <u>Pollution Legal Liability</u>. Pollution legal liability insurance applicable to bodily injury, property damage, including loss of use of damaged property or of property that has not

been physically injured or destroyed, cleanup costs, and defense, including costs and expenses incurred in the investigation, defense, or settlement of claims, all in connection with any loss arising from the Project. Coverage shall be maintained in an amount of at least \$10,000,000 per loss, with an annual aggregate of at least \$10,000,000. Coverage shall apply to sudden and non-sudden pollution conditions resulting from the escape or release of smoke, vapors, fumes, acids, alkalis, toxic chemicals, liquids, or gases, waste materials, or other irritants, contaminants, or pollutants. If coverage is written on a claims-made basis, Principal Project Company warrants that continuous coverage will be maintained or an extended discovery period will be exercised for a period of 10 years beginning from the Termination Date.

2.5. <u>Property Insurance</u>. A property insurance policy covering the full replacement value of the Project throughout the entire IFM Period. Coverage shall include the following: equipment breakdown, wind perils, collapse, terrorism, water damage including overflow, leakage, sewer backup or seepage, utility interruption, debris removal, extra expense, boiler and machinery and valuable papers. Coverage shall also include earthquake insurance at a limit of \$25,000,000 and flood insurance with a sublimit of not less than \$10,000,000. City shall be named a named insured. The Collateral Agent shall be the sole loss payee on the property throughout the entire IFM Period.

During the period of any Renewal Work or other significant capital maintenance or repair work that is not associated with a loss from a covered peril, Principal Project Company shall provide coverage either through a course of construction coverage extension under the property policy or insure through a separately placed Builder's Risk Insurance policy.

The property insurance policy shall also include coverage for:

- (a) Foundations and pilings;
- (b) Physical damage resulting from machinery accidents or mechanical or electrical breakdown including electrical substations but excluding normal and natural wear and tear, corrosion, erosion, inherent vice, or latent defect in the machinery;
- (c) Plans, blueprints, and specifications;
- (d) Demolition and debris removal coverage insuring the buildings, structures, machinery, equipment, materials, facilities, fixtures, and all other properties constituting a part of the Project, with a sublimit of no less than \$10,000,000;
- (e) Increased replacement cost due to any change in applicable Law;
- (f) Business interruption/time element coverage, including loss of revenue, including loss of any Availability Payments, expediting and extra expense, with a sublimit in the amount of \$\_\_\_\_\_; [NTD: AMOUNT TO BE DETERMINED BASED ON AP/AGREEMENT BEFORE COMMERCIAL CLOSE]
- (g) Building ordinance compliance, with the building ordinance exclusion deleted;
- (h) All buildings, fixtures, improvements, and equipment that are built or placed on the Project Site; and

(i) Coverage at least as broad as coverage provided by the most recent Insurance Services Office commercial property Special Cause of Loss form.

2.6. <u>Other.</u> Any other form of insurance and with such limits, in such form, in amounts and for risks as the City or SFMTA, acting reasonably, may require from time to time. Principal Project Company's compensation shall be adjusted to reflect the cost of any such additionally required insurance.

# FORM OF INTERFACE AGREEMENT

[NOTE TO PNC: TO FOLLOW FROM PNC ONCE FINALIZED]

# <u>EXHIBIT 9</u>

## **CHANGE PROCEDURES**

### 1. City Change Procedures

**1.1** The provisions of this <u>Section 1</u> (City Change Procedures) and of <u>Section 2</u> (Unilateral Change Orders) shall apply with respect to any City Change.

**1.2** If City desires to initiate or evaluate whether to initiate a City Change, then City may issue a proposed Change Order (a "**Proposed Change Order**"). The Proposed Change Order shall state the nature, extent and details of the contemplated City Change.

## 1.3 Response to Proposed Change Order

**1.3.1** As soon as possible, and in any case, within 20 days after City delivers to Principal Project Company a Proposed Change Order, Principal Project Company shall deliver to City a change proposal ("Change Proposal") prepared in accordance with this Section 1.3 (Response to Proposed Change Order). Except as expressly set forth in this Section 1.3.1, the obligation of Principal Project Company to provide a Change Proposal is not Extra Work and shall not entitle Principal Project Company to any additional compensation, time extension or other relief. Upon receipt of a Proposed Change Order from City, if Principal Project Company believes it will incur substantial out-of-pocket contracted design costs to prepare a Change Proposal, Principal Project Company, within 7 days of the receipt of the Proposed Change Order, provide Notice to City setting forth the reasonable out-of-pocket contracted design costs that Principal Project Company will incur to prepare such Change Proposal. Failure to provide such Notice within such 7 day period shall forfeit any Claim for payment for such out-of-pocket contracted design costs. Upon delivery of such Notice by Principal Project Company, Principal Project Company shall promptly meet with City, at City's request, to explain and discuss such potential costs and such discussions and negotiations shall occur on an Open Book Basis. City, in its sole discretion, may agree to compensate Principal Project Company for such specified and approved reasonable out-ofpocket contracted design costs as Extra Work Costs, in which case, Principal Project Company shall proceed with the Change Proposal as required in the Contract Documents. Payment of such approved amounts shall be the sole compensation to Principal Project Company and no other compensation, time extension or other relief shall be available or provided. Should City elect, in its sole discretion, to not approve such potential costs, the Proposed Change Order shall be deemed withdrawn. The Parties agree that the foregoing payment of reasonable out-of-pocket contracted design costs for preparation of a Change Proposal shall only apply to unusual circumstances involving substantial design and out-of-pocket contracted design costs.

- **1.3.2** Each Change Proposal shall include:
- (a) a detailed explanation of how the contemplated City Change would impact both the D&C Work and the IFM Services;
- (b) if the Proposed Change Order is issued before Substantial Completion, a detailed description of any suggested adjustments to the Project Schedule. This includes changes

to any Contract Deadline that would be necessary because of potential delays caused by implementing the contemplated City Change;

- (c) if adjustments to any Contract Deadline are suggested:
  - a time impact analysis that identifies Critical Path impacts (including activity numbers, durations, predecessor and successor activities, resources, costs and impact, if any, on Float). The time impact analysis must show how schedule changes or disruptions would affect the Contract Deadlines while complying with the requirements of <u>Section 1.2.1</u> of <u>Division 1</u> of the Technical Requirements;
  - (ii) an assessment of the feasibility of accelerating the Work to meet the original deadline or to reduce the total delay period; and
  - (iii) if acceleration is feasible, an estimate of the cost to accelerate, as well as information about any Extra Work Costs, Financing Delay Costs, and Delay Costs, if any, payable by City if the schedule is not accelerated;
- (d) a detailed, itemized estimate of any applicable Extra Work Costs, Financing Delay Costs or Delay Costs claimed; Principal Project Company shall provide all information on an Open Book Basis as specified in <u>Exhibit 13</u> (Costs Schedule);
- (e) an estimate of the cost savings, if any, resulting from the contemplated City Change, including reductions in direct labor, material and equipment, site overhead and home office overhead, operations and maintenance, and financing costs;
- (f) the effect (if any) of the contemplated City Change on Principal Project Company's ability to perform the IFM Services stated by Contract Year;
- (g) where relief from obligations under the Contract Documents is sought, the effect of the contemplated City Change on Principal Project Company's ability to perform any of its obligations under the Contract Documents that if not performed would result in the accrual of Noncompliance Points, the assessment of Deductions or the occurrence of a PPC Default, in each case including details of the relevant obligations, the effect on each such obligation, the likely duration of that effect and the specific relief sought;
- (h) a description of any additional consents or approvals required, including amendments, if any, of any Regulatory Approvals required to implement the contemplated City Change;
- a detailed description of the steps Principal Project Company will take to implement the contemplated City Change, including measures that Principal Project Company will take to mitigate the costs, delay and other consequences of the contemplated City Change; and
- (j) any other relevant information related to the contemplated City Change.

## 1.4 Negotiation of Change Order

**1.4.1** Following City's receipt of Principal Project Company's Change Proposal and City's further assessment of the cost, schedule and other impacts of the contemplated City Change, City and Principal Project Company, shall engage in good faith negotiations to reach agreement on the terms of a change order, including any Extra Work Costs, Financing Delay Costs or Delay Costs, adjustment of the Contract Deadlines or other relief to which Principal Project Company is entitled, and any net cost savings and schedule savings to which City is entitled.

**1.4.2** City may, by written Notice, modify or abandon a contemplated City Change at any time prior to the Parties reaching an agreement on the Change Order. Principal Project Company will, as soon as practicable but in no event more than 10 Business Days after receipt of a modification, provide Notice to City of any subsequent changes to the Change Proposal.

**1.4.3** A City Change will become effective upon mutual execution of a written change order (a "**Change Order**"). The Change Order shall be in a form provided by City, and shall specify, as applicable, the timing and method for payment or deduction of any Extra Work Costs, Financing Delay Costs, Delay Costs or Financing Costs, or for realizing any net savings in the cost of the Work.

## 1.5 Disagreement on Change Proposal

If the Parties do not agree on a Change Proposal, then City may seek to resolve any points of disagreement through the Contract Dispute Procedures without issuing a Unilateral Change Order, or City may issue a Unilateral Change Order.

#### 2. Unilateral Change Orders

**2.1** City may, at any time, in its sole discretion, issue a Unilateral Change Order to Principal Project Company regarding any matter for which a Change Order can be issued or, in the event of any Contract Dispute, regarding the scope of the Work or whether Principal Project Company has performed the Work in accordance with the requirements of the Contract Documents (a "**Unilateral Change Order**"). The Unilateral Change Order will state that it is issued under this Section 2 (Unilateral Change Orders), will describe the Work to be performed and will state the basis for determining compensation, if any, and schedule adjustment, if any. The Unilateral Change Order will either: (a) direct Principal Project Company to implement a City Change; or (b) state that disputed Work is within Principal Project Company's original scope of Work or is necessary to comply with the requirements of the Contract Documents.

**2.2** If the Unilateral Change Order does not state that the Work constitutes a City Change, Principal Project Company shall proceed with the Work as directed but may assert a Claim that a City Change has occurred under the procedures in <u>Article 12</u> (City Change Process; Unilateral Change Orders; Deviations) of this Agreement.

**2.3** If the Unilateral Change Order provides for the implementation of a City Change, Principal Project Company shall, within 21 Business Days after the issuance of the Unilateral Change Order, deliver to City a Change Proposal in accordance with <u>Section 1.3</u> (Response to Proposed Change Order) of this <u>Exhibit 9</u> (Change Procedures), and the Parties shall subsequently follow the procedures and provisions set forth in <u>Section 1.4</u> (Negotiation of Change Order) of this <u>Exhibit 9</u> (Change Procedures).

**2.4** If the Unilateral Change Order provides for the implementation of a City Change or Principal Project Company intends to assert a Claim under <u>Section 2.2</u>, Principal Project Company shall maintain records for such Work in accordance with <u>Section 2.2.9</u> of <u>Exhibit 13</u> (Costs Schedule), pending resolution of the Contract Dispute or execution of a Change Order.

**2.5** The fact that a Unilateral Change Order was issued by City shall not be considered evidence that a City Change has occurred. The determination whether a City Change has occurred shall be based on an analysis of the original requirements of the Contract Documents and a determination as to whether the Unilateral Change Order in fact constituted a change in those requirements.

# 3. Principal Project Company Change Requests

**3.1** Principal Project Company may, at any time, request a Change by submitting to City a written request for a Change ("**PPC Change Request**"). PPC Change Requests must comply with the requirements of <u>Section 1.3.2</u> of this <u>Exhibit 9</u> (Change Procedures) and be in a form approved by the City.

**3.2** City has the sole discretion to accept or reject any PPC Change Request. If the City accepts a PPC Change Request, Principal Project Company shall execute a Change Order and implement the proposed change in accordance with the Change Order, applicable Technical Requirements, the Project Management Plan, Good Industry Practice and all applicable Law. Acceptance is only valid when documented in a written Change Order signed by City's Authorized Representative or their designee appointed in writing.

**3.3** If City accepts a PPC Change Request, Principal Project Company shall be solely responsible for bearing any increase in the Extra Work Costs or other costs, and any additional risks that result from the accepted Change, except as may otherwise provided in the Change Order. Principal Project Company shall not be entitled to any extension of the Project Schedule or any Contract Deadline due to delays or other impacts that result from the City accepting the PPC Change Request, except as may otherwise provided in the Change Order.

**3.4** If a PPC Change Request accepted by City results in a net cost savings to Principal Project Company, City will be entitled to 50% of such savings. City will obtain its share of the savings in the manner described in <u>Section 1.4.3</u> and otherwise in accordance with <u>Exhibit 13</u> (Costs Schedule).

Certain minor changes without significant cost savings may be approved in writing by City as Deviations, and in such event, shall not require a Change Order. Any other change in the requirements of the Contract Documents shall require a Change Order.

#### **INITIAL DESIGNATION OF AUTHORIZED REPRESENTATIVES**

City Representative: ]

Chris Lazaro, Project Director San Francisco Municipal Transportation Agency 1 South Van Ness, 8th Floor San Francisco, CA 94103 Attn: Chris Lazaro Email: <u>Chris.Lazaro@sfmta.com</u> Telephone: 415-549-6572

Principal Project Company Representative:

[NAME], [TITLE] [ADDRESS] Email:\_\_\_\_\_ Telephone: [\_\_\_\_\_]

### SUBMITTALS REVIEW PROCESS

#### 1. Submittal Requirements

Each Submittal provided by Principal Project Company to City for information, review and comment, review and acceptance or approval shall:

- (a) be accurate, complete and in conformity with the Contract Documents;
- (b) include a completed transmittal form in form agreed between City and Principal Project Company; and
- (c) include all necessary information and documentation concerning the subject matter and any additional information reasonably requested by City.

#### 2. Submittal Types and Time Periods

**2.1** Submittals provided by Principal Project Company to City will consist of the following types:

- (a) <u>Submittals for information</u>. Submittals for City information do not include any deadline for City to respond. City may provide comments at any time or not at all.
- (b) <u>Submittals for review and comment</u>. For any Submittal subject to review and comment, City may respond at any time or not all. Principal Project Company is only required to resubmit a Submittal if City responds within 14 days. City responses include: (i) reviewed with no comments; (ii) reviewed with comments, resubmittal not required; or (iii) reviewed with comments, resubmittal required. Submittals are subject to City's review and comment unless either the Contract Documents or the City-accepted Submittal Schedule contemplates a different type of review.
- (c) <u>Submittals for review and acceptance</u>. For any Submittal subject to City review and acceptance, City will have 21 days to respond. City responses include: (i) reviewed and accepted; (ii) reviewed and accepted with comments, resubmittal not required; or (iii) reviewed and not accepted with comments, resubmittal required.
- (d) <u>Submittals for approval</u>. For any Submittal subject to City approval, City will have 21 days to respond. City responses include: (i) approved; (ii) approved with comments, resubmittal not required; or (iii) not approved with comments, resubmittal required.

**2.2** If any other provision of the Contract Documents expressly provides a longer or shorter period for City to act in response to a specific Submittal, then such period shall prevail over the time periods set forth in <u>Section 2.1</u>.

**2.3** The Parties shall agree in good faith upon any necessary extensions of the review period to accommodate particularly complex or comprehensive Submittals.

**2.4** If the number of Submittals delivered simultaneously to City exceeds the limits specified in <u>Section 1.3</u> of <u>Division 1</u> of the Technical Requirements, then City may extend the applicable period for it to act with respect to such Submittals to allow City a reasonable period to respond, and no such extension shall constitute a City-Caused Delay Event, a City-Caused Relief Event, or other basis for any Claim. Submittals are considered "delivered simultaneously" if the review time periods available to City under this <u>Article 2</u> (Submittal Types and Time Periods) for two or more Submittals entirely or partially overlap.

**2.5** Whenever City is in receipt of Submittals delivered simultaneously, Principal Project Company may provide Notice to City including a requested order of priority for processing such Submittals. Upon receipt of such Notice, City will use reasonable efforts to accommodate the requested order of priority; provided, however, that City will not be obligated to shorten the review times otherwise applicable under this <u>Article 2</u> (Submittal Types and Time Periods).

**2.6** All time periods for City to act under this <u>Article 2</u> (Submittal Types and Time Periods) shall be extended by the period of any delay in City's review caused by any Relief Event (other than a City-Caused Delay Event or City-Caused Relief Event) or any PPC Fault.

**2.7** During any time that City is entitled to increase the level of its Oversight under <u>Section 15.5</u> (Persistent PPC Default and Increased Oversight) of this Agreement, the applicable period for City to act on any Submittals received during such time and not related to addressing events that instigated the <u>Section 15.5</u> (Persistent PPC Default and Increased Oversight) of this Agreement action shall be extended as reasonably needed due to the increased level of Oversight, but not to exceed an additional 10 Business Days per Submittal. No such extension shall constitute a City-Caused Delay Event, City-Caused Relief Event or other basis for any Claim.

**2.8** Principal Project Company may, by Notice to City, request expedited action on a specific Submittal. City has no obligation to expedite any Submittal but upon receipt of such a request will use reasonable efforts to accommodate such request within the practical limitations (a) on availability of City personnel relevant to the request or (b) imposed by restrictions upon City's rights under agreements with Third Parties and Utility Owners. However, City's obligation to use reasonable efforts to accommodate such a request shall not apply with respect to the review periods described in <u>Section 2.7</u>.

## 3. City Actions Relevant to Submittals

If a Submittal is subject to City's acceptance or approval, Principal Project Company may not proceed without receiving City's acceptance or approval, as applicable.

## 4. City Objection, Rejection Binding

**4.1** Any exception, objection, rejection, non-acceptance or disapproval by City shall be deemed reasonable if, among other reasons, the determination is based on any of the following grounds:

- (a) the Submittal or subject provision fails to comply, or is inconsistent, with any applicable Standards and Specifications or any covenant, condition, requirement, term or provision of the Contract Documents, Project Management Plan;
- (b) the Submittal or subject provision does not at a minimum meet Good Industry Practice;
- (c) Principal Project Company has not provided all necessary information and documentation concerning the subject matter and any additional information reasonably requested relating to the Submittal (provided that Principal Project Company may subsequently resubmit the Submittal with the required or reasonably requested content or information); or
- (d) adoption of the Submittal or subject provision, or of any course of action proposed in the Submittal, would result in a conflict with or violation of any Law or Regulatory Approval.

**4.2** Principal Project Company shall respond in writing to all of City's comments (including any exceptions, objections, rejections and disapprovals) relating to a Submittal, and subject to <u>Sections 2.1(a)</u> and <u>(b)</u>, shall make modifications to the Submittal as necessary to fully reflect and resolve all such comments in accordance with the review processes in this <u>Exhibit 11</u> (Submittals Review Process), prior to executing the Work identified in the Submittal.

**4.3** If Principal Project Company does not accommodate or otherwise resolve any City comment, Principal Project Company shall, within 10 Business Days after receipt of City's comments, provide an explanation setting out:

- (a) why modifications based on City's comments are not required;
- (b) the facts, analyses and reasons that support Principal Project Company's conclusion; and
- (c) the basis for any belief that incorporating City's comments or resolving exceptions, objections, rejections or disapprovals that would render the Submittal erroneous, defective or reflective of less than Good Industry Practice.

**4.4** Promptly following delivery of Principal Project Company's explanation under <u>Section</u> <u>4.3</u>, Principal Project Company shall meet with City with the goal of reaching agreement regarding changes to be made to the Submittal. City may at any time issue a Unilateral Change Order, in which case Principal Project Company shall proceed in accordance with City's Unilateral Change Order with the right to seek resolution of the Contract Dispute under the Contract Dispute Procedures.

**4.5** If Principal Project Company fails to provide an explanation to City in accordance with <u>Section 4.3</u>, City may deliver to Principal Project Company a Notice setting out comments that have not been addressed and relevant dates for Principal Project Company to respond. If Principal Project Company fails to address such comments by the dates specified by City, Principal Project Company shall make all changes necessary to accommodate and resolve the comment and will be fully responsible for such changes without right to assert a City-Caused

Delay Event, City-Caused Relief Event or other basis for a Claim that City has assumed design or other liability.

### ENERGY MANAGEMENT

#### 1. DEFINITIONS

1.1 Capitalized terms used but not defined herein shall have the meaning given to them in <u>Exhibit 1</u> of the Agreement.

## 2. CALCULATION OF ANNUAL ENERGY TARGET

2.1 Annual Energy Target

- (a) After the second anniversary of the Final Acceptance Date, Principal Project Company and City shall appoint an Independent Energy Auditor to provide a report inclusive of operating assumptions, current Weather Data and energy model simulation files regarding the Infrastructure Facility's Energy performance and to establish an Annual Energy Target (the "Annual Energy Target").
- 2.2 Program or Variation Adjustments to the Annual Energy Target
  - 2.2.1 At any time commencing after the Energy Performance Commencement Date, City or PPC can apply an adjustment to the Annual Energy Target for the purposes of preparing the Energy Analysis Report to account for:
- (a) any Regulated Load change(s) that results from a Change Order;
  - (b) changes in Infrastructure Facility occupancy;
  - (c) changes to Core Hours; or
  - (d) changes in the utilization or other foreseen operating conditions of the Infrastructure Facility which materially differ from those assumed for the purpose of calculating the Annual Energy Target.

For greater certainty, Principal Project Company and City acknowledge that an adjustment to the Annual Energy Target shall be made only as a result of a circumstance specified in <u>Section 2.2.1</u>.

2.2.2 Upon an adjustment to the Annual Energy Target as a result of a circumstance specified in <u>Section 2.2.1</u>, the Parties shall appoint, subject to the other Party's approval (acting reasonably), a complete energy audit to be conducted by an Independent Energy Auditor. The costs associated with the Independent Energy Auditor shall be shared equally between the Parties. Each Party shall cooperate with (including providing any information requested by) the IEA. The IEA shall prepare a report making recommendations regarding adjustments to the Annual Energy Target; the report shall include a detailed computer simulation, or other mutually agreed energy analysis method, of Energy usage by function and a comprehensive evaluation of Energy usage patterns. The

Parties shall work in good faith to have such IEA deliver its report within 30 days of the IEA's appointment. In preparing the report, the IEA shall have access to the most recent Energy Analysis Report. For greater certainty, the recommended adjustments should only apply for that period of time that the circumstance(s) in Section 2.2.1 giving rise to such adjustments occur, continue or impacted. Both City and Principal Project Company, each acting reasonably, shall have 20 days following receipt of such report to provide notice to the other of (i) its agreement to the recommended adjustments to the Annual Energy Target set forth in such report; or (ii) its disagreement with such recommendations, together with its rationale for such disagreement. In the event that either Party provides notice of its disagreement with the IEA's recommendations, then either Party may refer the matter to the Contract Dispute Procedures. If both Parties are in agreement with the recommendations, then the Annual Energy Target shall be adjusted in accordance with such recommendations effective as of the date on which the circumstance(s) in Section 2.2.1 commenced.

## 3. MEASUREMENT

- 3.1 Without prejudice to Exhibit 18 Technical Requirements or anything else in this Exhibit 12, Principal Project Company shall measure the amount of Aggregate Actual Consumption to meet LEED BD+C Measurement and Verification mandatory credits and to meet LEED EBOM reporting requirements for the Infrastructure Facility in respect of each calendar month beginning on the Substantial Completion Date and ending on the earlier of expiry of Term or termination of the Agreement.
- 3.2 For each Contract Year after the Energy Performance Commencement Date, Principal Project Company shall provide City with a draft Energy Analysis Report within 60 days following the end of each Contract Year, which report shall include copies of all working papers to fully support the draft Energy Analysis Report. The draft Energy Analysis Report shall be consistent with the format and content requirements set out in <u>Section 3.7</u>.
- 3.3 As soon as practicable and in any event within 30 days after receipt by City of the draft Energy Analysis Report (or on such other date as may be agreed between City and Principal Project Company ), Principal Project Company and City shall convene an annual energy review meeting to be attended by the PPC Representative, the City Representative and such other individuals as may be agreed to by the Parties (the "Annual Energy Review Meeting"). At the Annual Energy Review Meeting, Principal Project Company shall present the draft Energy Analysis Report to City, and City and Principal Project Company shall discuss the Aggregate Actual Consumption for each discrete Energy Service for the preceding Contract Year.
- 3.4 Principal Project Company shall assist the City Representative and afford the City Representative such information and access to the Infrastructure Facility, building management system records, Utility meters, and Help Desk (as such services are described in <u>Exhibit 18</u> Technical Requirements) and by other means as may reasonably be required for the City Representative to review and confirm the content of the draft Energy Analysis Report provided by Principal Project Company.

- 3.5 City shall notify Principal Project Company within 21 days after the Annual Energy Review Meeting of the details of any disagreement of all or any aspect of the draft Energy Analysis Report, and the Parties shall then seek to agree to any matters in dispute. Where matters cannot be resolved within 20 days following receipt by Principal Project Company of notification by City of a disagreement (or such other period as may be otherwise agreed between the City Representative and the PPC Representative, acting reasonably), the disagreement shall be dealt with in accordance with the Dispute Resolution Procedure.
- 3.6 Subject to <u>Section 3.5</u>, within 20 days following each Annual Energy Review Meeting, or within such period as may be otherwise agreed between the City Representative and the PPC Representatives, acting reasonably, City shall confirm its acceptance of all or any aspects of the Energy Analysis Report.
- 3.7 Content and Format of the Energy Analysis Report
  - 3.7.1 The Energy Analysis Report shall present findings of Aggregate Actual Consumption for each separate Energy Service for the relevant Contract Year and calculation of the Corrected Aggregate Energy Consumption (CAEC) and shall include the following:
    - 3.7.1.1 a summary of (A) actual usage and breakdown by Utility in megajoules and cubic meters, or other Utility rate units; and (B) any exceptional changes in consumption or pattern of use since any previous Energy Analysis Report;
    - 3.7.1.2 accurate and precise consumption data;
    - 3.7.1.3 identification of potential cost savings in respect of Energy usage at the Infrastructure Facility and provide an estimate of potential Energy Service consumption savings broken down by fuel type, implementation costs, and projected savings, along with identifying potential risks associated with each proposed cost savings measure;
      - (i) presentation of Aggregate Actual Consumption for each individual Energy Service;
      - (ii) presentation of correlated energy Weather Data for the relevant Contract Year;
      - (iii) detailed analysis of all sub metered end-uses; and
      - (iv) calculation of the Corrected Aggregate Energy Consumption for each individual Energy Service;
    - 3.7.1.4 outline of any outstanding issues from any previous Energy Analysis Report; and,
    - 3.7.1.5 summary of adjustments to the Annual Energy Target, if any, as agreed upon in accordance with <u>Section 2.2</u>.

- 3.8 Principal Project Company shall deliver a Monthly Energy Report to City, following Substantial Completion, five business days following the end of each Contract Month.
- 3.9 Content and Format of the Monthly Energy Report
  - 3.9.1.1 The Energy Analysis Report shall show monthly Aggregate Actual Consumption and monthly Corrected Aggregate Energy Consumption data for each individual Energy Service
  - 3.9.1.2 For the monthly Corrected Aggregate Energy Consumption:
    - (i) A comparison against the previous month's value, as well as supporting narrative articulating reasons for any variance
    - (ii) A comparison against the value of the same month of the previous Contract Year, as well as supporting narrative articulating reasons for any variance
    - (iii) A comparison of current rolling 12-month average against the previous month's rolling 12-month average, as well as supporting narrative articulating reasons for any variance

#### 4. COMPARING CORRECTED AGGREGATE CONSUMPTION WITH ADJUSTED ANNUAL ENERGY TARGET

- 4.1 After the acceptance of the Energy Analysis Report as described in <u>Section 3.6</u> for each Contract Year, the CAEC for each Energy Service shall be compared to the Annual Energy Target or Adjusted Annual Energy Target for each Energy Service, and:
  - 4.1.1 if the CAEC in respect of any discrete Energy Service is greater than 115% of the AET or AAET in respect of such Energy Service, then City may request that the Principal Project Company prepare and submit an Energy Performance Action Plan ('EPAP'). For the purposes of an Energy Performance Action Plan in respect of this <u>Section 4.1.1</u>, the PAP will be due no later than 30 days after acceptance of the Energy Analysis Report; and
  - 4.1.2 if the CAEC in respect of any Energy Service is less than 115% of the AET or AAET in respect of such Energy Service, no further action is required.
- 4.2 If an EPAP is required pursuant to <u>Section 4.1.1</u>, the Energy Performance Action Plan will, at a minimum;
  - (a) identify the variances between the CAEC and the AET or AAET;
  - (b) summarize potential issues giving rise to the variances;
  - (c) include a root cause analysis, where possible, or a plan to develop root cause analysis if it cannot be established prior to issuance of the PAP; and

- (d) the steps or actions that will be undertaken by Principal Project Company to address the issues and specific and reasonable timeframes for such steps.
- 4.3 If an EPAP is required pursuant to <u>Section 4.1.1</u>, any missed submittals of Energy Performance Action Plans or failure to implement particular steps or actions of such EPAPs shall be addressed in accordance with <u>Section 6.2.2</u> of <u>Exhibit 4</u> (Payment Mechanism).

# 5. EXTERNAL ENERGY AUDITOR

- 5.1 In accordance with <u>Section 8.9.2</u> of the Agreement, City has the right to procure an External Energy Auditor to provide Infrastructure Facility operations and energy usage and consumption advice. This review would likely also include recommendations that cover PPC IFM Services and SFMTA O&M Services scope.
- 5.2 Similar to Section 4.1.1, if the CAEC is greater than 115% of the AET or AAET, then Principal Project Company will be required to comply with the recommendations of the External Energy Auditor, if procured, at its own cost. Such recommendations may include operational and maintenance recommendations, but may not require Principal Project Company to replace any material equipment or incur any material capital expenditure. Any requirement to replace material equipment or incur material capital expenditures shall be addressed in accordance with Section 12 of the Agreement. Additionally, Principal Project Company would not be required to comply with any recommendations that contradict specific advice provided by equipment manufacturers.

# COSTS SCHEDULE

#### 1.0 Overview and General Principles

- **1.1** This <u>Exhibit 13</u> (Costs Schedule) describes the methods for calculating:
  - (A) Extra Work Costs, Delay Costs, and Financing Delay Costs owing from City to Principal Project Company pursuant to <u>Article 12</u> (City Change Process; Unilateral Change Orders; Deviations) and <u>Article 14</u> (Compensation and Other Relief for Delay Events and Relief Events) of this Agreement; and
  - (B) Any other amount expressly payable by City or Principal Project Company under this Agreement.

**1.2** The following general principles apply to Extra Work Costs, Delay Costs, and Financing Delay Costs calculated under this <u>Exhibit 13</u> (Costs Schedule):

- Principal Project Company shall provide all information referred to in this <u>Exhibit 13</u> (Costs Schedule) or <u>Exhibit 9</u> (Change Procedures) on an Open Book Basis;
- (B) All payments or deductions made by City to Principal Project Company in accordance with this <u>Exhibit 13</u> (Costs Schedule) will be made as and when incurred or in arrears in accordance with <u>Section 7.0</u> (Form of Timing of Compensation) or as otherwise expressly provided under this Agreement;
- (C) In calculating Extra Work Costs, Delay Costs, and Financing Costs, the time value of money and timing of cash flows shall be accounted for. This means cash flows, whether they are costs incurred or payments received, shall be discounted or inflated to reflect when they occur (if applicable);
- (D) Extra Work Costs shall not include Delay Costs or Financing Delay Costs; and
- (E) No amounts shall be double counted.

**1.3** Principal Project Company's recovery for any Extra Work Costs, Delay Costs, and Financing Delay Costs under this <u>Exhibit 13</u> (Costs Schedule) is subject to Principal Project Company complying with the timeframes specified in <u>Article 14</u> (Compensation and Other Relief for Delay Events and Relief Events), and <u>Exhibit 9</u> (Change Procedures) as applicable, and otherwise in accordance with this Agreement.

#### 2.0 Extra Work Costs

#### 2.1 Methods of Determining Extra Work Costs

Extra Work Costs, payable in accordance with <u>Article 14</u> (Compensation and Other Relief for Delay Events and Relief Events) or <u>Exhibit 9</u> (Change Procedures) of this Agreement, shall be determined using the following methods:

- (A) negotiated lump sum;
- (B) unit prices; or
- (C) Force Account.

Each method is described below.

#### 2.2 Negotiated Lump Sum

If the City determines in its good faith discretion, Extra Work Costs will be determined on a negotiated lump sum basis. Subject to the City's right to issue a Unilateral Change Order, lump sum costs of Extra Work, negotiated and agreed with Principal Project Company shall be based on the direct costs to perform the Extra Work, as follows:

**2.2.1** <u>Direct Costs</u>. The City will pay Principal Project Company the sum of the direct costs of labor, materials, and equipment used to perform Extra Work as follows:

- (A) <u>Labor</u>. The directs costs of labor for workers that actually perform Extra Work. Such workers include superintendents and supervisory foremen only if they are at the Project Site planning, coordinating, or overseeing Extra Work. All other supervision costs shall be included in the markup defined in <u>Section 2.2.3</u> (Markup for Overhead and Profit). Whether the employer is Principal Project Company, a Contractor, a lower tier Contactor, or other forces, the direct cost of labor is the sum of the following:
  - (1) <u>Actual Wages</u>. The actual wages paid, including any actual payments by the employer for its workers' health and welfare, pension, vacation, training, and similar purposes.
  - (2) <u>Labor Surcharge</u>. The applicable labor surcharges in the California Department of Transportation's Labor Surcharge and Equipment Rental Rates publication in effect on the date the Extra Work is completed. These labor surcharges shall constitute full compensation for employer's payment of workers' compensation insurance, Social Security, Medicare, state and federal unemployment insurance, and state training taxes. City will not pay any other fixed labor burdens unless approved in writing by the City.
  - (3) <u>Subsistence and Travel Allowance</u>. The actual allowance paid to workers for subsistence and travel.

- (B) <u>Materials</u>. The direct costs of materials required and furnished specifically for Extra Work. This includes only the direct expenditure, including sales tax, borne by the purchaser from the Supplier and any transportation expenses (e.g., shipping fees, freight charges), except delivery charges unless specifically required for the Extra Work. If a genuine Supplier offers a trade discount to purchaser, Principal Project Company shall credit the City with this discount, even if the discount was not originally taken. If materials originate from a Supplier wholly or partially owned by a PPC-Related Entity, the City's payment therefor shall not exceed the current wholesale price, as determined by the City. The term "trade discount" includes the concept of cash discounting.
- (C) <u>Equipment</u>. The direct costs of equipment are the applicable rental rates for equipment in the California Department of Transportation's Labor Surcharge & Equipment Rental Rate Book (including its supplement Miscellaneous Equipment Rental Rates) in effect on the date the Extra Work is completed.
  - (1) As deemed appropriate, the City may adjust such rental rates and use them to compute payments for equipment, regardless of whether the equipment is under the control of a PPC-Related Entity through direct ownership, leasing, renting, or other method of acquisition; except that, for equipment rented or leased in arm's length transactions with outside vendors, the City will reimburse Principal Project Company at the actual rental or leased invoice rates when such rates are reasonably in line with the applicable rates specified in the Labor Surcharge & Equipment Rental Rate Book (including its supplement Miscellaneous Equipment Rental Rates) as determined by the City. Arm's length rental or lease transactions are those in which the firm involved in the rental or lease of such equipment is not associated with, owned by, have common management, directorship, facilities, or stockholders with the firm renting the equipment. Principal Project Company has the burden of proof to demonstrate that a rental or lease transaction was an arm's length transaction. Principal Project Company shall submit copies of all rental or lease invoices, and other information as requested by the City, if any, as supporting documentation with each Proposed Change Order cost proposal.
  - (2) For equipment that is not listed in the Labor Surcharge & Equipment Rental Rate Book, Principal Project Company shall provide to the City three separate quotes for rental of the applicable equipment for City's consideration.
  - (3) The City will pay for equipment based on daily, weekly, or monthly rates, whichever are lowest. The City will not pay for equipment based on hourly rates including operator. Unless otherwise specified, Principal Project Company shall use manufacturer's ratings and manufacturer-approved modifications to classify equipment for determination of applicable rental rates. If, however, equipment of unwarranted size or type and cost is used, Principal Project Company shall calculate the cost at the rental rate for equipment of proper size and type.

- (4) The City will pay for equipment only for the time the equipment is in productive operation on the Extra Work. The City will not pay for equipment while inoperative due to breakdown or for non-work days. In addition, the City will not pay for any equipment rental time required to move the equipment to and from the Project Site. The City will pay for equipment loading and transportation costs, in lieu of rental time, only if the equipment does not move under its own power and is utilized solely for the Extra Work. The City will not pay for mobilization or demobilization of equipment already on the Project Site. The City will reimburse Principal Project Company for equipment that is idle, non-operating, or in standby mode at the lesser of Caltrans' rates, as adjusted by Caltrans Delay Factor (defined in the Labor Surcharge & Equipment Rental Rate Book), as adjusted by its standby calculation, unless such equipment is rented or leased as provided above.
- (5) Individual pieces of equipment having a replacement value of \$1,000 or less are considered small tools or small equipment; City will not pay for such tools and equipment since the costs of these tools and equipment are included as part of markups for overhead and profit as defined in <u>Section 2.2.3</u> (Markup for Overhead and Profit).
- (6) Payment to Principal Project Company for the use of equipment as set forth in this <u>Exhibit 13</u> (Costs Schedule) shall constitute full compensation to Principal Project Company for the cost of fuel, power, oil, lubricants, supplies, small equipment, necessary attachments, repairs and maintenance of any kind, depreciation, storage, insurance, labor (except for equipment operators), and any and all costs to Principal Project Company incidental to the use of the equipment.

**2.2.2** Costs Included as Part of Markup for Overhead and Profit. To the total of the direct costs of labor, materials, and equipment computed as provided in Section 2.2.1 (Direct Costs), the City will pay Principal Project Company markups for overhead and profit, as specified in Section 2.2.3 (Markup for Overhead and Profit). These markups shall constitute full compensation for all direct and indirect overhead costs and profit, which shall be deemed to include all items of expense not specifically listed in Section 2.2.1 (Direct Costs) as direct costs. The City shall not be obligated to pay for any separate allowance or itemization for any overhead costs. The following is a list, not intended to be comprehensive, of the types of costs that are included in the markups for overhead and profit for all Extra Work:

- (A) Home office and field personnel including, principals, project managers, superintendents and supervisory foremen (unless they are at the Project Site planning, coordinating, or overseeing Extra Work), estimators, project engineers, detailers, draftspersons, schedulers, consultants, watchpersons, payroll clerks, administrative assistants, and secretaries.
- (B) All field and home office expenses, including: field trailers; parking; storage sheds; office equipment and supplies; telephone service at the Project Site; longdistance telephone calls; fax machines; computers and software; internet and email services; temporary utilities; sanitary facilities and services; janitorial

services; small tools and equipment with a cost under \$1,000 each; portable scaffolding; blocking; shores; appliances; job vehicles; security and fencing; conformance to all regulatory requirements including compliance with safety regulations, safety programs, and safety meetings; cartage; warranties; record documents; and all related maintenance costs.

- (C) Administrative functions, including reviewing, coordinating, distributing, processing, posting, recording, estimating, negotiating, scheduling, schedule updating and revising, expediting, surveying, engineering, drawing, detailing, revising shop drawings, preparing record drawings, carting, cleaning, protecting the Work, and other incidental Work related to the Change Order.
- (D) Bond and insurance costs.
- (E) All other costs and taxes required to be paid, but not included under direct costs as defined in <u>Section 2.2.1</u> (Direct Costs).

**2.2.3** <u>Markup for Overhead and Profit</u>. To the actual total direct costs of labor, material, and equipment used to perform Extra Work, the City will apply the following markups for overhead and profit:

(A)	Initial Markup. An initial maximum markup for the PPC-Related Entity that
	performs the Extra Work based on the direct cost categories, as follows:

Direct Cost Categories	Maximum Markup
Principal Project Company – labor *	10%
Principal Project Company – materials*	10%
Principal Project Company – equipment*	10%
Prime Contractor/Subcontractor (of any tier) – labor	10%
Prime Contractor/Subcontractor (of any tier) – materials	10%
Prime Contractor/Subcontractor (of any tier) – equipment	10%

\* Extra Work self-performed by Principal Project Company.

- (B) <u>Additional Markup</u>. An additional maximum markup for the administration of Extra Work performed by a Contractor, as follows:
  - (1) <u>Prime Contractor</u>. For Extra Work performed by a Prime Contractor, an additional maximum markup of 5% for Principal Project Company's administration of the Extra Work.

(2) <u>Subcontractor</u>. For Extra Work performed by a Subcontractor, additional maximum markups of 5% each for Principal Project Company, Prime Contractor, and, if applicable, the higher-tier Subcontractor(s) for their respective administration of the Extra Work. The total of these additional markups shall not exceed 10% regardless of the number of Subcontractor tiers involved with the Extra Work.

#### 2.2.4 Adjustments and Considerations for Extra Work Costs.

- (A) When both additions and credits are involved in any one Change Order, Principal Project Company's markup for direct costs shall be computed on the basis of its direct costs and labor productivity for the net change in the quantity of the Extra Work. For example, if a Change Order adds 14 units on one drawing and deletes 5 units on another drawing, the markup shall be based on the net addition of 9 units. No markup will be allowed if the deductive cost exceeds the additive cost.
- (B) If the City issues written notice of deletion of a portion of Extra Work after the commencement of such Extra Work or after Principal Project Company has ordered acceptable materials for such Work which cannot be cancelled, or if part or all of such Extra Work is not performed by Principal Project Company because it is unnecessary due to actual Project Site conditions, City will pay Principal Project Company for direct costs of such Work actually performed plus markup for overhead and profit as provided in <u>Section 2.2.3</u> (Markup for Overhead and Profit).
- (C) The City shall not be obligated to compensate Principal Project Company for costs incurred after Principal Project Company receives the City's written notice deleting the portion of Extra Work.
- (D) Materials ordered by Principal Project Company prior to the City's issuance of a notice of deletion and paid for by the City shall become the property of the City, and the City will pay for the actual cost of any further handling of such material. If the material is returnable to the vendor, and if the City so directs, Principal Project Company shall return the material and the City will pay Principal Project Company only for the actual charges made by the vendor for returning the material including restocking charges.
- (E) Principal Project Company shall be solely responsible for determining which of its Contractors and Suppliers are assigned Change Orders. The City will not provide additional compensation to Principal Project Company for the cost of its Contractors and Suppliers to review, post, coordinate, or perform related tasks to administer Change Orders.

**2.2.5** <u>Records</u>. Principal Project Company shall maintain its records in such a manner as to provide a clear distinction between the direct costs of Change Orders and the cost of original Work. This requirement pertains to all types of Change Orders, as well as the additions, deletions, revisions, and Claims initiated by Principal Project Company.

## 2.3 Unit Prices

If mutually agreed by the Parties or specified in the Contract Documents, the cost of Extra Work will be determined on a unit-cost basis.

**2.3.1** <u>Initial Estimate</u>. The City will calculate an initial cost estimate by multiplying the preagreed unit prices by the estimated quantities indicated in the corresponding Change Order.

**2.3.2** <u>Final Cost Calculation</u>. The actual cost payable to Principal Project Company for the Extra Work will be based on the real quantities of work completed, not the initial estimates.

**2.3.3** Overhead and Profit. Unit prices must include any overhead and profit, the calculation of which shall follow the guidelines provided in this Exhibit 13 (Costs Schedule).

### 2.4 Force Account

If the City determines in its good faith discretion, Extra Work Costs will be determined on a Force Account basis. In such a case, the City will direct Principal Project Company to proceed with the Extra Work on a Force Account basis, subject to a "not to exceed" budget established by the City.

**2.4.1** <u>General</u>. When the City pays Extra Work on a Force Account basis, all direct costs itemized in accordance with <u>Section 2.2.1</u> (Direct Costs) shall be subject to the approval of the City and compensation will be determined as set forth in this <u>Exhibit 13</u> (Costs Schedule).

- (A) All requirements for direct costs and markup for overhead and profit provided in <u>Section 2.2.3</u> (Markup for Overhead and Profit) shall apply to Force Account Work. However, the City will pay only the actual necessary costs verified in the field by the City on a daily basis.
- (B) Principal Project Company shall be responsible for all costs related to the documentation, data preparation, and administration of Force Account Work. Compensation for these costs shall be fully covered by the markup for overhead and profit markup as provided in <u>Section 2.3</u> (Unit Prices).
- (C) <u>Notification and Verification</u>. Before commencing any Force Account Work, Principal Project Company shall provide written notice to the City at least 24 hours before the scheduled Work. The Force Account Work must be witnessed, documented, and approved in writing by the City on the day the Work is performed. Failure by the Principal Project Company to provide timely notice to the City before initiating Force Account Work shall result in the City not being obligated to compensate for such Work.

In addition, Principal Project Company shall notify the City when the cumulative costs incurred by Principal Project Company for the Force Account Work equal 80% of the budget pre-established by the City. The City shall not be obligated to compensate Principal Project Company for Force Account Work exceeding the "not to exceed" budget amount if Principal Project Company fails to provide the required notice before exceeding 80% of the Force Account budget.

- (D) <u>Reports</u>. Principal Project Company shall diligently proceed with City-directed Force Account Work and shall submit to the City no later than 12:00 p.m. of the day following performance of Force Account Work a daily Force Account Work report on a form obtained from the City. The report shall provide an itemized, detailed account of the daily Force Account labor, material, and equipment, including names of the individuals and the specific pieces of equipment identified by manufacturer's model type and serial number. Principal Project Company's authorized representative shall complete and sign the report. The City shall not be obligated to compensate Principal Project Company for Force Account Work for which Principal Project Company does not timely complete and submit the aforementioned report to the City.
- (E) <u>Agreement</u>. If Principal Project Company and the City reach a negotiated, signed agreement on the cost of a Change Order while the Extra Work is proceeding on a Force Account basis, Principal Project Company's signed written reports shall be discontinued and all previously signed reports shall become invalid.

#### 3.0 Delay Costs

- (A) <u>General</u>. In the event of a City-Caused Delay Event, the City will pay Principal Project Company for incurred Delay Costs as specified in this <u>Section 3.0</u> (Delay Costs) to the extent (i) expressly allowed under <u>Section 14.1.4(b)</u> of this Agreement, and (ii) such costs have not been previously paid as allowed under <u>Section 2.4</u> (Force Account). Such payment constitutes full compensation for incurred Delay Costs.
- (B) The City will not pay for Delay Costs until Principal Project Company submits an itemized statement of those costs; Principal Project Company must provide the content specified in <u>Section 2.4</u> (Force Account), for the applicable items in this statement and as follows:
  - (1) Proof of cost of superintendent, or other project staff salaries, wages, and payroll taxes and insurance;
  - (2) Proof of cost of office rent, utilities, land rent, and office supplies;
  - (3) Proof of escalated cost for labor, equipment, and material;
  - (4) Proof of material storage costs; and
  - (5) Proof of other increased Delay Costs consistent with this <u>Section 3.0</u> (Delay Costs).
- (C) <u>Allowable Delay Costs</u>. Increases in cost for labor, equipment, and materials will be calculated as follows:
  - <u>Idle Labor</u>. Labor costs during delays must be calculated as specified in <u>Section 2.2.1(A)</u> for all non-salaried personnel remaining on the Project as required under collective bargaining agreements or for City-approved reasons.

- (2) <u>Escalated Labor</u>. Payments authorized for increases in labor costs will be based on (i) the difference between old and new labor rates established by a State or federal agency, or (ii) a project labor agreement or other agreement between the employee's and Principal Project Company's bargaining agency, which is accepted by City. Payment will be based on certified payrolls. Payment will also include the increases in fringe benefit rates and increases in payroll taxes that Principal Project Company is required to pay.
- (3) <u>Idle Equipment</u>. Payment may be allowed on a rental basis for the idled equipment if any of the following criteria is met:
  - a. The equipment is on the Project Site at the time of the delay, is required for the controlling operation, and cannot be used at other locations on the Project Site.
  - b. The equipment is specialized and directly related to the controlling operation, whether on or off the Project Site. This must be certified by Principal Project Company and verified by the City.
  - c. The rental rate for idled leased or rented equipment will be the leased or rented rate established in <u>Section 2.2.1(C)</u>. However, the City may direct Principal Project Company to return equipment and take it off rental.
  - d. The rental rate for idled equipment owned by PPC-Related Entities will be one-half the rate established in <u>Section 2.2.1(C)</u>. No payment will be allowed for operating costs.
  - e. Payment will be limited to the difference between the hours used and 8 hours in any one day and to the difference between the hours used and 40 hours in any one week. No additional compensation for overhead will be allowed.
  - f. Equipment demobilization and remobilization, if directed by City, will be paid in accordance with <u>Section 2.0</u> (Extra Work Costs).
- (4) <u>Material Escalation or Material Storage</u>. Payment for increased cost of materials will be based on differences in the invoice costs before and after the delay period. When requesting an increase in cost of materials, Principal Project Company shall document the increased costs due to the delay. The cost of materials storage during the delay will be the invoiced storage cost.

## 4.0 Financing Delay Costs

Financing Delay Costs, when allowed under this Agreement, will be paid in the form and timing as described in <u>Section 7.0</u> (Form and Timing of Compensation).

#### 5.0 Directed Acceleration

If the City orders Principal Project Company to accelerate the D&C Work in accordance with <u>Section 1.3.2(c)</u> of <u>Exhibit 9</u> (Change Procedures), and in the absence of agreed upon compensation, the City will compensate Principal Project Company for performance of the accelerated work in accordance with <u>Section 2.4</u> (Force Account).

#### 6.0 Unrecoverable Costs

Principal Project Company is not entitled to compensation for the following costs:

- (A) Loss of anticipated profit.
- (B) Consequential damages, including loss of bonding capacity, loss of bidding opportunities, insolvency, and the effects of force account work on other projects, or business interruption.
- (C) Indirect costs.
- (D) Attorneys' fees, claim preparation expenses, and the costs of litigation.
- (E) Unabsorbed or extended field or home office overhead or any damages using an Eichleay or similar equation, except as otherwise provided in the mark ups specified in <u>Section 2.2.2</u> (Costs Included as Part of Markup for Overhead and Profit).
- (F) The cost of project management services provided by Principal Project Company.

The following do not constitute cause for a Claim for Extra Work Costs or Delay Costs:

- (1) The inability to secure satisfactory materials, for reasons beyond Principal Project Company's control, from the source upon which the proposal was based, unless project specific single source Suppliers are specified by City; or
- (2) Changes in carrier rates or the alteration of transportation facilities for these materials during the Term.

#### 7.0 Form and Timing of Compensation

# 7.1 Payments or Deductions of Extra Work Costs, Delay Costs and Financing Delay Costs

If a Delay Event or Relief Event:

(A) results in an amount owing from Principal Project Company to City in accordance with this Agreement, City will deduct such amount from the Milestone Payment (only to the extent that such Delay Event or Relief Event affects the (1) D&C Work or (2) IFM Services prior to the Substantial Completion Date) or Availability Payments payable to Principal Project Company after the Delay Event or Relief Event, or if no subsequent Milestone Payment or Availability Payments are payable to Principal Project Company, such amount will be a debt due and payable by Principal Project Company to City;

- (B) results in an amount owing from City to Principal Project Company that is not financed by Principal Project Company in accordance with <u>Section 7.2</u> (Additional Funding), City will compensate Principal Project Company as follows:
  - (1) subject to Sections 7.1(B)(2) and 7.1(B)(3) in the form of:
    - a. an adjustment to the Milestone Payment or Availability Payments over the Term;
    - b. a lump sum payment;
    - c. progress or other periodic payments invoiced as Extra Work is completed or as other multiple payments over the Term; or
    - d. any combination of the above,

in accordance with the payment arrangements set out in the Change Order or Unilateral Change Order or otherwise as determined by the City in its sole discretion;

- (2) in respect of Extra Work Costs or Delay Costs, within 1 calendar month after the date of the receipt from Principal Project Company of the Change Order except to the extent that any Extra Work Costs or Delay Costs are disputed by City and referred for dispute resolution in accordance with <u>Article 18</u> (Contract Dispute Procedures); and
- (3) in respect of Financing Delay Costs, on the date which City would have paid the Milestone Payment or Availability Payment relating to those days of delay had Substantial Completion not been delayed by the relevant Compensable Delay Event; or
- (C) results in an amount owing from City to Principal Project Company that is financed by Principal Project Company in accordance with <u>Section 7.2</u> (Additional Funding), City will pay such amount to Principal Project Company in the same manner as <u>Section 7.1(B)</u>.

## 7.2 Additional Funding

Where City requests Principal Project Company obtain funding for a Delay Event or Relief Event, Principal Project Company shall use all reasonable endeavors to obtain such funding, including by:

- (A) using any savings resulting from other Delay Events or Relief Events which have resulted in amounts being available under the Financing Documents;
- (B) utilizing any standby facility that may be available to Principal Project Company;

- (C) arranging for additional funding under the Financing Documents and from other sources (if permitted under the Financing Documents); and
- (D) arranging other funding obtained on commercial terms for Principal Project Company by City (without any obligation on City to make any such arrangements).

To the extent Principal Project Company is able to obtain funding, the cost of the funding will be taken into consideration by the parties in the compensating Principal Project Company for the Delay Event or Relief Event in accordance with <u>Section 7.1</u> (Payments or Deductions of Extra Work Costs, Delay Costs and Financing Delay Costs). City shall pay Principal Project Company an amount equal to the reasonable out-of-pocket expenses incurred by Principal Project Company in seeking such financing, provided that City approved such expenses prior to Principal Project Company incurring them.

Where, having used all reasonable endeavors, Principal Project Company is unable to obtain funding or funding is on terms which are not satisfactory to City, City will, without limiting its rights under Exhibit 9 (Change Procedures), compensate Principal Project Company in accordance with Section 7.1(A)-(C).

### KEY CONTRACT PROVISIONS

#### I. Key Contracts

Each Key Contract shall:

- (a) Include a covenant to maintain all licenses required by applicable Law;
- (b) Require the Key Contractor to carry out its scope of work in accordance with applicable requirements of the Contract Documents, the Regulatory Approvals, applicable Law, and plans, systems and manuals developed and used by Principal Project Company under the Contract Documents;
- (c) Set forth representations, warranties, guaranties and liability provisions of the Key Contractor appropriate for work of a similar scope and scale;
- Expressly state that all warranties and guaranties remaining in effect upon the expiration of the Term or earlier termination of this Agreement, whether express or implied, shall inure to the benefit of City, its successors and assigns, and any Third Parties for whom Work is being performed;
- (e) Set forth a standard of professional responsibility or a standard for commercial practice (as applicable) equal to or better than the requirements of the Contract Documents and in accordance with Good Industry Practice for work of similar scope and scale;
- (f) To the extent applicable, if not obtained by Principal Project Company, require the Key Contractor to provide Payment Bond(s) and Performance Bond(s) as required under <u>Section 10.2</u> (Performance Security) of this Agreement before commencement of any work by or on behalf of the Key Contractor, and expressly require such Key Contractor to provide any surety Notices of loss or potential loss to Principal Project Company and City;
- (g) Preclude suspension of performance or demobilization by the Key Contractor unless and until it delivers to City Notice of the other contracting party's breach or default under such Key Contract and allows City the reasonable opportunity to cure such breach or default;
- (h) Not be assignable by the Key Contractor without Principal Project Company's and City's prior consent, provided that this provision shall not prohibit subcontracting of portions of the Work to qualified Subcontractors;
- (i) Include the requirements and provisions in this Agreement applicable to Contractors regarding title to and other Intellectual Property rights and licenses;
- (j) Require the Key Contractor to participate in meetings between Principal Project Company and City concerning matters pertaining to such Key Contractor, its work or the coordination of its work with Other Contractors in accordance with direction to such Key

Contractor provided by Principal Project Company or other party to the Key Contract, provided that City retains authority to give such direction or take such action as in its opinion is necessary to remove an immediate and present threat to the safety of life or property;

- (k) Require the Key Contractor to participate in, be subject to and give evidence in any dispute resolution proceeding under <u>Article 18</u> (Contract Dispute Procedures) of this Agreement, if such participation is requested by either City or Principal Project Company;
- (I) Without cost to Principal Project Company or City, and subject to the rights of the Collateral Agent under any Direct Agreement, permit assignment to City or its successors, assigns or designees of all Principal Project Company's or other contracting party's rights under the Key Contract (with such assignment to include the benefit of all Key Contractor warranties, indemnities, guarantees and professional responsibility), contingent only upon delivery of Notice from City following the Termination Date, allowing City or its successor, assign or designee to obtain the benefit of Principal Project Company's or other contracting party's rights with liability only for those remaining obligations of Principal Project Company or the other contracting party accruing after the date of delivery of said Notice from City, without extinguishing existing claims of the Key Contractor against Principal Project Company or the corresponding Claims of Principal Project Company against City;
- (m) Expressly state that any acceptance of assignment of the Key Contract by City, the Collateral Agent or either or their respective successors, assigns or designees shall not operate to make the assignee responsible or liable for any breach of the Key Contract by Principal Project Company or for any amounts due and owing under the Key Contract for work or services rendered before acceptance of the assignment;
- (n) Expressly include the Indemnitees as indemnitees, with direct right of enforcement, in any indemnity given by the Key Contractor under the Key Contract;
- (o) Expressly include an acknowledgement that, except to the extent of stop notice rights under State law, the Key Contractor has no right or claim to any lien or encumbrance upon the Project and Project Site for failure of the other contracting party to pay amounts due the Key Contractor, and a waiver of any such right or claim that may exist at Law or in equity;
- (p) Expressly include the right of Principal Project Company to terminate the Key Contract in whole or in part upon any termination of this Agreement without liability of Principal Project Company or City for the Key Contractor's lost profits or business opportunity;
- (q) Not contain any terms that do not comply or are inconsistent with the terms of the Contract Documents;
- (r) Include:
  - (i) a covenant acknowledging that, subject to the rights of the Collateral Agent under any Direct Agreement, upon receipt of written Notice from City, City is entitled to exercise step-in rights with respect to the Key Contract (where City is also

exercising its step-in rights under <u>Section 16.2.5</u> (City Step-in Rights) of this Agreement), without any necessity for a consent or approval from Principal Project Company or the making of a determination whether City validly exercised its step-in rights; and

- a waiver and release by Principal Project Company of any claim or cause of action against the Key Contractor arising out of, relating to or resulting from its recognition of City's step-in rights in reliance on any such written Notice from City;
- (s) Include a covenant that will survive termination of the Key Contract obligating the Key Contractor to promptly execute and deliver to City or its successor, assign or designee a new contract between the Key Contractor and City or its successor, assign or designee on the same terms as the Key Contract, if (i) the Key Contract is rejected by Principal Project Company in bankruptcy or is wrongfully terminated by Principal Project Company and (ii) City delivers a request for such new contract within 60 days following termination or expiration of this Agreement;
- (t) Include a covenant that will survive termination of the Key Contract to the effect that if the Key Contractor was a party to an escrow agreement for an IP Escrow and Principal Project Company terminates it, then the Key Contractor also shall execute and deliver to City, concurrently with such new contract, a new escrow agreement on the same terms as the terminated escrow agreement, and shall concurrently make the same deposits to the new IP Escrow as made or provided under the terminated escrow agreement. The obligation to include the same terms in each such new contract (including new IP Escrows) is subject to the following exceptions: (i) terms of a Key Contract or IP Escrow agreement rendered moot or inapplicable solely due to change in the identity of the contracting party; and (ii) terms of a Key Contract that must be adjusted due to schedule delay caused solely by Principal Project Company's rejection in bankruptcy or wrongful termination;
- (u) Require the Key Contractor to (i) maintain usual and customary Books and Records for the type and scope of operations of business in which it is engaged and retain such Books and Records for the period stated in <u>Section 21.1.1(e)</u> of this Agreement or other applicable period specified in the Contract Documents, (ii) permit audit of Books and Records by City and (iii) provide progress reports to Principal Project Company appropriate for the type of work it is performing sufficient to enable Principal Project Company to furnish reports required under this Agreement;
- Include a right of inspection for City, or City's designee(s), consistent with City's inspection rights under the Contract Documents;
- (w) Include provisions for Renewal Work during the last two years of the Term ensuring that warranties and guaranties under each Key Contract inure to the benefit of both City and Principal Project Company; and
- (x) Provide that any purported amendment contrary to the requirements of this <u>Exhibit 14</u> (Key Contract Provisions), without the prior written consent of City, shall be null and void.

## II. D&C Contract

Each D&C Contract shall:

- (b) Ensure D&C Contractor participation in and compliance with commissioning tasks, including the requirements under <u>Division 6</u> (Testing & Commissioning and Operational Readiness) of the Technical Requirements, to provide a properly functioning building that includes fundamental commissioning requirements; and
- (c) At a minimum, outlining the following commissioning requirements, in accordance with <u>Division 6</u> (Testing & Commissioning and Operational Readiness) of the Technical Requirements:
  - (i) Commissioning team roles and responsibilities;
  - (ii) Requirements for a communication protocol between Principal Project Company, D&C Contractor, and the Commissioning Provider;
  - (iii) Submittal requirements and review procedures;
  - (iv) Operation and maintenance documentation requirements;
  - (v) Meetings;
  - (vi) Construction verification procedures;
  - (vii) Cost of retesting;
  - (viii) Start-up, testing, adjusting and balancing documentation and verification;
  - (ix) Functional performance testing requirements;
  - (x) Systems Manual requirements;
  - (xi) Training of IFM Provider and City personnel;
  - (xii) Schedule and contractual milestones;
  - (xiii) End of warranty site visit; and
  - (xiv) Commissioning specifications are to be provided to the D&C Contractor.

## **CONDITIONS PRECEDENT**

- Exhibit 15A: Conditions to NTP 1 Commencement of Non-Construction Work
- Exhibit 15B: Conditions to NTP 2 Commencement of Construction Work
- Exhibit 15C: Conditions to Substantial Completion
- Exhibit 15D: Conditions to Final Acceptance

# EXHIBIT 15A

#### **CONDITIONS TO NTP1 - COMMENCEMENT OF NON-CONSTRUCTION WORK**

The conditions to NTP 1 are:

- (a) Financial Close has occurred;
- (b) City has accepted the following elements of the Project Management Plan: (i) the Design Management Plan; and (ii) the Quality Management Plan (Design);
- (c) all Insurance Policies required to be in effect at NTP 1 pursuant to <u>Exhibit 7</u> (Insurance Requirements) have been obtained and are in full force and effect and Principal Project Company has delivered to City verification thereof as required under <u>Section 10.1.2.4(a)</u> of this Agreement;
- (d) Principal Project Company has certified to City that all personnel who will perform D&C Work either hold all licenses, certifications, registrations, permits or approvals necessary for performance of the D&C Work or will obtain them before starting work;
- (e) Principal Project Company is not then in receipt of any Notice of PPC Default from City unless any such default has been cured or waived in writing by City;
- (f) Principal Project Company is not then in receipt of any Notice of default delivered pursuant to the Financing Documents unless any such default has been cured, and no Lender has otherwise indicated that it is unwilling or unable to presently fund Principal Project Company's costs of the Work; and
- (g) All representations and warranties of Principal Project Company in <u>Section 19.1</u> (Principal Project Company Representations and Warranties) of this Agreement shall be and remain true and correct in all material respects, and Principal Project Company has delivered to City a certificate certifying to the same.

# EXHIBIT 15B

### **CONDITIONS TO NTP 2 - COMMENCEMENT OF CONSTRUCTION WORK**

The conditions to NTP 2 are:

- (a) City has issued NTP 1;
- (b) City has accepted the Project Schedule;
- (c) City has accepted the Project Management Plan;
- (d) City's has accepted the Health and Safety Plan required pursuant to <u>Section 01 35 45</u> of <u>Division 10</u> of the Technical Requirements;
- (e) all Insurance Policies required to be in effect at NTP 1 pursuant to <u>Exhibit 7</u> (Insurance Requirements) remain in full force and effect, and Principal Project Company has delivered to City verification thereof as required under <u>Section 10.1.2.4(a)</u> of this Agreement;
- (f) all Regulatory Approvals necessary to begin the applicable portions of the Construction Work have been obtained and Principal Project Company has furnished to City fully executed copies of such Regulatory Approvals other than the CEQA Approval;
- (g) the Access Date has occurred and all rights of access necessary for commencement of Construction Work on the applicable portion of the Project Site have been obtained;
- (h) all applicable pre-construction requirements, as set forth in the final MMRP contained in <u>Section 01 35 50 (Appendix A)</u> of <u>Division 10</u> of the Technical Requirements and contained in <u>Section 01 35 43</u>, <u>Section 01 35 50</u>, <u>Section 02 80 13</u>, and <u>Section 02 81 10</u> of <u>Division 10</u> of the Technical Requirements, have been reviewed and confirmed to be completed by City in its regulatory capacity;
- (i) all applicable pre-construction requirements contained in any Regulatory Approvals, in each case for the applicable portion of the Construction Work, have been satisfied;
- (j) All Utility Adjustments have been completed and Principal Project Company certifies that, other than Utility Adjustments arising out of any Unidentified Utilities found after the Setting Date, no further Utility conflicts exist with respect to the Project;
- (k) Principal Project Company has obtained approvals from Authorities Having Jurisdiction required for, as well as City approval of, any proposed lane closures, and has taken other appropriate measures to ensure maintenance of traffic in the area affected by the Work;
- (I) Principal Project Company has delivered to City, and City has accepted or approved (as applicable), all Submittals relating to the applicable portion of the Construction Work required by the Project Management Plan and the Contract Documents to be accepted or

approved, in the form and content required by the Project Management Plan or Contract Documents;

- (m) Principal Project Company has delivered to City, and City has approved, the Final Commissioning Plan;
- Principal Project Company has obtained City approval of the Release for Construction Documents for the affected Construction Work in accordance with <u>Section 1.8.6</u> of <u>Division 1</u> of the Technical Requirements;
- (o) the guarantees in favor of City, if any, required under <u>Section 10.5</u> (Guarantees) of this Agreement have been executed, obtained and delivered to, and received by, City and are in full force and effect;
- (p) Principal Project Company is not then in receipt of any Notice of PPC Default from City unless any such default has been cured or waived in writing by City;
- (q) Principal Project Company is not then in receipt of any Notice of default delivered pursuant to the Financing Documents unless any such default has been cured, and no Lender has otherwise indicated that it is unwilling or unable to presently fund Principal Project Company's costs of the Work; and
- (r) All representations and warranties of Principal Project Company in <u>Section 19.1</u> (Principal Project Company Representations and Warranties) of this Agreement shall be and remain true and correct in all material respects, and Principal Project Company has delivered to City a certificate certifying to the same.

## EXHIBIT 15C

## CONDITIONS TO SUBSTANTIAL COMPLETION

The conditions to Substantial Completion are:

- (a) Principal Project Company has completed all D&C Work in accordance with the requirements of this Agreement and the Infrastructure Facility can be utilized safely for its intended purpose, including (i) full access to all points of entry and exit and (ii) completion of all Construction Work other than Punch List items approved by City;
- (b) each Authority Having Jurisdiction has issued a temporary certificate of occupancy for the Infrastructure Facility, to the extent such certificate is required by applicable Law, or has accepted the Infrastructure Facility, as applicable;
- (c) Principal Project Company has accepted the updated IFM Management Plan;
- (d) the Infrastructure Facility is in a condition of full operational functionality and operational readiness to allow the SFMTA's transit operations to relocate to the Infrastructure Facility as required in <u>Section 01 77 00</u> of <u>Division 10</u> and <u>Division 6</u> of the Technical Requirements, respectively including with all emergency testing and commissioning activities successfully completed, and with the Help Desk established and operating;
- (e) Principal Project Company has:
  - (i) completed training of Principal Project Company's IFM Services personnel in accordance with <u>Section 1.7</u> of <u>Division 1</u> of the Technical Requirements;
  - (ii) completed training of City's SFMTA O&M Services personnel in accordance with <u>Section 1.7</u> of <u>Division 1</u> of the Technical Requirements;
  - delivered to City a certificate, in form acceptable to City, executed by Principal Project Company that it and its Contractors are fully staffed with such trained personnel and are ready, willing and able to operate and maintain the Infrastructure Facility in accordance with the terms of the Contract Documents including the approved Project Management Plan;
  - (iv) delivered to City training records evidencing compliance with training requirements for both Principal Project Company's IFM Services personnel and City's SFMTA O&M Services personnel, including copies of course completion certificates issued to each of the subject personnel, and including SFMTA O&M Services training manuals and video recordings of training sessions;
  - (v) completed and documented completion of all training required to allow full access to the Project Site to those individuals designated by City in accordance <u>Section 1.7</u> of <u>Division 1</u> of the Technical Requirements; and
  - (vi) satisfactorily demonstrated integrated operational functionality through "live" coordinated responses (in conjunction with City staff including emergency response

personnel) to failure management and other emergency events during the operations of the Infrastructure Facility in accordance with <u>Division 6</u> of the Technical Requirements;

- (f) the relevant systems and equipment have passed all required tests and Principal Project Company has delivered to City all reports, data, and documentation relating to such tests;
- (g) Principal Project Company has delivered to City, and City has approved, the Systems Manual;
- (h) Principal Project Company has, in accordance with <u>Section 21.4</u> (Intellectual Property), granted to City all Base License Rights to PPC IP and Third Party IP, delivered to City all Developed IP, delivered to City all IP Materials, and made all deposits to the IP Escrow required with respect to the D&C Work;
- (i) Principal Project Company has prepared and submitted a Punch List in accordance with the Contract Documents and City has accepted such list;
- (j) Principal Project Company is not then in receipt of any Default Notice from City unless any such default has been cured or waived in writing by City;
- (k) Principal Project Company is not then in receipt of any Notice of default delivered pursuant to the Financing Documents unless any such default has been cured;
- Principal Project Company has delivered to City (i) all manufacturers' warranties required under, and in the form and content specified by the Technical Requirements (including <u>Division 3</u> of the Technical Requirements and <u>Section 01 78 36</u> of <u>Division 10</u> of the Technical Requirements) and (ii) all documents and other evidence of warranties under <u>Sections 7.9</u> (Final Acceptance) and <u>6.11</u> (Warranties) of this Agreement;
- (m) all Insurance Policies required to be in effect for the IFM Period pursuant to <u>Exhibit 7</u> (Insurance Requirements) have been obtained and are in full force and effect and Principal Project Company has delivered to City verification thereof as required under <u>Section 10.1.2.4(a)</u> of this Agreement; and
- (n) all Submittals required by the Project Management Plan or Contract Documents to be submitted, accepted and/or approved by City have been submitted to and accepted or approved by City, as applicable.

## EXHIBIT 15D

## **CONDITIONS TO FINAL ACCEPTANCE**

The conditions to Final Acceptance are:

- (a) Principal Project Company has completed all D&C Work in accordance with this Agreement;
- (b) City has issued a Certificate of Substantial Completion for the Infrastructure Facility;
- (c) all Punch List items have been completed to the reasonable satisfaction of City;
- Principal Project Company has delivered to City a reasonable inventory of all spare parts, spare components, spare equipment, special tools, materials, expendables and consumables necessary for the operation and maintenance of the Infrastructure Facility;
- (e) all Submittals that Principal Project Company is required by the Contract Documents to submit upon Final Acceptance have been submitted to City;
- (f) each Authority Having Jurisdiction has issued a certificate of occupancy, to the extent such certificate is required by applicable Law, and/or has provided other approvals required for operation of the Infrastructure Facility, as applicable;
- (g) City has received a complete set of the As-Built Documents and documentation for the Infrastructure Facility;
- (h) Principal Project Company has, in accordance with <u>Section 21.4</u> (Intellectual Property), granted to City all Base License Rights to PPC IP and Third Party IP, delivered to City all Developed IP, delivered to City all IP Materials, and made all deposits to the IP Escrow required at or prior to Final Acceptance;
- (i) all Insurance Policies required to be in effect for the IFM Period pursuant to <u>Exhibit 7</u> (Insurance Requirements) have been obtained and are in full force and effect and Principal Project Company has delivered to City verification thereof as required under <u>Section 10.1.2.4(a)</u> of this Agreement;
- (j) Principal Project Company has completed the Move-In, in accordance with <u>Section 7.13</u> (Move-In) of this Agreement.
- (k) Principal Project Company is not then in receipt of any Notice of PPC Default from City unless any such default has been cured or waived in writing by City; and
- (I) Principal Project Company is not then in receipt of any Notice of default delivered pursuant to the Financing Documents unless any such default has been cured.

## EXHIBIT 16

## FEDERAL, STATE AND CITY REQUIREMENTS

- Exhibit 16A: Federal Requirements
- Exhibit 16B: State Requirements
- Exhibit 16C: City Requirements
- Exhibit 16D: SFMTA's Surveillance Technology Policy

## EXHIBIT 16A

## FEDERAL REQUIREMENTS

## [NOTE TO PNC: FEDERAL PROVISIONS UNDER DEVELOPMENT AND SUBJECT TO ADDITIONAL REVISION]

#### 1. NO FEDERAL GOVERNMENT OBLIGATIONS TO THIRD PARTIES

Principal Project Company and City acknowledge and agree that, notwithstanding any concurrence by the federal government in, or approval of, the solicitation or award of this Agreement, absent the express written consent by the federal government, the federal government is not a party to this Agreement and shall not be subject to any obligations or liabilities to City, Principal Project Company or any other party pertaining to any matter resulting from this Agreement. Principal Project Company agrees to include the above clause in each Contract financed in whole or in part with federal assistance provided by the FTA. It is further agreed that the clause shall not be modified, except to identify the Contractor who will be subject to its provisions.

## 2. FALSE STATEMENTS OR CLAIMS – CIVIL OR CRIMINAL FRAUD

49 U.S.C. § 5323(I)(1) 31 U.S.C. §§ 3801-3812 18 U.S.C. § 1001 49 C.F.R. part 31

Principal Project Company acknowledges that the provisions of the Program Fraud Civil Remedies Act of 1986, as amended, 31 U.S.C. § 3801 et seq. and U.S. DOT regulations, "Program Fraud Civil Remedies," 49 C.F.R. part 31, apply to its actions pertaining to the Project. Upon execution of this Agreement, Principal Project Company certifies or affirms the truthfulness and accuracy of any statement it has made, it makes, it may make, or causes to be made, pertaining to this Agreement or the Project. In addition to other penalties that may be applicable, Principal Project Company further acknowledges that if it makes, or causes to be made, a false, fictitious, or fraudulent claim, statement, submission, or certification, the federal government reserves the right to impose the penalties of the Program Fraud Civil Remedies Act of 1986 on Principal Project Company to the extent the federal government deems appropriate. Principal Project Company also acknowledges that if it makes, or causes to be made, a false, fictitious, or fraudulent claim, statement, submission, or certification to the federal government under a contract connected with a project that is financed in whole or in part with federal assistance originally awarded by FTA under authority of 49 U.S.C. chapter 53, the Government reserves the right to impose the penalties of 18 U.S.C. § 1001 and 49 U.S.C. § 5323(I) on Principal Project Company, to the extent the federal government deems appropriate. Principal Project Company agrees to include the above two clauses in each Contract financed in whole or in part with federal assistance provided by FTA. It is further agreed that the clauses shall not be modified, except to identify the Contractor who will be subject to the provisions.

## 3. ACCESS TO THIRD PARTY CONTRACT RECORDS

49 U.S.C. § 5325(g) 2 C.F.R. § 200.333 49 C.F.R. part 633

- (a) Record Retention. Principal Project Company will retain, and will require its Contractors of all tiers to retain, complete and readily accessible records related in whole or in part to this Agreement, including, but not limited to, data, documents, reports, statistics, sub-agreements, leases, subcontracts, arrangements, other third party agreements of any type, and supporting materials related to those records.
- (b) Retention Period. Principal Project Company agrees to comply with the record retention requirements in accordance with 2 C.F.R. § 200.333. Principal Project Company shall maintain all books, records, accounts and reports required under this Agreement for a period of at not less than three (3) years after the date of termination or expiration of this Agreement, except in the event of litigation or settlement of claims arising from the performance of this Agreement, in which case records shall be maintained until the disposition of all such litigation, appeals, claims or exceptions related thereto.
- (c) Access to Records. Principal Project Company agrees to provide sufficient access to FTA and its contractors to inspect and audit records and information related to performance of this Agreement as reasonably may be required and to the U.S. Secretary of Transportation and the Comptroller General of the United states, the state, or their duly authorized representatives, access to all third party contract records (at any tier) as required under 49 U.S.C. § 5325(g).
- (d) Access to the Sites of Performance. Principal Project Company agrees to permit FTA and its contractors access to the sites of performance under this Agreement as reasonably may be required.

#### 4. CHANGES TO FEDERAL REQUIREMENTS

- (a) Principal Project Company shall at all times comply with all applicable FTA regulations, policies, procedures and directives, as these regulations, policies, procedures, and directives may be amended from time to time, including those listed directly or by reference in any Master Agreement between City and FTA. Principal Project Company's failure to so comply shall constitute a material breach of this Agreement.
- (b) Federal requirements that apply to City or the award, this Agreement, and any amendments thereto may change due to changes in federal law, regulation, other requirements, or guidance, or changes in the City's underlying agreement including any information incorporated by reference and made part of that underlying agreement.
- (c) Applicable changes to those federal requirements will apply to this Agreement and Parties thereto at any tier.

#### 5. TERMINATION

2 C.F.R. § 200.339 2 C.F.R. App. II(B) to part 200

See <u>Article 17</u> (Termination) of this Agreement.

#### 6. CIVIL RIGHTS

Principal Project Company is an Equal Opportunity Employer. As such, Principal Project Company agrees to comply with all applicable federal civil rights laws and implementing regulations. Apart from inconsistent requirements imposed by federal laws or regulations, Principal Project Company agrees to comply with the requirements of 49 U.S.C. § 5323(h)(3) by not using any federal assistance awarded by FTA to support procurements using exclusionary or discriminatory specifications. Under this Agreement, Principal Project Company shall at all times comply with the following requirements and shall include these requirements in each Contract entered into as part thereof.

- (a) **Nondiscrimination**. In accordance with federal transit law at 49 U.S.C. § 5332, Principal Project Company agrees that it will not discriminate against any employee or applicant for employment because of race, color, religion, national origin, sex, disability, or age; and that the Principal Project Company and each Contractor maintains no employee facilities segregated on the basis of race, color, religion or national origin. In addition, Principal Project Company agrees to comply with applicable federal implementing regulations and other implementing requirements FTA may issue.
- Race, Color, Religion, National Origin, Sex. In accordance with Title VII of the (b) Civil Rights Act, as amended, 42 U.S.C. § 2000e et seq., and federal transit laws at 49 U.S.C. § 5332, Principal Project Company agrees to comply with all applicable equal employment opportunity requirements of U.S. Department of Labor (U.S. DOL) regulations, "Office of Federal Contract Compliance Programs, Equal Employment Opportunity, Department of Labor," 41 C.F.R. chapter 60, and Executive Order No. 11246, "Equal Employment Opportunity in Federal Employment," September 24, 1965, 42 U.S.C. § 2000e note, as amended by any later Executive Order that amends or supersedes it, referenced in 42 U.S.C. § 2000e note. Principal Project Company agrees to take affirmative action to ensure that applicants are employed, and that employees are treated during employment, without regard to their race, color, religion, national origin, or sex (including sexual orientation and gender identity). Such action shall include, but not be limited to, the following: employment, promotion, demotion or transfer, recruitment or recruitment advertising, layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. In addition, Principal Project Company agrees to comply with any implementing requirements FTA may issue.
- (c) **Nondiscrimination Title VI of the Civil Rights Act**. Principal Project Company will:
  - (i) prohibit discrimination based on race, color, or national origin;

- (ii) comply with: (A) Title VI of the Civil Rights Act of 1964, as amended, 42 U.S.C. § 2000d, et seq.; (B) U.S. DOT regulations, "Nondiscrimination in Federally-Assisted Programs of the Department of Transportation Effectuation of Title VI of the Civil Rights Act of 1964," 49 C.F.R. Part 21; and (C) Federal transit law, specifically 49 U.S.C. § 5332; and
- (iii) follow: (A) The most recent edition of FTA Circular 4702.1, "Title VI Requirements and Guidelines for Federal Transit Administration Recipients," to the extent consistent with applicable federal laws, regulations, requirements, and guidance; (B) U.S. DOJ, "Guidelines for the enforcement of Title VI, Civil Rights Act of 1964," 28 C.F.R. § 50.3; and (C) All other applicable federal guidance that may be issued.

For more information on Title VI of the Civil Rights Act requirements, Principal Project Company shall review City's Title VI Program developed for the Project to comply with Title VI of the Civil Rights Act of 1964.

- (d) Age. In accordance with the Age Discrimination in Employment Act, 29 U.S.C. §§ 621- 634, U.S. Equal Employment Opportunity Commission (U.S. EEOC) regulations, "Age Discrimination in Employment Act," 29 C.F.R. part 1625, the Age Discrimination Act of 1975, as amended, 42 U.S.C. § 6101 et seq., U.S. Health and Human Services regulations, "Nondiscrimination on the Basis of Age in Programs or Activities Receiving Federal Financial Assistance," 45 C.F.R. part 90, and Federal transit law at 49 U.S.C. § 5332, Principal Project Company agrees to refrain from discrimination against present and prospective employees for reason of age. In addition, Principal Project Company agrees to comply with any implementing requirements FTA may issue.
- (e) Disabilities. In accordance with section 504 of the Rehabilitation Act of 1973, as amended, 29 U.S.C. § 794, the Americans with Disabilities Act of 1990, as amended, 42 U.S.C. § 12101 et seq., the Architectural Barriers Act of 1968, as amended, 42 U.S.C. § 4151 et seq., and Federal transit law at 49 U.S.C. § 5332, Principal Project Company agrees that it will not discriminate against individuals on the basis of disability. In addition, Principal Project Company agrees to comply with any implementing requirements FTA may issue.
- (f) **Department of Transportation Funding Requirements**. To ensure compliance with requirements applicable to agreements funded in whole or in part by USDOT funds the following requirements shall apply to this Agreement:
  - (i) Principal Project Company and each Contractor shall not discriminate on the basis of race, color, national origin, or sex in the award and performance of this Agreement.
  - (ii) Principal Project Company and each Contractor shall carry out applicable requirements of 49 C.F.R. Part 26, and shall take all necessary and reasonable steps under 49 C.F.R. Part 26 to ensure nondiscrimination in the award and administration of USDOT-assisted Contracts.

(iii) Failure by Principal Project Company and any of its Contractors to carry out the requirements of this <u>Section 6(f)</u> (Department of Transportation Funding Requirements) is a material breach of this Agreement, which may result in the termination of this Agreement or such other remedy as the City deems appropriate, which may include but which are not limited to: withholding of monthly payments, assessment of sanctions, liquidated damages, and/or disqualifying Principal Project Company from future work as non-responsible.

#### 7. SPECIAL DEPARTMENT OF LABOR EQUAL EMPLOYMENT OPPORTUNITY CLAUSE FOR CONSTRUCTION PROJECTS

During the performance of this Agreement, Principal Project Company agrees as follows:

- (a) Principal Project Company will not discriminate against any employee or applicant for employment because of race, color, religion, sex, sexual orientation, gender identity, or national origin. Principal Project Company will take affirmative action to ensure that applicants are employed, and that employees are treated during employment without regard to their race, color, religion, sex, sexual orientation, gender identity, or national origin. Such action shall include, but not be limited to the following: Employment, upgrading, demotion, or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. Principal Project Company agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided setting forth the provisions of this nondiscrimination clause.
- (b) Principal Project Company will, in all solicitations or advertisements for employees placed by or on behalf of Principal Project Company, state that all qualified applicants will receive consideration for employment without regard to race, color, religion, sex, sexual orientation, gender identity, or national origin.
- (c) Principal Project Company will not discharge or in any other manner discriminate against any employee or applicant for employment because such employee or applicant has inquired about, discussed, or disclosed the compensation of the employee or applicant or another employee or applicant. This provision shall not apply to instances in which an employee who has access to the compensation information of other employees or applicants as a part of such employee's essential job functions discloses the compensation of such other employees or applicants to individuals who do not otherwise have access to such information, unless such disclosure is in response to a formal complaint or charge, in furtherance of an investigation, proceeding, hearing, or action, including an investigation conducted by the employer, or is consistent with Principal Project Company's legal duty to furnish information.
- (d) Principal Project Company will send to each labor union or representative of workers with which it has a collective bargaining agreement or other contract or understanding, a notice to be provided advising the said labor union or workers' representatives of Principal Project Company's commitments under this section,

and shall post copies of the notice in conspicuous places available to employees and applicants for employment.

- (e) Principal Project Company will comply with all provisions of Executive Order 11246 of September 24, 1965, and of the rules, regulations, and relevant orders of the Secretary of Labor.
- (f) Principal Project Company will furnish all information and reports required by Executive Order 11246 of September 24, 1965, and by rules, regulations, and orders of the Secretary of Labor, or pursuant thereto, and will permit access to its books, records, and accounts by the administering agency and the Secretary of Labor for purposes of investigation to ascertain compliance with such rules, regulations, and orders.
- (g) In the event of Principal Project Company's noncompliance with the nondiscrimination clauses of this Agreement or with any of the said rules, regulations, or orders, this Agreement may be canceled, terminated, or suspended in whole or in part and Principal Project Company may be declared ineligible for further government contracts or federally assisted construction contracts in accordance with procedures authorized in Executive Order 11246 of September 24, 1965, and such other sanctions may be imposed and remedies invoked as provided in Executive Order 11246 of September 24, 1965, or by rule, regulation, or order of the Secretary of Labor, or as otherwise provided by law.
- (h) Principal Project Company will include the portion of the sentence immediately preceding paragraph (a) and the provisions of paragraphs (a) through (h) in every Contract or purchase order unless exempted by rules, regulations, or orders of the Secretary of Labor issued pursuant to section 204 of Executive Order 11246 of September 24, 1965, so that such provisions will be binding upon each Contractor or vendor. Principal Project Company will take such action with respect to any Contract or purchase order as the administering agency may direct as a means of enforcing such provisions, including sanctions for noncompliance: *Provided*, however, that in the event Principal Project Company becomes involved in, or is threatened with, litigation with a Contractor or vendor as a result of such direction by the administering agency, Principal Project Company may request the United States to enter into such litigation to protect the interests of the United States.

## 8. DISADVANTAGED BUSINESS ENTERPRISES (DBES)

#### 49 C.F.R. part 26

- (a) DBE Goals. City will establish a DBE Program and DBE Program goals for the Project, and will be required to report on DBE participation on a semi-annual basis so that its attainment efforts may be evaluated. Such requirements are in addition to all other equal opportunity employment requirements of this Agreement. Principal Project Company shall comply with the DBE goals that City establishes for the Project and this Agreement.
- (b) **Post-Award Compliance Monitoring**. City will conduct post-award monitoring of Principal Project Company's compliance with the DBE provisions of this

Agreement. Principal Project Company shall cooperate with City requests for assistance with post-award monitoring. Principal Project Company shall maintain records sufficient to document, on an ongoing basis, name of each DBE Contractor, work assignment of each DBE Contractor, DBE commitments, amounts paid to each DBE Contractor during the reporting period, amounts paid to each DBE Contractor during the total commitment to each DBE Contractor, etc., among other information. Principal Project Company will be required to submit supplemental reports on a monthly basis including a monthly DBE report and forecast chart showing planned and actual attainment of DBE Contractors.

- (c) **California Unified Certification Program**. Principal Project Company shall only use, and City shall only accept the use of, DBEs that are certified through the California Unified Certification Program.
- (d) Listed DBEs. Principal Project Company shall utilize the specific DBEs listed in this Agreement unless Principal Project Company obtains City's written consent to terminate or substitute a DBE Contractor. Unless City's consent is provided, Principal Project Company shall not be entitled to any payment for work or material unless it is performed or supplied by the listed DBE.
- (e) Access to Records. City will require Principal Project Company and its Contractors to maintain records and documents of payments to DBEs for three years following the performance of the contract. These records will be made available for inspection upon request by any authorized representative of City or the USDOT. This reporting requirement also extends to any certified DBE Contractor. The authorized representative(s) of City, the USDOT, the Comptroller General of the United States, shall have the right to inspect and audit all data and records of Principal Project Company relating to its performance under this <u>Article</u> <u>8</u> (Disadvantaged Business Enterprises (DBEs)).
- (f) **Special Requirements for a Transit Vehicle Manufacturer**. The transit vehicle manufacturer, as a condition of being authorized to bid or propose on FTA-assisted transit vehicle procurements, must certify that it has complied with the requirements of 49 C.F.R. part 26.
- (g) No Discrimination. Principal Project Company and its Contractors shall not discriminate on the basis of race, color, national origin, or sex in the performance of this Agreement. Principal Project Company shall carry out applicable requirements of 49 C.F.R. Part 26 in the award and administration of USDOTassisted contracts. Failure by Principal Project Company to carry out these requirements is a material breach of this Agreement, which may result in the termination of this Agreement or such other remedy as City deems appropriate, which may include, but is not limited to:
  - (i) Withholding monthly progress payments;
  - (ii) Assessing sanction;
  - (iii) Liquidated damages; and/or

(iv) Disqualifying Principal Project Company from future bidding as non-responsible (49 C.F.R. § 26.13(b).

## 9. INCORPORATION OF FTA TERMS

This Agreement includes, in part, certain standard terms and conditions required by the USDOT and FTA, whether or not expressly set forth in the preceding contract provisions. All contractual provisions required by the USDOT, as set forth in FTA Circular 4220.1F, dated November 1, 2008, as revised by Rev. 1, dated April 14, 2009, Rev. 2, dated July 1, 2010, Rev. 3, dated February 14, 2011, and Rev. 4, dated March 18, 2013, as may be amended, are hereby incorporated by reference. Principal Project Company shall comply with all applicable FTA regulations, policies, procedures and directives, including those listed directly in or referred to in the current FTA Master Agreement. Anything to the contrary herein notwithstanding, all FTA mandated terms shall be deemed to control in the event of a conflict with other provisions contained in this Agreement. Principal Project Company shall not perform any act, fail to perform any act, or refuse to comply with any City requests which would cause the City to be in violation of FTA terms and conditions. Principal Project Company will include these requirements in all Contracts issued pursuant to this Agreement.

#### 10. DEBARMENT AND SUSPENSION

2 C.F.R. part 180 2 C.F.R part 1200 2 C.F.R. § 200.213 2 C.F.R. part 200 Appendix II (I) Executive Order 12549 Executive Order 12689

Principal Project Company shall comply and facilitate compliance with USDOT regulations, "Non-procurement Suspension and Debarment," 2 C.F.R. part 1200, which adopts and supplements the U.S. Office of Management and Budget (U.S. OMB) "Guidelines to Agencies on Government wide Debarment and Suspension (Non-procurement)," 2 C.F.R. part 180. These provisions apply to each contract at any tier of \$25,000 or more, and to each contract at any tier for a federally required audit (irrespective of the contract amount), and to each contract at any tier that must be approved by an FTA official irrespective of the contract amount. As such, Principal Project Company shall verify that its principals, Affiliates, and Contractors are eligible to participate in this federally funded contract and are not presently declared by any federal department or agency to be:

- (i) debarred from participation in any federally assisted award;
- (ii) suspended from participation in any federally assisted award;
- (iii) proposed for debarment from participation in any federally assisted award;
- (iv) declared ineligible to participate in any federally assisted award;
- (v) voluntarily excluded from participation in any federally assisted award; or

(vi) disqualified from participation in ay federally assisted award.

The certification in this clause is a material representation of fact relied upon by City. If it is later determined by City that Principal Project Company knowingly rendered an erroneous certification, in addition to remedies available to City, the federal government may pursue available remedies, including but not limited to suspension and/or debarment. Principal Project Company agrees to comply with the requirements of 2 C.F.R. part 180, subpart C, as supplemented by 2 C.F.R. part 1200 throughout the period of any contract that may arise from this Agreement or the Project. Principal Project Company further agrees to include a provision requiring such compliance in its lower tier covered transactions and agreements.

#### 11. BUY AMERICA

49 U.S.C. 5323(j) 49 C.F.R. part 661 Pub. L. No. 117-58, §§ 70901-17

Principal Project Company agrees to comply with 49 U.S.C. 5323(j), 49 C.F.R. part 661 and Pub. L. No. 117-58, §§ 70901-17, which provide that federal funds may not be obligated unless all steel, iron, and manufactured products used in FTA funded projects are produced in the United States and designated construction materials are manufactured in the United States, unless a waiver has been granted by FTA or the product is subject to a general waiver. General waivers are listed in 49 C.F.R. § 661.7. Separate requirements for rolling stock are set out at 49 U.S.C. 5323(j)(2)(C) and 49 C.F.R. § 661.11.

The Buy America requirements flow down from the City to Principal Project Company, which is responsible for ensuring that lower tier Contractors are in compliance.

## In accordance with 49 C.F.R. § 661.6, for the procurement of steel, iron, manufactured products or construction materials, use the certifications below.

Certificate of Compliance with Buy America Requirements

The bidder or offeror hereby certifies that it will comply with the requirements of 49 U.S.C. 5323(j)(1), the applicable regulations in 49 C.F.R. part 661 and Pub. L. No. 117-58, §§ 70901-17.

Date:		
Signature:		
Company:		
Name:		
Title:		

Certificate of Non-Compliance with Buy America Requirements

The bidder or offeror hereby certifies that it cannot comply with the requirements of 49 U.S.C. 5323(j), but it may qualify for an exception to the requirement pursuant to 49 U.S.C. 5323(j)(2),

as amended, the applicable regulations in 49 C.F.R. § 661.7 and Pub. L. No. 117-58, §§ 70901-17.

Date:
Signature:
Company:
Name:
Title:

# In accordance with 49 C.F.R. § 661.12, for the procurement of rolling stock (including train control, communication, and traction power equipment), use the certifications below.

Certificate of Compliance with Buy America Rolling Stock Requirements

The bidder or offeror hereby certifies that it will comply with the requirements of 49 U.S.C. 5323(j), and the applicable regulations in 49 C.F.R. § 661.11.

Date:			
Signature: _			
Company: _			
Name:			
Title:			

Certificate of Non-Compliance with Buy America Rolling Stock Requirements

The bidder or offeror hereby certifies that it cannot comply with the requirements of 49 U.S.C. 5323(j), but it may qualify for an exception to the requirement pursuant to 49 U.S.C. 5323(j)(2)(C), as amended, and the applicable regulations in 49 C.F.R. § 661.7.



## 12. RESOLUTION OF DISPUTES, BREACHES, OR OTHER LITIGATION

See <u>Article 18</u> (Contract Dispute Procedures) of this Agreement.

## 13. LOBBYING RESTRICTIONS

31 U.S.C. § 1352 2 C.F.R. § 200.450 2 C.F.R. part 200 appendix II (J) 49 C.F.R. part 20

Principal Project Company certifies, to the best of its knowledge and belief, that:

- (a) no federally appropriated funds have been paid or will be paid, by or on behalf of Principal Project Company, to any person for influencing or attempting to influence an officer or employee of an agency, a member of Congress, an officer or employee of Congress, or an employee of a member of Congress in connection with the awarding of any federal contract, the making of any federal grant, the making of any federal loan, the entering into of any cooperative agreement, and the extension, continuation, renewal, amendment, or modification of any federal contract, grant, loan, or cooperative agreement.
- (b) if any funds other than federal appropriated funds have been paid or will be paid to any person for influencing or attempting to influence an officer or employee of any agency, a member of Congress, an officer or employee of Congress, or an employee of a member of Congress in connection with this federal contract, grant, loan, or cooperative agreement, Principal Project Company shall complete and submit Standard Form-LLL, "Disclosure Form to Report Lobbying," in accordance with its instructions.
- (c) Principal Project Company shall require that the language of this certification be included in the award documents for all sub-awards at all tiers (including Contracts, sub-grants, and Subcontracts under grants, loans, and cooperative agreements) and that all sub recipients shall certify and disclose accordingly.

This certification is a material representation of fact upon which reliance was placed when this transaction was made or entered into. Submission of this certification is a prerequisite for making or entering into this transaction imposed by section 1352, title 31, U.S. Code. Any person who fails to file the required certification shall be subject to a civil penalty of not less than \$10,000 and not more than \$100,000 for each such failure.

#### 14. CLEAN AIR ACT AND CLEAN WATER ACT

42 U.S.C. §§ 7401 – 7671q 33 U.S.C. §§ 1251-1387 2 C.F.R. part 200, Appendix II (G)

The Clean Air and Federal Water Pollution Control Act requirements apply to each contract and subcontract exceeding \$150,000.

Principal Project Company agrees:

(a) it will not use any violating facilities;

- (b) it will report the use of facilities placed on or likely to be placed on the U.S. EPA "List of Violating Facilities";
- (c) it will report violations of use of prohibited facilities to the FTA; and
- (d) it will comply with the inspection and other requirements of the Clean Air Act, as amended, (42 U.S.C. §§ 7401 7671q) and the Federal Water Pollution Control Act, as amended, (33 U.S.C. §§ 1251-1387).

#### 15. FLY AMERICA

49 U.S.C. § 40118 41 C.F.R. part 301-10 48 C.F.R. part 47.4

- (a) As used in this clause "international air transportation" means transportation by air between a place in the United States and a place outside the United States or between two places both of which are outside the United States. "United States" means the 50 States, the District of Columbia, and outlying areas. "U.S.-flag air carrier" means an air carrier holding a certificate under 49 U.S.C. Chapter 411.
- (b) When federal funds are used to fund travel, Section 5 of the International Air Transportation Fair Competitive Practices Act of 1974 (49 U.S.C. 40118) (Fly America Act) requires contractors, recipients, and others use U.S.-flag air carriers for U.S. government-financed international air transportation of personnel (and their personal effects) or property, to the extent that service by those carriers is available. It requires the Comptroller General of the United States, in the absence of satisfactory proof of the necessity for foreign-flag air transportation, to disallow expenditures from funds, appropriated or otherwise established for the account of the United States, for international air transportation secured aboard a foreign-flag air carrier if a U.S.-flag air carrier is available to provide such services.
- (c) if available, Principal Project Company, in performing work under this Agreement, shall use U.S.-flag carriers for international air transportation of personnel (and their personal effects) or property.
- (d) In the event that Principal Project Company selects a carrier other than a U.S.-flag air carrier for international air transportation, Principal Project Company shall include a statement on vouchers involving such transportation essentially as follows:

#### Statement of Unavailability of U.S.-Flag Air Carriers

International air transportation of persons (and their personal effects) or property by U.S.flag air carrier was not available or it was necessary to use foreign-flag air carrier service for the following reasons. See FAR § 47.403.

[State reasons]: \_\_\_\_

(End of statement)

(e) Principal Project Company shall include the substance of this clause, including this paragraph (e), in each Contract or purchase under this Agreement that may involve international air transportation.

## 16. EMPLOYEE PROTECTIONS

49 U.S.C. § 5333(a) 40 U.S.C. §§ 3141-3148 29 C.F.R. part 5 18 U.S.C. § 847 29 C.F.R. part 3 40 U.S.C. §§ 3701-3708 29 C.F.R. part 1926

## 16.1 PREVAILING WAGE AND COPELAND ANTI-KICKBACK ACT

For all prime construction, alteration or repair contracts in excess of \$2,000 awarded by FTA. Principal Project Company shall comply with the Davis-Bacon Act and the Copeland "Anti-Kickback" Act. Under 49 U.S.C. § 5333(a), prevailing wage protections apply to laborers and mechanics employed on FTA assisted construction, alteration, or repair projects. Principal Project Company will comply with the Davis-Bacon Act, 40 U.S.C. §§ 3141-3144, and 3146-3148 as supplemented by DOL regulations at 29 C.F.R. part 5, "Labor Standards Provisions Applicable to Contracts Governing Federally Financed and Assisted Construction." In accordance with the statute, Principal Project Company shall pay wages to laborers and mechanics at a rate not less than the prevailing wages specified in a wage determination made by the Secretary of Labor. In addition, Principal Project Company agrees to pay wages not less than once a week. Principal Project Company shall also comply with the Copeland "Anti-Kickback" Act (40 U.S.C. § 3145), as supplemented by DOL regulations at 29 C.F.R. part 3, "Contractors and Subcontractors on Public Building or Public Work Financed in Whole or in part by Loans or Grants from the United States." Principal Project Company is prohibited from inducing, by any means, any person employed in the construction, completion, or repair of public work, to give up any part of the compensation to which he or she is otherwise entitled.

## 16.2 CONTRACT WORK HOURS AND SAFETY STANDARDS ACT

For all contracts in excess of \$100,000 that involve the employment of mechanics or laborers, Principal Project Company shall comply with the Contract Work Hours and Safety Standards Act (40 U.S.C. §§ 3701-3708), as supplemented by the DOL regulations at 29 C.F.R. part 5. Under 40 U.S.C. § 3702 of the Act, Principal Project Company shall compute the wages of every mechanic and laborer, including watchmen and guards, on the basis of a standard work week of 40 hours. Work in excess of the standard work week is permissible provided that the worker is compensated at a rate of not less than one and a half times the basic rate of pay for all hours worked in excess of 40 hours in the work week. The requirements of 40 U.S.C. § 3704 and DOL regulations, "Recording and Reporting Occupational Injuries and Illnesses," 29 C.F.R. part 1904; "Occupational Safety and Health Standards," 29 C.F.R. part 1910; and "Safety and Health Regulations for Construction," 29 C.F.R. part 1926 are applicable to construction work and provide that no laborer or mechanic be required to work in surroundings or under working conditions which are unsanitary, hazardous or dangerous. These requirements do not apply to the purchase of supplies or materials or articles ordinarily available on the open market, or to contracts for transportation or transmission of intelligence.

In the event of any violation of the clause set forth herein, Principal Project Company and any Contractor responsible therefor shall be liable for the unpaid wages. In addition, Principal Project Company and Contractor shall be liable to the United States (in the case of work done under contract for the District of Columbia or a territory, to such District or to such territory), for liquidated damages. Such liquidated damages shall be computed with respect to each individual laborer or mechanic, including watchmen and guards, employed in violation of this clause in the sum of \$10 for each calendar day on which such individual was required or permitted to work in excess of the standard workweek of forty hours without payment of the overtime wages required by this clause.

The FTA shall upon its own action or upon written request of an authorized representative of the Department of Labor withhold or cause to be withheld, from any moneys payable on account of work performed by Principal Project Company or any Contractor under this Agreement or any Contract entered into in furtherance thereof or any other federal contract with Principal Project Company, or any other federally-assisted contract subject to the Contract Work Hours and Safety Standards Act, which is held by Principal Project Company, such sums as may be determined to be necessary to satisfy any liabilities of Principal Project Company or any Contractor for unpaid wages and liquidated damages as provided in this section.

Principal Project Company or any Contractor shall insert in any Contracts the clauses set forth in this section and also a clause requiring the Contractors to include these clauses in any lower tier Contracts. Principal Project Company shall be responsible for compliance by any Contractor or lower tier Contractor with the clauses set forth in this Agreement.

#### Contract Work Hours and Safety Standards for Awards Not Involving Construction

Principal Project Company shall comply with all federal laws, regulations, and requirements providing wage and hour protections for non-construction employees, in accordance with 40 U.S.C. § 3702, Contract Work Hours and Safety Standards Act, and other relevant parts of that Act, 40 U.S.C. § 3701 *et seq.*, and U.S. DOL regulations, "Labor Standards Provisions Applicable to Contracts Covering Federally Financed and Assisted Construction (also Labor Standards Provisions Applicable to Non-construction Contracts Subject to the Contract Work Hours and Safety Standards Act)," 29 C.F.R. part 5.

Principal Project Company shall maintain payrolls and basic payroll records during the course of the work and shall preserve them for a period of three (3) years from the completion of this Agreement for all laborers and mechanics, including guards and watchmen, working on this Agreement. Such records shall contain the name and address of each such employee, social security number, correct classifications, hourly rates of wages paid, daily and weekly number of hours worked, deductions made, and actual wages paid.

Such records maintained under this paragraph shall be made available by Principal Project Company for inspection, copying, or transcription by authorized representatives of the FTA and the Department of Labor, and Principal Project Company will permit such representatives to interview employees during working hours on the job.

Principal Project Company shall require the inclusion of the language of this clause within Contracts of all tiers.

#### 16.3 AWARDS INVOLVING COMMERCE

Principal Project Company agrees to comply with the Fair Labor Standards Act (FLSA), 29 U.S.C. § 201 *et seq.* to the extent that the FLSA applies to employees performing work involving commerce, and as the federal government otherwise determines applicable.

#### 17. BONDING REQUIREMENTS

2 C.F.R. 200.326

See <u>Section 10.2</u> (Performance Security) of this Agreement.

#### 18. SEISMIC SAFETY

42 U.S.C. 7701 *et seq.* 49 C.F.R. part 41 Executive Order (E.O.) 12699

Principal Project Company agrees that any new building or addition to an existing building will be designed and constructed in accordance with the standards for seismic safety required in Department of Transportation (DOT) seismic safety regulations 49 C.F.R. part 41 and will certify to compliance to the extent required by the regulation. Principal Project Company also agrees to ensure that all work performed under this Agreement, including work performed by a Contractor, is in compliance with the standards required by the seismic safety regulations and the certification of compliance issued on the project.

#### **19. ENERGY CONSERVATION**

42 U.S.C. 6321 et seq. 49 C.F.R. part 622, subpart C

These requirements apply to Principal Project Company and extend to all third-party contractors and their contracts at every tier and sub-recipients and their subcontracts at every tier.

Principal Project Company agrees to comply with mandatory standards and policies relating to energy efficiency, which are contained in the state energy conservation plan issued in compliance with the Energy Policy and Conservation Act.

#### 20. RECYCLED PRODUCTS

42 U.S.C. § 6962 40 C.F.R. part 247 2 C.F.R. part § 200.322

These requirements apply to Principal Project Company and extend to all third-party contractors and their contracts and subcontracts at every tier where the value of an EPA designated item exceeds \$10,000.

Principal Project Company agrees to provide a preference for those products and services that conserve natural resources, protect the environment, and are energy efficient by complying with

and facilitating compliance with Section 6002 of the Resource Conservation and Recovery Act, as amended, 42 U.S.C. § 6962, and U.S. Environmental Protection Agency (U.S. EPA), "Comprehensive Procurement Guideline for Products Containing Recovered Materials," 40 C.F.R. part 247.

## 21. ADA ACCESS

<u>Federal Protections for Individuals with Disabilities</u>. The Americans with Disabilities Act of 1990, as amended (ADA), 42 U.S.C. Sections 12101 *et seq.*, prohibits discrimination against qualified individuals with disabilities in programs, activities, and services, and imposes specific requirements on public and private entities. Principal Project Company must comply with its responsibilities under Titles I, II, III, IV, and V of the ADA in employment, public services, public accommodations, telecommunications, and other provisions, many of which are subject to regulations issued by other federal agencies.

## 22. SAFE OPERATION OF MOTOR VEHICLES

23 U.S.C. part 402 Executive Order No. 13043 Executive Order No. 13513 U.S. DOT Order No. 3902.10

The Safe Operation of Motor Vehicles requirements shall apply to Principal Project Company and all Contractors at every tier.

- (a) Seat Belt Use. Principal Project Company is encouraged to adopt and promote on-the-job seat belt use policies and programs for its employees and other personnel that operate company-owned vehicles, company-rented vehicles, or personally operated vehicles. The terms "company-owned" and "company-leased" refer to vehicles owned or leased either by Principal Project Company or City.
- (b) **Distracted Driving**. Principal Project Company agrees to adopt and enforce workplace safety policies to decrease crashes caused by distracted drivers, including policies to ban text messaging while using an electronic device supplied by an employer, and driving a vehicle the driver owns or rents, a vehicle Principal Project Company or Contactor owns, leases, or rents, or a privately-owned vehicle when on official business in connection with the work performed under this Agreement.

## 23. ENVIRONMENTAL PROTECTIONS

- (a) **General**. Principal Project Company will comply with all applicable environmental and resource use laws, regulations, and requirements, and follow applicable guidance, now in effect or that may become effective in the future, including state and local laws, ordinances, regulations, and requirements and follow applicable guidance.
- (b) **National Environmental Policy Act**. An award of federal assistance requires the full compliance with applicable environmental laws, regulations, and requirements. Accordingly, Principal Project Company shall:

- (i) comply and facilitate compliance with federal laws, regulations, and requirements, including, but not limited to: (A) federal transit laws, such as 49 U.S.C. § 5323(c)(2), and 23 U.S.C. § 139; (B) the National Environmental Policy Act of 1969 (NEPA), as amended, 42 U.S.C. §§ 4321, et seq., as limited by 42 U.S.C. § 5159, and CEQ's implementing regulations 40 C.F.R. Part 1500 1508; (C) Joint FHWA and FTA regulations, "Environmental Impact and Related Procedures," 23 C.F.R. Part 771 and 49 C.F.R. Part 622; (D) Executive Order No. 11514, as amended, "Protection and Enhancement of Environmental Quality," March 5, 1970, 42 U.S.C. § 4321 note (35 Fed. Reg. 4247); and (E) other federal environmental protection laws, regulations, and requirements applicable to City or the Project.
- (ii) follow the federal guidance identified herein to the extent that the guidance is consistent with applicable authorizing legislation: (A) Joint FHWA and FTA final guidance, "Interim Guidance on MAP-21 Section 1319, Accelerated Decision-making in Environmental Reviews," January 14, 2013; (B) Joint FHWA and FTA final guidance, "SAFETEA-LU Environmental Review Process (Public Law 109-59)," 71 Fed. Reg. 66576, November 15, 2006; and (C) other federal environmental guidance applicable to the Project.
- (c) **Environmental Justice**. Principal Project Company will promote environmental justice by following:
  - Executive Order No. 12898, "Federal Actions to Address Environmental Justice in Minority Populations and Low-Income Populations," February 11, 1994, 42 U.S.C. § 4321 note, (59 Fed. Reg. 7629, 3 C.F.R. 1994 Comp., p. 859) as well as facilitating compliance with that Executive Order;
  - (ii) U.S. DOT Order 5610.2(a), "Department of Transportation Updated Environmental Justice Order," 77 Fed. Reg. 27534, November 8, 2012; and
  - (iii) the most recent edition of FTA Circular 4703.1, "Environmental Justice Policy Guidance for Federal Transit Administration Recipients," August 15, 2012, to the extent consistent with applicable federal laws, regulations, requirements, and guidance.
- (d) Other Environmental Federal Laws. Principal Project Company will comply or facilitate compliance with all applicable federal laws, regulations, and requirements, and will follow applicable guidance, including, but not limited to, the Clean Air Act, Clean Water Act, Wild and Scenic Rivers Act of 1968 (16 U.S.C. §§ 1271-1287), Coastal Zone Management Act of 1972, the Endangered Species Act of 1973, Magnuson Stevens Fishery Conservation and Management Act, Resource Conservation and Recovery Act, Comprehensive Environmental Response, Compensation, and Liability Act, Executive Order No. 11990 relating to "Protection of Wetlands," and Executive Order No. 11988, as amended, "Floodplain Management."

- (e) **Use of Certain Public Lands**. Principal Project Company will comply with USDOT laws, specifically 49 U.S.C. § 303 (often referred to as "section 4(f)"), and joint FHWA and FTA regulations, "Parks, Recreation Areas, Wildlife and Waterfowl Refuges, and Historic Sites," 23 C.F.R. Part 774, and referenced in 49 C.F.R. Part 622.
- (f) **Historic Preservation**. Principal Project Company will:
  - comply with USDOT laws, including 49 U.S.C. § 303 (often referred to as "section 4(f)"), which requires certain findings be made before an award may be undertaken if it involves the use of any land from a historic site that is on or eligible for inclusion on the National Register of Historic Places;
  - (ii) encourage compliance with the federal historic and archaeological preservation requirements of section 106 of the National Historic Preservation Act, as amended, 54 U.S.C. § 306108;
  - (iii) comply with the Archeological and Historic Preservation Act of 1974, as amended, 54 U.S.C. § 312501, et seq.;
  - (iv) comply with U.S. Advisory Council on Historic Preservation regulations, "Protection of Historic Properties," 36 C.F.R. Part 800; and
  - (v) comply with federal requirements and follow federal guidance to avoid or mitigate adverse effects on historic properties.
- (g) Indian Sacred Sites. Principal Project Company will facilitate compliance with federal efforts to promote the preservation of places and objects of religious importance to American Indians, Eskimos, Aleuts, and Native Hawaiians, and facilitate compliance with the American Indian Religious Freedom Act, 42 U.S.C. § 1996, and Executive Order No. 13007, "Indian Sacred Sites," May 24, 1996, 42 U.S.C. § 3161 note (61 Fed. Reg. 26771).

## 24. FEDERAL TAX LIABILITY AND RECENT FELONY CONVICTIONS

- (a) **Transactions Prohibited**. Principal Project Company must certify that it:
  - (i) does not have any unpaid federal tax liability that has been assessed, for which all judicial and administrative remedies have been exhausted or have lapsed, and that is not being paid in a timely manner pursuant to an agreement with City responsible for collecting the tax liability; and
  - (ii) was not convicted of the felony criminal violation under any federal law within the preceding 24 months from the Effective Date.
- (b) If Principal Project Company cannot so certify, City agrees to refer the matter to FTA and not to enter into any contract with Principal Project Company without FTA's written approval. Principal Project Company shall include this requirement in Contracts at all lower tiers, without regard to the value of such Contracts.

#### 25. NOTIFICATION TO FTA

If a current or prospective legal matter that may affect the federal government emerges, Principal Project Company must notify City, who will promptly notify the FTA Chief Counsel and FTA Regional Counsel for the region in which City is located. Principal Project Company must include a similar notification requirement in its sub-agreements at every tier, for any agreement that is a "covered transaction" according to 2 C.F.R. §§ 180.220 and 1200.220.

- (a) **Types**. The types of legal matters that require notification include, but are not limited to, a major dispute, breach, default, litigation, or naming any federal government as a party to litigation or a legal disagreement in any forum for any reason.
- (b) **Matters**. Matters that may affect any federal government include, but are not limited to, the federal government's interests in the award, this Agreement, and any amendments thereto, or the any federal government's administration or enforcement of federal laws, regulations, and requirements.
- Additional Notice to U.S. DOT Inspector General. Principal Project Company (c) must notify City, who will promptly notify the USDOT Inspector General in addition to the FTA Chief Counsel or Regional Counsel for the Region in which City is located, if Principal Project Company has knowledge of potential fraud, waste, or abuse occurring on a project receiving assistance from FTA. The notification provision applies if a person has or may have submitted a false claim under the False Claims Act, 31 U.S.C. § 3729, et seq., or has or may have committed a criminal or civil violation of law pertaining to such matters as fraud, conflict of interest, bid rigging, misappropriation or embezzlement, bribery, gratuity, or similar misconduct involving federal assistance. This responsibility occurs whether the Project is subject to any agreement between City and FTA. Knowledge, as used in this paragraph, includes, but is not limited to, knowledge of a criminal or civil investigation by a federal, state, or local law enforcement or other investigative agency, a criminal indictment or civil complaint, or probable cause that could support a criminal indictment, or any other credible information in the possession of Principal Project Company. In this paragraph, "promptly" means to refer information without delay and without change.

## 26. PROHIBITION ON CERTAIN TELECOMMUNICATIONS AND VIDEO SURVEILLANCE SERVICES OR EQUIPMENT

- (a) Pursuant to section 889 of the National Defense Authorization Act of 2019 (H.R. 5515 at pp. 282-284; Pub. L. 115-232) (NDAA), and as promulgated at 2 C.F.R. § 200.216, Principal Project Company and any Contractor at any tier shall not procure or obtain the Covered Equipment and Services in the performance of work for the Project or in connection with this Agreement.
- (b) "Covered Equipment and Services" is defined to include any telecommunication or video surveillance equipment, systems, or services produced or provided by any of the following entities, or any subsidiary or affiliate of the following entities:
  - (i) Huawei Technologies Company;

- (ii) ZTE Corporation;
- (iii) Hytera Communications Corporation;
- (iv) Hangzhou Hikivision Digital Technology Company;
- (v) Dahua Technology Company; and
- (vi) any entity that the Secretary of Defense, in consultation with the Director of the National Intelligence or the Director of the Federal Bureau of Investigation, reasonably believes to be an entity owned or controlled by, or otherwise connected to, the government of a covered foreign country.
- (c) The burden of proof for the origin or place of production of telecommunications or video surveillance equipment, systems, or services is the responsibility of Principal Project Company.
- (d) Prior to the use of any telecommunication or video surveillance equipment, systems, or services pursuant to this Agreement, Principal Project Company shall furnish a certification to City stating that the telecommunication or video surveillance equipment, systems, or services are not Covered Equipment and Services pursuant to this <u>Article 26</u> (Prohibition on Certain Telecommunications and Video Surveillance Services or Equipment), 2 C.F.R. § 200.216, and the NDAA.

#### 27. SENSITIVE SECURITY INFORMATION

Principal Project Company must protect, and take measures to ensure that its Contractors at each tier protect "sensitive security information" made available during the administration of this Agreement or any Contract to ensure compliance with 49 U.S.C. Section 40119(b) and implementing USDOT regulations, "Protection of Sensitive Security Information," 49 C.F.R. Part 15, and with 49 U.S.C. Section 114(r) and implementing Department of Homeland Security regulations, "Protection of Sensitive Security Information," 49 C.F.R. Part 15, and with 49 U.S.C. Section 114(r) and implementing Department of Homeland Security regulations, "Protection of Sensitive Security Information," 49 C.F.R. Part 15, and with 49 U.S.C. Section 40 (r) and implementing Department of Homeland Security regulations, "Protection of Sensitive Security Information," 49 C.F.R. Part 15, and with 49 U.S.C. Section 40 (r) and implementing Department of Homeland Security regulations, "Protection of Sensitive Security Information," 49 C.F.R. Part 1520.

#### 28. FLOOD INSURANCE

Principal Project Company agree to comply with flood insurance laws and guidance as follows:

- (a) It will have flood insurance as required by the Flood Disaster Protection Act of 1973, 42 U.S.C. § 4012a(a), for any building located in a special flood hazard area (100-year flood zone), before accessing federal assistance to acquire, construct, reconstruct, repair, or improve that building.
- (b) Each such building and its contents will be covered by flood insurance in an amount at least equal to the federal investment (less estimated land cost) or to the maximum limit of coverage made available with respect to the particular type of property under the National Flood Insurance Act of 1968, 42 U.S.C. § 4001, et seq., whichever is less.

(c) It will follow FTA guidance, except to the extent FTA determines otherwise in writing.

#### 29. PROMPT PAYMENT AND RETENTION

(a) Payment of Contractors. Principal Project Company shall pay its Contractors within 30 calendar days from receipt of each payment made to Principal Project Company by City or any Lender. The 30 calendar days is applicable unless a shorter time period controls as provided in <u>Section 9.4</u> (Prompt Payment to Contractors) of this Agreement. Any delay or postponement of payment over 30 calendar days (or any controlling shorter period) may occur only for good cause and with the prior written approval of City.

#### (b) **Retention of Funds**.

- (i) No standard retention will be withheld by City from payments due to Principal Project Company.
- (ii) Any retention withheld by Principal Project Company or Contractors from Progress Payments due applicable Contractors shall be promptly paid in full to Contractors within 30 days (or any shorter time period that controls as provided in <u>Section 9.4</u> (Prompt Payment to Contractors) of this Agreement) after the Contractor's work is satisfactorily completed. Principal Project Company shall assure that each Contract contains a clause obligating Principal Project Company or the applicable Contractor to make prompt and full payment of any retention kept by Principal Project Company or the applicable Contractor to the Subcontractor within such time period. For this purpose, a Contractor's work is satisfactorily completed when all the tasks called for in the Contract have been accomplished and documented as required by City.
- (c) When City has made an incremental acceptance of a portion of the D&C Work, the Work of a Contractor covered by that acceptance is deemed to be satisfactorily completed. These requirements shall not be construed to limit or impair any contractual, administrative, or judicial remedies otherwise available to Principal Project Company or Contractor in the event of a dispute involving late payment or nonpayment by Principal Project Company, deficient Contractor performance, or noncompliance by a Contractor.
- (d) This Section applies to both DBE and non-DBE Contractors.

## EXHIBIT 16B

#### **STATE REQUIREMENTS**

Principal Project Company shall perform its obligations under the Contract Documents in accordance with the following requirements.

#### 1. LABOR CODE REQUIREMENTS

#### 1.1 Worker's Compensation

Principal Project Company shall comply with the provisions of Section 3700 of the California Labor Code which require every employer to be insured against liability for workers' compensation or to undertake self-insurance in accordance with the provisions of that Code, and to secure the payment of compensation to his or her employees. Before commencing the Work, Principal Project Company and Contractors will sign and file a certification with City under Labor Code Section 1861 stating the following:

I am aware of the provisions of Section 3700 of the Labor Code which require every employer to be insured against liability for worker's compensation or to undertake self-insurance in accordance with the provisions of that code, and I will comply with such provisions before commencing the performance of any work or services under the Agreement, any contract or subcontract.

## 1.2 Prevailing Wages

Pursuant to the provisions of Section 1773 of the State Labor Code. City has obtained the general prevailing rate of wages (which rate includes employer payments for health and welfare, pension, vacation, travel time and subsistence pay as provided for in Section 1773.8 of said Code, apprenticeship or other training programs authorized by Section 3093 of said Code, and similar purposes) applicable to the Work to be done, for straight time, overtime, Saturday, Sunday, and holiday work. The holiday wage rate listed shall be applicable to all holidays recognized in the collective bargaining agreement of the particular craft, classification or type of worker concerned. Said prevailing wage rates are incorporated herein by reference. These prevailing rates of wages will be furnished to Principal Project Company and other interested parties on request and are on file at City's offices. These wage rates are also available through the California State Department of Industrial Relations at http://www.dir.ca.gov. For crafts or classifications not shown on the prevailing wage determinations, Principal Project Company may be required to pay the wage rate of the most closely related craft or classification shown in such determinations for the Work. Principal Project Company shall post a copy of the prevailing wage rates at the jobsite or material staging area. Workers employed in the Work must be paid at the rates at least equal to the prevailing wage rates as adopted. This Agreement is also subject to federal requirements for payment of prevailing wages as determined by the Secretary of Labor. Where there are differences in the rates, the higher shall apply.

## **1.3** Hours of Work/Overtime Requirements

Eight hours labor constitutes a legal day's work. Neither Principal Project Company nor any Contractor shall require or permit any worker to work in excess of eight hours in any one

calendar day or in excess of 40 hours in any one calendar week (defined as seven sequential calendar days) unless such worker receives compensation at a rate not less than one and one-half times the basic rate of pay for all hours worked in excess of eight hours in any one calendar day or in excess of 40 hours in any one calendar week, whichever is greater. Failure to comply with the preceding requirements shall subject Principal Project Company to the penalties specified in Labor Code Section 1813.

## 1.4 Payroll Records

- (a) Principal Project Company and each Contractor performing any portion of the Work under this Agreement shall keep an accurate payroll record as required by Law (California Labor Code Section 1776), including showing the name, address, social security number, work classification, straight time and overtime hours for each day and week, and the actual per diem wages paid to each journeyman, apprentice, worker, professional, salaried, or other employee employed by him or her in connection with the Work.
- (b) The payroll records of Principal Project Company and each Contractor (including payroll records for professional or salaried employees) shall be certified and shall be available for inspection at the principal office of Principal Project Company.
- (c) Principal Project Company shall file a certified copy of the payroll records (including those applicable to professional and salaried employees) with City within 10 days after receipt of a written request from City.
- (d) Principal Project Company shall inform City of the location of said payroll records, including the street address, city and county, and shall, within five days, provide a Notice of change of location and address of said payroll records.
- (e) It shall be the responsibility of Principal Project Company to ensure compliance for itself and the Contractors with the provisions of this section.
- (f) In the event of noncompliance with the requirements of this section, Principal Project Company shall have 10 days in which to comply subsequent to receipt of written Notice specifying in what respect it must comply. Should noncompliance exist after the said 10-day period, Principal Project Company shall be subject to a fee of \$50.00 for each day, or portion thereof, for each worker to whom the noncompliance pertains, until strict compliance is effectuated. Principal Project Company acknowledges that, without limitation as to other remedies of enforcement available to City, upon the request of the Division of Apprenticeship Standards or the Division of Labor Standards Enforcement of the California Department of Industrial Relations, such penalties shall be withheld from payments due Principal Project Company.
- (g) Certified payroll records shall be submitted to City by Principal Project Company and all Contractors performing work on the Project regardless of dollar amount or type of contract.
- (h) The period covered shall be from the time Work starts until all Work is completed on the Project. Failure to submit said certified payrolls on time may result in the

withholding of payments to Principal Project Company and the assessment of penalties as set forth in the California Labor Code.

#### 1.5 Specific Labor Code Provisions

Principal Project Company's attention is directed to the following requirements of the Labor Code. Principal Project Company shall cause Contractors to insert in any Contracts a copy of each such Code section and shall also cause Subcontractors to include these clauses in any lower tier Subcontracts. Principal Project Company shall be responsible for the compliance by any Contractor or Subcontractor with the clauses set forth in this <u>Section 1.5</u> (Specific Labor Code Provisions), as may be amended and updated from time to time.

#### Labor Code Section 1725.5

1725.5. A contractor shall be registered pursuant to this section to be qualified to bid on, be listed in a bid proposal, subject to the requirements of Section 4104 of the Public Contract Code, or engage in the performance of any public work contract that is subject to the requirements of this chapter. For the purposes of this section, "contractor" includes a subcontractor as defined by Section 1722.1.

(a) To qualify for registration under this section, a contractor shall do all of the following:

(1) (A) Register with the Department of Industrial Relations in the manner prescribed by the department and pay an initial nonrefundable application fee of four hundred dollars (\$400) to qualify for registration under this section and an annual renewal fee on or before July 1 of each year thereafter. The annual renewal fee shall be in a uniform amount set by the Director of Industrial Relations, and the initial registration and renewal fees may be adjusted no more than annually by the director to support the costs specified in Section 1771.3.

(B) Beginning June 1, 2019, a contractor may register or renew according to this subdivision in annual increments up to three years from the date of registration. Contractors who wish to do so will be required to prepay the applicable nonrefundable application or renewal fees to qualify for the number of years for which they wish to preregister.

(2) Provide evidence, disclosures, or releases as are necessary to establish all of the following:

(A) Workers' compensation coverage that meets the requirements of Division 4 (commencing with Section 3200) and includes sufficient coverage for any worker whom the contractor employs to perform work that is subject to prevailing wage requirements other than a contractor who is separately registered under this section. Coverage may be evidenced by a current and valid certificate of workers' compensation insurance or certification of self-insurance required under Section 7125 of the Business and Professions Code.

(B) If applicable, the contractor is licensed in accordance with Chapter 9 (commencing with Section 7000) of the Business and Professions Code.

(C) The contractor does not have any delinquent liability to an employee or the state for any assessment of back wages or related damages, interest, fines, or penalties pursuant to any final judgment, order, or determination by a court or any federal, state, or local administrative agency, including a confirmed arbitration award. However, for purposes of this paragraph, the contractor shall not be disqualified for any judgment, order, or determination that is under appeal, provided that the contractor has secured the payment of any amount eventually found due through a bond or other appropriate means.

(D) The contractor is not currently debarred under Section 1777.1 or under any other federal or state law providing for the debarment of contractors from public works.

(E) The contractor has not bid on a public works contract, been listed in a bid proposal, or engaged in the performance of a contract for public works without being lawfully registered in accordance with this section, within the preceding 12 months or since the effective date of the requirements set forth in subdivision (e), whichever is earlier. If a contractor is found to be in violation of the requirements of this paragraph, the period of disqualification shall be waived if both of the following are true:

(i) The contractor has not previously been found to be in violation of the requirements of this paragraph within the preceding 12 months.

(ii) The contractor pays an additional nonrefundable penalty registration fee of two thousand dollars (\$2,000).

(b) Fees received pursuant to this section shall be deposited in the State Public Works Enforcement Fund established by Section 1771.3 and shall be used only for the purposes specified in that section.

(c) A contractor who fails to pay the renewal fee required under paragraph (1) of subdivision (a) on or before the expiration of any prior period of registration shall be prohibited from bidding on or engaging in the performance of any contract for public work until once again registered pursuant to this section. If the failure to pay the renewal fee was inadvertent, the contractor may renew its registration retroactively by paying an additional nonrefundable penalty renewal fee equal to the amount of the renewal fee within 90 days of the due date of the renewal fee.

(d) If, after a body awarding a contract accepts the contractor's bid or awards the contract, the work covered by the bid or contract is determined to be a public work to which Section 1771 applies, either as the result of a determination by the director pursuant to Section 1773.5 or a court decision, the requirements of this section shall not apply, subject to the following requirements:

(1) The body that awarded the contract failed, in the bid specification or in the contract documents, to identify as a public work that portion of the work that the determination or decision subsequently classifies as a public work.

(2) Within 20 days following service of notice on the awarding body of a determination by the Director of Industrial Relations pursuant to Section 1773.5 or a decision by a court that the contract was for public work as defined in this chapter, the contractor and any

subcontractors are registered under this section or are replaced by a contractor or subcontractors who are registered under this section.

(3) The requirements of this section shall apply prospectively only to any subsequent bid, bid proposal, contract, or work performed after the awarding body is served with of the determination or decision referred to in paragraph (2).

(e) The requirements of this section shall apply to any bid proposal submitted on or after March 1, 2015, to any contract for public work, as defined in this chapter, executed on or after April 1, 2015, and to any work performed under a contract for public work on or after January 1, 2018, regardless of when the contract for public work was executed.

(f) This section does not apply to work performed on a public works project of twenty-five thousand dollars (\$25,000) or less when the project is for construction, alteration, demolition, installation, or repair work or to work performed on a public works project of fifteen thousand dollars (\$15,000) or less when the project is for maintenance work.

#### Labor Code Section 1735

1735. A contractor shall not discriminate in the employment of persons upon public works on any basis listed in subdivision (a) of Section 12940 of the Government Code, as those bases are defined in Sections 12926 and 12926.1 of the Government Code, except as otherwise provided in Section 12940 of the Government Code. Every contractor for public works who violates this section is subject to all the penalties imposed for a violation of this chapter.

#### Labor Code Section 1771

1771. Except for public works projects of one thousand dollars (\$1,000) or less, not less than the general prevailing rate of per diem wages for work of a similar character in the locality in which the public work is performed, and not less than the general prevailing rate of per diem wages for holiday and overtime work fixed as provided in this chapter, shall be paid to all workers employed on public works.

This section is applicable only to work performed under contract, and is not applicable to work carried out by a public agency with its own forces. This section is applicable to contracts let for maintenance work.

## Labor Code Section 1771.1

1771.1 (a) A contractor or subcontractor shall not be qualified to bid on, be listed in a bid proposal, subject to the requirements of Section 4104 of the Public Contract Code, or engage in the performance of any contract for public work, as defined in this chapter, unless currently registered and qualified to perform public work pursuant to Section 1725.5. It is not a violation of this section for an unregistered contractor to submit a bid that is authorized by Section 7029.1 of the Business and Professions Code or by Section

10164 or 20103.5 of the Public Contract Code, provided the contractor is registered to perform public work pursuant to Section 1725.5 at the time the contract is awarded.

(b) Notice of the requirement described in subdivision (a) shall be included in all bid invitations and public works contracts, and a bid shall not be accepted nor any contract or subcontract entered into without proof of the contractor or subcontractor's current registration to perform public work pursuant to Section 1725.5.

(c) An inadvertent error in listing a subcontractor who is not registered pursuant to Section 1725.5 in a bid proposal shall not be grounds for filing a bid protest or grounds for considering the bid nonresponsive, provided that any of the following apply:

(1) The subcontractor is registered prior to the bid opening.

(2) Within 24 hours after the bid opening, the subcontractor is registered and has paid the penalty registration fee specified in subparagraph (E) of paragraph (2) of subdivision (a) of Section 1725.5.

(3) The subcontractor is replaced by another registered subcontractor pursuant to Section 4107 of the Public Contract Code.

(d) Failure by a subcontractor to be registered to perform public work as required by subdivision (a) shall be grounds under Section 4107 of the Public Contract Code for the contractor, with the consent of the awarding authority, to substitute a subcontractor who is registered to perform public work pursuant to Section 1725.5 in place of the unregistered subcontractor.

(e) The department shall maintain on its Internet Web site a list of contractors who are currently registered to perform public work pursuant to Section 1725.5.

(f) A contract entered into with any contractor or subcontractor in violation of subdivision (a) shall be subject to cancellation, provided that a contract for public work shall not be unlawful, void, or voidable solely due to the failure of the awarding body, contractor, or any subcontractor to comply with the requirements of Section 1725.5 or this section.

(g) If the Labor Commissioner or his or her designee determines that a contractor or subcontractor engaged in the performance of any public work contract without having been registered in accordance with this section, the contractor or subcontractor shall forfeit, as a civil penalty to the state, one hundred dollars (\$100) for each day of work performed in violation of the registration requirement, not to exceed an aggregate penalty of eight thousand dollars (\$8,000) in addition to any penalty registration fee assessed pursuant to clause (ii) of subparagraph (E) of paragraph (2) of subdivision (a) of Section 1725.5.

(h) (1) In addition to, or in lieu of, any other penalty or sanction authorized pursuant to this chapter, a higher tiered public works contractor or subcontractor who is found to have entered into a subcontract with an unregistered lower tier subcontractor to perform any public work in violation of the requirements of Section 1725.5 or this section shall be subject to forfeiture, as a civil penalty to the state, of one hundred dollars (\$100) for each

day the unregistered lower tier subcontractor performs work in violation of the registration requirement, not to exceed an aggregate penalty of ten thousand dollars (\$10,000).

(2) The Labor Commissioner shall use the same standards specified in subparagraph (A) of paragraph (2) of subdivision (a) of Section 1775 when determining the severity of the violation and what penalty to assess, and may waive the penalty for a first time violation that was unintentional and did not hinder the Labor Commissioner's ability to monitor and enforce compliance with the requirements of this chapter.

(3) A higher tiered public works contractor or subcontractor shall not be liable for penalties assessed pursuant to paragraph (1) if the lower tier subcontractor's performance is in violation of the requirements of Section 1725.5 due to the revocation of a previously approved registration.

(4) A subcontractor shall not be liable for any penalties assessed against a higher tiered public works contractor or subcontractor pursuant to paragraph (1). A higher tiered public works contractor or subcontractor may not require a lower tiered subcontractor to indemnify or otherwise be liable for any penalties pursuant to paragraph (1).

(i) The Labor Commissioner or his or her designee shall issue a civil wage and penalty assessment, in accordance with the provisions of Section 1741, upon determination of penalties pursuant to subdivision (g) and subparagraph (B) of paragraph (1) of subdivision (h). Review of a civil wage and penalty assessment issued under this subdivision may be requested in accordance with the provisions of Section 1742. The regulations of the Director of Industrial Relations, which govern proceedings for review of civil wage and penalty assessments under Article 1 (commencing with Section 1720) and Article 2 (commencing with Section 1770), shall apply.

(j) (1) Where a contractor or subcontractor engages in the performance of any public work contract without having been registered in violation of the requirements of Section 1725.5 or this section, the Labor Commissioner shall issue and serve a stop order prohibiting the use of the unregistered contractor or the unregistered subcontractor on all public works until the unregistered contractor or unregistered subcontractor is registered. The stop order shall not apply to work by registered contractors or subcontractors on the public work.

(2) A stop order may be personally served upon the contractor or subcontractor by either of the following methods:

(A) Manual delivery of the order to the contractor or subcontractor personally.

(B) Leaving signed copies of the order with the person who is apparently in charge at the site of the public work and by thereafter mailing copies of the order by first class mail, postage prepaid to the contractor or subcontractor at one of the following:

(i) The address of the contractor or subcontractor on file with either the Secretary of State or the Contractors' State License Board.

(ii) If the contractor or subcontractor has no address on file with the Secretary of State or the Contractors' State License Board, the address of the site of the public work.

(3) The stop order shall be effective immediately upon service and shall be subject to appeal by the party contracting with the unregistered contractor or subcontractor, by the unregistered contractor or subcontractor, or both. The appeal, hearing, and any further review of the hearing decision shall be governed by the procedures, time limits, and other requirements specified in subdivision (a) of Section 238.1.

(4) Any employee of an unregistered contractor or subcontractor who is affected by a work stoppage ordered by the commissioner pursuant to this subdivision shall be paid at his or her regular hourly prevailing wage rate by that employer for any hours the employee would have worked but for the work stoppage, not to exceed 10 days.

(k) Failure of a contractor or subcontractor, owner, director, officer, or managing agent of the contractor or subcontractor to observe a stop order issued and served upon him or her pursuant to subdivision (j) is guilty of a misdemeanor punishable by imprisonment in county jail not exceeding 60 days or by a fine not exceeding ten thousand dollars (\$10,000), or both.

(I) This section shall apply to any bid proposal submitted on or after March 1, 2015, and any contract for public work entered into on or after April 1, 2015. This section shall also apply to the performance of any public work, as defined in this chapter, on or after January 1, 2018, regardless of when the contract for public work was entered.

(m) Penalties received pursuant to this section shall be deposited in the State Public Works Enforcement Fund established by Section 1771.3 and shall be used only for the purposes specified in that section.

(n) This section shall not apply to work performed on a public works project of twenty-five thousand dollars (\$25,000) or less when the project is for construction, alteration, demolition, installation, or repair work or to work performed on a public works project of fifteen thousand dollars (\$15,000) or less when the project is for maintenance work.

#### Labor Code Section 1771.4

1771.4 (a) All of the following are applicable to all public works projects that are otherwise subject to the requirements of this chapter:

(1) The call for bids and contract documents shall specify that the project is subject to compliance monitoring and enforcement by the Department of Industrial Relations.

(2) The awarding body shall post or require the prime contractor to post job site notices, as prescribed by regulation.

(3) Each contractor and subcontractor shall furnish the records specified in Section 1776 directly to the Labor Commissioner, in the following manner:

(A) At least monthly or more frequently if specified in the contract with the awarding body.

(B) In a format prescribed by the Labor Commissioner.

(4) If the contractor or subcontractor is not registered pursuant to Section 1725.5 and is performing work on a project for which registration is not required because of subdivision (f) of Section 1725.5, the unregistered contractor or subcontractor is not required to furnish the records specified in Section 1776 directly to the Labor Commissioner but shall retain the records specified in Section 1776 for at least three years after completion of the work.

(5) The department shall undertake those activities it deems necessary to monitor and enforce compliance with prevailing wage requirements.

(b) The Labor Commissioner may exempt a public works project from compliance with all or part of the requirements of subdivision (a) if either of the following occurs:

(1) The awarding body has enforced an approved labor compliance program, as defined in Section 1771.5, on all public works projects under its authority, except those deemed exempt pursuant to subdivision (a) of Section 1771.5, continuously since December 31, 2011.

(2) The awarding body has entered into a collective bargaining agreement that binds all contractors performing work on the project and that includes a mechanism for resolving disputes about the payment of wages.

(c) The requirements of paragraph (1) of subdivision (a) shall only apply to contracts for public works projects awarded on or after January 1, 2015.

(d) The requirements of paragraph (3) of subdivision (a) shall apply to all contracts for public work, whether new or ongoing, on or after January 1, 2016.

#### Labor Code Section 1775

1775. (a) (1) The contractor and any subcontractor under the contractor shall, as a penalty to the state or political subdivision on whose behalf the contract is made or awarded, forfeit not more than two hundred dollars (\$200) for each calendar day, or portion thereof, for each worker paid less than the prevailing wage rates as determined by the director for the work or craft in which the worker is employed for any public work done under the contract by the contractor or, except as provided in subdivision (b), by any subcontractor under the contractor.

(2) (A) The amount of the penalty shall be determined by the Labor Commissioner based on consideration of both of the following:

(i) Whether the failure of the contractor or subcontractor to pay the correct rate of per diem wages was a good faith mistake and, if so, the error was promptly and voluntarily corrected when brought to the attention of the contractor or subcontractor.

(ii) Whether the contractor or subcontractor has a prior record of failing to meet its prevailing wage obligations.

(B) (i) The penalty may not be less than forty dollars (\$40) for each calendar day, or portion thereof, for each worker paid less than the prevailing wage rate, unless the failure of the contractor or subcontractor to pay the correct rate of per diem wages was a good faith mistake and, if so, the error was promptly and voluntarily corrected when brought to the attention of the contractor or subcontractor.

(ii) The penalty may not be less than eighty dollars (\$80) for each calendar day, or portion thereof, for each worker paid less than the prevailing wage rate, if the contractor or subcontractor has been assessed penalties within the previous three years for failing to meet its prevailing wage obligations on a separate contract, unless those penalties were subsequently withdrawn or overturned.

(iii) The penalty may not be less than one hundred twenty dollars (\$120) for each calendar day, or portion thereof, for each worker paid less than the prevailing wage rate, if the Labor Commissioner determines that the violation was willful, as defined in subdivision (c) of Section 1777.1.

(C) If the amount due under this section is collected from the contractor or subcontractor, any outstanding wage claim under Chapter 1 (commencing with Section 1720) of Part 7 of Division 2 against that contractor or subcontractor shall be satisfied before applying that amount to the penalty imposed on that contractor or subcontractor pursuant to this section.

(D) The determination of the Labor Commissioner as to the amount of the penalty shall be reviewable only for abuse of discretion.

(E) The difference between the prevailing wage rates and the amount paid to each worker for each calendar day or portion thereof for which each worker was paid less than the prevailing wage rate shall be paid to each worker by the contractor or subcontractor, and the body awarding the contract shall cause to be inserted in the contract a stipulation that this section will be complied with.

(b) If a worker employed by a subcontractor on a public works project is not paid the general prevailing rate of per diem wages by the subcontractor, the prime contractor of the project is not liable for any penalties under subdivision (a) unless the prime contractor had knowledge of that failure of the subcontractor to pay the specified prevailing rate of wages to those workers or unless the prime contractor fails to comply with all of the following requirements:

(1) The contract executed between the contractor and the subcontractor for the performance of work on the public works project shall include a copy of the provisions of Sections 1771, 1776, 1777.5, 1813, and 1815.

(2) The contractor shall monitor the payment of the specified general prevailing rate of per diem wages by the subcontractor to the employees, by periodic review of the certified payroll records of the subcontractor.

(3) Upon becoming aware of the failure of the subcontractor to pay his or her workers the specified prevailing rate of wages, the contractor shall diligently take corrective action to halt or rectify the failure, including, but not limited to, retaining sufficient funds due the subcontractor for work performed on the public works project.

(4) Prior to making final payment to the subcontractor for work performed on the public works project, the contractor shall obtain an affidavit signed under penalty of perjury from the subcontractor that the subcontractor has paid the specified general prevailing rate of per diem wages to his or her employees on the public works project and any amounts due pursuant to Section 1813.

(c) The Division of Labor Standards Enforcement shall notify the contractor on a public works project within 15 days of the receipt by the Division of Labor Standards Enforcement of a complaint of the failure of a subcontractor on that public works project to pay workers the general prevailing rate of per diem wages.

#### Labor Code Section 1777.5

1777.5. (a) This chapter does not prevent the employment of properly registered apprentices upon public works.

(b) (1) Every apprentice employed upon public works shall be paid the prevailing rate of per diem wages for apprentices in the trade to which he or she is registered and shall be employed only at the work of the craft or trade to which he or she is registered.

(2) Unless otherwise provided by a collective bargaining agreement, when a contractor requests the dispatch of an apprentice pursuant to this section to perform work on a public works project and requires the apprentice to fill out an application or undergo testing, training, an examination, or other preemployment process as a condition of employment, the apprentice shall be paid for the time spent on the required preemployment activity, including travel time to and from the required activity, if any, at the prevailing rate of per diem wages for apprentices in the trade to which he or she is registered. Unless otherwise provided by a collective bargaining agreement, a contractor is not required to compensate an apprentice for the time spent on preemployment activities if the apprentice is required to take a preemployment drug or alcohol test and he or she fails to pass that test.

(c) Only apprentices, as defined in Section 3077, who are in training under apprenticeship standards that have been approved by the Chief of the Division of Apprenticeship Standards and who are parties to written apprentice agreements under Chapter 4 (commencing with Section 3070) of Division 3 are eligible to be employed at the apprentice wage rate on public works. The employment and training of each apprentice shall be in accordance with either of the following:

(1) The apprenticeship standards and apprentice agreements under which he or she is training.

(2) The rules and regulations of the California Apprenticeship Council.

(d) If the contractor to whom the contract is awarded by the state or any political subdivision, in performing any of the work under the contract, employs workers in any apprenticeable craft or trade, the contractor shall employ apprentices in at least the ratio set forth in this section and may apply to any apprenticeship program in the craft or trade that can provide apprentices to the site of the public work for a certificate approving the contractor under the apprenticeship standards for the employment and training of apprentices in the area or industry affected. However, the decision of the apprenticeship program to approve or deny a certificate shall be subject to review by the Administrator of Apprenticeship. The apprenticeship program or programs, upon approving the contractor, shall arrange for the dispatch of apprentices to the contractor. A contractor covered by an apprenticeship program's standards shall not be required to submit any additional application in order to include additional public works contracts under that program. "Apprenticeable craft or trade," as used in this section, means a craft or trade determined as an apprenticeable occupation in accordance with rules and regulations prescribed by the California Apprenticeship Council. As used in this section, "contractor" includes any subcontractor under a contractor who performs any public works not excluded by subdivision (o).

(e) Before commencing work on a contract for public works, every contractor shall submit contract award information to an applicable apprenticeship program that can supply apprentices to the site of the public work. The information submitted shall include an estimate of journeyman hours to be performed under the contract, the number of apprentices proposed to be employed, and the approximate dates the apprentices would be employed. A copy of this information shall also be submitted to the awarding body, if requested by the awarding body. Within 60 days after concluding work on the contract, each contractor and subcontractor shall submit to the awarding body, if requested, and to the apprenticeship program a verified statement of the journeyman and apprentice hours performed on the contract. The information under this subdivision shall be public. The apprenticeship programs shall retain this information for 12 months.

(f) The apprenticeship program supplying apprentices to the area of the site of the public work shall ensure equal employment and affirmative action in apprenticeship for women and minorities.

(g) The ratio of work performed by apprentices to journeymen employed in a particular craft or trade on the public work may be no higher than the ratio stipulated in the apprenticeship standards under which the apprenticeship program operates if the contractor agrees to be bound by those standards. However, except as otherwise provided in this section, in no case shall the ratio be less than one hour of apprentice work for every five hours of journeyman work.

(h) This ratio of apprentice work to journeyman work shall apply during any day or portion of a day when any journeyman is employed at the jobsite and shall be computed on the basis of the hours worked during the day by journeymen so employed. Any work performed by a journeyman in excess of eight hours per day or 40 hours per week shall not be used to calculate the ratio. The contractor shall employ apprentices for the number of hours computed as above before the end of the contract or, in the case of a subcontractor, before the end of the subcontract. However, the contractor shall endeavor, to the greatest extent possible, to employ apprentices during the same time period that the journeymen in the same craft or trade are employed at the jobsite. When an hourly apprenticeship ratio is not feasible for a particular craft or trade, the Administrator of Apprenticeship, upon application of an apprenticeship program, may order a minimum ratio of not less than one apprentice for each five journeymen in a craft or trade classification.

(i) A contractor covered by this section who has agreed to be covered by an apprenticeship program's standards upon the issuance of the approval certificate, or who has been previously approved for an apprenticeship program in the craft or trade, shall employ the number of apprentices or the ratio of apprentices to journeymen stipulated in the applicable apprenticeship standards, but in no event less than the 1-to-5 ratio required by subdivision (g).

(j) Upon proper showing by a contractor that he or she employs apprentices in a particular craft or trade in the state on all of his or her contracts on an annual average of not less than one hour of apprentice work for every five hours of labor performed by journeymen, the Administrator of Apprenticeship may grant a certificate exempting the contractor from the 1-to-5 hourly ratio, as set forth in this section for that craft or trade.

(k) An apprenticeship program has the discretion to grant to a participating contractor or contractor association a certificate, which shall be subject to the approval of the Administrator of Apprenticeship, exempting the contractor from the 1-to-5 ratio set forth in this section when it finds that any one of the following conditions is met:

(1) Unemployment for the previous three-month period in the area exceeds an average of 15 percent.

(2) The number of apprentices in training in the area exceeds a ratio of 1 to 5.

(3) There is a showing that the apprenticeable craft or trade is replacing at least onethirtieth of its journeymen annually through apprenticeship training, either on a statewide basis or on a local basis.

(4) Assignment of an apprentice to any work performed under a public works contract would create a condition that would jeopardize his or her life or the life, safety, or property of fellow employees or the public at large, or the specific task to which the apprentice is to be assigned is of a nature that training cannot be provided by a journeyman.

(I) If an exemption is granted pursuant to subdivision (k) to an organization that represents contractors in a specific trade from the 1-to-5 ratio on a local or statewide basis, the member contractors shall not be required to submit individual applications for approval to local joint apprenticeship committees, if they are already covered by the local apprenticeship standards.

(m) (1) A contractor to whom a contract is awarded, who, in performing any of the work under the contract, employs journeymen or apprentices in any apprenticeable craft or trade shall contribute to the California Apprenticeship Council the same amount that the director determines is the prevailing amount of apprenticeship training contributions in the area of the public works site. A contractor may take as a credit for payments to the council any amounts paid by the contractor to an approved apprenticeship program that can supply apprentices to the site of the public works project. The contractor may add the amount of the contributions in computing his or her bid for the contract.

(2) At the conclusion of the 2002–03 fiscal year and each fiscal year thereafter, the California Apprenticeship Council shall distribute training contributions received by the council under this subdivision, less the expenses of the Department of Industrial Relations for administering this subdivision, by making grants to approved apprenticeship programs for the purpose of training apprentices. The funds shall be distributed as follows:

(A) If there is an approved multiemployer apprenticeship program serving the same craft or trade and geographic area for which the training contributions were made to the council, a grant to that program shall be made.

(B) If there are two or more approved multiemployer apprenticeship programs serving the same craft or trade and county for which the training contributions were made to the council, the grant shall be divided among those programs based on the number of apprentices from that county registered in each program.

(C) All training contributions not distributed under subparagraphs (A) and (B) shall be used to defray the future expenses of the Department of Industrial Relations for the administration and enforcement of apprenticeship standards and requirements under this code.

(3) All training contributions received pursuant to this subdivision shall be deposited in the Apprenticeship Training Contribution Fund, which is hereby created in the State Treasury. Upon appropriation by the Legislature, all moneys in the Apprenticeship Training Contribution Fund shall be used for the purpose of carrying out this subdivision and to pay the expenses of the Department of Industrial Relations.

(n) The body awarding the contract shall cause to be inserted in the contract stipulations to effectuate this section. The stipulations shall fix the responsibility of compliance with this section for all apprenticeable occupations with the prime contractor.

(o) This section does not apply to contracts of general contractors or to contracts of specialty contractors not bidding for work through a general or prime contractor when the contracts of general contractors or those specialty contractors involve less than thirty thousand dollars (\$30,000).

(p) An awarding body that implements an approved labor compliance program in accordance with subdivision (b) of Section 1771.5 may, with the approval of the director, assist in the enforcement of this section under the terms and conditions prescribed by the director.

#### Labor Code Section 1813

1813. The contractor or subcontractor shall, as a penalty to the state or political subdivision on whose behalf the contract is made or awarded, forfeit twenty-five dollars (\$25) for each worker employed in the execution of the contract by the respective contractor or subcontractor for each calendar day during which the worker is required or permitted to work more than 8 hours in any one calendar day and 40 hours in any one

calendar week in violation of the provisions of this article. In awarding any contract for public work, the awarding body shall cause to be inserted in the contract a stipulation to this effect. The awarding body shall take cognizance of all violations of this article committed in the course of the execution of the contract, and shall report them to the Division of Labor Standards Enforcement.

#### Labor Code Section 1815

1815. Notwithstanding the provisions of Sections 1810 to 1814, inclusive, of this code, and notwithstanding any stipulation inserted in any contract pursuant to the requirements of said sections, work performed by employees of contractors in excess of 8 hours per day, and 40 hours during any one week, shall be permitted upon public work upon compensation for all hours worked in excess of 8 hours per day at not less than 11/2 times the basic rate of pay.

#### 1.6 Excavation Safety

Principal Project Company shall comply with Labor Code Section 6705 while excavating. For an excavation five feet or more in depth, submit shop drawings for a protective system.

The drawings must show the design and details for providing worker protection from caving ground during excavation.

Shop drawings of protective systems for which the Construction Safety Orders issued by Cal/OSHA require design by a registered professional engineer must be sealed and signed by an engineer who is registered as a civil engineer in the State.

## 2. PUBLIC CONTRACT CODE REQUIREMENTS

## 2.1 Ineligible Contractors

Principal Project Company shall not enter into or permit entering into any Contract with a Contractor who is ineligible to perform work on the Project pursuant to Section 1777.1 or 1777.7 of the Labor Code.

#### 2.2 Assignment of Causes of Action

Principal Project Company's attention is directed to the following requirements in Public Contract Code Section 7103.5:

(b) In entering into a public works contract or a subcontract to supply goods, services, or materials pursuant to a public works contract, the contractor or subcontractor offers and agrees to assign to the awarding body all rights, title, and interest in and to all causes of action it may have under Section 4 of the Clayton Act (15 U.S.C. Sec. 15) or under the Cartwright Act (Chapter 2 (commencing with Section 16700) of Part 2 of Division 7 of the Business and Professions Code), arising from purchases of goods, services, or materials pursuant to the public works contract or the subcontract. This assignment shall be made and become effective at

the time the awarding body tenders final payment to the contractor, without further acknowledgment by the parties.

#### 2.3 [RESERVED]

#### 2.4 Specifications by Brand or Trade Names

Principal Project Company's attention is directed to the following requirements in Public Contract Code Section 3400:

(b) No agency of the state, nor any political subdivision, municipal corporation, or district, nor any public officer or person charged with the letting of contracts for the construction, alteration, or repair of public works, shall draft or cause to be drafted specifications for bids, in connection with the construction, alteration, or repair of public works, (1) in a manner that limits the bidding, directly or indirectly, to any one specific concern, or (2) calling for a designated material, product, thing, or service by specific brand or trade name unless the specification is followed by the words "or equal" so that bidders may furnish any equal material, product, thing, or service. In applying this section, the specifying agency shall, if aware of an equal product manufactured in this state, name that product in the specification. Specifications shall provide a period of time prior to or after, or prior to and after, the award of the contract for submission of data substantiating a request for a substitution of "an equal" item. If no time period is specified, data may be submitted any time within 35 days after the award of the contract.

(c) Subdivision (b) is not applicable if the awarding authority, or its designee, makes a finding that is described in the invitation for bids or request for proposals that a particular material, product, thing, or service is designated by specific brand or trade name for any of the following purposes:

(1) In order that a field test or experiment may be made to determine the product's suitability for future use.

(2) In order to match other products in use on a particular public improvement either completed or in the course of completion.

(3) In order to obtain a necessary item that is only available from one source.

(4)(A) In order to respond to an emergency declared by a local agency, but only if the declaration is approved by a four-fifths vote of the governing board of the local agency issuing the invitation for bid or request for proposals.

(B) In order to respond to an emergency declared by the state, a state agency, or political subdivision of the state, but only if the facts setting forth the reasons for the finding of the emergency are contained in the public records of the authority issuing the invitation for bid or request for proposals.

## 3. GOVERNMENT CODE REQUIREMENTS

#### 3.1 Removal, Relocation or Protection of Existing Utilities

Principal Project Company acknowledges and agrees that the provisions of <u>Article 14</u> (Compensation and Other Relief for Delay Events and Relief Events) of this Agreement satisfy City's obligations pursuant to Government Code Section 4215. Principal Project Company agrees that to the extent that Government Code Section 4215 may be construed to the contrary, Principal Project Company hereby waives the benefit of such statute.

#### 3.2 Nondiscrimination and Compliance Employment Programs

Principal Project Company shall comply with, and shall require Contractors to comply with, the provisions of:

- (b) the Fair Employment and Housing Act (Government Code section 12900 et seq.), and the applicable regulations promulgated thereunder (California Code of Regulations, Title 2, section 7285 et seq.); and
- (c) the Fair Employment and Housing Commission regulations implementing Government Code section 12990 (a-f) set forth in Chapter 5 of Division 4 of Title 2 of the California Code of Regulations,

which are incorporated into, and made a part of this Agreement as if set forth in full. Principal Project Company shall require each Contractor to include the compliance requirements under this <u>Section 3.2</u> (Nondiscrimination and Compliance Employment Programs) in its Contract, and give written Notice of such Contractor's obligations under this <u>Section 3.2</u> (Nondiscrimination and Compliance Employment Programs) to labor organizations with which it has a collective bargaining or other agreement, as appropriate.

## 4. BUSINESS AND PROFESSIONS CODE

Principal Project Company's attention is directed to the following requirements in Business and Professions Code Sections 7030(a) and 7108.5(a) through (d):

**7030.** (a) Except for contractors writing home improvement contracts pursuant to Section 7151.2 and contractors writing service and repair contracts pursuant to Section 7159.10, every person licensed pursuant to this chapter shall include the following statement in at least 10-point type on all written contracts with respect to which the person is a prime contractor:

"Contractors are required by law to be licensed and regulated by the Contractors' State License Board which has jurisdiction to investigate complaints against contractors if a complaint regarding a patent act or omission is filed within four years of the date of the alleged violation. A complaint regarding a latent act or omission pertaining to structural defects must be filed within 10 years of the date of the alleged violation. Any questions concerning a contractor may be referred to the Registrar, Contractors' State License Board, P.O. Box 26000, Sacramento, CA 95826."

**7108.5** (a) A prime contractor or subcontractor shall pay to any subcontractor, not later than seven days after receipt of each progress payment, unless otherwise agreed to in writing, the

respective amounts allowed the contractor on account of the work performed by the subcontractors, to the extent of each subcontractor's interest therein. In the event that there is a good faith dispute over all or any portion of the amount due on a progress payment from the prime contractor or subcontractor to a subcontractor, the prime contractor or subcontractor may withhold no more than 150 percent of the disputed amount.

(b) Any violation of this section shall constitute a cause for disciplinary action and shall subject the licensee to a penalty, payable to the subcontractor, of 2 percent of the amount due per month for every month that payment is not made.

(c) In any action for the collection of funds wrongfully withheld, the prevailing party shall be entitled to his or her attorney's fees and costs.

(d) The sanctions authorized under this section shall be separate from, and in addition to, all other remedies, either civil, administrative, or criminal.

# EXHIBIT 16C

## CITY REQUIREMENTS

# [NOTE TO PNC: Local Hire, First Source, and SFMTA Training Program requirements to be added]

#### 1. CITY REQUIREMENTS

Principal Project Company has reviewed, understands, and is ready, willing, and able to comply with the terms and conditions of this <u>Exhibit 16C</u> (City Requirements), which summarizes special City requirements as of the Effective Date, each of which is fully incorporated by reference. Principal Project Company acknowledges that City requirements in effect when any Contract Documents are executed will be incorporated into the Contract Documents, as applicable, and will apply to all Contractors, Subcontractors, and any other PPC-Related Entities, as applicable. City requirements of general applicability will apply to the Project even if not summarized below.

The following summary is for Principal Project Company's convenience only; Principal Project Company is obligated to become familiar with all applicable requirements and to comply with them fully as they are amended from time to time. City ordinances are currently available on the web at www.sfgov.org and at www.amlegal.com/codes/client/san-francisco\_ca. References to specific laws in this <u>Exhibit 16C</u> (City Requirements) refer to the San Francisco Municipal Code unless specified otherwise. Capitalized terms used in this <u>Exhibit 16C</u> (City Requirements) and not defined in this Agreement will have the meanings assigned to them in the applicable Section of the San Francisco Municipal Code.

#### 1.1. Nondiscrimination in City Contracts and Benefits Ordinance.

(a) <u>Non-Discrimination in Contracts</u>. Principal Project Company shall comply with the provisions of Chapters 12B and 12C of the Administrative Code, which are incorporated into this Agreement by this reference. Principal Project Company shall incorporate by reference in all Contractor Documents the provisions of Sections 12B.2(a), 12B.2(c)-(k), and 12C.3 of the Administrative Code and shall require all PPC-Related Entities to comply with such provisions. Principal Project Company is subject to the enforcement and penalty provisions in Chapters 12B and 12C.

(b) <u>Non-Discrimination in the Provision of Employee Benefits</u>. Principal Project Company does not as of Effective Date, and will not during the Term, in any of its operations in San Francisco, on real property owned by San Francisco, or where work is being performed for City elsewhere in the United States, discriminate in the provision of employee benefits between employees with domestic partners and employees with spouses and/or between the domestic partners and spouses of such employees, subject to the conditions set forth in San Francisco Administrative Code Section 12B.2.

**1.2.** Requiring Health Benefits for Covered Employees. All undefined, initiallycapitalized terms used in this <u>Section 1.2</u> (Requiring Health Benefits for Covered Employees) shall have the meanings given to them in Administrative Code Chapter 12Q (the "HCAO"). If the HCAO applies to this Agreement, Principal Project Company shall comply with the requirements of the HCAO. For each Covered Employee, Principal Project Company shall provide the appropriate health benefit set forth in Section 12Q.3 of the HCAO. If Principal Project Company chooses to offer the health plan option, such health plan shall meet the minimum standards set forth by the San Francisco Health Commission. Information about and the text of the HCAO, as well as the Health Commission's minimum standards, is available on the web at http://sfgov.org/olse/hcao. Principal Project Company is subject to the enforcement and penalty provisions in the HCAO. Any Contract entered into by Principal Project Company shall require any PPC-Related Entity with 20 or more employees to comply with the requirements of the HCAO and shall contain contractual obligations substantially the same as those set forth in this <u>Section 1.2</u> (Requiring Health Benefits for Covered Employees).

**1.3. Minimum Compensation Ordinance**. If San Francisco Administrative Code Chapter 12P applies to this Agreement, Principal Project Company shall pay covered employees no less than the minimum compensation required by San Francisco Administrative Code Chapter 12P ("**Chapter 12P**"), including a minimum hourly gross compensation, compensated time off, and uncompensated time off. Principal Project Company is subject to the enforcement and penalty provisions in Chapter 12P. Information about and the text of the Chapter 12P is available on the web at http://sfgov.org/olse/mco. Principal Project Company is required to comply with all of the applicable provisions of Chapter 12P, irrespective of the listing of obligations in this <u>Section 1.3</u> (Minimum Compensation Ordinance). By signing and executing this Agreement, Principal Project Company certifies that it complies with Chapter 12P.

## **1.4.** Prevailing Rate of Wages and Working Conditions.

(a) <u>Covered Services</u>. Principal Project Company agrees it will pay, and require the PPC-Related Entities to pay, the Prevailing Rate of Wages for any Work performed by Principal Project Company or the PPC-Related Entities ("**Covered Services**"), including any trade work for the Project performed by or for Principal Project Company during the Term. The provisions of Section 6.22(e) and 21C of the San Francisco Administrative Code are incorporated as provisions of this Agreement as if fully set forth herein and will apply to any Covered Services performed by Principal Project Company and the PPC-Related Entities.

(b) Determining the Prevailing Rate of Wages. The latest Prevailing Rate of Wages for private employment on public contracts as determined by the San Francisco Board of Supervisors and the Director of the California Department of Industrial Relations, as such prevailing wage rates may be changed during the Term, are hereby incorporated as provisions of this Agreement. Copies of the Prevailing Rate of Wages as fixed and determined by the Board of Supervisors are available from the Office of Labor Standards and Enforcement ("OLSE") and on the Internet at http://www.dir.ca.gov/DLSR/PWD and http://sfgov.org/olse/prevailing-wage. Principal Project Company agrees that it and the PPC-Related Entities will pay no less than the Prevailing Rate of Wages, as fixed and determined by the Board of Supervisors, to all workers who perform Covered Services and are employed by Principal Project Company or the PPC-Related Entities.

(c) <u>Subcontract Requirements</u>. As required by Section 6.22(e)(5) of the San Francisco Administrative Code, Principal Project Company shall require each Contractor to insert in every Subcontract or other arrangement, which it may make for the performance of Covered Services under this Agreement, a provision that said Subcontractor shall pay to all Persons performing labor in connection with Covered Services under said Subcontract or other arrangement not less than the highest general the Prevailing Rate of Wages as fixed and determined by the Board of Supervisors for such labor or services.

(d) <u>Posted Notices</u>. As required by Section 1771.4 of the California Labor Code, Principal Project Company shall post job site notices prescribed by the California Department of Industrial Relations ("**DIR**") at all job sites where Covered Services are to be performed.

(e) <u>Payroll Records</u>. As required by Section 6.22(e)(6) of the San Francisco Administrative Code and Section 1776 of the California Labor Code, Principal Project Company shall keep or cause to be kept complete and accurate payroll records for all trade workers performing Covered Services. Such records shall include the name, address and social security number of each worker who provided Covered Services on the project, including apprentices, his or her classification, a general description of the services each worker performed each day, the rate of pay (including rates of contributions for, or costs assumed to provide fringe benefits), daily and weekly number of hours worked, deductions made and actual wages paid. Every Contractor and Subcontractor who shall undertake the performance of any part of Covered Services shall keep a like record of each person engaged in the execution of Covered Services under the Contract or Subcontract. All such records shall at all times be available for inspection of and examination by City and its authorized representatives and the DIR.

(f) <u>Certified Payrolls</u>. Certified payrolls shall be prepared pursuant to San Francisco Administrative Code Section 6.22(e)(6) and California Labor Code Section 1776 for the period involved for all employees, including those of Subcontractors, who performed labor in connection with Covered Services. Principal Project Company and each Subcontractor performing Covered Services shall submit certified payrolls to City and to the DIR electronically. Principal Project Company shall submit payrolls to City via the reporting system selected by City. The DIR will specify how to submit certified payrolls to it. City will provide basic training in the use of the reporting system at a scheduled training session. Principal Project Company, all Contractors and all Subcontractors that will perform Covered Services must attend the training session. Principal Project Company, Contractors, and Subcontractors shall comply with electronic certified payroll requirements (including training) at no additional cost to City.

Compliance Monitoring. Covered Services to be performed under this (g) Agreement are subject to compliance monitoring and enforcement of prevailing wage requirements by the DIR and/or the OLSE. Principal Project Company, Contractors and Subcontractors performing Covered Services will cooperate fully with the DIR and/or the OLSE and other City employees and agents authorized to assist in the administration and enforcement of the prevailing wage requirements, and agrees to take the specific steps and actions as required by Section 6.22(e)(7) of the San Francisco Administrative Code. Steps and actions include but are not limited to requirements that: (i) Principal Project Company will cooperate fully with the Labor Standards Enforcement Officer and other City employees and agents authorized to assist in the administration and enforcement of the Prevailing Wage requirements and other labor standards imposed on Principal Project Company by the Charter and Chapter 6 of the San Francisco Administrative Code; (ii) Principal Project Company agrees that the Labor Standards Enforcement Officer and his or her designees, in the performance of their duties, shall have the right to engage in random inspections of job sites and to have access to the employees of Principal Project Company, employee time sheets, inspection logs, payroll records and employee paychecks: (iii) Contractors and Subcontractors shall maintain a sign-in and sign-out sheet showing which employees are present on the job site; (iv) Principal Project Company shall prominently post at each job-site a sign informing employees that the project is subject to the City's Prevailing Wage requirements and that these requirements are enforced by the Labor Standards Enforcement Officer; and (v) that the Labor Standards Enforcement Officer may audit such records of Principal Project Company as he or she reasonably deems necessary to determine compliance with the Prevailing Wage and other labor standards imposed by the

Charter and this Chapter and applicable to this Agreement. Failure to comply with these requirements may result in penalties and forfeitures consistent with analogous provisions of the California Labor Code, including Section 1776(g), as amended from time to time.

(h) <u>Remedies</u>. Should Principal Project Company, Contractors or Subcontractors who shall undertake the performance of any Covered Services, fail or neglect to pay to the persons who perform Covered Services under this Agreement, Contracts or Subcontracts or other arrangement for the Covered Services, the general prevailing rate of wages as herein specified, Principal Project Company shall forfeit, and in the case of any Subcontractor so failing or neglecting to pay said wage, Principal Project Company and the Subcontractor shall jointly and severally forfeit, back wages due plus the penalties set forth in San Francisco Administrative Code Section 6.22 (e) and/or California Labor Code Section 1775. City, when certifying any payment, which may become due under the terms of this Agreement, shall deduct from the amount that would otherwise be due on such payment the amount of said forfeiture.

**1.5. Prohibition on Use of Public Funds for Political Activity.** In performing the Work, Principal Project Company shall comply with San Francisco Administrative Code Chapter 12G, which prohibits funds appropriated by City for this Agreement from being expended to participate in, support, or attempt to influence any political campaign for a candidate or for a ballot measure. Principal Project Company is subject to the enforcement and penalty provisions in Chapter 12G.

**1.6. Consideration of Salary History.** Principal Project Company shall comply with San Francisco Administrative Code Chapter 12K, the Consideration of Salary History Ordinance or "Pay Parity Act." Principal Project Company is prohibited from considering current or past salary of an applicant in determining whether to hire the applicant or what salary to offer the applicant to the extent that such applicant is applying for employment to be performed on this Agreement or in furtherance of this Agreement, and whose application, in whole or part, will be solicited, received, processed or considered, whether or not through an interview, in City or on City property. The ordinance also prohibits employers from (1) asking such applicants about their current or past salary or (2) disclosing a current or former employee's salary history without that employee's authorization unless the salary history is publicly available. Principal Project Company is subject to the enforcement and penalty provisions in Chapter 12K. Information about and the text of Chapter 12K is available on the web at

https://sfgov.org/olse/consideration-salary-history. Principal Project Company is required to comply with all of the applicable provisions of 12K, irrespective of the listing of obligations in this <u>Section 1.6</u> (Consideration of Salary History).

## **1.7.** Consideration of Criminal History in Hiring and Employment Decisions.

(a) Principal Project Company agrees to comply fully with and be bound by all of the provisions of Chapter 12T, "City Contractor/Subcontractor Consideration of Criminal History in Hiring and Employment Decisions," of the San Francisco Administrative Code ("Chapter 12T"), including the remedies provided, and implementing regulations, as may be amended from time to time. The provisions of Chapter 12T are incorporated by reference and made a part of this Agreement as though fully set forth herein. The text of the Chapter 12T is available on the web at http://sfgov.org/olse/fco. Principal Project Company is required to comply with all of the applicable provisions of 12T, irrespective of the listing of obligations in this Section 1.7 (Consideration of Criminal History in Hiring and Employment Decisions). Capitalized terms used in this Section 1.7 (Consideration of Criminal History in Hiring and Employment

Decisions.) and not defined in this Agreement shall have the meanings assigned to such terms in Chapter 12T.

(b) The requirements of Chapter 12T shall only apply to a Principal Project Company's or its Agent's operations to the extent those operations are in furtherance of the performance of this Agreement, shall apply only to applicants and employees who would be or are performing work in furtherance of this Agreement, and shall apply when the physical location of the employment or prospective employment of an individual is wholly or substantially within the City of San Francisco. Chapter 12T shall not apply when the application in a particular context would conflict with federal or state law or with a requirement of a government agency implementing federal or state law.

**1.8. Resource Efficiency Requirements**. The Project will be subject to Chapter 7 of the San Francisco Environment Code. Accordingly, the Project must meet certain resource efficient requirements. Principal Project Company agrees that it will design the Project to comply with Chapter 7 of the San Francisco Environment Code, as may be amended from time to time, or any similar law.

**1.9. MacBride Principles Northern Ireland**. City urges companies doing business in Northern Ireland to move towards resolving employment inequities, and encourages such companies to abide by the MacBride Principles. City urges San Francisco companies to do business with corporations that abide by the MacBride Principles.

**1.10.** Notification of Limitations on Contributions. Principal Project Company acknowledges its obligations under Section 1.126 of the San Francisco Campaign and Governmental Conduct Code, which prohibits any person who contracts with, or is seeking a contract with, any department of City for the rendition of personal services, for the furnishing of any material, supplies or equipment, for the sale or lease of any land or building, for a grant, loan or loan guarantee, or for a development agreement, from making any campaign contribution to (i) a City elected official if the contract must be approved by that official, a board on which that official serves, or the board of a state agency on which an appointee of that official serves, (ii) a candidate for that City elective office, or (iii) a committee controlled by such elected official or a candidate for that office, at any time from the submission of a proposal for the contract until the later of either the termination of negotiations for such contract or twelve months after the date City approves the contract. The prohibition on contributions applies to (a) each prospective party to the contract, (b) each member of the contractor's board of directors, the contractor's chairperson, chief executive officer, chief financial officer and chief operating officer, (c) any person with an ownership interest of more than ten percent (10%) in the contractor, (d) any Subcontractor listed in the bid or contract, and (e) any committee that is sponsored or controlled by the contractor. Principal Project Company certifies that it has informed each such person of the limitation on contributions imposed by Section 1.126 by the time it submitted the Implementation Proposal, and has provided the names of the persons required to be informed to City.

**1.11. Sunshine Ordinance**. Principal Project Company acknowledges that the Contract Documents and all records related to its formation, Principal Project Company's performance of Work, and City's payment are subject to the California Public Records Act, (California Government Code § 6250 et. seq.), and the San Francisco Sunshine Ordinance, (San Francisco Administrative Code Chapter 67). Such records are subject to public inspection and copying unless exempt from disclosure under federal, state or local law.

**1.12. Conflicts of Interest**. Principal Project Company acknowledges that it is familiar with the provisions of San Francisco Charter, Article III, Chapter 2, Section 15.103 of the City's Campaign and Governmental Conduct Code, and California Government Code Sections 87100 *et seq.* and Sections 1090 *et seq.*, certifies that it does not know of any facts that would constitute a violation of these provisions, and agrees that if Principal Project Company becomes aware of any such fact during the Term, Principal Project Company will provide Notice to City immediately.

**1.13. Certification of Funds**. This Agreement is subject to the fiscal provisions of the City's Charter and the budget decisions of its Mayor and Board of Supervisors, each acting in its sole discretion. No funds will be available hereunder until prior written authorization certified by the City's Controller. The City's Controller cannot authorize payments unless funds have been certified as available in the budget or in a supplemental appropriation. City's obligations hereunder shall never exceed the amount certified by the City's Controller for the purpose and period stated in such certification. City, its employees and officers are not authorized to offer or promise any additional funding without City's Controller certification of such additional funding. Without such lawful approval and certification, City shall not be required to provide such additional funding.

**1.14.** Art Commission Design Review; Art Enrichment Allocation. The Facility will be subject to the requirements of San Francisco Charter Section 5.103 and Administrative Code Section 3.19. Principal Project Company must work with the San Francisco Arts Commission, in consultation with the City, to design and build the Facility in compliance with those requirements.

**1.15.** Tropical Hardwoods and Virgin Redwood Ban. City urges companies not to import, purchase, obtain, or use for any purpose, any tropical hardwood, tropical hardwood wood product, virgin redwood, or virgin redwood wood product, except as expressly permitted by the application of San Francisco Environment Code sections 802(b) and 803(b). Principal Project Company will not, except as permitted by the application of sections 802(b) and 803(b), use or incorporate any tropical hardwood, tropical hardwood wood product, virgin redwood, or virgin redwood wood product in the performance of this Agreement.

**1.16. City Business and Tax Regulations Code**. Principal Project Company acknowledges that under Section 6.10-2 of the San Francisco Business and Tax Regulations Code, the City Treasurer and Tax Collector may require the withholding of payments to any vendor that is delinquent in the payment of any amounts that the vendor is required to pay the City under the San Francisco Business and Tax Regulations Code. If, under that authority, any payment City is required to make to Principal Project Company under this Agreement is withheld, then City will not be in breach or default under this Agreement, and the Treasurer and Tax Collector will authorize release of any payments withheld under this paragraph to Principal Project Company, without interest, late fees, penalties, or other charges, upon Principal Project Company coming back into compliance with its San Francisco Business and Tax Regulations Code obligations.

**1.17. Contracting Requirements**. The following City requirements are incorporated into this Agreement and apply to the Principal Project Company, Contractors and Subcontractors, together with any other applicable City contract requirements that are in effect on the Effective Date.

(a) <u>Tobacco and Alcohol Products Advertising Ban</u>. Principal Project Company acknowledges and agrees that no advertising of cigarettes, tobacco products, or

alcoholic beverages is allowed on any real property owned by or under the control of City. This prohibition includes the placement of the name of a company producing, selling or distributing cigarettes, tobacco products, or alcoholic beverages or the name of any cigarette, tobacco product, or alcoholic beverages in any promotion of any event or product. This prohibition does not apply to any advertisement sponsored by a state, local, nonprofit, or other entity designed to: (a) communicate the health hazards of cigarettes and tobacco products or alcoholic beverages; (b) encourage people not to smoke or to stop smoking, or not to drink alcohol or to stop drinking alcohol; or (c) provide or publicize drug or alcohol treatment or rehabilitation services.

(b) <u>Alcohol and Drug-Free Workplace</u>. City reserves the right to deny access to, or require Principal Project Company to remove any PPC-Related Entities from City facilities if City has reasonable grounds to believe that person has engaged in alcohol abuse or illegal drug activity which in any way impairs City's ability to maintain safe work facilities or to protect the health and well-being of City employees and the general public. City shall have the right of final approval for the entry or re-entry of any that person previously denied access to, or removed from, City facilities. Illegal drug activity means possessing, furnishing, selling, offering, purchasing, using or being under the influence of illegal drugs or other controlled substances for which the person lacks a valid prescription. Alcohol abuse means possessing, furnishing, selling, selling, offering, or using alcoholic beverages, or being under the influence of alcohol.

(c) <u>Drug-Free Workplace</u>. Principal Project Company acknowledges that pursuant to the Federal Drug-Free Workplace Act of 1988 (41 U.S.C. §§ 8101 et seq.), the unlawful manufacture, distribution, possession or use of a controlled substance is prohibited on City property.

(d) <u>Food Service Waste Reduction Ordinance</u>. Principal Project Company agrees to comply fully with and be bound by the Food Service Waste Reduction Ordinance (San Francisco Environment Code Chapter 16), including but not limited to the remedies for noncompliance provided therein.

(e) <u>Sugar Sweetened Beverages and Packaged Water</u>. Principal Project Company agrees that it will not sell, provide, or otherwise distribute Sugar-Sweetened Beverages, as defined by San Francisco Administrative Code Chapter 101, as part of its performance of this Agreement. Principal Project Company agrees that it shall not sell, provide, or otherwise distribute Packaged Water, as defined by San Francisco Environment Code Chapter 24, as part of its performance of this Agreement.

**1.18. Preservative Treated Wood Products.** Principal Project Company shall comply with the provisions of San Francisco Environment Code Chapter 13, which requires that anyone purchasing preservative-treated wood products on behalf of the City, shall only purchase such products from the list of alternatives adopted by the Department of the Environment pursuant to Section 1302 of Chapter 13, unless otherwise granted an exemption by the terms of that Chapter.

**1.19. SFMTA Surveillance Technology Policy**. Principal Project Company must comply with the SFMTA's Surveillance Technology Policy, adopted under San Francisco Administrative Code Chapter 19B. A copy of the SFMTA's Surveillance Technology Policy is attached as <u>Exhibit 16D</u> (SFMTA's Surveillance Technology Policy).

**1.20.** Clean Construction. Principal Project Company agrees to comply fully with and be bound by the Clean Construction requirements set forth in Section 6.25 of the San

Francisco Administrative Code and Chapter 25 of the Environment Code. The provisions of Section 6.25 and Chapter 25 are incorporated into this Agreement by reference. Principal Project Company may seek waivers from the Clean Construction requirements as set forth in Chapter 25 of the Environment Code. By entering into this Agreement, Principal Project Company and City agree that if Principal Project Company uses off-road equipment and/or off-road engines in violation of the Clean Construction requirements set forth in Section 6.25 of the Administrative Code and Chapter 25 of the Environment Code, the City will suffer actual damages that will be impractical or extremely difficult to determine. Accordingly, Principal Project Company and the City agree that Principal Project Company shall pay the City the amount of \$100 per day per each piece of off-road equipment and each off-road engine used to complete work on the Project in violation of the Clean Construction requirements. Such amount shall not be considered a penalty, but rather agreed monetary damages sustained by City because of Principal Project Company's failure to comply with the Clean Construction requirements.

**1.21.** Compliance with Americans with Disabilities Act. Principal Project Company shall provide the Work in a manner that complies with the Americans with Disabilities Act (ADA), including but not limited to Title II's program access requirements, and all other applicable federal, state and local disability rights legislation.

**1.22. Government Code Claim Requirement.** No suit for money or damages may be brought against the City until a written claim therefor has been presented to and rejected by the City in conformity with the provisions of San Francisco Administrative Code Chapter 10 and California Government Code Section 900, et seq. Nothing set forth in this Agreement shall operate to toll, waive or excuse Contractor's compliance with the California Government Code Claim requirements set forth in San Francisco Administrative Code Chapter 10 and California Government Code Section 900, et seq.

**1.23. Restrictions on Use of Pesticides**. Chapter 3 of the San Francisco Environment Code (the Integrated Pest Management Program Ordinance or "IPM Ordinance") describes an integrated pest management ("IPM") policy to be implemented by all City departments. PCC-Related Entities shall not use or apply or allow the use or application of any pesticides on the Project Site or contract with any party to provide pest abatement or control services to the Project Site without first receiving City's written approval of an IPM plan. PCC-Related Entities will comply, and will require all of PCC-Related Entities' contractors to comply, with any IPM plan approved by the City and will comply with the requirements of sections 300(d), 302, 304, 305(f), 305(g), and 306 of the IPM Ordinance, as if PCC-Related Entities were a City department.

**1.24. All-Gender Toilet Facilities**. If applicable, PCC-Related Entities will comply with San Francisco Administrative Code Section 4.1-3 requiring at least one all-gender toilet facility in any new building constructed on City-owned land or that is constructed by or for the City where toilet facilities are required or provided. An "all-gender toilet facility" means a toilet that is not restricted to use by persons of a specific sex or gender identity by means of signage, design, or the installation of fixtures.

# EXHIBIT 16D

#### SFMTA'S SURVEILLANCE TECHNOLOGY POLICY

See the SFMTA policy at the attached link:

https://sf.gov/sites/default/files/2022-03/MTA\_Security%20Camera%20Policy%20Final%20Draft%202-25-21.pdf

## SECTION 3.6 INVOICE

Reference is made to that certain Design-Build-Finance-Operate-Maintain Agreement dated as of [\_\_\_\_\_], 2025 (the "**Agreement**"), by and between the City and County of San Francisco ("**City**"), a municipal corporation acting by and through the San Francisco Municipal Transportation Agency ("**SFMTA**"), and [\_\_\_\_\_] ("**Principal Project Company**").

Capitalized terms used but not defined herein have the meaning given to them in this Agreement.

Pursuant to <u>Section 3.6</u> of this Agreement, Principal Project Company hereby requests payment of [\_\_\_\_\_] U.S. dollars (\$[\_\_\_\_]), based on the following: (a) this Agreement has been terminated under <u>Section 3.6.1</u> or <u>Section 3.6.2</u> of this Agreement; (b) this invoice is submitted within 30 days of the termination; and (c) Principal Project Company has submitted to City all work product produced by Principal Project Company related to the Project, as required in <u>Section 3.6.5</u> of this Agreement.

Principal Project Company represents and warrants to City that: (a) Principal Project Company is eligible for payment pursuant to <u>Section 3.6</u> of this Agreement; and (b) Principal Project Company has attached documentation reasonably required by City sufficient to support such statement.

Principal Project Company acknowledges that: (a) such payment is the exclusive compensation payable by City to Principal Project Company under this Agreement for a termination under <u>Section 3.6.1</u> or <u>Section 3.6.2</u> of this Agreement; and (b) submission of this invoice, and payment by City of any amount in response to this invoice, is in all respects subject to the terms and conditions of this Agreement.

## **TECHNICAL REQUIREMENTS**

[SEE ATTACHED]

#### INITIAL SCHEDULE

[See attached]

[Note: Schedule to be attached before Commercial Close. Schedule shall be a critical path schedule, in Primavera 6 software, setting out all the relevant D&C Work activities from Commercial Close through Final Acceptance, with key interim milestones during that period of work, and at a minimum at Level 3 defined in AACE 91R-16 "Schedule Development"]

# LIST OF KEY PERSONNEL

- 1. Project Director
- 2. Project Manager
- 3. Deputy Project Manager
- 4. Equity Member's Project Principal
- 5. Engineer(s) of Record
- 6. Architect(s) of Record
- 7. Design Manager
- 8. Construction Manager
- 9. Quality Program Manager
- 10. Third Party and Utility Coordination Manager
- 11. Project Safety Representative
- 12. IFM Manager
- 13. IFM Quality Manager